UNIVERSITÀ DEGLI STUDI DI MILANO

Facoltà di Scienze e Tecnologie Corso di Dottorato in Matematica Ciclo XXVIII



Tesi di Dottorato di Ricerca

A FUNCTIONAL ANALYTIC FRAMEWORK FOR LOCAL ZETA REGULARIZATION AND THE SCALAR CASIMIR EFFECT

MAT/07

Advisor: Prof. Livio Pizzocchero Coordinator of PhD Studies: Prof. Bert van Geemen

> Elaborato Finale di: Davide Fermi Matricola: R09931

Anno Accademico 2014/2015

A FUNCTIONAL ANALYTIC FRAMEWORK FOR LOCAL ZETA REGULARIZATION AND THE SCALAR CASIMIR EFFECT

Davide Fermi

A thesis submitted for the degree of Doctor of Philosophy in Mathematics

Università degli Studi di Milano Department of Mathematics

January 2016

Advisor: Prof. L. Pizzocchero Coordinator of PhD studies: Prof. Bert van Geemen

Abstract

It is developed a functional analytic framework allowing to formulate a rigorous implementation of zeta regularization for a canonically quantized scalar field, living on an arbitrary spatial domain and interacting with a classical background potential. This framework relies on the construction of an infinite scale of graded Hilbert spaces associated to the real powers of some given, positive self-adjoint operator. When the latter is a Schrödinger-type differential operator, this formulation provides a natural language to study the integral kernels related to a large class of operators, fulfilling minimal regularity requirements; particular attention is devoted to the regularity of these kernels and to the construction of their analytic continuations with respect to some parameters. Within this framework, complex powers of the elliptic operator giving rise to the Klein-Gordon equation are used to define a zeta-regularized version of the Wightman field whose pointwise evaluation is well-posed. This regularized field determines regularized local observables (such as the stress-energy tensor), whose vacuum expectation values can be expressed in terms of the above mentioned integral kernels. This allows to make contact with the theory of the Casimir effect. Renormalization is achieved by analytic continuation, which is proved to give finite results for the previously mentioned expectation values in most cases of interest. Finally, to exhibit the computational efficiency of the above methods, some explicit examples are discussed.

To my family...

Acknowledgments

I wish to express my sincere appreciation and gratitude to those who have contributed to this thesis and supported me in one way or the other during this journey. Without the help of each one of them, this thesis would not have been possible.

First and foremost I want to thank my advisor, Professor Livio Pizzocchero. It has been an exciting and inspiring experience to be one of his first PhD students. Throughout the past six years, he provided sound advice, excellent teaching and good company. I appreciate all his contributions of time and ideas to make my academic adventure so far productive and stimulating.

A special thanks to my family. They have supported me in all my pursuits. I would not have made it this far without them. Words cannot express how grateful I am to my hard-working parents for all of the sacrifices that they have made on the behalf of me. I am indebted to my sister for patentiently tolerating me and enduring throughout the many years of our "room sharing". Last, but not least, I am grateful to my beloved wife Erika for being such a true and great supporter. She has unconditionally loved me during my good and bad times. I truly thank Erika for sticking by my side, especially during the final stages of this PhD, allowing me to spend most of the time on this thesis.

Ringraziamenti

Desidero esprimere il mio sincero riconoscimento e la mia gratitudine a tutti coloro che hanno contribuito a questa tesi e che mi hanno supportato in un modo o nell'altro durante questo percorso. Senza l'aiuto di ciascuno di loro, questa tesi non sarebbe stata possibile.

Innanzitutto vorrei ringraziare il mio relatore, il Professor Livio Pizzocchero. Essere uno dei suoi primi studenti di dottorato è stata una esperienza stimolante ed entusiasmante. Durante gli ultimi sei anni, Livio è stato fonte di buoni consigli, un ottimo insegnante, nonchè un buon amico. Apprezzo molto tutti i suoi contributi in termini di tempo e di idee, che hanno reso fino ad ora la mia avventura accademica produttiva e stimolante.

Devo un ringraziamento speciale alla mia famiglia. Loro mi hanno supportato nel raggiungimento di tutti i miei obbiettivi. Non sarei arrivato dove sono ora senza di loro. Non ci sono parole adatte ad esprimere la mia gratitudine nei confronti dei miei genitori per tutti i sacrifici che hanno fatto per me. Sono in debito con mia sorella per avermi pazientemente tollerato e per aver tenuto duro durante i molti anni in cui abbiamo "condiviso" la camera. Infine, ma non per importanza, sono grato alla mia amata moglie Erika per essere stata un punto di riferimento solido e sincero. Mi ha voluto bene incondizionatamente nei miei alti e bassi. Ringrazio sinceramente Erika per essere rimasta al mio fianco, soprattutto nell'ultimo periodo di questo PhD, permettendomi di dedicare la maggior parte del tempo a questa tesi.

Contents

Abstract						
A	Acknowledgments					
In	Introduction					
1	Mo	tivations and basic ideas	7			
	1.1	The reference classical theory	8			
	1.2	Second quantization.	9			
	1.3	Zeta regularization	11			
2	Fun	actional spaces, operators and kernels	13			
	2.1	Function spaces	13			
	2.2	Some known results on Banach spaces.	18			
	2.3	Operators on Hilbert spaces	21			
	2.4	Conjugations and Hilbert spaces	25			
	2.5	Scale of Hilbert spaces associated to a positive self-adjoint operator	27			
		Finite order spaces.	27			
		Infinite order spaces.	30			
		Duality relations.	35			
		Banach adjoints of continuous operators.	38			
		The case of a Hilbert space with conjugation	41			
		Further results on the complex powers of \mathcal{A}	42			
	2.6	Schrödinger-type operators: regularity and self-adjointness.	49			
		Regularity results for elliptic differential operators.	49			
		The general notion of admissible operator. Some examples	55			
		Embedding results for the scale of Hilbert spaces.	58			
	0.7	Dirac delta functions	60 67			
	2.7	Integral kernels.	67			
		General results on integral kernels.	69			
		The Dirichlet kernel. The heat, cylinder and modified cylinder kernels	74			
		Heat and cylinder kernels as solutions of differential problems	78			
		Asymptotic behaviours of the heat and cylinder kernels	80			

		Integral representations of the Dirichlet kernel in terms of the exponential	
		kernels.	82
	<u> </u>	I he case of a compact domain with smooth boundary.	82 02
	2.0	Concred definitions and some basic relations	92
		General definitions and some basic relations	92
		Analytic continuation of Menin transforms	90 100
		Analytic continuation of the Dirichlet kernels	$\frac{102}{107}$
3	Qua	antum field theory on spatial domains with boundaries	111
	3.1	Canonical quantization of an abstract Hermitian scalar field.	112
		The bosonic Fock space.	112
		Creation and annihilation operators.	115
		Segal quantization.	119
		The Wightman field at time zero.	121
		Time evolution for the Wightman field.	125
	3.2	Scalar quantum field on a spatial domain.	132
		Canonical quantization (revisited).	133
		The zeta-regularized Wightman field.	135
		The zeta-regularized stress-energy tensor and its VEV	140
		The zeta-regularized vacuum energy and pressure	145
		Renormalized observables.	149
		Anomalies	151
4	Some explicitly solvable models 15		
	4.1	The case with a background harmonic potential	156
	4.2	The case of a rectangular box	167
	4.3	The case of parallel planes with Robin boundary conditions	183
$\mathbf{A}_{]}$	ppen	dix. Some results for the cylinder kernel on a segment with Robin	
	bou	ndary conditions.	195
Bi	Bibliography		

Introduction

In quantum field theory (QFT) there often arises the necessity to give meaning to illdefined expressions; these appear typically in formal manipulations of distributions that give rise to infinities or divergent quantities when explicit computations are attempted. *Zeta regularization* (ZR) allows to deal with many of these problematic expressions by re-interpreting them as the analytic continuations with respect to a regulator parameter, introduced on purpose, of well-defined integral kernels (or of the corresponding traces) associated to complex powers of the elliptic differential operator appearing in the field equations.

The standard textbook example deals with the sum of positive integers $\sum_{n=1}^{+\infty} n$. Of course, this series is divergent in the sense of Cauchy; nevertheless, it can be re-interpreted in terms of the analytic continuation of the Riemann zeta function ζ , by introducing the regulator $s \in \mathbb{C}$ and setting

$$\sum_{n=1}^{+\infty} n \,\,"="\,\left[\sum_{n=1}^{+\infty} \frac{1}{n^s}\right]_{s=-1} \,\,"="\,\left[\zeta(s)\right]_{s=-1} = \,-\frac{1}{12} \,\,. \tag{1}$$

Here the first identity is purely formal, the second one only holds if $\Re s > 1$ and the third one has to be meant as the evaluation of the analytic continuation at s = -1 (¹).

The mathematical literature on this topic dates back to the early works of Minakshisundaram and Plejel [111, 112], and to the subsequent contributions of Seeley [143], Ray and Singer [128].

The first application to QFT was considered in the pioneering work of Dowker and Critchley [51]; soon after a more systematic formulation was developed by Hawking [86] and Wald [152]. Later on, the ZR approach was championed, revisited and widely extended by many authors; among them are, in particular, Cognola, Zerbini and Elizalde [40, 41] (see also [56, 57]), Actor, Svaiter et al. [3, 4, 6, 131, 132] and Moretti et al. [32, 88, 114, 117, 118] (see also the references in the cited works). All the authors mentioned above were mainly interested in the renormalization of local observables, such as the *vacuum expectation value* (VEV) of the stress-energy tensor, having in mind the ultimate purpose of a semiclassical treatment of quantum effects in general relativity (e.g., using the stress-energy VEV as a source in Einstein's equations).

¹The same arguments allow to infer, perhaps even more surprisingly, that $\sum_{n=1}^{+\infty} n^2$ "=" 0.

On the other hand, the zeta strategy can be applied as well to global observables, such as the VEV of the total energy and of the force on the boundary; in this "global version", it has become perhaps even more popular than its local counterpart. The literature on global ZR is enormous; here we only cite the classical papers [135, 136] by Zimerman et al., [22] by Blau, Visser, and Wipf and the monographies of Elizalde et al. [32, 56, 57], of Bordag et al. [24, 25], and of Kirsten [91] (see also the papers [36, 71, 72, 85, 92, 98] by the same author).

Both in the local and in the global versions mentioned above, zeta methods provide a very natural approach to study the theory of the *Casimir effect* (CE): in the present manuscript, with this terminology we refer generically to a class of quite notable physical phenomena which are related to the vacuum state of a quantum field interacting with classical boundaries and/or potentials, possibly living on curved and/or topologically non-trivial background spacetimes. Let us stress that no back-reaction is ever taken into account. The experimental confirmation [28, 93, 113, 147, 148] of these theoretical predictions has shown that operations such as a naive implementation of Wick's normal ordering (usually interpreted as a simple redefinition of the energy) are highly non-trivial, since they remove a priori the possibility to predict any vacuum effect.

In comparison with the original theoretical derivation of these effects by Casimir [35], and with other methods such as point-splitting [21, 31, 49, 59, 110, 126] (in particular, see [84, 115] for a comparison between point-splitting and the ZR approach) and the algebraic, microlocal approach (see, e.g., [16, 44, 75, 123, 127] and citations therein), the zeta strategy is competitive and, perhaps, more elegant. Nevertheless, in most of the works cited previously ZR is implemented within the framework of Euclidean formulation of QFT, often in connection with expressions which are derived by formal manipulations of functional integrals; moreover, apart from a few exceptions [3, 4, 6, 40, 114, 131, 132] dealing with specific spatial configurations, local aspects are scarcely taken into account when boundaries are also present.

In this manuscript a different, more systematic formulation is proposed for the ZR approach, which can be applied automatically in any specific case. The basic framework under consideration is that of canonical quantization for a Hermitian scalar field, living on an arbitrary spatial subset of Minkowski spacetime; prescribed boundary conditions are also taken into account, as well as the "effective interaction" with a classical, background scalar potential. The attention is mainly focused on the VEV of the stress-energy tensor, of which both the conformal and the non-conformal parts are considered; however, the total energy and boundary forces are also taken into account.

More in detail, the manuscript consists of four chapters, described hereafter.

Chapter 1 briefly reviews the approach to ZR developed in the previous works [63, 64, 65, 66, 67] (D.F., L. Pizzocchero) where, having in mind a direct application to some specific physical models, the language employed is less formal than the one developed in the subsequent chapters of the present work. Attention is focused on the case of a Hermitian scalar field living on a subset of Minkowski spacetime; time evolution is described by the Klein-Gordon equation with the addition of a classical background potential and suitable

spatial boundary conditions are prescribed. After a brief outline of the classical reference theory, the field is described in second quantization by means of the standard expansion in terms of annihilation and creation operators associated to a complete orthonormal system of eigenfunctions for the elliptic operator defining the "spatial part" of the Klein-Gordon equation. Complex powers of this elliptic operator are used to define a regularized version of the field; in turn, the latter is used to derive a regularized version of the propagator, which is related to the VEVs of many observables (such as the stress-energy tensor). The regularized propagator appears to be connected to the integral kernels associated to complex powers of the elliptic operator mentioned above. In the end, renormalization is defined in terms of the analytic continuation of these kernels.

In *Chapter 2* a more systematic functional analytic framework is developed, allowing to address properly topics such as the integral kernels associated to the complex powers of an elliptic differential operator, their regularity and their analytic continuations (without relying heavily on eigenfunction expansion techniques). This apparatus is conceived to be used in the subsequent Chapter 3 in connection with a more rigorous formulation of the field theory discussed in Chapter 1 and of its regularization. First an abstract framework is described, where a scale of indexed Hilbert spaces is constructed starting from an assigned separable Hilbert space \mathcal{H} and from a strictly positive, (essentially) self-adjoint operator \mathcal{A} on it. Next, this abstract framework is employed in the case where \mathcal{H} is the space of square-summable functions on a given spatial domain $\Omega \subset \mathbb{R}^d$ and \mathcal{A} is a Schrödingertype differential operator; it is shown that the elements of the scale of Hilbert spaces are related to Sobolev spaces and to spaces of differentiable functions. This formalism is used to describe the theory of integral kernels related to a class of suitable operators; regularity and other miscellaneous results are derived for these kernels. Particular attention is dedicated to the Dirichlet kernel (i.e., the kernel related to the complex power \mathcal{A}^{-s} , for suitable $s \in \mathbb{C}$) and to the heat and cylinder kernels (associated, respectively, to $e^{-t\mathcal{A}}$ and $e^{-t\sqrt{A}}$, for $\Re t > 0$; it is shown that the first can be expressed in terms of Mellin transforms of the latters. This fact is employed to construct the analytic continuation of the Dirichlet kernel and of its derivatives, by means of three different methods; these rely on suitable assumptions for the heat or cylinder kernels, which are well known to be fulfilled in most cases of interest.

In Chapter 3 the approach to local ZR discussed in the first chapter is developed within a fully rigorous framework for second quantization, based on Fock space techniques. The starting point is the bosonic Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$ on the single particle Hilbert space \mathcal{H} ; next, for $h \in \mathcal{H}$, the Segal field $\hat{\Phi}_S(h)$ and the conjugate momentum $\hat{\Pi}_S(h)$ are defined in terms of the annihilation and creation operators (which, in turn, are defined in the standard manner on the dense finite particle subspace of $\mathfrak{F}^{\vee}(\mathcal{H})$). Assuming \mathcal{A} to be a strictly positive and self-adjoint operator on \mathcal{H} , the Wightman field at time zero is described as the unique \mathbb{C} -linear extension of the \mathbb{R} -linear map $h \mapsto \hat{\Phi}_S(\mathcal{A}^{-1/4}h)$ (for $\overline{h} = h$). Time evolution is implemented applying the second quantization map Γ to the strongly continuous, one parameter unitary group $(e^{-it\sqrt{\mathcal{A}}})_{t\in\mathbb{R}}$ (where $\sqrt{\mathcal{A}}$ plays the role of the single particle Hamiltonian); it is shown that, under suitable assumptions on h, the time evolved field $\hat{\varphi}_t(h)$ is differentiable and fulfills a strong form of the Klein-Gordon equation. ZR is implemented using the powers $\mathcal{A}_{\kappa}^{-u/4} := (\mathcal{A}/\kappa^2)^{-u/4}$ ($u \in \mathbb{C}$; $\kappa > 0$ a given mass parameter), to define a family of regularized Dirac deltas $\delta_{\mathbf{x}}^u := \mathcal{A}_{\kappa}^{-u/4} \delta_{\mathbf{x}}$ and setting, for $x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega$,

$$\hat{\varphi}^u(x) := \hat{\varphi}_t(\delta^u_{\mathbf{x}}) ; \qquad (2)$$

this is shown to be a regular function of x for $\Re u$ large enough and it is used to define a zeta-regularized version $\hat{T}^u_{\mu\nu}(x)$ of the stress-energy tensor operator, by analogy with the corresponding classical observable. Moreover, the VEV of $\hat{T}^u_{\mu\nu}(x)$ is proven to be an analytic function of u which can be expressed in terms of the Dirichlet kernel and of its derivatives; the analytic continuation of these kernels can be obtained using one of the methods described in Chapter 2, relating to Mellin transforms. In the end, the renormalized stress-energy VEV is defined by analytic continuation at u = 0. Related observables, such as the total energy and the pressure on the boundary, are also discussed by similar methods.

In the conclusive *Chapter* 4 the framework developed previously is applied to some specific configurations. First of all, some of the results obtained for special cases in our antecedent works [66, 67] are briefly recalled: in particular, attention is focused on the cases of a massless field either interacting with a background harmonic potential or confined within a rectangular box. Next, another case is analyzed which, to the author's knowledge, has never been addressed before using zeta techniques; this is the case of a scalar field confined between two parallel planes π_0, π_a , on which a particular type of Robin boundary conditions are prescribed. More precisely, for $\beta \in \mathbb{R}$, it is required

$$(1+\beta \,\partial_{\mathbf{n}})\,\hat{\varphi}\Big|_{\pi_0} = 0 \,, \qquad (1-\beta \,\partial_{\mathbf{n}})\,\hat{\varphi}\Big|_{\pi_a} = 0 \tag{3}$$

(where $\partial_{\mathbf{n}}$ indicates the normal derivative at points of π_0 and π_a , respectively). To treat this case, an integral representation is derived for the cylinder kernel corresponding to the reduced 1-dimensional problem of a segment with the boundary conditions descending from those in Eq. (3). This integral representation allows to derive an explicit expression for the regularized stress-energy VEV, which can be evaluated explicitly at u = 0, thus giving the renormalized VEV of the stress-energy tensor.

Basic notations.

We write \mathbb{N} for the set of nonnegative integers $n = 0, 1, 2, \dots$

Throughout this manuscript, we indicate with \cdot and | | the standard inner product on \mathbb{R}^d $(d \in \{1, 2, 3, ...\})$ and the corresponding norm, respectively. Points of \mathbb{R}^d are indicated with boldface symbols (such as $\mathbf{x}, \mathbf{y}, \mathbf{z}$, etc).

The Lebesgue measure on \mathbb{R}^d is denoted with $d\mathbf{x}$ (or $d\mathbf{y}$); we indicate with |E| the Lebesgue measure of any measurable subset $E \subset \mathbb{R}^d$.

CONTENTS

We often use *d*-dimensional multi-indeces, which are sequences $\alpha = (\alpha_1, ..., \alpha_d) \in \mathbb{N}^d$; the order of any such multi-index is $|\alpha| := \alpha_1 + ... + \alpha_d$. We use the standard notation for partial derivatives and set $\partial^{\alpha} := \partial_x {}^{\alpha_1} ... \partial_x {}^{\alpha_d}$. For functions of two (or more) sets of variables such as $\psi : \mathbb{R}^d \times \mathbb{R}^d \to \mathbb{C}$, $(\mathbf{x}, \mathbf{y}) \mapsto \psi(\mathbf{x}, \mathbf{y})$, given any pair of multi-indices $\alpha, \beta \in \mathbb{N}^d$, we use the self-evident notations $\partial_1^{\alpha} \partial_2^{\beta} \psi \equiv \partial_{\mathbf{x}}^{\alpha} \partial_{\mathbf{y}}^{\beta} \psi$ to indicate the corresponding partial derivatives.

About complex powers.

Throughout the present manuscript, the following conventions are employed. i. $\ln : (0, +\infty) \to \mathbb{R}$ is the elementary logarithm. ii. For any $s \in \mathbb{C}$, we systematically refer to the standard definition

$$x^{s} := e^{s \ln x} \qquad \text{for all } x \in (0, +\infty) . \tag{4}$$

iii. For any $s \in \mathbb{C}$ and for any z in a convenient subset \mathbb{C}^{\times} of the complex plane, we put

$$z^s := e^{s\ln|z| + is\arg z} , (5)$$

where $\arg : \mathbb{C}^{\times} \to \mathbb{R}$ is some determination of the argument; this determination depends on the domain \mathbb{C}^{\times} and must be specified in each case of interest. Unless otherwise stated, hereafter we always put

$$\mathbb{C}^{\times} := \mathbb{C} \setminus [0, +\infty) ; \tag{6}$$

arg := the unique determination of the argument with values in $(0, 2\pi)$. (6)

Chapter 1

Motivations and basic ideas

In the present chapter we outline the main heuristic ideas motivating the more rigorous analysis to be discussed in detail throughout the subsequent chapters of this work, where a language more formal and precise from a mathematical point of view is employed. To this purpose, hereafter we briefly recall some results which were first presented more extensively in our previous work [64] (see also [63]). In particular, we consider an approach to ZR which works in the framework of canonical quantization; this makes a major difference in comparison with the more common version of this regularization scheme, which relies substantially on the Euclidean formulation of QFT.

We restrict the attention to the case of a scalar field; this is assumed to live on an assigned spatial domain of arbitrary dimension, to fulfill suitable boundary conditions and, possibly, to interact with a classical background potential (including, in some cases, a mass term). In Section 1.1 we review the reference classical theory, introducing the physical observables of main interest for the subsequent analysis; namely, the stress-energy tensor, the total energy and the pressure on the boundary. In Section 1.2 we consider the canonical quantization of the classical theory described previously; the quantized scalar field is expanded in terms of creation and annihilation operators, using a complete orthonormal set of eigenfunctions of the elliptic operator appearing in the field equations. In the final Section 1.3 we describe the approach to ZR which was first proposed in [63, 64]: the basic idea is to use the complex powers of the previously mentioned elliptic operator to define a regularized version of the field at a point. The latter determines, in turn, a regularized propagator, whose diagonal evaluation is strictly connected to certain integral kernels, to be analyzed in detail in Chapter 2.

In passing, we take the chance to fix some conventions and notations to which we will refer throughout the entire manuscript.

Before proceeding let us stress that, in both the classical and quantum versions of the field theory under analysis, the above mentioned spatial domain, its boundary and the background potential are treated as purely classical objects, which are assigned once and for all and possess no dynamical evolution in time. Therefore, the interaction between these objects and the (quantum) field is described in a purely effective fashion; in particular, no back-reaction is ever taken into account in the analysis to be described in the present Chapter, as well as in the remainder of this work.

1.1 The reference classical theory.

We use natural units, so that

$$c = 1 \qquad \text{and} \qquad \hbar = 1 , \qquad (1.1)$$

and work in (d + 1)-dimensional Minkowski spacetime \mathcal{M}_{d+1} (with $d \in \{1, 2, 3, ..\}$ arbitrary); this is identified with \mathbb{R}^{d+1} using a set of inertial coordinates

$$x = (x^{\mu})_{\mu=0,1,\dots,d} \equiv (x^{0}, \mathbf{x}) \equiv (t, \mathbf{x}) : \mathcal{M}_{d+1} \to \mathbb{R}^{d+1} , \qquad (1.2)$$

in which the Minkowski metric $\eta \equiv (\eta_{\mu\nu})$ has coefficients (¹)

$$(\eta_{\mu\nu}) = \operatorname{diag}(-1, \underbrace{1, \dots, 1}_{d \text{ times}}) .$$
(1.3)

We fix a spatial domain $\Omega \subset \mathbb{R}^d$ and refer to the classical theory of a scalar field

$$\varphi : \mathbb{R} \times \Omega \to \mathbb{R} , \qquad x \mapsto \varphi(x) \equiv \varphi(t, \mathbf{x}) ;$$
 (1.4)

this is assumed to be at least twice differentiable, to fulfill prescribed conditions (e.g., of Dirichlet, Neumann or Robin type) on the boundary $\mathbb{R} \times \partial \Omega$ and to decay rapidly at time (and possibly spatial, if Ω is unbounded) infinity. Analogous settings are considered, e.g., in [24, 25, 32, 75].

The action functional describing the theory under analysis is

$$\mathcal{S}[\varphi] := \int_{\mathbb{R} \times \Omega} dt \, d\mathbf{x} \, \frac{1}{2} \left(\partial^{\mu} \varphi(t, \mathbf{x}) \partial_{\mu} \varphi(t, \mathbf{x}) + V(\mathbf{x}) \, \varphi^{2}(t, \mathbf{x}) \right) \,, \tag{1.5}$$

where $V : \Omega \to \mathbb{R}$, $\mathbf{x} \mapsto V(\mathbf{x})$ is a suitable smooth potential; here and elsewhere, unless otherwise stated, we use *Einstein's summation convention*.

The field time evolution is described by the Klein-Gordon equation with external potential

$$0 = (-\partial^{\mu}\partial_{\mu} + V)\varphi = (\partial_{tt} - \triangle + V)\varphi$$
(1.6)

 $(\Delta := \sum_{i=1}^{d} \partial_{x^{i}x^{i}}$ is the *d*-dimensional laplacian); of course, initial data for the field and for its time derivative must be provided as well in order to obtain a well-posed evolution differential problem.

The stress-energy tensor has components (for $\mu, \nu \in \{0, ..., d\}$)

$$T_{\mu\nu} := (1 - 2\xi) \,\partial_{\mu}\varphi \,\partial_{\nu}\varphi - \left(\frac{1}{2} - 2\xi\right) \eta_{\mu\nu} (\partial^{\lambda}\varphi \,\partial_{\lambda}\varphi + V \,\varphi^2) - 2\xi \,\varphi \,\partial_{\mu\nu}\varphi \;; \tag{1.7}$$

¹Equivalently, the line element corresponding to the Minkowski metric η is assumed to be

$$ds^{2} = \eta_{\mu\nu} dx^{\mu} dx^{\nu} = -(dx^{0})^{2} + (dx^{1})^{2} + \ldots + (dx^{d})^{2} .$$

1.2. SECOND QUANTIZATION.

the parameter $\xi \in \mathbb{R}$ is fixed arbitrarily. The above mentioned regularity assumptions for the field φ suffice to infer the continuity of the functions $T_{\mu\nu} : \mathbb{R} \times \Omega \to \mathbb{R}, x \mapsto T_{\mu\nu}(x) \equiv$ $T_{\mu\nu}(t, \mathbf{x})$ (let us stress that all the bilinear expressions in the field in Eq. (1.7) are to be evaluated along the diagonal; so, e.g., $(\partial_{\mu}\varphi \partial_{\nu}\varphi)(x) = \partial_{\mu}\varphi(x) \partial_{\nu}\varphi(x)$). The expression in Eq. (1.7) is usually referred to as "improved" stress-energy tensor; this is a well-known modification of the canonical stress-energy tensor (i.e., the conserved Noether current associated with the spacetime translational symmetries of the action functional (1.5)) with an additive term proportional to the real parameter ξ , that does not alter its divergence (for more details, see Appendix A of [64]). The improved stress-energy tensor was first proposed by Callan, Coleman and Jackiw [34] in order to deal with some pathologies arising in perturbation theory for the corresponding quantized version of the field theory under analysis; later on, this tensor was reinterpreted in terms of the Minkowskian limit for a scalar field coupled to gravity via the curvature scalar [21, 32, 51, 121, 123] (see also the footnote on page 4 of [63]).

Other observables can be defined in terms of the stress-energy tensor. More precisely, the *total energy* at time $t \in \mathbb{R}$ is the integral over the spatial domain Ω of the energy density $T_{00}(t, \mathbf{x})$, i.e.,

$$\mathcal{E}(t) := \int_{\Omega} d\mathbf{x} \ T_{00}(t, \mathbf{x}) \ ; \tag{1.8}$$

on the other hand, the *pressure* at a point on the boundary of the spatial domain $\mathbf{x} \in \partial \Omega$ (at times $t \in \mathbb{R}$), is the vector $\mathbf{p}(x) = (p^i(t, \mathbf{x}))$ (i = 1, ..., d), whose components are

$$p^{i}(t, \mathbf{x}) := T^{i}{}_{j}(t, \mathbf{x}) n^{j}(\mathbf{x}) , \qquad (1.9)$$

where $\mathbf{n}(\mathbf{x}) = (n^i(\mathbf{x}))$ is the unit normal to $\partial \Omega$ at \mathbf{x} .

1.2 Second quantization.

In order to fix some notations, let us now briefly address the second quantization of the classical field theory described in the previous section. Within this framework, we consider the single particle Hilbert space $\mathcal{H} = L^2(\Omega)$, with inner product $\langle | \rangle_{L^2}$, and the bosonic Fock space $\mathfrak{F}^{\vee}(\mathcal{H}) \equiv \mathfrak{F}$ on it, with inner product (|).

The classical field $\varphi:\mathbb{R}\times\Omega\to\mathbb{R}$ is replaced by the map

$$\hat{\varphi} : \mathbb{R} \times \Omega \to \mathcal{L}_{sa}(\mathfrak{F}) , \qquad x \mapsto \hat{\varphi}(x) \equiv \hat{\varphi}(t, \mathbf{x})$$
 (1.10)

(²), where we are referring to the space $\mathcal{L}(\mathfrak{F})$ of linear operators on \mathfrak{F} and to the subset $\mathcal{L}_{sa}(\mathfrak{F})$ of self-adjoint operators. Next, let us put

$$\mathcal{A} := -\bigtriangleup + V , \qquad (1.11)$$

² Of course the notation $\hat{\varphi} : \mathbb{R} \times \Omega \to \mathcal{L}_{sa}(\mathfrak{F}), (\mathbf{x},t) \mapsto \hat{\varphi}(\mathbf{x},t)$ is used here in connection with a *generalized* operator valued function; in fact, as well known, $\hat{\varphi}$ is an operator valued *distribution*. We defer to Chapter 3 a more rigorous analysis.

intending that the boundary conditions on $\partial \Omega$ (if any) are accounted for in the above definition; we assume the framework under analysis to grant that \mathcal{A} is a *strictly positive*, *selfadjoiont* operator on \mathcal{H} .

To proceed, consider a complete orthonormal set $(F_k)_{k\in\mathcal{K}}$ of (possibly generalized) eigenfunctions of \mathcal{A} (³), indexed by an unspecified set of labels \mathcal{K} , and write the corresponding eigenvalues in the form $(\omega_k^2)_{k\in\mathcal{K}}$ ($\omega_k \geq \varepsilon$ for some $\varepsilon > 0$ and for all $k \in \mathcal{K}$); thus

$$F_k : \Omega \to \mathbb{C} , \qquad \mathcal{A}F_k = \omega_k^2 F_k ,$$

$$\langle F_k | F_h \rangle_{L^2} = \delta(k, h) \qquad \text{for all } k, h \in \mathcal{K} .$$
 (1.12)

Any eigenfunction label $k \in \mathcal{K}$ can include different parameters, both discrete and continuous. Besides, we generically write $\int_{\mathcal{K}} dk$ to indicate summation over all labels (i.e., literal summation over discrete parameters and integration over continuous parameters, with respect to a suitable measure); $\delta(h, k)$ is the Dirac delta function for the label space \mathcal{K} (this reduces to the Krönecker symbol in the case of discrete parameters).

The set of eigenfunctions described above allows us to derive for the quantized field a normal modes expansion of the form

$$\hat{\varphi}(t,\mathbf{x}) = \int_{\mathcal{K}} \frac{dk}{\sqrt{2\omega_k}} \left[\hat{a}_k \, e^{-i\omega_k t} \, F_k(\mathbf{x}) + \hat{a}_k^{\dagger} \, e^{i\omega_k t} \, \overline{F}_k(\mathbf{x}) \right] \tag{1.13}$$

(with $\bar{}$ indicating complex conjugation). Here we are considering the destruction and creation operators $\hat{a}_k, \hat{a}_k^{\dagger} \in \mathcal{L}(\mathfrak{F})$ ($k \in \mathcal{K}$) associated to the set of eigenfunctions $(F_k)_{k \in \mathcal{K}}$; these fulfill the canonical commutation relations

$$[\hat{a}_h, \hat{a}_k] = [\hat{a}_h^{\dagger}, \hat{a}_k^{\dagger}] = \hat{\mathbb{O}}_{\mathfrak{F}} , \qquad [\hat{a}_h, \hat{a}_k^{\dagger}] = \delta(h, k) \,\hat{\mathbb{I}}_{\mathfrak{F}} , \qquad (1.14)$$

where $\hat{\mathbb{O}}_{\mathfrak{F}}$, $\mathbb{I}_{\mathfrak{F}} \in \mathcal{L}(\mathfrak{F})$ denote the null and the identity operator on \mathfrak{F} , respectively. Moreover, indicating with $\boldsymbol{v} \in \mathfrak{F}$ the *vacuum state* (of unit norm), there holds the annihilation condition

$$\hat{a}_k \, \boldsymbol{v} = \boldsymbol{o} \tag{1.15}$$

 $(o \in \mathfrak{F}$ is the null element of \mathfrak{F} , not to be confused with v). Next let us consider the *propagator*, i.e., the vacuum expectation value (VEV)

$$(\boldsymbol{v}|\,\hat{\varphi}(x)\,\hat{\varphi}(y)\,\boldsymbol{v}) \qquad (x,y\in\mathbb{R}\times\Omega);$$
(1.16)

this allows to make connection with the Casimir effect. In fact, one can define second quantized versions of the classical observables mentioned in the previous subsection (such as the stress-energy tensor (1.7)) and compute the corresponding VEVs; all these VEVs can be expressed in terms of the propagator (1.16) and of its derivatives, evaluated along the diagonal y = x. We defer to Chapter 3 a more detailed analysis.

³For a fully rigorous discussion of generalized eigenfunctions see, e.g., Chapter IV of [78].

1.3. ZETA REGULARIZATION.

The assumption of strict positivity for \mathcal{A} (granting that $\omega_k \geq \varepsilon$, for some $\varepsilon > 0$ and for all $k \in \mathcal{K}$) excludes the presence of any infrared divergence (⁴). On the other hand, the propagator is known to be plagued with ultraviolet divergences, which appear along the diagonal; therefore, expressions such as $(\boldsymbol{v} | \hat{\varphi}(x)^2 \boldsymbol{v}) \equiv (\boldsymbol{v} | \hat{\varphi}(x) \hat{\varphi}(x) \boldsymbol{v})$ are merely formal ways to indicate ill-defined quantities.

Our purpose in the next subsection is to redefine the field operator via a suitable regularization scheme, ultimately yielding finite values for the propagator also along the diagonal (and for other related observables).

1.3 Zeta regularization.

Let $\kappa > 0$ denote a parameter, to which we attribute the dimension of a mass (or of an inverse length, since we have fixed $\hbar = 1$). Because of this, κ will be called the *mass scale*; it is introduced for dimensional reasons and it plays the role of a normalization scale. The final, renormalized results appear to depend on κ only when singularities appear in the analytic continuations involved in the following construction. See [22, 32, 56, 88, 114] for further comments regarding this parameter and its presence or absence in the renormalized observables related to the field.

The zeta strategy, in the version proposed in [63, 64] (and employed systematically in [65, 66, 67]), relies on the powers

$$\mathcal{A}_{\kappa}^{-u/4} := (\mathcal{A}/\kappa^2)^{-u/4} , \qquad (1.17)$$

where $\mathcal{A} = -\Delta + V$ is the operator (1.11) and $u \in \mathbb{C}$; these operators are employed to introduce the *zeta-regularized field* operator

$$\hat{\varphi}^u := \mathcal{A}_{\kappa}^{-u/4} \,\hat{\varphi} \,\,, \tag{1.18}$$

depending on the complex parameter u and coinciding with the usual field operator $\hat{\varphi}$ for u = 0. We provisionally accept the following heuristic definition of $\mathcal{A}_{\kappa}^{-u/4} \hat{\varphi}$ (⁵): this is the operator-valued function constructed expanding $\hat{\varphi}$ as in Eq. (1.13) via a complete orthonormal system of eigenfunctions $(F_k)_{k\in\mathcal{K}}$ of \mathcal{A} and letting $\mathcal{A}_{\kappa}^{-u/4}$ act on each eigenfunction as $\mathcal{A}_{\kappa}^{-u/4}F_k = \kappa^{u/2}\omega_k^{-u/2}F_k$. In this way, Eq. (1.18) yields

$$\hat{\varphi}^{u}(t,\mathbf{x}) = \kappa^{u/2} \int_{\mathcal{K}} \frac{dk}{\sqrt{2}\,\omega_{k}^{(u+1)/2}} \left[\hat{a}_{k} \, e^{-i\omega_{k}t} F_{k}(\mathbf{x}) + \hat{a}_{k}^{\dagger} \, e^{i\omega_{k}t} \overline{F}_{k}(\mathbf{x}) \right]. \tag{1.19}$$

Note that, in the limit $\omega_k \to +\infty$, the term $1/\omega_k^{(u+1)/2}$ in the above integral vanishes rapidly if $\Re u$ is large; this is a manifestation of the regularizing effect of the operator $\mathcal{A}_{\kappa}^{-u/4}$ for large $\Re u$.

 $^{^4\}mathrm{In}$ Section 5 of [64], we also considered some variations of the present setting allowing to deal with infrared issues as well.

⁵We defer to Chapter 3 of this manuscript a more rigorous formulation.

The regularized field $\hat{\varphi}^u$ allows to construct a *regularized propagator*

$$(\boldsymbol{v} | \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \boldsymbol{v}) \qquad (x, y \in \mathbb{R} \times \Omega) , \qquad (1.20)$$

which can be used, in turn, to define regularized versions for the VEV of the other previously mentioned observables as well. In fact, it appears that the regularized propagator (1.20) (along with its derivatives) is regular also along the diagonal y = x if $\Re u$ is large enough; this means that the ultraviolet divergences of $(\boldsymbol{v} | \hat{\varphi}(x) \hat{\varphi}(y) \boldsymbol{v})$ have been cured by the regularization scheme described above.

In particular, using the expansion (1.19) along with the commutation relations (1.14), the diagonal regularized propagator $(\boldsymbol{v} | (\hat{\varphi}^u(t, \mathbf{x}))^2 \boldsymbol{v}) \equiv (\boldsymbol{v} | \hat{\varphi}^u(x) \hat{\varphi}^u(x) \boldsymbol{v})$ can be computed explicitly. The final result is

$$(\boldsymbol{v} | (\hat{\varphi}^{u}(t, \mathbf{x}))^{2} \boldsymbol{v}) = \frac{\kappa^{u}}{2} \int_{\mathcal{K}} \frac{dk}{\omega_{k}^{u+1}} F_{k}(\mathbf{x}) \overline{F}_{k}(\mathbf{x}) ; \qquad (1.21)$$

notice that the right-hand side above does not depend on the time coordinate t. Moreover, the integral appearing in Eq. (1.21) can be interpreted as the eigenfunction expansion of the integral kernel associated to the complex power $\mathcal{A}^{-\frac{u+1}{2}}$ evaluated along the diagonal, i.e., as $\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{x})$; more precisely there holds

$$(\boldsymbol{v} | (\hat{\varphi}^{u}(t, \mathbf{x}))^{2} \boldsymbol{v}) = \frac{\kappa^{u}}{2} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{x}) .$$
(1.22)

Typically, the zeta regularized observables derived in the above framework are analytic functions of the parameter u, for $\Re u$ sufficiently large. Then, according to the (either restricted or extended) zeta approach described in [64], the renormalized versions of this observables are defined in terms of the analytic continuations (or, rather, of their regular parts) at u = 0 of the regularized counterparts; so, for example, we put

$$(\boldsymbol{v} \mid (\hat{\varphi}(t, \mathbf{x}))^2 \, \boldsymbol{v})_{ren} := RP \Big|_{\boldsymbol{u}=0} \, \frac{\kappa^u}{2} \, \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{x}) \tag{1.23}$$

(here $RP|_{u=0}$ indicates the evaluation in u = 0 of the regular part of the related analytic continuation). As exemplified in Eq. (1.23), the renormalized observables are strictly related to the diagonal values of the integral kernels (and of the corresponding derivatives) associated to complex powers of the differential operator $\mathcal{A} = -\Delta + V$.

In view of the above considerations, in order to give a more rigorous and systematic formulation of the approach described above (a formulation to be discussed in Chapter 3), it is first necessary to develop a functional analytic framework allowing to address properly the theory of integral kernels. This topic will be discussed in the forthcoming Chapter 2 were attention is focused, in particular, on the regularity properties of the mentioned kernels and on the methods allowing to construct their analytic continuations.

Chapter 2

Functional spaces, operators and kernels

In this chapter we describe the basic functional analytic framework to which we will refer throughout the remainder of the present manuscript. We first recall some well-known facts about distributions, Sobolev spaces and interpolation of Banach spaces. Next, we describe an abstract setting based on a scale of Hilbert spaces associated to the real powers of a given strictly positive, self-adjoint operator. The case were the mentioned operator is an elliptic Schrödinger-type differential operator is then considered in more detail; the results obtained in this setting allow, among else, to give an alternative formulation of the theory of integral kernels related to a suitable type of operators. Finally, we briefly review some results about Mellin transforms and their analytic continuations.

2.1 Function spaces.

Throughout the entire manuscript Ω denotes a *domain* in \mathbb{R}^d , meaning that $\Omega \subset \mathbb{R}^d$ is an open connected subset. We will mainly focus the attention on the settings i) and ii) described hereafter:

i) Ω is an arbitrary domain in \mathbb{R}^d , with no restrictions regarding either regularity or boundedness; in fact, Ω could also be unbounded with a boundary which is not even continuous (the case $\Omega = \mathbb{R}^d$ is not excluded as well).

ii) Ω is a bounded domain, with compact boundary $\partial \Omega$ of class C^{∞} .

In both cases i) and ii), $\overline{\Omega}$ indicates the closure of Ω , i.e., $\overline{\Omega} := \Omega \cup \partial \Omega$.

Possible variations of the above settings are sometimes taken into account as well, in order to hint at simple generalizations of some results to be discussed in the following.

Remark 2.1. Unless otherwise stated, Ω is always assumed to be as in item i) above.

In the subsequent paragraphs we fix some standards about the main function spaces on Ω to be considered in this work.

Test functions and distributions.

A test function on Ω is a smooth, compactly supported function $f : \Omega \to \mathbb{C}$; these functions form the vector space $\mathcal{D}(\Omega)$, which can be equipped with the well known inductive limit topology (see, e.g., [151]). With respect to this topology, $\mathcal{D}(\Omega)$ is a so-called LF-space, that is a countable, strict inductive limit of Fréchet spaces.

The topological dual space $\mathcal{D}'(\Omega)$ is the locally convex space of Schwartz distributions on Ω ; we write $\langle f, \varphi \rangle$ for the action of a distribution f on a test function φ . Unless otherwise stated, all derivatives considered in the sequel are to be understood in the sense of distributions.

The spaces $L^p(\Omega), L^p_{loc}(\Omega)$.

For any $p \in (0, +\infty)$, these spaces consist of measurable functions $f : \Omega \to \mathbb{C}$ such that $|f|^p$ is Lebesgue integrable on Ω or (in the local case) on any compact subset of Ω . The standard norm on the Banach space $L^p(\Omega)$ is indicated with $\| \|_{L^p}$; we regard $L^2(\Omega)$ as a Hilbert space endowed with the inner product $\langle f|g\rangle_{L^2} := \int_{\Omega} d\mathbf{x} \ \overline{f}(\mathbf{x}) \ g(\mathbf{x}) \equiv \int_{\Omega} \overline{f} \ g$. We use systematically the embeddings $L^p(\Omega) \hookrightarrow L^p_{\text{loc}}(\Omega) \hookrightarrow \mathcal{D}'(\Omega)$.

Sobolev spaces

Let $m \in \mathbb{N}$; we will often consider the Sobolev space (of L^2 type and) of integer order m

$$H^{m}(\Omega) := \{ f \in \mathcal{D}'(\Omega) \mid \partial^{\alpha} f \in L^{2}(\Omega) \text{ for } \alpha \in \mathbb{N}^{d}, \, |\alpha| \leq m \} .$$

$$(2.1)$$

This is a complex Hilbert space with the inner product

$$\langle f|g\rangle_{H^m} := \sum_{\alpha \in \mathbb{N}^d, \, |\alpha| \le m} \langle \partial^{\alpha} f|\partial^{\alpha} g\rangle_{L^2} , \qquad (2.2)$$

inducing the norm $||f||_{H^m} := \sqrt{\langle f|f\rangle_{H^m}}$; of course, $H^0(\Omega) = L^2(\Omega)$. Let $r \in [0, +\infty) \setminus \mathbb{N}$; consider the integer part $[r] \in \mathbb{N}$ and put $\rho := r - [r] \in (0, 1)$. The Sobolev space of fractional order r is

$$H^{r}(\Omega) := \left\{ f \in \mathcal{D}'(\Omega) \mid \partial^{\alpha} f \in L^{2}(\Omega) \text{ for } \alpha \in \mathbb{N}^{d}, \, |\alpha| \leq [r], \text{ and} \right.$$
$$\left. \int_{\Omega \times \Omega} d\mathbf{x} \, d\mathbf{y} \, \frac{|\partial^{\alpha} f(\mathbf{x}) - \partial^{\alpha} f(\mathbf{y})|^{2}}{|\mathbf{x} - \mathbf{y}|^{d+2\rho}} \, < +\infty \text{ for } \alpha \in \mathbb{N}^{d}, \, |\alpha| = [r] \right\};$$
(2.3)

this is also a complex Hilbert space with the inner product

$$\langle f|g\rangle_{H^r} := \tag{2.4}$$

$$\sum_{\alpha \in \mathbb{N}^{d}, |\alpha| \leq [r]} \langle \partial^{\alpha} f | \partial^{\alpha} g \rangle_{L^{2}} + \sum_{\alpha \in \mathbb{N}^{d}, |\alpha| = [r]} \int_{\Omega \times \Omega} d\mathbf{x} \, d\mathbf{y} \, \overline{\frac{(\partial^{\alpha} f(\mathbf{x}) - \partial^{\alpha} f(\mathbf{y}))}{|\mathbf{x} - \mathbf{y}|^{d + 2\rho}}} \, (\partial^{\alpha} g(\mathbf{x}) - \partial^{\alpha} g(\mathbf{y})) \, d\beta_{\alpha} \, d\beta_{\alpha$$

inducing the norm $||f||_{H^r} := \sqrt{\langle f|f\rangle_{H^r}}$.

2.1. FUNCTION SPACES.

For any $r \in [0, +\infty)$ (integer or not), we denote with $H_0^r(\Omega)$ the closure of $\mathcal{D}(\Omega)$ in the space $H^r(\Omega)$ (with respect to the associated norm); $H_0^r(\Omega)$ is itself a Hilbert space with the inner product (2.4) inherited from $H^r(\Omega)$. We denote with $H^{-r}(\Omega)$ its topological dual; so, each $f \in H^{-r}(\Omega)$ is a continuous linear form on $H_0^r(\Omega)$ and, by restriction to $\mathcal{D}(\Omega)$, it can be identified with a distribution on Ω . By the Riesz theorem, there is a unique antilinear isomorphism $i_r : H_0^r(\Omega) \to H^{-r}(\Omega)$ such that $\langle i_r f, g \rangle = \langle f | g \rangle_{H^r}$ for all $g \in H^r(\Omega)$; we define an inner product on $H^{-r}(\Omega)$ setting

$$\langle f|g\rangle_{H^{-r}} := \langle i_r^{-1}g|i_r^{-1}f\rangle_{H^r}$$
 (2.5)

Poincaré type inequalities.

Two inequalities of this kind will be considered in this work.

i) Assume the (otherwise arbitrary) domain $\Omega \subset \mathbb{R}^d$ to be bounded along a direction; this means that there exist a unit vector $\mathbf{n} \in \mathbb{R}^d$ and two constants $a \in \mathbb{R}$, $h \in (0, +\infty)$ such that Ω is contained in the strip $\{\mathbf{x} \in \mathbb{R}^d \mid a \leq \mathbf{n} \cdot \mathbf{x} \leq a + h\}$ (of course, h represents the width of the strip). Under these conditions, it is known that there is a constant $c_{\Omega} \in (0, +\infty)$ such that

$$\int_{\Omega} |f|^2 \le c_{\Omega} \int_{\Omega} |\nabla f|^2 \qquad \text{for all } f \in H^1_0(\Omega)$$
(2.6)

(here $|\nabla f|^2 := \sum_{i=1}^d |\partial_{x^i} f|^2$); as well known (see, e.g., [54, 109]), one can take $c_{\Omega} = h^2/2$. ii) Another inequality we will refer to involves functions of zero mean (rather than functions vanishing on the boundary of the domain) with domain consisting of any ball $B(\mathbf{x}_0, r)$ in \mathbb{R}^d of center \mathbf{x}_0 and radius r. It is known that there exists a constant $c_B \in (0, +\infty)$, depending only on d, such that for any $\mathbf{x}_0 \in \mathbb{R}^d$ and $r \in (0, +\infty)$

$$\int_{B(\mathbf{x}_0,r)} |f|^2 \le c_B r^2 \int_{B(\mathbf{x}_0,r)} |\nabla f|^2 \quad \text{for all } f \in H^1(B(\mathbf{x}_0,r)) \text{ s.t. } \int_{B(\mathbf{x}_0,r)} f = 0 ; \quad (2.7)$$

one can take $c_B = 2^{2d}$ (see [79], page 164, Eq. (7.45)).

Local Sobolev spaces.

For any $r \in \mathbb{R}$, we consider the local Sobolev space of order r (see [38])

$$H^{r}_{\text{loc}}(\Omega) := \{ f \in \mathcal{D}'(\Omega) \mid \varphi f \in H^{r}(\Omega) \text{ for any } \varphi \in \mathcal{D}(\Omega) \} .$$
(2.8)

This carries the locally convex topology induced by the family of seminorms

$$f \mapsto \|\varphi f\|_{H^r} \qquad (\varphi \in \mathcal{D}(\Omega)) ,$$
 (2.9)

with $\| \|_{H^r}$ indicating the H^r norm.

Spaces of differentiable functions and Sobolev embeddings.

Let $j \in \mathbb{N}$; we denote with $C^{j}(\Omega)$ the space of functions $f : \Omega \to \mathbb{C}$ which are continuous on Ω along with all their partial derivatives $\partial^{\alpha} f$ of orders $|\alpha| \leq j$. We endow $C^{j}(\Omega)$ with the Fréchet topology induced by the family of seminorms

$$f \mapsto |f|_{C^{j},K} := \max_{\alpha \in \mathbb{N}^{d}, |\alpha| \le j} \sup_{\mathbf{x} \in K} |\partial^{\alpha} f(\mathbf{x})| \qquad (K \subset \Omega \text{ compact})$$
(2.10)

(¹); of course, this family is equivalent to the set of seminorms

$$f \mapsto |f|_{C^{j,\varphi}} := \max_{\alpha \in \mathbb{N}^{d}, \, |\alpha| \le j} \, \sup_{\mathbf{x} \in \Omega} |\varphi(\mathbf{x}) \, \partial^{\alpha} f(\mathbf{x})| \qquad (\varphi \in \mathcal{D}(\Omega)) \, . \tag{2.11}$$

On the other hand, we indicate with $C^{j}(\overline{\Omega})$ the linear subspace of $C^{j}(\Omega)$ consisting of functions f that, along with all their derivatives $\partial^{\alpha} f$ of order $|\alpha| \leq j$, admit continuous extensions to $\overline{\Omega}$; f, $\partial^{\alpha} f$ are often identified with these extensions, which are unique. If Ω is bounded, $C^{j}(\overline{\Omega})$ is a Banach space with respect to the norm

$$||f||_{C^j} := \max_{\alpha \in \mathbb{N}^d, \, |\alpha| \le j} \, \sup_{\mathbf{x} \in \Omega} |\partial^{\alpha} f(\mathbf{x})| \,.$$
(2.12)

We shall also need the spaces of functions whose derivatives satisfy a Hölder condition for some exponent $\lambda \in (0, 1]$. More precisely, we write $C^{j,\lambda}(\Omega)$ for the space of functions $f \in C^{j}(\Omega)$ whose derivatives $\partial^{\alpha} f$ of order $|\alpha| = j$ fulfill, for any compact subset $K \subset \Omega$,

$$\sup_{\mathbf{x},\mathbf{y}\in K,\,\mathbf{x}\neq\mathbf{y}}\frac{|\partial^{\alpha}f(\mathbf{x})-\partial^{\alpha}f(\mathbf{y})|}{|\mathbf{x}-\mathbf{y}|^{\lambda}}<+\infty\;;$$
(2.13)

we endow this space with the Fréchet topology induced by family of seminorms

$$|f|_{C^{j,\lambda},K} := |f|_{C^{j,K}} + \max_{\alpha \in \mathbb{N}^d, \, |\alpha| = j} \sup_{\mathbf{x}, \mathbf{y} \in K, \, \mathbf{x} \neq \mathbf{y}} \frac{|\partial^{\alpha} f(\mathbf{x}) - \partial^{\alpha} f(\mathbf{y})|}{|\mathbf{x} - \mathbf{y}|^{\lambda}} \quad (K \subset \Omega \text{ compact}) \text{ . } (2.14)$$

For Ω bounded, we indicate with $C^{j,\lambda}(\overline{\Omega})$ the space of functions $f \in C^j(\overline{\Omega})$ whose derivatives $\partial^{\alpha} f$ of order $|\alpha| = j$ fulfill

$$\sup_{\mathbf{x},\mathbf{y}\in\Omega,\,\mathbf{x}\neq\mathbf{y}}\frac{|\partial^{\alpha}f(\mathbf{x})-\partial^{\alpha}f(\mathbf{y})|}{|\mathbf{x}-\mathbf{y}|^{\lambda}}<+\infty;$$
(2.15)

this becomes a Banach space if equipped with the norm

$$\|f\|_{C^{j,\lambda}} := \|f\|_{C^j} + \max_{\alpha \in \mathbb{N}^d, \, |\alpha| = j} \sup_{\mathbf{x}, \mathbf{y} \in \Omega, \, \mathbf{x} \neq \mathbf{y}} \frac{|\partial^{\alpha} f(\mathbf{x}) - \partial^{\alpha} f(\mathbf{y})|}{|\mathbf{x} - \mathbf{y}|^{\lambda}} \,. \tag{2.16}$$

There hold the following well-known results.

¹In fact, this family is equivalent to the countable set of seminorms which corresponds to any given sequence of compact subsets $K_0 \subset K_1 \subset K_2 \subset ...$ such that $\Omega = \bigcup_{n \in \mathbb{N}} K_n$.

Theorem 2.1. (Sobolev Embedding) Let $r \in \mathbb{R}$ and $j \in \mathbb{N}$ be such that r > j + d/2; then, there hold the following continuous embeddings. *i*) For any domain $\Omega \subset \mathbb{R}^d$, there holds

$$H^r_{loc}(\Omega) \hookrightarrow C^j(\Omega) ;$$
 (2.17)

more precisely, for any $\lambda \in (0,1)$ such that $r > j + d/2 + \lambda$, it is

$$H^r_{loc}(\Omega) \hookrightarrow C^{j,\lambda}(\Omega) \hookrightarrow C^j(\Omega)$$
 . (2.18)

ii) If $\Omega \subset \mathbb{R}^d$ is a bounded domain, with compact boundary $\partial \Omega$ of class C^{∞} , there holds

$$H^r(\Omega) \hookrightarrow C^j(\overline{\Omega})$$
; (2.19)

more precisely, for any $\lambda \in (0,1)$ such that $r > j + d/2 + \lambda$, it is

$$H^r(\Omega) \hookrightarrow C^{j,\lambda}(\overline{\Omega}) \hookrightarrow C^j(\overline{\Omega})$$
 . (2.20)

Remark 2.2. We refer to Proposition 2.13 of [38] and to Corollary 9.1 on page 46 of [97] for the proof of statements i) and ii), respectively. See also [7], Theorems 5.4 and 6.2, for generalizations of the above theorem dealing also with configurations involving unbounded domains whose boundary only fulfill regularity conditions much weaker than the ones in statement ii).

The case $\Omega = \mathbb{R}^d$: tempered distributions, Fourier transforms and another characterization of Sobolev spaces.

We write $\mathcal{S}(\mathbb{R}^d)$ for the Schwartz space of smooth functions $f : \mathbb{R}^d \to \mathbb{C}$ rapidly decreasing at infinity with all their derivatives; this space has well known topology [151], and its topological dual $\mathcal{S}'(\mathbb{R}^d)$ is called the space of tempered distributions. Any $f \in \mathcal{S}'(\mathbb{R}^d)$, after restriction to the space $\mathcal{D}(\Omega)$, can be identified with an element of $\mathcal{D}'(\Omega)$. A Fourier transform can be defined

$$\mathfrak{F}: \mathcal{S}'(\mathbb{R}^d) \to \mathcal{S}'(\mathbb{R}^d) , \qquad f \mapsto \mathfrak{F}f ;$$
 (2.21)

we use for \mathfrak{F} the normalization such that

$$(\mathfrak{F}f)(\mathbf{k}) = \frac{1}{(2\pi)^{d/2}} \int_{\mathbb{R}^d} d\mathbf{x} \ f(\mathbf{x}) \ e^{-i\mathbf{k}\cdot\mathbf{x}} \qquad (f \in \mathcal{S}(\mathbb{R}^d); \ \mathbf{k} \in \mathbb{R}^d) \ . \tag{2.22}$$

For any $r \in \mathbb{R}$ we have the following, equivalent characterization of the spaces $H^r(\mathbb{R}^d)$:

$$H^{r}(\mathbb{R}^{d}) = \{ f \in \mathcal{S}'(\mathbb{R}^{d}) \mid (1 + |\mathbf{k}|^{2})^{r/2} \mathfrak{F} f \in L^{2}(\mathbb{R}^{d}) \}$$
(2.23)

(where $(1 + |\mathbf{k}|^2)^{r/2} \mathfrak{F} f$ indicates multiplication of the Fourier transform $\mathfrak{F} f$ by the smooth function $\mathbb{R}^d \to \mathbb{R}$, $\mathbf{k} \mapsto (1 + |\mathbf{k}|^2)^{r/2}$; the quantity

$$\langle f|g\rangle_{H^r(\mathbb{R}^d)} := \langle (1+|\mathbf{k}|^2)^{r/2} \mathfrak{F} \mid (1+|\mathbf{k}|^2)^{r/2} \mathfrak{F} g\rangle_{L^2}$$
(2.24)

yields an inner product on $H^r(\mathbb{R}^d)$ equivalent to the inner products introduced previously for any space $H^r(\Omega)$ (considering separately the cases $r \in \mathbb{N}$, $r \in [0, +\infty) \setminus \mathbb{N}$, and $r \in (-\infty, 0)$). Moreover, there holds the continuous embedding (²)

$$H^u(\mathbb{R}^d) \hookrightarrow H^r(\mathbb{R}^d) \quad \text{for } r, u \in \mathbb{R} \text{ with } u \ge r .$$
 (2.25)

2.2 Some known results on Banach spaces.

If \mathcal{G} is any topological vector space, we write \mathcal{G}' for the dual vector space, made of the continuous linear forms $\alpha : \mathcal{G} \to \mathbb{C}, g \mapsto \langle \alpha, g \rangle$. If \mathcal{G} is a Banach space with a norm $|| \, ||, \mathcal{G}'$ is as well a Banach space with the norm $|| \alpha ||' := \sup_{g \in \mathcal{G}, g \neq 0} \frac{|\langle \alpha, g \rangle|}{||g||}$; in particular, for each $r \in \mathbb{R}, (\mathcal{H}^r)'$ is a Banach space with the norm $|| \, ||'_r$ induced by $|| \, ||_r$.

Notations for operators. Banach adjoints.

Let X, Y be any two Banach spaces, endowed with the standard norm topology; we denote with X', Y' their topological duals, and with \langle , \rangle the bilinear duality pairing between X' and X (resp. Y' and Y).

An operator \mathcal{B} : Dom $(\mathcal{B}) \subset X \to Y$ is a linear map whose domain Dom (\mathcal{B}) is a linear subspace of X; if \mathcal{B} is injective, we write \mathcal{B}^{-1} for the inverse. We indicate with $\mathfrak{B}(X,Y)$ the Banach space of continuous (i.e., bounded) operators from X to Y; this is equipped with the standard operator norm $\| \|_{\mathfrak{B}(X,Y)}$ (and with the induced topology).

The Banach adjoint of a continuous operator $\mathcal{B} \in \mathfrak{B}(X, Y)$ is the unique continuous operator $\mathcal{B}^* \in \mathfrak{B}(Y', X')$ such that

$$\langle \mathcal{B}^*g, f \rangle_X = \langle g, \mathcal{B}f \rangle_Y \quad \text{for all } f \in X, \ g \in Y;$$
 (2.26)

we refer to the standard theory analyzed, e.g., in [42, 129]. It can be easily proved that the adjoint map $*: \mathfrak{B}(X,Y) \to \mathfrak{B}(Y',X'), \mathcal{B} \mapsto \mathcal{B}^*$ is an isometric isomorphism (³); moreover, if X and Y are reflexive (i.e., X'' = X and Y'' = Y), any continuous operator $\mathcal{B} \in \mathfrak{B}(X,Y)$ coincides with its double adjoint $\mathcal{B}^{**} \equiv (\mathcal{B}^*)^* \in \mathfrak{B}(X'',Y'')$. Given a third Banach space Z and any two continuous operators $\mathcal{B} \in \mathfrak{B}(X,Y), \mathcal{C} \in \mathfrak{B}(Y,Z)$, the composition $\mathcal{CB}: X \to Z$ is also continuous and its adjoint $(\mathcal{CB})^*$ fulfills $(\mathcal{CB})^* = \mathcal{B}^*\mathcal{C}^* \in \mathfrak{B}(Z',X')$.

We will employ the above mentioned facts in subsection 2.5.

$$\|f\|_{H^r(\mathbb{R}^d)}^2 = \|(1+|\mathbf{k}|^2)^{r/2} \mathfrak{F}f\|_{L^2}^2 \le \|(1+|\mathbf{k}|^2)^{u/2} \mathfrak{F}f\|_{L^2}^2 = \|f\|_{H^u(\mathbb{R}^d)}^2.$$

³For a proof of this fact, see, e.g., [129], page 186, Theorem VI.2.

² In fact, using the norm $\| \|_{H^r(\mathbb{R}^d)}$ induced by inner product (2.24), for any $f \in H^u(\mathbb{R}^d)$ one has

Integration of Banach-valued functions.

Let us consider a real (or complex) Banach space X and its topological dual X', equipped with the standard norm topology; moreover, let $(\Omega, \mathfrak{M}_{\Omega}, \mu)$ be a measure space (with \mathfrak{M}_{Ω} a σ -algebra of parts of Ω). A function $f : \Omega \to X$ is said to be weakly measurable if, for each $\alpha \in X'$, the function $\langle \alpha, f \rangle : \Omega \to \mathbb{R}$ (or \mathbb{C}), $\mathbf{x} \mapsto \langle \alpha, f(\mathbf{x}) \rangle$ is measurable in the usual sense. The function f is said to be weakly integrable (or integrable in the Gelfand-Pettis sense) if it is weakly measurable, $\langle \alpha, f \rangle$ is Lebesgue integrable for all $\alpha \in X'$ and there is an element of X, indicated with $\int_{\Omega} f(\mathbf{x}) d\mu(\mathbf{x}) \equiv \int_{\Omega} f d\mu$, such that

$$\langle \alpha, \int_{\Omega} f \, d\mu \rangle = \int_{\Omega} \langle \alpha, f \rangle \, d\mu \quad \text{for all } \alpha \in X' ;$$
 (2.27)

the above element of X is unique if it exists, and it is called the *integral of* f *in the weak* (or Gelfand-Pettis) sense.

In the following developments we will need the forthcoming result.

Theorem 2.2. Let X, Y be any two real (or complex) Banach spaces, $f : \Omega \to X$ a weakly integrable function and let $\mathcal{B} : X \to Y$ be a continuous operator; then, the function $\mathcal{B}f : \Omega \to Y, \mathbf{x} \mapsto (\mathcal{B}(f(\mathbf{x})) \equiv (\mathcal{B}f)(\mathbf{x}) \text{ is weakly integrable and}$

$$\mathcal{B}\left(\int_{\Omega} f \, d\mu\right) = \int_{\Omega} (\mathcal{B}f) \, d\mu \ . \tag{2.28}$$

See, e.g., [50] and [125] for more details about the topics discussed in the present paragraph (in particular, for the proof of Theorem 2.2).

Abstract interpolation of complex Banach spaces.

Two complex vector spaces X_0, X_1 are said to be interpolable if their set theoretical intersection $X_0 \cap X_1$ is a linear subspace of both of them. In this case there exists a vector space $X_0 + X_1$ that contains X_0 and X_1 as subspaces, and is generated by them; $X_0 + X_1$ is unique up to linear isomorphisms that preserve the elements of X_0 and X_1 .

Now let us consider two complex Banach spaces X_0 and X_1 , with norms $|| ||_{X_0}$ and $|| ||_{X_1}$, respectively; these are said to be interpolable if the underlying vector spaces are so. In this case, both the vector spaces $X_0 \cap X_1$ and $X_0 + X_1$ are Banach spaces with respect to the norms $||x||_{X_0\cap X_1} := \max(||x||_{X_0}, ||x||_{X_1})$ and $||x||_{X_0+X_1} := \inf\{||x_0||_{X_0} + ||x_1||_{X_1} | x_0 \in$ $X_0, x_1 \in X_1, x_0 + x_1 = x\}$, respectively. It turns out that $X_0, X_1, X_0 \cap X_1 \hookrightarrow X_0 + X_1$ (continuous embeddings).

To go on, let us consider the strip $S := \{z \in \mathbb{C} \mid 0 < \Re z < 1\}$ in the complex plane \mathbb{C} , and its closure \overline{S} (that is the union of S with the lines $s_0 := \{\Re z = 0\}$ and $s_1 := \{\Re z = 1\}$. Let us introduce the function space

$$\mathfrak{F}(X_0, X_1) := \{F \in \operatorname{Hol}(S, X_0 + X_1) \cap C(\overline{S}, X_0 + X_1) \mid (F \upharpoonright s_i) \in C_B(s_i, X_i) \text{ for } i = 0, 1\},$$
(2.29)

where Hol, C, C_B mean, respectively, holomorphic, continuous, continuous and bounded. $\mathfrak{F}(X_0, X_1)$ is a vector space and it becomes a Banach space when it is equipped with the norm

$$||F||_{\mathfrak{F}(X_0,X_1)} := \max\left(\sup_{z \in s_0} ||F(z)||_{X_0}, \sup_{z \in s_1} ||F(z)||_{X_1}\right).$$
(2.30)

For $\theta \in (0, 1)$, the interpolation space of order θ between X_0 and X_1 is

$$[X_0, X_1]_{\theta} := \{ x \in X_0 + X_1 \mid x = F(\theta) \text{ for some } F \in \mathfrak{F}(X_0, X_1) ; \qquad (2.31)$$

this is a linear subspace of $X_0 + X_1$, and it is a Banach space with respect to the norm

$$\|x\|_{[X_0,X_1]_{\theta}} := \inf\{\|F\|_{\mathfrak{F}(X_0,X_1)} \mid F \in \mathfrak{F}(X_0,X_1), \ F(\theta) = x\}.$$
(2.32)

It turns out that $X_0 \cap X_1 \hookrightarrow [X_0, X_1]_{\theta} \hookrightarrow X_0 + X_1$. In particular, assume $X_1 \hookrightarrow X_0$. Then the two spaces X_0, X_1 are interpolable; in terms of linear structures we have $X_0 \cap X_1 = X_1$, and we can take $X_0 + X_1 = X_0$. Moreover the norms defined previously for $X_0 \cap X_1$ and $X_0 + X_1$ are equivalent to the norms of X_1 and X_0 , respectively. Summing up, in this case, $X_1 \hookrightarrow [X_0, X_1]_{\theta} \hookrightarrow X_0$.

The main theorem in interpolation theory states the following [1, 19, 101]:

Theorem 2.3. Let X_0, X_1 and Y_0, Y_1 be two arbitrary pairs of interpolable Banach spaces and consider two continuous linear operators $\mathcal{B}_0 : X_0 \to Y_0$ and $\mathcal{B}_1 : X_1 \to Y_1$ such that $\mathcal{B}_0 \upharpoonright X_0 \cap X_1 = \mathcal{B}_1 \upharpoonright X_0 \cap X_1$. Let $\mathcal{B} : X_0 + X_1 \to Y_0 + Y_1$ be the unique linear map such that $\mathcal{B} \upharpoonright X_0 = \mathcal{B}_0$ and $\mathcal{B} \upharpoonright X_1 = \mathcal{B}_1$. Then, for each $\theta \in (0, 1)$, $\mathcal{B}_\theta := \mathcal{B} \upharpoonright X_\theta$ maps continuously $[X_0, X_1]_\theta$ to $[Y_0, Y_1]_\theta$ and

$$\|\mathcal{B}_{\theta}\|_{\mathfrak{B}([X_{0},X_{1}]_{\theta},[Y_{0},Y_{1}]_{\theta})} \leq (\|\mathcal{B}_{0}\|_{\mathfrak{B}(X_{0},Y_{0})})^{1-\theta} (\|\mathcal{B}_{1}\|_{\mathfrak{B}(X_{1},Y_{1})})^{\theta} .$$
(2.33)

Many applications of interpolation theory are related to L^p spaces. More precisely, consider a measure space $(K, \mathfrak{M}_K, \mu) \equiv K$, with μ a positive measure on some σ -algebra \mathfrak{M}_K of subsets of K; let us write $\operatorname{Mis}(K)$ for the vector space of complex measurable functions $K \to \mathbb{C}$ (or, more precisely, of the equivalence classes of such functions with respect to equality almost everywhere for the measure μ). Let $p \in [1, +\infty]$, and $\omega \in \operatorname{Mis}(K)$ be such that $\omega(k) > 0$ for almost every (a.e.) $k \in K$; consider the Banach space $L^p(K, \omega \, d\mu)$, made of functions $f \in \operatorname{Mis}(K)$ which are L^p with respect to the measure $\mathfrak{M}_K \ni A \mapsto \int_A \omega \, d\mu$. Let $p_0, p_1 \in [1, +\infty]$ and $\omega_0, \omega_1 \in \operatorname{Mis}(K)$, with $\omega_0, \omega_1 > 0$ almost everywhere. Then, the

Let $p_0, p_1 \in [1, +\infty]$ and $\omega_0, \omega_1 \in Mis(K)$, with $\omega_0, \omega_1 > 0$ almost everywhere. Then, the Banach spaces $L^{p_0}(K, \omega_0 d\mu)$ and $L^{p_1}(K, \omega_1 d\mu)$ are both interpolable, and their sum can be realized naturally as a linear subspace of Mis(K); moreover, for all $\theta \in (0, 1)$ one has (see [1], Theorem 2.12)

$$[L^{p_0}(K,\omega_0 d\mu), L^{p_1}(K,\omega_1 d\mu)]_{\theta} = L^p(K,\omega d\mu) , \qquad (2.34)$$

where $p \in [1, +\infty]$ and $\omega \in Mis(K)$ are defined, respectively, by

$$\frac{1}{p} = \frac{1-\theta}{p_0} + \frac{\theta}{p_1} , \qquad \omega(k) := \omega_0(k)^{\frac{p}{p_0}(1-\theta)} \omega_1(k)^{\frac{p}{p_1}\theta} \text{ for a.e. } k \in K .$$
(2.35)

2.3. OPERATORS ON HILBERT SPACES.

The equality in Eq. (2.34) must be intended as follows: the left and the right-hand side coincide as vector spaces, and carry the same norm.

A subcase of this result, of interest for us in the sequel, is the one with $p_0 = p_1 \equiv p$ and $\omega_0 = w^{r_0}$, $\omega_1 = w^{r_1}$ for some $w \in \text{Mis}(K)$, with w > 0 almost everywhere, and $r_0, r_1 \in \mathbb{R}$. In this case, Eq.s (2.34) and (2.35) give

$$[L^{p}(K, w^{r_{0}}d\mu), L^{p}(K, w^{r_{1}}d\mu)]_{\theta} = L^{p}(K, w^{(1-\theta)r_{0}+\theta r_{1}}d\mu) .$$
(2.36)

Standard interpolation results for the Sobolev spaces $H^n(\mathbb{R}^d)$ can be derived by Eq. (2.36); indeed, for each $r \in \mathbb{R}$, the space $H^r(\mathbb{R}^d)$ of Eq. (2.23) can be identified (via the Fourier transform) with $L^2(\mathbb{R}^d, w^r d\mathbf{k})$, where $w(\mathbf{k}) := 1 + |\mathbf{k}|^2$ and $d\mathbf{k}$ is the standard Lebesgue measure. So, for all $r_0, r_1 \in \mathbb{R}$ and $\theta \in (0, 1)$, one infers from Eq. (2.36) that

$$[H^{r_0}(\mathbb{R}^d), H^{r_1}(\mathbb{R}^d)]_{\theta} = H^{(1-\theta)r_0 + \theta r_1}(\mathbb{R}^d) .$$
(2.37)

A result analogous to Eq. (2.37) holds as well for the Sobolev spaces $H^r(\Omega)$ $(r \in \mathbb{R})$, under minimal regularity assumptions for the domain $\Omega \subset \mathbb{R}^d$; in particular, if Ω is bounded with compact boundary of class C^{∞} , for any $r_0, r_1 \in \mathbb{R}$ and for any $\theta \in (0, 1)$ one has $(^4)$

$$[H^{r_0}(\Omega), H^{r_1}(\Omega)]_{\theta} = H^{(1-\theta)r_0 + \theta r_1}(\Omega) .$$
(2.38)

Let us stress that, contrary to the previous examples (2.34-2.37), the spaces on the two sides of the above equality are in general endowed with distinct norms (⁵): Eq. (2.38) must be meant to hold algebraically and topologically, in the sense that the mentioned norms are equivalent.

For a more comprehensive analysis of the topics reviewed briefly in the present paragraph, we refer, e.g., to the already cited books [1, 19, 101].

2.3 Operators on Hilbert spaces.

In the sequel we consider a complex, separable Hilbert space \mathcal{H} ; we write $\langle | \rangle$ for the inner product of \mathcal{H} (antilinear in the left argument, linear in the right one) and || || for the induced norm.

Notations for operators.

Similarly to the case of Banach spaces (compare with Section 2.2), an operator on \mathcal{H} is a linear map $\mathcal{A} : \text{Dom}(\mathcal{A}) \subset \mathcal{H} \to \mathcal{H}$, whose domain $\text{Dom}(\mathcal{A})$ is a linear subspace of \mathcal{H} ; $\mathfrak{B}(\mathcal{H}) \equiv \mathfrak{B}(\mathcal{H}, \mathcal{H})$ is the space of continuous (i.e., bounded) operators from \mathcal{H} into \mathcal{H} , and $\| \|_{\mathfrak{B}} \equiv \| \|_{\mathfrak{B}(\mathcal{H},\mathcal{H})}$ indicates the usual operator norm on this space.

 $^{{}^{4}}$ Eq. (2.38) contains a well-known result on Sobolev spaces; see, e.g., Theorem 7.48 of [7], Proposition 2.7 of [96] and Theorem 9.6 of [97] for more details.

⁵In fact, the space $[H^{r_0}(\Omega), H^{r_1}(\Omega)]_{\theta}$ carries the interpolation norm defined according to Eq. (2.32) while, if $r_0, r_1 \geq 0, H^{(1-\theta)r_0+\theta r_1}(\Omega)$ has the standard norm induced by the inner product (2.4).

If $\text{Dom}(\mathcal{A})$ is dense in \mathcal{H} , we can define the *Hilbert adjoint* operator \mathcal{A}^{\dagger} (see, e.g., [119, 129]). Let us stress that if \mathcal{A} is continuous and $\text{Dom}(\mathcal{A}) = \mathcal{H}$ the notion of Banach adjoint defined in Section 2.2 reduces to that of Hilbert adjoint considered here; in fact, viewing \mathcal{H} as an (anti-)self-dual Banach space, one has $\mathcal{A}^* = \mathcal{A}^{\dagger} \in \mathfrak{B}(\mathcal{H})$. In the remainder of this work, for the sake of brevity, we use the bare adjective "adjoint" as short for "Hilbert adjoint".

Given a pair of linear operators $\mathcal{A} : \text{Dom}(\mathcal{A}) \subset \mathcal{H} \to \mathcal{H}, \mathcal{B} : \text{Dom}(\mathcal{B}) \subset \mathcal{H} \to \mathcal{H}$ and a constant $c \in \mathbb{C}$, the operators $c\mathcal{A}, \mathcal{A} + \mathcal{B}$ and $\mathcal{B}\mathcal{A}$ are obviously defined on the domains $\text{Dom}(c\mathcal{A}) := \text{Dom}(\mathcal{A}), \text{Dom}(\mathcal{A}+\mathcal{B}) := \text{Dom}(\mathcal{A}) \cap \text{Dom}(\mathcal{B})$ and $\text{Dom}(\mathcal{B}\mathcal{A}) := \{f \in \mathcal{H} \mid f \in \text{Dom}(\mathcal{A}), \mathcal{A}f \in \text{Dom}(\mathcal{B})\}$. If \mathcal{C} is another operator on \mathcal{H} we define $\mathcal{A}+\mathcal{B}+\mathcal{C} := (\mathcal{A}+\mathcal{B})+\mathcal{C}, \mathcal{C}\mathcal{B}\mathcal{A} := \mathcal{C}(\mathcal{B}\mathcal{A})$, and so on iteratively. We write $\mathcal{B} \supset \mathcal{A}$ if \mathcal{B} is an extension of \mathcal{A} .

In the sequel we will write $\sigma(\mathcal{A})$ for the *spectrum* of an operator \mathcal{A} , defined as in [119, 129].

The case $\mathcal{H} = L^2(K, \mu)$. Multiplication operators.

Let us consider again a measure space $(K, \mathfrak{M}_K, \mu) \equiv (K, \mu)$, with μ a positive measure on some σ -algebra \mathfrak{M}_K of subsets of K; let us recall that we write k for a generic point of Kand $\operatorname{Mis}(K)$ for the space of measurable functions on K (see subsection 2.2). The space $L^2(K, \mu)$ of complex, square integrable, measurable functions on K is a Hilbert space with the inner product

$$\langle f|g\rangle := \int_{K} \overline{f}(k) g(k) d\mu(k) \equiv \int_{K} \overline{f} g d\mu , \qquad (2.39)$$

inducing the norm

$$||f||^2 = \int_K |f|^2 \, d\mu \; . \tag{2.40}$$

To go on, let us consider a function $w \in Mis(K)$; the multiplication operator by w is

$$M_w: \operatorname{Dom}(M_w) \subset L^2(K,\mu) \to L^2(K,\mu), \qquad f \mapsto M_w f := wf$$
(2.41)

with

$$Dom(M_w) := \{ f \in L^2(K, \mu) \mid wf \in L^2(K, \mu) \} .$$
(2.42)

This is a linear operator with dense domain, whose adjoint is

$$(M_w)^{\dagger} = M_{\overline{w}} ; \qquad (2.43)$$

in particular, M_w is self-adjoint if w is real valued. The operator M_w is defined on the whole space $L^2(K,\mu)$ and bounded (i.e., $M_w \in \mathfrak{B}(L^2(K,\mu))$) if and only if w is essentially bounded (i.e., bounded up to sets of zero μ measure); in this case

$$\|M_w\|_{\mathfrak{B}} = \operatorname{ess\,sup}_{k \in K} |w(k)| \ . \tag{2.44}$$

For any given $w \in Mis(K)$, the spectrum of M_w is as follows:

$$\sigma(M_w) = \operatorname{EssIm} w , \qquad (2.45)$$

2.3. OPERATORS ON HILBERT SPACES.

where the right-hand side indicates the essential image of w, defined by

EssIm
$$w := \{\lambda \in \mathbb{C} \mid \mu(w^{-1}(B(\lambda, \epsilon))) > 0 \text{ for each } \epsilon > 0\}$$
 (2.46)

(with $B(\lambda, \epsilon) := \{\lambda' \in \mathbb{C} \mid |\lambda' - \lambda| < \epsilon\}$).

If w is real valued, in which case M_w is self-adjoint, one infers from Eq. (2.45) that, for all $a \in \mathbb{R}$,

$$w \ge a \quad \mu\text{-almost everywhere} \Leftrightarrow \sigma(M_w) \subset [a, +\infty) .$$
 (2.47)

Given any pair of functions $v, w \in Mis(K)$ one has

$$M_{v+w} \supset M_w + M_w , \qquad M_{vw} \supset M_v M_w , \qquad (2.48)$$

with \supset replaced by = if v and w are essentially bounded. Moreover, for $c \in \mathbb{C} \setminus \{0\}$, it is

$$M_{cw} = c M_w , \qquad (2.49)$$

while $M_{0 \cdot w} \supset 0 \cdot M_w$.

Spectral theorem and functional calculus for self-adjoint operators.

Let us consider an abstract, separable Hilbert space \mathcal{H} and a self-adjoint operator \mathcal{A} : Dom $(\mathcal{A}) \subset \mathcal{H} \to \mathcal{H}$. The "multiplication operator version" of the spectral theorem (see [129]) states that there exist a measure space (K, μ) , a measurable function $w : K \to \mathbb{R}$ and a Hilbertian isomorphism $\mathcal{I} : \mathcal{H} \to L^2(K, \mu)$ such that

$$\mathcal{A} = \mathcal{I}^{-1} M_w \mathcal{I} . \tag{2.50}$$

Obviously enough, $\mathcal{I}^{-1}M_w\mathcal{I}$ indicates the linear operator $f \mapsto \mathcal{I}^{-1}M_w\mathcal{I}f$ with domain $\mathcal{I}^{-1}(\text{Dom}(M_w))$. The set (K, μ, w, \mathcal{I}) is not uniquely determined by \mathcal{A} , but this is no cause of concern for the relevant constructions associated to this representation and, in particular, for the functional calculus described hereafter. For future use, we mention that one can choose (K, μ) so that $\mu(K) < +\infty$ (see [129], page 260, Theorem VIII.4). Now, let us consider a measurable function

$$\phi: \sigma(\mathcal{A}) \subset \mathbb{R} \to \mathbb{C} . \tag{2.51}$$

As well known, the functional calculus for self-adjoint operators allows to associate to ϕ an operator $\phi(\mathcal{A}) : \text{Dom}(\phi(\mathcal{A})) \subset \mathcal{H} \to \mathcal{H}$ with dense domain, which is everywhere defined and bounded if ϕ is bounded; this has adjoint $\phi(\mathcal{A})^{\dagger} = \overline{\phi}(\mathcal{A})$ where $\overline{\phi} : \sigma(\mathcal{A}) \to \mathbb{C}$ is the complex conjugate function. If (K, μ, w, \mathcal{I}) is a set as above, the composition $\phi \circ w \in \text{Mis}(K)$ makes sense, because $w(k) \in \sigma(\mathcal{A})$ for μ -a.e. $k \in K$; it turns out that

$$\phi(\mathcal{A}) = \mathcal{I}^{-1} M_{\phi \circ w} \mathcal{I} : \mathcal{I}^{-1}(\text{Dom}(M_{\phi \circ w})) \subset \mathcal{H} \to \mathcal{H}$$
(2.52)

(6). As an example, for $z \in \mathbb{C}$, $e^{-z\mathcal{A}}$ is defined as above choosing $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$, $\phi(\lambda) := e^{-z\lambda}$; this operator is self-adjoint if $z \in \mathbb{R}$, and $e^{-z\mathcal{A}} \in \mathfrak{B}(\mathcal{H})$ if $\sigma(\mathcal{A}) \subset [0, +\infty)$ and $\Re z \ge 0$.

⁶As a matter of fact, one could define $\phi(\mathcal{A})$ as the unique operator on \mathcal{H} fulfilling Eq. (2.52), for each (K, μ, w, \mathcal{I}) . The previously mentioned condition of boundedness for $\phi(\mathcal{A})$ and the expression given for its adjoint are made evident by the representation (2.52).

Complex powers of a strictly positive, self-adjoint operator.

In the rest of this paragraph we consider a self-adjoint operator $\mathcal{A} : \text{Dom}(\mathcal{A}) \subset \mathcal{H} \to \mathcal{H}$, assuming

$$\sigma(\mathcal{A}) \subset [\varepsilon, +\infty) \quad \text{for some } \varepsilon > 0 \ . \tag{2.53}$$

Up to Hilbertian isomorphisms, it can be assumed that

$$\mathcal{H} = L^2(K,\mu) , \qquad (2.54)$$

and that \mathcal{A} is the multiplication operator by a measurable function (⁷)

$$w: K \to [\varepsilon, +\infty) ; \qquad (2.55)$$

so,

Dom
$$(\mathcal{A}) = \{ f \in L^2(K,\mu) \mid wf \in L^2(K,\mu) \} , \qquad \mathcal{A}f = wf .$$
 (2.56)

Now, let $s \in \mathbb{C}$ and consider the operator \mathcal{A}^{-s} , which can be defined applying the standard functional calculus with $\phi(\lambda) := \lambda^{-s}$.

In the realization (2.54-2.56) of \mathcal{H} and \mathcal{A} , assumed as a standard in the sequel, for any $s \in \mathbb{C}$ one has

$$Dom(\mathcal{A}^{-s}) = \{ f \in L^2(K,\mu) \mid w^{-s}f \in L^2(K,\mu) \} , \qquad \mathcal{A}^{-s}f = w^{-s}f .$$
 (2.57)

If $\Re s \ge 0$, one has $|w^{-s}| = w^{-\Re s} \le \varepsilon^{-\Re s}$; so,

$$\Re s \ge 0 \qquad \Rightarrow \qquad \mathcal{A}^{-s} \in \mathfrak{B}(\mathcal{H}) \quad \text{and} \quad \|\mathcal{A}^{-s}\|_{\mathfrak{B}} \le \varepsilon^{-\Re s} .$$
 (2.58)

One readily checks that $\mathcal{A}^0 = \mathbb{I}$ (with \mathbb{I} the identity operator on \mathcal{H}), that $\mathcal{A}^1 = \mathcal{A}$ and that \mathcal{A}^{-1} coincides with the inverse of the injective operator \mathcal{A} ; for $n \in \mathbb{N}$ one has

$$\mathcal{A}^{n} = \underbrace{\mathcal{A} \dots \mathcal{A}}_{n \text{ times}}, \qquad \mathcal{A}^{-n} = \underbrace{\mathcal{A}^{-1} \dots \mathcal{A}^{-1}}_{n \text{ times}}.$$
(2.59)

Trace class and Hilbert-Schmidt operators.

Let $C \in \mathfrak{B}(\mathcal{H})$ be any nonnegative self-adjoint operator (so that, in particular, $\langle f | Cf \rangle \geq 0$ for all $f \in \mathcal{H}$). The trace of C is

$$\operatorname{Tr} \mathcal{C} := \sum_{n \in \mathbb{N}} \langle u_n | \mathcal{C} u_n \rangle \in [0, +\infty] , \qquad (2.60)$$

where $(u_n)_{n\in\mathbb{N}}$ is any orthonormal basis of \mathcal{H} ; the trace is in fact independent of the basis.

⁷Indeed, the assumptions made on $\sigma(\mathcal{A})$ ensure that \mathcal{A} can be realized as the multiplication by a real function w such that $w(k) \geq \varepsilon$ for a.e. $k \in K$. Redefining w on a set of measure zero, if necessary, we obtain $w(k) \geq \varepsilon$ for all k.
Given any $C \in \mathfrak{B}(\mathcal{H})$, we can associate to it the nonnegative self-adjoint operator $|\mathcal{C}| := \sqrt{C^{\dagger}C}$ (⁸). As well known, C is said to be of *trace class* if $\operatorname{Tr} |\mathcal{C}| < +\infty$; in this case we can define

$$\operatorname{Tr} \mathcal{C} := \sum_{n \in \mathbb{N}} \langle u_n | \mathcal{C} u_n \rangle \in \mathbb{C} , \qquad (2.61)$$

where $(u_n)_{n\in\mathbb{N}}$ is any orthonormal basis of \mathcal{H} (the above series is shown to converge, with a sum independent of the basis). In the sequel, we indicate with $\mathfrak{B}_1(\mathcal{H})$ the set of trace class operators on \mathcal{H} , which is found to be a two-sided ideal of $\mathfrak{B}(\mathcal{H})$.

Let us also recall that an operator $\mathcal{B} \in \mathfrak{B}(\mathcal{H})$ is said to be *Hilbert-Schmidt* if $\operatorname{Tr}(\mathcal{B}^{\dagger}\mathcal{B}) < +\infty$; the set $\mathcal{B}_{2}(\mathcal{H})$ of these operators is a two-sided ideal of $\mathfrak{B}(\mathcal{H})$, and $\mathcal{B}_{2}(\mathcal{H}) \supset \mathcal{B}_{1}(\mathcal{H})$. To go on we assume the Hilbert space to be $\mathcal{H} = L^{2}(K, \mu)$, with μ a σ -finite, separable measure on K. Then, as well known (see, e.g., [119]), $\mathfrak{B}_{2}(\mathcal{H})$ can be identified isomorphically with the Hilbert space $L^{2}(K \times K, \mu \otimes \mu)$; in particular, this means that for any $\mathcal{B} \in \mathfrak{B}_{2}(\mathcal{H})$ there exists a unique *integral kernel* $\mathcal{B}(\cdot, \cdot) \in L^{2}(\Omega \times \Omega)$ such that, for any $f \in \mathcal{H}$,

$$(\mathcal{B}f)(h) = \int_{K} d\mu(k) \ \mathcal{B}(h,k) f(k) \qquad \text{(for μ-a.e. $k \in K$)}.$$
(2.62)

Moreover, for any given orthonormal basis $(u_n)_{n \in \mathbb{N}}$ of \mathcal{H} (recall that \mathcal{H} is assumed to be separable), there holds the L^2 -convergent expansion

$$\mathcal{B}(h,k) = \sum_{n,m\in\mathbb{N}} \langle u_m | \mathcal{B} u_n \rangle \ u_m(h) \ \overline{u_n(k)} \qquad \text{(for μ-a.e. $h, k \in K$)} .$$
(2.63)

2.4 Conjugations and Hilbert spaces

Generalities

Let us consider a complex vector space \mathcal{H} ; a *conjugation* on \mathcal{H} is an antilinear, involutive map $\mathcal{J} : \mathcal{H} \to \mathcal{H}$; so, for all $\alpha, \beta \in \mathbb{C}$ and for all $f, g \in \mathcal{H}$, there hold

$$\mathcal{J}(\alpha f + \beta g) = \overline{\alpha} \, \mathcal{J}f + \overline{\beta} \, \mathcal{J}g \,, \qquad \mathcal{J}^2 f = f \,, \qquad (2.64)$$

(where $\bar{}$ indicates the usual complex conjugation). Given the conjugation \mathcal{J} , we can introduce the sets

$$\mathcal{H}_{\pm} := \{ f \in \mathcal{H} \mid \mathcal{J}f = \pm f \} ; \qquad (2.65)$$

for obvious reasons, the elements of \mathcal{H}_+ and \mathcal{H}_- are called, respectively, the \mathcal{J} -real and the \mathcal{J} -imaginary vectors of \mathcal{H} . One readily checks that \mathcal{H}_{\pm} are real vector subspaces of \mathcal{H} , and that

$$\mathcal{H}_{\mp} = i \,\mathcal{H}_{\pm} \; ; \tag{2.66}$$

moreover, one finds that

$$\mathcal{H} = \mathcal{H}_+ \oplus \mathcal{H}_- \tag{2.67}$$

⁸This is, the power of exponent 1/2 of the nonnegative self-adjoint operator $\mathcal{C}^{\dagger}\mathcal{C}$.

(direct sum of real vector spaces) and that the projections $\mathcal{P}_{\pm} : \mathcal{H} \to \mathcal{H}_{\pm}$ corresponding to the above decomposition are given by

$$\mathcal{P}_{+} := \frac{\mathbb{I} + \mathcal{J}}{2} , \qquad \mathcal{P}_{-} := \frac{\mathbb{I} - \mathcal{J}}{2}$$
(2.68)

(of course $\mathcal{P}_{\pm}\mathcal{P}_{\mp} = \mathbb{O}$ and $\mathcal{P}_{+} + \mathcal{P}_{-} = \mathbb{I}$, with \mathbb{O} and \mathbb{I} indicating the null and the identity operators on \mathcal{H} , respectively). Let us also mention that, for all $f \in \mathcal{H}$, there hold the following identities:

$$\mathcal{P}_{\pm}(if) = i \mathcal{P}_{\mp}f , \qquad f = \mathcal{P}_{+}f + i \mathcal{P}_{+}(-if) . \qquad (2.69)$$

Now, let us consider a (\mathbb{C} -linear) operator \mathcal{B} : Dom(\mathcal{B}) $\subset \mathcal{H} \to \mathcal{H}$, with domain on a (complex) vector suspace; \mathcal{B} is said to be \mathcal{J} -real if [138, 155]

$$\mathcal{J}\mathrm{Dom}(\mathcal{B}) \subset \mathrm{Dom}(\mathcal{B})$$
 and $\mathcal{J}\mathcal{B}f = \mathcal{B}\mathcal{J}f$ for all $f \in \mathrm{Dom}(\mathcal{B})$. (2.70)

Since \mathcal{J} is in particular an involution, the second relation in Eq. (2.70) can be restated as follows, for all $f \in \text{Dom}(\mathcal{B})$:

$$\mathcal{JB}\mathcal{J}^{-1}f = \mathcal{B}f \ . \tag{2.71}$$

To conclude, we stipulate the following: a conjugation on a complex *topological* vector space \mathcal{H} ; is a conjugation on \mathcal{H} as a vector space, which is also a homeomorphism in the given topology. In this case, the subspaces \mathcal{H}_{\pm} defined as before are closed subsets of \mathcal{H} .

Coniugations on a Hilbert space.

A conjugation on a complex *Hilbert* space \mathcal{H} (equipped with inner product $\langle | \rangle$) is a conjugation \mathcal{J} in the vector space \mathcal{H} with the additional property

$$\langle \mathcal{J}f|\mathcal{J}g\rangle = \overline{\langle f|g\rangle} \tag{2.72}$$

for all $f, g \in \mathcal{H}$; this indicates that \mathcal{J} is an antiunitary operator. One easily checks that condition (2.72) has the equivalent formulation

$$\langle \mathcal{J}f|g\rangle = \overline{\langle f|\mathcal{J}g\rangle} ,$$
 (2.73)

sometimes used in the sequel $(^{9})$.

When a conjugation \mathcal{J} is given on the Hilbert space \mathcal{H} , we can then introduce the real vector subspaces \mathcal{H}_{\pm} , defined as in the previous subsection, and check that (2.72) implies $\langle f|g \rangle \in \mathbb{R}$ if f, g are both in \mathcal{H}_+ or both in \mathcal{H}_- .

To go on, let us consider a self-adjoint operator $\mathcal{A} : \text{Dom}(\mathcal{A}) \subset \mathcal{H} \to \mathcal{H}$ and recall that (for \mathcal{H} separable) the spectral theorem ensures the existence of a Hilbertian isomorphism $\mathcal{I} : \mathcal{H} \to L^2(K, \mu)$ such that $\mathcal{I}\mathcal{A}\mathcal{I}^{-1}$ is the multiplication operator by a measurable

⁹Applying Eq. (2.72) with g replaced by $\mathcal{J}g$, and using $\mathcal{J}^2g = g$, one infers (2.73). One uses similar arguments to infer (2.72) from (2.73)

function $w : K \to \mathbb{R}$. In addition let us assume that \mathcal{A} is \mathcal{J} -real, in the sense of the previous subsection (see Eq. (2.70)); then, the system (K, μ, w, \mathcal{I}) mentioned before can be chosen so that

$$\mathcal{I}\mathcal{J}\mathcal{I}^{-1} = \begin{tabular}{c} & (2.74) \end{tabular}$$

where $: L^2(K,\mu) \to L^2(K,\mu), f \mapsto \overline{f}$ is the usual pointwise complex conjugation (¹⁰). Let us also mention the relation

$$\mathcal{J}\phi(\mathcal{A})\,\mathcal{J}^{-1} = \overline{\phi}(\mathcal{A}) \tag{2.75}$$

holding for each measurable function $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$: this is self-evident if we represent \mathcal{A} as a multiplication operator using a system (K, μ, w, \mathcal{I}) with the property (2.74).

2.5 Scale of Hilbert spaces associated to a positive self-adjoint operator.

Let us consider an abstract Hilbert space $(\mathcal{H}, \langle | \rangle) \equiv \mathcal{H}$ and let $\mathcal{A} : \text{Dom}(\mathcal{A}) \subset \mathcal{H} \to \mathcal{H}$ be a self-adjoint operator on it with spectrum as in Eq. (2.53) (i.e., $\sigma(\mathcal{A}) \subset [\varepsilon, +\infty)$ for some $\varepsilon > 0$). Unless otherwise stated, \mathcal{H} is endowed with no further structure.

In the present section we assume that, when a normed space X is a linear subspace of another normed space Y, their completions are realized so that \overline{X} is a linear subspace of \overline{Y} (such a realization is always possible).

Finite order spaces.

Hereafter we introduce a family of spaces, associated to the real powers of \mathcal{A} , i.e., to the operators \mathcal{A}^r for $r \in \mathbb{R}$.

Proposition 2.4. For any $r \in \mathbb{R}$, the following statements hold. i) Define $\langle | \rangle_r : Dom(\mathcal{A}^{r/2}) \times Dom(\mathcal{A}^{r/2}) \to \mathbb{C}$ and $|| ||_r : Dom(\mathcal{A}^{r/2}) \to [0, +\infty)$ by

$$\langle g|f\rangle_r := \langle \mathcal{A}^{r/2}g|\mathcal{A}^{r/2}f\rangle , \qquad \|f\|_r^2 := \|\mathcal{A}^{r/2}f\|^2 .$$
 (2.76)

¹⁰This statement is derived reconsidering the standard proof of the "multiplication operator form" for the spectral theorem on self-adjoint operators, as given in [129]; one must perform a rather simple adaptation of the argument to the case where \mathcal{H} carries a conjugation \mathcal{J} and the operator \mathcal{A} is \mathcal{J} -real. The main points in this adaptation are the following:

i) the spectral measure P of \mathcal{A} is as well \mathcal{J} -real: $\mathcal{J}P(M)\mathcal{J}^{-1} = P(M)$ for each Borel subset of the spectrum of \mathcal{A} . Due to this, for each measurable function $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ one has $\mathcal{J}\phi(\mathcal{A})\mathcal{J}^{-1} = \overline{\phi}(\mathcal{A})$ where $\overline{\phi}$ is the usual complex conjugate;

ii) In the standard proof of the "multiplication operator" theorem, one lets the functions of \mathcal{A} act on a maximal family of cyclic vectors of \mathcal{H} , whose existence is proved via the Zorn lemma. In the present adaptation one must use a maximal family of \mathcal{J} -real cyclic vectors, whose existence is again established by Zornication.

Then, $\langle | \rangle_r$ is an inner product and $|| ||_r$ is the induced norm; moreover, in the realization (2.54-2.56) of \mathcal{H} and \mathcal{A} , one has

$$\langle g|f\rangle_r = \int_K w^r \,\overline{g} \, f \, d\mu \,, \qquad \|f\|_r^2 = \int_K w^r \, |f|^2 \, d\mu \,.$$
 (2.77)

ii) Let us denote with \mathcal{H}^r the Hilbert space obtained completing $(Dom(\mathcal{A}^{r/2}), \langle | \rangle_r)$. If \mathcal{H} and \mathcal{A} are realized as in (2.54-2.56), this Hilbert space can be represented as

$$\mathcal{H}^{r} = \{ f \in Mis(K,\mu) \mid w^{r/2}f \in L^{2}(K,\mu) \} ; \qquad (2.78)$$

its inner product and norm, denoted again with $\langle | \rangle_r$ and $|| ||_r$, can be expressed as in Eq. (2.77) for all $f, g \in \mathcal{H}^r$. In other terms

$$\mathcal{H}^r = L^2(K, w^r d\mu) . \tag{2.79}$$

iii) There holds

 $\mathcal{H}^r = Dom(\mathcal{A}^{r/2}) \qquad for \ r \in \mathbb{R}, \ r \ge 0 \ .$ (2.80)

iv) Consider the linear subspace $Dom(e^{\mathcal{A}}) \subset \mathcal{H}$ that, in the realization (2.54-2.56) of \mathcal{H} and \mathcal{A} , is given by

$$Dom(e^{\mathcal{A}}) = \{ f \in L^2(K,\mu) \mid e^w f \in L^2(K,\mu) \} \subset \mathcal{H} .$$

$$(2.81)$$

For all $r \in \mathbb{R}$ there holds

$$Dom(e^{\mathcal{A}}) \subset Dom(\mathcal{A}^{r/2}) \subset \mathcal{H}^r$$
, (2.82)

and $Dom(e^{\mathcal{A}})$ is dense in $(\mathcal{H}^r, || ||_r)$.

Proof. Point i) is obvious. Now, provisionally regard \mathcal{H}^r to be defined by Eq. (2.78), and intend $\langle | \rangle_r : \mathcal{H}^r \times \mathcal{H}^r \to \mathbb{C}$ to be defined as in Eq. (2.77); due to (2.79), it is evident that \mathcal{H}^r is a Hilbert space with this inner product and that the norm induced by this inner product, denoted with $|| ||_r$, can be expressed as in Eq. (2.77). As well, it appears that $\mathcal{H}^r \supset \text{Dom}(\mathcal{A}^{r/2})$ and that the inner product $\langle | \rangle_r$ of \mathcal{H}^r extends the inner product on $\text{Dom}(\mathcal{A}^{r/2})$ of item i). In the sequel we will prove in order points iii), iv) and finally ii). iii) For any $f \in \mathcal{H}^r$ $(r \geq 0)$ one has

$$+\infty > \int_{K} w^{r} |f|^{2} d\mu \geq \varepsilon^{r} \int_{\mathcal{H}} |f|^{2} d\mu ,$$

whence $f \in L^2(K, \mu)$. By comparison with the representation of Eq. (2.57), one concludes that $f \in \text{Dom}(\mathcal{A}^{r/2})$, so that iii) follows.

iv) For any $r \in \mathbb{R}$, there exists a constant $C_r > 0$ such that $x^r \leq C_r e^x$ for all $x \in \mathbb{R}$ such that $x \geq \varepsilon$; this implies $w^r \leq C_r e^w$ so that, for any $f \in \text{Dom}(e^{\mathcal{A}})$, one has

$$\int_K w^r |f|^2 d\mu \leq C_r \int_K e^w |f|^2 d\mu < +\infty ,$$

2.5. SCALE OF HILBERT SPACES ASSOCIATED TO A POSITIVE SELF-ADJOINT OPERATOR.

and the inclusions in Eq. (2.82) follow.

Let us now pass to prove the density of $\text{Dom}(e^{\mathcal{A}})$ in \mathcal{H}^r , choosing arbitrarily $f \in \mathcal{H}^r$ and constructing a sequence $(f_n)_{n \in \mathbb{N}}$ in $\text{Dom}(e^{\mathcal{A}})$ such that $\|f - f_n\|_r \to 0$ for $n \to +\infty$. Define

$$f_n := (\chi_{[0,n]} \circ w)^{1/2} f$$
,

where $\chi_{[0,n]} : \mathbb{R} \to \mathbb{R}$ is the indicator function of the interval [0,n] (¹¹); we will prove that

$$f_n \in \text{Dom}(e^{\mathcal{A}}) \quad \text{for all } n \in \mathbb{N} .$$
 (2.83)

Let us first show that $f_n \in L^2(K,\mu)$ for all $n \in \mathbb{N}$. In fact, for $r \geq 0$ it is $\mathcal{H}^r = \text{Dom}(\mathcal{A}^{r/2}) \subset L^2(K,\mu)$; thus $f \in L^2(K,\mu)$, and the obvious inequality $|f_n| \leq |f|$ implies $f_n \in L^2(K,\mu)$. For r < 0, from the definition of f_n it follows that $|f_n| \leq (w/n)^{r/2} |f|$, so that

$$\int_{K} |f_{n}|^{2} d\mu \leq \frac{1}{n^{r}} \int_{K} w^{r} |f|^{2} d\mu = \frac{1}{n^{r}} ||f||_{r}^{2} < +\infty ;$$

thus, $f_n \in L^2(K,\mu)$ for all $n \in \mathbb{N}$. Now note that $e^w |f_n| \leq e^n |f_n|$, whence

$$\int_{K} e^{2w} |f_n|^2 \, d\mu \, \le \, e^{2n} \int_{K} |f_n|^2 \, d\mu \, < \, +\infty \, ,$$

so that Eq. (2.83) is proven.

Let us pass to evaluate the quantity

$$||f - f_n||_r^2 = \int_K w^r |f - f_n|^2 d\mu$$
.

By construction, we have $f_n(k) \to f(k)$ for a.e. $k \in K$ in the limit $n \to +\infty$, whence $w^r |f - f_n|^2 \to 0$ μ -almost everywhere. Moreover $w^r |f - f_n|^2 = w^r (1 - (\chi_{[0,n]} \circ w)^{1/2})^2 |f|^2 \leq w^r |f|^2 \in L^1(K,\mu)$ so, by Lebesgue's theorem on dominated convergence (see, e.g., [47], page 417, Theorem 2.1),

$$\int_{K} w^{r} |f - f_{n}|^{2} d\mu \to 0 \quad \text{for } n \to +\infty .$$

Summing up, we have proven that $Dom(e^{\mathcal{A}})$ is dense in \mathcal{H}^r for any $r \in \mathbb{R}$.

ii) For any $r \in \mathbb{R}$, to state that \mathcal{H}^r is the completion of $\text{Dom}(\mathcal{A}^{r/2})$, one has to prove that $\text{Dom}(\mathcal{A}^{r/2})$ is dense in \mathcal{H}^r . For $r \geq 0$ this holds trivially, since $\text{Dom}(\mathcal{A}^{r/2}) = \mathcal{H}^r$ by point iii); for r < 0 the thesis follows from the density of $\text{Dom}(e^{\mathcal{A}}) \subset \text{Dom}(\mathcal{A}^{r/2})$ in \mathcal{H}^r (see point iv)).

$$\chi_J : \mathbb{R} \to \{0, 1\}, \qquad t \mapsto \chi_J(t) := \begin{cases} 1 & \text{for } t \in J \\ 0 & \text{for } t \notin J \end{cases}$$

¹¹For any subset $J \subset \mathbb{R}$, the indicator function is

Remark 2.3. In the sequel we will keep the notations \mathcal{H}^r , $\langle | \rangle_r$, and so on for the spaces considered in the previous proposition. Of course, since $\mathcal{A}^0 = \mathbb{I}$ we have $\mathcal{H}^0 = \mathcal{H}$ and $\langle | \rangle_0 = \langle | \rangle$.

Proposition 2.5. Let $r, r' \in \mathbb{R}$ with $r' \geq r$; then the following statements hold, showing in particular that $\mathcal{H}^{r'} \stackrel{dense}{\hookrightarrow} \mathcal{H}^{r}$.

i) $Dom(\mathcal{A}^{r'/2})$ (resp. $\mathcal{H}^{r'})$ is a linear subspace of $Dom(\mathcal{A}^{r/2})$ (resp., $\mathcal{H}^{r})$ and

$$||f||_r \le \varepsilon^{-(r'-r)/2} ||f||_{r'} \quad for \ all \ f \in \mathcal{H}^{r'};$$
 (2.84)

so, the continuous embedding $\mathcal{H}^{r'} \hookrightarrow \mathcal{H}^r$ holds. ii) $\mathcal{H}^{r'}$ is dense in \mathcal{H}^r .

Proof. i) All the above statements follow easily using representations of the form (2.57) for $\text{Dom}(\mathcal{A}^{r'/2})$, $\text{Dom}(\mathcal{A}^{r/2})$, (2.78) for $\mathcal{H}^{r'}$, \mathcal{H}^{r} , and (2.77) for $\| \|_{r'}$, $\| \|_{r}$; it must be taken into account that $w^{r} = w^{r'}/w^{r'-r} \leq w^{r'}/\varepsilon^{r'-r}$, as well.

ii) It has already been proved that $\text{Dom}(e^{\mathcal{A}}) \subset \mathcal{H}^{r'}$ is dense in \mathcal{H}^r ; of course, this suffices to grant the density of $\mathcal{H}^{r'}$ in \mathcal{H}^r . \Box

Proposition 2.6. Let $r_0, r_1 \in \mathbb{R}$, $\theta \in (0, 1)$, and consider the interpolation space $[\mathcal{H}^{r_0}, \mathcal{H}^{r_1}]_{\theta}$; then

$$[\mathcal{H}^{r_0}, \mathcal{H}^{r_1}]_{\theta} = \mathcal{H}^{(1-\theta)r_0 + \theta r_1} .$$

$$(2.85)$$

Proof. It suffices to use the realization $\mathcal{H}^r = L^2(K, w^r d\mu)$ of Eq. (2.79), for each $r \in \mathbb{R}$, and to recall the identity in Eq. (2.36).

Infinite order spaces.

Let us now pass to discuss two natural spaces: the first one is contained in all the spaces \mathcal{H}^r $(r \in \mathbb{R})$ and the second one contains them all.

Definition 2.7. We put

$$\mathcal{H}^{+\infty} := \bigcap_{r \in \mathbb{R}} \mathcal{H}^r , \qquad (2.86)$$

and equip this linear space with the locally convex topology $\mathcal{T}^{+\infty}$ induced by the family of norms $\| \|_r$ $(r \in \mathbb{R})$.

Remark 2.4. i) The topological space $(\mathcal{H}^{+\infty}, \mathcal{T}^{+\infty})$ is complete, due to the completeness of each space \mathcal{H}^r in the corresponding norm topology induced by $\| \|_r$.

ii) For any $r \in \mathbb{R}$, there exists an $n \in \mathbb{N}$ such that $n \geq r$ so that, by Proposition 2.5, $\| \|_r \leq \varepsilon^{-(n-r)/2} \| \|_n$; therefore, the topology induced on $\mathcal{H}^{+\infty}$ by the family $(\| \|_r)_{r \in \mathbb{R}}$ coincides with the topology induced by the countable subfamily $(\| \|_n)_{n \in \mathbb{N}}$. In conclusion, $\mathcal{T}^{+\infty}$ is a Fréchet topology.

Proposition 2.8. For each $r \in \mathbb{R}$ there holds $\mathcal{H}^{+\infty} \stackrel{dense}{\hookrightarrow} \mathcal{H}^r$.

Proof. By construction, it is $\mathcal{H}^{+\infty} \hookrightarrow \mathcal{H}^r$. The density follows from the fact that $\text{Dom}(e^{\mathcal{A}})$ is contained in $\mathcal{H}^{+\infty}$ and dense in \mathcal{H}^r .

Definition 2.9. We put

$$\mathcal{H}^{-\infty} := \bigcup_{r \in \mathbb{R}} \mathcal{H}^r , \qquad (2.87)$$

and equip this linear space with the inductive limit topology $\mathcal{T}^{-\infty}$ corresponding to the family of normed subspaces $(\mathcal{H}^r, || \parallel_r)$ $(r \in \mathbb{R})$.

Remark 2.5. i) For a general definition of inductive limit topologies, see e.g. [42]. In few words, $\mathcal{T}^{-\infty}$ is the finest locally convex topology on $\mathcal{H}^{-\infty}$ such that $\mathcal{H}^r \hookrightarrow \mathcal{H}^{-\infty}$ for all $r \in \mathbb{R}$.

For any given locally convex space $(\mathcal{Y}, \mathcal{T}_{\mathcal{Y}})$, a linear operator $\mathcal{B} : \mathcal{H}^{-\infty} \to \mathcal{Y}$ is continuous in the topologies $\mathcal{T}^{-\infty}$ and $\mathcal{T}_{\mathcal{Y}}$ if and only if the restriction $\mathcal{B} \upharpoonright \mathcal{H}^r$ is continuous in the topologies \mathcal{T}^r and $\mathcal{T}_{\mathcal{Y}}$ for each $r \in \mathbb{R}$ (see, again, [42]).

ii) Let \mathbb{L} denote any subset of \mathbb{R} such that $\inf \mathbb{L} = -\infty$ (e.g., $\mathbb{L} := \{-1, -2, -3, ...\}$). Then, for each $r \in \mathbb{R}$, there is an $\ell \in \mathbb{L}$ such that $\mathcal{H}^r \hookrightarrow \mathcal{H}^\ell$ (just take any ℓ such that $r \geq \ell$); of course, it also holds that for each $\ell \in \mathbb{L}$ there is $r \in \mathbb{R}$ such that $\mathcal{H}^\ell \hookrightarrow \mathcal{H}^r$ (just take $r = \ell$). Due to these facts we have

$$\mathcal{H}^{-\infty} = \bigcup_{\ell \in \mathbb{L}} \mathcal{H}^{\ell} \tag{2.88}$$

and the topology $\mathcal{T}^{-\infty}$ coincides with the inductive limit topology on $\mathcal{H}^{-\infty}$ given by the family of normed subspaces $(\mathcal{H}^{\ell}, \| \|_{\ell})$ $(\ell \in \mathbb{L})$ (¹²).

Proposition 2.10. $(\mathcal{H}^{-\infty}, \mathcal{T}^{-\infty})$ is a Hausdorff space.

Proof. Choose representations of \mathcal{H} and \mathcal{A} as in Eq.s (2.54-2.56) such that the measure space (K, μ) satisfies $\mu(K) < +\infty$ (we already mentioned that such a representation always exists); then

$$\mathcal{H}^{-\infty} = \{ f \in \operatorname{Mis}(K) \mid w^{r/2} f \in L^2(K, \mu) \text{ for some } r \in \mathbb{R} \} .$$
(2.89)

Now, for each measurable subset $M \subset K$, let us put

$$\alpha_M : \mathcal{H}^{-\infty} \to \mathbb{C} , \qquad f \mapsto \langle \alpha_M, f \rangle := \int_M e^{-w} f \, d\mu .$$

The integral defining $\langle \alpha_M, f \rangle$ exists for any $f \in \mathcal{H}^{-\infty}$; in fact, if $r \in \mathbb{R}$ is such that $w^{r/2}f \in L^2(K,\mu)$, it follows that

$$\begin{split} &\int_{M} |e^{-w}f| \, d\mu = \int_{M} e^{-w} w^{-r/2} \, w^{r/2} |f| \, d\mu \leq \\ &\sqrt{\int_{M} e^{-2w} w^{-r} \, d\mu} \, \sqrt{\int_{M} w^{r} |f|^{2} \, d\mu} \leq \sqrt{C_{r} \, \mu(M)} \, \sqrt{\int_{K} w^{r} |f|^{2} \, d\mu} < +\infty \ , \end{split}$$

¹²Concerning the last statement, see e.g. [42], page 118, Proposition 5.8.

where we have put $C_r := \sup_{\lambda \in [\epsilon, +\infty)} e^{-2\lambda} \lambda^{-r}$ (so that $e^{-2w} w^{-r} \leq C_r$). Clearly, α_M is a linear form on $\mathcal{H}^{-\infty}$; moreover, for each $r \in \mathbb{R}$, the previous manipulations also imply $|\langle \alpha_M, f \rangle| \leq \sqrt{C_r \,\mu(M)} \, ||f||_r$ for all $f \in \mathcal{H}^r$, so that $\alpha_M \upharpoonright \mathcal{H}^r$ is continuous. In conclusion, α_M is a continuous linear form on $(\mathcal{H}^{-\infty}, \mathcal{T}^{-\infty})$ (see the comments in Remark 2.5). Now observe that, for each $f \in \mathcal{H}^{-\infty}$,

$$\langle \alpha_M, f \rangle = 0$$
 for any measurable subset $M \subset K \Rightarrow$
 $\Rightarrow e^{-w}f = 0$ a.e. in $K \Rightarrow f = 0$ a.e. in K .

To go on, let $f, g \in \mathcal{H}^{-\infty}$ and $f \neq g$; then there is a measurable subset $M \subset K$ such that $\delta := |\langle \alpha_M, f - g \rangle| > 0$ (otherwise we would have f - g = 0). Put $\mathcal{U}_f := \{f' \in \mathcal{H}^{-\infty} | |\langle \alpha_M, f' - f \rangle| < \delta/3 \}$ and $\mathcal{U}_g := \{g' \in \mathcal{H}^{-\infty} | |\langle \alpha_M, g' - g \rangle| < \delta/3 \}$; then, it appears that \mathcal{U}_f and \mathcal{U}_g are open subsets of $\mathcal{H}^{-\infty}$ containing f and g, respectively, and $\mathcal{U}_f \cap \mathcal{U}_g = \emptyset$.

Remark 2.6. The spaces \mathcal{H}^r $(r \in \mathbb{R})$ and $\mathcal{H}^{+\infty}$ are also Hausdorff for obvious reasons; this fact follows trivially from Proposition 2.10 since $\mathcal{H}^{+\infty} \subset \mathcal{H}^r \subset \mathcal{H}^{-\infty}$ and any subset of a Hausdorff space is itself Hausdorff (see, e.g., [26], page 77). Summing up, \mathcal{H}^r is a Hausdorff space for any $r \in [-\infty, +\infty]$.

Proposition 2.11. Each one of the sets $Dom(e^{\mathcal{A}})$, $Dom(\mathcal{A}^{r/2})$ and \mathcal{H}^r $(r \in \mathbb{R})$ is dense in $(\mathcal{H}^{-\infty}, \mathcal{T}^{-\infty})$.

Proof. Since the chain of inclusions $\text{Dom}(e^{\mathcal{A}}) \subset \text{Dom}(\mathcal{A}^{r/2}) \subset \mathcal{H}^r$ holds, it suffices to prove the density of $\text{Dom}(e^{\mathcal{A}})$ in $(\mathcal{H}^{-\infty}, \mathcal{T}^{-\infty})$; to this purpose, let us consider any $f \in \mathcal{H}^{-\infty}$ and prove that there exists a sequence $(f_n)_{n\in\mathbb{N}}$ of elements of $\text{Dom}(e^{\mathcal{A}})$ such that $f_n \to f$ for $n \to +\infty$ in $(\mathcal{H}^{-\infty}, \mathcal{T}^{-\infty})$. Indeed, let r be such that $f \in \mathcal{H}^r$; then, due to Proposition 2.4, there is a sequence $(f_n)_{n\in\mathbb{N}}$ in $\text{Dom}(e^{\mathcal{A}})$ such that $f_n \to f$ in $(\mathcal{H}^r, || \parallel_r)$ for $n \to +\infty$. From here and from $\mathcal{H}^r \hookrightarrow \mathcal{H}^{-\infty}$ it follows that $f_n \to f$ in $(\mathcal{H}^{-\infty}, \mathcal{T}^{-\infty})$ for $n \to +\infty$. \Box

Proposition 2.12. Let $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ be a measurable function and assume that, for some $b \in \mathbb{R}$,

$$\sup_{\lambda \in \sigma(\mathcal{A})} \lambda^b |\phi(\lambda)| < +\infty .$$
(2.90)

Then, the following statements hold:

i) The operator $\phi(\mathcal{A}) : Dom(\phi(\mathcal{A})) \subset \mathcal{H} \to \mathcal{H}$ has a unique continuous linear extension, denoted with the same symbol, $\phi(\mathcal{A}) : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$.

2.5. SCALE OF HILBERT SPACES ASSOCIATED TO A POSITIVE SELF-ADJOINT OPERATOR.

ii) Let $r, r' \in \mathbb{R}$ be such that $r' - r \geq -2b$. Then, the extension $\phi(\mathcal{A})$ sends continuously $\mathcal{H}^{r'}$ into \mathcal{H}^r ; moreover, for all $f \in \mathcal{H}^{r'}$ one has (¹³)

$$\|\phi(\mathcal{A})f\|_{r} \le Q_{rr'} \|f\|_{r'} , \qquad Q_{rr'} := \sup_{\lambda \in \sigma(\mathcal{A})} \lambda^{-(r'-r)/2} |\phi(\lambda)| < +\infty .$$
 (2.91)

iii) In particular assume that, for some $b \in \mathbb{R}$, it is

$$|\phi(\lambda)| = \lambda^{-b} \qquad \text{for all } \lambda \in \sigma(\mathcal{A}) ; \qquad (2.92)$$

then the extension $\phi(\mathcal{A})$ of item i) is a Hilbertian isomorphism between \mathcal{H}^r and \mathcal{H}^{r+2b} for each $r \in \mathbb{R}$.

Proof. i) ii) First of all, let us prove that the mentioned continuous extension is unique, assuming it to exist. In fact, consider two distinct continuous extensions $\phi(\mathcal{A}), \phi'(\mathcal{A}) : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$. By construction, $\phi(\mathcal{A})$ and $\phi'(\mathcal{A})$ coincide on the set $\text{Dom}(\phi(\mathcal{A}))$ which is dense in \mathcal{H} and, consequently, in $\mathcal{H}^{-\infty}$ (see Proposition 2.11); so $\phi(\mathcal{A}) = \phi'(\mathcal{A})$ everywhere on $\mathcal{H}^{-\infty}$ by continuity.

In order to prove the existence of the continuous linear extension, represent $\mathcal{H}^{-\infty}$ as in Eq. (2.89) and define

$$\phi(\mathcal{A}): \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty} , \qquad f \mapsto \phi(\mathcal{A})f := (\phi \circ w)f$$
(2.93)

(recalling that $w(k) \in \sigma(\mathcal{A}) = \text{Dom}(\phi)$ for a.e. $k \in K$, so that $\phi \circ w$ is defined almost everywhere in K). Let us prove that the map (2.93) is well defined. to this purpose, let $f \in \mathcal{H}^{-\infty}$; then $f \in \mathcal{H}^{r'}$ for some $r' \in \mathbb{R}$. If we consider any $r \in \mathbb{R}$ such that $r' - r \geq -2b$, writing (almost everywhere in K)

$$w^{r/2} \phi(\mathcal{A}) f = w^{-(r'-r)/2} (\phi \circ w) w^{r'/2} f$$
,

we infer

$$w^{r/2} |\phi(\mathcal{A})f| \le Q_{rr'} w^{r'/2} |f| ,$$
 (2.94)

with $Q_{r'r}$ as in Eq. (2.91). From Eq. (2.94) and from $w^{r'/2}f \in L^2(K,\mu)$ it follows that $w^{r/2}\phi(\mathcal{A})f \in L^2(K,\mu)$, which amounts to state that $\phi(\mathcal{A})f \in \mathcal{H}^r \hookrightarrow \mathcal{H}^{-\infty}$.

Summing up, we have shown that the map in Eq. (2.93) actually sends $\mathcal{H}^{-\infty}$ into $\mathcal{H}^{-\infty}$; the linearity of this map is evident. The previous argument also indicates that $\phi(\mathcal{A})\mathcal{H}^{r'} \subset \mathcal{H}^r$ for $r' - r \geq -2b$; moreover, the inequality in Eq. (2.94) can be used to infer the norm inequality in Eq. (2.91). This also implies the continuity of $\phi(\mathcal{A})$ as a map $\mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$. iii) The thesis is almost evident; the main point in the proof is the (obvious) identity $w^{r/2+b}|\phi(\mathcal{A})f| = w^{r/2}f$.

$$\begin{split} \lambda^{-(r'-r)/2} |\phi(\lambda)| &= \varepsilon^{-(r'-r)/2} (\lambda/\varepsilon)^{-(r'-r)/2} |\phi(\lambda)| \le \\ &\le \varepsilon^{-(r'-r)/2} (\lambda/\varepsilon)^b |\phi(\lambda)| = \varepsilon^{-(r'-r+2b)/2} \lambda^b |\phi(\lambda)| \;, \end{split}$$

and Eq. (2.90) gives the thesis.

¹³Note that the supremum defining $Q_{rr'}$ is finite. In fact, for any $\lambda \in \sigma(\mathcal{A}) \subset [\varepsilon, +\infty)$, we have

Corollary 2.13. *i)* For any $s \in \mathbb{C}$, the operator $\mathcal{A}^{-s} : Dom(\mathcal{A}^{-s}) \subset \mathcal{H}^{-\infty} \to \mathcal{H} \subset \mathcal{H}^{-\infty}$ has a unique, continuous linear extension (indicated with the same symbol) $\mathcal{A}^{-s} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$.

ii) The extension of item *i*) is such that, for any $r \in \mathbb{R}$,

$$\mathcal{A}^{-s}(\mathcal{H}^r) = \mathcal{H}^{r+2\Re s} , \qquad \|\mathcal{A}^{-s}f\|_{r+2\Re s} = \|f\|_r \quad \text{for all } f \in \mathcal{H}^r .$$

iii) Concerning the continuous extensions to $\mathcal{H}^{-\infty}$, the following statements hold: $\mathcal{A}^{0} = \mathbb{I}_{\mathcal{H}^{-\infty}}, \ \mathcal{A}^{-(s+s')} = \mathcal{A}^{-s}\mathcal{A}^{-s'}$ for any $s, s' \in \mathbb{C}$, and \mathcal{A}^{s} is the inverse operator of \mathcal{A}^{-s} for any $s \in \mathbb{C}$.

Proof. i) ii) The thesis follows by statements i) and iii) of Proposition 2.12. Indeed, for any $s \in \mathbb{C}$, the map $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$, $\lambda \mapsto \lambda^{-s}$ is measurable and fulfills Eq. (2.92) for $b = \Re s$.

iii) All these statements follow easily noting that, for any $s \in \mathbb{C}$, the linear continuous extension $\mathcal{A}^{-s} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ in item i) is defined according to Eq. (2.93), so that $\mathcal{A}^{-s}f = w^{-s}f$ for all $f \in \mathcal{H}^{-\infty}$.

Corollary 2.14. i) For any $\mathbf{t} \in \mathbb{C}$ with $\Re \mathbf{t} \geq 0$ consider the operators $e^{-\mathbf{t}\mathcal{A}}$, $e^{-\mathbf{t}\sqrt{\mathcal{A}}}$, $(e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})$ mapping the respective domains $Dom(e^{-\mathbf{t}\mathcal{A}})$, $Dom(e^{-\mathbf{t}\sqrt{\mathcal{A}}})$, $Dom(e^{-\mathbf{t}\sqrt{\mathcal{A}}}) \subset \mathcal{H} \subset \mathcal{H}^{-\infty}$ into $\mathcal{H} \subset \mathcal{H}^{-\infty}$. Each of them admits a unique continuous linear extension (indicated with the same symbol)

$$e^{-\mathbf{t}\mathcal{A}}, e^{-\mathbf{t}\sqrt{\mathcal{A}}}, \frac{e^{-\mathbf{t}\sqrt{\mathcal{A}}}}{\sqrt{\mathcal{A}}} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$$
 (2.96)

ii) Assume $\Re t > 0$ and consider the extensions of item i); these are such that, for any $r, r' \in \mathbb{R}$, and for all $f \in \mathcal{H}^{r'}$, one has

$$\|e^{-\mathbf{t}\mathcal{A}}f\|_{r} \leq \begin{cases} \left(\frac{r-r'}{2e\,\Re\mathbf{t}}\right)^{(r-r')/2} \|f\|_{r'} & \text{for } 0 < \Re\mathbf{t} \leq \frac{r-r'}{2\varepsilon} \\ \varepsilon^{(r-r')/2} e^{-\varepsilon\,\Re\mathbf{t}} \|f\|_{r'} & \text{for } \Re\mathbf{t} > \frac{r-r'}{2\varepsilon} \end{cases},$$
(2.97)

$$\|e^{-\mathbf{t}\sqrt{\mathcal{A}}}f\|_{r} \leq \begin{cases} \left(\frac{r-r'}{e\,\Re\mathbf{t}}\right)^{r-r'} \|f\|_{r'} & \text{for } 0 < \Re\mathbf{t} \leq \frac{r-r'}{\sqrt{\varepsilon}} \\ \varepsilon^{(r-r')/2} e^{-\sqrt{\varepsilon}\,\Re\mathbf{t}} \|f\|_{r'} & \text{for } \Re\mathbf{t} > \frac{r-r'}{\sqrt{\varepsilon}} \end{cases},$$
(2.98)

$$\|\left(e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}\right)f\|_{r} \leq \begin{cases} \left(\frac{r-r'-1}{e\,\Re\mathbf{t}}\right)^{r-r'-1} \|f\|_{r'} & \text{for } 0<\Re\mathbf{t}\leq\frac{r-r'-1}{\sqrt{\varepsilon}}\\ \varepsilon^{(r-r'-1)/2} e^{-\sqrt{\varepsilon}\,\Re\mathbf{t}} \|f\|_{r'} & \text{for } \Re\mathbf{t}>\frac{r-r'-1}{\sqrt{\varepsilon}} \end{cases}$$
(2.99)

Moreover, each of the extensions $e^{-t\mathcal{A}}$, $e^{-t\sqrt{\mathcal{A}}}$, $(e^{-t\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})$ maps continuously $\mathcal{H}^{-\infty}$ into $\mathcal{H}^{+\infty}$.

iii) Assume $\Re \mathbf{t} = 0$, so that $\mathbf{t} = it$ for some $t \in \mathbb{R}$. Then the extensions $e^{-it\mathcal{A}}$, $e^{-it\sqrt{\mathcal{A}}}$ of item i) are Hilbertian automorphism of \mathcal{H}^r , for each $r \in \mathbb{R}$; in particular, there holds

$$\|e^{-it\mathcal{A}}f\|_{r} = \|e^{-it\sqrt{\mathcal{A}}}f\|_{r} = \|f\|_{r} .$$
(2.100)

2.5. SCALE OF HILBERT SPACES ASSOCIATED TO A POSITIVE SELF-ADJOINT OPERATOR.

Moreover, the extension $e^{-it\sqrt{A}}/\sqrt{A}$ is an Hilbertian isomorphism of \mathcal{H}^r into \mathcal{H}^{r+1} , for any $r \in \mathbb{R}$; in particular,

$$\|(e^{-it\sqrt{A}}/\sqrt{A})f\|_{r} = \|f\|_{r+1} .$$
(2.101)

Proof. Hereafter we report the proof only for the operator $e^{-t\mathcal{A}}$; the proofs of the analogous statements for the other two operators $e^{-t\sqrt{\mathcal{A}}}$, $e^{-t\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}$ can be derived by similar arguments.

i) ii) The thesis follows by items i) and ii) of Proposition 2.12. Indeed, for any $t \in \mathbb{C}$ with $\Re \mathbf{t} > 0$, the map $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$, $\lambda \mapsto e^{-\lambda \mathbf{t}}$ is measurable and fulfills Eq. (2.91) with

$$Q_{rr'} = \sup_{\lambda \in \sigma(\mathcal{A})} \lambda^{-(r'-r)/2} e^{-\lambda \Re \mathbf{t}} = \begin{cases} \left(\frac{r-r'}{2e\Re \mathbf{t}}\right)^{(r-r')/2} & \text{for } 0 < \Re \mathbf{t} \le \frac{r-r'}{2\varepsilon} \\ \varepsilon^{(r-r')/2} e^{-\varepsilon \Re t} & \text{for } \Re \mathbf{t} > \frac{r-r'}{2\varepsilon} \end{cases}$$

Moreover, for all $f \in \mathcal{H}^{-\infty}$, there exists an $r' \in \mathbb{R}$ such that $f \in \mathcal{H}^{r'}$; then, by Item ii), for any $\mathbf{t} \in \mathbb{C}$ with $\Re \mathbf{t} > 0$, one has $e^{-\mathbf{t}\mathcal{A}}f \in \mathcal{H}^r$ for all $r \in \mathbb{R}$; thus $e^{-\mathbf{t}\mathcal{A}}f \in \mathcal{H}^{+\infty}$, that is the thesis. The continuity of $e^{-\mathbf{t}\mathcal{A}}$ from $\mathcal{H}^{-\infty}$ to $\mathcal{H}^{+\infty}$ is proved by similar arguments. iv) Consider the elementary identities $|e^{-it\lambda}| = |e^{-it\sqrt{\lambda}}| = 1$ and $|e^{-it\sqrt{\lambda}}/\sqrt{\lambda}| = \lambda^{-1/2}$, for $\lambda \in \sigma(\mathcal{A})$; then, the thesis follows straightforwardly from item iii) of Proposition 2.12. \Box

Duality relations.

In the present paragraph we discuss duality relations holding for both the finite $(\mathcal{H}^r, r \in \mathbb{R})$ and infinite order $(\mathcal{H}^{\pm \infty})$ spaces considered in the two previous subsections.

Definition 2.15. We consider the set

$$\mathcal{H}^{(2)} := \bigcup_{r \in \mathbb{R}} \mathcal{H}^{-r} \times \mathcal{H}^r .$$
(2.102)

Remark 2.7. Note that the space $\mathcal{H}^{(2)}$ contains, in particular, $\mathcal{H} \times \mathcal{H} = \mathcal{H}^0 \times \mathcal{H}^0$.

Proposition 2.16. The inner product $\langle | \rangle$ of \mathcal{H} has a unique extension

$$\langle | \rangle : \mathcal{H}^{(2)} \to \mathbb{C}$$
 (2.103)

that is continuous when restricted to any of the products $\mathcal{H}^{-r} \times \mathcal{H}^r$ $(r \in \mathbb{R})$ (¹⁴). For each $r \in \mathbb{R}$, the map (2.103) restricted to $\mathcal{H}^{-r} \times \mathcal{H}^r$ is a sesquilinear Hermitian form; moreover, for all $f \in \mathcal{H}^{-r}$ and all $g \in \mathcal{H}^r$, it fulfills

$$|\langle f|g\rangle| \le ||f||_{-r} ||g||_r .$$
(2.104)

¹⁴Of course, $\mathcal{H}^{-r} \times \mathcal{H}^{r}$ $(r \in \mathbb{R})$ is equipped with the product topology.

Proof. Let us first prove that the extension $\langle | \rangle$ with the above requirements, if it exists, is unique; to this purpose, it suffices to show that any extension with the above properties is uniquely determined on any product $\mathcal{H}^{-r} \times \mathcal{H}^r$ $(r \in \mathbb{R})$. In fact, if $r \geq 0$, $\mathcal{H}^r \times \mathcal{H}^r$ is dense in $\mathcal{H}^{-r} \times \mathcal{H}^r$ and the extension must agree with the usual inner product of \mathcal{H} on $\mathcal{H}^r \times \mathcal{H}^r$ $(\subset \mathcal{H} \times \mathcal{H})$; similar arguments hold if r < 0 considering the space $\mathcal{H}^{-r} \times \mathcal{H}^{-r}$ in place of $\mathcal{H}^r \times \mathcal{H}^r$.

To prove existence of the extension (2.103), consider the realization of \mathcal{H} and \mathcal{A} given in Eq.s (2.54-2.56) and define

$$\langle f|g\rangle := \int_{K} \overline{f}g \, d\mu \quad \text{for all } (f,g) \in \mathcal{H}^{(2)} .$$
 (2.105)

The above integral exists; in fact, if $r \in \mathbb{R}$ is such that $(f, g) \in \mathcal{H}^{-r} \times \mathcal{H}^{r}$, we have

$$\int_{K} |\overline{f}g| \, d\mu = \int_{K} w^{-r/2} |f| \, w^{r/2} |g| \, d\mu \le$$

$$\leq \sqrt{\int_{K} w^{-r} |f|^2 \, d\mu} \sqrt{\int_{K} w^{r} |g|^2 \, d\mu} = \|f\|_{-r} \|g\|_{r} < +\infty \,. \tag{2.106}$$

It is clear that (2.105) defines an extension of the inner product on $\mathcal{H} \times \mathcal{H}$. For all $r \in \mathbb{R}$, the map (2.105) is a sesquilinear Hermitian form on $\mathcal{H}^{-r} \times \mathcal{H}^{r}$; moreover, due to the estimate in Eq. (2.106), it fulfills the inequality (2.104), which implies its continuity. \Box

In the sequel we keep the notation $\langle | \rangle$ for the extension (2.103) of the inner product on \mathcal{H} . In the forthcoming Proposition 2.18 this extension is used to prove that \mathcal{H}^{-r} can be identified isomorphically with the dual of the Banach space \mathcal{H}^r ; in the subsequent Proposition 2.20 it is shown that $\mathcal{H}^{-\infty}$ can be identified with the dual of the Fréchet space $\mathcal{H}^{+\infty}$, as well.

Definition 2.17. For $r \in \mathbb{R}$, we define the map

$$\mathcal{I}^r : \mathcal{H}^{-r} \to (\mathcal{H}^r)', \quad f \mapsto \mathcal{I}^r f \quad \text{such that} \quad \langle \mathcal{I}^r f, g \rangle := \langle f | g \rangle \text{ for all } g \in \mathcal{H}^r.$$
 (2.107)

Proposition 2.18. The following statements hold. i) The map \mathcal{I}^r in Eq. (2.107) is well defined, antilinear, bijective and fulfills

$$\|\mathcal{I}^r f\|'_r = \|f\|_{-r} \qquad \text{for all } f \in \mathcal{H}^{-r} .$$
 (2.108)

(and thus, summing up, it is an antilinear isomorphism of Banach spaces). ii) Let $r, r' \in \mathbb{R}$ with $r' \geq r$, so that $\mathcal{H}^{r'} \hookrightarrow \mathcal{H}^r$ and $\mathcal{H}^{-r} \hookrightarrow \mathcal{H}^{-r'}$; then

$$\mathcal{I}^{r'}f = \mathcal{I}^r f \upharpoonright \mathcal{H}^{r'} \qquad for \ all \ f \in \mathcal{H}^{-r} \,. \tag{2.109}$$

2.5. SCALE OF HILBERT SPACES ASSOCIATED TO A POSITIVE SELF-ADJOINT OPERATOR.

Proof. i) Fix $r \in \mathbb{R}$; due to Proposition 2.16 it appears that, for each $f \in \mathcal{H}^{-r}$, the element $\mathcal{I}^r f$ defined in Eq. (2.107) is a continuous linear form on \mathcal{H}^r , with the norm bound (recall Eq. (2.104))

$$\|\mathcal{I}^r f\|'_r \le \|f\|_{-r} . \tag{2.110}$$

It is evident as well that \mathcal{I}^r is a continuous antilinear map from \mathcal{H}^{-r} to $(\mathcal{H}^r)'$. In the sequel we use again the representations (2.54-2.56) for \mathcal{H} and \mathcal{A} , so that

$$\langle \mathcal{I}^r f, g \rangle = \int_K \overline{f} g \, d\mu \qquad \text{for all } f \in \mathcal{H}^{-r}, \, g \in \mathcal{H}^r \,.$$
 (2.111)

Let us prove Eq. (2.108), for a given $f \in \mathcal{H}^{-r}$. To this purpose, put $g = w^{-r}f$; then $\int_{K} w^{r} |g|^{2} d\mu = \int_{K} w^{-r} |f|^{2} d\mu$, which implies $g \in \mathcal{H}^{r}$ and $||g||_{r} = ||f||_{-r}$. For this g it is $\langle \mathcal{I}^{r}f, g \rangle = \int_{K} w^{-r} |f|^{2} d\mu = ||f||_{-r} ||f||_{-r} = ||f||_{-r} ||g||_{r}$, whence $||\mathcal{I}^{r}f||_{r} \geq ||f||_{-r}$. From here and from Eq. (2.110) the thesis (2.108) follows.

To conclude, let us prove that \mathcal{I}^r is bijective between \mathcal{H}^{-r} and \mathcal{H}^r . Due to Eq. (2.108), \mathcal{I}^r has zero kernel, so it is injective. To prove surjectivity, let $\alpha \in (\mathcal{H}^r)'$; since \mathcal{H}^r is a Hilbert space with the inner product $\langle | \rangle_r$, by the Riesz theorem there exists an $h \in \mathcal{H}^r$ such that

$$\langle \alpha, g \rangle = \langle h | g \rangle_r = \int_K w^r \,\overline{h} \, g \, d\mu \qquad \text{for all } g \in \mathcal{H}^r \,.$$
 (2.112)

Put $f := w^r h$; then $\int_K w^{-r} |f|^2 d\mu = \int_K w^r |h|^2 d\mu < +\infty$, so that $f \in \mathcal{H}^{-r}$. Moreover, Eq. (2.112) implies $\langle \alpha, g \rangle = \int_K \overline{f} g d\mu = \langle \mathcal{I}^r f, g \rangle$ for all $g \in \mathcal{H}^r$; this means $\alpha = \mathcal{I}^r f$, thus the surjectivity of \mathcal{I}^r is proved.

ii) The thesis follows immediately from the definitions of \mathcal{I}^r and $\mathcal{I}^{r'}$.

Consider now $(\mathcal{H}^{+\infty})'$, the topological dual of the space $\mathcal{H}^{+\infty}$. Recall that the topology on the Fréchet space $\mathcal{H}^{+\infty}$ is induced by the family of norms $\| \|_r \upharpoonright \mathcal{H}^r$ $(r \in \mathbb{R})$; so, a linear form $\alpha : \mathcal{H}^{+\infty} \to \mathbb{C}$ is continuous if and only if there exist $r \in \mathbb{R}$ and $C_r \in [0, +\infty)$ such that $|\langle \alpha, f \rangle| \leq C_r |\|f\|_r$ for all $f \in \mathcal{H}^{+\infty}$. This condition is equivalent to state that α is the restriction of a continuous linear form on \mathcal{H}^r ; this form on \mathcal{H}^r is unique, due to the density of $\mathcal{H}^{+\infty}$ in \mathcal{H}^r . Therefore, by identifing the elements of $(\mathcal{H}^{+\infty})'$ with their restrictions to \mathcal{H}^r , one has

$$(\mathcal{H}^{+\infty})' = \bigcup_{r \in \mathbb{R}} (\mathcal{H}^r)' .$$
(2.113)

Definition 2.19. From now on, $(\mathcal{H}^{+\infty})'$ will be equipped with the inductive limit topology corresponding to the family of Banach spaces $((\mathcal{H}^r)', || ||'_r)$.

Remark 2.8. The above topology is the finest locally convex topology on $(\mathcal{H}^{+\infty})'$ that makes continuous the injection of each space $((\mathcal{H}^r)', || \parallel'_r)$ in $(\mathcal{H}^{+\infty})'$.

The following statement is easily proved using, where necessary, the already known facts on the Banach anti-isomorphisms $\mathcal{I}^r : \mathcal{H}^{-r} \to (\mathcal{H}^r)'$.

Proposition 2.20. $\mathcal{H}^{-\infty} \times \mathcal{H}^{+\infty}$ is contained in the set $\mathcal{H}^{(2)}$ defined in Eq. (2.102); let us put

$$\mathcal{I}^{\infty}: \mathcal{H}^{-\infty} \to (\mathcal{H}^{+\infty})', \quad f \mapsto \mathcal{I}^{\infty}f \quad such \ that \quad \langle \mathcal{I}^{\infty}f, g \rangle := \langle f|g \rangle \quad for \ all \ g \in \mathcal{H}^{+\infty}.$$
(2.114)

The map \mathcal{I}^{∞} is well defined, antilinear and bijective; it is a homeomorphism with respect to the inductive limit topologies of $\mathcal{H}^{-\infty}$ and $(\mathcal{H}^{+\infty})'$ introduced in the Definitions 2.9 and 2.19. For any $r \in \mathbb{R}$ and any $f \in \mathcal{H}^{-r}$, there holds

$$\mathcal{I}^{\infty}f = \mathcal{I}^r f \upharpoonright \mathcal{H}^{+\infty} . \tag{2.115}$$

Proof. Let $(f,g) \in \mathcal{H}^{-\infty} \times \mathcal{H}^{+\infty}$; then $f \in \mathcal{H}^{-r}$ for some $r \in [0, +\infty)$. Since $g \in \mathcal{H}^r$, we have $(f,g) \in \mathcal{H}^{-r} \times \mathcal{H}^r \subset \mathcal{H}^{(2)}$; this proves the inclusion $\mathcal{H}^{-\infty} \times \mathcal{H}^{+\infty} \subset \mathcal{H}^{(2)}$.

Now, let $f \in \mathcal{H}^{-\infty}$, and define $\mathcal{I}^{\infty}f : \mathcal{H}^r \to \mathbb{C}$ following (2.114). Clearly, $\mathcal{I}^{\infty}f$ is a linear map. If $r \in \mathbb{R}$ is such that $f \in \mathcal{H}^{-r}$, it is evident that $\mathcal{I}^{\infty}f = \mathcal{I}^rf \upharpoonright \mathcal{H}^{+\infty}$, and this restriction is continuous on $\mathcal{H}^{+\infty}$ since \mathcal{I}^rf is continuous on \mathcal{H}^r and $\mathcal{H}^{+\infty} \hookrightarrow \mathcal{H}^r$.

In conclusion $\mathcal{I}^{\infty}f \in (\mathcal{H}^{+\infty})'$ and we have a well defined map $\mathcal{I}^{\infty} : \mathcal{H}^{-\infty} \to (\mathcal{H}^{+\infty})'$, $f \to \mathcal{I}^{\infty}f$. The antilinearity of this map is self-evident; hereafter we show that \mathcal{I}^{∞} is bijective.

Indeed, let $f, f' \in \mathcal{H}^{-\infty}$ be such that $\mathcal{I}^{\infty}f = \mathcal{I}^{\infty}f'$. There exists $r \in \mathbb{R}$ such that $f, f' \in \mathcal{H}^{-r}$; this implies $\mathcal{I}^r f \upharpoonright \mathcal{H}^{+\infty} = \mathcal{I}^r f' \upharpoonright \mathcal{H}^{+\infty}$ which gives (by the density of $\mathcal{H}^{+\infty}$ in \mathcal{H}^r) $\mathcal{I}^r f = \mathcal{I}^r f'$. Thus, we infer that f = f', which proves the injectivity of \mathcal{I}^{∞} .

To prove surjectivity, let us take $\alpha \in (\mathcal{H}^{+\infty})'$ and show that $\alpha = \mathcal{I}^{\infty} f$ for some $f \in \mathcal{H}^{-\infty}$. Indeed, there are $r \in \mathbb{R}$ and $\alpha_r \in (\mathcal{H}^r)'$ such that $\alpha = \alpha_r \upharpoonright \mathcal{H}^{+\infty}$; if $f \in \mathcal{H}^{-r}$ is such that $\alpha_r = \mathcal{I}^r f$, we have $\mathcal{I}^{\infty} f = \mathcal{I}^r f \upharpoonright \mathcal{H}^{+\infty} = \alpha$.

To infer that \mathcal{I}^{∞} is in fact a homeomorphism (i.e., continuous with continuous inverse), one can use the continuity of the maps $\mathcal{I}^r : \mathcal{H}^{-r} \to (\mathcal{H}^r)' \ (r \in \mathbb{R})$ (see Proposition 2.18), recalling the fundamental properties of the inductive limit topologies on $\mathcal{H}^{-\infty}$ and on $(\mathcal{H}^{+\infty})'$ (¹⁵).

Banach adjoints of continuous operators.

Consider the scale of spaces \mathcal{H}^r $(r \in \mathbb{R})$ discussed previously in this section; even though each of these spaces is in fact a Hilbert space (with inner product $\langle | \rangle_r$; see Proposition 2.4), in the present subsection we shall only consider the underlying Banach space structure. In particular, hereafter we discuss the notion of adjoint for a continuous operator acting between any two of the Banach spaces \mathcal{H}^r $(r \in \mathbb{R})$.

Recall that for any $r \in \mathbb{R}$ (or $r = +\infty$) the topological dual of the Banach space \mathcal{H}^r , i.e. $(\mathcal{H}^r)'$, is anti-linearly isomorphic \mathcal{H}^{-r} (see Proposition 2.18); moreover, the duality pairing

¹⁵For example, recall that the map $\mathcal{I}^{\infty} : \mathcal{H}^{-\infty} \to (\mathcal{H}^{+\infty})$ is continuous if and only if its restriction $\mathcal{I}^{\infty} \upharpoonright \mathcal{H}^{-r} : \mathcal{H}^{-r} \to (\mathcal{H}^{r})'$ is so, for all $r \in \mathbb{R}$; on the other hand, the continuity of $\mathcal{I}^{\infty} \upharpoonright \mathcal{H}^{-r}$ can be infered noting that $\mathcal{I}^{\infty} \upharpoonright \mathcal{H}^{-r} \equiv \mathcal{I}^{r} : \mathcal{H}^{-r} \to (\mathcal{H}^{r})'$ (which is continuous) and that the embedding $(\mathcal{H}^{r})' \hookrightarrow (\mathcal{H}^{+\infty})'$ is continuous, since $(\mathcal{H}^{+\infty})'$ is endowed with the inductive limit topology.

between these spaces is described by the restriction to $\mathcal{H}^{-r} \times \mathcal{H}^{r}$ of the map $\langle | \rangle : \mathcal{H}^{(2)} \to \mathbb{C}$, introduced in Proposition 2.16. In view of these facts, the notion of Banach adjoint of a continuous operator (see Section 2.2) can be rephrased as follows within the framework under analysis.

Definition 2.21. Let $r, r' \in \mathbb{R}$ and consider a continuous linear operator $\mathcal{B} : \mathcal{H}^r \to \mathcal{H}^{r'}$. The *Banach adjoint* of \mathcal{B} is the unique continuous operator $\mathcal{B}^* : \mathcal{H}^{-r'} \to \mathcal{H}^{-r}$ such that

$$\langle \mathcal{B}^* g | f \rangle = \langle g | \mathcal{B} f \rangle$$
 for all $f \in \mathcal{H}^r, g \in \mathcal{H}^{-r'}$. (2.116)

Remark 2.9. Let us point out a few facts concerning the above definition.

i) The symbol $\langle | \rangle$ on the left-hand side of Eq. (2.116) is used to indicate the restriction of the map (2.103) to $\mathcal{H}^{-r} \times \mathcal{H}^{r}$; on the other hand, the same symbol is used on the right-hand side of the cited equation to denote the restriction of the map (2.103) to $\mathcal{H}^{-r'} \times \mathcal{H}^{r'}$.

ii) Recall that the map $\langle | \rangle : \mathcal{H}^{(2)} \to \mathbb{C}$ of Proposition 2.16 is an extension of the inner product on \mathcal{H} ; thus, given a continuous operator $\mathcal{B} : \mathcal{H} \to \mathcal{H}$ ($\mathcal{H} \equiv \mathcal{H}^0$), the corresponding Banach adjoint $\mathcal{B}^* : \mathcal{H} \to \mathcal{H}$ (defined according to Eq. (2.116)) does in fact coincide with the standard Hilbert adjoint \mathcal{B}^{\dagger} . On the other hand, despite the fact that \mathcal{H}^r and $\mathcal{H}^{r'}$ are also Hilbert spaces, the adjoint of an operator $\mathcal{B} : \mathcal{H}^r \to \mathcal{H}^{r'}$ described above is different from the notion of Hilbert adjoint with respect to the inner products $\langle | \rangle_r$ (of \mathcal{H}^r) and $\langle | \rangle_{r'}$ (of $\mathcal{H}^{r'}$). Within the present manuscript Hilbert adjoints with respect to the products $\langle | \rangle_r$ and $\langle | \rangle_{r'}$ will *never* be considered.

iii) Let $\mathcal{B}: \mathcal{H}^r \to \mathcal{H}^{+\infty}$ be a continuous operator (for some $r \in \mathbb{R}$); due to the results of Proposition 2.20, we can generalize the Definition 2.21 of adjoint operator to include this case. In analogy to Definition 2.21, we say that the adjoint of \mathcal{B} is the unique continuous operator $\mathcal{B}^*: \mathcal{H}^{-\infty} \to \mathcal{H}^{-r}$ fulfilling Eq. (2.116) for all $f \in \mathcal{H}^{-\infty}$ and for all $g \in \mathcal{H}^r$; of course, the expression in the right-hand side of the cited equation must be intended here in terms of the isomorphism $\mathcal{I}^{\infty}: \mathcal{H}^{-\infty} \to (\mathcal{H}^{+\infty})'$ introduced in Eq. (2.114). (¹⁶)

The results on Banach adjoints already mentioned in Section 2.2 can be straightforwardly translated in the present framework; we enumerate them in the subsequent Lemma.

Lemma 2.22. Let $r, r', r'' \in \mathbb{R}$; then, there hold the following results. *i*) The adjoint map

$$^{*}:\mathfrak{B}(\mathcal{H}^{r},\mathcal{H}^{r'})\to\mathfrak{B}(\mathcal{H}^{-r'},\mathcal{H}^{-r}),\qquad \mathcal{B}\mapsto\mathcal{B}^{*}$$
(2.117)

is an isometric anti-isomorphism $(^{17})$.

ii) Any continuous operator $\mathcal{B}: \mathcal{H}^r \to \mathcal{H}^{r'}$ coincides with its double adjoint $\mathcal{B}^{**} \equiv (\mathcal{B}^*)^*: \mathcal{H}^r \to \mathcal{H}^{r'}$, that is

$$\mathcal{B}^{**} = \mathcal{B} \ . \tag{2.118}$$

¹⁶On the contrary, it is not so easy to extend the notion of adjoint operator to the case of a continuous operator $\mathcal{B}: \mathcal{H}^r \to \mathcal{H}^{-\infty}$ $(r \in \mathbb{R})$ within this contex since, in general, there only holds the strict inclusion $(\mathcal{H}^{-\infty})' \supset \mathcal{H}^{+\infty}$ (and not the equality).

¹⁷Recall that, for any pair of Banach spaces X and Y, $\mathfrak{B}(X, Y)$ indicates the Banach space of linear continuous operators from X to Y.

iii) Let $\mathcal{B} : \mathcal{H}^r \to \mathcal{H}^{r'}$ and $\mathcal{C} : \mathcal{H}^{r'} \to \mathcal{H}^{r''}$ be any pair of continuous operators and consider their composition $\mathcal{CB} : \mathcal{H}^r \to \mathcal{H}^{r''}$; this is also a continuous operator and its adjoint $(\mathcal{CB})^* : \mathcal{H}^{-r''} \to \mathcal{H}^{-r}$ fulfills

$$(\mathcal{CB})^* = \mathcal{B}^* \mathcal{C}^* . \tag{2.119}$$

Proof. All the statements in the present Lemma are obtained as trivial reformulations of well-known results on Banach adjoint operators in the framework under analysis. \Box

Proposition 2.23. Let $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ be any measurable function fulfilling the assumption (2.90) for some $b \in \mathbb{R}$; for any pair $r, r' \in \mathbb{R}$ with $r' - r \geq -2b$, consider the continuous extension $\phi(\mathcal{A}) : \mathcal{H}^{r'} \to \mathcal{H}^{r}$ introduced in Proposition 2.12. Then, the corresponding adjoint operator $\phi(\mathcal{A})^* : \mathcal{H}^{-r} \to \mathcal{H}^{-r'}$ is given by

$$\phi(\mathcal{A})^* = \overline{\phi}(\mathcal{A}) \tag{2.120}$$

 $(\overline{\phi}: \sigma(\mathcal{A}) \to \mathbb{C} \text{ denotes the complex conjugate of the function } \phi; \text{ in the right-hand side of } Eq. (2.120) we are considering the extension of } \overline{\phi}(\mathcal{A}) \text{ to a continuous map } \mathcal{H}^{-r} \to \mathcal{H}^{-r'}, \text{ which exists since } (-r) - (-r') = r' - r \geq -2b).$

Remark 2.10. In the case where $\phi(\mathcal{A})$ maps \mathcal{H}^{-r} into \mathcal{H}^r continuously, for some given $r \in \mathbb{R}$ with $r \leq b$, using the previous results with r' = -r we infer

$$\phi(\mathcal{A})^* = \overline{\phi}(\mathcal{A}) \in \mathfrak{B}(\mathcal{H}^{-r}, \mathcal{H}^r) .$$
(2.121)

In particular, if ϕ is real-valued it follows that

$$\phi(\mathcal{A})^* = \phi(\mathcal{A}) ; \qquad (2.122)$$

then, in analogy with the theory of operators on Hilbert spaces, we refer to $\phi(\mathcal{A})$ as a generalized self-adjoint operator.

Proof. Consider the representations of \mathcal{H} and \mathcal{A} given in Eq.s (2.54-2.56); in particular, recall that the sesquilinear form $\langle | \rangle : \mathcal{H}^{-r} \times \mathcal{H}^r \to \mathbb{C}$ (for all $r \in \mathbb{R}$) can be expressed as in Eq. (2.105) and that $\phi(\mathcal{A})$ coincides with the multiplication operator $M_{\phi(w)} : \mathcal{H} \to \mathcal{H}$, $f \mapsto M_{\phi(w)}f := \phi(w)f$ (which also defines a unique continuous extension $\phi(\mathcal{A}) : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ according to the prescription (2.93)). Using the basic identity (2.116) for adjoint operators we infer, for all $g \in \mathcal{H}^{-r}$ and all $f \in \mathcal{H}^{r'}$,

$$\langle \phi(\mathcal{A})^* g | f \rangle = \langle g | \phi(\mathcal{A}) f \rangle = \int_K d\mu \, \overline{g} \, (\phi(w)f) = \int_K d\mu \, \overline{(\overline{\phi}(w)g)} \, f = \langle \overline{\phi}(\mathcal{A})g | f \rangle \; ;$$

due to the arbitrariness of f and g, the above chain of equalities implies the thesis. \Box

Corollary 2.24. There hold the following results.

i) Let $r \in \mathbb{R}$, $s \in \mathbb{C}$ and consider the continuous operator $\mathcal{A}^{-s} : \mathcal{H}^r \to \mathcal{H}^{r+2\Re s}$; its adjoint is the continuous operator

$$(\mathcal{A}^{-s})^* = (\mathcal{A}^{-\bar{s}} \upharpoonright \mathcal{H}^{-(r+2\Re s)}) : \mathcal{H}^{-(r+2\Re s)} \to \mathcal{H}^{-r} .$$
(2.123)

ii) Let $r \in \mathbb{R}$, $\mathbf{t} \in \mathbb{C}$ with $\Re \mathbf{t} > 0$ and consider the restriction of the continuous operator $e^{-\mathbf{t}\mathcal{A}} : \mathcal{H}^{-\infty} \to \mathcal{H}^{+\infty}$ to the space \mathcal{H}^r ; its adjoint is the continuous operator

$$(e^{-\mathbf{t}\mathcal{A}})^* = e^{-\bar{\mathbf{t}}\mathcal{A}} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-r} .$$
(2.124)

If $\Re \mathbf{t} = 0$, so that $\mathbf{t} = it$ for some $t \in \mathbb{R}$, the adjoint of the continuous operator $e^{-it\mathcal{A}}$: $\mathcal{H}^r \to \mathcal{H}^r$ is

$$(e^{-it\mathcal{A}})^* = (e^{it\mathcal{A}} \upharpoonright \mathcal{H}^{-r}) : \mathcal{H}^{-r} \to \mathcal{H}^{-r} .$$
(2.125)

Analogous results hold for the exponential operators $e^{-t\sqrt{A}}$ and $e^{-t\sqrt{A}}/\sqrt{A}$.

Proof. Both items i) and ii) follow straightforwardly from Proposition 2.23; in particular, for ii) recall the considerations in Remarks 2.10. \Box

The case of a Hilbert space with conjugation

Let us now assume that the basic Hilbert space \mathcal{H} is endowed with a *conjugation* \mathcal{J} : $\mathcal{H} \to \mathcal{H}$ (see Section 2.4 and, expecially, the second paragraph therein); moreover, let us assume that the operator \mathcal{A} used in the previous subsections to construct the scale of spaces \mathcal{H}^r ($r \in [-\infty, +\infty]$) is \mathcal{J} -real (so that \mathcal{A} and \mathcal{J} commute).

Proposition 2.25. The map $\mathcal{J} : \mathcal{H} \to \mathcal{H}$ has a unique continuous extension to the topological vector space $\mathcal{H}^{-\infty}$. This extension, denoted from now on with $\mathcal{J} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$, possesses the following properties:

i) \mathcal{J} is a conjugation on the topological vector space $\mathcal{H}^{-\infty}$; ii) $\mathcal{J}(\mathcal{H}^r) = \mathcal{H}^r$ for all $r \in [-\infty, +\infty]$;

iii) for all $f \in \mathcal{H}^{-\infty}$ and $g \in \mathcal{H}^{+\infty}$ one has

$$\langle \mathcal{J}f|\mathcal{J}g\rangle = \overline{\langle f|g\rangle} \tag{2.126}$$

or, equivalently,

$$\langle \mathcal{J}f|g\rangle = \overline{\langle f|\mathcal{J}g\rangle} \tag{2.127}$$

(where $\langle | \rangle$ is the extension (2.103) of the inner product of \mathcal{H}); iv) for each $r \in \mathbb{R}$, $\mathcal{J} \upharpoonright \mathcal{H}^r$ is a conjugation on the Hilbert space \mathcal{H}^r :

$$\langle \mathcal{J}f|\mathcal{J}g\rangle_r = \overline{\langle f|g\rangle_r}$$
 (2.128)

v) let $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ be a measurable function fulfilling the assumption (2.90) for some $b \in \mathbb{R}$ and consider the continuous extension $\phi(\mathcal{A}) : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ introduced in Proposition 2.12 and the analogous map $\overline{\phi}(\mathcal{A}) : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$. Then, the relation

$$\mathcal{J}\phi(\mathcal{A})\,\mathcal{J}^{-1} = \overline{\phi}(\mathcal{A}) \tag{2.129}$$

holds in the space of operators from $\mathcal{H}^{-\infty}$ to $\mathcal{H}^{-\infty}$.

Proof. The continuous extension of \mathcal{J} to $\mathcal{H}^{-\infty}$, if it exists, is unique due to the density of \mathcal{H} in $\mathcal{H}^{-\infty}$. In the sequel we prove that such a continuous extension exists and fulfills statements i)-v).

To this purpose, let us recall the formulation of the spectral theorem given in the second paragraph of Section 2.4 for a \mathcal{J} -real self-adjoint operator; due to this formulation, it suffices to prove existence and i)-v) when $\mathcal{H} = L^2(K,\mu)$, \mathcal{A} is the multiplication operator by measurable function $w : K \to [\varepsilon, +\infty)$ and $\mathcal{J} : L^2(K,\mu) \to L^2(K,\mu)$ is the usual complex conjugation: $\mathcal{J}f = \overline{f}$. In this case, let us define

$$\mathcal{J}f := \overline{f} \qquad \text{for all } f \in \mathcal{H}^{-\infty} \subset \operatorname{Mis}(K) ; \qquad (2.130)$$

then, one readily checks that \mathcal{J} maps $\mathcal{H}^{-\infty}$ to $\mathcal{H}^{-\infty}$ extending continuously the conjugation of $\mathcal{H} = L^2(K, \mu)$, and that all statements i)-v) hold. \Box

Corollary 2.26. Consider the map $\mathcal{J} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ defined by Proposition 2.25 and the operators $\mathcal{A}^{-s}, \mathcal{A}^{-\bar{s}}, e^{\mathbf{t}\mathcal{A}}, e^{-\bar{\mathbf{t}}\mathcal{A}} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ for any $s \in \mathbb{C}$, $\mathbf{t} \in \mathbb{C}$ with $\Re \mathbf{t} \geq 0$; then

$$\mathcal{J}\mathcal{A}^{-s}\mathcal{J}^{-1} = \mathcal{A}^{-\bar{s}} , \qquad (2.131)$$

$$\mathcal{J}e^{-\mathbf{t}\mathcal{A}}\mathcal{J}^{-1} = e^{-\bar{\mathbf{t}}\mathcal{A}} , \qquad (2.132)$$

and analogous results hold for the exponential operators $e^{-t\sqrt{A}}$, $e^{-t\sqrt{A}}/\sqrt{A}$: $\mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$.

Proof. Use item v) of the Proposition 2.25 with $\phi(\lambda) := \lambda^{-s}$, $\phi(\lambda) := e^{-t\lambda}$ and so on. \Box

Further results on the complex powers of \mathcal{A} .

Let us keep all the assumptions and the notations of the previous subsections; in particular, we consider the (extensions of the) powers $\mathcal{A}^{-s} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$, for $s \in \mathbb{C}$. Moreover, for any fixed $r \in \mathbb{R}$, let us define the strip

$$\Sigma_r := \{ s \in \mathbb{C} \mid \Re s > r \} . \tag{2.133}$$

Lemma 2.27. Let $r_1, r_2 \in \mathbb{R}$; for all $s \in \Sigma_{(r_1+r_2)/2}$, the operator \mathcal{A}^{-s} maps continuously \mathcal{H}^{-r_1} into \mathcal{H}^{r_2} .

Proof. Recall that, for any $s \in \mathbb{C}$, \mathcal{A}^{-s} maps continuously \mathcal{H}^{-r_1} to $\mathcal{H}^{-r_1+2\Re s}$ (see Corollary 2.13). Moreover, for any $s \in \Sigma_{(r_1+r_2)/2}$, we have $-r_1 + 2\Re s > r_2$ so that $\mathcal{H}^{-r_1+2\Re s} \hookrightarrow \mathcal{H}^{r_2}$; this suffices to infer the thesis.

Let us recall once more that, for any $r_1, r_2 \in \mathbb{R}$, $\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})$ indicates the Banach space of continuous linear operators from \mathcal{H}^{-r_1} to \mathcal{H}^{r_2} , with the usual operator norm.

Proposition 2.28. For any $r_1, r_2 \in \mathbb{R}$, the following function is holomorphic:

$$\Sigma_{(r_1+r_2)/2} \to \mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2}), \qquad s \mapsto \mathcal{A}^{-s} \upharpoonright \mathcal{H}^{-r_1}.$$
 (2.134)

2.5. SCALE OF HILBERT SPACES ASSOCIATED TO A POSITIVE SELF-ADJOINT OPERATOR.

Proof. Let us fix $s_0 \in \Sigma_{(r_1+r_2)/2}$ arbitrarily. For any $s \in \Sigma_{(r_1+r_2)/2}$, \mathcal{A}^{-s} can be represented as the multiplication operator by the function $w^{-s} : K \to \mathbb{C}$; the latter can be re-expressed, in turn, as follows:

$$w^{-s} = w^{-s_0} w^{-(s-s_0)} = w^{-s_0} e^{-(s-s_0)\ln w} =$$
$$= w^{-s_0} \sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} (-\ln w)^n = \sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} w^{-s_0} (-\ln w)^n$$

Hereafter we will prove the following statements, yielding the thesis.

i) For each $n \in \mathbb{N}$, the operator $\mathcal{A}^{-s_0}(-\ln \mathcal{A})^n$ maps \mathcal{H}^{-r_1} into \mathcal{H}^{r_2} , and $\mathcal{C}_n := \mathcal{A}^{-s_0}(-\ln \mathcal{A})^n \upharpoonright \mathcal{H}^{-r_1}$ is continuous from \mathcal{H}^{-r_1} to \mathcal{H}^{r_2} .

ii) The series $\sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} C_n$ is convergent in $\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})$ for all s inside the disc $D(s_0, \Re s_0 - (r_1 + r_2)/2) \subset \Sigma_{(r_1 + r_2)/2}.$

iii) For all $s \in D(s_0, \Re s_0 - (r_1 + r_2)/2)$, the sum of the series in item ii) equals $\mathcal{A}^{-s} \upharpoonright \mathcal{H}^{-r_1}$. These statements are proved in the following Steps 1, 2 and 3.

Step 1 - Statement i) holds and, for each $n \in \mathbb{N}$, the operator \mathcal{C}_n has norm

$$\|\mathcal{C}_n\|_{\mathfrak{B}(\mathcal{H}^{-r_1},\mathcal{H}^{r_2})} \le \max\left(\varepsilon^{-(\Re s_0 - \frac{r_1 + r_2}{2})} |\ln \varepsilon|^n, \left(\frac{n}{e}\right)^n \left(\Re s_0 - \frac{r_1 + r_2}{2}\right)^{-n}\right).$$
(2.135)

Let us fix $n \in \mathbb{N}$; moreover, let $f \in \mathcal{H}^{-r_1}$ so that $w^{-r_1/2}f \in L^2(K,\mu)$. Then

$$w^{r_2/2} \left[w^{-s_0} (-\ln w)^n \right] f = \left[w^{-(s_0 - \frac{r_1 + r_2}{2})} (-\ln w)^n \right] w^{-r_1/2} f ,$$

so that we have $w^{-s_0}(-\ln w)^n f \in \mathcal{H}^{r_2}$ if the function $w^{-(s_0-\frac{r_1+r_2}{2})}(-\ln w)^n$ is essentially bounded on K. To go on, recall that $\Re s_0 - (r_1+r_2)/2 > 0$ (since $s_0 \in \Sigma_{(r_1+r_2)/2}$) and that $w(k) \geq \varepsilon$ for a.e. $k \in K$; then, analyzing by elementary means the function $F_n: [\varepsilon, +\infty) \to \mathbb{R}, z \mapsto F_n(z) := z^{-(\Re s_0 - \frac{r_1+r_2}{2})} |\ln z|^n$, we obtain

$$\sup_{K} \left| w^{-(s_0 - \frac{r_1 + r_2}{2})} (-\ln w)^n \right| \le \sup_{z \in [\varepsilon, +\infty)} F_n(z) = \max\left(F_n(\varepsilon), \ F_n(e^{n/(\Re s_0 - \frac{r_1 + r_2}{2})}) \right) = \\ = \max\left(\varepsilon^{-(\Re s_0 - \frac{r_1 + r_2}{2})} |\ln \varepsilon|^n, \ \left(\frac{n}{e}\right)^n \left(\Re s_0 - \frac{r_1 + r_2}{2}\right)^{-n} \right) < +\infty .$$

Summing up, we have shown that $\mathcal{A}^{-s_0}(-\ln \mathcal{A})^n f = w^{-s_0}(-\ln w)^n f \in \mathcal{H}^{r_2}$, and that $\mathcal{C}_n := \mathcal{A}^{-s_0}(-\ln \mathcal{A})^n \upharpoonright \mathcal{H}^{-r_1} : \mathcal{H}^{-r_1} \to \mathcal{H}^{r_2}$ fulfills

$$\|\mathcal{C}_n f\|_{r_2} \le \max\left(\varepsilon^{-(\Re s_0 - \frac{r_1 + r_2}{2})} |\ln \varepsilon|^n, \left(\frac{n}{e}\right)^n \left(\Re s_0 - \frac{r_1 + r_2}{2}\right)^{-n}\right) \|f\|_{-r_1},$$

for all $f \in \mathcal{H}^{-r_1}$; so, \mathcal{C}_n is continuous with norm fulfilling Eq. (2.135). Step 2 - Statement ii) holds. The convergence radius of the series $\sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} \mathcal{C}_n$ in the Banach space $\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})$ is $\rho \in [0, +\infty]$, where

$$\frac{1}{\rho} := \limsup_{n \to +\infty} \left(\frac{1}{n!} \| \mathcal{C}_n \|_{\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})} \right)^{1/n}$$

Let us show that $\rho \geq \Re s_0 - (r_1 + r_2)/2$; in fact, due to Eq. (2.135) we have

$$\frac{1}{\rho} \le \max\left(\lim_{n \to +\infty} \left(\frac{1}{n!} \varepsilon^{-(\Re s_0 - \frac{r_1 + r_2}{2})} |\ln \varepsilon|^n\right)^{1/n}, \lim_{n \to +\infty} \left(\frac{1}{n!} \left(\frac{n}{e}\right)^n \left(\Re s_0 - \frac{r_1 + r_2}{2}\right)^{-n}\right)^{1/n}\right) = \frac{1}{\Re s_0 - \frac{r_1 + r_2}{2}}$$

(to compute the above limits, we have used the Stirling formula $n! = (n/e)^n \sqrt{2\pi n} u_n$, with $u_n \to 1$ for $n \to +\infty$).

Step 3 - Proof of statement iii). Let $s \in D(s_0, \Re s_0 - (r_1 + r_2)/2)$; for all $f \in \mathcal{H}^{-r_1}$, the convergence of the series $\sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} C_n$ in the space $\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})$ and the definition of C_n imply the following

$$\left(\sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} \,\mathcal{C}_n\right) f = \sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} \,\mathcal{C}_n f = \sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} \,w^{-s_0} (-\ln w)^n f$$

and the last series converges in \mathcal{H}^{r_2} . On the other hand, it can be checked that the series $\sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} w^{-s_0} (-\ln w)^n$ converges in $L^{\infty}(K,\mu)$ to the function w^{-s} ; in conclusion,

$$\left(\sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} \mathcal{C}_n\right) f = w^{-s} f$$

for all $f \in \mathcal{H}^{-r_1}$, that is, $\sum_{n=0}^{+\infty} \frac{(s-s_0)^n}{n!} \mathcal{C}_n = \mathcal{A}^{-s} \upharpoonright \mathcal{H}^{-r_1}$.

Proposition 2.29. For any $r_1, r_2 \in \mathbb{R}$, the following maps are holomorphic:

$$\Sigma_0 \to \mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2}) , \qquad \mathbf{t} \mapsto e^{-\mathbf{t}\mathcal{A}} \upharpoonright \mathcal{H}^{-r_1}, \ e^{-\mathbf{t}\sqrt{\mathcal{A}}} \upharpoonright \mathcal{H}^{-r_1}, \ (e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}) \upharpoonright \mathcal{H}^{-r_1} .$$
 (2.136)

Proof. We report the proof only for the map $\mathbf{t} \mapsto e^{-\mathbf{t}A}$; the derivation of same results for the other exponential functions is analogous. We move along the same lines as in the proof of the previous Proposition 2.28.

So, let $r_1, r_2 \in \mathbb{R}$ and let us fix $\mathbf{t}_0 \in \Sigma_0$. Recall that, for all $\mathbf{t} \in \Sigma_0$, $e^{-\mathbf{t}\mathcal{A}}$ can be represented as the multiplication operator by the function $e^{-\mathbf{t}w} : K \to \mathbb{C}$; the latter can be re-expressed, in turn, as

$$\begin{split} e^{-\mathbf{t}w} &= e^{-\mathbf{t}_0 w} \, e^{-(\mathbf{t}-\mathbf{t}_0)w} = \\ &= e^{-\mathbf{t}_0 w} \sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \, (-w)^n = \sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \, e^{-\mathbf{t}_0 w} \, (-w)^n \ . \end{split}$$

Hereafter we will prove the following statements, yielding the thesis.

i) For each $n \in \mathbb{N}$ and for any $r_1, r_2 \in \mathbb{R}$, $e^{-\mathbf{t}_0 \mathcal{A}} (-\mathcal{A})^n$ maps \mathcal{H}^{-r_1} into \mathcal{H}^{r_2} and $\mathcal{C}_n := e^{-\mathbf{t}_0 \mathcal{A}} (-\mathcal{A})^n \upharpoonright \mathcal{H}^{-r_1}$ is continuous from \mathcal{H}^{-r_1} to \mathcal{H}^{r_2} .

ii) The series $\sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} C_n$ converges in $\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})$ for all \mathbf{t} belonging to the disc $D(\Re \mathbf{t}_0, \Re \mathbf{t}_0) \subset \Sigma_0$.

iii) The sum of the series in item ii) equals $e^{-\mathbf{t}\mathcal{A}} \upharpoonright \mathcal{H}^{-r_1}$, for all $\mathbf{t} \in D(\Re \mathbf{t}_0, \Re \mathbf{t}_0)$.

These statements are proved in the following Steps 1, 2 and 3.

Step 1 - Statement i) holds and, for each $n \in \mathbb{N}$ and for any $r_1, r_2 \in \mathbb{R}$, the operator \mathcal{C}_n has norm

$$\|\mathcal{C}_n\|_{\mathfrak{B}(\mathcal{H}^{-r_1},\mathcal{H}^{r_2})} \le \max\left(\varepsilon^{n+\frac{r_1+r_2}{2}}e^{-\Re \mathbf{t}_0\varepsilon}, \left(\frac{n+\frac{r_1+r_2}{2}}{e\,\Re \mathbf{t}_0}\right)^{n+\frac{r_1+r_2}{2}}\right) .$$
(2.137)

Let us fix $n \in \mathbb{N}$ and let $f \in \mathcal{H}^{-r_1}$, so that $w^{-r_1/2}f \in L^2(K,\mu)$; then

$$w^{r_2/2} \left[e^{-\mathbf{t}_0 w} (-w^n) \right] f = \left[e^{-\mathbf{t}_0 w} (-1)^n w^{n + \frac{r_1 + r_2}{2}} \right] w^{-r_1/2} f .$$

Therefore, it suffices to show that the function $e^{-\mathbf{t}_0 w} w^{n+\frac{r_1+r_2}{2}}$ is essentially bounded on K to infer that $e^{-\mathbf{t}_0 w}(-w)^n f \in \mathcal{H}^{r_2}$. To this purpose, let us first recall that $\Re \mathbf{t}_0 > 0$ (since $\mathbf{t}_0 \in \Sigma_0$) and that $w(k) \geq \varepsilon$ for a.e. $k \in K$; then, analyzing by elementary means the function $F_n : [\varepsilon, +\infty) \to \mathbb{R}, z \mapsto F_n(z) := e^{-\mathbf{t}_0 z} z^{n+\frac{r_1+r_2}{2}}$, we obtain

$$\operatorname{ess\,sup}_{K} |e^{-\mathbf{t}_{0}w}(-1)^{n}w^{n+\frac{r_{1}+r_{2}}{2}}| \leq \sup_{z \in [\varepsilon, +\infty)} F_{n}(z) = \\ = \max\left(F_{n}(\varepsilon), F_{n}\left(\frac{n+\frac{r_{1}+r_{2}}{2}}{2}\right)\right) = \max\left(e^{-\Re\mathbf{t}_{0}\,\varepsilon}\varepsilon^{n+\frac{r_{1}+r_{2}}{2}}, \left(\frac{n+\frac{r_{1}+r_{2}}{2}}{e\,\Re\mathbf{t}_{0}}\right)^{n+\frac{r_{1}+r_{2}}{2}}\right) < +\infty.$$

As stated above, this implies that $e^{-\mathbf{t}_0 \mathcal{A}} (-\mathcal{A})^n f = e^{-\mathbf{t}_0 w} (-w)^n f \in \mathcal{H}^{r_2}$ which also ensures that the operator $\mathcal{C}_n := e^{-\mathbf{t}_0 \mathcal{A}} (-\mathcal{A})^n \upharpoonright \mathcal{H}^{-r_1}$ maps \mathcal{H}^{-r_1} to \mathcal{H}^{r_2} . Moreover, the arguments employed also give

$$\|\mathcal{C}_n f\|_{r_2} \le \max\left(e^{-\Re \mathbf{t}_0 \varepsilon} \varepsilon^{n + \frac{r_1 + r_2}{2}}, \left(\frac{n + \frac{r_1 + r_2}{2}}{e \,\Re \mathbf{t}_0}\right)^{n + \frac{r_1 + r_2}{2}}\right) \|f\|_{-r_1},$$

for all $f \in \mathcal{H}^{-r_1}$; so, \mathcal{C}_n is continuous with norm fulfilling Eq. (2.137). Step 2 - Statement ii) holds. Indeed, the convergence radius of the series $\sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \mathcal{C}_n$ in the Banach space $\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})$ is $\rho \in [0, +\infty]$, where

$$\frac{1}{\rho} := \limsup_{n \to +\infty} \left(\frac{1}{n!} \| \mathcal{C}_n \|_{\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})} \right)^{1/n}$$

Next note that, due to Eq. (2.137), we have

$$\frac{1}{\rho} \le \max\left(\lim_{n \to +\infty} \left(\frac{1}{n!} e^{-\Re \mathbf{t}_0 \varepsilon} \varepsilon^{n + \frac{r_1 + r_2}{2}}\right)^{1/n}, \lim_{n \to +\infty} \left(\frac{1}{n!} \left(\frac{n + \frac{r_1 + r_2}{2}}{e \, \Re \mathbf{t}_0}\right)^{n + \frac{r_1 + r_2}{2}}\right)^{1/n}\right) = \frac{1}{\Re \mathbf{t}_0}$$

(the above limits have been comptuted using again the Stirling formula $n! = (n/e)^n \sqrt{2\pi n} u_n$, with $u_n \to 1$ for $n \to +\infty$); so, $\rho \ge \Re \mathbf{t}_0$.

Step 3 - Proof of statement iii). Let $\mathbf{t} \in D(\Re \mathbf{t}_0, \Re \mathbf{t}_0)$. For all $f \in \mathcal{H}^{-r_1}$, the convergence of the series $\sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \mathcal{C}_n$ in the space $\mathfrak{B}(\mathcal{H}^{-r_1}, \mathcal{H}^{r_2})$ and the definition of \mathcal{C}_n imply the following

$$\left(\sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \,\mathcal{C}_n\right) f = \sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \,\mathcal{C}_n f = \sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \,e^{-\mathbf{t}_0 w} (-w)^n f \;;$$

the last series converges in \mathcal{H}^{r_2} . On the other hand, it can be checked that the series $\sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} e^{-\mathbf{t}_0 w} (-w)^n$ converges in $L^{\infty}(K,\mu)$ to the function $e^{-\mathbf{t}w}$; summing up, we have shown that

$$\left(\sum_{n=0}^{+\infty} \frac{(\mathbf{t} - \mathbf{t}_0)^n}{n!} \mathcal{C}_n\right) f = e^{-\mathbf{t}w} f$$

$$\sum_{n=0}^{\infty} \frac{(\mathbf{t} - \mathbf{t}_0)^n}{n!} \mathcal{C}_n = e^{-\mathbf{t}\mathcal{A}} \upharpoonright \mathcal{H}^{-r_1}.$$

for all $f \in \mathcal{H}^{-r_1}$, that is, $\sum_{n=0}^{+\infty} \frac{(\mathbf{t}-\mathbf{t}_0)^n}{n!} \mathcal{C}_n = e^{-\mathbf{t}\mathcal{A}} \upharpoonright \mathcal{H}^{-r_1}$.

Proposition 2.30. For any $r \in \mathbb{R}$ and for any $n \in \mathbb{N}$, the following maps are of class C^n :

$$\mathbb{R} \to \mathfrak{B}(\mathcal{H}^{r+2(n+1)}, \mathcal{H}^{r}) , \qquad t \mapsto e^{-it\mathcal{A}} \upharpoonright \mathcal{H}^{r+2(n+1)} ,$$
$$\mathbb{R} \to \mathfrak{B}(\mathcal{H}^{r+n+1}, \mathcal{H}^{r}) , \qquad t \mapsto e^{-it\sqrt{\mathcal{A}}} \upharpoonright \mathcal{H}^{r+n+1} , \qquad (2.138)$$
$$\mathbb{R} \to \mathfrak{B}(\mathcal{H}^{r+n}, \mathcal{H}^{r}) , \qquad t \mapsto (e^{-it\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}) \upharpoonright \mathcal{H}^{r+n} .$$

Proof. As an example, we consider the map $t \mapsto e^{-it\mathcal{A}}$ and show the existence and continuity of its first derivative (arguments similar to those presented hereafter can be employed to discuss the higher order derivatives). Let us stress that, in view of the identity $e^{-it\mathcal{A}} = e^{-it_0\mathcal{A}}e^{-i(t-t_0)\mathcal{A}}$ and of the fact that $e^{-it_0\mathcal{A}}$ is an Hilbertian isomorphism on \mathcal{H}^r (for any $t_0 \in \mathbb{R}$ and for any $r \in \mathbb{R}$), it suffices to discuss the differentiability in t = 0. Therefore, to infer the existence of $\frac{d}{dt}e^{-it\mathcal{A}}$, it suffices to show that

$$\frac{e^{-it\mathcal{A}} - 1 + it\mathcal{A}}{t} \to 0 \qquad \text{in } \mathfrak{B}(\mathcal{H}^{r+4}, \mathcal{H}^r) \text{ for } t \to 0.$$
(2.139)

Notice that, by definition, there holds

$$\left\|\frac{e^{-it\mathcal{A}} - 1 + it\mathcal{A}}{t}\right\|_{\mathfrak{B}(\mathcal{H}^{r+4},\mathcal{H}^r)} = \sup_{f \in \mathcal{H}^{r+4}} \frac{\left\|\frac{e^{-it\mathcal{A}} - 1 + it\mathcal{A}}{t}f\right\|_r}{\|f\|_{r+4}};$$

moreover, using representations of \mathcal{H} and \mathcal{A} as in Eq.s (2.54-2.56), for any $f \in \mathcal{H}^{r+2(n+1)}$ we have

$$\left\|\frac{e^{-it\mathcal{A}} - 1 + it\mathcal{A}}{t}f\right\|_{r}^{2} = \int_{K} d\mu \ w^{r} \left|\frac{e^{-itw} - 1 + iwt}{t}f\right|^{2} = \int_{K} d\mu \ w^{r} \ \frac{2}{t^{2}} \left(1 - \cos(wt) - tw\sin(wt) + \frac{1}{2}w^{2}t^{2}\right)|f|^{2}$$

2.5. SCALE OF HILBERT SPACES ASSOCIATED TO A POSITIVE SELF-ADJOINT OPERATOR.

On the other hand, it can be proved by elementary means that $\frac{2}{t^2}(1-\cos(w t)-t w \sin(w t)+\frac{1}{2}w^2t^2) \leq \frac{1}{4}w^4t^2$ for |t| small enough; this allows to infer

$$\left\| \frac{e^{-it\mathcal{A}} - 1 + it\mathcal{A}}{t} f \right\|_{r}^{2} \le \frac{t^{2}}{4} \int_{K} d\mu \ w^{r+4} \ |f|^{2} = \frac{t^{2}}{4} \ \|f\|_{r+4}^{2} \ .$$

Summing up, the above considerations imply

$$\left\|\frac{e^{-it\mathcal{A}} - 1 + it\mathcal{A}}{t}\right\|_{\mathfrak{B}(\mathcal{H}^{r+4},\mathcal{H}^r)} \leq \frac{1}{2} |t| ,$$

which, in view of Eq. (2.139), proves the existence of the map $\mathbb{R} \to \mathfrak{B}(\mathcal{H}^{r+4}, \mathcal{H}^r)$, $t \mapsto \frac{d}{dt} e^{-it\mathcal{A}} = -i\mathcal{A} e^{-it\mathcal{A}}$. The latter can also be shown to be continuous by arguments analogous to those employed above, thus yielding the thesis.

Before moving on, let us prove the following proposition, relating the complex powers \mathcal{A}^{-s} $(s \in \mathbb{C})$ to the Mellin transforms of the exponential operators $e^{-t\mathcal{A}}, e^{-t\sqrt{\mathcal{A}}}, e^{-t\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}$.

Proposition 2.31. Let $\mathbf{t} \in (0, +\infty)$ and consider the operators $e^{-\mathbf{t}\mathcal{A}}$, $e^{-\mathbf{t}\sqrt{\mathcal{A}}}$,

$$\mathcal{A}^{-s}f = \frac{1}{\Gamma(s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, e^{-\mathbf{t}\mathcal{A}} f \,, \qquad (2.140)$$

$$\mathcal{A}^{-s}f = \frac{1}{\Gamma(2s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{2s-1} \, e^{-\mathbf{t}\sqrt{\mathcal{A}}} f \,, \qquad (2.141)$$

$$\mathcal{A}^{-s}f = \frac{1}{\Gamma(2s-1)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{2s-2} \left(e^{-\mathbf{t}\sqrt{\mathcal{A}}} / \sqrt{\mathcal{A}} \right) f \,. \tag{2.142}$$

In the above, all integrals are intended in the Gelfand-Pettis sense and involve functions from $(0, +\infty)$ to \mathcal{H}^{r_2} (in fact, according to Corollary 2.14, $e^{-\mathbf{t}\mathcal{A}}f$, $e^{-\mathbf{t}\sqrt{\mathcal{A}}}f$, $(e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})f \in \mathcal{H}^{+\infty} \hookrightarrow \mathcal{H}^{r_2}$).

Proof. We report the proof only for the operator $e^{-t\mathcal{A}}$ (the proof for the other operators is analogous). Fix $r_1, r_2 \in \mathbb{R}$ with $r_1 + r_2 > 0$; for any $s \in \Sigma_{(r_1+r_2)/2}$ and any $f \in \mathcal{H}^{-r_1}$, one has $\mathcal{A}^{-s}f \in \mathcal{H}^{2\Re s - r_1} \hookrightarrow \mathcal{H}^{r_2}$ (see Corollary 2.13). Next, recall that all linear forms on \mathcal{H}^{r_2} have the form $\langle g | \rangle$, for some $g \in \mathcal{H}^{-r_2}$ (with $\langle | \rangle$ denoting the extended inner product of Proposition 2.16). Then, by the definition of the Gelfand-Pettis integral, proving Eq. (2.140) amounts to showing the following: for any $g \in \mathcal{H}^{-r_2}$, the function $\mathbf{t} \mapsto \mathbf{t}^{s-1} \langle g | e^{-\mathbf{t}\mathcal{A}}f \rangle$ is integrable and

$$\langle g | \mathcal{A}^{-s} f \rangle = \frac{1}{\Gamma(s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, \langle g | e^{-\mathbf{t}\mathcal{A}} f \rangle \,.$$
 (2.143)

Now, consider the representation (2.54-2.56) for \mathcal{H} and \mathcal{A} ; then, \mathcal{A}^{-s} and $e^{-t\mathcal{A}}$ correspond to the multiplication operators by the functions $w^{-s} : K \to \mathbb{C}$ and $e^{-tw} : K \to \mathbb{R}$, respectively. So, Eq. (2.143) (to be proved) can be rephrased as

$$\int_{K} d\mu \ w^{-s} \,\overline{g} \, f = \frac{1}{\Gamma(s)} \int_{0}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \int_{K} d\mu \ e^{-\mathbf{t}w} \,\overline{g} \, f \ . \tag{2.144}$$

Indeed, recall that $w(k) \in [\varepsilon, +\infty)$ for a.e. $k \in K$ and that $\Re s > 0$, since $s \in \Sigma_{(r_1+r_2)/2}$ and $r_1+r_2 > 0$; in consequence of this, we have (see [122], page 139, Eq. 5.9.1) $w(k)^{-s} = \frac{1}{\Gamma(s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, e^{-\mathbf{t}w(k)}$ for a.e. $k \in K$. Therefore Eq. (2.144) is, in turn, equivalent to

$$\frac{1}{\Gamma(s)} \int_{K} d\mu \left(\int_{0}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, e^{-\mathbf{t}w} \right) \overline{g} \, f = \frac{1}{\Gamma(s)} \int_{0}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \left(\int_{K} d\mu \, e^{-\mathbf{t}w} \, \overline{g} \, f \right)$$

Summing up, the thesis follows if we can show that the order of integration on K and $(0, +\infty)$ can be interchanged; this statement is ensured by the Fubini-Tonelli theorem [134] if

$$I := \int_{(0,+\infty)\times K} d\mathbf{t} \otimes d\mu \, \left| \mathbf{t}^{s-1} \, e^{-\mathbf{t}w} \, \overline{g} \, f \right| \, < \, +\infty \, . \tag{2.145}$$

To prove Eq. (2.145), first re-express the integral therein as

$$I = \int_{(0,+\infty)\times K} d\mathbf{t} \otimes d\mu \, \mathbf{t}^{\Re s - 1} \left(e^{-\mathbf{t}w} \, w^{\frac{r_1 + r_2}{2}} \right) \left| w^{-r_2/2} g \right| \left| w^{-r_1/2} f \right| \, .$$

Then, consider the function $F: (0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto F(\mathbf{t}) := \sup_{w \in [\varepsilon, +\infty)} |w^{\frac{r_1+r_2}{2}} e^{-\mathbf{t}w}|$; by elementary methods, we obtain

$$F(\mathbf{t}) = \left(\frac{r_1 + r_2}{2e\,\mathbf{t}}\right)^{\frac{r_1 + r_2}{2}} \chi_{(0,(r_1 + r_2)/(2\varepsilon)]}(\mathbf{t}) + \left(\varepsilon^{\frac{r_1 + r_2}{2}} e^{-\varepsilon\mathbf{t}}\right) \chi_{((r_1 + r_2)/(2\varepsilon), +\infty)}(\mathbf{t})$$

(where χ denotes the indicator function; see the footnote 11 of page 29). It follows that

$$I \le \left(\int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{\Re s - 1} F(\mathbf{t})\right) \left(\int_K d\mu \, |w^{-r_2/2}g| \, |w^{-r_1/2}f|\right) \,. \tag{2.146}$$

On the one hand, due to Hölder's inequality we have $\int_K d\mu |w^{-r_2/2}g||w^{-r_1/2}f| \leq ||g||_{-r_2} ||f||_{-r_1}$, so that the second integral in Eq. (2.146) is finite. On the other hand, concerning the first integral in the same equation, we have

$$\int_{0}^{+\infty} d\mathbf{t} \, \mathbf{t}^{\Re s - 1} F(\mathbf{t}) = \left(\frac{r_1 + r_2}{2e}\right)^{\frac{r_1 + r_2}{2}} \int_{0}^{\frac{r_1 + r_2}{2\varepsilon}} d\mathbf{t} \, \mathbf{t}^{\Re s - \frac{r_1 + r_2}{2} - 1} + \varepsilon^{\frac{r_1 + r_2}{2}} \int_{\frac{r_1 + r_2}{2\varepsilon}}^{+\infty} d\mathbf{t} \, \mathbf{t}^{\Re s - 1} \, e^{-\mathbf{t}\varepsilon} ;$$

both the integrals on the right-hand side are finite since $s \in \Sigma_{(r_1+r_2)/2}$ and $\varepsilon > 0$. Summing up, Eq. (2.146) proves Eq. (2.145), thus yielding the thesis.

2.6 Schrödinger-type operators: regularity and self-adjointness.

So far, we have been considering an abstract framework where \mathcal{A} was some self-adjoint operator acting on a given Hilbert space \mathcal{H} . In this section we analyze more in detail the case where the basic Hilbert space is $\mathcal{H} = L^2(\Omega)$, with $\Omega \subset \mathbb{R}^d$ a suitable domain, and \mathcal{A} is determined by some differential operator on it.

Regularity results for elliptic differential operators.

Local results.

Let us first assume that Ω is an arbitrary domain in \mathbb{R}^d and let us choose a "potential"

$$V \in C^{\infty}(\Omega, \mathbb{R}) ; \qquad (2.147)$$

we will make more specific hypotheses on Ω and V when necessary. Moreover, let us consider the second-order, proper (see [38, 107, 156]) elliptic differential operator of Schrödinger type

$$A := -\bigtriangleup + V : \mathcal{D}'(\Omega) \to \mathcal{D}'(\Omega)$$
(2.148)

(recall that \triangle stands for the *d*-dimensional Laplacian, i.e., $\triangle := \sum_{i=1}^{d} \partial_{x^{i}x^{i}}$). For any $n \in \mathbb{N}$ and any $r \in \mathbb{R}$, it is easily checked that

$$A^{n}H^{r+2n}_{\text{loc}}(\Omega) \subset H^{r}_{\text{loc}}(\Omega)$$
(2.149)

and that $A^n \upharpoonright H^{r+2n}_{loc}(\Omega)$ maps continuously $H^{r+2n}_{loc}(\Omega)$ into $H^r_{loc}(\Omega)$ (see [38], page 98, Proposition 2.13). Furthermore, the standard theory of local regularity for elliptic differential operators allows to derive the following result.

Theorem 2.32. For $n \in \mathbb{N}$, the following statements hold. *i)* One has

$$H^{2n}_{loc}(\Omega) = \{ f \in \mathcal{D}'(\Omega) \mid A^m f \in L^2_{loc}(\Omega) \text{ for } m = 0, ..., n \} .$$
 (2.150)

ii) Consider the standard topology of $H^{2n}_{loc}(\Omega)$, based on the family of seminorms $f \mapsto \|\varphi f\|_{H^{2n}}$ ($\varphi \in \mathcal{D}(\Omega)$). This coincides with the topology induced by the family of seminorms

$$f \mapsto \|\varphi A^m f\|_{L^2} \qquad (\varphi \in \mathcal{D}(\Omega), \ m \in \{0, ..., n\}) \ . \tag{2.151}$$

Proof. We give the proof of the statements i) and ii) in several steps.

Step 1 - Proof of i). Let us provisionally denote with $\mathcal{H}^{2n}(\Omega)$ the right-hand side of Eq. (2.150); it is evident that $H^{2n}_{loc}(\Omega) \subset \mathcal{H}^{2n}(\Omega)$. The proof of the reverse inclusion relies on the following well-known result:

$$m \in \mathbb{N}, \ r \in \mathbb{R}, \ f \in \mathcal{D}'(\Omega), \ A^m f \in H^r_{\text{loc}}(\Omega) \Rightarrow f \in H^{r+2m}_{\text{loc}}(\Omega)$$
 (2.152)

(see, e.g., [141] and Theorem 3.2 of [97]). Now, let $f \in \mathcal{H}^{2n}(\Omega)$; the condition in Eq. (2.150) with m = n reads $A^n f \in L^2_{loc}(\Omega) \equiv H^0_{loc}(\Omega)$ and this implies $f \in H^{2n}_{loc}(\Omega)$.

Step 2 - An inequality. For any $\ell \in \mathbb{Z}$ and any compact subset $K \subset \Omega$, let us define the set $H_K^{\ell}(\Omega) := \{f \in H^{\ell}(\Omega) \mid \operatorname{supp} f \subset K\}$; then $AH_K^{\ell+2}(\Omega) \subset H_K^{\ell}(\Omega)$. Since A is a proper elliptic operator of degree 2 in the sense of [38] (see also [107, 156]), for each $\ell \in \mathbb{Z}$ and each compact subset $K \subset \Omega$ there exists a constant $C_{\ell,K} > 0$ such that

$$||f||_{H^{\ell+2}} \le C_{\ell,K}(||Af||_{H^{\ell}} + ||f||_{L^2}) \quad \text{for all } f \in H_K^{\ell+2}(\Omega)$$
(2.153)

(see [38], Theorem 8.11; the inequality written above follows from Eq. (8.11.3) therein, setting m = 2 and t = 0 (¹⁸)).

Step 3 - Another inequality. Let $\ell \in \mathbb{Z}$, $\varphi \in \mathcal{D}(\Omega)$. We claim there is a constant $C_{\ell,\varphi}$ such that, for all $f \in H^{\ell+2}_{\text{loc}}(\Omega)$,

$$\|\varphi f\|_{H^{\ell+2}} \leq C_{\ell,\varphi} \left(\|\varphi(Af)\|_{H^{\ell}} + \sum_{i=1}^{d} \|(\partial_{x^{i}}\varphi)f\|_{H^{\ell+1}} + \|(\triangle\varphi)f\|_{H^{\ell}} + \|\varphi f\|_{L^{2}} \right).$$
(2.154)

Indeed, let ℓ, φ be as above, and set $K := \operatorname{supp} \varphi$; in the sequel, "const." indicates a positive constant, depending only on ℓ and φ . Let $f \in H^{\ell+2}_{\operatorname{loc}}(\Omega)$; then $\varphi f \in H^{\ell+2}_{K}(\Omega)$ and the inequality in Eq. (2.153) for this function gives

$$\|\varphi f\|_{H^{\ell+2}} \le \text{const.}(\|A(\varphi f)\|_{H^{\ell}} + \|\varphi f\|_{L^2}) .$$
(2.155)

On the other hand $A(\varphi f) = \varphi(Af) - 2\sum_{i=1}^{d} \partial_{x^i}[(\partial_{x^i}\varphi)f] + (\triangle \varphi)f$, which allows to infer

$$\|A(\varphi f)\|_{H^{\ell}} \leq$$

$$\|\varphi(Af)\|_{H^{\ell}} + \text{const.} \sum_{i=1}^{d} \|(\partial_{x^{i}}\varphi)f\|_{H^{\ell+1}} + \|(\bigtriangleup\varphi)f\|_{H^{\ell}} .$$

$$(2.156)$$

Moreover, recall that φ is smooth and has compact support. These facts and the inequalities in Eq.s (2.155) (2.156) yield the thesis (2.154).

Step 4 - Proof of ii). Let us show that each of the topologies based respectively on the family of seminorms $f \mapsto \|\varphi A^m f\|_{L^2}$ and $f \mapsto \|\varphi f\|_{H^{2n}}$ ($\varphi \in \mathcal{D}(\Omega)$) is finer than the other. To this purpose we will prove that, for all $\varphi \in \mathcal{D}(\Omega)$ and for all $n \in \mathbb{N}$, there exist two constants $C_{n,\varphi}, C'_{n,\varphi} > 0$ and two finite sets $I_n, I'_{n,\varphi}$ such that

$$\|\varphi f\|_{H^{2n}} \le C_{n,\varphi} \sum_{(m,\alpha)\in I_n} \|(\partial^{\alpha}\varphi)A^m f\|_{L^2} , \qquad (2.157)$$

¹⁸As a matter of fact, the cited Eq. (8.11.3) of [38] is stated for functions in $C^{\infty}(\Omega)$ with support in a given compact subset, but can be readily extended to functions in $H_{K}^{\ell+2}(\Omega)$ by elementary density arguments.

2.6. SCHRÖDINGER-TYPE OPERATORS: REGULARITY AND SELF-ADJOINTNESS.

$$\|\varphi A^{n}f\|_{L^{2}} \leq C_{n,\varphi}' \sum_{\psi \in I_{n,\varphi}'} \|\psi f\|_{H^{2n}} .$$
(2.158)

The set I_n appearing above has elements of the form (m, α) , where $m \in \{0, ..., n\}$ and α a multi-index with $|\alpha| \leq 2n$; on the contrary, $I'_{n,\varphi}$ is a subset of $\mathcal{D}(\Omega)$. Let us begin proving Eq. (2.157). By Eq. (2.154) in Step 3 of the present proof (here

employed with $\ell = 2(n-1)$, for all $f \in H^{2n}_{loc}(\Omega)$ one has

$$\|\varphi f\|_{H^{2n}} \leq C_{2(n-1),\varphi} \left(\|\varphi(Af)\|_{H^{2(n-1)}} + \sum_{i=1}^{d} \|(\partial_{x^{i}}\varphi)f\|_{H^{2n-1}} + \|(\triangle\varphi)f\|_{H^{2(n-1)}} + \|\varphi f\|_{L^{2}} \right);$$

note that all the terms on the right-hand side of this inequality are of the form $\|(\partial^{\alpha}\varphi)v\|_{H^{k}}$ for some multi-index α of order $|\alpha| \leq 2$ and some $v \in H^{k}_{loc}(\Omega)$ $(k \in \mathbb{N}, k < 2n)$. Each of these terms can be estimated in the same way, employing once more Eq. (2.154) (with φ replaced by $\partial^{\alpha}\varphi$). This procedure can be iterated *n* times, until only norms of order 0 or less are left on the right-hand side; clearly, each term remaining is of the form $\|(\partial^{\alpha}\varphi)A^{m}f\|_{H^{k}}$ for some $m \in \{0, ..., n\}, k \in \{-1, 0\}$ and α a multi-index with $|\alpha| \leq 2n$. Then, the inequality (2.157) follows since $\|v\|_{H^{-1}} \leq \text{const.} \|v\|_{H^{0}} = \text{const.} \|v\|_{L^{2}}$ for all $v \in L^{2}(\Omega)$ with compact support (¹⁹).

Let us now pass to justify Eq. (2.158); indeed, this inequality is a just a reformulation of an already mentioned fact, i.e., the continuity of the differential operator $(A^n \upharpoonright H^{2n}_{\text{loc}}(\Omega))$: $H^{2n}_{\text{loc}}(\Omega) \to L^2_{\text{loc}}(\Omega)$ (let us recall again reference [38]).

Global results.

Making stricter assumptions on the domain Ω , on its boundary $\partial\Omega$ and on the regularity of the potential V defining the elliptic operator A of Eq. (2.148), one can infer stronger versions of Theorem 2.32 which allow to deal with the behaviour of the functions under analysis up to the boundary of Ω .

For example, from here to the end of this paragraph, let us consider the case where

$$\Omega \subset \mathbb{R}^d \text{ is a bounded domain with compact boundary } \partial\Omega \text{ of class } C^{\infty}$$

and $V : \Omega \to \mathbb{R}$ has a C^{∞} extension $V : \overline{\Omega} = \Omega \cup \partial\Omega \to \mathbb{R}$; (2.159)

these hypotheses suffice to infer the continuity of the differential operator $(A^n \upharpoonright H^{\ell+2n}(\Omega))$: $H^{\ell+2n}(\Omega) \to H^{\ell}(\Omega)$ for any $n \in \mathbb{N}$ and any $\ell \in \mathbb{Z}$ (compare with Eq. (2.149)).

In this situation, it is natural to consider the behaviour of functions on the boundary of Ω . As well known (see, e.g., [109]) there is a linear, continuous operation of *trace*

$$H^1(\Omega) \to L^2(\partial\Omega) , \qquad f \mapsto f \restriction \partial\Omega$$
 (2.160)

 $^{^{19}}$ See [97], Chapter 2, Section 3, pages 121–122.

that coincides with the usual operation of restriction to $\partial\Omega$ on any function $f \in C^1(\overline{\Omega})$; moreover the space $H_0^1(\Omega)$ (defined, we recall it, as the closure of $\mathcal{D}(\Omega)$ in $H^1(\Omega)$) admits in the present case the representation

$$H_0^1(\Omega) = \{ f \in H^1(\Omega) \mid f \upharpoonright \partial \Omega = 0 \} .$$

$$(2.161)$$

The prescription $f \upharpoonright \partial \Omega = 0$ is a *Dirichlet boundary condition*; in the sequel we will be especially interested in analysing the differential operator A (and its powers) with these boundary conditions, indicated by the symbol \mathfrak{D} . In view of this, following [109] (see pages 228–229), we introduce the following space:

Definition 2.33. For $n \in \mathbb{N}$ we put

$$H_{\mathfrak{D}}^{2n}(\Omega) := \{ f \in H^{2n}(\Omega) \mid (A^m f) \restriction \partial \Omega = 0 \text{ for } m = 0, ..., n-1 \} .$$
(2.162)

Remark 2.11. If $f \in H^{2n}(\Omega)$ and $m \in \{0, ..., n-1\}$, then $A^m f \in H^{2n-2m}(\Omega) \subset H^1(\Omega)$, so it makes sense to speak of the trace $(A^m f) \upharpoonright \partial \Omega$. Elementary considerations related to the linearity and continuity of the operators $A^m, \upharpoonright \partial \Omega$ ensure that $H^{2n}_{\mathfrak{D}}(\Omega)$ is a closed linear subspace of $H^{2n}(\Omega)$ and, in particular, it is itself a Banach space with the norm $\| \|_{H^{2n}}$; furthermore, it appears that $\mathcal{D}(\Omega)$ and, consequently, its closure $H^{2n}_0(\Omega)$ are linear subspaces of $H^{2n}_{\mathfrak{D}}(\Omega)$.

In analogy to Theorem 2.32, one can infer the forthcoming result.

Theorem 2.34. For $n \in \mathbb{N}$, the following statements hold. *i)* One has

$$H_{\mathfrak{D}}^{2n}(\Omega) = \left\{ f \in \mathcal{D}'(\Omega) \mid \begin{array}{c} A^n f \in L^2(\Omega) \quad and \\ A^m f \in H_0^1(\Omega) \quad for \ m = 0, ..., n-1 \end{array} \right\}$$
(2.163)

ii) The standard topology of $H^{2n}_{\mathfrak{D}}(\Omega)$, inherited from $H^{2n}(\Omega)$, coincides with the topology induced by the norm

$$f \mapsto \sum_{m=0}^{n} \|A^m f\|_{L^2}$$
 (2.164)

Proof. The thesis can be proved moving along the same lines as in the proof of Theorem 2.32; hereafter we only point out the main steps.

The essential argument is the following, well-known result of regularity (see [62] page 323, Theorem 5):

$$k \in \mathbb{N}, \ g \in H_0^1(\Omega), \ Ag \in H^k(\Omega) \quad \Rightarrow$$
$$g \in H^{k+2}(\Omega) \quad \text{and} \quad \|g\|_{H^{k+2}} \le \text{const.}(\|Ag\|_{H^k} + \|g\|_{L^2}) \ . \tag{2.165}$$

In order to show statement i), let us temporarily indicate with $\mathcal{H}_{\mathfrak{D}}^{2n}(\Omega)$ the space in the right-hand side of Eq. (2.163); of course, $H_{\mathfrak{D}}^{2n}(\Omega) \subset \mathcal{H}_{\mathfrak{D}}^{2n}(\Omega)$. On the other hand, for any $f \in \mathcal{H}_{\mathfrak{D}}^{2n}(\Omega)$, we have $A(A^{n-1}f) = A^n f \in L^2(\Omega)$ and $A^{n-1}f \in H_0^1(\Omega)$; by (2.165) with

 $g = A^{n-1}f$ and k = 0, this implies that $A^{n-1}f \in H^2(\Omega)$. Iterating these arguments one obtains $A^{n-\ell}f \in H^{2\ell}(\Omega)$ for $\ell = 0, ..., n$ and, in particular, $f \in H^{2n}(\Omega)$, which proves the inclusion $\mathcal{H}^{2n}_{\mathfrak{D}}(\Omega) \subset H^{2n}_{\mathfrak{D}}(\Omega)$.

Next, since Ω is bounded and the potential V is smooth up to the boundary (whence, bounded along with its derivatives), it can be easily checked that there exists a positive constant "const." such that, for all $f \in H^{2n}_{\mathfrak{D}}(\Omega)$,

$$\sum_{m=0}^{n} \|A^{m}f\|_{L^{2}} \le \text{const.} \|f\|_{H^{2n}} ; \qquad (2.166)$$

on the other hand for $f \in H^{2n}_{\mathfrak{D}}(\Omega)$, repeated use of the inequality in (2.165) gives $||f||_{H^{2n}} \leq \text{const.}(||Af||_{H^{2n-2}} + ||f||_{L^2}) \leq \text{const.}(||A^2f||_{H^{2n-4}} + ||Af||_{L^2} + ||f||_{L^2})$ and, more generally, $||f||_{H^{2n}} \leq \text{const.}(||A^\ell f||_{H^{2n-2\ell}} + \sum_{m=0}^{\ell} ||A^m f||_{L^2})$ for $\ell = 0, ..., n$; in particular, with $\ell = n$ we have

$$||f||_{H^{2n}} \le \text{const.} \sum_{m=0}^{n} ||A^m f||_{L^2} .$$
 (2.167)

Summing up, Eq.s (2.166) (2.167) prove statement ii).

Some possible generalizations.

Results analogous to those discussed in the previous Theorems 2.32 and 2.34 can be derived under more general hypotheses for the differential operator A, for the domain Ω and for the boundary conditions on $\partial\Omega$. The proofs of these results closely resemble those of the cited theorems, and do in fact rely on essentially the same arguments; for this reason, these proofs will not be reported in this manuscript. We will limit ourself to simply mention some variations of Theorems 2.32, 2.34 and to hint at some of their possible, further generalizations. Let us notice that, as a matter of fact, some of these generalizations are not of direct interest for the physical applications to be considered in Chapter 3 of this work. Nevertheless, they are likely to have an interest of their own and in connection with other topics, which are not dealt with in the present manuscript; for these reasons we report them here anyway for completeness.

First of all, let us stress that if A is any second order, formally self-adjoint, proper elliptic differential operator $(^{20})$

$$A := \sum_{|\alpha| \le 2} a_{\alpha} \partial^{\alpha} : \mathcal{D}'(\Omega) \to \mathcal{D}'(\Omega)$$
(2.168)

²⁰Formal self-adjointness of the operator A in (2.168) means that A coincides with its Lagrange adjoint [39]

$$A^{\dagger} := \sum_{|\alpha| \le 2} (-\partial)^{\alpha} \bar{a}_{\alpha} : \mathcal{D}'(\Omega) \to \mathcal{D}'(\Omega) .$$

We refer again to [38, 107, 156] for the definition of proper differential operator.

for some smooth coefficients $a_{\alpha} : \Omega \to \mathbb{C}$ ($\alpha \in \mathbb{N}^d$, $|\alpha| \leq 2$), then the results of Theorem 2.32 continue to hold, understanding A to be as in Eq. (2.168); similarly, if the domain Ω and its boundary are as in Eq. (2.159), assuming that $a_{\alpha} \in C^{\infty}(\overline{\Omega})$ and that Dirichlet boundary conditions are prescribed on $\partial\Omega$ one can infer statements analogous to those of Theorem 2.34 (²¹).

Next, let us note that one could make weaker assumptions on the regularity of the boundary $\partial\Omega$ and of the coefficients defining the differential operator A. For example, let us consider the case where A is as in Eq. (2.168) with $a_{\alpha} \in C^{2j}(\Omega)$ for some $j \in \mathbb{N}$; then, Theorem 2.32 continues to hold for all $n \in \mathbb{N}$ with $n \leq j$. On the other hand, Theorem 2.34 is still valid for the same values of n, if Ω is bounded with boundary $\partial\Omega$ of class C^{2j+2} and if $a_{\alpha} \in C^{2j}(\overline{\Omega})$ ($|\alpha| \leq 2$).

Finally, let us draw the attention to the fact that boundary conditions different from those of Dirichlet type could be taken into account, as well. To this purpose, let us consider the case described in Eq. (2.159), where Ω is bounded with boundary of class C^{∞} and $A = -\Delta + V$ with $V \in C^{\infty}(\overline{\Omega})$; hereafter we briefly analyze as examples the cases where either *Neumann* or *Robin boundary conditions* (respectively indicated with the symbols \mathfrak{N} and \mathfrak{R}) are imposed on $\partial\Omega$. Following the Definition 2.33 of $H^{2n}_{\mathfrak{D}}(\Omega)$ $(n \in \mathbb{N})$, we consider the spaces described hereafter (see again [109], pages 228–229).

Definition 2.35. For any $n \in \mathbb{N}$, we introduce the spaces (compare with Eq. (2.162))

$$H^{2n}_{\mathfrak{N}}(\Omega) := \{ f \in H^{2n}(\Omega) \mid \partial_{\mathbf{n}}(A^m f) \restriction \partial \Omega = 0 \text{ for } m = 0, ..., n-1 \} , \qquad (2.169)$$

$$H_{\mathfrak{R}}^{2n}(\Omega) := \{ f \in H^{2n}(\Omega) \mid (h + \partial_{\mathbf{n}})(A^m f) \restriction \partial\Omega = 0 \text{ for } m = 0, ..., n-1 \} ; \qquad (2.170)$$

here $\mathbf{n} : \partial \Omega \to \mathbb{R}^d$ is the outer normal versor, $\partial_{\mathbf{n}} = \sum_{i=1}^d n^i \partial_i$ indicates the normal derivative at points of $\partial \Omega$ and $h : \partial \Omega \to \mathbb{C}$ is an assigned function of class C^{∞} (²²).

$$A := \sum_{|\alpha| \le p} a_{\alpha} \, \partial^{\alpha} : \mathcal{D}'(\Omega) \to \mathcal{D}'(\Omega) \; .$$

For example, in place of Eq. (2.150) it can be proved that

$$H_{\text{loc}}^{pn}(\Omega) = \{ f \in \mathcal{D}'(\Omega) \mid A^m f \in L^2_{\text{loc}}(\Omega) \text{ for all } m \in \{0, ..., n\} \}$$

and that the topology determined by the family of seminorms $f \mapsto \|\varphi A^m f\|_{L^2}$ ($\varphi \in \mathcal{D}(\Omega), m \in \{0, ..., n\}$) is equivalent to the usual one of $H^{pn}_{\text{loc}}(\Omega)$. Moreover, results analogous to those of Theorem 2.34 could be derived assuming that Ω is bounded with smooth boundary $\partial\Omega$ and that the coefficients a_{α} ($|\alpha| \leq 2p$) are of class C^{∞} on $\overline{\Omega}$. Nevertheless, the case of a differential operator A of order different from p = 2mentioned within this footnote is not strictly relevant for the purposes of the present work; therefore, it will never be considered in the remainder of this manuscript.

²²For m = 0, ..., n - 1, the above prescriptions on the normal derivatives of $A^m f$ make sense because (for i = 1, ..., d) one has $\partial_i (A^m f) \in H^1(\Omega)$, which allows to define the trace of this function on $\partial\Omega$.

²¹ In fact, results analogous to Theorems 2.32 and 2.34 can be derived for any proper elliptic differential operator of arbitrary order $p \in \mathbb{N}$, with smooth coefficients a_{α} ($|\alpha| \leq p$):

Let us assume that \mathbf{n}, h have extensions $\mathbf{n} \in C^{\infty}(\overline{\Omega}, \mathbb{R}^d), h \in C^{\infty}(\overline{\Omega}, \mathbb{C})$ (allowing to define $\partial_{\mathbf{n}} := \sum_{i=1}^d n^i \partial_i$ and $h + \partial_{\mathbf{n}}$ everywhere on Ω). Then, in analogy to Theorem 2.34, one can easily infer the forthcoming result, which we report here without proof.

Theorem 2.36. For $n \in \mathbb{N}$, the following statements hold. *i)* One has

$$H^{2n}_{\mathfrak{N}}(\Omega) := \left\{ f \in \mathcal{D}'(\Omega) \mid \begin{array}{c} A^n f \in L^2(\Omega) \quad and \\ \partial_n(A^m f) \in H^1_0(\Omega) \quad for \ m = 0, \dots, n-1 \end{array} \right\},$$
(2.171)

$$H^{2n}_{\mathfrak{R}}(\Omega) := \left\{ f \in \mathcal{D}'(\Omega) \mid \begin{array}{c} A^n f \in L^2(\Omega) \quad and \\ (h + \partial_n)(A^m f) \in H^1_0(\Omega) \quad for \ m = 0, ..., n-1 \end{array} \right\}.$$
(2.172)

ii) The topology of induced by $H^{2n}(\Omega)$ on $H^{2n}_{\mathfrak{R}}(\Omega)$ and $H^{2n}_{\mathfrak{R}}(\Omega)$ coincides with the one determined by the norm $f \mapsto \sum_{m=0}^{n} \|A^m f\|_{L^2}$.

Remark 2.12. As a matter of fact, most of the results to be derived in the remainder of the present Chapter 2 continue to hold under more general assumptions; for brevity, this fact will not be restated every once in a while, although it is implicitly understood most of the times.

In order to avoid misunderstandings, let us stress that in the following we will always assume A to be the differential operator of Eq. (2.148). Moreover, we will generically speak of *suitable boundary conditions* (*s.b.c.* for short) when the prescribed conditions on $\partial\Omega$ are of Dirichlet, Neumann or Robin type; in fact, we will typically consider boundary conditions of Dirichlet type, even though most of the results presented in the following continue to hold if Dirichlet conditions are replaced with s.b.c. (also of Neumann or Robin type).

The general notion of admissible operator. Some examples.

Let us consider the Schrödinger-type differential operator $A := -\Delta + V$ introduced in Eq. (2.148), where $V \in C^{\infty}(\Omega)$. The definition that follows will be referred to systematically in the sequel.

Definition 2.37. Consider a linear subspace $\mathcal{D}_{\mathcal{A}}$ of the Hilbert space $L^2(\Omega)$. We say that $\mathcal{D}_{\mathcal{A}}$ is an *admissible domain for* A if: i) $A \mathcal{D}_{\mathcal{A}} \subset L^2(\Omega)$ and the operator

$$\mathcal{A} := (A \upharpoonright \mathcal{D}_{\mathcal{A}}) : \mathcal{D}_{\mathcal{A}} \subset L^2(\Omega) \to L^2(\Omega)$$
(2.173)

is self-adjoint;

ii) $\sigma(\mathcal{A}) \subset (0, +\infty)$ (strict inclusion), where $\sigma(\mathcal{A})$ indicates the spectrum of \mathcal{A} . Under the above conditions, \mathcal{A} will be referred to as an *admissible operator*.

Herefter we review some known cases where $\mathcal{D}_{\mathcal{A}}$ verifies the admissibility conditions. In these examples $\mathcal{D}_{\mathcal{A}}$ is contained in $H_0^1(\Omega)$ which means that, if $\partial \Omega \neq \emptyset$, Dirichlet boundary conditions are being considered.

Given $\mathcal{D}_{\mathcal{A}}$ and V with suitable features, the propositions reviewed hereafter contain sufficient conditions for the self-adjointness of \mathcal{A} and also localize the spectrum $\sigma(\mathcal{A})$ within proper intervals, thus allowing to prove in certain cases that $\sigma(A) \subset (0, +\infty)$. It would not be difficult to extend our list of examples considering cases where the boundary conditions are of either Neumann or Robin type (see the comments at the end of the

Proposition 2.38. i) Assume

previous subsection).

$$\Omega \subset \mathbb{R}^d \text{ to be an arbitrary domain,} V(\mathbf{x}) \ge W \text{ for all } \mathbf{x} \in \Omega \ (W \in \mathbb{R})$$

$$(2.174)$$

and define

$$\mathcal{D}_{\mathcal{A}} := \{ f \in H_0^1(\Omega) \mid Af \in L^2(\Omega) \} .$$

$$(2.175)$$

Then $\mathcal{A} := \mathcal{A} \upharpoonright \mathcal{D}_{\mathcal{A}}$ is self-adjoint and

$$\sigma(A) \subset [W, +\infty) . \tag{2.176}$$

In particular, \mathcal{A} is an admissible operator if W > 0. ii) In the particular case $\Omega = \mathbb{R}^d$, the definition (2.175) of $\mathcal{D}_{\mathcal{A}}$ is equivalent to

$$\mathcal{D}_{\mathcal{A}} := \{ f \in L^2(\mathbb{R}^d) \mid Af \in L^2(\mathbb{R}^d) \} .$$
(2.177)

Proof. i) Assume first W = 0; in this case \mathcal{A} is *m*-accretive (see [89], Theorem I). Since \mathcal{A} is symmetric, this is equivalent to say that \mathcal{A} is self-adjoint and $\sigma(\mathcal{A}) \subset [0, +\infty)$ (see [90], page 279, Problem 3.32).

Consider now the case of an arbitrary $W \in \mathbb{R}$. Let $\tilde{V}(\mathbf{x}) := V(\mathbf{x}) - W$, and \tilde{A} , $\text{Dom}(\tilde{\mathcal{A}})$, $\tilde{\mathcal{A}}$ be defined respectively as A, $\mathcal{D}_{\mathcal{A}}$ and \mathcal{A} replacing V with \tilde{V} ; since $\tilde{V}(\mathbf{x}) \geq 0$, the result proved above for W = 0 implies that $\tilde{\mathcal{A}}$ is self-adjoint and $\sigma(\tilde{\mathcal{A}}) \subset [0, +\infty)$. On the other hand, one easily ckecks that $\mathcal{D}_{\mathcal{A}} = \text{Dom}(\tilde{\mathcal{A}})$ and that $\mathcal{A} = \tilde{\mathcal{A}} + W\mathbb{I}$ (with \mathbb{I} indicating the identity operator on $L^2(\Omega)$), whence the thesis about \mathcal{A} .

ii) Let $\Omega = \mathbb{R}^d$ and recall that $H_0^1(\mathbb{R}^d) = H^1(\mathbb{R}^d)$ (see [7], page 56, Corollary 3.19); to get the thesis, it must be shown that $f \in L^2(\mathbb{R}^d)$ and $Af \in L^2(\mathbb{R}^d)$ imply $f \in H^1(\mathbb{R}^d)$. Indeed, the assumptions on f and Proposition (2.32) give $f \in H^2_{loc}(\mathbb{R}^d)$, so the first and second distributional derivatives of f are ordinary functions of class $L^2_{loc}(\mathbb{R}^d)$. To prove that $f \in H^1(\mathbb{R}^d)$ it remains to show that $\int_{\mathbb{R}^d} |\nabla f|^2 = \sum_{i=1}^d \int_{\mathbb{R}^d} |\partial_i f|^2 < +\infty$; let us only sketch the argument, based on the manipulations reported hereafter and on the remark that $V - W \ge 0$:

$$+\infty > \int_{\mathbb{R}^d} \overline{f}(-\bigtriangleup f + Vf) - W \int_{\mathbb{R}^d} |f|^2 = \int_{\mathbb{R}^d} |\nabla f|^2 + \int_{\mathbb{R}^d} (V - W) |f|^2 \ge \int_{\mathbb{R}^d} |\nabla f|^2 .$$

Proposition 2.39. Assume

$$\Omega \subset \mathbb{R}^d \text{ to be a domain bounded along a direction}, V(\mathbf{x}) \ge W \text{ for all } \mathbf{x} \in \Omega \ (W \in \mathbb{R})$$

$$(2.178)$$

and consider the self-adjoint operator \mathcal{A} with domain $\mathcal{D}_{\mathcal{A}}$ as in Eq. (2.175). Then

$$\sigma(\mathcal{A}) \subset \left[\frac{1}{c_{\Omega}} + W, +\infty\right) , \qquad (2.179)$$

where $c_{\Omega} > 0$ is a constant fulfilling the Poincaré inequality of Eq. (2.6). In particular, \mathcal{A} is an admissible operator whenever $W > -1/c_{\Omega}$.

Proof. It suffices to show that

$$\langle f | \mathcal{A} f \rangle_{L^2} \ge \left(\frac{1}{c_{\Omega}} + W\right) \| f \|_{L^2}^2 \quad \text{for all } f \in \mathcal{D}_{\mathcal{A}} .$$
 (2.180)

In fact, given $f \in \mathcal{D}_{\mathcal{A}}$ we have

$$\langle f | \mathcal{A} f \rangle_{L^2} = \int_{\Omega} \overline{f} (-\bigtriangleup f + V f) = \int_{\Omega} |\nabla f|^2 + \int_{\Omega} V |f|^2 \ge \frac{1}{c_{\Omega}} \int_{\Omega} |f|^2 + W \int_{\Omega} |f|^2 ,$$

whence the thesis of Eq. (2.180).

Proposition 2.40. Let

$$\Omega = \mathbb{R}^d , \qquad V(\boldsymbol{x}) \ge W \text{ for all } \boldsymbol{x} \in \mathbb{R}^d \ (W \in \mathbb{R}) , \qquad (2.181)$$

and consider the self-adjoint operator \mathcal{A} with domain $\mathcal{D}_{\mathcal{A}}$ as in Eq. (2.177). Besides, assume that there exist $r_0 \in (0, +\infty)$, $p \in (1, +\infty)$ and $a, b \in (0, +\infty)$ (depending on V) such that, for all $\mathbf{x}_0 \in \mathbb{R}^d$,

$$\frac{1}{|B(\boldsymbol{x}_0, r_0)|} \int_{B(\boldsymbol{x}_0, r_0)} (V - W) \ge a , \qquad (2.182)$$

$$\left(\frac{1}{|B(\boldsymbol{x}_{0},r_{0})|}\int_{B(\boldsymbol{x}_{0},r_{0})}(V-W)^{p}\right)^{\frac{1}{p}} \leq \frac{b}{|B(\boldsymbol{x}_{0},r_{0})|}\int_{B(\boldsymbol{x}_{0},r_{0})}(V-W) ; \qquad (2.183)$$

finally, let $c_B \in (0, +\infty)$ denote a constant fulfilling the Poincaré-type inequality of Eq. (2.7). Then

$$\sigma(\mathcal{A}) \subset [F + W, +\infty) , \qquad F := \frac{a}{2^{2d} (2c_B^2 a \, b^{\frac{p}{p-1}} r_0^2 + 1) \, \max(2, 2^{\frac{1}{p-1}})} > 0 . \tag{2.184}$$

In particular, \mathcal{A} is an admissible operator if W > -F.

Proof. In the case W = 0, the thesis is proved in [145] (see Theorem A of this work; the constants c_i , i = 1, 2, 3, 4, mentioned therein satisfy $c_1 = c_B$, $c_2 = 2^d$, $c_3 \ge a$, $c_4 = b$). The case of an arbitrary $W \in \mathbb{R}$ is treated applying the results of [145] with V replaced by $\tilde{V} := V - W$; then, as in the proof of the Proposition 2.38, one must note that the operator $\tilde{\mathcal{A}}$ corresponding to \tilde{V} has the same domain $\mathcal{D}_{\mathcal{A}}$ as \mathcal{A} and is related to it by $\mathcal{A} = \tilde{\mathcal{A}} + W\mathbb{I}$ (again, recall that \mathbb{I} is the identity operator on $L^2(\Omega)$).

Let us add to the previous examples a statement on the integer powers \mathcal{A}^n $(n \in \mathbb{N})$ of a suitable admissible operator \mathcal{A} ; let us recall that by item iii) in Proposition 2.4, the space $\text{Dom}(\mathcal{A}^n)$ coincides with the abstract space \mathcal{H}^{2n} associated to \mathcal{A} ; the latter carries the Hilbertian norm $\|f\|_{2n} := \|\mathcal{A}^n f\|_{L^2}$.

Proposition 2.41. Assume \mathcal{A} to be an admissible operator with domain $\mathcal{D}_{\mathcal{A}} = \{f \in H_0^1(\Omega) \mid Af \in L^2(\Omega)\}$. Then the following statements i) and ii) hold. i) For all $n \in \mathbb{N}$

$$Dom(\mathcal{A}^n) = \{ f \in \mathcal{D}'(\Omega) \mid A^m f \in H^1_0(\Omega) \text{ for } m = 0, ..., n-1, A^n f \in L^2(\Omega) \} .$$
 (2.185)

ii) If in addition the assumptions (2.159) are fulfilled (so that $H_0^1(\Omega) = \{f \in H^1(\Omega) \mid f \upharpoonright \partial \Omega = 0\}$ and \mathcal{D}_A can be interpreted in terms of Dirichlet boundary conditions), then, for all $n \in \mathbb{N}$,

$$Dom(\mathcal{A}^n) = \{ f \in H^{2n}(\Omega) \mid A^m f \upharpoonright \partial \Omega = 0 \text{ for } m = 0, ..., n-1 \} ; \qquad (2.186)$$

 $Dom(\mathcal{A}^n)$ is a closed subspace of $H^{2n}(\Omega)$, and the topology that it inherits from $H^{2n}(\Omega)$ coincides with the one induced by the norm $\|f\|_{2n} = \|\mathcal{A}f\|_{L^2}$.

Proof. i) This statement follows by a simple verification by recurrence, based on the definition $\text{Dom}(\mathcal{A}^{n+1}) = \{f \in \text{Dom}(\mathcal{A}^n) \mid \mathcal{A}^n f \in \mathcal{D}_{\mathcal{A}}\}.$

ii) To prove the representation (2.186) of $\text{Dom}(\mathcal{A}^n)$, we must show that the right-hand sides of Eq.s (2.185) and (2.186) coincide; in fact, we already know this from subsection 2.6 (see Definition 2.33 and Theorem 2.34). In the cited subsection we have already noted that the space described by (2.186) is a closed subspace of $H^{2n}(\Omega)$, and we have shown that the topology that it inherits from $H^{2n}(\Omega)$ agrees with the one given by the norm $f \mapsto \sum_{m=0}^{n} \|\mathcal{A}^m f\|_{L^2}$ (see again Theorem 2.34). In the present case where admissibility requires $\text{Dom}(\mathcal{A}) \subset [\varepsilon, +\infty)$ for some $\varepsilon > 0$, we have $\|\mathcal{A}^m f\|_{L^2} \leq \varepsilon^{-(n-m)} \|\mathcal{A}^n f\|_{L^2}$ for m = 0, ..., n (see Eq. (2.84)), so that $\|\mathcal{A}^n f\|_{L^2}$ is a norm equivalent to $\sum_{m=0}^{n} \|\mathcal{A}^m f\|_{L^2}$. \Box

Embedding results for the scale of Hilbert spaces.

Assume again $\Omega \subset \mathbb{R}^d$ to be an arbitrary domain and A to be a Schrödinger-type operator as in Eq. (2.148). Let $\mathcal{D}_{\mathcal{A}} \subset L^2(\Omega)$ denote any admissible domain for A and consider the spaces $(\mathcal{H}^r, || \, ||_r)$ associated to the admissible operator \mathcal{A} (see Proposition 2.4). Recall that $||f||_r = ||\mathcal{A}^{r/2}f||_{L^2}$, for $f \in \text{Dom}(\mathcal{A}^{r/2})$; besides, \mathcal{H}^r coincides with $\text{Dom}(\mathcal{A}^{r/2})$ for $r \geq 0$, and it is its completion for r < 0. **Proposition 2.42.** Let $r \in [0, +\infty)$; then \mathcal{H}^r is a linear subspace of $H^r_{loc}(\Omega)$, and there holds the continuous embedding

$$\mathcal{H}^r \hookrightarrow H^r_{loc}(\Omega) \ . \tag{2.187}$$

Proof. The proof is divided in several steps.

Step 1 - The thesis holds in the case r = 2n, for any $n \in \mathbb{N}$. Let $f \in \mathcal{H}^{2n} = \text{Dom}(\mathcal{A}^n)$; then $A^m f \in L^2(\Omega) \subset L^2_{loc}(\Omega)$ for all $m \in \{0, ..., n\}$. This implies $f \in H^{2n}_{loc}(\Omega)$, due to item i) of Theorem 2.32. On the one hand, by item ii) of the same theorem, the topology of $H^{2n}_{loc}(\Omega)$ is induced by the family of seminorms $f \mapsto \|\varphi A^m f\|_{L^2}$, for $\varphi \in \mathcal{D}(\Omega)$ and $m \in \{0, ..., n\}$. On the other hand, for such φ, m and for all $f \in \mathcal{H}^{2n}$, one has

$$\|\varphi A^m f\|_{L^2} = \|\varphi \mathcal{A}^m f\|_{L^2} \le \le \left(\sup_{\Omega} |\varphi|\right) \|\mathcal{A}^m f\|_{L^2} = \left(\sup_{\Omega} |\varphi|\right) \|f\|_{2m} \le \left(\sup_{\Omega} |\varphi|\right) \varepsilon^{-(n-m)} \|f\|_{2m}$$

(see Proposition 2.5 for the last inequality); this proves the continuity of the embedding $\mathcal{H}^{2n} \hookrightarrow H^{2n}_{\text{loc}}(\Omega)$.

Step 2 - Reformulation of the statement to be proved, for arbitrary r. For each $\varphi \in \mathcal{D}(\Omega)$, let us introduce the linear operator of multiplication $M_{\varphi}: \mathcal{D}'(\Omega) \to \mathcal{D}'(\mathbb{R}^d), f \mapsto M_{\varphi}f := \varphi f$. Consider any $r \in [0, +\infty)$; then, the definition of $H^r_{\text{loc}}(\Omega)$ can be rephrased as $H^r_{\text{loc}}(\Omega) = \{f \in \mathcal{D}'(\Omega) | M_{\varphi}f \in H^r(\mathbb{R}^d) \text{ for all } \varphi \in \mathcal{D}(\Omega)\}$. Recall that the topology of $H^r_{\text{loc}}(\Omega)$ is, by construction, the initial topology with respect to the family of maps $M_{\varphi}: H^r_{\text{loc}}(\Omega) \to H^r(\mathbb{R}^d) \ (\varphi \in \mathcal{D}(\Omega))$. Then, due to the characteristic property of the initial topology, the statement $\mathcal{H}^r \to H^r_{\text{loc}}(\Omega)$ is equivalent to the following: "for each $\varphi \in D(\Omega), M_{\varphi}$ maps continuously \mathcal{H}^r into $H^r(\mathbb{R}^d)$ ".

Step 3 - Proof of the thesis for arbitrary $r \in [0, +\infty)$. Given such an r, let n denote the unique natural number such that 2n < r < 2n + 2; then there is a unique $\theta \in$ (0,1) such that $r = 2n(1-\theta) + (2n+2)\theta$. Due to Step 1, we know that $\mathcal{H}^{2n} \hookrightarrow$ $\mathcal{H}^{2n}_{loc}(\Omega)$ and $\mathcal{H}^{2n+2} \hookrightarrow \mathcal{H}^{2n+2}_{loc}(\Omega)$; so, due to Step 2, for each $\varphi \in \mathcal{D}(\Omega)$ the operator M_{φ} maps continuously \mathcal{H}^{2n} into $\mathcal{H}^{2n}(\mathbb{R}^d)$, and \mathcal{H}^{2n+2} into $\mathcal{H}^{2n+2}(\mathbb{R}^d)$. By the fundamental theorem of interpolation (Theorem 2.3), for each φ the operator M_{φ} maps continuously $[\mathcal{H}^{2n}, \mathcal{H}^{2n+2}]_{\theta}$ into $[\mathcal{H}^{2n}(\mathbb{R}^d), \mathcal{H}^{2n+2}(\mathbb{R}^d)]_{\theta}$; however, using Eq.s (2.37) (2.85) with $r_0 = 2n$, $r_1 = 2n + 2$, we obtain $[\mathcal{H}^{2n}, \mathcal{H}^{2n+2}]_{\theta} = \mathcal{H}^r$ and $[\mathcal{H}^{2n}(\mathbb{R}^d), \mathcal{H}^{2n+2}(\mathbb{R}^d)] = \mathcal{H}^r(\mathbb{R}^d)$. In conclusion, for any $\varphi \in \mathcal{D}(\Omega)$ the operator M_{φ} maps continuously \mathcal{H}^r to $\mathcal{H}^r(\mathbb{R}^d)$; due to Step 2, this proves that $\mathcal{H}^r \hookrightarrow \mathcal{H}^r_{loc}(\Omega)$.

Corollary 2.43. Let $r \in \mathbb{R}$ and $j \in \mathbb{N}$ be such that r > j + d/2; then there holds the continuous inclusion

$$\mathcal{H}^r \hookrightarrow C^j(\Omega)$$
 . (2.188)

More precisely, $\mathcal{H}^r \hookrightarrow C^{j,\lambda}(\Omega) \hookrightarrow C^j(\Omega)$ for each $\lambda \in (0,1)$ such that $r > j + d/2 + \lambda$.

Proof. In fact $\mathcal{H}^r \hookrightarrow H^r_{\text{loc}}(\Omega)$, (for $r \ge 0$) by the previous proposition, and $H^r_{\text{loc}}(\Omega) \hookrightarrow C^{j,\lambda}(\Omega) \hookrightarrow C^j(\Omega)$ by the Sobolev embedding theorem (see Theorem 2.1 of page 17). \Box

Similarly to the analysis performed in subsection 2.6, hereafter we are going to show that stronger versions of Proposition 2.42 and of the related Corollary 2.43 can be derived making stricter hypotheses on the regularity of the domain Ω , of its boundary $\partial\Omega$ and of the Schrödinger-type operator $A = -\Delta + V$.

For example, let us make the assumptions described in Eq. (2.159), so that Ω is a bounded domain with boundary of class C^{∞} and $A = -\Delta + V$ with $V \in C^{\infty}(\overline{\Omega})$, and consider the Sobolev spaces $H^{r}(\Omega)$ $(r \in \mathbb{R})$. Then, there hold the following results.

Proposition 2.44. Let the assumptions (2.159) hold with Dirichlet boundary conditions prescribed on $\partial\Omega$; then, for any $r \in [0, +\infty)$, \mathcal{H}^r is a linear subspace of $H^r(\Omega)$ and there holds the continuous embedding

$$\mathcal{H}^r \hookrightarrow H^r(\Omega)$$
 . (2.189)

Proof. The proof closely resembles that of Proposition 2.42, and it is divided in several steps likewise.

Step 1 - The thesis holds in the case r = 2n, for any $n \in \mathbb{N}$. This is granted by item ii) of Proposition 2.41, dealing with $\text{Dom}(\mathcal{A}^n) = \mathcal{H}^{2n}$.

Step 2 - Proof of the thesis for arbitrary $r \in [0, +\infty)$. Let $n \in \mathbb{N}$ be such that 2n < r < 2n + 2; then, there exists a unique $\theta \in (0, 1)$ such that $r = 2n(1 - \theta) + (2n + 2)\theta$. Due to the interpolation relations (2.85) and (2.38), we have $[\mathcal{H}^{2n}, \mathcal{H}^{2n+2}]_{\theta} = \mathcal{H}^r$ and $[H^{2n}(\Omega), H^{2n+2}(\Omega)]_{\theta} = H^r(\Omega)$. on the other hand, due to Step 1, we already know that the identity maps linearly and continuously \mathcal{H}^{2n} into $H^{2n}(\Omega)$ and \mathcal{H}^{2n+2} into $H^{2n+2}(\Omega)$. So, by the fundamental theorem of interpolation (see Theorem 2.3), the identity maps linearly and continuously \mathcal{H}^r into $H^r(\Omega)$.

Corollary 2.45. Let the assumptions (2.159) hold with Dirichlet boundary conditions prescribed on $\partial\Omega$; moreover, let $r \in \mathbb{R}$ and $j \in \mathbb{N}$ be such that r > j + d/2. Then there holds the continuous inclusion

$$\mathcal{H}^r \hookrightarrow C^j(\overline{\Omega})$$
 . (2.190)

More precisely, $\mathcal{H}^r \hookrightarrow C^{j,\lambda}(\overline{\Omega}) \hookrightarrow C^j(\overline{\Omega})$ for each $\lambda \in (0,1)$ such that $r > j + d/2 + \lambda$.

Proof. In fact $\mathcal{H}^r \hookrightarrow H^r(\Omega)$ (for $r \ge 0$) by the previous proposition, and $H^r(\Omega) \hookrightarrow C^{j,\lambda}(\overline{\Omega}) \hookrightarrow C^j(\overline{\Omega})$ by the Sobolev embedding (2.19) of Theorem 2.1.

Dirac delta functions.

The Dirac delta at interior points.

Let once more Ω be an arbitrary domain and let $V : \Omega \to \mathbb{R}$ be a smooth potential on it; we indicate with **x** an arbitrary point in the interior of Ω , i.e.,

$$\mathbf{x} \in \Omega \ . \tag{2.191}$$

Corollary 2.43, discussed in the previous subsection, has the following implications.
Proposition 2.46. Let $x \in \Omega$; then, there hold the subsequent statements. i) There exists a unique element $\delta_x \in \mathcal{H}^{-\infty}$ such that, for all r > d/2, $\delta_x \in \mathcal{H}^{-r}$ and

$$\langle \delta_{\boldsymbol{x}} | f \rangle = f(\boldsymbol{x}) \qquad \text{for all } f \in \mathcal{H}^r$$
. (2.192)

ii) Let α denote a multi-index of order $|\alpha|$; then, there exists a unique element $\partial^{\alpha} \delta_{x} \in \mathcal{H}^{-\infty}$ such that, for all $r > |\alpha| + \frac{d}{2}$, $\partial^{\alpha} \delta_{x} \in \mathcal{H}^{-r}$ and

$$\langle \partial^{\alpha} \delta_{\boldsymbol{x}} | f \rangle = (-1)^{|\alpha|} \partial^{\alpha} f(\boldsymbol{x}) \qquad \text{for all } f \in \mathcal{H}^r$$
. (2.193)

Remark 2.13. Let us stress that the left-hand sides of Eq.s (2.192) (2.193) both contain the extension (2.103) of the inner product $\langle | \rangle$ on $\mathcal{H} \equiv L^2(\Omega)$. To define the right-hand side of Eq. (2.192), notice that $\mathcal{H}^r \hookrightarrow C^0(\Omega)$ for any r > d/2 and evaluate at \mathbf{x} the continuous function f; on the other hand, recall that $\mathcal{H}^r \hookrightarrow C^j(\Omega)$ for r > j + d/2, so that the the right-hand side of Eq. (2.193) can be interpreted in terms of the α -th derivative at \mathbf{x} of the C^j function f (for any $|\alpha| \leq j$).

Proof. i) First note that the evaluation map

$$e_{\mathbf{x}}: C^{0}(\Omega) \to \mathbb{C} , \qquad f \mapsto \langle e_{\mathbf{x}}, f \rangle := f(\mathbf{x})$$

is a continuous linear form. Now, let us fix r > d/2 arbitrarily; since $\mathcal{H}^r \hookrightarrow C^0(\Omega)$, $e_{\mathbf{x}} \upharpoonright \mathcal{H}^r$ is an element of the topological dual $(\mathcal{H}^r)'$. Then, by Proposition 2.18 there is a unique element of \mathcal{H}^{-r} , which we denote provisionally with $\delta_{\mathbf{x}}^r$, such that $\langle \delta_{\mathbf{x}}^r | f \rangle = \langle e_{\mathbf{x}}, f \rangle = f(\mathbf{x})$ for all $f \in \mathcal{H}^r$.

To go on, note that $\delta_{\mathbf{x}}^s = \delta_{\mathbf{x}}^r$ for all $s, r \in \mathbb{R}$ such that $s \geq r > d/2$; in fact, for any such s and r, one has $\delta_{\mathbf{x}}^s \in \mathcal{H}^{-s}$, $\delta_{\mathbf{x}}^r \in \mathcal{H}^{-r} \subset \mathcal{H}^{-s}$ and $\langle \delta_{\mathbf{x}}^s | f \rangle = f(\mathbf{x}) = \langle \delta_{\mathbf{x}}^r | f \rangle$ for all $f \in \mathcal{H}^s \subset \mathcal{H}^r$, so that $\delta_{\mathbf{x}}^s$ and $\delta_{\mathbf{x}}^r$ are the same element of \mathcal{H}^{-s} .

In conclusion, all the elements $\delta_{\mathbf{x}}^r$ (r > d/2) are copies of a same element, that we indicate with $\delta_{\mathbf{x}}$; this element belongs to \mathcal{H}^{-r} for any r > d/2, and it fulfills Eq. (2.192) by construction. Finally, it is easy to check that an element of $\mathcal{H}^{-\infty}$ possessing the same properties coincides with this $\delta_{\mathbf{x}}$, thus granting its uniqueness.

ii) The proof follows by arguments similar to the ones employed to show statement i) for the special case $|\alpha| = 0$. For any $j \in \mathbb{N}$ and for any multi-index α of order $\leq j$, one uses the continuity of the linear form

 $e_{\mathbf{x}}^{\alpha}: C^{j}(\Omega) \to \mathbb{C}$, $f \mapsto \langle e_{\mathbf{x}}^{\alpha}, f \rangle := (\partial^{\alpha} f)(\mathbf{x})$ (2.194)

together with the continuous embedding $\mathcal{H}^r \hookrightarrow C^j(\Omega)$ for r > j + d/2; the element $\partial^{\alpha} \delta_{\mathbf{x}} \in \mathcal{H}^{-\infty}$ is the one corresponding to the linear form $(-1)^{|\alpha|} e_{\mathbf{x}}^{\alpha}$.

Apart from the use of the spaces \mathcal{H}^r (for r > j + d/2), Eq.s (2.192) and (2.193) closely resemble the definitions given in the standard theory of Schwartz distributions for the Dirac delta function at a point **x** and for its derivatives, respectively; this justifies the following definition. **Definition 2.47.** In the sequel, $\delta_{\mathbf{x}}$ and $\partial^{\alpha} \delta_{\mathbf{x}}$ will be referred to as the *Dirac delta* at \mathbf{x} and its α -th derivative, respectively.

Proposition 2.48. Let $j \in \mathbb{N}$ and $r \in \mathbb{R}$ be such that r > j + d/2; consider the map

$$\delta: \Omega \to \mathcal{H}^{-r} \qquad \boldsymbol{x} \mapsto \delta_{\boldsymbol{x}}$$
 (2.195)

This map is of class C^j from Ω to the Banach space \mathcal{H}^{-r} ; more precisely, $\delta \in C^{j,\lambda}(\Omega, \mathcal{H}^{-r})$ for any $\lambda \in (0,1)$ such that $r > j + d/2 + \lambda$.

Moreover, for any multi-index α of order $\leq j$, the corresponding partial derivative of δ at a point $\mathbf{x} \in \Omega$ is given by

$$(\partial^{\alpha}\delta)(\boldsymbol{x}) = (-1)^{|\alpha|} \partial^{\alpha}\delta_{\boldsymbol{x}} .$$
(2.196)

Proof. In this proof, for any locally convex space X, the topological dual space X' is equipped with the strong topology (i.e., with the topology of uniform convergence on the bounded subsets of X); this topology is induced by the family of seminorms $p_{\mathcal{B}}$ ($\mathcal{B} \subset X$ bounded) where, for each $\lambda \in X'$, we set

$$p_{\mathcal{B}}(\lambda) := \sup_{x \in \mathcal{B}} |\langle \lambda, x \rangle| .$$
(2.197)

In particular, if X is a Banach space, the strong topology on X' is just the usual norm topology. The rest of the proof is divided in several steps.

Step 1 - The maps e^{α} on $C^{j,\lambda}(\Omega)$. Let $j \in \mathbb{N}$, $\lambda \in (0,1)$, and consider a multi-index α of order $\leq j$. For each $\mathbf{x} \in \Omega$ we put

$$e^{\alpha}_{\mathbf{x}}: C^{j,\lambda}(\Omega) \to \mathbb{C}, \qquad f \mapsto \langle e^{\alpha}_{\mathbf{x}}, f \rangle := (\partial^{\alpha} f)(\mathbf{x}) ;$$

in particular, let us stress that

$$e^0_{\mathbf{x}} \equiv e_{\mathbf{x}}$$

is the usual evaluation map on $C^{0,\lambda}(\Omega)$. Clearly, $e_{\mathbf{x}}^{\alpha} \in (C^{j,\lambda}(\Omega))'$; so, we have the map

$$e^{\alpha}: \Omega \to C^{j,\lambda}(\Omega)', \qquad \mathbf{x} \mapsto e^{\alpha}_{\mathbf{x}}.$$
 (2.198)

Step 2 - Let $j \in \mathbb{N}$, $\lambda \in (0, 1)$ and consider a multi-index α of order $\leq j$. Then the map (2.198) is of class $C^{0,1}$ (i.e., locally Lipschitz, hence continuous) if $|\alpha| < j$ and of class $C^{0,\lambda}$ (hence continuous) if $|\alpha| = j$. For the sake of brevity, hereafter we put

$$\tilde{\lambda} := \begin{cases} 1 & \text{if } |\alpha| < j \\ \tilde{\lambda} := \lambda & \text{if } |\alpha| = j \end{cases}.$$

Let us consider any compact subset $K \subset \Omega$. If $f \in C^{j,\lambda}(\Omega)$ and $\mathbf{x}, \mathbf{y} \in K$ (with $\mathbf{y} \neq \mathbf{x}$) we have

$$\langle e_{\mathbf{y}}^{\alpha} - e_{\mathbf{x}}^{\alpha}, f \rangle = \partial^{\alpha} f(\mathbf{y}) - \partial^{\alpha} f(\mathbf{x}) = \frac{\partial^{\alpha} f(\mathbf{y}) - \partial^{\alpha} f(\mathbf{x})}{|\mathbf{y} - \mathbf{x}|^{\tilde{\lambda}}} |\mathbf{y} - \mathbf{x}|^{\tilde{\lambda}}$$

2.6. SCHRÖDINGER-TYPE OPERATORS: REGULARITY AND SELF-ADJOINTNESS.

that implies

$$|\langle e_{\mathbf{y}}^{\alpha} - e_{\mathbf{x}}^{\alpha}, f \rangle| \leq |f|_{K}^{\alpha, \tilde{\lambda}} |\mathbf{y} - \mathbf{x}|^{\tilde{\lambda}} \text{ for } \mathbf{x}, \mathbf{y} \in K, \quad |f|_{K}^{\alpha, \tilde{\lambda}} := \sup_{\mathbf{x}, \mathbf{y} \in K, \ \mathbf{y} \neq \mathbf{x}} \frac{|\partial^{\alpha} f(\mathbf{y}) - \partial^{\alpha} f(\mathbf{x})|}{|\mathbf{y} - \mathbf{z}|^{\tilde{\lambda}}} \quad (2.199)$$

(if $|\alpha| < j$, so that $\tilde{\lambda} = 1$, the finiteness of $|f|_K^{\alpha,1}$ follows from the fact that $\partial^{\alpha} f$ is a C^1 function). For any bounded subset \mathcal{B} of $C^{j,\lambda}(\Omega)$, let us put

$$M_{K,\mathcal{B}}^{\alpha,\tilde{\lambda}} := \sup_{f \in \mathcal{B}} |f|_K^{\alpha,\tilde{\lambda}} < +\infty$$

(the sup over \mathcal{B} is finite, because $f \mapsto |f|_{K}^{\alpha,\tilde{\lambda}}$ is a continuous seminorm on $C^{j,\lambda}(\Omega)$; compare with Eq. (2.14)). Returning to Eq. (2.199) we see that

$$|\langle e_{\mathbf{y}}^{\alpha} - e_{\mathbf{x}}^{\alpha}, f \rangle| \leq M_{K,\mathcal{B}}^{\alpha,\tilde{\lambda}} |\mathbf{y} - \mathbf{x}|^{\lambda} \quad \text{for } \mathbf{x}, \mathbf{y} \in K, f \in \mathcal{B} ;$$

therefore, if $p_{\mathcal{B}}$ is the seminorm on $(C^{j,\lambda}(\Omega))'$ defined by Eq. (2.197), we have

$$p_{\mathcal{B}}(e_{\mathbf{y}}^{\alpha} - e_{\mathbf{x}}^{\alpha}) \le M_{K,\mathcal{B}}^{\alpha,\tilde{\lambda}} |\mathbf{y} - \mathbf{x}|^{\tilde{\lambda}} \quad \text{for } \mathbf{x}, \mathbf{y} \in K$$

Since the above relation holds for any bounded subset $\mathcal{B} \subset C^{j,\lambda}(\Omega)$, we have proved that e^{α} is of class $C^{0,\tilde{\lambda}}$.

Step 3 - Let $j \in \mathbb{N}, \lambda \in (0, 1)$. The map

$$e: \Omega \to (C^{j,\lambda}(\Omega))', \qquad \mathbf{x} \mapsto e_{\mathbf{x}}$$

is of class $C^{j,\lambda}$; for each multi-index α of order $\leq j$ and each $\mathbf{x} \in \Omega$, the α -th partial derivative of the map e at \mathbf{x} is

$$(\partial^{\alpha} e)(\boldsymbol{x}) = e^{\alpha}_{\boldsymbol{x}} \ . \tag{2.200}$$

To exemplify the necessary arguments, hereafter we give the proof for j = 1; so, our purpose is to prove that

$$e: \Omega \to (C^{1,\lambda}(\Omega))', \qquad \mathbf{x} \mapsto e_{\mathbf{x}}$$

is of class $C^{1,\lambda}$ with partial derivatives

$$(\partial_i e)(\mathbf{x}) = e^i_{\mathbf{x}}$$
 $(i = 1, ..., d; \langle e^i_{\mathbf{x}}, f \rangle := \partial_i f(\mathbf{x}) \text{ for each } f \in C^{1,\lambda}(\Omega)).$

Due to Step 1, we already know that the maps $e^i : \Omega \to C^{1,\lambda}(\Omega)$, $\mathbf{x} \mapsto e^i_{\mathbf{x}}$ are of class $C^{0,\lambda}$; therefore, to get the thesis it suffices to prove the following, at each $\mathbf{x} \in \Omega$:

$$\frac{e_{\mathbf{y}} - e_{\mathbf{x}} - \sum_{i=1}^{d} (y^{i} - x^{i}) e_{\mathbf{x}}^{i}}{|\mathbf{y} - \mathbf{x}|} \to 0 \quad \text{in } (C^{1,\lambda}(\Omega))' \quad \text{for } \mathbf{y} \in \Omega, \, \mathbf{y} \to \mathbf{x} \;.$$
(2.201)

Let us fix $\mathbf{x} \in \Omega$; for $f \in C^{1,\lambda}(\Omega)$ and $\mathbf{y} \in \Omega$ we have

$$\langle e_{\mathbf{y}} - e_{\mathbf{x}} - \sum_{i=1}^{d} (y^{i} - x^{i}) e_{\mathbf{x}}^{i}, f \rangle = f(\mathbf{y}) - f(\mathbf{x}) - \sum_{i=1}^{d} (y^{i} - x^{i}) (\partial_{i} f)(\mathbf{x}) .$$
 (2.202)

To go on, we let us consider a radius $\rho > 0$ such that $\overline{B(\mathbf{x}, \rho)} \subset \Omega$. Then, for $\mathbf{y} \in \overline{B(\mathbf{x}, \rho)}$ we can write

$$f(\mathbf{y}) - f(\mathbf{x}) = \sum_{i=1}^{d} \left(y^{i} - x^{i} \right) \int_{0}^{1} d\tau \left(\partial_{i} f \right) \left(\mathbf{x} + \tau (\mathbf{y} - \mathbf{x}) \right) ;$$

inserting this result into Eq. (2.202) we easily obtain

$$\langle e_{\mathbf{y}} - e_{\mathbf{x}} - \sum_{i=1}^{d} (y^{i} - x^{i}) e_{\mathbf{x}}^{i}, f \rangle = \sum_{i=1}^{d} (y^{i} - x^{i}) \int_{0}^{1} d\tau \left[(\partial_{i} f) (\mathbf{x} + \tau(\mathbf{y} - \mathbf{x})) - (\partial_{i} f) (\mathbf{x}) \right]. \quad (2.203)$$

On the other hand, for $\mathbf{y} \in \overline{B(\mathbf{x}, \rho)}$ and $\tau \in [0, 1]$ we have

$$\left| (\partial_i f)(\mathbf{x} + \tau(\mathbf{y} - \mathbf{x})) - (\partial_i f)(\mathbf{x}) \right| \le |f|_{\mathbf{x},\rho}^{i,\lambda} |\mathbf{y} - \mathbf{x}|^{\lambda}, \quad |f|_{\mathbf{x},\rho}^{i,\lambda} := \sup_{\mathbf{z},\mathbf{w}\in\overline{B(\mathbf{x},\rho)}, \mathbf{z}\neq\mathbf{w}} \frac{|\partial_i f(\mathbf{z}) - \partial_i f(\mathbf{w})|}{|\mathbf{z} - \mathbf{w}|^{\lambda}}. \quad (2.204)$$

From here and from Eq. (2.203) we obtain

$$\begin{split} \left| \langle e_{\mathbf{y}} - e_{\mathbf{x}} - \sum_{i=1}^{d} (y^{i} - x^{i}) e_{\mathbf{x}}^{i}, f \rangle \right| &\leq \left(\sum_{i=1}^{d} |y^{i} - x^{i}| |f|_{\mathbf{x},\rho}^{i,\lambda} \right) |\mathbf{y} - \mathbf{x}|^{\lambda} \\ &\leq \sqrt{\sum_{i=1}^{d} (|f|_{\mathbf{x},\rho}^{i,\lambda})^{2}} |\mathbf{y} - \mathbf{x}|^{1+\lambda} \quad \text{for } f \in C^{1,\lambda}(\Omega), \, \mathbf{y} \in \overline{B(\mathbf{x},\rho)} \;. \end{split}$$

To go on, for any bounded subset \mathcal{B} of $C^{1,\lambda}(\Omega)$, let us put

$$M_{\mathbf{x},\rho}^{\mathcal{B},1,\lambda} := \sup_{f \in \mathcal{B}} \sqrt{\sum_{i=1}^{d} (|f|_{\mathbf{x},\rho}^{i,\lambda})^2} < +\infty .$$

Returning to Eq. (2.204) we see that

$$\left| \langle e_{\mathbf{y}} - e_{\mathbf{x}} - \sum_{i=1}^{d} (y^{i} - x^{i}) e_{\mathbf{x}}^{i}, f \rangle \right| \leq M_{\mathbf{x},\rho}^{\mathcal{B},1,\lambda} |\mathbf{y} - \mathbf{x}|^{1+\lambda} \quad \text{for } \mathbf{y} \in \overline{B(\mathbf{x},\rho)}, f \in \mathcal{B} ;$$

therefore, if $p_{\mathcal{B}}$ is the seminorm on $(C^{1,\lambda}(\Omega))'$ defined by Eq. (2.197), we have

$$p_{\mathcal{B}}\left(e_{\mathbf{y}} - e_{\mathbf{x}} - \sum_{i=1}^{d} (y^{i} - x^{i})e_{\mathbf{x}}^{i}\right) \le M_{\mathbf{x},\rho}^{\mathcal{B},1,\lambda} |\mathbf{y} - \mathbf{x}|^{1+\lambda} \quad \text{for } \mathbf{y} \in \overline{B(\mathbf{x},\rho)} .$$
(2.205)

2.6. SCHRÖDINGER-TYPE OPERATORS: REGULARITY AND SELF-ADJOINTNESS.

By the arbitrariness of the bounded subset $\mathcal{B} \subset C^{1,\lambda}(\Omega)$, this suffices to obtain the thesis (2.201).

Step 4 - Let $r \in \mathbb{R}$, $j \in \mathbb{N}$ and $\lambda \in (0, 1)$ be such that $r > j + d/2 + \sigma$. Then the map (2.195) $\delta : \Omega \to \mathcal{H}^{-r}$, $\mathbf{x} \mapsto \delta_{\mathbf{x}}$ is of class $C^{j,\lambda}$ and its derivatives are as in Eq. (2.196), for each multi-index α of order $\leq j$ and each $\mathbf{x} \in \Omega$. First of all let us recall the embedding $\mathcal{H}^r \hookrightarrow C^{j,\lambda}(\Omega)$; this induces a continuous linear map

$$\mathcal{R}^r : (C^{j,\lambda}(\Omega))' \to (\mathcal{H}^r)', \qquad \alpha \mapsto \mathcal{R}^r \alpha := \alpha \restriction \mathcal{H}^r .$$
(2.206)

Let as also recall that there is a Banach antilinear isomorphism $\mathcal{I}^r : \mathcal{H}^{-r} \to (\mathcal{H}^r)'$ such that $\langle \mathcal{I}^r g, f \rangle := \langle g | f \rangle$ for $g \in \mathcal{H}^{-r}, f \in \mathcal{H}^r$. Now, for any $\mathbf{x} \in \Omega, e_{\mathbf{x}} \in (C^{j,\lambda}(\Omega))'$ and $\delta_{\mathbf{x}} \in \mathcal{H}^{-r}$ are such that

$$\langle e_{\mathbf{x}}, f \rangle = f(\mathbf{x}) \text{ for } f \in C^{j,\lambda}(\Omega) , \quad \langle \delta_{\mathbf{x}} | f \rangle = f(\mathbf{x}) \text{ for } f \in \mathcal{H}^r$$

(recall Step 1 and Eq. (2.192)). Therefore $\mathcal{R}^r e_{\mathbf{x}} = \mathcal{I}^r \delta_{\mathbf{x}}$, i.e.,

$$\delta_{\mathbf{x}} = (\mathcal{I}^r)^{-1} \mathcal{R}^r e_{\mathbf{x}} . \tag{2.207}$$

The map $(\mathcal{I}^r)^{-1}\mathcal{R}^r : (C^{j,\lambda}(\Omega))' \to \mathcal{H}^r$ is antilinear and continuous, and $e : \mathbf{x} \in \Omega \mapsto e_{\mathbf{x}} \in (C^{j,\sigma}(\Omega))'$ is of class $C^{j,\lambda}$ according to Step 3; therefore $\delta : \mathbf{x} \in \Omega \mapsto \delta_{\mathbf{x}} \in \mathcal{H}^r$ is also of class $C^{j,\lambda}$. Moreover, for $|\alpha| \leq j$ and $\mathbf{x} \in \Omega$ we have

$$(\partial^{\alpha}\delta)(\mathbf{x}) = (\mathcal{I}^{r})^{-1}\mathcal{R}^{r}(\partial^{\alpha}e)(\mathbf{x}) = (\mathcal{I}^{r})^{-1}\mathcal{R}^{r}e^{\alpha}_{\mathbf{x}}$$
(2.208)

where the last equality is again due to Step 3. Summing up, for each $f \in \mathcal{H}^r$ we have

$$\langle (\partial^{\alpha} \delta)(\mathbf{x}) | f \rangle = \langle e_{\mathbf{x}}^{\alpha}, f \rangle = (\partial^{\alpha} f)(\mathbf{x}),$$

and comparing this result with the relation (2.193) $\langle \partial^{\alpha} \delta_{\mathbf{x}} | f \rangle = (-1)^{|\alpha|} (\partial^{\alpha} f)(\mathbf{x})$ we finally obtain Eq. (2.196), i.e., the thesis.

Extension of the previous results up to the boundary.

Let us remark that so far in this subsection we have always assumed \mathbf{x} to be a point in the interior of the open set Ω (see Eq. (2.191)). Nonetheless, it appears from the proofs of Propositions 2.46 and 2.48 that, making stronger regularity assumptions, the same results stated in these propositions continue to hold as well for points on the boundary $\partial\Omega$.

For example, let us make the stricter hypotheses (2.159), so that Ω is a bounded domain with boundary of class C^{∞} and $\mathcal{A} = -\Delta + V$ with $V \in C^{\infty}(\overline{\Omega})$; moreover, assume Dirichlet boundary conditions are imposed for the operator \mathcal{A} , which is also supposed to be strictly positive. Thus,

$$Dom(\mathcal{A}) \equiv \mathcal{D}_{\mathcal{A}} = \{ f \in H_0^1(\Omega) \mid (-\Delta + V) f \in L^2(\Omega) \} ,$$

$$\sigma(\mathcal{A}) \subset [\varepsilon, +\infty) \text{ for some } \varepsilon > 0 .$$
(2.209)

Throughout the present paragraph, **x** denotes any point in the closure $\overline{\Omega}$ of the domain Ω , that is

$$\mathbf{x} \in \overline{\Omega} = \Omega \cup \partial \Omega \ . \tag{2.210}$$

In this situation, we have the continous embeddings $\mathcal{H}^r \hookrightarrow C^{j,\lambda}(\overline{\Omega}) \hookrightarrow C^j(\overline{\Omega})$ for $r \in \mathbb{R}$, $j \in \mathbb{N}, \lambda \in (0, 1)$ s.t. $r > j + d/2 + \lambda$. This yields the following variants of Propositions 2.46 and 2.48, that are easily proved replacing $C^j(\Omega)$ with $C^j(\overline{\Omega})$, and $C^{j,\lambda}(\Omega)$ with $C^{j,\lambda}(\overline{\Omega})$.

Proposition 2.49. Under the assumptions (2.159) and (2.209), both statements i) and ii) of Proposition 2.46 hold for any $\mathbf{x} \in \overline{\Omega}$, allowing to define $\delta_{\mathbf{x}}$ and $\partial^{\alpha} \delta_{\mathbf{x}}$ even for $\mathbf{x} \in \partial \Omega$.

Proposition 2.50. Consider the assumptions (2.159) and (2.209); moreover, let $j \in \mathbb{N}$, $r \in \mathbb{R}$ be such that r > j + d/2. Then, the map

$$\delta: \Omega \to \mathcal{H}^{-r} \qquad \boldsymbol{x} \mapsto \delta_{\boldsymbol{x}} , \qquad (2.211)$$

is of class $C^{j}(\overline{\Omega}, \mathcal{H}^{-r})$; more precisely, it is of class $C^{j,\lambda}(\overline{\Omega}, \mathcal{H}^{-r})$ for each $\lambda \in (0,1)$ s.t. $r > j + d/2 + \lambda$. The continuous extension of (2.211) to $\overline{\Omega}$ is the map $\boldsymbol{x} \mapsto \delta_{\boldsymbol{x}}$ defined according to Proposition 2.49; furthermore, for any multi-index α of order $\leq j$, the (continuous extension to $\overline{\Omega}$ of the) partial derivative $\partial^{\alpha}\delta$ fulfills Eq. (2.196) at all points $\boldsymbol{x} \in \overline{\Omega}$.

Weak integrability of Dirac delta functions.

Let us return to the case where Ω is an arbitrary domain. The statement that follows refers to the notion of weak integrability for Banach valued functions (see the paragraph in Section 2.2; we refer, in particular, to Theorem 2.2).

Proposition 2.51. Let $r \in \mathbb{R}$, r > d/2 and $f \in L^2(\Omega)$. Then the map $\Omega \to \mathcal{H}^{-r}$, $\boldsymbol{x} \mapsto f(\boldsymbol{x}) \delta_{\boldsymbol{x}}$ is weakly measurable and integrable with respect to the Lebesgue measure $d\boldsymbol{x}$; moreover,

$$\int_{\Omega} d\boldsymbol{x} f(\boldsymbol{x}) \,\delta_{\boldsymbol{x}} = f \,\,. \tag{2.212}$$

Remark 2.14. The integral in Eq. (2.212) should produce an element of \mathcal{H}^{-r} ; in fact, $f \in L^2(\Omega) \equiv \mathcal{H}^0 \subset \mathcal{H}^{-r}$.

Proof. Considerations about weak measurability and weak integrability involve the dual space of \mathcal{H}^{-r} , which is antilinearly isomorphic to \mathcal{H}^r via the map $\mathcal{H}^r \to (\mathcal{H}^{-r})', g \mapsto \langle g | \rangle$ (as usual, $\langle | \rangle$ indicates the extension (2.103) of the inner product of $\mathcal{H} \equiv L^2(\Omega)$ to $\mathcal{H}^{(2)}$). Let us now fix $g \in \mathcal{H}^r$ arbitrarily. Then, for a.e. $\mathbf{x} \in \Omega$ we have $\langle g | f(\mathbf{x}) \delta_{\mathbf{x}} \rangle = f(\mathbf{x}) \overline{\langle \delta_{\mathbf{x}} | g \rangle} =$ $f(\mathbf{x}) \overline{g(\mathbf{x})}$, i.e., the map $\Omega \to \mathbb{C}, \mathbf{x} \mapsto \langle g | f(\mathbf{x}) \delta_{\mathbf{x}} \rangle$ coincides a.e. with $f \overline{g}$; this map is clearly measurable and integrable (since $f \in L^2(\Omega)$ and $g \in \mathcal{H}^r \hookrightarrow \mathcal{H} \equiv L^2(\Omega)$). Therefore, the map $\Omega \to \mathcal{H}^{-r}, \mathbf{x} \mapsto f(\mathbf{x}) \delta_{\mathbf{x}}$ is weakly measurable; moreover, for all $g \in \mathcal{H}^r$, it is $\int_{\Omega} d\mathbf{x} \langle g | f(\mathbf{x}) \delta_{\mathbf{x}} \rangle = \int_{\Omega} d\mathbf{x} f(\mathbf{x}) g(\mathbf{x}) = \langle g | f \rangle$. This means that $\mathbf{x} \to f(\mathbf{x}) \delta_{\mathbf{x}}$ is weakly integrable, with integral f.

Complex conjugate of Dirac delta functions.

Let us consider the general framework outlined in subsection 2.5 for an abstract Hilbert space \mathcal{H} with a conjugation \mathcal{J} . Of course, in the setting considered in the present section, there is a natural conjugation on the basic Hilbert space $\mathcal{H} = L^2(\Omega)$: namely, the complex conjugation

$$\mathcal{J}: L^2(\Omega) \to L^2(\Omega) , \qquad f \mapsto \mathcal{J}f := \overline{f} .$$
 (2.213)

Needless to say, in this case the projectors \mathcal{P}_+ and \mathcal{P}_- (see Eq. (2.68)) associated to \mathcal{J} are simply the maps associating to any given complex function $f: \Omega \to \mathbb{C}$ its real part $\Re f$ and its imaginary part $\Im f$ multiplied by the imaginary unit, respectively; so,

$$\mathcal{P}_{+}f = \Re f , \qquad \mathcal{P}_{-}f = i\,\Im f . \qquad (2.214)$$

Furthermore, it appears that \mathcal{J} commutes with the admissible operator \mathcal{A} associated to any given Schrödinger operator $A := -\Delta + V$ ($V : \Omega \to \mathbb{R}$); i.e., \mathcal{A} is \mathcal{J} -real or, briefly, *real*. Therefore, we can consider the corresponding extension $\mathcal{J} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$, defined according to Proposition 2.25.

There holds the following result concerning the Dirac delta, which will be useful for the physical applications to be discussed in the next chapter.

Lemma 2.52. Let $\mathbf{x} \in \Omega$. Then, for any multi-index α , the α -th derivative of the Dirac delta $\partial^{\alpha} \delta_{\mathbf{x}} (\in \mathcal{H}^{-r}, \text{ for } r > |\alpha| + d/2)$ is invariant under the extended complex conjugation $\mathcal{J} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$, i.e.,

$$\mathcal{J}\,\partial^{\alpha}\delta_{\boldsymbol{x}} = \partial^{\alpha}\delta_{\boldsymbol{x}} \,. \tag{2.215}$$

Proof. Let us choose $j \in \mathbb{N}$, $r \in \mathbb{R}$ such that $|\alpha| \leq j$ and r > j + d/2; then $\partial^{\alpha} \delta_{\mathbf{x}} \in \mathcal{H}^{-r}$. Let us recall that the extension \mathcal{J} can be characterized as follows: for each $g \in \mathcal{H}^{-r}$, $\mathcal{J}g$ is the unique element of \mathcal{H}^{-r} such that $\langle \mathcal{J}g|f \rangle = \overline{\langle g|\mathcal{J}f \rangle}$ for all $f \in \mathcal{H}^r$. On the other hand, for any $f \in \mathcal{H}^r \hookrightarrow C^j(\Omega)$, one has

$$\langle \mathcal{J}\partial^{\alpha}\delta_{\mathbf{x}}|f\rangle = \overline{\langle \partial^{\alpha}\delta_{\mathbf{x}}|\mathcal{J}f\rangle} = \overline{(-1)^{|\alpha|}\partial^{\alpha}\overline{f}(\mathbf{x})} = (-1)^{|\alpha|}\partial^{\alpha}f(\mathbf{x}) = \langle \partial^{\alpha}\delta_{\mathbf{x}}|f\rangle ,$$

which suffices to infer the thesis.

Remark 2.15. Lemma 2.52 continues to hold as well for points on the boundary $\partial\Omega$ whenever the stricter assumptions (2.159) and (2.209) are met.

2.7 Integral kernels.

Let us consider the framework of the previous section; in particular, let $\mathcal{D}_{\mathcal{A}} \subset L^2(\Omega)$ denote any admissible domain for the differential operator $A = -\Delta + V$, and consider the spaces $(\mathcal{H}^r, || \, ||_r)$ associated to the admissible operator \mathcal{A} (see Proposition 2.4). Next, let us consider a linear operator

$$\mathcal{B}: \operatorname{Dom}(\mathcal{B}) \subset \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty} ; \qquad (2.216)$$

in the sequel of this work we will be mainly interested in the case where \mathcal{B} fulfills the following condition, for some $j \in \mathbb{N}$:

there exists $\vartheta \in (0, +\infty)$ such that, for all $j_1, j_2 \in \mathbb{N}$ with $j_1 + j_2 \leq j$, \mathcal{B} maps continuously $\mathcal{H}^{-(j_2+d/2+\vartheta)}$ to $\mathcal{H}^{j_1+d/2+\vartheta}$ (2.217)

(²³). We refer to the above equation as condition $(2.217)_j$. Remark 2.16. i) Since $\mathcal{H}^{j-j_2+d/2+\vartheta} \hookrightarrow \mathcal{H}^{j_1+d/2+\vartheta}$ for all j_1, j_2 as in Eq. (2.217)_j, it appears that this condition is equivalent to the following one:

there exists
$$\vartheta \in (0, +\infty)$$
 such that, for all $j_2 \in \mathbb{N}$ with $j_2 \leq j$,
 \mathcal{B} maps continuously $\mathcal{H}^{-(j_2+d/2+\vartheta)}$ to $\mathcal{H}^{j-j_2+d/2+\vartheta}$. (2.218)

ii) In particular, condition $(2.217)_0$ reads

there exists
$$\vartheta \in (0, +\infty)$$
 such that
 \mathcal{B} maps continuously $\mathcal{H}^{-(d/2+\vartheta)}$ to $\mathcal{H}^{d/2+\vartheta}$. (2.219)

iii) Of course, $(2.217)_j$ implies $(2.217)_i$ for all $i \in \{0, ..., j\}$.

Definition 2.53. If the condition $(2.217)_0$ holds, the *integral kernel* associated to the operator \mathcal{B} is the map

$$\mathcal{B}(\ ,\):\Omega\times\Omega\to\mathbb{C}\ ,\qquad (\mathbf{x},\mathbf{y})\mapsto\mathcal{B}(\mathbf{x},\mathbf{y}):=\langle\delta_{\mathbf{x}}|\mathcal{B}\,\delta_{\mathbf{y}}\rangle \tag{2.220}$$

(where $\langle | \rangle$ indicates the extension (2.103) of the inner product on \mathcal{H} . Note that, for ϑ as in $(2.217)_0$, one has $\delta_{\mathbf{x}}, \delta_{\mathbf{y}} \in \mathcal{H}^{-(d/2+\vartheta)}$ and $\mathcal{B}\delta_{\mathbf{y}} \in \mathcal{H}^{d/2+\vartheta}$).

Shortly afterwards, we will prove that the function (2.220) is continuous on $\Omega \times \Omega$, and even more regular if \mathcal{B} fulfills condition $(2.217)_j$ for some j > 0 ($j \in \mathbb{N}$). The following Lemma will be useful for the sequel.

Lemma 2.54. Suppose $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ is a measurable function fulfilling condition (2.90) (i.e., $\sup_{\lambda \in \sigma(\mathcal{A})} \lambda^b |\phi(\lambda)| < +\infty$) for some $b \in \mathbb{R}$. If b > (j+d)/2 for some $j \in \mathbb{N}$, then the operator $\mathcal{B} := \phi(\mathcal{A})$ (extended to $\mathcal{H}^{-\infty}$) fulfills condition (2.217)_j.

Proof. The thesis follows by an elementary application of Proposition 2.12. \Box

In the forthcoming subsection 2.7, we present a series of results describing some notable features of the integral kernels associated to a suitable class of operators. For brevity we choose to discuss these results only in the general setting corresponding to an arbitrary domain Ω with V a smooth potential on it. Nonetheless, it appears from the related proofs that, whenever the stricter hypotheses (2.159) are fulfilled and suitable boundary

²³Let us stress that, within the general setting considered within the present work, there holds $\mathcal{H}^{j_1+d/2+\vartheta} \hookrightarrow C^{j_1}(\Omega)$ (see Corollary 2.43); on the other hand, under the stricter assumptions (2.159), one can infer the stronger result $\mathcal{H}^{j_1+d/2+\vartheta} \hookrightarrow C^{j_1}(\overline{\Omega})$ (see Corollary 2.45).

conditions (see, e.g., Eq. (2.209)) are prescribed, the results to be presented in the sequel can be easily generalized to describe the behaviour of the integral kernels up to the boundary. We refer to the subsequent subsection 2.7 for more details and further results related to this case.

General results on integral kernels.

Let us present several results for the integral kernels associated to operators fulfilling condition $(2.217)_j$ (for some $j \in \mathbb{N}$). These results apply, in particular, to operators of the form $\mathcal{B} = \phi(\mathcal{A})$ as in Lemma 2.54.

Lemma 2.55. Let $(2.217)_j$ hold for some $j \in \mathbb{N}$. Then, $\mathcal{B}(,) \in C^j(\Omega \times \Omega)$; moreover, for any pair of multi-indices α, β such that $|\alpha| + |\beta| \leq j$ and for all $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, one has

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{B}(\boldsymbol{x}, \boldsymbol{y}) = (-1)^{|\alpha| + |\beta|} \langle \partial^{\alpha} \delta_{\boldsymbol{x}} | \mathcal{B} \, \partial^{\beta} \delta_{\boldsymbol{y}} \rangle \tag{2.221}$$

(where ∂_1 and ∂_2 represent derivatives with respect to the first and second argument, respectively).

Proof. First notice that each derivative of order $\leq j$ of the kernel $\mathcal{B}(,)$ involves (in an arbitrary order) differentiating α_i times with respect to x^i and β_i times with respect to y^i (for i = 1, ..., d), where $\alpha = (\alpha_1, ..., \alpha_d)$ and $\beta = (\beta_1, ..., \beta_d)$ are such that $|\alpha| + |\beta| \leq j$. We generically write $\partial^{\gamma} \mathcal{B}(,)$ for such a derivative. The thesis holds if we can show that $\partial^{\gamma} \mathcal{B}(,) : \Omega \times \Omega \to \mathbb{C}$ exists, is continuous and has the explicit expression in the right-hand side of Eq. (2.221).

To this purpose, let us recall that both the extended inner product $\langle | \rangle$ and the (linear) operator \mathcal{B} are continuous on the corresponding domains of definition; so, by linearity $\partial^{\gamma} \mathcal{B}(,)$ exists and, for all $(\mathbf{x}, \mathbf{y}) \in \Omega \times \Omega$, we can write

$$\partial^{\gamma} \mathcal{B}(\mathbf{x}, \mathbf{y}) = \langle (\partial^{\alpha} \delta)(\mathbf{x}) | \mathcal{B}(\partial^{\beta} \delta)(\mathbf{y}) \rangle .$$
(2.222)

In consequence of this, the function $\partial^{\gamma} \mathcal{B}(\ ,\)$ can be viewed as the composition of the maps

$$\begin{split} \delta^{(\alpha,\beta)} &: \Omega \times \Omega \to \mathcal{H}^{-(|\alpha|+d/2+\vartheta)} \times \mathcal{H}^{-(|\beta|+d/2+\vartheta)} , \qquad (\mathbf{x},\mathbf{y}) \mapsto ((\partial^{\alpha}\delta)(\mathbf{x}), (\partial^{\beta}\delta)(\mathbf{y})) , \\ \mathfrak{B}^{(\alpha,\beta)} &: \mathcal{H}^{-(|\alpha|+d/2+\vartheta)} \times \mathcal{H}^{-(|\beta|+d/2+\vartheta)} \to \mathbb{C} , \qquad (f,g) \mapsto \langle f | \mathcal{B}g \rangle \end{split}$$

for some $\vartheta \in (0, +\infty)$ such that Eq. (2.217) holds $({}^{24})$. On the one hand, since the map $\Omega \to \mathcal{H}^{-r}$, $\mathbf{x} \mapsto \delta_{\mathbf{x}}$ is of class C^{j_0} for all $r > j_0 + d/2$ (see Propositions 2.46 and 2.48), it appears that $\delta^{(\alpha,\beta)}$ is continuous. On the other hand, let us recall that the operator $\mathcal{B}: \mathcal{H}^{-(|\beta|+d/2+\vartheta)} \to \mathcal{H}^{|\alpha|+d/2+\vartheta}$ and the bilinear map $\langle | \rangle : \mathcal{H}^{-(|\alpha|+d/2+\vartheta)} \times \mathcal{H}^{|\alpha|+d/2+\vartheta} \to \mathbb{C}$ are both continuous; this suffices to infer that $\mathfrak{B}^{(\alpha,\beta)}$ is continuous as well. Summing

²⁴Of course, $\mathcal{H}^{-(|\alpha|+d/2+\vartheta)} \times \mathcal{H}^{-(|\beta|+d/2+\vartheta)}$ is equipped with the product of the Banach topologies of its factors.

up, the above considerations show that $\partial^{\gamma} \mathcal{B}(,)$ is continuous since it is given by the composition of continuous functions.

To conclude, let us notice that Eq. (2.222) and the identity (2.196) for the derivatives $(\partial^{\alpha}\delta)(\mathbf{x}), (\partial^{\beta}\delta)(\mathbf{y})$ yield for $\partial^{\gamma}\mathcal{B}(,)$ the expression in the right-hand side of Eq. (2.221), thus proving the thesis.

Proposition 2.56. Let $(2.217)_j$ hold for some $j \in \mathbb{N}$ and let α, β be any two multi-indices, each one of order $\leq j$.

i) Let $f \in L^2(\Omega)$ and consider the (C^j) map $\mathcal{B}f : \Omega \to \mathbb{C}$. Then, for any $\mathbf{x} \in \Omega$, the map $\Omega \to \mathbb{C}$, $\mathbf{y} \mapsto \partial^{\alpha} \mathcal{B}(\mathbf{x}, \mathbf{y})$ is integrable and

$$(\partial^{\alpha} \mathcal{B} f)(\boldsymbol{x}) = \int_{\Omega} d\boldsymbol{y} \, (\partial_{1}^{\alpha} \mathcal{B})(\boldsymbol{x}, \boldsymbol{y}) \, f(\boldsymbol{y}) ; \qquad (2.223)$$

in particular, if $\alpha = \emptyset$, this gives

$$(\mathcal{B}f)(\boldsymbol{x}) = \int_{\Omega} d\boldsymbol{y} \, \mathcal{B}(\boldsymbol{x}, \boldsymbol{y}) \, f(\boldsymbol{y}) \, . \tag{2.224}$$

ii) For any fixed $\boldsymbol{x} \in \Omega$, consider the (continuous) map $\partial_1^{\alpha} \mathcal{B}(\boldsymbol{x},): \Omega \to \mathbb{C}, \ \boldsymbol{y} \mapsto \partial_1^{\alpha} \mathcal{B}(\boldsymbol{x}, \boldsymbol{y});$ then $\partial_1^{\alpha} \mathcal{B}(\boldsymbol{x},) \in L^2(\Omega).$

iii) For any fixed $\mathbf{y} \in \Omega$, consider the (continuous) map $\partial_2^\beta \mathcal{B}(\mathbf{x}, \mathbf{y}) : \Omega \to \mathbb{C}, \mathbf{x} \mapsto \partial_2^\beta \mathcal{B}(\mathbf{x}, \mathbf{y});$ this is given by

$$\partial_2^\beta \mathcal{B}(\ , \boldsymbol{y}) = (-1)^{|\beta|} \, \mathcal{B} \, \partial^\beta \delta_{\boldsymbol{y}} \,. \tag{2.225}$$

Moreover, there exists $\vartheta \in (0, +\infty)$ such that $\partial_2^\beta \mathcal{B}(\cdot, y) \in \mathcal{H}^{j-|\beta|+d/2+\vartheta}$.

Remark 2.17. The last statement in item iii) of the above proposition allows to infer, in particular, that $\partial_2^{\beta} \mathcal{B}(, \mathbf{y}) \in L^2(\Omega)$ and $\partial_2^{\beta} \mathcal{B}(, \mathbf{y}) \in C^{j-|\beta|}(\Omega)$. Furthemore, if $j - |\beta| + d/2 + \vartheta \geq 2$, there holds $\partial_2^{\beta} \mathcal{B}(, \mathbf{y}) \in \mathcal{H}^2 = \text{Dom}(\mathcal{A})$, so that $\partial_2^{\beta} \mathcal{B}(, \mathbf{y})$ fulfills (at least in weak sense) the possible boundary conditions included in the definition of \mathcal{A} (see, e.g., Eq. (2.209)); we will return to this topic in subsection 2.7, where stricter assumptions are made for the domain Ω and for the potential V.

Proof. In the sequel we always assume $\vartheta \in (0, +\infty)$ and $j \in \mathbb{N}$ to be as in Eq. (2.217). i) Let $f \in L^2(\Omega)$. Due to Proposition 2.51 (here employed with $r = d/2 + \vartheta > d/2$), the map $\Omega \to \mathcal{H}^{-(d/2+\vartheta)}$, $\mathbf{y} \mapsto f(\mathbf{y}) \, \delta_{\mathbf{y}}$ is weakly integrable and $f = \int_{\Omega} d\mathbf{y} f(\mathbf{y}) \delta_{\mathbf{y}}$. From here and from the continuity of \mathcal{B} between $\mathcal{H}^{-(d/2+\vartheta)}$ and $\mathcal{H}^{j+d/2+\vartheta}$, we infer the weak integrability of the map $\Omega \to \mathcal{H}^{j+d/2+\vartheta}$, $\mathbf{y} \mapsto f(\mathbf{y}) \mathcal{B} \delta_{\mathbf{y}}$ and the relation $\mathcal{B}f = \int_{\Omega} d\mathbf{y} f(\mathbf{y}) \mathcal{B} \delta_{\mathbf{y}}$ proved in Theorem 2.2. Now, fix a point $\mathbf{x} \in \Omega$ and a multi-index α of order $\leq j$. Since $\langle \partial^{\alpha} \delta_{\mathbf{x}} | \rangle$ is a continuous linear form on $\mathcal{H}^{j+d/2+\vartheta}$, it follows that the map $\Omega \to \mathbb{C}$, $\mathbf{y} \mapsto f(\mathbf{y}) \langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B} \delta_{\mathbf{y}} \rangle$ is integrable, and $\langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B}f \rangle = \int_{\Omega} d\mathbf{y} f(\mathbf{y}) \langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B} \delta_{\mathbf{y}} \rangle$. Making explicit the definitions of $\partial^{\alpha} \delta_{\mathbf{x}}$ and $\partial_1^{\alpha} \mathcal{B}(,)$, we can say that the map $\Omega \to \mathbb{C}$, $\mathbf{y} \mapsto \partial_1^{\alpha} \mathcal{B}(\mathbf{x}, \mathbf{y}) f(\mathbf{y})$ is integrable, and that Eq. (2.224) holds.

ii) From item i) we know that $\partial_1^{\alpha} \mathcal{B}(\mathbf{x}, f)$ is integrable for each $f \in L^2(\Omega)$. The considerations in the proof of item i) also show that the map $L^2(\Omega) \to \mathbb{C}, f \mapsto \int_{\Omega} d\mathbf{y} \partial_1^{\alpha} \mathcal{B}(\mathbf{x}, \mathbf{y}) f(\mathbf{y})$ is continuous (because $\mathcal{B} \upharpoonright L^2(\Omega) : L^2(\Omega) \hookrightarrow \mathcal{H}^{-(d/2+\vartheta)} \to \mathcal{H}^{j+d/2+\vartheta}$ is continuous); so, $\partial_1^{\alpha} \mathcal{B}(\mathbf{x}, \cdot) \in L^2(\Omega)$ due to the Riesz representation theorem.

iii) First recall that $\partial^{\beta} \delta_{\mathbf{y}} \in \mathcal{H}^{-(|\beta|+d/2+\vartheta)}$ and $\mathcal{B} : \mathcal{H}^{-(|\beta|+d/2+\vartheta)} \to \mathcal{H}^{j-|\beta|+d/2+\vartheta}$, so that $\mathcal{B}\partial^{\beta} \delta_{\mathbf{y}} \in \mathcal{H}^{j-|\beta|+d/2+\vartheta}$. On the other hand, due to Eq. (2.221) and to the definition of $\delta_{\mathbf{x}}$, for all $\mathbf{x} \in \Omega$ we have $\partial_{2}^{\beta} \mathcal{B}(\mathbf{x}, \mathbf{y}) = (-1)^{|\beta|} \langle \delta_{\mathbf{x}} | \mathcal{B} \partial^{\beta} \delta_{\mathbf{y}} \rangle = (-1)^{|\beta|} (\mathcal{B} \partial^{\beta} \delta_{\mathbf{y}})(\mathbf{x})$, whence the thesis (2.225).

Corollary 2.57. Let \mathcal{B} fulfill the condition $(2.217)_j$ for some $j \in \mathbb{N}$ and assume $\mathcal{B}_{\mathcal{H}} := (\mathcal{B} \upharpoonright \mathcal{H}) : \mathcal{H} \to \mathcal{H}$ to be of Hilbert-Schmidt type, i.e., $\mathcal{B}_{\mathcal{H}} \in \mathfrak{B}_2(\mathcal{H})$ (²⁵); moreover, let $\mathcal{B}_{\mathcal{H}}(,)$ and $\mathcal{B}(,)$ denote, respectively, the corresponding Hilbert-Schmidt kernel and the integral kernel introduced in Eq. (2.220). Then, there holds

$$\mathcal{B}(\boldsymbol{x}, \boldsymbol{y}) = \mathcal{B}_{\mathcal{H}}(\boldsymbol{x}, \boldsymbol{y}) \qquad for \ a.e. \ \boldsymbol{x}, \boldsymbol{y} \in \Omega \ . \tag{2.226}$$

Proof. First of all, let us recall that the Hilbert-Schmidt kernel $\mathcal{B}_{\mathcal{H}}(,) \in L^2(\Omega \times \Omega)$ of $\mathcal{B}_{\mathcal{H}}$ is the unique function such that, for all $f \in \mathcal{H}$, there holds (see Eq. (2.62))

$$(\mathcal{B}_{\mathcal{H}}f)(\mathbf{x}) = \int_{\Omega} d\mathbf{y} \ \mathcal{B}_{\mathcal{H}}(\mathbf{x}, \mathbf{y}) f(\mathbf{y}) \qquad \text{(for a.e. } \mathbf{x} \in \Omega) \ .$$

Then, the thesis follows easily from the above relation and from Eq. (2.224) of Proposition 2.56, by the arbitrariness of $f \in \mathcal{H}$ in both these identities.

Remark 2.18. The above Corollary states that the usual notion of integral kernel for Hilbert-Schmidt operators and the different definition we introduced previously (using the Dirac delta elements $\delta_{\mathbf{x}}, \delta_{\mathbf{y}} \in \mathcal{H}^{-r}$) do in fact coincide when both are well-defined.

Proposition 2.58. Let \mathcal{B} be a linear continuous operator fulfilling the assumption $(2.217)_j$ for some $j \in \mathbb{N}$ and some $\vartheta \in (0, +\infty)$. Consider a linear operator $\mathcal{C} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ and assume that, for all $j_1 \in \mathbb{N}$ with $j_1 \leq j$, there exists $\vartheta' \in (0, +\infty)$ such that \mathcal{C} sends continuously $\mathcal{H}^{j_1+d/2+\vartheta}$ into $\mathcal{H}^{j_1+d/2+\vartheta'}$. Then, for any multi-index β of order $\leq j$ and for all $\mathbf{y} \in \Omega$, there holds

$$\mathcal{C}(\partial_2^\beta \mathcal{B}(\ ,\boldsymbol{y})) = \partial_2^\beta (\mathcal{CB})(\ ,\boldsymbol{y})$$
(2.227)

(this identity is meant to hold in $\mathcal{H}^{j-|\beta|+d/2+\vartheta'} \hookrightarrow C^{j-|\beta|}(\Omega)$).

Proof. Eq. (2.227) can be formally derived by the following chain of equalities

$$\mathcal{C}(\partial_2^{\beta}\mathcal{B}(\mathbf{y})) = (-1)^{|\beta|} \mathcal{C}\mathcal{B} \,\partial^{\beta} \delta_{\mathbf{y}} = \partial_2^{\beta} (\mathcal{C}\mathcal{B})(\mathbf{y}) + \mathcal{C}(\partial_2^{\beta}\mathcal{B}(\mathbf{y}))$$

where the fundamental relation (2.225) has been used in both passages (for $|\beta| \leq j_2$ and $j_1 + j_2 \leq j$). More precisely, if $\vartheta' > \vartheta$ one must consider the integral kernels corresponding to the continuous operators $\mathcal{B} : \mathcal{H}^{-j_2+d/2+\vartheta} \to \mathcal{H}^{j_1+d/2+\vartheta}$ and $\mathcal{CB} : \mathcal{H}^{-(j_2+d/2+\vartheta)}$

²⁵Here, we are referring to the general theory reviewed in the final paragraph of Section 2.3, on page 24.

 $\rightarrow (\mathcal{H}^{j_1+d/2+\vartheta'} \hookrightarrow) \mathcal{H}^{j_1+d/2+\vartheta}; \text{ on the other hand, if } \vartheta' < \vartheta \text{ one must use the identity (2.225)}$ for the restrictions $(\mathcal{B} \upharpoonright \mathcal{H}^{-j_2+d/2+\vartheta'}) : \mathcal{H}^{-(j_2+d/2+\vartheta')}(\hookrightarrow \mathcal{H}^{-(j_2+d/2+\vartheta)}) \rightarrow (\mathcal{H}^{j_1+d/2+\vartheta} \hookrightarrow)$ $(\mathcal{H}^{j_1+d/2+\vartheta'}) : \mathcal{H}^{-(j_2+d/2+\vartheta')} \rightarrow \mathcal{H}^{j_1+d/2+\vartheta'}.$

From now on, given any operator $\mathcal{B}: \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ fulfilling condition $(2.217)_j$ for some $j \in \mathbb{N}$ and some $\vartheta \in (0, +\infty)$, we always consider its restriction (indicated with the same symbol) $\mathcal{B} \equiv \mathcal{B} \upharpoonright \mathcal{H}^{-(j+d/2+\vartheta)}$; this restriction sends continuously $\mathcal{H}^{-(j+d/2+\vartheta)}$ into $\mathcal{H}^{d/2+\vartheta}$. Then, following the general considerations of subsection 2.5, we can consider the (Banach) adjoint of \mathcal{B} , i.e.,

$$\mathcal{B}^*: \mathcal{H}^{-(j+d/2+\vartheta)} \to \mathcal{H}^{d/2+\vartheta} .$$
(2.228)

The forthcoming Propositions 2.59 and 2.60 refer to this operator and to its kernel.

Proposition 2.59. Let $(2.217)_j$ hold for some $j \in \mathbb{N}$; then there hold the following results. i) The adjoint operator \mathcal{B}^* (see Eq. (2.228)) also fulfills the condition $(2.217)_j$; therefore, Lemma 2.55 and Proposition 2.56 continue to hold with \mathcal{B} replaced by \mathcal{B}^* .

ii) Let α, β be any pair of multi-indices with $|\alpha| + |\beta| \leq j$ and consider the kernels $\mathcal{B}(,)$ and $\mathcal{B}^*(,)$; then, for all $\mathbf{x}, \mathbf{y} \in \Omega$, there holds

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{B}^*(\boldsymbol{x}, \boldsymbol{y}) = \overline{\partial_1^{\beta} \partial_2^{\alpha} \mathcal{B}(\boldsymbol{y}, \boldsymbol{x})} .$$
(2.229)

Proof. i) Let ϑ be as in Eq. (2.217); for any $j_1, j_2 \in \mathbb{N}$ with $j_1 + j_2 \leq j$, we can consider the restrictions $\mathcal{B} \upharpoonright \mathcal{H}^{-(j_2+d/2+\vartheta)}$ sending continuously $\mathcal{H}^{-(j_2+d/2+\vartheta)}$ into $\mathcal{H}^{j_1+d/2+\vartheta}$. By definition, the corresponding Banch adjoints $(\mathcal{B} \upharpoonright \mathcal{H}^{-(j_2+d/2+\vartheta)})^*$ send $\mathcal{H}^{-(j_1+d/2+\vartheta)}$ into $\mathcal{H}^{j_2+d/2+\vartheta}$; moreover, by construction, these maps coincide with the restrictions of the Banach adjoint (2.228) $\mathcal{B}^* \upharpoonright \mathcal{H}^{-(j_1+d/2+\vartheta)}$. This suffices to infer the thesis.

ii) As pointed out above, due to statement i) in the present proposition, Lemma 2.55 also holds for the kernel $\mathcal{B}^*(\ ,\)$; in particular, the general identity (2.221) implies both $\partial_1^{\alpha} \partial_2^{\beta} \mathcal{B}^*(\mathbf{x}, \mathbf{y}) = (-1)^{|\alpha|+|\beta|} \langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B}^* \partial^{\beta} \delta_{\mathbf{y}} \rangle$ and $\partial_1^{\beta} \partial_2^{\alpha} \mathcal{B}(\mathbf{y}, \mathbf{x}) = (-1)^{|\alpha|+|\beta|} \langle \partial^{\beta} \delta_{\mathbf{y}} | \mathcal{B} \partial^{\alpha} \delta_{\mathbf{x}} \rangle$. Next, let us recall that the bilinear form $\langle | \rangle$ is hermitian, so that $\langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B}^* \partial^{\beta} \delta_{\mathbf{y}} \rangle = \overline{\langle \mathcal{B}^* \partial^{\beta} \delta_{\mathbf{y}} | \partial^{\alpha} \delta_{\mathbf{x}} \rangle}$. The thesis follows noting that the basic relation (2.116) for the adjoint operator implies $\langle \mathcal{B}^* \partial^{\beta} \delta_{\mathbf{y}} | \partial^{\alpha} \delta_{\mathbf{x}} \rangle = \langle \partial^{\beta} \delta_{\mathbf{y}} | \mathcal{B} \partial^{\alpha} \delta_{\mathbf{x}} \rangle$.

Proposition 2.60. Let \mathcal{B} and \mathcal{C} be two operators both fulfilling the assumption $(2.217)_j$ for some given $j \in \mathbb{N}$, and consider the corresponding integral kernels. Then, for any pair of multi-indices α, β with $|\alpha| + |\beta| \leq j$ and for all $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, there hold the following identities:

$$\langle \partial_2^{\alpha} \mathcal{B}(, \boldsymbol{x}) | \partial_2^{\beta} \mathcal{C}(, \boldsymbol{y}) \rangle = \partial_1^{\alpha} \partial_2^{\beta} (\mathcal{B}^* \mathcal{C})(\boldsymbol{x}, \boldsymbol{y}) , \qquad (2.230)$$

$$\langle \partial_1^{\alpha} \mathcal{B}(\boldsymbol{x},) | \partial_1^{\beta} \mathcal{C}(\boldsymbol{y},) \rangle = \partial_2^{\alpha} \partial_1^{\beta} (\mathcal{C} \mathcal{B}^*) (\boldsymbol{y}, \boldsymbol{x}) .$$
 (2.231)

Remark 2.19. i) Due to Proposition 2.56, the functions $\partial_2^{\alpha} \mathcal{B}(, \mathbf{x}), \ \partial_1^{\alpha} \mathcal{B}(\mathbf{x},), \ \partial_2^{\alpha} \mathcal{C}(, \mathbf{x}), \ \partial_1^{\alpha} \mathcal{C}(\mathbf{x},)$ (for any fixed $\mathbf{x} \in \Omega$) all belong to $L^2(\Omega)$; so, the expressions $\langle | \rangle$ in the left-hand sides of Eq.s (2.230) (2.231) can both to be intended as usual inner products in $L^2(\Omega)$. ii) Let us point out that the expressions in the right-hand sides of Eq.s (2.230) (2.231) does in fact make sense under the weaker assumption that the operator $\mathcal{B}^*\mathcal{C}$ (or \mathcal{CB}^*)

2.7. INTEGRAL KERNELS.

possesses the property $(2.217)_j$. This fact can be used to give meaning to the left-hand sides of the cited equations also for less regular operators \mathcal{B}, \mathcal{C} ; we will return on this topic in the following (see Remark 2.20).

Proof. Let us first prove Eq. (2.230); to this purpose, note that due to item iii) of Proposition 2.56 there holds

$$\langle \partial_2^{\alpha} \mathcal{B}(\mathbf{x}) | \partial_2^{\beta} \mathcal{C}(\mathbf{y}) \rangle = (-1)^{|\alpha| + |\beta|} \langle \mathcal{B} \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{C} \partial^{\beta} \delta_{\mathbf{y}} \rangle .$$
(2.232)

Now, consider the assumption $(2.217)_j$ for \mathcal{B} and \mathcal{C} ; for any $i_1, i_2, j_1, j_2 \in \mathbb{N}$ with $i_1 + i_2 \leq j$ and $j_1 + j_2 \leq j$, this assumption grants the existence of some $\vartheta \in (0, +\infty)$ such that the maps $\mathcal{B} : \mathcal{H}^{-(i_2+d/2+\vartheta)} \to \mathcal{H}^{i_1+d/2+\vartheta} (\hookrightarrow \mathcal{H}^{-(j_1+d/2+\vartheta)})$ and $\mathcal{C} : \mathcal{H}^{-(j_2+d/2+\vartheta)} \to \mathcal{H}^{j_1+d/2+\vartheta}$ $(\hookrightarrow \mathcal{H}^{-(j_2+d/2+\vartheta)})$ are both continuous (²⁶).

Due to the above considerations, the expression $\langle | \rangle$ in the right-hand side of Eq. (2.232) can be interpreted in terms of the extension (2.103) of the inner product on \mathcal{H} (see Proposition 2.16) acting on $\mathcal{H}^{-(j_1+d/2+\vartheta)} \times \mathcal{H}^{j_1+d/2+\vartheta}$. Then, recalling that $\mathcal{B} = \mathcal{B}^{**} \equiv (\mathcal{B}^*)^*$ (see item ii) of Lemma 2.22) and using the basic identity (2.116) for adjoint operators, it follows that

$$\langle \mathcal{B}\partial^{\alpha}\delta_{\mathbf{x}}|\mathcal{C}\,\partial^{\beta}\delta_{\mathbf{y}}\rangle = \langle (\mathcal{B}^{*})^{*}\partial^{\alpha}\delta_{\mathbf{x}}|\mathcal{C}\,\partial^{\beta}\delta_{\mathbf{y}}\rangle = \langle \partial^{\alpha}\delta_{\mathbf{x}}|\mathcal{B}^{*}\mathcal{C}\,\partial^{\beta}\delta_{\mathbf{y}}\rangle \ . \tag{2.233}$$

Next, note that $\mathcal{B}^*\mathcal{C} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ also fulfills the assumption $(2.217)_j$: in fact, $\mathcal{B}^*\mathcal{C}$ sends continuously $\mathcal{H}^{-(j_2+d/2+\vartheta)}$ into $\mathcal{H}^{j_1+d/2+\vartheta}$ for all $j_1 + j_2 \leq j$ (²⁷). So, the last expression in Eq. (2.233) can be reformulated in terms of integral kernels giving

$$\langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B}^{*} \mathcal{C} \, \partial^{\beta} \delta_{\mathbf{y}} \rangle = (-1)^{|\alpha| + |\beta|} \partial_{1}^{\alpha} \partial_{2}^{\beta} \, (\mathcal{B}^{*} \mathcal{C})(\mathbf{x}, \mathbf{y}) \,. \tag{2.234}$$

Summing up, Eq.s (2.232-2.234) yield Eq. (2.230).

In order to prove Eq. (2.231) let us first notice that, explicitating the inner product in $L^2(\Omega)$ and using the identity (2.229) of Proposition 2.59, one infers

$$\langle \partial_1^{\alpha} \mathcal{B}(\mathbf{x}, \) | \partial_1^{\beta} \mathcal{C}(\mathbf{y}, \) \rangle = \int_{\Omega} d\mathbf{z} \ \overline{\partial_1^{\alpha} \mathcal{B}(\mathbf{x}, \mathbf{z})} \ \partial_1^{\beta} \mathcal{C}(\mathbf{y}, \mathbf{z}) = \int_{\Omega} d\mathbf{z} \ \partial_2^{\alpha} \mathcal{B}^*(\mathbf{z}, \mathbf{x}) \ \overline{\partial_2^{\beta} \mathcal{C}^*(\mathbf{z}, \mathbf{y})} = \langle \partial_2^{\beta} \mathcal{C}^*(\ , \mathbf{y}) | \partial_2^{\alpha} \mathcal{B}^*(\ , \mathbf{x}) \rangle .$$

$$(2.235)$$

Then, the thesis follows using the previously discussed identity (2.230), making obvious substitutions and recalling that $C^{**} = C$.

²⁶Of course, the embeddings indicated within round brackets hold since they involve spaces of positive and negative orders on the left and right-hand sides respectively. Let us also point out that \mathcal{B} and \mathcal{C} may fulfill condition (2.217)_j for different parameters $\theta_{\mathcal{B}} \neq \theta_{\mathcal{C}} \in (0, +\infty)$; in this case, it suffices to put $\theta := \min\{\theta_{\mathcal{B}}, \theta_{\mathcal{C}}\}$ in the previous considerations.

²⁷To prove this statement, recall once more that the maps $\mathcal{C}: \mathcal{H}^{-(j_2+d/2+\vartheta)} \to \mathcal{H}^{j_1+d/2+\vartheta}$ and $\mathcal{B}^*: \mathcal{H}^{-(i_1+d/2+\vartheta)} \to \mathcal{H}^{i_2+d/2+\vartheta}$ are continuous and that there holds the continuous embedding $\mathcal{H}^{j_1+d/2+\vartheta} \to \mathcal{H}^{-(j_2+d/2+\vartheta)}$ for all $i_1, i_2, j_1, j_2 \in \mathbb{N}$ as above.

Let us now consider the extension $\mathcal{J}: \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ of the complex conjugation $\mathcal{J}: \mathcal{H} \to \mathcal{H}, f \mapsto \mathcal{J}f := \overline{f}$ (see Proposition 2.25). Then, there holds the following result, which we report here only for completeness.

Lemma 2.61. Let $(2.217)_j$ hold for some $j \in \mathbb{N}$. Then, for any multi-index β of order $\leq j$ and for all $y \in \Omega$, there holds

$$\partial_2^{\beta} \mathcal{B}(, \mathbf{y}) = \partial_2^{\beta} (\mathcal{J} \mathcal{B} \mathcal{J}^{-1})(, \mathbf{y})$$
(2.236)

(of course, also in this case this identity is meant to hold in $\mathcal{H}^{j-|\beta|+d/2+\vartheta} \hookrightarrow C^{j-|\beta|}(\Omega)$; see below Eq. (2.227)).

Proof. The thesis follows straightforwardly from the chain of equalities

$$\overline{\partial_2^{\beta} \mathcal{B}(\mathbf{y})} = (-1)^{|\beta|} \mathcal{J} \mathcal{B} \mathcal{J}^{-1} \mathcal{J} \partial^{\beta} \delta_{\mathbf{y}} = (-1)^{|\beta|} \mathcal{J} \mathcal{B} \mathcal{J}^{-1} \partial^{\beta} \delta_{\mathbf{y}} = \partial_2^{\beta} (\mathcal{J} \mathcal{B} \mathcal{J}^{-1})(\mathbf{y})$$

which can be easily derived recalling the properties of the extended complex conjugation \mathcal{J} , along with the results of Proposition 2.56 and of Lemma 2.52.

The Dirichlet kernel. The heat, cylinder and modified cylinder kernels.

Let us keep all the notations of the previous sections; for any $r \in \mathbb{R}$, Σ_r denotes again the strip $\{s \in \mathbb{C} \mid \Re s > r\}$ (see Eq. (2.133)). Note that, for any $s \in \Sigma_{d/2}$ there exists $\vartheta > 0$ such that $\Re s > d/2 + \vartheta$, so that the operator \mathcal{A}^{-s} maps continuously $\mathcal{H}^{-(d/2+\vartheta)}$ to $\mathcal{H}^{d/2+\vartheta}$. Then, the results obtained previously in the present section allow us to define the kernel

$$\mathcal{A}^{-s}(,) \in C^{0}(\Omega \times \Omega), \qquad (\mathbf{x}, \mathbf{y}) \mapsto \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y}) := \langle \delta_{\mathbf{x}} | \mathcal{A}^{-s} \delta_{\mathbf{y}} \rangle.$$
(2.237)

Definition 2.62. For $s \in \Sigma_{d/2}$, the function $\mathcal{A}^{-s}(,)$ is referred to as the *Dirichlet kernel* of \mathcal{A} of order s.

Next, let $\mathbf{t} \in (0, +\infty)$; each of the operators $e^{-\mathbf{t}\mathcal{A}}$, $e^{-\mathbf{t}\sqrt{\mathcal{A}}}$ and $e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}$ maps continuously $\mathcal{H}^{-(d/2+\vartheta)}$ to $\mathcal{H}^{d/2+\vartheta}$ for any $\vartheta \in \mathbb{R}$ (in particular, for $\vartheta > 0$). Again, the previous results of this section allow us to define the kernels

$$e^{-\mathbf{t}\mathcal{A}}(,), e^{-\mathbf{t}\sqrt{\mathcal{A}}}(,), (e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(,) \in C^{0}(\Omega \times \Omega);$$
 (2.238)

these are such that, for example,

$$e^{-\mathbf{t}\mathcal{A}}(\mathbf{x},\mathbf{y}) := \langle \delta_{\mathbf{x}} | e^{-\mathbf{t}\mathcal{A}} \delta_{\mathbf{y}} \rangle$$
 for all $\mathbf{x}, \mathbf{y} \in \Omega$. (2.239)

Definition 2.63. Let $\mathbf{t} \in (0, +\infty)$. $e^{-\mathbf{t}\mathcal{A}}(,), e^{-\mathbf{t}\sqrt{\mathcal{A}}}(,)$ and $(e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(,)$ are referred to, respectively, as the *heat*, cylinder (or Poisson) and modified cylinder kernel of \mathcal{A} at \mathbf{t} .

Let us point out some properties of the above kernels.

Proposition 2.64. (Properties of the Dirichlet kernel). Let $j \in \mathbb{N}$ and let α, β be any pair of multi-indices such that $|\alpha| + |\beta| \leq j$.

i) For all $s \in \Sigma_{(j+d)/2}$, the kernel $\mathcal{A}^{-s}(,)$ is in $C^{j}(\Omega \times \Omega)$ and there hold all the statements of Proposition 2.56. Moreover, for any $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, one has

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-\overline{s}}(\boldsymbol{x}, \boldsymbol{y}) = \partial_1^{\beta} \partial_2^{\alpha} \mathcal{A}^{-\overline{s}}(\boldsymbol{y}, \boldsymbol{x}) = \overline{\partial_1^{\beta} \partial_2^{\alpha} \mathcal{A}^{-s}(\boldsymbol{y}, \boldsymbol{x})} ; \qquad (2.240)$$

ii) Let $\vartheta \in (0, +\infty)$ and put

$$r_{(\alpha\beta,\vartheta)} := \frac{|\alpha| + |\beta| + d}{2} + \vartheta ; \qquad (2.241)$$

then, for all $s \in \Sigma_{(j+d)/2+\vartheta}$, there holds the pointwise bound (for $\boldsymbol{x}, \boldsymbol{y} \in \Omega$)

$$\left|\partial_{1}^{\alpha}\partial_{2}^{\beta}\mathcal{A}^{-s}(\boldsymbol{x},\boldsymbol{y})\right| \leq \varepsilon^{-(\Re s - r_{(\alpha\beta,\vartheta)})} \left\|\partial^{\alpha}\delta_{\boldsymbol{x}}\right\|_{-(|\alpha| + d/2 + \vartheta)} \left\|\partial^{\beta}\delta_{\boldsymbol{y}}\right\|_{-(|\beta| + d/2 + \vartheta)} .$$
(2.242)

iii) Let $s_1 \in \Sigma_{(j+d)/2}$, so that \mathcal{A}^{-s_1} fulfills the assumption $(2.217)_j$ for some $\vartheta \in (0, +\infty)$; moreover, let $s_2 \in \Sigma_{-\vartheta/2}$. Then, for any fixed $\boldsymbol{y} \in \Omega$, there holds

$$\mathcal{A}^{-s_2}(\partial_2^{\beta}\mathcal{A}^{-s_1}(,\boldsymbol{y})) = \partial_2^{\beta}\mathcal{A}^{-(s_1+s_2)}(,\boldsymbol{y})$$
(2.243)

(for any $\vartheta' \in (0, +\infty)$), Eq. (2.243) can be meant to hold in $\mathcal{H}^{j-|\beta|+d/2+\vartheta'} \hookrightarrow C^{j-|\beta|}(\Omega)$). iv) For all $s_1, s_2 \in \Sigma_{(j+d)/2}$ and for all $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, $\partial_1^{\alpha} \mathcal{A}^{-s_1}(\boldsymbol{x}, \cdot), \partial_2^{\beta} \mathcal{A}^{-s_1}(\cdot, \boldsymbol{y}) \in L^2(\Omega)$ and

$$\langle \partial_2^{\alpha} \mathcal{A}^{-s_1}(\boldsymbol{x}) | \partial_2^{\beta} \mathcal{A}^{-s_2}(\boldsymbol{y}) \rangle_{L^2} = \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-(\bar{s}_1 + s_2)}(\boldsymbol{x}, \boldsymbol{y}) , \qquad (2.244)$$

$$\langle \partial_1^{\alpha} \mathcal{A}^{-s_1}(\boldsymbol{x}, \) | \partial_1^{\beta} \mathcal{A}^{-s_2}(\boldsymbol{y}, \) \rangle_{L^2} = \partial_2^{\alpha} \partial_1^{\beta} \mathcal{A}^{-(\overline{s}_1 + s_2)}(\boldsymbol{y}, \boldsymbol{x}) \ . \tag{2.245}$$

v) For any fixed $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, the map

$$\Sigma_{(j+d)/2} \to \mathbb{C} , \qquad s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y})$$
 (2.246)

is analytic. Moreover, for any $n \in \mathbb{N}$ and $s \in \Sigma_{(j+d)/2}$, the kernel $(\partial_s^n \mathcal{A}^{-s})(,) \equiv (\mathcal{A}^{-s}(-\ln \mathcal{A})^n)(,)$ belongs to $C^j(\Omega \times \Omega)$ and

$$\partial_1^{\alpha} \partial_2^{\beta} (\partial_s^n \mathcal{A}^{-s})(\boldsymbol{x}, \boldsymbol{y}) = \partial_s^n \left(\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) \right) \,. \tag{2.247}$$

vi) Let $s \in \Sigma_{(j+d)/2}$ and β be a multi-index of order $\leq j$ such that $j - |\beta| + d/2 > 2$; then, for any fixed $\boldsymbol{y} \in \Omega$, $\partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{y}) \in Dom(\mathcal{A})$ (²⁸).

Remark 2.20. Regarding the statements of item iv), recall the general considerations of item ii) in Remark 2.19; in the present setting, the right-hand sides of both Eq.s (2.244) (2.245) do also make sense under the assumption $s_1, s_2 \in \mathbb{C}$ with $s_1+s_2 \in \Sigma_{(j+d)/2}$, which is weaker than the hypothesis $s_1, s_2 \in \Sigma_{(j+d)/2}$. This fact and other analogous considerations will be used in the following applications to construct the analytic continuation of certain expressions which can be represented in the form (2.244) (2.245); see Chapter 3.

²⁸In particular, this allows to infer that the map $\Omega \to \mathbb{C}$, $\mathbf{x} \mapsto \partial_2^\beta \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})$ fulfills (in weak sense) the possible boundary conditions included in the definition of \mathcal{A} .

Proof. i) Let $s \in \Sigma_{(j+d)/2}$; then, there exists $\vartheta \in (0, +\infty)$ such that $s > (j+d)/2 + \vartheta$ which also implies $s > (j_1+j_2+d)/2+\vartheta$ for all $j_1, j_2 \in \mathbb{N}$ with $j_1+j_2 \leq j$. Since \mathcal{A}^{-s} maps continuously $\mathcal{H}^{-(j_2+d/2+\vartheta)}$ into $\mathcal{H}^{j_1+d/2+\vartheta}$ for all $j_1+j_2 \leq j$, Lemma 2.55 and Proposition 2.56 yield the first part of the thesis. On the other hand, the first identity in Eq. (2.240) follows straightforwardly from Proposition 2.59 and Corollary 2.24; the second identity in the cited equation follows from Corollary 2.26 and from Lemma 2.61, in view of the regularity of the kernels involved.

ii) Let again s, ϑ be as in the proof of item i). Recall that due to Lemma 2.55, there holds $\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y}) = (-1)^{|\alpha|+|\beta|} \langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{A}^{-s} \partial^{\beta} \delta_{\mathbf{y}} \rangle$; keeping in mind that the bilinear form $\langle | \rangle : \mathcal{H}^{-(|\alpha|+d/2+\vartheta)} \times \mathcal{H}^{|\alpha|+d/2+\vartheta} \to \mathbb{C}$ is continuous (see Eq. (2.104)), this allows to infer

$$\begin{aligned} &|\partial_1^{\alpha}\partial_2^{\beta}\mathcal{A}^{-s}(\mathbf{x},\mathbf{y})|\\ \leq \|\partial^{\alpha}\delta_{\mathbf{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\mathcal{A}^{-s}\partial^{\beta}\delta_{\mathbf{y}}\|_{|\alpha|+d/2+\vartheta} = \|\partial^{\alpha}\delta_{\mathbf{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\partial^{\beta}\delta_{\mathbf{y}}\|_{|\alpha|+d/2+\vartheta-2\Re s} \ ,\end{aligned}$$

where the last equality can be derived using the second relation in Eq. (2.95). Then, the thesis follows using the bound (2.84) of Proposition 2.5.

iii) For all $j_1 \in \mathbb{N}$ with $j_1 \leq j$, item i) of Corollary 2.13 implies $\mathcal{A}^{-s_2}(\mathcal{H}^{j_1+d/2+\vartheta}) = \mathcal{H}^{j_1+d/2+\vartheta+2\Re s_2}$. Furthermore, the hypothesis on s_2 grants that $\vartheta + 2\Re s_2 > \vartheta'$ for some $\vartheta' > 0$, so that $\mathcal{H}^{j_1+d/2+\vartheta+2\Re s_2} \hookrightarrow \mathcal{H}^{j_1+d/2+\vartheta'}$; this allows to infer that the map \mathcal{A}^{-s_2} : $\mathcal{H}^{j_1+d/2+\vartheta} \to \mathcal{H}^{j_1+d/2+\vartheta'}$ is continuous. Then, the thesis follows straightforwardly from Proposition 2.58.

iv) The statement $\partial_1^{\alpha} \mathcal{A}^{-s_1}(\mathbf{x}, \cdot), \partial_2^{\beta} \mathcal{A}^{-s_1}(\cdot, \mathbf{y}) \in L^2(\Omega)$ follows from items ii) and iii) of Proposition 2.56, while Eq.s (2.244) (2.245) can be derived using Proposition 2.60.

v) It suffices to show that the map $s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})$ is analytic on the strip $\Sigma_{(j+d)/2+\vartheta}$ for each $\vartheta \in (0, +\infty)$. Recall that $\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y}) = (-1)^{|\alpha|+|\beta|} \langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{A}^{-s} \partial^{\beta} \delta_{\mathbf{y}} \rangle$, with $\partial^{\alpha} \delta_{\mathbf{x}} \in \mathcal{H}^{-(|\alpha|+d/2+\vartheta)}$ and $\partial^{\beta} \delta_{\mathbf{y}} \in \mathcal{H}^{-(|\beta|+d/2+\vartheta)}$ (see item ii) of the present proof). So, the map $\Sigma_{(j+d)/2} \ni s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})$ can be represented as the composition of the map $\Sigma_{(|\alpha|+|\beta|+d)/2+\vartheta} \to \mathfrak{B}(\mathcal{H}^{-(|\beta|+d/2+\vartheta)}, \mathcal{H}^{|\alpha|+d/2+\vartheta}), s \mapsto \mathcal{A}^{-s} \upharpoonright \mathcal{H}^{-(|\beta|+d/2+\vartheta)}$, which is analytic due to Proposition 2.28, with the linear (continuous) form $\mathfrak{B}(\mathcal{H}^{-(|\beta|+d/2+\vartheta)}, \mathcal{H}^{|\alpha|+d/2+\vartheta}) \to \mathbb{C}, \mathcal{B} \mapsto \langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B} \partial^{\beta} \delta_{\mathbf{y}} \rangle$; this is sufficient to infer the thesis.

vi) The thesis follows easily from item iii) of Proposition 2.56 and from the considerations of the related Remark 2.17, noting that $\partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{y}) \in \mathcal{H}^{j-|\beta|+d/2+\vartheta}$ for some $\vartheta \in (0, +\infty)$.

Proposition 2.65. (Properties of the exponential kernels). Let $j \in \mathbb{N}$ and let α, β be any two multi-indices with $|\alpha| + |\beta| \leq j$.

i) For all $\mathbf{t} \in \Sigma_0$, the kernel $e^{-\mathbf{t}\mathcal{A}}(,)$ is in $C^{\infty}(\Omega \times \Omega)$ and there hold all the statements of Proposition 2.56 for any $j \in \mathbb{N}$; moreover, for any $\mathbf{x}, \mathbf{y} \in \Omega$, one has

$$\partial_1^{\alpha} \partial_2^{\beta} e^{-\bar{\mathbf{t}}\mathcal{A}}(\boldsymbol{x}, \boldsymbol{y}) = \partial_1^{\beta} \partial_2^{\alpha} e^{-\bar{\mathbf{t}}\mathcal{A}}(\boldsymbol{y}, \boldsymbol{x}) = \partial_1^{\beta} \partial_2^{\alpha} e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{y}, \boldsymbol{x}) .$$
(2.248)

ii) Let $\vartheta \in (0, +\infty)$ and put again (see Eq. (2.241))

$$r_{(\alpha\beta,\vartheta)} := \frac{|\alpha| + |\beta| + d}{2} + \vartheta ;$$

2.7. INTEGRAL KERNELS.

then, there holds the pointwise bound (for $\boldsymbol{x}, \boldsymbol{y} \in \Omega$)

$$\begin{aligned} &|\partial_{1}^{\alpha}\partial_{2}^{\beta}e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{x},\boldsymbol{y})| \leq \\ &\left\{ \begin{array}{l} \left(\frac{r_{(\alpha\beta,\vartheta)}}{e\,\Re\mathbf{t}}\right)^{r_{(\alpha\beta,\vartheta)}} \|\partial^{\alpha}\delta_{\boldsymbol{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\partial^{\beta}\delta_{\boldsymbol{y}}\|_{-(|\beta|+d/2+\vartheta)} & \text{for } 0<\Re\mathbf{t}\leq\frac{r_{(\alpha\beta,\vartheta)}}{\varepsilon} \\ \varepsilon^{r_{(\alpha\beta,\vartheta)}}e^{-\varepsilon\,\Re\mathbf{t}} \|\partial^{\alpha}\delta_{\boldsymbol{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\partial^{\beta}\delta_{\boldsymbol{y}}\|_{-(|\beta|+d/2+\vartheta)} & \text{for } \Re\mathbf{t}>\frac{r_{(\alpha\beta,\vartheta)}}{\varepsilon} \end{array} \right\}. \end{aligned}$$
(2.249)

iii) For all $\mathbf{t}_1, \mathbf{t}_2 \in \Sigma_0$ and for all fixed $\mathbf{y} \in \Omega$, there holds

$$e^{-\mathbf{t}_2\mathcal{A}}(\partial_2^\beta e^{-\mathbf{t}_1\mathcal{A}}(\mathbf{y})) = \partial_2^\beta e^{-(\mathbf{t}_1 + \mathbf{t}_2)\mathcal{A}}(\mathbf{y}) .$$
(2.250)

iv) For all $\mathbf{t}_1, \mathbf{t}_2 \in \Sigma_0$ and for all $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, $\partial_1^{\alpha} e^{-\mathbf{t}_1 \mathcal{A}}(\boldsymbol{x}, \cdot), \partial_2^{\beta} e^{-\mathbf{t}_1 \mathcal{A}}(\cdot, \boldsymbol{y}) \in L^2(\Omega)$ and

$$\langle \partial_2^{\alpha} e^{-\mathbf{t}_1 \mathcal{A}} (, \mathbf{x}) | \partial_2^{\beta} e^{-\mathbf{t}_2 \mathcal{A}} (, \mathbf{y}) \rangle = \partial_1^{\alpha} \partial_2^{\beta} e^{-(\bar{\mathbf{t}}_1 + \mathbf{t}_2) \mathcal{A}} (\mathbf{x}, \mathbf{y}) , \qquad (2.251)$$

$$\langle \partial_1^{\alpha} e^{-\mathbf{t}_1 \mathcal{A}}(\boldsymbol{x}, \) | \partial_1^{\beta} e^{-\mathbf{t}_2 \mathcal{A}}(\boldsymbol{y}, \) \rangle = \partial_2^{\alpha} \partial_1^{\beta} e^{-(\overline{\mathbf{t}}_1 + \mathbf{t}_2) \mathcal{A}}(\boldsymbol{y}, \boldsymbol{x}) \ .$$
(2.252)

v) For any fixed $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, the map

$$\Sigma_0 \to \mathbb{C} , \qquad \mathbf{t} \mapsto \partial_1^{\alpha} \partial_2^{\beta}(e^{-\mathbf{t}\mathcal{A}})(\boldsymbol{x}, \boldsymbol{y})$$
 (2.253)

is analytic.

vi) Let $\mathbf{t} \in \Sigma_0$ and β be a multi-index; then, for any fixed $\mathbf{y} \in \Omega$, $\partial_2^{\beta} e^{-\mathbf{t} \mathcal{A}}(\mathbf{y}) \in Dom(\mathcal{A})$ $(^{29}).$

Analogous results also hold for the kernels $e^{-t\sqrt{A}}(,), (e^{-t\sqrt{A}}/\sqrt{A})(,)$. In particular, in place of Eq. (2.249), for $\boldsymbol{x}, \boldsymbol{y} \in \Omega$ there hold

$$\left|\partial_{1}^{\alpha}\partial_{2}^{\beta}e^{-\mathbf{t}\sqrt{\mathcal{A}}}\left(\boldsymbol{x},\boldsymbol{y}\right)\right| \leq \tag{2.254}$$

$$\begin{cases} \left(\frac{2r_{(\alpha\beta,\vartheta)}}{e\,\Re\mathbf{t}}\right)^{2r_{(\alpha\beta,\vartheta)}} \|\partial^{\alpha}\delta_{\boldsymbol{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\partial^{\beta}\delta_{\boldsymbol{y}}\|_{-(|\beta|+d/2+\vartheta)} & \text{for } 0<\Re\mathbf{t}\leq\frac{2r_{(\alpha\beta,\vartheta)}}{\sqrt{\varepsilon}}\\ \varepsilon^{r_{(\alpha\beta,\vartheta)}}e^{-\sqrt{\varepsilon}\,\Re\mathbf{t}} \|\partial^{\alpha}\delta_{\boldsymbol{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\partial^{\beta}\delta_{\boldsymbol{y}}\|_{-(|\beta|+d/2+\vartheta)} & \text{for } \Re\mathbf{t}>\frac{2r_{(\alpha\beta,\vartheta)}}{\sqrt{\varepsilon}} \end{cases};$$

$$\left|\partial_{1}^{\alpha}\partial_{2}^{\beta}(e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(\boldsymbol{x},\boldsymbol{y})\right| \leq$$
(2.255)

$$\begin{aligned} |\partial_{1}^{\alpha}\partial_{2}^{\beta}(e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(\boldsymbol{x},\boldsymbol{y})| &\leq \qquad (2.255) \\ \left\{ \left(\frac{2r_{(\alpha\beta,\vartheta)}-1}{e\,\Re\mathbf{t}}\right)^{2r_{(\alpha\beta,\vartheta)}-1} \|\partial^{\alpha}\delta_{\boldsymbol{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\partial^{\beta}\delta_{\boldsymbol{y}}\|_{-(|\beta|+d/2+\vartheta)} \quad for \ 0 < \Re\mathbf{t} \leq \frac{2r_{(\alpha\beta,\vartheta)}-1}{\sqrt{\varepsilon}} \\ \varepsilon^{r_{(\alpha\beta,\vartheta)}-1/2}e^{-\sqrt{\varepsilon}\Re\mathbf{t}} \|\partial^{\alpha}\delta_{\boldsymbol{x}}\|_{-(|\alpha|+d/2+\vartheta)} \|\partial^{\beta}\delta_{\boldsymbol{y}}\|_{-(|\beta|+d/2+\vartheta)} \quad for \ \Re\mathbf{t} > \frac{2r_{(\alpha\beta,\vartheta)}-1}{\sqrt{\varepsilon}} \end{aligned} \right\}.$$

²⁹In particular, this allows to infer that the map $\Omega \to \mathbb{C}$, $\mathbf{x} \mapsto \partial_2^\beta e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y})$ fulfills (in weak sense) the possible boundary conditions included in the definition of \mathcal{A} .

Proof. i) Fix $\mathbf{t} \in (0, +\infty)$ and $j \in \mathbb{N}$ arbitrarily; the operator $e^{-\mathbf{t}\mathcal{A}}$ maps continuously $\mathcal{H}^{-(j_2+d/2+\vartheta)}$ into $\mathcal{H}^{j_1+d/2+\vartheta}$ for each $\vartheta \in (0, +\infty)$ and for all $j_1, j_2 \in \mathbb{N}$ (in particular, if $j_1 + j_2 \leq j$). So, the integral kernel $(e^{-\mathbf{t}\mathcal{A}})(,)$ is of class C^j by Lemma 2.55, and Proposition 2.56 holds for such j. Then, the first part of the thesis (i.e., $e^{-\mathbf{t}\mathcal{A}}(,) \in C^{\infty}(\Omega \times \Omega)$) follows from the arbitrariness of $j \in \mathbb{N}$. The first identity in Eq. (2.248) can be easily proved using Lemma 2.59 and Corollary 2.24; also in this case, the second identity in the cited equation follows from Corollary 2.26 and from Lemma 2.61, in view of the regularity of the kernels involved.

ii) First recall that, due to Lemma 2.55, we have $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y}) = (-1)^{|\alpha|+|\beta|} \langle \partial^{\alpha} \delta_{\mathbf{x}} | e^{-\mathbf{t}\mathcal{A}} \partial^{\beta} \delta_{\mathbf{y}} \rangle$; due to Eq. (2.104) for the bilinear form $\langle | \rangle : \mathcal{H}^{-(|\alpha|+d/2+\vartheta)} \times \mathcal{H}^{|\alpha|+d/2+\vartheta} \to \mathbb{C}$ for any $\vartheta \in (0, +\infty)$, we have

$$\left|\partial_{1}^{\alpha}\partial_{2}^{\beta}(e^{-\mathbf{t}\mathcal{A}})(\mathbf{x},\mathbf{y})\right| \leq \left\|\partial^{\alpha}\delta_{\mathbf{x}}\right\|_{-(|\alpha|+d/2+\vartheta)} \left\|e^{-\mathbf{t}\mathcal{A}}\partial^{\beta}\delta_{\mathbf{y}}\right\|_{|\alpha|+d/2+\vartheta}$$

The thesis (2.249) follows easily noting that the bound (2.97) of Corollary 2.14 gives (with $r_{(\alpha\beta,\vartheta)}$ as in Eq. (2.241))

$$\|e^{-\mathbf{t}\mathcal{A}}\partial^{\beta}\delta_{\mathbf{y}}\|_{|\alpha|+d/2+\vartheta} \leq \begin{cases} \left(\frac{r_{(\alpha\beta,\vartheta)}}{e\,\mathfrak{N}\mathbf{t}}\right)^{r_{(\alpha\beta,\vartheta)}} \|\partial^{\beta}\delta_{\mathbf{y}}\|_{-(|\beta|+d/2+\vartheta)} & \text{for } 0<\Re\mathbf{t}\leq\frac{r_{(\alpha\beta,\vartheta)}}{\varepsilon}\\ \varepsilon^{\frac{|\alpha|+|\beta|+d}{2}+\vartheta} e^{-\varepsilon\,\Re\mathbf{t}} \|\partial^{\beta}\delta_{\mathbf{y}}\|_{-(|\beta|+d/2+\vartheta)} & \text{for } \Re\mathbf{t}>\frac{|\alpha|+|\beta|+d+2\vartheta}{2\varepsilon} \end{cases}$$

iii) The thesis follows from Proposition 2.58 since $e^{-\mathbf{t}\mathcal{A}}(\mathcal{H}^{-\infty}) = \mathcal{H}^{+\infty}$ for all $\mathbf{t} \in \Sigma_0$ (see Corollary 2.14) and $\mathcal{H}^{+\infty} \hookrightarrow \mathcal{H}^r$ for any $r \in \mathbb{R}$ (see Proposition 2.8).

iv) Once more, the square-integrability of $\partial_1^{\alpha} e^{-\mathbf{t}_1 \mathcal{A}}(\mathbf{x}, \cdot)$ and $\partial_2^{\beta} e^{-\mathbf{t}_1 \mathcal{A}}(\cdot, \mathbf{y})$ follows from items ii) and iii) of Proposition 2.56, while Eq.s (2.251) (2.252) can be derived using Proposition 2.60.

v) Due to item ii) of the present proof we have $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y}) = (-1)^{|\alpha|+|\beta|} \langle \partial^{\alpha} \delta_{\mathbf{x}} | e^{-\mathbf{t}\mathcal{A}} \partial^{\beta} \delta_{\mathbf{y}} \rangle$, with $\partial^{\alpha} \delta_{\mathbf{x}} \in \mathcal{H}^{-(|\alpha|+d/2+\vartheta)}$, $\partial^{\beta} \delta_{\mathbf{y}} \in \mathcal{H}^{-(|\beta|+d/2+\vartheta)}$ for any r > j + d/2. So, the map $\mathbf{t} \mapsto \partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y})$ can be viewed as the composition of the map $\Sigma_0 \to \mathfrak{B}(\mathcal{H}^{-(|\beta|+d/2+\vartheta)}, \mathcal{H}^{|\alpha|+d/2+\vartheta})$, $\mathbf{t} \mapsto e^{-\mathbf{t}\mathcal{A}} \upharpoonright \mathcal{H}^{-(|\beta|+d/2+\vartheta)}$, which is analytic due to Proposition 2.29, with the linear (continuous) form $\mathfrak{B}(\mathcal{H}^{-(|\beta|+d/2+\vartheta)}, \mathcal{H}^{|\alpha|+d/2+\vartheta}) \to \mathbb{C}, \ \mathcal{B} \mapsto \langle \partial^{\alpha} \delta_{\mathbf{x}} | \mathcal{B} \partial^{\beta} \delta_{\mathbf{y}} \rangle$; this is sufficient to infer the thesis.

vi) Also in this case, the thesis follows noting that $\partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\mathbf{y}) \in \mathcal{H}^{+\infty} \hookrightarrow \mathcal{H}^2 \equiv \text{Dom}(\mathcal{A})$ (see, again, item iii) of Proposition 2.56 and Remark 2.17).

Heat and cylinder kernels as solutions of differential problems.

Assume again $\Omega \subset \mathbb{R}^d$ to be an arbitrary domain and let \mathcal{A} be the admissible operator obtained restricting the differential operator $A = -\Delta + V$ ($V \in C^{\infty}(\Omega)$) to an admissible domain $\mathcal{D}_{\mathcal{A}} \equiv \text{Dom}(\mathcal{A}) \subset L^2(\Omega)$ (³⁰). In the following, we will consider differential equations where more than one variable appear at the same time (namely, \mathbf{x} and $\mathbf{y} \in \Omega$);

 $^{^{30}}$ As a matter of fact, the results to be discussed in the present subsection continue to hold also if A is an otherwise generic, elliptic differential operator with smooth coefficients as in the footnote 21 of page 54.

2.7. INTEGRAL KERNELS.

in order to avoid confusion, we specify the variable to which the differential operator A is referred to by introducing the notation

$$A_{\mathbf{x}} := -\Delta_{\mathbf{x}} + V(\mathbf{x}) \qquad (\mathbf{x} \in \Omega)$$
(2.256)

where, of course, $\Delta_{\mathbf{x}}$ indicates the laplacian involving derivatives with respect to \mathbf{x} . Now, consider the heat and cylinder kernels which were introduced in the previous subsection as the integral kernels associated to the exponential operators $e^{-\mathbf{t}\mathcal{A}}$ and $e^{-\mathbf{t}\sqrt{\mathcal{A}}}$ (for $\mathbf{t} \in (0, +\infty)$).

Definition 2.66. For any fixed $\mathbf{y} \in \Omega$, let us define the maps

$$K_{\mathbf{y}}: (0, +\infty) \times \Omega \to \mathbb{R}, \qquad (\mathbf{t}, \mathbf{x}) \mapsto K_{\mathbf{y}}(\mathbf{t}; \mathbf{x}) := e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y}), \qquad (2.257)$$

$$T_{\mathbf{y}}: (0, +\infty) \times \Omega \to \mathbb{R}, \qquad (\mathbf{t}, \mathbf{x}) \mapsto T_{\mathbf{y}}(\mathbf{t}; \mathbf{x}) := e^{-\mathbf{t}\sqrt{\mathcal{A}}}(\mathbf{x}, \mathbf{y}).$$
 (2.258)

Remark 2.21. For any $\mathbf{t} \in (0, +\infty)$, there hold the following identities:

$$K_{\mathbf{y}}(\mathbf{t}\,;\,\,) = e^{-\mathbf{t}\mathcal{A}}\delta_{\mathbf{y}}\,,\qquad T_{\mathbf{y}}(\mathbf{t}\,;\,\,) = e^{-\mathbf{t}\sqrt{\mathcal{A}}}\delta_{\mathbf{y}}\,;\qquad(2.259)$$

note that the expressions on the right-hand sides of the above relations are, in fact, elements of $\mathcal{H}^{+\infty} \hookrightarrow C^{\infty}(\Omega)$. The fact that $K_{\mathbf{y}}(\mathbf{t};), T_{\mathbf{y}}(\mathbf{t};) \in \mathcal{H}^{+\infty} \hookrightarrow \mathcal{H}^2 = \text{Dom}(\mathcal{A})$ also indicates that these functions fulfill the boundary conditions encoded in the definition of $\text{Dom}(\mathcal{A})$.

Proposition 2.67. Fix $y \in \Omega$. Then, K_y and T_y possess the following properties:

$$\begin{cases} (\partial_{\mathbf{t}} + A_{\mathbf{x}}) K_{\mathbf{y}}(\mathbf{t} ; \mathbf{x}) = 0 & for (\mathbf{t}, \mathbf{x}) \in (0, +\infty) \times \Omega \\ \lim_{t \to 0^{+}} K_{\mathbf{y}}(0 ; \cdot) = \delta_{\mathbf{y}} in \mathcal{H}^{-d/2 + \vartheta} & for any \ \vartheta \in (0, +\infty) \end{cases} ; \qquad (2.260) \\ \begin{cases} (\partial_{\mathbf{tt}} + A_{\mathbf{x}}) T_{\mathbf{y}}(\mathbf{t} ; \mathbf{x}) = 0 & for (\mathbf{t}, \mathbf{x}) \in (0, +\infty) \times \Omega \\ \lim_{t \to 0^{+}} T_{\mathbf{y}}(0 ; \cdot) = \delta_{\mathbf{y}} in \mathcal{H}^{-d/2 + \vartheta} & for any \ \vartheta \in (0, +\infty) \end{cases} . \qquad (2.261) \end{cases}$$

Remark 2.22. Let us stress that the first equations in both Eq.s (2.260) and (2.261) can be meant to hold pointwisely, since both the kernels $K_{\mathbf{y}}$ and $T_{\mathbf{y}}$ are smooth for $(\mathbf{t}, \mathbf{x}) \in (0, +\infty) \times \Omega$ (see Remark 2.21 along with Proposition 2.65).

Proof. We are going to prove the equations in (2.260) (the proof of the analogous relations in (2.261) is similar and will not be reported here). We refer to Remark 2.21 for the proof of the second equation. On the other hand, let us fix $\vartheta \in (0, +\infty)$ arbitrarily; then the first equation follows if we can rigorously justfy the following chain of (formal) equalities

$$\begin{split} \partial_{\mathbf{t}} K_{\mathbf{y}}(\mathbf{t}\,;\mathbf{x}) &= \partial_{\mathbf{t}} \langle \delta_{\mathbf{x}} | e^{-\mathbf{t}\mathcal{A}} \delta_{\mathbf{y}} \rangle \stackrel{1)}{=} \langle \delta_{\mathbf{x}} | \partial_{\mathbf{t}} e^{-\mathbf{t}\mathcal{A}} \delta_{\mathbf{y}} \rangle \stackrel{2)}{=} - \langle \delta_{\mathbf{x}} | \mathcal{A} \, e^{-\mathbf{t}\mathcal{A}} \delta_{\mathbf{y}} \rangle = \\ \stackrel{3)}{=} -A_{\mathbf{x}} \langle \delta_{\mathbf{x}} | e^{-\mathbf{t}\mathcal{A}} \delta_{\mathbf{y}} \rangle = -A_{\mathbf{x}} K_{\mathbf{y}}(\mathbf{t}\,;\mathbf{x}) \; . \end{split}$$

As in the proof of statement ii) of Proposition 2.65, notice that for any fixed $\mathbf{x}, \mathbf{y} \in \Omega$ the map $\Sigma_0 \to \mathbb{C}, \mathbf{t} \mapsto \langle \delta_{\mathbf{x}} | e^{-\mathbf{t}\mathcal{A}} \delta_{\mathbf{y}} \rangle$ can be interpreted as the composition of the analytic function $\Sigma_0 \to \mathfrak{B}(\mathcal{H}^{-(d/2+\vartheta)}, \mathcal{H}^{d/2+\vartheta}), \mathbf{t} \mapsto e^{-\mathbf{t}\mathcal{A}}$ with the linear form $\mathfrak{B}(\mathcal{H}^{-(d/2+\vartheta)}, \mathcal{H}^{d/2+\vartheta}) \to \mathbb{C}, \mathcal{B} \mapsto \langle \delta_{\mathbf{x}} | \mathcal{B} \delta_{\mathbf{y}} \rangle$; then, 1) follows by the linearity and by the continuity of the second map. Equality 2) follows easily from Proposition 2.29. To prove equality 3), just notice that for any $f \in \mathcal{H}^{2+d/2+\vartheta} \hookrightarrow C^2(\Omega)$ there holds $\langle \delta_{\mathbf{x}} | \mathcal{A} f \rangle = (\mathcal{A} f)(\mathbf{x}) = (\mathcal{A} f)(\mathbf{x}) = \mathcal{A}_{\mathbf{x}} f(\mathbf{x}) = \mathcal{A}_{\mathbf{x}} \langle \delta_{\mathbf{x}} | f \rangle$; then, the thesis follows since $e^{-\mathbf{t}\mathcal{A}} \delta_{\mathbf{y}} \in \mathcal{H}^{+\infty} \hookrightarrow \mathcal{H}^{2+d/2+\vartheta}$ (for any $\mathbf{t} \in (0, +\infty)$).

Asymptotic behaviours of the heat and cylinder kernels.

Hereafter we report some facts on the asymptotic behaviour of the heat and cylinder kernels $e^{-t\mathcal{A}}(,)$, $e^{-t\sqrt{\mathcal{A}}}(,)$ in the limits of small and large values of the parameter $\mathbf{t} \in (0, +\infty)$. Some of these facts can be easily infered within the framework presented in this manuscript, in view of the results proved in the previous subsection 2.7. Moreover, we also take the chance to point out some more properties of the above mentioned kernels, which in the literature are well-known to hold and are closely related to the interpretation of $e^{-t\mathcal{A}}(,)$ and $e^{-t\sqrt{\mathcal{A}}}(,)$ as solutions of the differential problems (2.260) (2.261) discussed previously in Proposition 2.67.

First of all, recall the bounds reported in Eq.s (2.249), (2.254) and (2.255).

The cited equations state the well-known exponential decay for $\Re t \to +\infty$ of the heat, cylinder and modified cylinder kernels, respectively. As a matter of fact, by a slight generalization of Proposition 2.65, the same behaviour can be easily proved to hold for both the spatial and *t*-derivatives of the kernels cited above.

The equations cited above also give bounds on the small-t behaviour of the exponential kernels; these bounds have been derived using only the abstract functional analytic tools developed in the previous sections for the self-adjoint operator \mathcal{A} . As a matter of fact, since \mathcal{A} is a Schrödinger type differential operator $-\Delta + V$, acting in a (sufficiently regular) domain $\Omega \subset \mathbb{R}^d$ (and, possibly, with suitable boundary conditions on $\partial\Omega$), much more can be said about the asymptotic behaviour of the heat kernel for $t \to 0^+$. We refer to the monographies by Berline et al. [20], Calin et al. [33], Chavel [37], Davies [46], Gilkey [80] and Grigor'yan [82] (and to the works cited therein) for a complete and throughout analysis of this topic.

In the sequel we limit ourself to consider a particular case of Theorem 2.4 on page 39 in [32] (see also Lemma 2.1 in [116]): we focus the attention to the configurations described by the assumptions (2.159) and (2.209).

Theorem 2.68. Let the assumptions (2.159) and (2.209) be fulfilled (so that Dirichlet boundary conditions are taken into account). Then, there exists a unique sequence of realvalued (smooth) functions $a_n : \Omega \times \Omega \to \mathbb{R}$ $(n \in \mathbb{N}; a_0(\mathbf{x}, \mathbf{y}) \equiv 1)$ such that, in the limit $\mathbf{t} \to 0^+$ for all $\boldsymbol{x}, \boldsymbol{y} \in \Omega$, there holds

$$\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{x}, \boldsymbol{y}) =$$

$$\partial_1^{\alpha} \partial_2^{\beta} \left(\frac{1}{(4\pi \mathbf{t})^{d/2}} e^{-\frac{|\boldsymbol{x}-\boldsymbol{y}|^2}{4\mathbf{t}}} \sum_{n=1}^N a_n(\boldsymbol{x}, \boldsymbol{y}) \mathbf{t}^n \right) + \frac{e^{-\frac{\eta |\boldsymbol{x}-\boldsymbol{y}|^2}{4\mathbf{t}}}}{(4\pi \mathbf{t})^{d/2}} t^{N-|\alpha|-|\beta|} O_{\eta,N}^{(\alpha,\beta)}(\mathbf{t}; \boldsymbol{x}, \boldsymbol{y})$$
(2.262)

for any pair of multi-indeces $\alpha, \beta, N \in \mathbb{N}$ with $N > 2(|\alpha| + |\beta|) + d/2, \eta \in (0, 1)$ and for some continuous function $O_{\eta,N}^{(\alpha,\beta)} : [0, +\infty) \times \Omega \times \Omega \to \mathbb{C}$.

Remark 2.23. i) The coefficients a_n $(n \in \mathbb{N})$ introduced in the above theorem are usually referred to as *HMDS coefficients*, which is short for Hadamard-Minakshisundaram-DeWitt-Seeley after the authors who gave the earliest and most significant contributions to the study of the asymptotic behaviour of the heat kernel.

ii) Let us stress that the expression

$$K_0(\mathbf{t}; \mathbf{x}, \mathbf{y}) := \frac{1}{(4\pi \mathbf{t})^{d/2}} e^{-\frac{|\mathbf{x}-\mathbf{y}|^2}{4\mathbf{t}}}$$
(2.263)

in Eq. (2.262) is just the heat kernel associated to the laplacian on the whole space \mathbb{R}^d . iii) In view of points i) and v) of Proposition 2.65, Theorem 2.68 suggests the existence of a smooth function $H: [0, +\infty) \times \Omega \times \Omega \to \mathbb{R}$ such that

$$e^{-\mathbf{t}\mathcal{A}}(\mathbf{x},\mathbf{y}) = \frac{1}{(4\pi\mathbf{t})^{d/2}} e^{-\frac{|\mathbf{x}-\mathbf{y}|^2}{4\mathbf{t}}} H(\mathbf{t};\mathbf{x},\mathbf{y}) \quad \text{for all } \mathbf{t} \in (0,+\infty), \, \mathbf{x},\mathbf{y} \in \Omega \;.$$
(2.264)

This actually occurs in many interesting examples. In the sequel, when necessary, we will assume existence of such a smooth H.

iv) Following [32], we refer to [152] and [37] for some generalizations of this theorem, also dealing with some settings involving unbounded domains and configurations where Ω is a (curved) Riemannian manifold.

The corresponding small-t analysis for the cylinder kernel is more involved and less well-known. Starting from the expansion (2.262) for the heat kernel, Fulling [60, 73, 74, 76] (see also [13]) proved by means of Riesz summation methods the following result.

Theorem 2.69. Let the assumptions (2.159) and (2.209) be fulfilled (so that Dirichlet boundary conditions are being considered). Then, there exists a unique pair of sequences of real-valued functions $e_n, f_n : \Omega \to \mathbb{R}$ $(n \in \mathbb{N})$ such that for any $N \in \mathbb{N}$, in the limit $\mathbf{t} \to 0^+$ ($\mathbf{t} \in (0, +\infty)$), $\mathbf{x}, \mathbf{y} \in \Omega$ interior points), there holds

$$e^{-\mathbf{t}\sqrt{\mathcal{A}}}(\boldsymbol{x},\boldsymbol{x}) = \frac{1}{\mathbf{t}^d} \left(\sum_{n=0}^N e_n(\boldsymbol{x}) \, \mathbf{t}^n + \sum_{\substack{n=d+1\\n-d \ odd}}^N f_n(\boldsymbol{x}) \, \mathbf{t}^n \ln \mathbf{t} + O(\mathbf{t}^{N+1} \ln \mathbf{t}) \right).$$
(2.265)

Before proceeding, let us mention that the heat and cylinder kernels often admit asymptotic expansions of the forms (2.262), (2.265), even under assumptions weaker than those made in corresponding Theorems 2.68, 2.69. For these reason, in the following, expressions of the form (2.262) and (2.265) will often be assumed to hold as hypotheses.

Integral representations of the Dirichlet kernel in terms of the exponential kernels.

Consider the Dirichlet kernel along with the heat, cylinder and modified cylinder kernels. Hereafter we use the methods developed in the previous subsections to give an alternative derivation of some identities known in the literature, relating $\mathcal{A}^{-s}(,)$ to some integral transforms of the exponential kernels $e^{-t\mathcal{A}}(,)$, $e^{-t\sqrt{\mathcal{A}}}(,)$, $(e^{-t\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(,)$. More precisely, there holds the following proposition.

Proposition 2.70. Let $j \in \mathbb{N}$; then, for all $s \in \Sigma_{(j+d)/2}$ and for any pair of multi-indices α, β with $|\alpha| + |\beta| \leq j$, the following relations hold pointwisely for all $\boldsymbol{x}, \boldsymbol{y} \in \Omega$:

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) = \frac{1}{\Gamma(s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, \partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{x}, \boldsymbol{y}) \,, \qquad (2.266)$$

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) = \frac{1}{\Gamma(2s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{2s-1} \, \partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\sqrt{\mathcal{A}}}(\boldsymbol{x}, \boldsymbol{y}) \,, \qquad (2.267)$$

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) = \frac{1}{\Gamma(2s-1)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{2s-2} \, \partial_1^{\alpha} \partial_2^{\beta} (e^{-\mathbf{t}\sqrt{\mathcal{A}}} / \sqrt{\mathcal{A}})(\boldsymbol{x}, \boldsymbol{y}) \,. \tag{2.268}$$

Proof. The thesis follows easily from Proposition 2.31 and from the relation (2.221) of Lemma 2.55, recalling that $\partial^{\alpha} \delta_{\mathbf{x}} \in \mathcal{H}^{-(|\alpha|+d/2+\vartheta)}$ and $\partial^{\beta} \delta_{\mathbf{x}} \in \mathcal{H}^{-(|\beta|+d/2+\vartheta)}$ for any pair of multi-indeces α, β with $|\alpha| + |\beta| \leq j$ and any $\vartheta \in (0, +\infty)$.

Remark 2.24. The above proposition allows to express the Dirichlet kernel and its spatial derivatives in terms of the Mellin transforms of the exponential kernels and of the corresponding derivatives. We review the theory of Mellin transforms and the methods allowing to construct their analytic continuations in the following Section 2.8.

The case of a compact domain with smooth boundary.

In this subsection we restrict the attention to settings which fulfill the stronger regularity hypotheses (2.159), so that Ω is bounded with boundary of class C^{∞} and $A = -\Delta + V$ with $V \in C^{\infty}(\overline{\Omega})$; moreover, we assume (2.209) to hold, so that Dirichlet conditions are prescribed on the boundary $\partial\Omega$.

Let $\mathcal{B} : \text{Dom}(\mathcal{B}) \subset \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ be an operator fulfilling the condition $(2.217)_j$, for some $j \in \mathbb{N}$, so that

there exists $\vartheta \in (0, +\infty)$ such that, for all $j_1, j_2 \in \mathbb{N}$ with $j_1 + j_2 \leq j$, \mathcal{B} maps continuously $\mathcal{H}^{-(j_2+d/2+\vartheta)}$ to $\mathcal{H}^{j_1+d/2+\vartheta}$.

Keeping in mind the results discussed in the previous Section 2.6 for configurations of the above type, it can be easily proved that in this case the integral kernel $\mathcal{B}(,)$ can be extended up to the boundary $\partial\Omega$ of the domain Ω ; more precisely, one can consider the map (compare with Eq. (2.220))

$$\mathcal{B}(,): \Omega \times \Omega \to \mathbb{C}, \qquad (\mathbf{x}, \mathbf{y}) \mapsto \mathcal{B}(\mathbf{x}, \mathbf{y}) := \langle \delta_{\mathbf{x}} | \mathcal{B} \, \delta_{\mathbf{y}} \rangle.$$
 (2.269)

Moreover, all the regularity results derived in subsection 2.7 can be easily generalized to analogous formulations describing the boundary behaviour of the kernel (2.269). In particular, we have the following variant of Lemma 2.55.

Lemma 2.71. Let $(2.217)_j$ hold for some $j \in \mathbb{N}$. Then, $\mathcal{B}(,) \in C^j(\overline{\Omega} \times \overline{\Omega})$ (and Eq. (2.221) for the derivatives of $\mathcal{B}(,)$ holds everywhere on Ω).

Other results on the integral kernel $\mathcal{B}(,)$ are reported in the subsequent paragraphs.

Boundary conditions.

Recall the results stated in item iii) of Proposition 2.56 and the facts mentioned in the related Remark 2.17. We mentioned therein that, generally, the kernel $\mathcal{B}(,)$ fulfills in weak sense the possible boundary conditions which are taken into account in the definition of the domain $\text{Dom}(\mathcal{A})$.

The forthcoming Proposition 2.72 and the subsequent Remark 2.25 show that the the boundary conditions prescribed on $\partial\Omega$ are, in fact, satisfied pointwisely.

Proposition 2.72. Let the assumptions (2.159) and (2.209) hold; moreover, let $(2.217)_j$ hold for some $j \in \mathbb{N}$. Then, for any multi-index α of order $\leq j$ and for any $\mathbf{y} \in \overline{\Omega}$, there hold

$$\partial_2^{\alpha} \mathcal{B}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{x} \in \partial\Omega} = 0 \quad and \quad \partial_1^{\alpha} \mathcal{B}(\boldsymbol{y}, \boldsymbol{x}) \Big|_{\boldsymbol{x} \in \partial\Omega} = 0 .$$
 (2.270)

Proof. First of all note that, in view of the symmetry relation (2.229), it suffices to prove the first identity in Eq. (2.270). So, let us consider the expression $\partial_2^{\alpha} \mathcal{B}(\mathbf{y})$ and notice that, due to Proposition 2.56 (see, in particular, Eq. (2.225)), there holds $\partial_2^{\alpha} \mathcal{B}(\mathbf{y}) =$ $(-1)^{|\alpha|} \mathcal{B} \partial^{\alpha} \delta_{\mathbf{y}}$ for any $\mathbf{y} \in \overline{\Omega}$. On the other hand, recall that $\partial^{\alpha} \delta_{\mathbf{y}} \in \mathcal{H}^{-(|\alpha|+d/2+\vartheta)}$ for any $\vartheta > 0$; this, along with the assumption $(2.217)_j$, allows to infer that $\partial_2^{\alpha} \mathcal{B}(\mathbf{y}) \in$ $\mathcal{H}^{j-|\alpha|+d/2+\vartheta} \hookrightarrow \mathcal{H}^{d/2+\vartheta}$. Then, the thesis follows recalling that $\mathcal{H}^{d/2+\vartheta} \hookrightarrow \mathcal{H}^{d/2+\vartheta}(\Omega) \hookrightarrow$ $C^0(\overline{\Omega})$ (see Proposition 2.44 and the related Corollary 2.45), along with the results of item iii) of Proposition 2.56 (see also the related Remark 2.17).

Remark 2.25. Analogous results can be derived for other kind of boundary conditions. More precisely, if Neumann conditions are prescribed on $\partial\Omega$, for any multi-index α of order $\leq j-1$ and for any $\mathbf{y} \in \overline{\Omega}$, there holds

$$\partial_{\mathbf{n}(\mathbf{x})} \partial_2^{\alpha} \mathcal{B}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{x} \in \partial\Omega} = 0 \quad \text{and} \quad \partial_{\mathbf{n}(\mathbf{x})} \partial_1^{\alpha} \mathcal{B}(\mathbf{y}, \mathbf{x}) \Big|_{\mathbf{x} \in \partial\Omega} = 0 , \quad (2.271)$$

where $\partial_{\mathbf{n}(\mathbf{x})}$ indicates the normal derivative at $\mathbf{x} \in \partial \Omega$.

On the other hand, when Robin boundary conditions are imposed on $\partial\Omega$ (for some given smooth function $h: \partial\Omega \to \mathbb{R}$), for any multi-index α of order $\leq j-1$ and for any $\mathbf{y} \in \overline{\Omega}$, there holds

$$(h(\mathbf{x}) + \partial_{\mathbf{n}(\mathbf{x})}) \partial_2^{\alpha} \mathcal{B}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{x} \in \partial \Omega} = 0 \quad \text{and} \\ (h(\mathbf{x}) + \partial_{\mathbf{n}(\mathbf{x})}) \partial_1^{\alpha} \mathcal{B}(\mathbf{y}, \mathbf{x}) \Big|_{\mathbf{x} \in \partial \Omega} = 0 .$$
 (2.272)

Eigenfunction expansions and traces.

Let us first consider the topic *eigenfunction expansions*; to this purpose, let us recall that the notions of Hilbert-Schmidt kernel and of integral kernel introduced previously (using the Dirac delta elements), associated to a given suitable operator, do in fact coincide when both are well-defined (see Corollary 2.57). On the other hand, the kernels related to Hilbert-Schmidt operators can be expanded as in Eq. (2.63) using any assigned orthonormal basis of $L^2(\Omega)$. In the present setting it is natural to choose this basis as a set of eigenfunctions of \mathcal{A} . In fact, since Ω is bounded and suitable boundary conditions (e.g., of Dirichlet, Neumann or Robin type) are prescribed on $\partial\Omega$, the operator $\mathcal{A} := -\Delta + V$ has purely discrete spectrum and possesses such a basis of eigenfunctions. For the sake of definiteness, hereafter we will perform our analysis in the case of Dirichlet conditions; thus, from here to the end of the paragraph, besides boundedness of the domain Ω and smoothness of its boundary $\partial\Omega$, we assume \mathcal{A} to have the features (2.209):

$$Dom(\mathcal{A}) \equiv \mathcal{D}_{\mathcal{A}} = \{ f \in H_0^1(\Omega) \mid (-\triangle + V) f \in L^2(\Omega) \} ,$$

$$\sigma(\mathcal{A}) \subset [\varepsilon, +\infty) \text{ for some } \varepsilon > 0 .$$

In this case, \mathcal{A} is known to admit a complete orthonormal system of eigenfunctions $(F_n)_{n \in \mathbb{N}}$ with positive eigenvalues, denoted with ω_n^2 where $\omega_n > 0$:

$$\mathcal{A}F_n = \omega_n^2 F_n \ . \tag{2.273}$$

The eigenvalues can be ordered so that

$$0 < \omega_0 \le \omega_1 \le \omega_2 \le \dots$$
 and $\lim_{n \to +\infty} \omega_n = +\infty$, (2.274)

where some of the inequalities in the first relation could in fact be equalities in order to deal with degenerate eigenvalues (see, e.g., [138], page 94, Proposition 5.12).

In this situation $\sigma(\mathcal{A}) = \{\omega_n^2 | n \in \mathbb{N}\}$. For each function $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ (which is automatically measurable), $\phi(\mathcal{A})$ is the operator with eigenfunctions F_n and eigenvalues $\phi(\omega_n^2)$ defined on the linear subspace of $L^2(\Omega)$ on which the eigenfunction expansion converges; more precisely,

$$\operatorname{Dom}(\phi(\mathcal{A})) = \left\{ f \in L^{2}(\Omega) \mid \sum_{n=0}^{+\infty} |\phi(\omega_{n}^{2})|^{2} |\langle F_{n}|f \rangle|^{2} < +\infty \right\},$$

$$\phi(\mathcal{A})f = \sum_{n=0}^{+\infty} \phi(\omega_{n}^{2}) \langle F_{n}|f \rangle F_{n} \quad \text{for } f \in \operatorname{Dom}(\phi(\mathcal{A})) \ (L^{2} \text{ convergence})$$

$$(2.275)$$

 $(^{31})$. As a matter of fact, much more can be said about the eigenfunctions and eigenvalues of the admissible operator \mathcal{A} .

³¹It is hardly the case to say how this setting can be used to represent \mathcal{A} as a multiplication operator: one equips \mathbb{N} with the counting measure # and introduces the Hilbertian isomorphism $\mathcal{I} : L^2(\Omega) \to$

Theorem 2.73. i) For any $n \in \mathbb{N}$, there holds $F_n \in C^{\infty}(\overline{\Omega})$; in addition, for any $j \in \mathbb{N}$ and for any $\vartheta \in (0, +\infty)$, there exists a positive constant $\Lambda_{j,\vartheta}$ such that

$$\|F_n\|_{C^j} \le \Lambda_{j,\vartheta} \,\omega_n^{j+d/2+\vartheta} \tag{2.276}$$

(where $\| \|_{C^j}$ is the norm defined in Eq. (2.12)). ii) There holds the Weyl asymptotic relation

$$\lim_{n \to +\infty} \frac{\omega_n^2}{n^{2/d}} = C_d , \qquad C_d := \frac{4\pi}{|\Omega|^{-2/d}} \Gamma\left(\frac{d}{2} + 1\right)^{2/d} . \tag{2.277}$$

Proof. i) Let $n \in \mathbb{N}$. For any $r \in \mathbb{R}$ we have $F_n \in \text{Dom}(\mathcal{A}^{r/2}) \subset \mathcal{H}^r$ and $\mathcal{A}^{r/2}F_n = \omega_n^r F_n$, whence $\|F_n\|_r = \omega_n^r \|F_n\|_{L^2} = \omega_n^r$. Now let $j \in \mathbb{N}$, $\vartheta \in (0, +\infty)$ and put $r = j + d/2 + \vartheta$; then $\mathcal{H}^r \hookrightarrow C^j(\overline{\Omega})$, so there is a constant $\Lambda_{j,\vartheta} > 0$ such that $\| \|_{C^j} \leq \Lambda_{j,\vartheta} \| \|_r$ on \mathcal{H}^r . In particular $F_n \in C^j(\overline{\Omega})$ and $\|F_n\|_{C^j} \leq \Lambda_{j,\vartheta} \|F_n\|_r = \Lambda_{j,\vartheta} \omega_n^r$; this justifies Eq. (2.276). In these considerations j is any natural number; from $F_n \in C^j(\overline{\Omega})$ for all $j \in \mathbb{N}$ it follows that $F_n \in C^{\infty}(\overline{\Omega})$.

ii) See, e.g., [109], Theorem 5, page 189 and [55], §8.2, pages 99–101 for elementary derivations of Eq. (2.277). $\hfill \Box$

Now, let us consider a bounded function $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$; then $\phi(\mathcal{A}) \in \mathfrak{B}(\mathcal{H})$. The operators $\mathcal{A}^{\dagger}\mathcal{A}$ and $|\mathcal{A}|$ have eigenvalues $|\phi(\omega_n^2)|^2$ and $|\phi(\omega_n^2)|$, respectively, (with eigenfunctions F_n), so that

$$\operatorname{Tr}(\mathcal{A}^{\dagger}\mathcal{A}) = \sum_{n=1}^{+\infty} |\phi(\omega_n^2)|^2 , \qquad (2.278)$$

$$\operatorname{Tr}(|\phi(\mathcal{A})|) = \sum_{n=1}^{+\infty} |\phi(\omega_n^2)| . \qquad (2.279)$$

Thus, $\phi(\mathcal{A})$ is of Hilbert-Schmidt class (resp., trace class) if and only if the series (2.278) (resp., then series (2.279)) converges.

In the Hilbert-Schmidt case, $\phi(\mathcal{A})$ has a kernel of class L^2 ; this admits the expansion

$$\phi(\mathcal{A})(\mathbf{x}, \mathbf{y}) = \sum_{n=0}^{+\infty} \phi(\omega_n^2) F_n(\mathbf{x}) \overline{F}_n(\mathbf{y}) \qquad (\mathbf{x}, \mathbf{y} \in \Omega) , \qquad (2.280)$$

 $L^2(\mathbb{N}, \#)$ sending a function f into the sequence $\mathcal{I}f$ of elements $(\mathcal{I}f)_n = \langle F_n | f \rangle$ $(n \in \mathbb{N})$. Then, \mathcal{IAI}^{-1} is the operator of multiplication by the sequence $(\omega_n^2)_{n \in \mathbb{N}}$ and, for each measurable function $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$, $\mathcal{I}\phi(\mathcal{A})\mathcal{I}^{-1}$ is the operator of multiplication by the sequence $(\phi(\omega_n^2))_{n \in \mathbb{N}}$.

All these statements are rather obvious; the purpose of this footnote is just to connect the case under analyis to the general "multiplication operator formalism" that we have employed to build the scale of Hilbert spaces associated to any positive self-adjoint operator.

converging in $L^2(\Omega \times \Omega)$ (see subsection 2.3, especially Eq. (2.63)). In the trace class case we can of course define the trace of $\phi(\mathcal{A})$, given by

$$\operatorname{Tr} \phi(\mathcal{A}) = \sum_{n=0}^{+\infty} \phi(\omega_n^2) . \qquad (2.281)$$

Hereafter we infer convergence results for the previous series, assuming ϕ to fulfill a suitable decay condition at infinity, and using the results of Theorem 2.73. Our decay assumptions on $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ involve condition (2.90) (for suitable values of the real exponent b), i.e.,

$$\sup_{\lambda\in\sigma(\mathcal{A})}\lambda^b|\phi(\lambda)|<+\infty.$$

Let us recall that kernel $\phi(\mathcal{A})(,)$ is of class $C^{j}(\overline{\Omega} \times \overline{\Omega})$ $(j \in \mathbb{N})$ if (2.90) holds for a b such that b > (j + d)/2 (see Lemma 2.54 and Lemma 2.55).

Proposition 2.74. Assume $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ to fulfill Eq. (2.90) for some real b; then the following holds.

i) If b > d/4, then $\phi(\mathcal{A})$ is of Hilbert-Schmidt class; so, the eigenfunction expansion (2.280) for the kernel $\phi(\mathcal{A})(,)$ converges in $L^2(\Omega \times \Omega)$.

ii) If b > j/2 + d for some $j \in \mathbb{N}$, then the eigenfunction expansion (2.280) for the kernel $\phi(\mathcal{A})(,)$ converges absolutely in $C^{j}(\overline{\Omega} \times \overline{\Omega})$.

iii) If b > d/2, then $\phi(\mathcal{A})$ is of trace class. iv) If b > d, then

$$Tr\phi(\mathcal{A}) = \int_{\Omega} d\boldsymbol{x} \,\phi(\mathcal{A})(\boldsymbol{x}, \boldsymbol{x})$$
(2.282)

(note that the integrand function in the right-hand side is of class $C^0(\overline{\Omega})$).

Proof. First of all, let us note that the assumption (2.90) and the Weyl asymptotic relation (2.277) imply the existence of some positive constants c, c' such that

$$\begin{aligned} |\phi(\omega_n^2)| &\leq c \; \omega_n^{-2b} \quad \text{for all } n \in \mathbb{N} \;, \\ \omega_n &\geq c' \; n^{1/d} \quad \text{for all } n \in \mathbb{N} \setminus \{0\} \;. \end{aligned}$$
(2.283)

Keeping in mind these facts, let us prove statements i)-iv). i) Let b > d/4. To prove the thesis, on the grounds of Eq. (2.278) we must show that

$$\sum_{n=0}^{+\infty} |\phi(\omega_n^2)|^2 < +\infty \; .$$

On the other hand, due to Eq. (2.283), for each $n \ge 1$ we have

$$|\phi(\omega_n^2)|^2 \le \text{const.} \, {\omega_n}^{-4b} \le \text{const.} \, n^{-4b/d}$$

and the series $\sum_{n=1}^{+\infty} n^{-4b/d}$ is convergent.

ii) Let b > j/2 + d. To infer the thesis, it suffices prove the absolute convergence in $C^{j}(\overline{\Omega} \times \overline{\Omega})$ of the series (2.280), i.e., that

$$\sum_{n=0}^{+\infty} |\phi(\omega_n^2)| \|F_n(\cdot)\overline{F}_n(\cdot)\|_{C^j} < +\infty ; \qquad (2.284)$$

here, in analogy with Eq. (2.12) we have set

$$\|F_n(\cdot)\overline{F}_n(\cdot)\|_{C^j} := \max_{|\alpha|+|\beta| \le j, \mathbf{x}, \mathbf{y} \in \overline{\Omega}} |\partial^{\alpha} F_n(\mathbf{x}) \overline{\partial^{\beta} F_n}(\mathbf{y})| .$$
(2.285)

If Eq. (2.284) is proved, we have convergence in $C^j(\overline{\Omega} \times \overline{\Omega})$ of the series (2.280); but $C^j(\overline{\Omega} \times \overline{\Omega}) \hookrightarrow L^2(\Omega \times \Omega)$, so the sum of the series in C^j equals its sum in L^2 , that we know to be the kernel of $\phi(\mathcal{A})$.

In order to derive Eq. (2.284), let us point out that that definition (2.285) yields automatically $||F_n(\cdot)\overline{F}_n(\cdot)||_{C^j} \leq \max_{j_1+j_2\leq j} ||F_n||_{C^{j_1}} ||F_n||_{C^{j_2}}$; so, for any $\vartheta \in (0, +\infty)$, the bound in Eq. (2.276) gives

$$\|F_n(\cdot)\overline{F}_n(\cdot)\|_{C^j} \le \max_{j_1+j_2\le j} \left(\Lambda_{j_1,\vartheta}\Lambda_{j_2,\vartheta}\,\omega_n^{j_1+j_2+d+2\vartheta}\right)\,.$$

Moreover, recalling that $\omega_n \geq \omega_0 > 0$ (for all $n \in \mathbb{N}$), one has $\omega_n^{j_1+j_2} = \omega_0^{j_1+j_2} (\omega_n/\omega_0)^{j_1+j_2} \leq \omega_0^{j_1+j_2} (\omega_n/\omega_0)^j = \omega_n^j/\omega_0^{j_1-j_1-j_2}$, so that

$$\|F_n(\cdot)\overline{F}_n(\cdot)\|_{C^j} \le \left(\max_{j_1+j_2\le j} \frac{\Lambda_{j_1,\vartheta}\Lambda_{j_2,\vartheta}}{\omega_0^{j-j_1-j_2}}\right)\omega_n^{j+d+2\vartheta} .$$
(2.286)

Eq.s (2.283) and (2.286) imply the following, for each $n \ge 1$:

$$|\phi(\omega_n^2)| \|F_n(\cdot)\overline{F}_n(\cdot)\|_{C^j} \le \text{const. } \omega_n^{-2b-j-d-2\vartheta} \le \text{const. } n^{-\frac{2b-j-d-2\vartheta}{d}}$$

Now the thesis follows noting that the series $\sum_{n=1}^{+\infty} n^{-\frac{2b-j-d-2\vartheta}{d}}$ converges for any $\vartheta \in (0, b-j/2-d)$.

iii) Let b > d/2. To prove the thesis, on the grounds of Eq. (2.279) we must show that

$$\sum_{n=0}^{+\infty} |\phi(\omega_n^2)| < +\infty ;$$

to this purpose, arguments very similar to the ones employed for item i) can be used. iv) Let b > d. By item ii) with j = 0, the expansion (2.280) of the kernel $\phi(\mathcal{A})(,)$ converges in $C^0(\overline{\Omega} \times \overline{\Omega})$. In particular, along the diagonal we have the expansion

$$\phi(\mathcal{A})(\mathbf{x}, \mathbf{x}) = \sum_{n=0}^{+\infty} \phi(\omega_n^2) |F_n(\mathbf{x})|^2 , \qquad (2.287)$$

converging in $C^0(\overline{\Omega})$. Of course the map $C^0(\overline{\Omega}) \ni f \mapsto \int_{\Omega}$ is a continuous linear form with respect to the topology of $C^0(\overline{\Omega})$; so, we can integrate term by term Eq. (2.287) and infer

$$\int_{\Omega} d\mathbf{x} \, \phi(\mathcal{A})(\mathbf{x}, \mathbf{x}) = \sum_{n=0}^{+\infty} \phi(\omega_n^2) \, \int_{\Omega} d\mathbf{x} \, |F_n(\mathbf{x})|^2 = \sum_{n=0}^{+\infty} \phi(\omega_n^2) = \operatorname{Tr}(\mathcal{A}) \, .$$

Corollary 2.75. i) For all $s \in \Sigma_{d/4}$, the operator \mathcal{A}^{-s} is of Hilbert-Schmidt class; so, its integral kernel $\mathcal{A}^{-s}(,)$ admits the L²-convergent eigenfunction expansion

$$\mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) = \sum_{n=0}^{+\infty} \omega_n^{-2s} F_n(\boldsymbol{x}) \overline{F}_n(\boldsymbol{y}) \qquad (\boldsymbol{x}, \boldsymbol{y} \in \Omega) .$$
(2.288)

ii) For any $j \in \mathbb{N}$ and for all $s \in \Sigma_{j/2+d}$, the eigenfunction expansion (2.288) for $\mathcal{A}^{-s}(,)$ converges absolutely in $C^{j}(\overline{\Omega} \times \overline{\Omega})$.

iii) For all $s \in \Sigma_{d/2}$, \mathcal{A}^{-s} is a trace class operator whose trace is given by the absolutely convergent series

$$Tr \mathcal{A}^{-s} = \sum_{n=0}^{+\infty} \omega_n^{-2s}$$
 (2.289)

In addition, the map $\Sigma_{d/2} \ni s \mapsto Tr \mathcal{A}^{-s}$ is analytic (³²). iv) For all $s \in \Sigma_d$, there holds

$$Tr \mathcal{A}^{-s} = \int_{\Omega} d\boldsymbol{x} \, \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{x}) \,.$$
(2.290)

Proof. i)-ii) The thesis follows from items i) (resp. ii)) of Proposition 2.74 noting that, for any $s \in \Sigma_{d/4}$ (resp. $s \in \Sigma_{j/2+d}$), there exists some b > d/4 (resp. b > j/2 + d) such that $\sup_{\lambda \in \sigma(\mathcal{A})} \lambda^{b-\Re s} < +\infty$ (e.g., choose $b = d/4 + (\Re s - d/4)/2$, resp. $b = j/2 + d + (\Re s - j/2 - d)/2$).

iii) The series representation (2.289) and its absolute convergence can be inferred by arguments similar to those employed in the proof of items i) and ii), using item iii) of Proposition 2.74. To prove the analyticity of the map $\Sigma_{d/2} \ni s \mapsto \operatorname{Tr} \mathcal{A}^{-s}$, it suffices to show the existence of the complex derivative $\frac{d}{ds} \operatorname{Tr} \mathcal{A}^{-s}$; in view of the series expansion (2.289), the last statement follows from Lebesgue dominated convergence theorem if there exists a summable, local dominant for $\frac{d}{dt} \omega_n^{-2s} = -2 \omega_n^{-2s} \ln \omega_n$. To obtain such a dominant, let us fix $s \in \Sigma_{d/2}$ and choose a constant $\sigma \in (d/2, \Re s)$ (note that $\Re s - \sigma > d/2$); then, for any $s' \in \Sigma_{(\Re s - \sigma, \Re s + \sigma)}$, it can be proved by elementary methods that $|\omega_n^{-2s'} \ln \omega_n| \leq c \omega_n^{-2(\Re s - \sigma)}$ for some positive constant c, (depending on ω_0 and on the fixed parameters s, σ). On the other hand, since $\Re s - \sigma > d/2$, the Weyl asymptotic relation (2.277) allows to infer that

³²We also refer to [56, 84, 115, 116] for alternative derivations of this and other related results.

 $\sum_{n=0}^{+\infty} \omega_n^{-2(\Re s - \sigma)} < +\infty$ (by arguments similar to those employed to prove the convergence of the series (2.289)). Summing up, $c \omega_n^{-2(\Re s - \sigma)}$ yields the required local dominant, thus concluding the proof.

iv) The thesis follows by arguments similar to those employed in the proof of items i), ii), using item iv) of Proposition 2.74. $\hfill \Box$

Corollary 2.76. The following statements hold for all $\mathbf{t} \in \Sigma_0$.

i) The operator e^{-tA} is of Hilbert-Schmidt class; so, its integral kernel $e^{-tA}(,)$ admits the L^2 -convergent eigenfunction expansion

$$e^{-\mathbf{t}\mathcal{A}}(\mathbf{x},\mathbf{y}) = \sum_{n=0}^{+\infty} e^{-\mathbf{t}\omega_n^2} F_n(\mathbf{x}) \overline{F}_n(\mathbf{y}) \qquad (\mathbf{x},\mathbf{y}\in\Omega) .$$
(2.291)

ii) The eigenfunction expansion (2.291) for $e^{-t\mathcal{A}}(,)$ converges absolutely in $C^{\infty}(\overline{\Omega} \times \overline{\Omega})$. iii) $e^{-t\mathcal{A}}$ is a trace class operator whose trace is given by the absolutely convergent series

$$Tr e^{-\mathbf{t}\mathcal{A}} = \sum_{n=0}^{+\infty} e^{-\mathbf{t}\omega_n^2} . \qquad (2.292)$$

In addition, the map $\Sigma_0 \ni \mathbf{t} \mapsto Tre^{-\mathbf{t}\mathcal{A}}$ is analytic. iv) There holds

$$Tr e^{-\mathbf{t}\mathcal{A}} = \int_{\Omega} d\boldsymbol{x} e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{x}, \boldsymbol{x}) . \qquad (2.293)$$

Results analogous to statements i)-iv) hold, respectively, for the exponential operators $e^{-t\sqrt{A}}$, $e^{-t\sqrt{A}}/\sqrt{A}$.

Proof. i)-ii) The thesis follows from items i) and ii) of Proposition 2.74 noting that, for any $\mathbf{t} \in \Sigma_0$ and for all $b \in \mathbb{R}$ there holds $\sup_{\lambda \in \sigma(\mathcal{A})} \lambda^b e^{-\Re t\lambda} < +\infty$.

iii) Also in this case, arguments similar to those employed in the proof of items i) and ii) allow to obtain the series representation (2.292) and its absolute convergence, using again item iii) of Proposition 2.74. Next, let us prove the existence of the complex derivative $\frac{d}{dt} \operatorname{Tr} e^{-t\mathcal{A}}$, granting the analyticity of the map $\Sigma_0 \ni \mathbf{t} \mapsto \operatorname{Tr} e^{-t\mathcal{A}}$. Notice that, in view of the series expansion (2.292), this fact follows from Lebesgue dominated convergence theorem if there exists a summable, local dominant for $\frac{d}{dt} e^{-t\omega_n^2} = -\omega_n^2 e^{-t\omega_n^2}$. To obtain such a dominant let us fix, for any $\mathbf{t} \in \Sigma_0$, a constant $T \in (0, \Re \mathbf{t})$; then, for any $\mathbf{t}' \in \Sigma_{T_0}$, there holds $|\omega_n^2 e^{-t'\omega_n^2}| \leq \omega_n^2 e^{-T\omega_n^2}$. Now notice that $\sum_{n=0}^{+\infty} \omega_n^2 e^{-T\omega_n^2} < +\infty$, due to the Weyl asymptotic relation (2.277). Summing up, $\omega_n^2 e^{-T\omega_n^2}$ yields the required dominant, thus yielding the thesis.

iv) Again, the thesis follows by arguments similar to those employed in the proof of items i), ii), using item iv) of Proposition 2.74. $\hfill \Box$

For completeness, let us mention that the topic of eigenfunction expansions for integral kernels associated to self-adjoint operators could also be addressed resorting to the theory of rigged Hilbert spaces, within a much more general framework [78, 106, 17, 18]. As well-known, assuming there to exist a nuclear space $\mathcal{N} \subset \mathcal{H}$ such that $\mathcal{N} \hookrightarrow \mathcal{H} \hookrightarrow \mathcal{N}'$ is a Gelfand triplet, any self-adjoint operator \mathcal{B} on \mathcal{H} admits a complete system of generalized eigenfunctions belonging to \mathcal{N}' (see, e.g., [78], page 190, theorem 1) which can be used to give an expansion of the integral kernel associated to \mathcal{B} . Within this framework, much more can be said in the case of a self-adjoint operator \mathcal{A} given by the closure of an elliptic differential operator (such as the Schrödinger operator to be considered in Section 2.6) defined on any (not necessarily bounded) domain $\Omega \subset \mathbb{R}^d$, with minimal assumptions of regularity for its coefficients and for the boundary conditions. In particular, the corresponding generalized eigenfunctions are in fact sufficiently smooth; moreover, they can be used to give an expansion of the integral kernel associated to the operator $\phi(\mathcal{A})$ for any function $\phi : \sigma(\mathcal{A}) \to \mathbb{C}$ decreasing rapidly enough (see [17], page 398, Theorem 2.1 for more details).

Asymptotic behaviours of the heat and cylinder traces

The results reported in this paragraph constitute the global counterparts of the local analysis described in subsection 2.7, where the topic of asymptotic behaviour of the heat and cylinder kernels was addressed.

More precisely, hereafter we report some results describing the small and large t expansions of the heat and cylinder traces $\operatorname{Tr} e^{-t\mathcal{A}}$, $\operatorname{Tr} e^{-t\sqrt{\mathcal{A}}}$.

Proposition 2.77. The map $\Sigma_0 \to \mathbb{C}$, $\mathbf{t} \mapsto Tre^{-\mathbf{t}A}$ decays exponentially for $\Re \mathbf{t} \to +\infty$; more precisely, for any fixed $T \in (0, +\infty)$ and for all $\mathbf{t} \in \Sigma_T$, there holds

$$Tre^{-\mathbf{t}\mathcal{A}} \le C_T e^{-\Re \mathbf{t}\,\omega_0^2}$$
, $C_T := Tre^{-T(\mathcal{A}-\omega_0^2\mathbb{I})} < +\infty$ (2.294)

(recall that $\omega_0 > 0$ is the minimum eigenvalue of \mathcal{A} ; \mathbb{I} is the identity operator on \mathcal{H}). Analogous results hold for the other exponential traces $Tre^{-t\sqrt{\mathcal{A}}}$, $Tr(e^{-t\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})$.

Proof. Consider again the series expansion (2.292) and recall that $\omega_n \geq \omega_0$, for all $n \in \mathbb{N}$; therefore, for all T > 0 and for all $\mathbf{t} \in \Sigma_T$, there holds $|e^{-\mathbf{t}\omega_n^2}| = e^{-\Re\mathbf{t}\omega_0^2}e^{-\Re\mathbf{t}(\omega_n^2-\omega_0^2)} \leq e^{-\Re\mathbf{t}\omega_0^2}e^{-T(\omega_n^2-\omega_0^2)}$. Then, the thesis (2.294) follows noting that the Weyl asymptotic relation (2.277) implies $\operatorname{Tr} e^{-T(\mathcal{A}-\omega_0^2\mathbb{I})} = \sum_{n=0}^{+\infty} e^{-T(\omega_n^2-\omega_0^2)} < +\infty$, by a slight variation of the arguments used in the proof of Corollary 2.76.

Clearly, the previous proposition only deals with the large t behaviour of the exponential traces. The small t asymptotics of these traces is described instead by the forthcoming Theorems 2.78 2.79; these theorems contain results analogous to those reported in Theorems 2.68 2.69 of subsection 2.7 and, also in this case, we refer to [20, 80, 91] and to [73, 74, 76] for their respective proofs.

Theorem 2.78. Let the assumptions (2.159) and (2.209) be fulfilled; moreover, assume V to be positive definite. Then there exists a unique sequence of real coefficients $a_n \in \mathbb{R}$

2.7. INTEGRAL KERNELS.

 $(n \in \mathbb{N})$ such that for any $N \in \mathbb{N}$, in the limit $\mathbf{t} \to 0^+$ $(\mathbf{t} \in (0, +\infty))$, there holds

$$Tr e^{-\mathbf{t}\mathcal{A}} = \frac{1}{\mathbf{t}^{d/2}} \left(\sum_{n=1}^{N} a_n \, \mathbf{t}^{n/2} + O(\mathbf{t}^{\frac{N+1}{2}}) \right).$$
(2.295)

Theorem 2.79. Let the assumptions (2.159) and (2.209) be fulfilled; moreover, assume V to be positive definite. Then, there exist a unique pair of sequences $e_n, f_n \in \mathbb{R}$ $(n \in \mathbb{N})$ such that, for any $N \in \mathbb{N}$ and for $\mathbf{t} \to 0^+$ $(\mathbf{t} \in (0, +\infty))$, there holds

$$Tre^{-\mathbf{t}\sqrt{\mathcal{A}}} = \frac{1}{\mathbf{t}^d} \left(\sum_{n=0}^N e_n \, \mathbf{t}^n + \sum_{\substack{n=d+1\\n-d \text{ odd}}}^N f_n \, \mathbf{t}^n \ln \mathbf{t} + O(\mathbf{t}^{N+1} \ln \mathbf{t}) \right).$$
(2.296)

Integral representations for $\operatorname{Tr} \mathcal{A}^{-S}$.

The forthcoming Proposition 2.80 allows to express the trace $\operatorname{Tr} \mathcal{A}^{-s}$ in terms of some integral transforms related to the exponential traces $\operatorname{Tr} e^{-t\mathcal{A}}$, $\operatorname{Tr} e^{-t\sqrt{\mathcal{A}}}$, $\operatorname{Tr} (e^{-t\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})$, thus givining a global analogous of Proposition 2.70.

Proposition 2.80. Let the assumptions (2.159) and (2.209) be fulfilled (so that Dirichlet boundary conditions are prescribed on $\partial\Omega$); then, for all $s \in \Sigma_{d/2}$ there hold

$$Tr\mathcal{A}^{-s} = \frac{1}{\Gamma(s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, Tre^{-\mathbf{t}\mathcal{A}} \,, \qquad (2.297)$$

$$Tr\mathcal{A}^{-s} = \frac{1}{\Gamma(2s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{2s-1} \, Tre^{-\mathbf{t}\sqrt{\mathcal{A}}} \,, \qquad (2.298)$$

$$Tr\mathcal{A}^{-s} = \frac{1}{\Gamma(2s-1)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{2s-2} \, Tr(e^{-\mathbf{t}\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}) \,. \tag{2.299}$$

Proof. As an example, we show how to derive Eq. (2.297). To this purpose, recall once more the series expansions (2.289) (2.292) for Tr \mathcal{A}^{-s} and Tr $e^{-t\mathcal{A}}$, resectively; these allow to rephrase Eq. (2.297) as

$$\sum_{n=0}^{+\infty} \omega_n^{-2s} = \frac{1}{\Gamma(s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, \sum_{n=0}^{+\infty} e^{-\mathbf{t}\omega_n^2} \,. \tag{2.300}$$

Therefore, in view of the identity $z^{-s} = \frac{1}{\Gamma(s)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} e^{-tz}$ (see [122], page 139, Eq. 5.9.1), the thesis follows if the integral and the series in Eq. (2.300) can be interchanged. In fact, the mentioned operation can be performed in consequence of the Fubini-Tonelli theorem [134] since

$$\sum_{n=0}^{+\infty} \int_0^{+\infty} d\mathbf{t} \, |\mathbf{t}^{s-1} \, e^{-\mathbf{t}\omega_n^2}| = \sum_{n=0}^{+\infty} \omega_n^{-2\Re s} < +\infty \ ,$$

where the finiteness of the last expression is granted by item ii) of Corollary 2.75. \Box

2.8 Mellin transforms and their analytic continuation.

General definitions and some basic relations.

Let us first fix a few notations to be employed throughout the present section. We indicate with $\overline{\mathbb{R}}$ the system of extended real numbers, i.e.,

$$\overline{\mathbb{R}} := \mathbb{R} \cup \{\pm \infty\} \equiv [-\infty, +\infty] ; \qquad (2.301)$$

moreover, for any pair $r_1, r_2 \in \mathbb{R}$ with $r_1 < r_2$, we put

$$\Sigma_{(r_1, r_2)} := \{ s \in \mathbb{C} \mid r_1 < \Re s < r_2 \} .$$
(2.302)

If $r_2 = +\infty$, in accordance with Eq. (2.133), we use the short-hand notation

$$\Sigma_{r_1} \equiv \Sigma_{(r_1, +\infty)} . \tag{2.303}$$

Definition 2.81. Let $r_1, r_2 \in \mathbb{R}$ with $r_1 < r_2$. A measurable function $F : (0, +\infty) \to \mathbb{C}$ is said to be of *type* (r_1, r_2) if the map $(0, +\infty) \to \mathbb{C}$, $\mathbf{t} \mapsto \mathbf{t}^{s-1}F(\mathbf{t})$ is integrable for all $s \in \Sigma_{(r_1, r_2)}$; we use the notation

$$\mathcal{M}_{(r_1, r_2)} := \{ F : (0, +\infty) \to \mathbb{C} \mid F \text{ is of type } (r_1, r_2) \} \quad (\mathcal{M}_{r_1} \equiv \mathcal{M}_{(r_1, +\infty)}) . \quad (2.304)$$

For $F \in \mathcal{M}_{(r_1, r_2)}$ and $s \in \Sigma_{(r_1, r_2)}$, we put

$$\mathfrak{M}[F](s) := \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} F(\mathbf{t}) , \qquad (2.305)$$

and refer to the map $\mathfrak{M}[F]: \Sigma_{(r_1,r_2)} \to \mathbb{C}, s \mapsto \mathfrak{M}[F](s)$ as the Mellin transform of F; the complex domain $\Sigma_{(r_1,r_2)}$ will be called the strip of definition of $\mathfrak{M}[F]$.

Remark 2.26. The whole theory of Mellin transforms could be formulated with much more generality; for example, one could only replace the function F with a distributionlike object acting on some suitable space of test functions. As a matter of fact, many of the results discussed in the present subsection continue to hold also in this generalized framework (for further details on this topic, see, e.g., [157, 30]). Nevertheless, such a broader generality is unecessary in the present work and will therefore be neglected: we only consider the Mellin transform of ordinary, Lebesgue-integrable functions, since this is the case of interest for the applications to be studied in the following.

Lemma 2.82. (Operational relations). Let $F : (0, +\infty) \to \mathbb{C}$. Then, for all $r_1, r_2 \in \mathbb{R}$ with $r_1 < r_2$ the following results are fulfilled. i) Let $a \in (0, +\infty)$ and consider the function

$$F_a: (0, +\infty) \to \mathbb{C}$$
, $t \mapsto F_a(\mathbf{t}) := F(a \mathbf{t})$. (2.306)

Then, there hold

$$F_a \in \mathcal{M}_{(r_2, r_1)} \quad \Leftrightarrow \quad F \in \mathcal{M}_{(r_1, r_2)} ;$$

$$(2.307)$$

$$\mathfrak{M}[F_a](s) = a^{-s} \mathfrak{M}[F](s) \quad \text{for all } s \in \Sigma_{(r_1, r_2)} .$$
(2.308)

ii) Let $a \in \mathbb{R} \setminus \{0\}$ and consider the function

$$G_a: (0, +\infty) \to \mathbb{C}$$
, $\mathbf{t} \mapsto G_a(\mathbf{t}) := F(\mathbf{t}^a)$. (2.309)

Then, there hold

$$G_a \in \mathcal{M}_{(a\,r_2,\,a\,r_1)} \quad \Leftrightarrow \quad F \in \mathcal{M}_{(r_1,r_2)} ;$$
 (2.310)

$$\mathfrak{M}[G_a](s) = \frac{1}{|a|} \mathfrak{M}[F]\left(\frac{s}{a}\right) \qquad \text{for all } s \in \Sigma_{(ar_1, ar_2)} .$$

$$(2.311)$$

iii) Let $\alpha \in \mathbb{C}$ and consider the function

$$H_{\alpha}: (0, +\infty) \to \mathbb{C} , \quad \mathbf{t} \mapsto H_{\alpha}(\mathbf{t}) = \mathbf{t}^{\alpha} F(\mathbf{t}) .$$
 (2.312)

Then, there hold

$$H_{\alpha} \in \mathcal{M}_{(r_1 - \Re \alpha, r_2 - \Re \alpha)} \quad \Leftrightarrow \quad F \in \mathcal{M}_{(r_1, r_2)} ; \qquad (2.313)$$

$$\mathfrak{M}[H_{\alpha}](s) = \mathfrak{M}[F](s+\alpha) \quad \text{for all } s \in \Sigma_{(r_1-\Re\alpha, r_2-\Re\alpha)} .$$
(2.314)

Proof. All the statements i)-iii) follow from Definition 2.81 by simple changes of the variable of integration, suitably defined in view of the explicit expressions (2.306), (2.309) and (2.312) for the functions F_a , G_a and H_α in terms of F.

As an example, let us give a few more details about the derivation of Eq. (2.311). First note that, Eq. (2.309) and the definition (2.305) of Mellin transform give $\mathfrak{M}[G_a](s) = \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} F(\mathbf{t}^a)$; making the change of variable $\mathbf{t} \mapsto \mathbf{t}^{1/a}$ (³³), one infers $\mathfrak{M}[G_a](s) = \frac{1}{|a|} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{\frac{s}{a}-1} F(\mathbf{t})$. Then, Eq. (2.311) follows recalling again the definition (2.305). \Box

Lemma 2.83. Let $F \in L^1_{loc}(0, +\infty)$ and assume that, for some $r_1, r_2 \in \mathbb{R}$ with $r_1 < r_2$,

$$F(\mathbf{t}) = \begin{cases} O(\mathbf{t}^{-r_1}) & \text{for } \mathbf{t} \to 0^+, \\ O(\mathbf{t}^{-r_2}) & \text{for } \mathbf{t} \to +\infty \end{cases}$$
(2.315)

Then, $F \in \mathcal{M}_{(r_1,r_2)}$.

Remark 2.27. In particular, if $r_1 = -\infty$ or $r_2 = +\infty$, the notations $F(\mathbf{t}) = O(\mathbf{t}^{+\infty})$ for $\mathbf{t} \to 0^+$ and $F(\mathbf{t}) = O(\mathbf{t}^{-\infty})$ for $\mathbf{t} \to +\infty$ in Eq. (2.315) mean, respectively, $F(\mathbf{t}) = O(\mathbf{t}^r)$ and $F(\mathbf{t}) = O(\mathbf{t}^{-r})$ for each $r \in (0, +\infty)$.

³³Of course, the restriction on the real parameter a in both items i) and ii) are necessary in order to guarantee that the change of variables to be employed in the proof (such as $\mathbf{t} \mapsto \mathbf{t}^{1/a}$) are well defined.

Proof. First note that, for any $T_1, T_2 \in (0, +\infty)$, there holds the following decomposition:

$$\int_{0}^{+\infty} d\mathbf{t} \, |\mathbf{t}^{s-1}F(\mathbf{t})| = \int_{0}^{T_{1}} d\mathbf{t} \, |\mathbf{t}^{s-1}F(\mathbf{t})| + \int_{T_{1}}^{T_{2}} d\mathbf{t} \, |\mathbf{t}^{s-1}F(\mathbf{t})| + \int_{T_{2}}^{+\infty} d\mathbf{t} \, |\mathbf{t}^{s-1}F(\mathbf{t})| \, . \quad (2.316)$$

By definition, F is of type (r_1, r_2) if and only if the integral in the left-hand side of the above equation is finite for all $s \in \Sigma_{(r_1, r_2)}$; this happens if and only if each of the integrals in the right-hand side of Eq. (2.316) are finite, since the integrand functions are all positive. Hereafter we prove the last statement which, according to the previous considerations, gives the thesis.

Of course, the assumption (2.315) means that there exist $T_1^*, T_2^* \in (0, +\infty)$ and a pair of positive constants M_1, M_2 such that

$$|F(\mathbf{t})| \le \begin{cases} M_1 \, \mathbf{t}^{-r_1} & \text{for } 0 < \mathbf{t} < T_1^* \\ M_2 \, \mathbf{t}^{-r_2} & \text{for } \mathbf{t} > T_2^* \end{cases};$$

this fact allows to infer easily the bounds

$$\int_{0}^{T_{1}^{*}} d\mathbf{t} |\mathbf{t}^{s-1}F(\mathbf{t})| \leq M_{1} \frac{(T_{1}^{*})^{\Re s-r_{1}}}{\Re s-r_{1}} \quad \text{for all } s \in \mathbb{C} \text{ with } \Re s > r_{1} ,$$
$$\int_{T_{2}^{*}}^{+\infty} d\mathbf{t} |\mathbf{t}^{s-1}F(\mathbf{t})| \leq M_{2} \frac{(T_{2}^{*})^{\Re s-r_{2}}}{r_{2}-\Re s} \quad \text{for all } s \in \mathbb{C} \text{ with } \Re s < r_{2} .$$

Next note that, for $T_1^*, T_2^* \in (0, +\infty)$ as above and for all $s \in \mathbb{C}$, there holds

$$\int_{T_1^*}^{T_2^*} d\mathbf{t} \, |\mathbf{t}^{s-1} F(\mathbf{t})| \le \max\left((T_1^*)^{\Re s-1}, (T_2^*)^{\Re s-1} \right) \int_{T_1^*}^{T_2^*} d\mathbf{t} \, |F(\mathbf{t})| < +\infty \; ,$$

where the finiteness of the integral on the right-hand side is granted by the assumption $F \in L^1_{loc}(0, +\infty)$.

Summing up, we have shown that the three integrals on the right-hand side of Eq. (2.316) are all finite at the same time for $s \in \mathbb{C}$ with $r_1 < \Re s < r_2$, i.e., for $s \in \Sigma_{(r_1,r_2)}$; as anticipated above, this yields the thesis.

Remark 2.28. The arguments presented in the proof of Lemma 2.83 allow to infer the results discussed hereafter, giving a partial converse of the cited lemma.

On the one hand, the assumption $F \in L^1_{loc}(0, +\infty)$ is in fact a necessary condition in order to have $F \in \mathcal{M}_{(r_1, r_2)}$ for some $r_1, r_2 \in \mathbb{R}$ $(r_1 < r_2)$ (³⁴).

³⁴To prove this fact note that, for any $T_1, T_2 \in (0, +\infty)$ and for all $s \in \mathbb{C}$, there holds

$$\int_{T_1}^{T_2} d\mathbf{t} |F(\mathbf{t})| \le \max\left(T_1^{1-\Re s}, T_2^{1-\Re s}\right) \int_{T_1}^{T_2} d\mathbf{t} |\mathbf{t}^{s-1} F(\mathbf{t})| .$$

On the other hand, it appears that the restrictions $\Re s > r_1$ and $\Re s < r_2$ descend, respectively, from the small and large **t** behaviour of F; more precisely, the following bounds must hold:

$$r_1 \ge \inf \left\{ r \in \mathbb{R} \mid \mathbf{t}^{r-1} f(\mathbf{t}) \in L^1(0, T) \text{ for some } T > 0 \right\} ;$$

$$r_2 \le \sup \left\{ r \in \mathbb{R} \mid \mathbf{t}^{r-1} f(\mathbf{t}) \in L^1(T, +\infty) \text{ for some } T > 0 \right\} .$$
(2.317)

Proposition 2.84. Let $r_1, r_2 \in \mathbb{R}$ with $r_1 < r_2$ and let $F \in \mathcal{M}_{(r_1, r_2)}$. Then, the Mellin transform $\mathfrak{M}[F]$ of F is analytic in the strip $\Sigma_{(r_1, r_2)}$; moreover, for any $k \in \mathbb{N}$ and for all $s \in \Sigma_{(r_1, r_2)}$, there holds

$$\frac{d^k \mathfrak{M}[F]}{ds^k}(s) = \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \left(\ln \mathbf{t}\right)^k F(\mathbf{t}) \,. \tag{2.318}$$

Proof. It suffices to prove that Eq. (2.318) holds for any $k \in \mathbb{N}$ and for all $s \in \Sigma_{(r_1,r_2)}$; indeed, for k = 1 this fact shows, in particular, that $\mathfrak{M}[f]$ is holomorphic on the strip $\Sigma_{(r_1,r_2)}$, hence analytic therein.

To derive Eq. (2.318), first consider the definition (2.305) of $\mathfrak{M}[f]$ and note that $\partial_s^k \mathbf{t}^{s-1} = \mathbf{t}^{s-1}(\ln \mathbf{t})^k$; then, it appears that Eq. (2.318) follows from the Lebesgue dominated convergence theorem if, locally for $s \in \Sigma_{(r_1,r_2)}$, the map $(0, +\infty) \to \mathbb{C}$, $\mathbf{t} \mapsto \mathbf{t}^{s-1}(\ln \mathbf{t})^k F(\mathbf{t})$ admits an integrable dominant function (independent of s). To prove the last statement, let us fix $s_0 \in (r_1, r_2)$ arbitrarily and choose $\vartheta > 0$ so that $s_0 \pm 2\vartheta \in (r_1, r_2)$ (such a ϑ always exists since (r_1, r_2) is open); for any $k \in \mathbb{N}$, there exits a pair of positive constants C_1, C_2 (depending on ϑ, k) such that (3^5)

$$\left| (\ln \mathbf{t})^k \right| \le \begin{cases} C_1 \, \mathbf{t}^{-\vartheta} & \text{for } \mathbf{t} \in (0,1) \,, \\ C_2 \, \mathbf{t}^\vartheta & \text{for } \mathbf{t} \in (1,+\infty) \,. \end{cases}$$

Then, for all $s \in \Sigma_{(s_0 - \vartheta, s_0 + \vartheta)}$ it follows that

$$\left|\mathbf{t}^{s-1}(\ln \mathbf{t})^{k}F(\mathbf{t})\right| \leq C_{1}\left|\mathbf{t}^{s_{0}-2\vartheta-1}F(\mathbf{t})\right| + C_{2}\left|\mathbf{t}^{s_{0}+2\vartheta-1}F(\mathbf{t})\right|.$$
 (2.319)

Since $F \in \mathcal{M}_{(r_1, r_2)}$ and we have chosen $\vartheta > 0$ such that $s_0 \pm 2\vartheta \in (r_1, r_2)$, the right-hand side of the inequality (2.319) gives a dominant function which is integrable over $(0, +\infty)$; due to the above considerations, this completes the proof.

Remark 2.29. The previous Proposition 2.84 grants the analyticity of the Mellin transform $\mathfrak{M}[F]$ of any given function $F \in \mathcal{M}_{(r_1,r_2)}$ inside the open strip $\Sigma_{(r_1,r_2)}$. Then, it is natural to try and extend $\mathfrak{M}[F]$ to a wider region of the complex plane by means of analytic continuation, a topic we discuss in the remainder of the present section; more precisely, we present three main methods allowing to construct the mentioned analytic continuation of $\mathfrak{M}[F]$, under suitable (increasingly stricter) hypotheses for the function F.

$$C_1 = C_2 = \left(\frac{k}{\vartheta e}\right)^k$$
.

 $^{^{35}}$ It can be show by elementary methods that the optimal choice for both C_1 and C_2 is

Analytic continuation of Mellin transforms.

Proposition 2.85. (First method for analytic continuation: asymptotic expansion). Let $F \in L^1_{loc}(0, +\infty)$. Assume there exist $N, P \in \mathbb{N}$ and two families of coefficients $\mathfrak{a}_n, \mathfrak{f}_{np}$ $(n \in \{0, ..., N+1\}, p \in \{0, ..., P\})$ with

$$\mathfrak{a}_n \in \mathbb{R}$$
, $\mathfrak{a}_0 < \mathfrak{a}_1 < \ldots < \mathfrak{a}_N < \mathfrak{a}_{N+1}$ and $\mathfrak{f}_{np} \in \mathbb{C}$, (2.320)

such that there holds the asymptotic expansion

$$F(\mathbf{t}) = \sum_{n=0}^{N} \sum_{p=0}^{P} \mathfrak{f}_{np} \, \mathbf{t}^{\mathfrak{a}_{n}} (\ln \mathbf{t})^{p} + O(\mathbf{t}^{\mathfrak{a}_{N+1}}) \qquad for \, \mathbf{t} \to 0^{+} ; \qquad (2.321)$$

moreover, assume there exists $r \in \overline{\mathbb{R}}$ with $r > -\mathfrak{a}_0$ such that

$$F(\mathbf{t}) = O(\mathbf{t}^{-r}) \qquad for \ \mathbf{t} \to +\infty \ .$$
 (2.322)

Then $F \in \mathcal{M}_{(-\mathfrak{a}_0,r)}$ and its Mellin transform $\mathfrak{M}[F]$ can be analytically continued to a function which is meromorphic in the strip $\Sigma_{(-\mathfrak{a}_{N+1},r)}$, possibly with pole singularities at $s \in \{-\mathfrak{a}_0, ..., -\mathfrak{a}_N\}$. Furthermore, the analytic continuation of $\mathfrak{M}[F]$ (indicated with the same symbol) can be represented as follows, for all $s \in \Sigma_{(-\mathfrak{a}_{N+1},r)}$:

$$\mathfrak{M}[F](s) = \sum_{n=0}^{N} \sum_{p=0}^{P} \sum_{j=0}^{p} \left(\frac{(-1)^{j} p!}{(p-j)!} \right) \frac{\mathfrak{f}_{np} \, T^{s+\mathfrak{a}_{n}} \, (\ln T)^{p-j}}{(s+\mathfrak{a}_{n})^{j}} + \int_{0}^{T} d\mathbf{t} \, \mathbf{t}^{s-1} \left(F(\mathbf{t}) - \sum_{n=0}^{N} \sum_{p=0}^{P} \mathfrak{f}_{np} \, \mathbf{t}^{\mathfrak{a}_{n}} (\ln \mathbf{t})^{p} \right) + \int_{T}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, F(\mathbf{t}) \; .$$

$$(2.323)$$

Proof. First note that $F(\mathbf{t}) = O(\mathbf{t}^{\mathfrak{a}_0})$ for $\mathbf{t} \to 0^+$ in consequence of the assumption (2.321); this fact and the other hypotheses on F made in the present proposition grant, in view of Lemma 2.83, that $F \in \mathcal{M}_{(-\mathfrak{a}_0,r)}$.

Next, let us pass to the construction of the analytic continuation of the Mellin transform $\mathfrak{M}[F]$. To this purpose, recall that the expression $O(\mathbf{t}^{\mathfrak{a}_{N+1}})$ for $\mathbf{t} \to 0^+$ by definition means that

$$\exists T, M > 0 \text{ such that } |O(\mathbf{t}^{\mathfrak{a}_{N+1}})| \le M \mathbf{t}^{\mathfrak{a}_{N+1}} \text{ for all } 0 < \mathbf{t} < T ; \qquad (2.324)$$

for any such T and for all $s \in \Sigma_{(-\mathfrak{a}_0,r)}$, using the expansion (2.321), the definition (2.305) of $\mathfrak{M}[F]$ can be rephrased as

$$\mathfrak{M}[F](s) = \sum_{n=0}^{N} \sum_{p=0}^{P} \mathfrak{f}_{np} \int_{0}^{T} d\mathbf{t} \, \mathbf{t}^{s+\mathfrak{a}_{n}-1} \, (\ln \mathbf{t})^{p} + \int_{0}^{T} d\mathbf{t} \, \mathbf{t}^{s-1} \, O(\mathbf{t}^{\mathfrak{a}_{N+1}}) + \int_{T}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, F(\mathbf{t}) \, .$$
(2.325)

Hereafter we discuss in detail each of the terms in the right-hand side of Eq. (2.325); the conclusion of this analysis will be the proof of Eq. (2.323), giving the analytic continuation of $\mathfrak{M}[F]$ as stated in the proposition.
First, let us fix $n \in \{0, ..., N\}$, $p \in \{0, ..., P\}$ and consider the corresponding integral in the first sum on the right-hand side of Eq. (2.325); this integral can be evaluated explicitly, for all $s \in \Sigma_{-\mathfrak{a}_n}$, giving $\binom{36}{5}$

$$\int_0^T d\mathbf{t} \, \mathbf{t}^{s+\mathfrak{a}_n-1} \, (\ln \mathbf{t})^p = T^{s+\mathfrak{a}_n} \sum_{j=0}^p \left(\frac{(-1)^j p!}{(p-j)!} \right) \frac{(\ln T)^{p-j}}{(s+\mathfrak{a}_n)^j} \,. \tag{2.326}$$

Despite the above identity is derived under the assumption $\Re s > -\mathfrak{a}_n$, it can be interpreted to give the analytic continuation of the integral on the left-hand side to a function which is meromorphic on the whole complex plane and whose only singularity is a pole of order p at $s = -\mathfrak{a}_n$.

Concerning the second and the third integral in Eq. (2.325), we state that these integrals are finite and determine analytic functions of s for all $s \in \mathbb{C}$ with $\Re s > -\mathfrak{a}_{N+1}$ and $\Re s < r$, respectively. In fact, the convergence of $\int_0^T d\mathbf{t} \, \mathbf{t}^{s-1} O(\mathbf{t}^{\mathfrak{a}_{N+1}})$ follows easily using the bound (2.324), while the integral $\int_T^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} F(\mathbf{t})$ can be shown to be finite moving along the same lines as in the proof of Lemma 2.83; the analyticity of these expressions can be proved by arguments similar to the ones presented in the proof of Proposition 2.84. So, as anticipated before, Eq.s (2.325) (2.326) imply the representation (2.323) which gives the analytic continuation of $\mathfrak{M}[F]$ to a function meromorphic in the strip $\Sigma_{(-\mathfrak{a}_{N+1},r)}$, with poles at $s \in \{-\mathfrak{a}_0, ..., -\mathfrak{a}_N\}$.

Corollary 2.86. Let $k \in \mathbb{N}$ and $F \in C^{k+1}([0, +\infty))$; moreover assume that, for some $r \in \mathbb{R}$ with r > 0, there holds

$$F(\mathbf{t}) = O(\mathbf{t}^{-r}) \qquad for \ \mathbf{t} \to +\infty \ .$$
 (2.327)

Then $F \in \mathcal{M}_{(0,r)}$ and its Mellin transform $\mathfrak{M}[F]$ can be analytically continued to a function meromorphic on the strip $\Sigma_{(-(k+1),r)}$, possibly with simple poles at $s \in \{0, -1, ..., -k\}$; in particular, for $s \in \Sigma_{(-(k+1),r)}$ there holds

$$\mathfrak{M}[F](s) = \sum_{n=0}^{k} \frac{F^{(n)}(0)}{n!(s+n)} T^{s+n} + \int_{0}^{T} d\mathbf{t} \mathbf{t}^{s-1} \left(F(\mathbf{t}) - \sum_{n=0}^{k} \frac{F^{(n)}(0)}{n!} \mathbf{t}^{n} \right) + \int_{T}^{+\infty} d\mathbf{t} \mathbf{t}^{s-1} F(\mathbf{t}) .$$
(2.328)

$$\int_0^T d\mathbf{t} \, \mathbf{t}^{s+\mathfrak{a}_n-1} \, (\ln \mathbf{t})^p = \frac{d^p}{ds^p} \int_0^T d\mathbf{t} \, \mathbf{t}^{s+\mathfrak{a}_n-1} = \frac{d^p}{ds^p} \left(\frac{T^{s+\mathfrak{a}_n}}{s+\mathfrak{a}_n} \right) \; .$$

Eq. (2.326) follows straightforwardly from the last expression above using the general Leibnitz rule.

³⁶In order to prove Eq. (2.326), first note that $\mathbf{t}^{s+\mathfrak{a}_n-1}(\ln \mathbf{t})^p = \partial_s^p(\mathbf{t}^{s+\mathfrak{a}_n-1})$; then, for all $s \in \mathbb{C}$ with $\Re s > -\mathfrak{a}_n$, by arguments similar to those presented in the proof of Proposition 2.84 one can resort to the Lebesgue dominated convergence theorem to infer

Proof. The assumption $F \in C^{k+1}([0, +\infty))$ ensures, in particular, that both $F \in L^1_{loc}(0, +\infty)$ and

$$F(\mathbf{t}) = \sum_{n=0}^{k-1} \frac{F^{(n)}(0)}{n!} \mathbf{t}^n + O(\mathbf{t}^k) \quad \text{for } \mathbf{t} \to 0^+ ;$$

then, the thesis follows straightforwardly from Proposition 2.85.

Corollary 2.87. Let $F \in L^1_{loc}(0, +\infty)$. Assume there exist $N, P \in \mathbb{N}$, and two families of coefficients $\mathfrak{c}_n, \mathfrak{f}_{np}$ $(n \in \{0, ..., N\}, p \in \{0, ..., P\})$ with

$$\mathfrak{c}_n \in \mathbb{R}$$
, $\mathfrak{c}_0 < \mathfrak{c}_1 < \ldots < \mathfrak{c}_N < \mathfrak{c}_{N+1}$ and $\mathfrak{f}_{np} \in \mathbb{C}$, (2.329)

such that there holds the asymptotic expansion

$$F(\mathbf{t}) = \sum_{n=0}^{N} \sum_{p=0}^{P} \mathfrak{f}_{np} \, \mathbf{t}^{-\mathfrak{c}_n} (\ln \mathbf{t})^p + O(\mathbf{t}^{-\mathfrak{c}_{N+1}}) \qquad for \, \mathbf{t} \to +\infty \; ; \qquad (2.330)$$

moreover, assume there exist $r \in \mathbb{R}$ with $r > -\mathfrak{c}_0$ such that

$$F(\mathbf{t}) = O(\mathbf{t}^r) \qquad for \, \mathbf{t} \to 0^+ \;.$$
 (2.331)

Then $F \in \mathcal{M}_{(-r,\mathfrak{c}_0)}$ and its Mellin transform $\mathfrak{M}[F]$ can be analytically continued to a function which is meromorphic on the strip $\Sigma_{(-r,\mathfrak{c}_{N+1})}$, possibly with pole singularities at $s \in {\mathfrak{c}_0, ..., \mathfrak{c}_N}$. Furthermore, the analytic continuation of $\mathfrak{M}[F]$ (indicated with the same symbol) can be represented as follows, for all $s \in \Sigma_{(-r,\mathfrak{c}_{N+1})}$:

$$\mathfrak{M}[F](s) = \sum_{n=0}^{N} \sum_{p=0}^{P} \sum_{j=0}^{p} \left(\frac{(-1)^{j} p!}{(p-j)!} \right) \frac{\mathfrak{f}_{np} T^{s-\mathfrak{c}_{n}} (\ln T)^{p-j}}{(s-\mathfrak{c}_{n})^{j}} + \int_{T}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \left(F(\mathbf{t}) - \sum_{n=0}^{N} \sum_{p=0}^{P} \mathfrak{f}_{np} \, \mathbf{t}^{\mathfrak{c}_{n}} (\ln \mathbf{t})^{p} \right) + \int_{0}^{T} d\mathbf{t} \, \mathbf{t}^{s-1} F(\mathbf{t}) \; .$$
(2.332)

Proof. Consider the function $F_{-1}: (0, +\infty) \to \mathbb{C}$, $\mathbf{t} \mapsto F_{-1}(\mathbf{t}) := F(1/\mathbf{t})$ (see Eq. (2.306), with a = -1). Of course, $F_{-1} \in L^{1}_{loc}(0, +\infty)$; moreover, there hold the following relations:

$$F_{-1}(\mathbf{t}) = \begin{cases} \sum_{n=0}^{N} \sum_{p=0}^{P} f_{np} \mathbf{t}^{\mathfrak{c}_{n}} (-\ln \mathbf{t})^{p} + O(\mathbf{t}^{\mathfrak{d}}) & \text{for } \mathbf{t} \to 0^{+} \\ O(\mathbf{t}^{-r}) & \text{for } \mathbf{t} \to +\infty \end{cases}$$

Therefore, due to Proposition 2.85, we have $F_{-1} \in \mathcal{M}_{(-\mathfrak{c},r)}$; furthermore, the Mellin transform $\mathfrak{M}[F_{-1}]$ can be analytically continued to the strip $\Sigma_{(-\mathfrak{d},r)}$ and therein it admits the following representation, for any $T_{\star} \in (0, +\infty)$:

$$\mathfrak{M}[F_{-1}](s) = \sum_{n=0}^{N} \sum_{p=0}^{P} \sum_{j=0}^{p} \left(\frac{(-1)^{p-j} p!}{(p-j)!} \right) \frac{\mathfrak{f}_{np} T_{\star}^{s+\mathfrak{c}_{n}} (\ln T_{\star})^{p-j}}{(s+\mathfrak{c}_{n})^{j}} + \int_{0}^{T_{\star}} d\mathbf{t} \, \mathbf{t}^{s-1} \left(F_{-1}(\mathbf{t}) - \sum_{n=0}^{N} \sum_{p=0}^{P} (-1)^{p} \mathfrak{f}_{np} \, \mathbf{t}^{+\mathfrak{c}_{n}} (\ln \mathbf{t})^{p} \right) + \int_{T_{\star}}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} F_{-1}(\mathbf{t}) \,.$$

$$(2.333)$$

The thesis can be easily inferred using statement i) in Lemma 2.82; in particular, Eq. (2.332) follows from Eq. (2.333) recalling that $\mathfrak{M}[F](s) = \mathfrak{M}[F_{-1}](-s)$ (see Eq. (2.308), again with a = -1), performing the change of variable $\mathbf{t} \mapsto 1/\mathbf{t}$ in the integrals appearing in Eq. (2.333) and setting $T := 1/T_{\star}$.

Proposition 2.88. (Second method for analytic continuation: integration by parts). Let $F \in C^k([0, +\infty))$ for some $k \in \mathbb{N}$; moreover, assume there exist $r \in \mathbb{R}$ with r > 0 and a family of coefficients r_n $(n \in \{0, ..., k-1\})$ with

$$r_n \ge r+n \ , \tag{2.334}$$

such that

$$F^{(n)}(\mathbf{t}) = O(\mathbf{t}^{-r_n}) \qquad \text{for } \mathbf{t} \to +\infty .$$
(2.335)

Then $F \in \mathcal{M}_{(0,r)}$ and its Mellin transform $\mathfrak{M}[F]$ can be analytically continued to a function which is meromorphic on the strip $\Sigma_{(-k,r_k-k)}$, possibly with simple poles at $s \in \{0, -1, ..., -(k-1)\}$; in particular, for $s \in \Sigma_{(-k,r_k-k)}$ there holds

$$\mathfrak{M}[F](s) = \frac{(-1)^k}{s(s+1)\dots(s+k-1)} \ \mathfrak{M}\left[F^{(k)}\right](s+k) \ . \tag{2.336}$$

Proof. First note that the function F fulfills all the hypotheses of Corollary 2.86, since $F \in C^k([0, +\infty))$ and $F(\mathbf{t}) = O(\mathbf{t}^{-r_0})$ in the limit $\mathbf{t} \to +\infty$ for some $r_0 \ge r > 0$; in particular, this suffices to infer that $F \in \mathcal{M}_{(0,r)}$.

In order to derive the representation (2.336) for the Mellin transform $\mathfrak{M}[F]$, consider the general definition (2.305); integrating by parts k times, we obtain

$$\mathfrak{M}[F](s) = \tag{2.337}$$

$$\sum_{n=0}^{k-1} \frac{(-1)^n}{s(s+1)\dots(s+n)} \left[\mathbf{t}^{s+n} F^{(n)}(\mathbf{t}) \right]_0^{+\infty} + \frac{(-1)^k}{s(s+1)\dots(s+k-1)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s+k-1} \, F^{(k)}(\mathbf{t}) \, .$$

Now, fix $n \in \{0, ..., k-1\}$ and consider the boundary term $[\mathbf{t}^{s+n}F^{(n)}(t)]_0^{+\infty}$. On the one hand, since $F \in C^k([0, +\infty))$ $(k \ge n)$, there holds

$$\lim_{\mathbf{t}\to 0^+} \left(\mathbf{t}^{s+n} F^{(n)}(\mathbf{t}) \right) = 0 \quad \text{for } s \in \mathbb{C} \text{ with } \Re s > -n ;$$

on the other hand, the assumption (2.335) allows to infer

$$\lim_{\mathbf{t} \to +\infty} \left(\mathbf{t}^{s+n} F^{(n)}(\mathbf{t}) \right) = 0 \quad \text{for } s \in \mathbb{C} \text{ with } \Re s < r_n - n .$$

The hypothesis (2.334) grants that the above results do indeed hold for all $s \in \Sigma_{(0,r)}$; thus, within this strip Eq. (2.337) reduces to $\mathfrak{M}[F](s) = \frac{(-1)^k}{s(s+1)\dots(s+k-1)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s+k-1} F^{(k)}(\mathbf{t})$, which is equivalent to Eq. (2.336) by the definition (2.305) of Mellin transform.

To conclude note that, since F fulfills the hypotheses of Corollary 2.86 (here employed with k+1 replaced by k), there holds $F^{(k)} \in \mathfrak{M}_{(0,r_k)}$; therefore, the integral in Eq. (2.336) converges for all $s \in \Sigma_{(-k,r_k-k)}$. Since $\Sigma_{(-k,r_k-k)} \supset \Sigma_{(0,r)}$, the representation (2.336) gives the analytic continuation of $\mathfrak{M}[F]$ within this strip to a meromorphic function with possible poles at $s \in \{0, -1, ..., -(k-1)\}$.

Proposition 2.89. (Third method for analytic continuation: complex integration). Let $F: (0, +\infty) \to \mathbb{C}$ be a function admitting an analytic extension (indicated with the same symbol) to a complex open neighbour $\mathcal{U} \subset \mathbb{C}$, such that

$$[0, +\infty) \subset \mathcal{U}$$
 and $\sup_{z \in \mathcal{U}} |\Im z| > 0;$ (2.338)

moreover, assume that, for some $r \in \overline{\mathbb{R}}$ with r > 0,

$$F(\mathbf{t}) = O\left((\Re \mathbf{t})^{-r}\right) \qquad \text{for } \Re \mathbf{t} \to +\infty .$$
 (2.339)

Then $F \in \mathcal{M}_{(0,r)}$ and its Mellin transform $\mathfrak{M}[F]$ can be analytically continued to a function meromorphic on the infinite left strip $\Sigma_{(-\infty,r)}$, possibly with simple poles at $s \in \mathbb{Z} \cap (-\infty,r)$; in particular, for $s \in \Sigma_{(-\infty,r)} \setminus \{\mathbb{Z} \cap (-\infty,r)\}$ there holds

$$\mathfrak{M}[F](s) = \frac{e^{-i\pi s}}{2i\sin(\pi s)} \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{s-1} F(\mathbf{t}) \,, \qquad (2.340)$$

where $\mathfrak{H} \subset \mathcal{U}$ denotes the Hankel contour, that is a simple path contained in the open neighbour \mathcal{U} that starts in the upper half-plane near $+\infty$, encircles the origin counterclockwise and returns to $+\infty$ in the lower half-plane (see Fig. 2.1 below).

Proof. Also in this case the assumptions made for the function F are sufficient to infer that $F \in \mathfrak{M}_{(0,r)}$, by Corollary 2.86.

Next, let us show how to derive the representation (2.340) for the Mellin transform $\mathfrak{M}[F]$; to this purpose, fix $s \in \Sigma_{(0,r)}$ and consider the integral

$$\mathfrak{N}[F](s) := \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{s-1} F(\mathbf{t}) \tag{2.341}$$

(note that the assumption $\Re s < r$ and the hypotheses made for F ensure that the above integral is well-defined). To proceed note that, for any δ with $0 < \delta < \sup_{z \in \mathcal{U}} |\Im z|$, the



Figure 2.1: The Hankel contour \mathfrak{H} .

Hankel contour \mathfrak{H} is homotopic to the path \mathfrak{H}_{δ} described as follows:

$$egin{aligned} &\mathfrak{H}_{\delta} = \mathfrak{H}_{\delta}^{+} \cup \mathfrak{H}_{\delta}^{0} \cup \mathfrak{H}_{\delta}^{-} \ , \qquad ext{with} \ &\mathfrak{H}_{\delta}^{\pm} := \{ \mathbf{t} \in \mathbb{C} \mid \mathbf{t} = \tau \pm i\delta, \ \tau \in [0, +\infty) \} \ , \ &\mathfrak{H}_{\delta}^{0} := \{ \mathbf{t} \in \mathbb{C} \mid \mathbf{t} = \delta \, e^{i\theta}, \ \theta \in (\pi/2, 3\pi/2) \} \ . \end{aligned}$$

Due to this fact and to the analyticity of F, the path \mathfrak{H} in Eq. (2.341) can be replaced with \mathfrak{H}_{δ} ; thus,

$$\mathfrak{N}[F](s) = \mathfrak{N}_{\delta}^{+}[F](s) + \mathfrak{N}_{\delta}^{0}[F](s) + \mathfrak{N}_{\delta}^{-}[F](s) ,$$

$$\mathfrak{N}_{\delta}^{\pm}[F](s) := \mp \int_{0}^{+\infty} d\tau \, (\tau \pm i\delta)^{s-1} F(\tau \pm i\delta) ,$$

$$\mathfrak{N}_{\delta}^{0}[F](s) := i \int_{\pi/2}^{3\pi/2} d\theta \, (\delta \, e^{i\theta})^{s} \, F(\delta \, e^{i\theta}) .$$
(2.342)

Consider the limit $\delta \to 0^+$. Notice that, in this limit, $(\tau+i\delta)^{s-1} \to \tau^{s-1}$ while $(\tau-i\delta)^{s-1} \to e^{2i\pi(s-1)}\tau^{s-1} = e^{2i\pi s}\tau^{s-1}$; moreover, $F(\tau \pm i\delta) \to F(\tau)$. Due to these results, by Lebesgue dominated convergence theorem it follows that $\lim_{\delta \to 0^+} \mathfrak{N}^+_{\delta}[F](s) = -\int_0^{+\infty} d\tau \ \tau^{s-1}F(\tau)$ and $\lim_{\delta \to 0^+} \mathfrak{N}^-_{\delta}[F](s) = e^{2i\pi s}\int_0^{+\infty} d\tau \ \tau^{s-1}F(\tau)$. Concerning the integral $\mathfrak{N}^0_{\delta}[F](s)$, note that the analyticity of F implies the existence of a positive constant C such that $|F(\delta e^{i\theta})| \leq C$ for all $\theta \in (0, 2\pi)$; thus, for $\Re s > 0$, one has

$$|\mathfrak{N}^0_{\delta}[F](s)| \le C \,\delta^{\Re s} \int_{\pi/2}^{3\pi/2} d\theta \ e^{-(\Im s)\theta} \to 0 \qquad \text{for } \delta \to 0^+ \ . \tag{2.343}$$

Summing up, in the limit $\delta \to 0^+$ Eq. (2.342) gives

$$\mathfrak{N}[F](s) = (e^{2i\pi s} - 1) \int_0^{+\infty} d\tau \ \tau^{s-1} F(\tau) \ . \tag{2.344}$$

Noting that $e^{2i\pi s} - 1 = 2ie^{i\pi s}\sin(\pi s)$ and recalling the definitions (2.305) (2.341) of $\mathfrak{M}[F]$, $\mathfrak{N}[F]$, Eq. (2.344) allow us to obtain the representation (2.340) of the Mellin transform $\mathfrak{M}[F]$ for all $s \in \Sigma_{(0,r)} \setminus \{\mathbb{Z} \cap (0,r)\}$.

To conclude, note that the hypothesis $\Re s > 0$ was employed only to justify the intermediate result (2.343); nevertheless, to grant the finiteness of the integral $\Re[F]$ of Eq. (2.341) it suffices to require $\Re s < r$. Therefore, the representation (2.340) makes sense for all $s \in \Sigma_{(-\infty,r)}$ and can be interpreted as the analytic continuation of $\mathfrak{M}[F]$ to this strip, yielding the thesis.

Remark 2.30. i) Let $F : (0, +\infty) \to \mathbb{C}$ be an analytic function fulfilling all the hypotheses of Proposition 2.89; for any $q \in \mathbb{Z}$, consider the analytic extension of the function introduced in Eq. (2.312):

$$H_q: \mathcal{U} \to \mathbb{C} , \qquad \mathbf{t} \mapsto H_q(\mathbf{t}) := \mathbf{t}^q F(\mathbf{t}) .$$
 (2.345)

Proposition 2.89 and the statement iii) of Lemma 2.82 allow to infer that H_q can be analytically continued to a function which is meromorphic on the infinite left strip $\Sigma_{(-\infty,r-q)}$; due to the results in Eq.s (2.314) (2.340), this analytic continuation can be represented as

$$\mathfrak{M}[H_q](s) = \frac{e^{-i\pi s}}{2i\sin(\pi s)} \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{s-1} H_q(\mathbf{t}) \,, \qquad (2.346)$$

where it was also used the fact that

$$e^{-i\pi q} / \sin(\pi(s+q)) = 1 / \sin(\pi s)$$
 for $q \in \mathbb{Z}$. (2.347)

ii) Consider the very well-known *reflection relation* (see, e.g., [122], page 138, Eq. 5.5.3)

$$\Gamma(s)\Gamma(1-s) = \frac{\pi}{\sin(\pi s)}$$
 for $s \in \mathbb{C} \setminus \mathbb{Z}$. (2.348)

Using this identity, Eq. (2.340) can be reformulated as follows, for all $s \in \mathbb{C} \setminus \mathbb{Z}$:

$$\frac{1}{\Gamma(s)} \mathfrak{M}[F](s) = \frac{e^{-i\pi s} \Gamma(1-s)}{2\pi i} \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{s-1} F(\mathbf{t}) , \qquad (2.349)$$

In some cases of interest for the applications to be considered in the following, Eq. (2.349) yields a very effective method for computing explicitly the analytic continuation of the Mellin transform of certain functions; in fact, for suitable $s \in \mathbb{Z}$, the integral in the right-hand side of Eq. (2.349) can be easily computed via the residue theorem, since F is assumed to be analytic in the region enclosed by the Hankel contour \mathfrak{H} .

Analytic continuation of the Dirichlet kernels.

Let us consider the framework of Sections 2.6 and 2.7; so, assume again $\Omega \subset \mathbb{R}^d$ to be an arbitrary domain and let \mathcal{A} be the admissible operator obtained restricting the differential operator $A = -\Delta + V$ ($V \in C^{\infty}(\Omega)$) to an admissible domain $\text{Dom}(\mathcal{A}) \subset L^2(\Omega)$. Consider the heat, cylinder and modified cylinder kernels associated to \mathcal{A} ; recall that the features possessed by these kernels have been discussed extensively in subsection 2.7. The results obtained therein allow, in particular, to derive the following lemma.

102

Lemma 2.90. Let $j \in \mathbb{N}$ and let α, β be any pair of multi-indices such that $|\alpha| + |\beta| \leq j$. *j.* For any fixed $\mathbf{x}, \mathbf{y} \in \Omega$, consider the maps $(0, +\infty) \to \mathbb{C}$, $\mathbf{t} \mapsto \partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t} \cdot \mathcal{A}}(\mathbf{x}, \mathbf{y})$, $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t} \sqrt{\mathcal{A}}}(\mathbf{x}, \mathbf{y}), \ \partial_1^{\alpha} \partial_2^{\beta}(e^{-\mathbf{t} \sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(\mathbf{x}, \mathbf{y})$; each one of these maps admits a Mellin transform. More precisely, there holds

$$\partial_{1}^{\alpha}\partial_{2}^{\beta}e^{-\bullet \mathcal{A}}(\boldsymbol{x},\boldsymbol{y}) \in \mathcal{M}_{(j+d)/2} ,$$

$$\partial_{1}^{\alpha}\partial_{2}^{\beta}e^{-\bullet \sqrt{\mathcal{A}}}(\boldsymbol{x},\boldsymbol{y}) \in \mathcal{M}_{j+d} ,$$

$$\partial_{1}^{\alpha}\partial_{2}^{\beta}(e^{-\bullet \sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(\boldsymbol{x},\boldsymbol{y}) \in \mathcal{M}_{j+d-1} .$$
(2.350)

Moreover, for all $s \in \Sigma_{(j+d)/2}$, the derivative $\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y})$ of the Dirichlet kernel can be expressed as follows in terms of the Mellin transforms of the above maps:

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) = \frac{1}{\Gamma(s)} \mathfrak{M}[\partial_1^{\alpha} \partial_2^{\beta} e^{-\bullet \mathcal{A}}(\boldsymbol{x}, \boldsymbol{y})](s) , \qquad (2.351)$$

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) = \frac{1}{\Gamma(2s)} \mathfrak{M}[\partial_1^{\alpha} \partial_2^{\beta} e^{-\bullet \sqrt{\mathcal{A}}}(\boldsymbol{x}, \boldsymbol{y})](2s) , \qquad (2.352)$$

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) = \frac{1}{\Gamma(2s-1)} \mathfrak{M}[\partial_1^{\alpha} \partial_2^{\beta}(e^{-\bullet\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})(\boldsymbol{x}, \boldsymbol{y})](2s-1) .$$
(2.353)

Remark 2.31. The identities in Eq.s (2.351-2.353) must be meant to hold pointwisely, for all fixed $\mathbf{x}, \mathbf{y} \in \Omega$ (and for all $s \in \Sigma_{(j+d)/2}$).

Proof. Recall the small and large t bounds derived in point ii) of Proposition 2.65 for the exponential kernels; see, in particular, Eq.s (2.249), (2.254) and (2.255). Then, the first part of the thesis (namely, Eq. (2.350)) follows easily from Lemma 2.83.

On the other hand, the identities in Eq.s (2.351-2.353) are simply a restatement of Eq.s (2.266-2.268) (see Proposition 2.70) in the language of Mellin transforms.

Let us anticipate that, for the physical applications to be considered in the subsequent Chapters 3 and 4, it is of utmost interest the evaluation along the diagonal $(\mathbf{y} = \mathbf{x})$ of the Dirichlet kernel and of its derivatives, as well as the computation of their analytic continuations with respect to the complex parameter defining their order.

Having in mind these developments, hereafter we proceed to construct the required analytic continuations, making suitable hypotheses for the small \mathbf{t} asymptotic behaviour of the corresponding derivatives of the heat and cylinder kernels (to be evaluated along the diagonal, as well). The main results are contained in the forthcoming Theorems 2.91, 2.92 and 2.93; in the related Remarks 2.32, 2.33 and 2.34 we will comment on the hypotheses mentioned above for the exponential kernels, pointing out that they are typically fulfilled in most cases of interest.

Before proceeding, let us recall once more that, for any $\mathbf{x} \in \Omega$ and for any pair of multiindexes α, β , the maps $(0, +\infty) \ni \mathbf{t} \mapsto \partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t} \cdot \mathcal{A}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}, \partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t} \cdot \sqrt{\mathcal{A}}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ are analytic and decay exponentially for $\mathbf{t} \to +\infty$ (see Proposition 2.65). On the other hand, all the hypotheses made in Theorems 2.91, 2.92 and 2.93 contain only assumptions about the asymptotics of the above maps in the limit $\mathbf{t} \to 0^+$. **Theorem 2.91.** Let α, β be any pair of multi-indexes and consider the derivative $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}A}(\mathbf{x}, \mathbf{y})$ of the heat kernel, for $\mathbf{t} \in (0, +\infty)$ and $\mathbf{x}, \mathbf{y} \in \Omega$. Assume there exist $N \in \mathbb{N}$ and a family of coefficients $b_n : \Omega \to \mathbb{R}$ ($n \in \{0, ..., N\}$) such that along the diagonal $\mathbf{y} = \mathbf{x}$, there holds the following asymptotic expansion for $\mathbf{t} \to 0^+$:

$$\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} = \frac{1}{\mathbf{t}^{(|\alpha|+|\beta|+d)/2}} \left(\sum_{n=1}^N b_n(\boldsymbol{x}) \, \mathbf{t}^n + O(\mathbf{t}^{N+1}) \right).$$
(2.354)

Then, the map $\Sigma_{(|\alpha|+|\beta|+d)/2} \to \mathbb{C}$, $s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{x})$ (for all $\boldsymbol{x} \in \Omega$) can be analytically continued to a function which is meromorphic in the strip $\Sigma_{(|\alpha|+|\beta|+d)/2-N}$ and possesses only possible simple pole singularities at $s = \frac{|\alpha|+|\beta|+d}{2} - n$, for $n \in \{0, ..., N\}$. More precisely, the analytic continuation is given by

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} = \frac{1}{\Gamma(s)} \left[\sum_{n=0}^N \frac{b_n(\boldsymbol{x}) \, T^{s-d/2+n}}{s - \frac{|\alpha| + |\beta| + d}{2} + n} \right] + (2.355)$$

$$+ \int_{0}^{T} d\mathbf{t} \, \mathbf{t}^{s-1} \left(\partial_{1}^{\alpha} \partial_{2}^{\beta} e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} - \frac{1}{\mathbf{t}^{(|\alpha|+|\beta|+d)/2}} \sum_{n=0}^{N} b_{n}(\boldsymbol{x}) \, \mathbf{t}^{n} \right) + \int_{T}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, \partial_{1}^{\alpha} \partial_{2}^{\beta} e^{-\mathbf{t}\mathcal{A}}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} \right]$$

Proof. First of all, recall that the results of Proposition 2.65 allow to infer the following estimate, for any fixed $\mathbf{x} \in \Omega$:

$$\left|\partial_1^{\alpha}\partial_2^{\beta}e^{-\mathbf{t}\mathcal{A}}(\mathbf{x},\mathbf{y})\right|_{\mathbf{y}=\mathbf{x}}\right| \leq C_{\alpha,\beta}(\mathbf{x}) \ e^{-\varepsilon \mathbf{t}} \qquad \text{for } \mathbf{t} > T_{\alpha,\beta}(\mathbf{x})$$

where the positive constants $C_{\alpha,\beta}(\mathbf{x}), T_{\alpha,\beta}(\mathbf{x})$ can be determined explicitly according to Eq. (2.249) and $\varepsilon > 0$ is such that $\sigma(\mathcal{A}) \subset [\varepsilon, +\infty)$. Then, due to Proposition 2.85, the map $s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ can be analytically continued to the (right-infinite) strip $\Sigma_{((|\alpha|+|\beta|+d)/2-N,+\infty)} \equiv \Sigma_{(|\alpha|+|\beta|+d)/2-N}$. Moreover, the analytic continuation can be determined explicitly starting with Eq. (2.351); it suffices to keep in mind the asymptotic expansion (2.354) and to resort to the general relation (2.323), here employed with P = 0and $\mathfrak{a}_n = n - (|\alpha| + |\beta| + d)/2, (n = 0, ..., N)$. This suffices to infer the thesis. \Box

Remark 2.32. i) As anticipated previously, the hypotheses made in the above theorem are well-known to be fulfilled in many cases of interest. For example, expansions of the form (2.354) for the diagonal heat kernel derivatives can be easily derived for arbitrary $N \in \mathbb{N}$ under the assumptions (2.159) for the domain Ω and for the potential V, starting from the expression in Eq. (2.262); in this case the coefficients b_n in Eq. (2.354) can all be determined in terms of the HDSM-coefficients a_n and of their derivatives. Analgous results for the heat kernel also hold in case the domain is a (possibly unbounded) subset \mathbb{R}^d delimited by flat boundaries (namely, parallel or perpendicular planes), on which suitable boundary conditions are prescribed (³⁷). Let us also mention that, when the domain Ω

³⁷This statement can be checked by direct computations, starting with the heat kernel $K_0(\mathbf{t}; \mathbf{x}, \mathbf{y}) := \frac{1}{(4\pi \mathbf{t})^{d/2}} e^{-\frac{|\mathbf{x}-\mathbf{y}|^2}{4\mathbf{t}}}$ (see Eq. (2.263)) and using the method of images.

is the whole space \mathbb{R}^d and the potential V is either a mass term ($V = m^2$, for some $m \in \mathbb{R}$) or an harmonic-type background ($V(\mathbf{x}) = \lambda^4 |\mathbf{x}|^2$, for some $\lambda \in \mathbb{R}$), the well-known explicit expressions for the corresponding heat kernels are easily seen to possess asymptotic expansions of the type (2.354).

ii) Whenever the hypotheses of Theorem 2.91 are fulfilled, it can be proved by Riesz means methods (see, e.g., [60, 73, 76]) that the derivative $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\sqrt{\mathcal{A}}}(\mathbf{x}, \mathbf{y})$ of the cylinder kernel evaluated along the diagonal admits a corresponding asymptotic expansion, for $\mathbf{t} \to 0^+$, given by (compare with (2.265))

$$\partial_{1}^{\alpha}\partial_{2}^{\beta}e^{-\mathbf{t}\sqrt{\mathcal{A}}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = \frac{1}{\mathbf{t}^{d+|\alpha|+|\beta|}} \left(\sum_{n=0}^{N} g_{n}(\mathbf{x})\mathbf{t}^{n} + \sum_{\substack{n=d+|\alpha|+|\beta|+1\\n-(d+|\alpha|+|\beta|) \text{ odd}}}^{N} h_{n}(\mathbf{x})\mathbf{t}^{n}\ln\mathbf{t} + O(\mathbf{t}^{N+1}\ln\mathbf{t})\right), \qquad (2.356)$$

for some $N \in \mathbb{N}$ and for two families of coefficients $g_n, h_n : \Omega \to \mathbb{R}$ $(n \in \{0, ..., N\})$ (³⁸). Then, a simple variation of the above Theorem 2.91 can be derived starting from the asymptotic expansion (2.356); since this alternative formulation will never be used in the applications to be considered in the following, it will not be reported in the present work.

In most of the cases mentioned in item i) of Remark 2.32 much more can be said about the small **t** behaviour of either the heat or cylinder kernel and of their derivatives. Having in mind these considerations, we state the following Theorems 2.92, 2.93; therein, under slightly stronger assumptions, we present other methods which allow to construct the analytic continuations of the maps $s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$, typically in a more efficient way from a computational point of view.

Theorem 2.92. Let α, β be any pair of multi-indexes. Assume there exist $N \in \mathbb{N}$ and a function $H^{(\alpha,\beta)} : [0, +\infty) \times \Omega \to \mathbb{R}$, $(\mathbf{t}, \mathbf{x}) \mapsto H^{(\alpha,\beta)}(\mathbf{t}; \mathbf{x})$ such that, for any fixed $\mathbf{x} \in \Omega$: i) the map $\mathbf{t} \mapsto H^{(\alpha,\beta)}(\mathbf{t}; \mathbf{x})$ is of class C^N ;

ii) for $\mathbf{t} \in (0, +\infty)$, the derivative $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t} \mathcal{A}}(\mathbf{x}, \mathbf{y})$ of the heat kernel evaluated along the diagonal is given by

$$\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{y}=\mathbf{x}} = \frac{1}{\mathbf{t}^{(|\alpha|+|\beta|+d)/2}} H^{(\alpha,\beta)}(\mathbf{t}; \mathbf{x}) .$$
(2.357)

Then, for any fixed $\mathbf{x} \in \Omega$, the map $\sum_{(d+|\alpha|+|\beta|)/2} \to \mathbb{C}$, $s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ can be analytically continued to a function which is meromorphic in the strip $\sum_{(d+|\alpha|+|\beta|)/2-N}$ and possesses only possible simple pole singularities at $s = \frac{|\alpha|+|\beta|+d}{2} - n$, for $n \in \{0, ..., N\}$. More precisely, the analytic continuation is given by

³⁸As pointed out by Fulling in the already cited works [60, 73, 76], some but not all of the coefficients g_n, h_n can be determined in terms of the heat kernel coefficients b_n .

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} =$$

$$\frac{(-1)^N}{\Gamma(s)(s - \frac{|\alpha| + |\beta| + d}{2}) \dots (s - \frac{|\alpha| + |\beta| + d}{2} - 1 + N)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s - \frac{|\alpha| + |\beta| + d}{2} - 1 + N} \, \partial_{\mathbf{t}}^N H^{(\alpha, \beta)}(\mathbf{t} \, ; \boldsymbol{x}) \, .$$

$$(2.358)$$

Proof. The thesis can be derived following the same arguments presented in the proof of Theorem 2.91, using Proposition 2.88 (and the corresponding Eq. (2.336)) in place of Proposition 2.85.

Remark 2.33. The hypotheses made in the above theorem are known to hold when Ω is bounded with smooth boundary and V = 0 (³⁹), or when $\Omega = \mathbb{R}^d$ and the potential is either $V = m^2$ ($m \in \mathbb{R}$) or $V(\mathbf{x}) = \lambda^4 |\mathbf{x}|^2$ ($\lambda \in \mathbb{R}$).

Theorem 2.93. Let α, β be any pair of multi-indexes. Assume there to exist a function $J^{(\alpha,\beta)}: (0,+\infty) \times \Omega \to \mathbb{R}, (\mathbf{t}, \mathbf{x}) \mapsto J^{(\alpha,\beta)}(\mathbf{t}; \mathbf{x})$ such that, for any fixed $\mathbf{x} \in \Omega$:

i) the map $\mathbf{t} \mapsto J^{(\alpha,\beta)}(\mathbf{t}; \mathbf{x})$ admits an analytic extension to a complex open neighbour $\mathcal{U} \subset \mathbb{C}$ of the positive real semi-axis $[0, +\infty)$;

ii) for $\mathbf{t} \in (0, +\infty)$, the derivative $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\sqrt{A}}(\mathbf{x}, \mathbf{y})$ of the cylinder kernel evaluated along the diagonal is given by

$$\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\sqrt{\mathcal{A}}} \left(\boldsymbol{x}, \boldsymbol{y} \right) \Big|_{\boldsymbol{y}=\boldsymbol{x}} = \frac{1}{\mathbf{t}^{|\alpha|+|\beta|+d}} J^{(\alpha,\beta)}(\mathbf{t}; \boldsymbol{x}) .$$
(2.359)

Then, the map $\Sigma_{(|\alpha|+|\beta|+d)/2} \to \mathbb{C}$, $s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y})|_{\boldsymbol{y}=\boldsymbol{x}}$ can be analytically continued to a function, which is meromorphic on the whole complex plane and possesses only possible simple pole singularities at s = k/2, for $k \in \{1, ..., |\alpha| + |\beta| + d\}$. More precisely, the analytic continuation is given by

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} = \frac{e^{-2i\pi s} \Gamma(1-2s)}{2\pi i} \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{2s-|\alpha|-|\beta|-d-1} J^{(\alpha,\beta)}(\mathbf{t} \, ; \boldsymbol{x}) \; ; \qquad (2.360)$$

in particular, for s = -k/2, $k \in \mathbb{N}$, there holds

$$\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{k/2}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} = (-1)^k \, \Gamma(k+1) \, \operatorname{Res} \left(\mathbf{t}^{-(k+|\alpha|+|\beta|+d+1)} J^{(\alpha,\beta)}(\mathbf{t}\,;\boldsymbol{x})\,;\,\, \mathbf{t}=0 \right) \,. \tag{2.361}$$

Proof. Also in this case the thesis can be derived following the same arguments presented in the proof of Theorem 2.91; this time, in place of Proposition 2.85, one must resort to Proposition 2.89 and to the related Remark 2.30 (see, in particular, item i) therein which must be employed here with $q = |\alpha| + |\beta| + d$ and $F(\mathbf{t}) = J^{(\alpha,\beta)}(\mathbf{t};\mathbf{x})$. Finally, Eq. (2.361) follows easily by the residue theorem, recalling that the map $\mathbf{t} \mapsto J^{(\alpha,\beta)}(\mathbf{t};\mathbf{x})$ is analytic for all $\Re \mathbf{t} > 0$.

³⁹In fact, in this case all the HDMS coefficients of any order vanish except for $a_0(\mathbf{x}, \mathbf{x}) = 1$.

Remark 2.34. i) The hypotheses made in Theorem 2.93 can be easily checked by direct computation in many cases of interest. For example, they are fulfilled when the potential is null (V = 0) and the domain is a subset of \mathbb{R}^d delimited by flat boundaries consisting, namely, of parallel and/or perpendicular planes on which either Dirichlet or Neumann conditions are prescribed (see [65] and the forthcoming Section 4.3 of Chapter 4). ii) Consider the analytic continuation of $\partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ determined according to Eq. (2.360). It appears that the singular behaviour at the points $s = k/2, k \in \{1, ..., |\alpha| +$ $|\beta| + d$ descendes from the pole singularity of the Gamma function evaluated at nonpositive integers. Due to the same considerations, one could expect any of the points s = k/2, with $k \in \mathbb{N}$ unrestricted, to be singular; however, it is known a priory that the map $s \mapsto \partial_1^{\alpha} \partial_2^{\beta} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ is analytic for $s \in \Sigma_{(|\alpha|+|\beta|+d)/2}$, so that no singularity exists therein (nor can it arise by analytic continuation). Let us stress that there is no contradiction; in fact, for s = k/2 with $k \in \mathbb{N}$ and $k > |\alpha| + |\beta| + d$, the integral along the Hankel contour in Eq. (2.360) can be easily seen to vanish via the residue theorem so that there appears an indeterminate form of the type $\infty \cdot 0$, which has to be evaluated by alternative methods.

Before moving on, let us point out that the results derived in the above Theorems 2.91, 2.92 and 2.93 continue to hold also for boundary points $\mathbf{x} \in \partial \Omega$ whenever stronger regularity assumptions (such as those in Eq. (2.159)) are fulfilled by the domain Ω and by the potential V, and suitable boundary conditions are prescribed on $\partial \Omega$.

Analytic continuation of the trace $\operatorname{Tr} \mathcal{A}^{-s}$.

Assume the stricter regularity hypotheses (2.159) and (2.209) to be fulfilled (so that Ω is bounded with boundary of class C^{∞} , $A = -\Delta + V$ with $V \in C^{\infty}(\overline{\Omega})$, and Dirichlet boundary conditions are prescribed on $\partial\Omega$).

Let us recall that, under the above assumptions, the heat and cylinder traces $\operatorname{Tr} e^{-\mathbf{t}\mathcal{A}}$, $\operatorname{Tr} e^{-\mathbf{t}\sqrt{\mathcal{A}}}$ exist and are analytic for $\mathbf{t} \in \Sigma_0$; moreover, these traces decay exponentially for $\Re \mathbf{t} \to +\infty$ (see Proposition 2.77). These facts can be employed, along with suitable hypotheses about the small \mathbf{t} behaviour of the mentioned traces, to construct the analytic continuation of the complex trace $\operatorname{Tr} \mathcal{A}^{-s}$ (recall that this is granted to exists and to be analytic for $s \in \Sigma_{d/2}$; see Corollary 2.75).

The forthcoming Theorems 2.94, 2.95 and 2.96 essentially contain global analogues of Theorems 2.91, 2.92 and 2.93, discussed in the previous subsection.

Theorem 2.94. Consider the heat trace $Tr e^{-tA}$, for $\mathbf{t} \in (0, +\infty)$. Assume there exist $N \in \mathbb{N}$ and a family of coefficients $b_n \in \mathbb{R}$ $(n \in \{0, ..., N\})$ such that, for $\mathbf{t} \to 0^+$, this trace possesses an asymptotic expansion of the form

$$Tr \, e^{-\mathbf{t}\mathcal{A}} = \frac{1}{\mathbf{t}^{d/2}} \left(\sum_{n=1}^{N} b_n \, \mathbf{t}^{n/2} + O(\mathbf{t}^{(N+1)/2}) \right).$$
(2.362)

Then $Tr e^{-\bullet A} \in \mathcal{M}_{d/2}$, and there holds

$$Tr\mathcal{A}^{-s} = \frac{1}{\Gamma(s)} \mathfrak{M}[Tr e^{-\bullet \mathcal{A}}](s) . \qquad (2.363)$$

Moreover, the map $\Sigma_{d/2} \to \mathbb{C}$, $s \mapsto Tr \mathcal{A}^{-s}$ can be analytically continued to a function which is meromorphic on the strip $\Sigma_{(d-N)/2}$ and possesses only possible simple pole singularities at s = (d-n)/2, for $n \in \{0, ..., N\}$; more precisely, the analytic continuation is given by

$$Tr\mathcal{A}^{-s} = \tag{2.364}$$

$$\frac{1}{\Gamma(s)} \left[\sum_{n=0}^{N} \frac{b_n T^{s-d/2+n}}{s - \frac{d}{2} + \frac{n}{2}} + \int_0^T d\mathbf{t} \, \mathbf{t}^{s-1} \left(Tr \, e^{-\mathbf{t}\mathcal{A}} - \frac{1}{\mathbf{t}^{d/2}} \sum_{n=0}^N b_n \, \mathbf{t}^{n/2} \right) + \int_T^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, Tr \, e^{-\mathbf{t}\mathcal{A}} \right] \, .$$

Proof. Recall again that the heat trace $\operatorname{Tr} e^{-t\mathcal{A}}$ is analytic for $\mathbf{t} \in \Sigma_0$ and vanishes exponentially for $\Re \mathbf{t} \to +\infty$ (see Proposition 2.77); these facts and the asymptotic expansion (2.362) suffice to infer that $\operatorname{Tr} e^{-\bullet\mathcal{A}} \in \mathcal{M}_{d/2}$, due to Lemma 2.83. Eq. (2.363) is just a restatement of the integral relation (2.297).

Also in this case, the explicit expression (2.364) for the analytic continuation of Tr \mathcal{A}^{-s} can be derived resorting to the general relation (2.323), here employed with P = 0 and $\mathfrak{a}_n = (n - |\alpha| + |\beta| + d)/2$, (n = 0, ..., N).

Theorem 2.95. Consider the heat trace $Tr e^{-tA}$, for $t \in (0, +\infty)$. Assume there exist $N \in \mathbb{N}$ and a function $H \in C^N([0, +\infty); \mathbb{R})$ such that

$$Tr e^{-\mathbf{t}\mathcal{A}} = \frac{1}{\mathbf{t}^{d/2}} H(\mathbf{t}) . \qquad (2.365)$$

Then $Tr e^{-\bullet A} \in \mathcal{M}_{d/2}$, and there holds Eq. (2.363), i.e.,

$$Tr\mathcal{A}^{-s} = \frac{1}{\Gamma(s)} \mathfrak{M}[Tr e^{-\bullet \mathcal{A}}](s) .$$

Moreover, the map $\Sigma_{d/2} \to \mathbb{C}$, $s \mapsto Tr \mathcal{A}^{-s}$ can be analytically continued to a function which is meromorphic on the strip $\Sigma_{(d-N)/2}$ and possesses only possible simple pole singularities at s = (d-n)/2, for $n \in \{0, ..., N\}$; more precisely, the analytic continuation is given by

$$Tr\mathcal{A}^{-s} = \frac{(-1)^N}{\Gamma(s)(s - \frac{d}{2})\dots(s - \frac{d-N}{2} - 1)} \int_0^{+\infty} d\mathbf{t} \, \mathbf{t}^{s - \frac{d-N}{2} - 1} \, \partial_{\mathbf{t}}^N H(\mathbf{t}) \,. \tag{2.366}$$

Proof. The thesis can be derived following the same arguments presented in the proof of Theorem 2.94, using Proposition 2.88 (and the corresponding Eq. (2.336)) in place of Proposition 2.85.

Theorem 2.96. Consider the cylinder trace $Tr e^{-\mathbf{t}\sqrt{A}}$, for $\mathbf{t} \in (0, +\infty)$. Assume there exists a function $J : (0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto J(\mathbf{t})$ admitting an analytic extension to a complex neighbour $\mathcal{U} \subset \mathbb{C}$ of the positive real semi-axis $[0, +\infty)$, such that

$$Tr e^{-\mathbf{t}\sqrt{\mathcal{A}}} = \frac{1}{\mathbf{t}^d} J(\mathbf{t}) . \qquad (2.367)$$

Then $Tr e^{-\bullet\sqrt{A}} \in \mathcal{M}_d$, and there holds Eq. (2.363), i.e.,

$$Tr \mathcal{A}^{-s} = \frac{1}{\Gamma(2s)} \mathfrak{M}[Tr e^{-\bullet\sqrt{\mathcal{A}}}](2s)$$

Moreover, the map $\Sigma_d \to \mathbb{C}$, $s \mapsto Tr \mathcal{A}^{-s}$ can be analytically continued to a function, which is meromorphic on the whole complex plane and possesses only possible simple pole singularities at s = k/2, for $k \in \{1, ..., d\}$. The analytic continuation is given by

$$Tr \mathcal{A}^{-s} = \frac{e^{-2i\pi s} \Gamma(1-2s)}{2\pi i} \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{2s-d-1} J(\mathbf{t}) \; ; \qquad (2.368)$$

in particular, for s = -k/2, $k \in \mathbb{N}$, there holds

$$Tr \mathcal{A}^{k/2} = (-1)^k \Gamma(k+1) \operatorname{Res}\left(\mathbf{t}^{-(k+d+1)} J(\mathbf{t}) \; ; \; \mathbf{t} = 0\right) \; . \tag{2.369}$$

Proof. Also in this case the thesis can be derived following the same arguments presented in the proof of Theorem 2.94; this time, in place of Proposition 2.85, one must resort to Proposition 2.89 and to the related Remark 2.30 (see, in particular, item i) therein which must be employed here with q = d and $F(\mathbf{t}) = J(\mathbf{t})$. Finally, Eq. (2.361) follows easily by the residue theorem, recalling that J is analytic.

Remark 2.35. Of course, the results derived in the above theorems continue to hold whenever the heat and cylinder traces are well-posed, decay exponentially at infinity and possess small **t** asymptotic behaviours such as those described in Eq.s (2.362), (2.365) and (2.367). In this sense, the assumptions (2.159) are sufficient but not necessary; for example, it can be checked by explicit computations that the heat trace is well-defined, analytic and exponentially vanishing at infinity if $\Omega = \mathbb{R}^d$ and $\mathcal{A} = -\Delta + V$, with $V(\mathbf{x}) = \lambda^4 |\mathbf{x}|^2$ ($\lambda \in \mathbb{R}$) (see Section 4.1).

Chapter 3

Quantum field theory on spatial domains with boundaries

In the present chapter we review the theory of canonical quantization for a Hermitian scalar field living on a suitable spatial domain Ω (which is assigned once and for all and has arbitrary dimension); possible conditions prescribed on the boundary of the domain and the interaction with a classical background potential are also taken into account. The implementation of local zeta regularization (ZR) within this framework is discussed in detail. The physical setting is the one which was described in Chapter 1 of the present manuscript, of which we retain the same notations and conventions; however, the language adopted here is more rigorous and precise from a mathematical point of view.

In Section 3.1 we review some well-known Fock space techniques, refering in particular to the Segal approach to quantization (see, e.g., [43, 58, 129, 130]). We give an abstract formulation of these topics, using the framework developed in Chapter 2 in terms of scales of Hilbert spaces associated to the real powers of a given strictly positive, self-adjoint operator (acting on the single particle space). In Section 3.2, this abstract framework is specialized to the case where the single particle Hilbert space consists of square-integrable functions on a fixed spatial domain $\Omega \subset \mathbb{R}^d$ and the positive, self-adjoint operator is a Schrödinger-type differential operator of the form $\mathcal{A} = -\Delta + V$, with V a smooth potential on Ω . Next, we use the complex powers \mathcal{A}^{-s} ($s \in \mathbb{C}$) to define a zeta-regularized Wightman field; this is employed, in turn, to construct a zeta-regularized version of the propagator and of some related observables (in particular, of the stress-energy VEV), thus making connection with the theory of the Casimir effect. In the end, we describe the zeta approach to renormalization in the most general formulation proposed in [64]; it is shown that this approach is granted to give finite values for the renormalized observables in many cases of interest.

Before proceeding, let us recall that we consider only a purely classical description of the above mentioned spatial domain Ω , of its boundary $\partial\Omega$ and of the background potential V; in particular, no back-reaction effect produced by the interaction with the quantum field is ever taken into account.

3.1 Canonical quantization of an abstract Hermitian scalar field.

In this section we review the theory of canonical quantization for an Hermitian scalar field. We develop an abstract formulation of this subject, making only minimal assumptions on the objects to be employed in the subsequent constructions. To this purpose, and in order to fix the notations and some conventions, we first recall some basic definitions and some well-known results, rephrasing them in the language developed in Chapter 2 of this work. For further details on the topics discussed hereafter (and, in particular, for the proofs of some statements), we refer mainly to the books of Reed and Simon [129] and of Moretti [119].

The bosonic Fock space.

Let $(\mathcal{H}, \langle | \rangle)$ be an abstract, separable Hilbert space and let $n \in \{1, 2, 3, ...\}$ be any given positive integer.

Consider the *n*-th tensor power $\mathcal{H}^{\otimes n} = \mathcal{H} \otimes ... \otimes \mathcal{H}$ (*n* times); this is the Hilbert space obtained by completing the algebraic tensor product $\otimes_{alg}^{n} \mathcal{H}$ of the underlying vector spaces with respect to the Hermitian inner product $(|)_{n} : \otimes_{alg}^{n} \mathcal{H} \times \otimes_{alg}^{n} \mathcal{H} \to \mathbb{C}$, defined as the unique sesquilinear extension (¹) of the map which, for any pair of factorized elements $f^{(n)} = f_1 \otimes ... \otimes f_n, g^{(n)} = g_1 \otimes ... \otimes g_n \in \otimes_{alg}^{n} \mathcal{H}$ ($f_i, g_i \in \mathcal{H}$ for all i = 1, ..., n), gives

$$(f^{(n)}|g^{(n)})_n = \prod_{i=1}^n \langle f_i|g_i \rangle .$$
 (3.1)

In order to avoid confusion, we will indicate with o the null element of \mathcal{H} and with $o^{(n)}$ the null vector of $\mathcal{H}^{\otimes n}$.

Next, consider the group of permutations of n elements, which we indicate with P_n . For any $\pi \in P_n$, let $U_{\pi} \in \mathfrak{B}(\mathcal{H}^{\otimes n})$ denote the unitary operator which, for any $f_1, \ldots, f_n \in \mathcal{H}$, gives

$$U_{\pi}(f_1 \otimes \ldots \otimes f_n) = f_{\pi^{-1}(1)} \otimes \ldots \otimes f_{\pi^{-1}(n)} ; \qquad (3.2)$$

it can be easily checked that the map $U : \mathbb{P}_n \to \mathfrak{B}(\mathcal{H}^{\otimes n}), \pi \mapsto U_{\pi}$ yields a faithful unitary representation of \mathbb{P}_n (see, e.g., Proposition 13.41 on page 661 of [119]).

The *n*-th totally symmetric tensor power $\mathcal{H}^{\vee n}$ is, by definition, the closed linear subspace of $\mathcal{H}^{\otimes n}$ formed by the elements which are U_{π} -invariant for all $\pi \in P_n$. Of course, $\mathcal{H}^{\vee n}$ is itself a Hilbert space with the inner product $(|)_n$ inherited from $\mathcal{H}^{\otimes n}$ and its null element is $o^{(n)}$.

¹This extension is defined by antilinearity in the left argument and by linearity in the right one, respectively; for the proof of its uniqueness, see, e.g., [129], page 49, Proposition 1 or [119], page 451, Proposition 10.23.

For any $f_1, \ldots, f_n \in \mathcal{H}$, let us define the symmetrized tensor product

$$f_1 \vee \ldots \vee f_n := \frac{1}{n!} \sum_{\pi \in \mathbf{P}_n} f_{\pi(1)} \otimes \ldots \otimes f_{\pi(n)} ; \qquad (3.3)$$

as well-known, $\mathcal{H}^{\vee n}$ coincides with the closed subspace of $\mathcal{H}^{\otimes n}$ generated by products of the form (3.3) (which explains the notation $^{\vee n}$).

Let us point out that, following the usual convention, we set $\mathcal{H}^{\otimes 0} := \mathbb{C}$ (with null vector $o^{(0)} := 0 \in \mathbb{C}$); consequently, we also have $\mathcal{H}^{\vee 0} = \mathbb{C}$.

Remark 3.1. The factor 1/n! on the right-hand side of Eq. (3.3) is just an arbitrary normalization choice; the subsequent items i) and ii) point out some facts descending from this choice.

i) Let $f \in \mathcal{H}$ and consider the totally symmetric element $f \vee \ldots \vee f \in \mathcal{H}^{\vee n}$. It follows straightforwardly from the definition (3.3) (here employed with $f_i = f$ for all i = 1, ..., n) that

$$f \vee \ldots \vee f = f \otimes \ldots \otimes f ; \qquad (3.4)$$

in particular, one has $o \lor \ldots \lor o = o^{(n)}$.

ii) Let $f^{(n)} = f_1 \vee \ldots \vee f_n$, $g^{(n)} = g_1 \vee \ldots \vee g_n \in \mathcal{H}^{\vee n}$ $(f_i, g_i \in \mathcal{H} \text{ for all } i = 1, ..., n)$ be any pair of totally symetric elements. Then, using Eq.s (3.1) (3.3), it can be easily inferred that the inner product $(f^{(n)}|g^{(n)})_n$ can be expressed as follows:

$$(f^{(n)}|g^{(n)})_n = \frac{1}{n!} \sum_{\pi \in \mathbf{P}_n} \prod_{i=1}^n \langle f_i | g_{\pi(i)} \rangle = \frac{1}{n!} \sum_{\pi \in \mathbf{P}_n} \prod_{i=1}^n \langle f_{\pi(i)} | g_i \rangle .$$
(3.5)

In particular, if $f^{(n)} = f \lor \ldots \lor f$ for some $f \in \mathcal{H}$, one has

$$(f^{(n)}|f^{(n)})_n = \langle f|f\rangle^n .$$
(3.6)

Now, consider the infinite collection of all totally symmetric Hilbert tensor powers of \mathcal{H} :

$$\mathcal{H}^{\vee n}$$
 $(n \in \{0, 1, 2, ...\})$. (3.7)

Definition 3.1. The symmetric Fock space over \mathcal{H} is

$$\mathfrak{F}^{\vee}(\mathcal{H}) := \bigoplus_{n=0}^{+\infty} \mathcal{H}^{\vee n} , \qquad (3.8)$$

where \bigoplus indicates the Hilbert direct sum. The corresponding inner product is denoted with (\mid).

Remark 3.2. According to the theory of Hilbert direct sums $(^2)$, $(\mathfrak{F}^{\vee}(\mathcal{H}), (|))$ is (up to isomorphisms) the unique Hilbert space possessing the forthcoming features i) and ii).

²For a precise definition of Hilbert tensor product and Hilbert direct sum in the case of infinitely many countable or uncountable spaces (to be intended properly in the sense of nets), see, e.g., [27, 119].

CHAPTER 3. QUANTUM FIELD THEORY ON SPATIAL DOMAINS WITH BOUNDARIES

i) For any $n \in \{0, 1, 2, ...\}$, $\mathcal{H}^{\vee n}$ is a closed subspace of $\mathfrak{F}^{\vee}(\mathcal{H})$ and the corresponding inner product $(|)_n$ coincides with the restriction of (|) to $\mathcal{H}^{\vee n}$.

ii) The summands $\mathcal{H}^{\vee n}$ $(n \in \{0, 1, 2, ...\})$ are mutually orthogonal with respect to (|) and the closed subspace they generate coincides with the whole Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$.

It should be noted that any element $f \in \mathfrak{F}^{\vee}(\mathcal{H})$ has a unique representation

$$\boldsymbol{f} = \sum_{n=0}^{+\infty} f^{(n)} \qquad (f^{(n)} \in \mathcal{H}^{\vee n}) ; \qquad (3.9)$$

for any $n \in \{0, 1, 2, ...\}$, $f^{(n)}$ is called the *n*-th *component* of \boldsymbol{f} . Let us point out that, for any \boldsymbol{f} as above and for any $\boldsymbol{g} = \sum_n g^{(n)} \in \mathfrak{F}^{\vee}(\mathcal{H})$, there holds

$$(\boldsymbol{f}|\boldsymbol{g}) = \sum_{n=0}^{+\infty} (f^{(n)}|g^{(n)})_n .$$
(3.10)

We put

$$\boldsymbol{o} := \text{the null vector of } \mathfrak{F}^{\vee}(\mathcal{H}) ; \qquad (3.11)$$

on the other hand, we will refer to the normalized vector

$$\boldsymbol{v} := 1 \in \mathbb{C} \equiv \mathcal{H}^{\vee 0} \subset \mathfrak{F}^{\vee}(\mathcal{H})$$
(3.12)

(³). Moreover, we indicate with $\hat{\mathbb{O}}_{\mathfrak{F}}$ and $\hat{\mathbb{I}}_{\mathfrak{F}}$ the null and the identity operators on $\mathfrak{F}^{\vee}(\mathcal{H})$, respectively.

In the sequel we will be often interested in the space

$$\mathfrak{D}^{\vee}(\mathcal{H}) := \{ \boldsymbol{f} = \sum_{n} f^{(n)} \in \mathfrak{F}^{\vee}(\mathcal{H}) \mid f^{(n)} = 0 \text{ for almost all } n \in \{0, 1, 2, ...\} \} ; \quad (3.13)$$

the above expression means that there exists $N_f \in \mathbb{N}$ (depending on f) such that $f^{(n)} = 0$ for all $n \geq N_f$. Of course, $\mathfrak{D}^{\vee}(\mathcal{H})$ is a dense linear subspace of the Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$; whenever we will need a topology on $\mathfrak{D}^{\vee}(\mathcal{H})$, we will use the (non-complete) topology induced by the restriction of the inner product (|).

$$f \equiv (f^{(n)}) = (f^{(0)}, f^{(1)}, f^{(2)}, \dots) ,$$

with $f^{(n)} \in \mathcal{H}^{\vee n}$ for each $n \in \{0, 1, 2, ...\}$, such that $\sum_{n} (f^{(n)}|f^{(n)})_n < +\infty$. This is a Hilbert space with the inner product (|), defined according to Eq. (3.10). An element $f^{(n)} \in \mathcal{H}^{\vee n}$ can be identified with the sequence $(o^{(0)}, ..., o^{(n-1)}, f^{(n)}, o^{(n+1)}, ...) \in \mathfrak{F}^{\vee}(\mathcal{H})$, where $o^{(n)}$ is the null element of $\mathcal{H}^{\vee n}$ $(n \in \{0, 1, 2, ...\})$. In this sense $\mathcal{H}^{\vee n}$ is a linear subspace of $\mathfrak{F}^{\vee}(\mathcal{H})$, for each $n \in \{0, 1, 2, ...\}$, and Eq. (3.9) holds for any \boldsymbol{f} as above; moreover, using the language of the present footnote, Eq.s (3.11) and (3.12) can be rephrased, respectively, as

$$\boldsymbol{o} := (o^{(0)}, o^{(1)}, o^{(2)}, \dots) , \qquad \boldsymbol{v} := (1, o^{(1)}, o^{(2)}, \dots) .$$

114

³ Let us mention that $\mathfrak{F}^{\vee}(\mathcal{H})$ could be defined as the set of sequences of the form

Needless to say, both \boldsymbol{o} and \boldsymbol{v} (see Eq.s (3.11) (3.12)) do, in fact, belong to $\mathfrak{D}^{\vee}(\mathcal{H})$. The space of all linear (unbounded) operators on $\mathfrak{D}^{\vee}(\mathcal{H})$ is $\mathcal{L}(\mathfrak{D}^{\vee}(\mathcal{H}))$. In particular, we indicate with $\hat{\mathbb{O}}_{\mathfrak{D}}$ and $\hat{\mathbb{I}}_{\mathfrak{D}}$ ($\in \mathcal{L}(\mathfrak{D}^{\vee}(\mathcal{H}))$) the restrictions to $\mathfrak{D}^{\vee}(\mathcal{H})$ of the null and the identity operators on $\mathfrak{F}^{\vee}(\mathcal{H})$ (i.e., of $\hat{\mathbb{O}}_{\mathfrak{F}}$ and $\hat{\mathbb{I}}_{\mathfrak{F}}$), respectively; so, for any $\boldsymbol{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$, there hold

$$\hat{\mathbb{O}}_{\mathfrak{D}}\boldsymbol{f} = \boldsymbol{o} , \qquad \hat{\mathbb{I}}_{\mathfrak{D}}\boldsymbol{f} = \boldsymbol{f} .$$
(3.14)

One can also consider the *particle number operator*; this is the essentially self-adjoint operator $\hat{\mathcal{N}} : \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{F}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ which is defined by components setting, for any $\boldsymbol{f} = \sum_{n} f^{(n)}$,

$$\hat{\mathcal{N}}\boldsymbol{f} := \sum_{n} n f^{(n)} . \tag{3.15}$$

In the following, we adopt systematically the standard terminology indicated hereafter.

Definition 3.2. For any $n \in \{1, 2, 3, ...\}$, $\mathcal{H}^{\vee n}$ is called the *n*-particle Hilbert space (\mathcal{H} is the single particle space) and its elements are the totally symmetric states of *n* particles. $\mathcal{H}^{\vee 0}$ is the Fock vacuum; this is generated by the normalized vector $\boldsymbol{v} \in \mathfrak{F}^{\vee}(\mathcal{H})$ (see Eq. (3.12)) which is called the vacuum state. $\mathfrak{D}^{\vee}(\mathcal{H})$ is the finite-particle subspace and $\mathfrak{F}^{\vee}(\mathcal{H})$ is the bosonic Fock space on \mathcal{H} .

Remark 3.3. In this work we will only consider a *scalar* field theory; therefore, since no confusion shall arise, the adjectives "totally symmetric" and "bosonic" will often be omitted and implicitly understood.

Finally, let us point out that in the physical applications that we are going to discuss in the following, we shall mainly be interested with expressions of the form described hereafter.

Definition 3.3. Let $\hat{\mathcal{O}}$: $\text{Dom}(\hat{\mathcal{O}}) \subset \mathfrak{F}^{\vee}(\mathcal{H}) \to \mathfrak{F}^{\vee}(\mathcal{H})$ be any operator with domain containing (at least) the vacuum state \boldsymbol{v} ($\boldsymbol{v} \in \text{Dom}(\hat{\mathcal{O}})$). The vacuum expectation value (in brief, VEV) of $\hat{\mathcal{O}}$ is the complex number ($\boldsymbol{v} | \hat{\mathcal{O}} \boldsymbol{v} \in \mathbb{C}$.

Creation and annihilation operators.

Let again $(\mathcal{H}, \langle | \rangle)$ be any abstract Hilbert space and consider the Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$, in the formulation given in the previous subsection; hereafter, using the language described therein, we introduce the creation and annihilation operators on $\mathfrak{F}^{\vee}(\mathcal{H})$ and analyze their main features.

Definition 3.4. For any given $h \in \mathcal{H}$, the (h-smeared) creation and annihilation operators are, respectively, the \mathbb{R} -linear maps

$$\hat{a}^+(h), \ \hat{a}^-(h) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$$

$$(3.16)$$

CHAPTER 3. QUANTUM FIELD THEORY ON SPATIAL DOMAINS WITH BOUNDARIES

defined "by components" as follows (⁴). For any $n \in \{1, 2, 3, ...\}$, $\hat{a}^+(h)$ and $\hat{a}^-(h)$ are the unique linear maps sending $\mathcal{H}^{\vee n}$ respectively into $\mathcal{H}^{\vee (n+1)}$ and $\mathcal{H}^{\vee (n-1)}$ which, for any element $f^{(n)} \in \mathcal{H}^{\vee n}$ of the form $f^{(n)} = f_1 \vee ... \vee f_n$ ($f_i \in \mathcal{H}$ for i = 1, ..., n), give

$$\hat{a}^{+}(h)(f_1 \vee \ldots \vee f_n) = \sqrt{n+1} \quad h \vee f_1 \vee \ldots \vee f_n , \qquad (3.17)$$

$$\hat{a}^{-}(h)(f_1 \vee \ldots \vee f_n) = \frac{1}{\sqrt{n}} \sum_{i=1}^n \langle h | f_i \rangle \ f_1 \vee \ldots \vee f_i \vee \ldots \vee f_n ; \qquad (3.18)$$

here the notation $f_1 \vee ... \vee f'_i \vee ... \vee f_n$ indicates the symmetrized tensor product of the n-1 elements obtained eliminating f_i from the collection $\{f_1, ..., f_n\}$. In particular, for n = 1 and $f^{(1)} \equiv f_1$, Eq. (3.18) must be meant to hold in the sense that

$$\hat{a}^{-}(h)f^{(1)} = \langle h|f_1 \rangle \ \boldsymbol{v} \ .$$
 (3.19)

Moreover, we put by convention

$$\hat{a}^{-}(h) \boldsymbol{v} := \boldsymbol{o} , \qquad \hat{a}^{+}(h) \boldsymbol{v} := h \in \mathcal{H} \equiv \mathcal{H}^{\vee 1} , \qquad (3.20)$$

with o and v indicating as usual the null vector and the vacuum state, respectively.

As well-known [130], for any $h \in \mathcal{H}$, both $\hat{a}^+(h)$ and $\hat{a}^-(h)$ are closable unbounded linear operators on $\mathfrak{F}^{\vee}(\mathcal{H})$, defined on the dense domain $\mathfrak{D}^{\vee}(\mathcal{H})$; unless otherwise stated, from here on the symbols $\hat{a}^+(h)$ and $\hat{a}^-(h)$ will be used to indicate the closures of the operators defined in Eq.s (3.17) and (3.18), respectively.

Let us also point out a fact made evident by the definitions (3.17) (3.18): namely, that the maps $h \mapsto \hat{a}^+(h)$ and $h \mapsto \hat{a}^-(h)$ are \mathbb{C} -linear and \mathbb{C} -antilinear, representively.

Proposition 3.5. For any $h \in \mathcal{H}$, the creation and annihilation operators $\hat{a}^+(h), \hat{a}^-(h)$ are the adjoints of each other; in particular, for all $\mathbf{f}, \mathbf{g} \in \mathfrak{D}^{\vee}(\mathcal{H})$, there holds

$$(\hat{a}^{+}(h) \mathbf{f} | \mathbf{g}) = (\mathbf{f} | \hat{a}^{-}(h) \mathbf{g}) .$$
 (3.21)

Proof. Due to the definition of the operators $\hat{a}^+(h), \hat{a}^-(h)$, it suffices to show that, for any $n \in \mathbb{N}$ the maps $\hat{a}^+(h) : \mathcal{H}^{\vee n} \to \mathcal{H}^{\vee(n+1)}$ and $\hat{a}^-(h) : \mathcal{H}^{\vee(n+1)} \to \mathcal{H}^{\vee n}$ are (Banach) adjoints of each other; since $\mathcal{H}^{\vee n}$ and $\mathcal{H}^{\vee(n+1)}$ are respectively spanned by elements of the form $f^{(n)} = f_1 \vee \ldots \vee f_n$ and $g^{(n+1)} = g_1 \vee \ldots \vee g_{n+1} \in \mathcal{H}^{\vee(n+1)}$, this amounts to prove that for any such pair $f^{(n)}, g^{(n+1)}$, there holds

$$(\hat{a}^{+}(h) f^{(n)} | g^{(n+1)})_{n+1} = (f^{(n)} | \hat{a}^{-}(h) g^{(n+1)})_{n}$$

where $(|)_n$ and $(|)_{n+1}$ indicate the inner products defined according to Eq. (3.1). On the other hand, the above identity can be easily checked keeping in mind Eq.s (3.17) (3.18) for $\hat{a}^+(h), \hat{a}^-(h)$ and recalling the identities in Eq. (3.5) for the inner product $(|)_n$ (and the corresponding ones for $(|)_{n+1}$).

⁴Recall the considerations below Eq. (3.3) on page 113. In particular, let us recall that the finite particle subspace $\mathfrak{D}^{\vee}(\mathcal{H})$ (as well as the Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$) is spanned by elements of the form $f_1 \vee \ldots \vee f_n$ $(f_i \in \mathcal{H}, \text{ for } i = 1, ..., n, n \in \{0, 1, 2, ...\})$.

To go on let us notice that, for any $n \in \{1, 2, 3, ...\}$ and for any family $h_i \in \mathcal{H}$ (i = 1, ..., n), the multiple product $\hat{a}^+(h_1) \dots \hat{a}^+(h_n)$ and the analogous expressions obtained replacing some of the \hat{a}^+ with \hat{a}^- are all well-defined operators on the finite particle subspace $\mathfrak{D}^{\vee}(\mathcal{H})$. In particular, there holds the subsequent proposition, where we introduce the notation

$$[X, Y] := X Y - Y X , \qquad (3.22)$$

to be respected thorughout the remainder of this manuscript.

Proposition 3.6. There hold the results enumerated hereafter. *i)* For any $n \in \{1, 2, 3, ...\}$, the set $\{\hat{a}^+(h_1) ... \hat{a}^+(h_n) v \mid h_i \in \mathcal{H}, i = 1, ..., n, n \in \mathbb{N}\}$ spans $\mathfrak{D}^{\vee}(\mathcal{H})$, (*i.e.*, it is total in $\mathfrak{F}^{\vee}(\mathcal{H})$).

ii) For any $h, k \in \mathcal{H}$, there hold the canonical commutation relations (CCR)

$$\begin{aligned} [\hat{a}^{+}(h), \hat{a}^{+}(k)] &= \hat{\mathbb{O}}_{\mathfrak{D}} ,\\ [\hat{a}^{-}(h), \hat{a}^{-}(k)] &= \hat{\mathbb{O}}_{\mathfrak{D}} ,\\ [\hat{a}^{-}(h), \hat{a}^{+}(k)] &= \langle h|k\rangle \,\hat{\mathbb{I}}_{\mathfrak{D}} . \end{aligned}$$
(3.23)

iii) For any $h, k \in \mathcal{H}$, the VEVs of monomials of degree two of the creation and annihilation operators are given by $(^{5})$

$$(\mathbf{v} | \hat{a}^{+}(h)\hat{a}^{+}(k) \mathbf{v}) = (\mathbf{v} | \hat{a}^{+}(h)\hat{a}^{-}(k) \mathbf{v}) = (\mathbf{v} | \hat{a}^{-}(h)\hat{a}^{-}(k) \mathbf{v}) = 0 , (\mathbf{v} | \hat{a}^{-}(h)\hat{a}^{+}(k) \mathbf{v}) = \langle h | k \rangle .$$
(3.24)

Remark 3.4. Statement i) can be expressed in other terms saying that the vacuum state \boldsymbol{v} is cyclic for the algebra generated by the identity operator $\hat{\mathbb{I}}_{\mathfrak{D}}$ and by the set of all creation and annihilation operators $\hat{a}^+(h), \hat{a}^-(h)$ $(h \in \mathcal{H})$.

Proof. i) Recalling the definition (3.17) of $\hat{a}^+(h)$ $(h \in \mathcal{H})$, it can be trivially proved by induction on $n \in \{1, 2, 3, ...\}$ that

$$\hat{a}^{+}(h_1)\dots\hat{a}^{+}(h_n)\,\boldsymbol{v} = \sqrt{(n+1)!} \ h_1 \vee \dots \vee h_n ;$$
 (3.25)

in view of the definition of $\mathfrak{D}^{\vee}(\mathcal{H})$, the above identity allows to infer the thesis.

ii) The thesis follows if one can prove the component-wise versions of the relations in Eq. (3.23). To this purpose, first notice that, due to the relations in Eq. (3.20), there hold $[\hat{a}^+(h), \hat{a}^+(k)] \boldsymbol{v} = \sqrt{2} (h \vee k - k \vee h) = \boldsymbol{o}, [\hat{a}^-(h), \hat{a}^-(k)] \boldsymbol{v} = \boldsymbol{o}$ and $[\hat{a}^-(h), \hat{a}^+(k)] \boldsymbol{v} = \hat{a}^-(h) k = \langle h|k \rangle$. Next, let us show that, for any $n \in \{1, 2, 3, ...\}$ and for any factorized element $f^{(n)} = f_1 \vee ... \vee f_n \in \mathcal{H}^{\vee n}$, there holds

$$[\hat{a}^{+}(h), \hat{a}^{+}(k)] f^{(n)} = 0^{(n)} , \qquad (3.26)$$

⁵Of course, with some computational effort, this result could be generalized by Wick's theorem [142, 146] to a monomial of any order in the creation and annihilation operators; we do not report the general result here for the pursue of brevity, since we will only need the relations in Eq. (3.24) for the developments to be discussed in the following.

CHAPTER 3. QUANTUM FIELD THEORY ON SPATIAL DOMAINS WITH BOUNDARIES

$$[\hat{a}^{-}(h), \hat{a}^{-}(k)] f^{(n)} = 0^{(n)} , \qquad (3.27)$$

$$[\hat{a}^{-}(h), \hat{a}^{+}(k)] f^{(n)} = \langle h|k\rangle f^{(n)} .$$
(3.28)

Eq. (3.26) follows trivially from the definition of symmetrized tensor product (see Eq. (3.3)) noting that $\hat{a}^+(h) \hat{a}^+(k) f^{(n)} = \sqrt{(n+1)(n+2)} h \lor k \lor f_1 \lor \ldots \lor f_n$. Once Eq. (3.26) has been extablished, one can easily infer Eq. (3.27) recalling that $\hat{a}^-(h)$ is the adjoint of $\hat{a}^+(h)$ for all $h \in \mathcal{H}$ (see Proposition 3.5). Finally, keeping in mind the definitions (3.17) (3.18) of the creation and annihilation operators, one obtains

$$\hat{a}^{-}(h)\hat{a}^{+}(k)f^{(n)} = \hat{a}^{-}(h)(\sqrt{n+1} \quad k \lor f_1 \lor \dots \lor f_n) =$$

$$\frac{1}{n+1} \left(\langle h|k \rangle \ f_1 \lor \dots \lor f_n + \sum_{i=1}^n \langle h|f_i \rangle \ k \lor f_1 \lor \dots \lor f_i \lor \dots \lor f_n \right),$$

$$\hat{a}^{+}(k)\hat{a}^{-}(h)f^{(n)} = \frac{1}{n}\sum_{i=1}^n \langle h|f_i \rangle \ \hat{a}^{+}(k)(f_1 \lor \dots \lor f_i \lor \dots \lor f_n) =$$

$$\frac{1}{n}\sum_{i=1}^n \langle h|f_i \rangle \ k \lor f_1 \lor \dots \lor f_i \lor \dots \lor f_n ;$$

the thesis (3.28) follows by taking the difference of the above expressions. iii) First notice that, due to Proposition 3.5, there holds

$$(\boldsymbol{v} \mid \hat{a}^{+}(h)\hat{a}^{+}(k) \boldsymbol{v}) = (\hat{a}^{-}(k)\hat{a}^{-}(h) \boldsymbol{v} \mid \boldsymbol{v}) ;$$

then, all the expressions in the first line of Eq. (3.24) are easily seen to vanish because of the convention (3.20). On the other hand, due to (the already proved) item ii) of the present theorem, there holds $\hat{a}^{-}(h) \hat{a}^{+}(k) = [\hat{a}^{-}(h), \hat{a}^{+}(k)] + \hat{a}^{+}(k) \hat{a}^{-}(h)$; then, the identity in the second line of Eq. (3.24) follows using again item ii) of the present theorem and recalling once more the convention (3.20).

For completeness, let us also give the following result.

Lemma 3.7. Let $h \in \mathcal{H}$ and consider the creation and annihilation operators $\hat{a}^+(h)$, $\hat{a}^-(h)$, along with the particle number operator $\hat{\mathcal{N}}$ (⁶). Then, there hold

$$[\hat{\mathcal{N}}, \hat{a}^+(h)] = \hat{a}^+(h) , [\hat{\mathcal{N}}, \hat{a}^-(h)] = -\hat{a}^-(h) .$$
(3.29)

Proof. Also in this case the thesis follows by proving component-wise versions of the relations in Eq. (3.29). First notice that, due to Eq. (3.20), $[\hat{\mathcal{N}}, \hat{a}^+(h)] \boldsymbol{v} = \hat{\mathcal{N}}h = h = \hat{a}^+(h)\boldsymbol{v}$ and $[\hat{\mathcal{N}}, \hat{a}^-(h)]\boldsymbol{v} = \boldsymbol{o} = \hat{a}^-(h)\boldsymbol{v}$. Then, it suffices to show that, for all $n \in \{1, 2, 3, ...\}$ and for all $f^{(n)} \in \mathcal{H}^{\vee n}$ of the form $f^{(n)} = f_1 \vee ... \vee f_n$, one has $[\hat{\mathcal{N}}, \hat{a}^+(h)] f^{(n)} = \hat{a}^+(h) f^{(n)}$ and $[\hat{\mathcal{N}}, \hat{a}^-(h)] f^{(n)} = -\hat{a}^-(h) f^{(n)}$. In fact, the latter identities can be inferred by straightforward computations, recalling the definitions (3.15) and (3.17) (3.18) of $\hat{\mathcal{N}}$ and $\hat{a}^+(h), \hat{a}^-(h)$; this yields the thesis.

⁶Recall that all these operators are defined on the common dense domain $\mathfrak{D}^{\vee}(\mathcal{H})$.

Segal quantization.

Let again $\mathfrak{F}^{\vee}(\mathcal{H})$ denote the Fock space constructed on any given, abstract Hilbert space \mathcal{H} , with inner product $\langle | \rangle$. In accordance with Segal's approach to field quantization (see, e.g., [43, 58, 130]), we give the following definition.

Definition 3.8. For any $h \in \mathcal{H}$, the (h-smeared) Segal field and conjugate momentum are the (unbounded) operators $\hat{\Phi}_S(h)$, $\hat{\Pi}_S(h) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ given, respectively, by

$$\hat{\Phi}_S(h) := \frac{1}{\sqrt{2}} \left(\hat{a}^-(h) + \hat{a}^+(h) \right) \,, \tag{3.30}$$

$$\hat{\Pi}_{S}(h) := \frac{1}{i\sqrt{2}} \left(\hat{a}^{-}(h) - \hat{a}^{+}(h) \right) \,. \tag{3.31}$$

Remark 3.5. It can be readily checked that, for any $h \in \mathcal{H}$, there holds

$$\Pi_S(h) = \Phi_S(ih) ; \qquad (3.32)$$

so, the Segal momentum is not strictly necessary within the framework considered in this subsection. For this reason, most of the forthcoming results will be formulated only in terms of the Segal field $\hat{\Phi}_S(h)$. Nevertheless, let us anticipate that the Segal momentum $\hat{\Pi}_S(h)$ turns out to be useful in order to motivate the definition of Wightman conjugate momentum, as well as in certain related considerations, to be discussed in the following subsection 3.1.

Recall that the maps $h \mapsto \hat{a}^+(h)$, $h \mapsto \hat{a}^-(h)$ are, respectively, \mathbb{C} -linear and \mathbb{C} -antilinear (see the comments below Definition 3.4); in consequence of this, the map $h \mapsto \hat{\Phi}_S(h)$ (usually referred to as *Segal quantization*) is neither \mathbb{C} -linear nor \mathbb{C} -antilinear, but only \mathbb{R} -linear.

It is well-known [43, 130] that, for any $h \in \mathcal{H}$, $\hat{\Phi}_S(h)$ extends to an essentially self-adjoint (closed) operator of $\mathfrak{F}^{\vee}(\mathcal{H})$ with domain of self-adjointness $\mathfrak{D}^{\vee}(\mathcal{H})$. Moreover, for any $n \in \{1, 2, 3, ...\}$ and for any family $h_i \in \mathcal{H}^1$, i = 1, ..., n, the finite product $\hat{\Phi}_S(h_1) ... \hat{\Phi}_S(h_n)$ is well-defined on the finite particle subspace $\mathfrak{D}^{\vee}(\mathcal{H})$.

Proposition 3.9. There hold the following results.

i) For any $n \in \{1, 2, 3, ...\}$, the set $\{\hat{\Phi}_S(h_1) ... \hat{\Phi}_S(h_n) v | h_i \in \mathcal{H}, \text{ for } i = 1, ..., n, n \in \{1, 2, 3, ...\}\}$ spans $\mathfrak{D}^{\vee}(\mathcal{H})$.

ii) For any $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$, the map $\mathcal{H} \to \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{F}^{\vee}(\mathcal{H})$, $h \mapsto \hat{\Phi}_{S}(h)\mathbf{f}$ is continuous. iii) For any $h, k \in \mathcal{H}$, there holds the commutation relation

$$[\hat{\Phi}_S(h), \hat{\Phi}_S(k)] = i \Im \langle h|k \rangle \,\hat{\mathbb{I}}_{\mathfrak{D}} \tag{3.33}$$

(here $\Im\langle h|k\rangle$ indicates the immaginary part of the inner product $\langle h|k\rangle \in \mathbb{C}$). iv) For any $h, k \in \mathcal{H}$, the VEV of the second-order monomial $\hat{\Phi}_{S}(h)\hat{\Phi}_{S}(k)$ is

$$(\boldsymbol{v} | \hat{\Phi}_S(h) \hat{\Phi}_S(k) \boldsymbol{v}) = \frac{1}{2} \langle h | k \rangle .$$
(3.34)

Remark 3.6. In consequence of statement i), the vacuum state \boldsymbol{v} is cyclic for the algebra generated by the identity operator $\hat{\mathbb{I}}_{\mathfrak{D}}$ and by the set of Segal field operators $\hat{\Phi}_{S}(h)$ $(h \in \mathcal{H})$.

Proof. i) Recalling the definition (3.20) and the identity (3.25), also in this case the thesis can be proved by induction on $n \in \{1, 2, 3, ...\}$, with some computational effort. ii) This is a well-known fact; see, e.g., the proof of item (d) of Theorem X.41 in [130]. iii) First notice that, in view of the definition (3.30) of the Segal field, one easily infers

$$\left[\hat{\Phi}_{S}(h), \hat{\Phi}_{S}(k)\right] = \frac{1}{2} \left(\left[\hat{a}^{-}(h), \hat{a}^{-}(k)\right] + \left[\hat{a}^{-}(h), \hat{a}^{+}(k)\right] + \left[\hat{a}^{+}(h), \hat{a}^{-}(k)\right] + \left[\hat{a}^{+}(h), \hat{a}^{+}(k)\right] \right);$$

due to the CCR (3.23), the above identity implies $[\hat{\Phi}_S(h), \hat{\Phi}_S(k)] = (\langle h|k \rangle - \langle k|h \rangle)/2 \hat{\mathbb{I}}_{\mathfrak{D}}$, which yields Eq. (3.33).

iv) Recalling again the definition (3.30) of $\hat{\Phi}_{S}(h)$ $(h \in \mathcal{H})$, one has

$$(\boldsymbol{v}|\hat{\Phi}_{S}(h)\hat{\Phi}_{S}(k)\boldsymbol{v}) = \frac{1}{2}\left((\boldsymbol{v}|\hat{a}^{-}(h)\hat{a}^{-}(k)\boldsymbol{v}) + (\boldsymbol{v}|\hat{a}^{-}(h)\hat{a}^{+}(k)\boldsymbol{v}) + (\boldsymbol{v}|\hat{a}^{+}(h)\hat{a}^{-}(k)\boldsymbol{v}) + (\boldsymbol{v}|\hat{a}^{+}(h)\hat{a}^{+}(k)\boldsymbol{v})\right);$$

now, Eq. (3.34) follows straightforwardly using the CCR (3.23).

Now, let us recall some well-known facts about second quantization; these facts will be employed in the following to introduce a notion of time evolution for the physical theory of a scalar field, to be described later on. We refer to [43, 130] for the proof of the forthcoming Proposition 3.11 (as well as for more general formulations, which can be derived under much weaker assumptions).

Definition 3.10. Consider the single particle Hilbert space \mathcal{H} and let \mathcal{U} be any unitary operator on it; the *second quantization* of \mathcal{U} is the unitary operator $\Gamma(\mathcal{U}) : \mathfrak{F}^{\vee}(\mathcal{H}) \to \mathfrak{F}^{\vee}(\mathcal{H})$ defined so that, for all $n \in \{1, 2, 3, ...\}$, there holds

$$\Gamma(\mathcal{U}) \upharpoonright \mathcal{H}^{\vee n} = \underbrace{\mathcal{U} \otimes \dots \otimes \mathcal{U}}_{n \text{ times}} .$$
(3.35)

Remark 3.7. Note that $\Gamma(\mathcal{U}) \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{D}^{\vee}(\mathcal{H})$.

Proposition 3.11. Let \mathcal{U} be any unitary operator on \mathcal{H} ; for any $h \in \mathcal{H}$ and for all $f \in \mathfrak{D}^{\vee}$, the Segal field $\hat{\Phi}_{S}(h)$ fulfills

$$\Gamma(\mathcal{U})\,\hat{\Phi}_S(h)\,\Gamma(\mathcal{U})^{-1}\boldsymbol{f} = \hat{\Phi}_S(\mathcal{U}h)\,\boldsymbol{f}\,. \tag{3.36}$$

Proof. First of all notice that, for any $n \in \{1, 2, 3, ...\}$ and for any $f^{(n)} \in \mathcal{H}^{\vee n}$ of the form $f^{(n)} = f_1 \vee ... \vee f_n$ $(f_i \in \mathcal{H}, i = 1, ..., n)$, there holds

$$\Gamma(\mathcal{U}) \hat{a}^{+}(h) \Gamma(\mathcal{U})^{-1} f^{(n)} = \Gamma(\mathcal{U}) \hat{a}^{+}(h) \left(\mathcal{U}^{-1} f_{1} \vee ... \vee \mathcal{U}^{-1} f_{n}\right) =$$
$$= \sqrt{n+1} \Gamma(\mathcal{U}) \left(h \vee \mathcal{U}^{-1} f_{1} \vee ... \vee \mathcal{U}^{-1} f_{n}\right) = \sqrt{n+1} \left(\mathcal{U} h\right) \vee f_{1} \vee ... \vee f_{n} = \hat{a}^{+}(\mathcal{U} h) f^{(n)} ;$$

by the usual linearity and density arguments, the above chain of equalities allows to infer that $\Gamma(\mathcal{U}) \hat{a}^+(h) \Gamma(\mathcal{U})^{-1} = \hat{a}^+(\mathcal{U}h)$ on \mathfrak{D}^{\vee} . On the other hand, keeping in mind Proposition 3.5 and taking the adjoint, one readily infers $\Gamma(\mathcal{U}) \hat{a}^-(h) \Gamma(\mathcal{U})^{-1} = \hat{a}^-(\mathcal{U}h)$ on \mathfrak{D}^{\vee} . Then the thesis follows recalling the definition (3.30) of $\hat{\Phi}_S(H)$. \Box

The Wightman field at time zero.

Let $(\mathcal{H}, \langle | \rangle)$ be any abstract Hilbert space endowed with a conjugation $\mathcal{J} : \mathcal{H} \to \mathcal{H}$ (see subsection 2.5); moreover, let \mathcal{A} be some strictly positive, self-adjoint operator on \mathcal{H} (with spectrum $\sigma(\mathcal{A}) \subset [\varepsilon, +\infty)$, for some $\varepsilon > 0$). We assume \mathcal{A} to be \mathcal{J} -real.

On the one hand, following the general construction developed in Chapter 2 (see, in particular, Proposition 2.4 of Section 2.5), one can consider the scale of Hilbert spaces $\mathcal{H}^r \equiv (\mathcal{H}^r, \langle | \rangle_r) \ (r \in [-\infty, +\infty])$ associated to the real powers $\mathcal{A}^{r/2}$. On the other hand, the conjugation over \mathcal{H} admits a continuous extension $\mathcal{J} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$, (see Proposition 2.25). So, we can introduce the linear subspaces

$$\mathcal{H}_{\pm}^{-\infty} := \{ f \in \mathcal{H}^{-\infty} \mid \mathcal{J}f = \pm f \} , \qquad (3.37)$$

fulfilling

$$\mathcal{H}^{-\infty} = \mathcal{H}^{-\infty}_+ \oplus \mathcal{H}^{-\infty}_- , \qquad \mathcal{H}^{-\infty}_- = i\mathcal{H}^{-\infty}_+ .$$
(3.38)

The projectors of $\mathcal{H}^{-\infty}$ onto $\mathcal{H}^{-\infty}_{\pm}$ are

$$\mathcal{P}_{\pm} := \frac{\mathbb{I} \pm \mathcal{J}}{2} : \mathcal{H}^{-\infty} \to \mathcal{H}_{\pm}^{-\infty}$$
(3.39)

and, for each $h \in \mathcal{H}^{-\infty}$, one has

$$h = \mathcal{P}_{+}h + \mathcal{P}_{-}h$$
, $\mathcal{P}_{-}h = i \mathcal{P}_{+}(-ih)$. (3.40)

Again from Proposition 2.25, we know that $\mathcal{JH}^r = \mathcal{H}^r$ for each $r \in [-\infty, +\infty]$. Of course, we can define the subspaces $\mathcal{H}^r_{\pm} := \{f \in \mathcal{H}^r \mid \mathcal{J}f = \pm f\}$ and write an analogue of Eq. (3.38) with $-\infty$ replaced by r; the projections of \mathcal{H}^r onto \mathcal{H}^r_{\pm} are the maps $\mathcal{P}_{\pm} \upharpoonright \mathcal{H}^r$.

In view of the forthcoming physical applications, we choose $\mathcal{H} \equiv \mathcal{H}^0$ as the fundamental, single particle Hilbert space and consider the bosonic Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$ on it. Nonetheless, we will often employ the spaces $\mathcal{H}^{\pm 1/2}$, as well as spaces of other orders; in particular, let us recall that $\mathcal{H}^{1/2} \stackrel{dense}{\hookrightarrow} \mathcal{H} \stackrel{dense}{\hookrightarrow} \mathcal{H}^{-1/2}$ (see item ii) of Proposition 2.5 in Chapter 2). Next, consider the Segal field $\hat{\Phi}_S(h)$ along with the conjugate momentum $\hat{\Pi}_S(h)$ ($h \in \mathcal{H}$); hereafter we will use them to define the so-called Wightman field and its conjugate momentum, which are more closely related to the Wightman axioms for field quantization (and, in particular, are \mathbb{C} -linear in the "test function" h, as required by the Wightman axioms). These fields will be the subject of the forthcoming Definition 3.13, which is preceeded by the following proposition. **Proposition 3.12.** Consider the \mathbb{R} -linear maps

$$\hat{\varphi}: \mathcal{H}_{+}^{-1/2} \to \mathfrak{L}(\mathfrak{D}^{\vee}(\mathcal{H})) , \qquad h \mapsto \hat{\varphi}(h) := \hat{\Phi}_{S}(\mathcal{A}^{-1/4}h) , \qquad (3.41)$$

$$\hat{\pi} : \mathcal{H}^{1/2}_+ \to \mathfrak{L}(\mathfrak{D}^{\vee}(\mathcal{H})) , \qquad h \mapsto \hat{\pi}(h) := \hat{\Pi}_S(\mathcal{A}^{1/4}h) .$$
 (3.42)

These possess unique \mathbb{C} -linear extensions

$$\hat{\varphi} : \mathcal{H}^{-1/2} \to \mathfrak{L}(\mathfrak{D}^{\vee}(\mathcal{H})) , \qquad \hat{\pi} : \mathcal{H}^{1/2} \to \mathfrak{L}(\mathfrak{D}^{\vee}(\mathcal{H})) , \qquad (3.43)$$

given by

$$\hat{\varphi}(h) = \hat{\Phi}_{S}(\mathcal{P}_{+}(\mathcal{A}^{-1/4}h)) + i\,\hat{\Phi}_{S}(\mathcal{P}_{+}(-i\,\mathcal{A}^{-1/4}h)) \qquad \text{for } h \in \mathcal{H}^{-1/2} , \qquad (3.44)$$

$$\hat{\pi}(h) = \hat{\Pi}_{S}(\mathcal{P}_{+}(\mathcal{A}^{1/4}h)) + i\,\hat{\Pi}_{S}(\mathcal{P}_{+}(-i\,\mathcal{A}^{1/4}h)) \qquad \text{for } h \in \mathcal{H}^{1/2} .$$
(3.45)

Remark 3.8. i) The assumptions $h \in \mathcal{H}^{-1/2}$ and $h \in \mathcal{H}^{1/2}$ in Eq.s (3.44) and (3.45), repsectively, are both necessary and sufficient in order for the expressions in the cited equations to make sense. In fact, due to Corollary 2.13 (see, in particular, Eq. (2.95)), one has $\mathcal{A}^{-1/4}h \in \mathcal{H}^0 \equiv \mathcal{H}$ and $\mathcal{A}^{1/4}h \in \mathcal{H}^0 \equiv \mathcal{H}$ if and only if $h \in \mathcal{H}^{-1/2}$ and $h \in$ $\mathcal{H}^{1/2}$, respectively; this grants the well-posedness of the Segal fields and of the conjugate momentums appearing in Eq. (3.44) (3.45).

ii) Of course, the maps $\mathcal{H}^{-1/2} \ni h \mapsto \hat{\varphi}(h)$ and $\mathcal{H}^{1/2} \ni h \mapsto \hat{\pi}(h)$ are both \mathbb{C} -linear. iii) Keeping in mind the identity $\mathcal{P}_{+}h = i \mathcal{P}_{-}(-ih)$ (see Eq. (3.40)) and recalling that $\hat{\Pi}_{S}(\mathcal{A}^{1/4}h) = \hat{\Phi}_{S}(i\mathcal{A}^{1/4}h)$ (see Eq. (3.32)), Eq. (3.45) can be easily re-expressed in terms of the Segal field as

$$\hat{\pi}(h) = i \Big(\hat{\Phi}_S(\mathcal{P}_-(\mathcal{A}^{1/4}h)) + i \, \hat{\Phi}_S(\mathcal{P}_-(-i \, \mathcal{A}^{1/4}h)) \Big) \quad \text{for } h \in \mathcal{H}^{1/2} .$$
(3.46)

Proof. . We are going to show how to derive Eq. (3.44) under the only assumption that $\mathcal{H}^{-1/2} \ni h \mapsto \hat{\varphi}(h)$ is a \mathbb{C} -linear extension of the map $\mathcal{H}^{-1/2}_+ \ni h \mapsto \hat{\Phi}_S(\mathcal{A}^{-1/4}h)$; of course, this automatically grants that the map under analysis exists and is uniquely determined. The analogous statement for $h \mapsto \hat{\pi}(h)$ can be proved similarly, in view of the identity $\hat{\Pi}_S(\mathcal{A}^{1/4}h) = \hat{\Phi}_S(i\mathcal{A}^{1/4}h) \ (h \in \mathcal{H}^{1/2}).$

So, let us first recall that $h = \mathcal{P}_+ h + i \mathcal{P}_+ (-i h)$ for all $h \in \mathcal{H}^{-1/2}$ (see Eq. (3.40)); of course $\mathcal{P}_+ h, \mathcal{P}_+ (-i h) \in \mathcal{H}_+^{-1/2}$. Next, let $\hat{\varphi} : \mathcal{H}^{-1/2} \to \mathfrak{L}(\mathfrak{D}^{\vee}(\mathcal{H}))$ be any \mathbb{C} -linear map; in view of the previous considerations, for any $h \in \mathcal{H}^{-1/2}$, by complex-linearity it follows that $\hat{\varphi}(h) = \hat{\varphi}(\mathcal{P}_+ h) + i \hat{\varphi}(\mathcal{P}_+ (-i h))$. Moreover, requiring $\hat{\varphi}$ to fulfill $\hat{\varphi}(h_+) = \hat{\Phi}_S(\mathcal{A}^{-1/4}h_+)$ for all $h_+ \in \mathcal{H}_+^{-1/2}$, one has $\hat{\varphi}(\mathcal{P}_+ h) + i \hat{\varphi}(\mathcal{P}_+ (-i h)) = \hat{\Phi}_S(\mathcal{A}^{-1/4}(\mathcal{P}_+ h)) + i \hat{\Phi}_S(\mathcal{A}^{-1/4}(\mathcal{P}_+ (-i h)))$. Since \mathcal{P}_+ commutes with any real function of \mathcal{A} (see Corollary 2.26), Eq. (3.44) follows easily. In conclusion, the arbitrariness of $h \in \mathcal{H}^{-1/2}$ yields the thesis. \Box

Definition 3.13. The (h-smeared) Wightman field and the related conjugate momentum at time zero are, respectively, the (unbounded) operators $\hat{\varphi}(h) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ (for $h \in \mathcal{H}^{-1/2}$) and $\hat{\pi}(h) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ (for $h \in \mathcal{H}^{1/2}$) introduced in Proposition 3.12.

122

Remark 3.9. The nomenclature at "time zero" might seem unclear at this stage; its meaning will become apparent in the next subsection, when a notion of *time evolution* for the field theory discussed here will be described.

Lemma 3.14. For any $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$, the maps $\mathcal{H}^{-1/2} \to \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{F}^{\vee}(\mathcal{H})$, $h \mapsto \hat{\varphi}(h)\mathbf{f}$ and $\mathcal{H}^{1/2} \to \mathfrak{D}^{\vee}(\mathcal{H})$, $h \mapsto \hat{\pi}(h)\mathbf{f}$ are continuous.

Proof. First consider the representations (3.44) and (3.45) of the Wightman field and of the conjugate momentum, respectively. Next, recall that the maps $\mathcal{A}^{\pm 1/4} : \mathcal{H}^{\pm 1/2} \to \mathcal{H}$ are Hilbertian isomorphisms (see Corollary 2.13); whence, in particular, they are continuous. On the other hand, in view of the continuity of the extension $\mathcal{J} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ of the conjugation operator (see Proposition 2.25), the restriction of the projector $\mathcal{P}_+ : \mathcal{H}^r \to \mathcal{H}^r$ is also continuous for all $r \in \mathbb{R}$. Then, the thesis follows from the continuity of the map $\mathcal{H} \ni h \mapsto \hat{\Phi}_S(h) \mathbf{f}$ (see item ii) of Proposition 3.9).

Now, let us give an auxiliary result which allows to make contact with the standard literature on the Wightman field [130]; this result also allows to derive easier proofs of the subsequent statements.

Lemma 3.15. The Wightman field and conjugate momentum operators at time zero possess, respectively, the following representations in terms of the creation and annihilation operators $\hat{a}^+(), \hat{a}^-()$:

$$\hat{\varphi}(h) = \frac{1}{\sqrt{2}} \left(\hat{a}^{-} (\mathcal{J}(\mathcal{A}^{-1/4}h)) + \hat{a}^{+} (\mathcal{A}^{-1/4}h) \right) \qquad (h \in \mathcal{H}^{-1/2}) , \qquad (3.47)$$

$$\hat{\pi}(h) = \frac{1}{i\sqrt{2}} \left(\hat{a}^{-} (\mathcal{J}(\mathcal{A}^{1/4}h)) - \hat{a}^{+} (\mathcal{A}^{1/4}h) \right) \qquad (h \in \mathcal{H}^{1/2}) .$$
(3.48)

Proof. As an example, we show how to derive Eq. (3.47); the analogous relation (3.48) can be derived by similar means. First of all, consider the identity (3.44) for the Wightman field $\hat{\varphi}(h)$ ($h \in \mathcal{H}^{-1/2}$); expressing the Segal field $\hat{\Phi}_S$ appearing therein in terms of the creation and annihilation operators $\hat{a}^{\pm}()$ according to Eq. (3.30), it can be readily inferred that

$$\hat{\varphi}(h) = \frac{1}{\sqrt{2}} \left(\hat{a}^{-}(\mathcal{P}_{+}(\mathcal{A}^{-1/4}h)) + \hat{a}^{+}(\mathcal{P}_{+}(\mathcal{A}^{-1/4}h)) + i\,\hat{a}^{-}(-i\,\mathcal{P}_{-}(\mathcal{A}^{-1/4}h)) + i\,\hat{a}^{+}(-i\,\mathcal{P}_{-}(\mathcal{A}^{-1/4}h)) \right)$$

Then, recalling that the maps $h \mapsto \hat{a}^+(h)$ and $h \mapsto \hat{a}^-(h)$ are, respectively, \mathbb{C} -linear and \mathbb{C} -antilinear, it follows that

$$\hat{\varphi}(h) = \frac{1}{\sqrt{2}} \left(\hat{a}^{-} ((\mathcal{P}_{+} - \mathcal{P}_{-})\mathcal{A}^{-1/4}h) + \hat{a}^{+} ((\mathcal{P}_{+} + \mathcal{P}_{-})\mathcal{A}^{-1/4}h) \right);$$

the above identity proves the thesis (3.47) since, in view of the definitions of \mathcal{P}_{\pm} in Eq. (3.39), $\mathcal{P}_{+} - \mathcal{P}_{-} = \mathcal{J}$ and $\mathcal{P}_{+} + \mathcal{P}_{-} = \mathbb{I}$ (with \mathbb{I} indicating the identity operator on \mathcal{H}). \Box

Proposition 3.16. There hold the following commutation relations:

$$\begin{aligned} [\hat{\varphi}(h), \hat{\varphi}(k)] &= \hat{\mathbb{O}}_{\mathfrak{D}} & \text{for } h, k \in \mathcal{H}^{-1/2} ,\\ [\hat{\varphi}(h), \hat{\pi}(k)] &= i \langle \mathcal{J}h | k \rangle \,\hat{\mathbb{I}}_{\mathfrak{D}} & \text{for } h \in \mathcal{H}^{-1/2} , \, k \in \mathcal{H}^{1/2} ,\\ [\hat{\pi}(h), \hat{\pi}(k)] &= \hat{\mathbb{O}}_{\mathfrak{D}} & \text{for } h, k \in \mathcal{H}^{1/2} \end{aligned}$$
(3.49)

where $\langle | \rangle$ indicates the extension (2.103) to $\mathcal{H}^{-1/2} \times \mathcal{H}^{1/2}$ of the inner product on \mathcal{H} .

Remark 3.10. Let $k \in \mathcal{H}^{1/2}$ be any given element; then, if h belongs to of the subspace $\mathcal{H}^{-1/2}_+ \subset \mathcal{H}^{-1/2}$ (which is invariant under the action of \mathcal{J}), the second relation in Eq. (3.49) reduces to

$$\left[\hat{\varphi}(h), \hat{\pi}(k)\right] = i \left\langle h|k\right\rangle \hat{\mathbb{I}}_{\mathfrak{D}} . \tag{3.50}$$

Proof. Let us compute, for example, the only non-vanishing commutator in Eq. (3.49), i.e., $[\hat{\varphi}(h), \hat{\pi}(k)]$. Keeping in mind the expressions derived in Lemma 3.15 (giving the Wightman field and the conjugate momentum in terms of creation and annihilation operators) and recalling the CCR (3.23), one easily infers that

$$\left[\hat{\varphi}(h), \hat{\pi}(k)\right] = -\frac{1}{2i} \left(\langle \mathcal{J}(\mathcal{A}^{-1/4}h) | \mathcal{A}^{1/4}k \rangle + \langle \mathcal{J}(\mathcal{A}^{1/4}k) | \mathcal{A}^{-1/4}h \rangle \right) \hat{\mathbb{I}}_{\mathfrak{D}}$$

where $\langle | \rangle$ indicates the (non-extended) inner product on \mathcal{H} . Since this map is sesquilinear, the basic properties (2.64) of the conjugation \mathcal{J} yield

$$\langle \mathcal{J}(\mathcal{A}^{1/4}k) | \mathcal{A}^{-1/4}h \rangle = \overline{\langle \mathcal{A}^{1/4}k | \mathcal{J}(\mathcal{A}^{-1/4}h) \rangle} = \langle \mathcal{J}(\mathcal{A}^{-1/4}h) | \mathcal{A}^{1/4}k \rangle .$$

Next, notice that $\mathcal{J}\mathcal{A}^{-1/4} = \mathcal{A}^{-1/4}\mathcal{J}$, due to Corollary 2.26; moreover, interpreting the expression $\langle | \rangle$ as the extension (2.103) to $\mathcal{H}^{-1/2} \times \mathcal{H}^{1/2}$ of the inner product on \mathcal{H} and keeping in mind the results of Corollary 2.24 on the Banach adjoints of powers of \mathcal{A} , it follows that $\langle \mathcal{A}^{-1/4}\mathcal{J}h | \mathcal{A}^{1/4} k \rangle = \langle \mathcal{J}h | k \rangle$. Summing up, the above mentioned facts prove the thesis.

Lemma 3.17. The VEVs of the second order monomials in the Wightman field $\hat{\varphi}(h)$ and in the conjugate momentum $\hat{\pi}(h)$ are the following:

$$\begin{aligned} (\boldsymbol{v} | \hat{\varphi}(h) \hat{\varphi}(k) \boldsymbol{v}) &= \frac{1}{2} \langle \mathcal{J}(\mathcal{A}^{-1/4}h) | \mathcal{A}^{-1/4}k \rangle & \text{for } h, k \in \mathcal{H}^{-1/2} , \\ (\boldsymbol{v} | \hat{\varphi}(h) \hat{\pi}(k) \boldsymbol{v}) &= -(\boldsymbol{v} | \hat{\pi}(h) \hat{\varphi}(k) \boldsymbol{v}) = \frac{i}{2} \langle \mathcal{J}h | k \rangle & \text{for } h \in \mathcal{H}^{-1/2}, \ k \in \mathcal{H}^{1/2} , \quad (3.51) \\ (\boldsymbol{v} | \hat{\pi}(h) \hat{\pi}(k) \boldsymbol{v}) &= \frac{1}{2} \langle \mathcal{J}(\mathcal{A}^{1/4}h) | \mathcal{A}^{1/4}k \rangle & \text{for } h, k \in \mathcal{H}^{1/2} \end{aligned}$$

where $\langle | \rangle$ indicates the usual inner product on \mathcal{H} .

Proof. As an example, let us evaluate the VEV $(\boldsymbol{v} | \hat{\varphi}(h) \hat{\varphi}(k) \boldsymbol{v})$. Recalling once more Eq.s (3.47) and (3.48) for the Wightman field and the conjugate momentum, respectively, one easily obtains

$$\begin{split} (\boldsymbol{v} \,|\, \hat{\varphi}(h) \,\hat{\varphi}(k) \,\boldsymbol{v}) = \\ &\frac{1}{2} \left((\boldsymbol{v} | \hat{a}^{-} (\mathcal{J}(\mathcal{A}^{-1/4}h) \,) \hat{a}^{-} (\mathcal{J}(\mathcal{A}^{-1/4}k) \,) \boldsymbol{v}) + (\boldsymbol{v} | \hat{a}^{-} (\mathcal{J}(\mathcal{A}^{-1/4}h) \,) \hat{a}^{+}(k) \boldsymbol{v}) + \right. \\ &\left. + (\boldsymbol{v} | \hat{a}^{+} (\mathcal{A}^{-1/4}h) \hat{a}^{-} (\mathcal{J}(\mathcal{A}^{-1/4}k) \,) \boldsymbol{v}) + (\boldsymbol{v} | \hat{a}^{+} (\mathcal{A}^{-1/4}h) \hat{a}^{+} (\mathcal{A}^{-1/4}k) \boldsymbol{v}) \right) \,. \end{split}$$

Due to item iii) of Proposition 3.6 (see, in particular, Eq. (3.24)), all the VEVs on the right-hand side vanish except for $(\boldsymbol{v}|\hat{a}^{-}(\mathcal{J}(\mathcal{A}^{-1/4}h))\hat{a}^{+}(\mathcal{A}^{-1/4}k)\boldsymbol{v}) = \langle \mathcal{J}(\mathcal{A}^{-1/4}h)|\mathcal{A}^{-1/4}k\rangle$ which gives the thesis.

Let us also point out that, in order to derive the explicit expression in Eq. (3.51) for $(\boldsymbol{v} | \hat{\varphi}(h) \hat{\pi}(k) \boldsymbol{v}), (\boldsymbol{v} | \hat{\pi}(h) \hat{\varphi}(k) \boldsymbol{v}),$ one should also recall that $\mathcal{J}\sqrt{\mathcal{A}} = \sqrt{\mathcal{A}}\mathcal{J}$ (see Corollary 2.26).

Time evolution for the Wightman field.

Let \mathcal{A} denote again a strictly positive, self-adjoint operator on some given Hilbert space \mathcal{H} , which we assume to be endowed with a conjugation \mathcal{J} (such that \mathcal{A} is \mathcal{J} -real).

In order to introduce a notion of "time evolution" for the Wightman field, let us first assume that the space \mathcal{H} is used to describe a physical theory where the role of the *single* particle Hamiltonian is played by the square root of the admissible operator \mathcal{A} , i.e.,

$$\sqrt{\mathcal{A}}$$
: Dom $(\mathcal{A}^{1/2}) \equiv \mathcal{H}^1 \subset \mathcal{H} \to \mathcal{H}$. (3.52)

Due to the abstract formulation we are considering, this assumption might appear a bit obscure and short of any real justification. Nevertheless, the motivations lying behind it will soon become clear: in fact, we will show in the forthcoming Corollary 3.23 that in consequence of this choice the time evolution of the Wightman field fulfills, as expected, the Klein-Gordon evolution equation (in a suitable strong sense).

Due to Stone's theorem [119], the time evolution of the non-interacting single-particle theory is implemented on \mathcal{H} in the Schrödinger picture by the strongly continuous one-parameter unitary group

$$\mathcal{U}_t := e^{-it\sqrt{\mathcal{A}}} : \mathcal{H} \to \mathcal{H} , \qquad (3.53)$$

where the (otherwise arbitrary) parameter $t \in \mathbb{R}$ plays the role of *time*.

Remark 3.11. The time parameter t shall not be confused with the other variable $\mathbf{t} \in \mathbb{R}$ used in Chapter 2 to define the exponential operators $e^{-\mathbf{t}\mathcal{A}}$, $e^{-\mathbf{t}\sqrt{\mathcal{A}}}$,

⁷Nonetheless, let us point out that, with the Wick rotation $t = -i\mathbf{t}$, one has $\mathcal{U}_{-i\mathbf{t}} = e^{-\mathbf{t}\sqrt{\mathcal{A}}}$, i.e., $\mathcal{U}_{-i\mathbf{t}}$ coincides with the cylinder operator.

CHAPTER 3. QUANTUM FIELD THEORY ON SPATIAL DOMAINS WITH BOUNDARIES

Before moving on, let us point out a couple of facts on the family of operators \mathcal{U}_t $(t \in \mathbb{R})$ and on the related operators $\cos(\sqrt{\mathcal{A}}t) := (\mathcal{U}_t + \mathcal{U}_{-t})/2$, $\sin(\sqrt{\mathcal{A}}t)/\sqrt{\mathcal{A}} := (\mathcal{U}_t - \mathcal{U}_{-t})/2i\sqrt{\mathcal{A}} : \mathcal{H} \to \mathcal{H}$, which will be employed in the forthcoming developments. Due to Proposition 2.12 in Chapter 2, all the operators mentioned above (for any $t \in \mathbb{R}$) possess unique, continuous linear extensions to $\mathcal{H}^{-\infty}$, indicated with the same symbols:

$$\mathcal{U}_t \ , \ \cos(\sqrt{\mathcal{A}} t) \ , \ (\sin(\sqrt{\mathcal{A}} t)/\sqrt{\mathcal{A}}) \ : \ \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty} \ .$$
 (3.54)

In particular, due to item ii) of Proposition 2.12, for any $r \in \mathbb{R}$ and for any $f \in \mathcal{H}^r$ there hold

$$\|\mathcal{U}_t f\|_r = \|f\|_r ,$$

$$\overline{\mathcal{A}} t f\|_r \le \|f\|_r , \qquad \|(\sin(\sqrt{\mathcal{A}} t)/\sqrt{\mathcal{A}})f\|_r \le \|f\|_{r-1} ;$$
(3.55)

moreover, item iii) of the same proposition (here employed with b = 0) allows to infer that \mathcal{U}_t is an Hilbertian automorphism of \mathcal{H}^r , for any $t \in \mathbb{R}$ and for any $r \in \mathbb{R}$.

Next, following the standard *second quantization* approach (see, Definition 3.10), we introduce the one-parameter unitary group given by

$$\Gamma(\mathcal{U}_t): \mathfrak{F}^{\vee}(\mathcal{H}) \to \mathfrak{F}^{\vee}(\mathcal{H}) \qquad (t \in \mathbb{R}) ;$$

$$(3.56)$$

then, from Definition 3.10, it follows straightforwardly that

 $\|\cos(v)\|$

$$\Gamma(\mathcal{U}_t)^{-1} = \Gamma(\mathcal{U}_t^{-1}) \tag{3.57}$$

 $(^{8})$. In the spirit of Heisenberg picture for time evolution, we give the following definition.

Definition 3.18. The Wightman field and conjugate momentum at time $t \in \mathbb{R}$ are the (unbounded) operators $\hat{\varphi}_t(h) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ and $\hat{\pi}_t(h) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ defined, respectively, as

$$\hat{\varphi}_t(h) := \Gamma(\mathcal{U}_t)^{-1} \,\hat{\varphi}(h) \,\Gamma(\mathcal{U}_t) \qquad \text{for } h \in \mathcal{H}^{-1/2} \,, \tag{3.58}$$

$$\hat{\pi}_t(h) := \Gamma(\mathcal{U}_t)^{-1} \,\hat{\pi}(h) \, \Gamma(\mathcal{U}_t) \qquad \text{for } h \in \mathcal{H}^{1/2} \,. \tag{3.59}$$

⁸For completeness, let us also mention that, setting

$$\mathfrak{D}_1 := \{ \boldsymbol{f} \sum_n f^{(n)} \in \mathfrak{D}^{\vee}(\mathcal{H}) \mid f^{(n)} \in (\mathrm{Dom}(\sqrt{\mathcal{A}}))^{\vee n} \equiv (\mathcal{H}^1)^{\vee n}, \text{ for all } n \in \mathbb{N} \} \}$$

one could also consider the *free Hamiltonian* for the Hermitian scalar field; this is the densely defined operator $d\Gamma(\sqrt{\mathcal{A}}) : \mathfrak{D}_1 \subset \mathfrak{F}^{\vee}(\mathcal{H}) \to \mathfrak{F}^{\vee}(\mathcal{H})$ which, for all $n \in \mathbb{N}$, fulfills

$$d\Gamma(\sqrt{\mathcal{A}}) \upharpoonright (\mathfrak{D}_1 \cap \mathcal{H}^{\vee n}) = \sum_{i=1}^n (\sqrt{\mathcal{A}})^{\delta_{i,1}} \otimes \dots \otimes (\sqrt{\mathcal{A}})^{\delta_{i,n}}$$

where δ indicates the Kroneker delta ($\delta_{ij} = 1$ if i = j, $\delta_{ij} = 0$ if $i \neq j$) and, by convention, we put $(\sqrt{\mathcal{A}})^0 := \mathbb{I}_{\mathcal{H}^{-1/2}}$ (i.e., the identity operator on $\mathcal{H}^{-1/2}$). As well-known [130], $d\Gamma(\sqrt{\mathcal{A}})$ is essentially self-adjoint on \mathfrak{D}_1 and there holds

$$\Gamma(\mathcal{U}_t) = e^{-it \, d\Gamma(\sqrt{\mathcal{A}}\,)}$$

126

Remark 3.12. Of course, for t = 0 there holds $\mathcal{U}_0 = \mathbb{I}$ (the identity operator on \mathcal{H}); on the other hand, it follows straightforwardly from Definition 3.10 that $\Gamma(\mathbb{I}) = \hat{\mathbb{I}}_{\mathfrak{F}}$ (the identity on the Fock space). Therefore, there holds

$$\hat{\varphi}_0(h) = \hat{\varphi}(h) \quad (h \in \mathcal{H}^{-1/2}) , \qquad \hat{\pi}_0(h) = \hat{\pi}(h) \quad (h \in \mathcal{H}^{1/2}) ;$$
 (3.60)

this justifies the nomenclature "at time zero" adopted in Definition 3.13 for the Wightman field $\hat{\varphi}(h)$ and for the conjugate momentum $\hat{\pi}(h)$.

Proposition 3.19. The Wightman field $\hat{\varphi}_t(h)$ and conjugate momentum $\hat{\pi}_t(h)$ at any time $t \in \mathbb{R}$ can be expressed as follows in terms of their time zero analogues $\hat{\varphi}(h), \hat{\pi}(h)$:

$$\hat{\varphi}_t(h) = \hat{\varphi}\left(\cos(\sqrt{\mathcal{A}}\,t)\,h\right) + \hat{\pi}\left(\left(\sin(\sqrt{\mathcal{A}}\,t)/\sqrt{\mathcal{A}}\,\right)\,h\right) \qquad (h \in \mathcal{H}^{-1/2}) \,\,, \tag{3.61}$$

$$\hat{\pi}_t(h) = \hat{\pi} \Big(\cos(\sqrt{\mathcal{A}} t) h \Big) - \hat{\varphi} \Big(\mathcal{A} \left(\sin(\sqrt{\mathcal{A}} t) / \sqrt{\mathcal{A}} \right) h \Big) \qquad (h \in \mathcal{H}^{1/2}) .$$
(3.62)

Remark 3.13. In analogy with the considerations of Remark 3.8 (see, in particular, item i) therein), it appears that the expressions (3.61) and (3.62) do in fact make sense for the respective choices of h indicated therein. In fact, $\cos(\sqrt{A}t)h \in \mathcal{H}^{-1/2}$ and $(\sin(\sqrt{A}t)/\sqrt{A})h \in \mathcal{H}^{1/2}$ for all $h \in \mathcal{H}^{-1/2}$, while $\cos(\sqrt{A}t)h \in \mathcal{H}^{1/2}$ and $\mathcal{A}(\sin(\sqrt{A}t)/\sqrt{A})h \in \mathcal{H}^{-1/2}$ for all $h \in \mathcal{H}^{1/2}$; therefore, all the Wightman fields and conjugate momentums at time zero appearing in the cited equations are well-defined.

Proof. We only show, as an example, how to derive Eq. (3.61). To this purpose, first recall the identity (3.44), expressing the Wightman field in terms of the Segal field, and the definition (3.58), describing the time evolution of the Wightman field. Then, in view of Proposition 3.11 and of Eq. (3.57), it can be easily inferred that

$$\hat{\varphi}_t(h) = \hat{\Phi}_S(\mathcal{U}_t^{-1}\mathcal{P}_+(\mathcal{A}^{-1/4}h)) + i\,\hat{\Phi}_S(\mathcal{U}_t^{-1}\mathcal{P}_+(-i\,\mathcal{A}^{-1/4}h))) \,.$$

Next, notice that $\mathcal{U}_t^{-1} = e^{it\sqrt{A}} = \cos(\sqrt{A}t) + i\sin(\sqrt{A}t)$, for all $t \in \mathbb{R}$; moreover, both the operators $\cos(\sqrt{A}t)$ and $\sin(\sqrt{A}t)$ are \mathcal{J} -real due to Proposition 2.25, so that $\cos(\sqrt{A}t)\mathcal{J} = \mathcal{J}\cos(\sqrt{A}t)$ and $\sin(\sqrt{A}t)\mathcal{J} = \mathcal{J}\sin(\sqrt{A}t)$. Therefore, by the \mathbb{R} -linearity of the Segal field $\hat{\Phi}_S$ (recalling that $\mathcal{P}_{\pm}(if) = i\mathcal{P} \mp f$; see Eq. (3.40)), one has

$$\begin{aligned} \hat{\varphi}_t(h) &= \\ \hat{\Phi}_S(\mathcal{P}_+(\cos(\sqrt{\mathcal{A}}\,t)\mathcal{A}^{-1/4}h)\,) + i\,\hat{\Phi}_S(\mathcal{P}_+(-i\,\cos(\sqrt{\mathcal{A}}\,t)\mathcal{A}^{-1/4}h)\,) + \\ &+ \hat{\Phi}_S(i\,\mathcal{P}_+(\sin(\sqrt{\mathcal{A}}\,t)\mathcal{A}^{-1/4}h)\,) + i\,\hat{\Phi}_S(i\,\mathcal{P}_+(-i\,\sin(\sqrt{\mathcal{A}}\,t)\mathcal{A}^{-1/4}h)\,) ; \end{aligned}$$

then, the thesis follows using, again, the identities (3.44) (3.45) for the Wightman field and the conjugate momentum at time zero (notice that $\hat{\Phi}_S(i \mathcal{P}_+(\mathcal{A}^{-1/4}h)) + i \hat{\Phi}_S(i \mathcal{P}_-(-i \mathcal{A}^{-1/4}h))$ $= \hat{\pi}(\mathcal{A}^{-1/2}h)$, for all $h \in \mathcal{H}^{-1} \supset \mathcal{H}^{-1/2}$).

For completeness, let us also report the following result, giving the commutation relations for the Wightman field and the conjugate momentum at unequal times. **Proposition 3.20.** For any $t, t' \in \mathbb{R}$, there hold the commutation relations at unequal times $(t \neq t')$

$$\begin{aligned} &[\hat{\varphi}_t(h), \hat{\varphi}_{t'}(k)] = -i \langle \mathcal{J}h | (\sin(\sqrt{\mathcal{A}} (t-t'))/\sqrt{\mathcal{A}}) k \rangle \, \hat{\mathbb{I}}_{\mathfrak{D}} & \text{for } h, k \in \mathcal{H}^{-1/2} \,, \\ &[\hat{\varphi}_t(h), \hat{\pi}_{t'}(k)] = +i \langle \mathcal{J}h | \cos(\sqrt{\mathcal{A}} (t-t')) k \rangle \, \hat{\mathbb{I}}_{\mathfrak{D}} & \text{for } h \in \mathcal{H}^{-1/2} \,, \ k \in \mathcal{H}^{1/2} \,, \\ &[\hat{\pi}_t(h), \hat{\pi}_{t'}(k)] = -i \langle \mathcal{J}h | (\sin(\sqrt{\mathcal{A}} (t-t'))/\sqrt{\mathcal{A}}) k \rangle \, \hat{\mathbb{I}}_{\mathfrak{D}} & \text{for } h, k \in \mathcal{H}^{1/2} \,. \end{aligned}$$

Proof. We only compute, as an example, the commutator $[\hat{\varphi}_t(h), \hat{\varphi}_{t'}(k)]$. To this purpose, recall the expression (3.61) for the time evolution of the Wightman field, along with the commutation relations at time zero (3.49); these facts allow to infer by elementary computations that

$$[\hat{\varphi}_t(h), \hat{\varphi}_{t'}(k)] = i \langle \mathcal{J}(\cos(\sqrt{\mathcal{A}} t) h) | (\sin(\sqrt{\mathcal{A}} t')/\sqrt{\mathcal{A}}) k \rangle - i \langle \mathcal{J}(\cos(\sqrt{\mathcal{A}} t') k) | (\sin(\sqrt{\mathcal{A}} t)/\sqrt{\mathcal{A}}) h \rangle \hat{\mathbb{I}}_{\mathfrak{D}}$$

Next notice that, $\langle \mathcal{J}(\cos(\sqrt{\mathcal{A}}t')k)|(\sin(\sqrt{\mathcal{A}}t)/\sqrt{\mathcal{A}})h\rangle = \langle \mathcal{J}(\sin(\sqrt{\mathcal{A}}t)/\sqrt{\mathcal{A}})h|\cos(\sqrt{\mathcal{A}}t')k\rangle$ (use the properties of the conjugation \mathcal{J} , keeping in mind the sesquilinearity of the inner product $\langle | \rangle$). Moreover, recall that $\cos(\sqrt{\mathcal{A}}t)$ and $\sin(\sqrt{\mathcal{A}}t')/\sqrt{\mathcal{A}}$ both commute with \mathcal{J} in consequence of Proposition 2.25; so, due to the results of Corollary 2.24 on Banach adjoints, it follows that

$$\left[\hat{\varphi}_t(h), \hat{\varphi}_{t'}(k)\right] = i \left\langle \mathcal{J}h \right| \left(\left(\cos(\sqrt{\mathcal{A}} t) \sin(\sqrt{\mathcal{A}} t') - \cos(\sqrt{\mathcal{A}} t') \sin(\sqrt{\mathcal{A}} t)\right) / \sqrt{\mathcal{A}} \right) k \right\rangle \hat{\mathbb{I}}_{\mathfrak{D}} .$$

Then, the thesis follows by elementary functional calculus $(^9)$.

Proposition 3.21. Let $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$ and $n \in \mathbb{N}$. Then, the maps $\mathbb{R} \mapsto \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{F}^{\vee}(\mathcal{H})$, $t \mapsto \hat{\varphi}_t(h) \mathbf{f}$ and $\mathbb{R} \mapsto \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{F}^{\vee}(\mathcal{H})$, $t \mapsto \hat{\pi}_t(h) \mathbf{f}$ are well-posed and of class C^n for all $h \in \mathcal{H}^{-1/2+n}$ and all $h \in \mathcal{H}^{1/2+n}$, respectively. In particular, for any $h \in \mathcal{H}^{3/2}$, the map $t \mapsto \hat{\varphi}_t(h) \mathbf{f}$ is of class C^2 and its first and second derivative are, respectively, given by

$$\frac{d}{dt}\left(\hat{\varphi}_t(h)\boldsymbol{f}\right) = \hat{\pi}_t(h)\boldsymbol{f}, \qquad \frac{d^2}{dt^2}\left(\hat{\varphi}_t(h)\boldsymbol{f}\right) = -\hat{\varphi}_t(\mathcal{A}\,h)\boldsymbol{f}. \tag{3.64}$$

Proof. As an example, we only discuss the differentiability of the map $t \mapsto \hat{\varphi}_t(h) \boldsymbol{f}$. First of all, recall that the Wightman field at any time $t \in \mathbb{R}$ can be expressed as $\hat{\varphi}_t(h) = \hat{\varphi}(\cos(\sqrt{A}t)h) + \hat{\pi}((\sin(\sqrt{A}t)/\sqrt{A})h)$ (see Eq. (3.61)). On the other hand, Proposition 2.30 grants that both the maps $\mathbb{R} \to \mathcal{H}^{-1/2}$, $t \mapsto \cos(\sqrt{A}t)h$ and $\mathbb{R} \to \mathcal{H}^{1/2}$, $t \mapsto (\sin(\sqrt{A}t)/\sqrt{A})h$ are of class C^n , for any $h \in \mathcal{H}^{-1/2+n}$. Moreover, due to Lemma 3.14, both the \mathbb{C} -linear maps $\mathcal{H}^{-1/2} \ni h \mapsto \hat{\varphi}(h)\boldsymbol{f}$ and $\mathcal{H}^{1/2} \ni h \mapsto \hat{\pi}(h)\boldsymbol{f}$ are continuous,

⁹One should recall the trigonometric identity $\cos a \sin b - \cos a \sin b = -\sin(a-b)$, for any $a, b \in \mathbb{R}$.

whence analytic, for any $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$. Summing up, the map $t \mapsto \hat{\varphi}_t(h) \mathbf{f}$ is the composition of functions which are either smooth or of class C^n , which proves the first part of the thesis.

Finally note that, by linearity and continuity, there hold

$$\frac{d^n}{dt^n} \left(\hat{\varphi}(\cos(\sqrt{A}t) h) \boldsymbol{f} \right) = \hat{\varphi} \left(\frac{d^n}{dt^n} \cos(\sqrt{A}t) h \right) \boldsymbol{f} ,$$
$$\frac{d^n}{dt^n} \left(\hat{\pi}(\left(\sin(\sqrt{A}t) / \sqrt{A} \right) h \right) \boldsymbol{f}) = \hat{\varphi} \left(\frac{d^n}{dt^n} \left(\sin(\sqrt{A}t) / \sqrt{A} \right) h \right) \boldsymbol{f} ;$$

then, the identities in Eq. (3.64) can be easily derived computing (via functional calculus) the derivatives $\frac{d^n}{dt^n} \cos(\sqrt{A}t)$, $\frac{d^n}{dt^n} (\sin(\sqrt{A}t)/\sqrt{A})$ and keeping in mind the identities in Eq.s (3.61) and (3.62).

Remark 3.14. Starting from the identities in Eq. (3.64) and working by recursion, one can derive explicit expressions also for the higher order derivatives of both the maps $t \mapsto \hat{\varphi}_t(h) \mathbf{f}, \hat{\pi}_t(h) \mathbf{f}$ (making suitable assumptions for h, in order to grant their existence). We will not report these expressions here because for the following developments we shall only need at most second order "time derivatives" of the Wightman field.

In view of the results extablished in the above Proposition 3.21, we give the forthcoming definition.

Definition 3.22. For any $n \in \mathbb{N}$ and for all $h \in \mathcal{H}^{-1/2+n}$, the (h-smeared) *n-th derivative* of the Wightman field at time t is the unbounded operator $\partial_t^n \hat{\varphi}_t(h) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ which, for all $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$, fulfills

$$\partial_t^n \hat{\varphi}_t(h) \boldsymbol{f} = \frac{d^n}{dt^n} \left(\hat{\varphi}_t(h) \boldsymbol{f} \right) \,. \tag{3.65}$$

Corollary 3.23. For any $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$ and for any $h \in \mathcal{H}^{3/2}$, the time-evolved Wightman field fulfills the strong form of the abstract Klein-Gordon evolution equation

$$\begin{cases} \left(\partial_{tt}\hat{\varphi}_{t}(h) + \hat{\varphi}_{t}(\mathcal{A}h)\right)\boldsymbol{f} = \boldsymbol{o} ,\\ \hat{\varphi}_{t}(h)\boldsymbol{f}\Big|_{t=0} = \hat{\varphi}(h)\boldsymbol{f} ,\\ \partial_{t}\hat{\varphi}_{t}(h)\boldsymbol{f}\Big|_{t=0} = \hat{\pi}(h)\boldsymbol{f} \end{cases}$$
(3.66)

where $o \in \mathfrak{D}^{\vee}(\mathcal{H})$ is the null element of the Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$.

Proof. The thesis follows straightforwardly from the results derived in Proposition 3.21, keeping in mind the considerations of Remark 3.12 and of item i) of Remark 3.14. \Box

Remark 3.15. i) It appears that Eq. (3.66) can be interpreted as the second quantized version of the classical Klein-Gordon equation, that is the evolution equation

$$\begin{cases} (\partial_{tt} + \mathcal{A})\varphi_t = 0 ,\\ \varphi_t|_{t=0} = \varphi ,\\ \partial_t \varphi_t|_{t=0} = \pi \end{cases}$$
(3.67)

where \mathcal{A} is some suitable, second order elliptic differential operator (e.g., of Schrödinger type), $t \mapsto \varphi_t$ is a function-valued map and φ, π are prescribed initial data. In view of this correspondence, it appears that the parameter t can in fact be interpreted as the time also in the abstract Klein-Gordon equation.

ii) Consider the ordinary differential equation corresponding to Eq. (3.67)

$$\begin{cases} (\partial_{tt} + \lambda)\varphi(t) = 0 ,\\ \varphi(t)|_{t=0} = \varphi_0 ,\\ \partial_t \varphi(t)|_{t=0} = \pi_0 \end{cases}$$
(3.68)

where $\omega > 0$ and $\varphi_0, \pi_0 \in \mathbb{R}$ are assigned and $t \mapsto \varphi(t)$ is an at least twice-differentiable function; by elementary arguments it follows that the solution of Eq. (3.68) is given by

$$\varphi(t) = \varphi_0 \cos(\sqrt{\omega} t) + \pi_0 \sin(\sqrt{\omega} t) / \sqrt{\omega} . \qquad (3.69)$$

In view of this fact, working backwords, one could have started with an ansatz and define the Wightman field $\hat{\varphi}_t(h)$ at time $t \in \mathbb{R}$ $(h \in \mathcal{H}^1)$ according to Eq. (3.61) (which in our approach is instead a consequence derived by more primitive definitions).

Using the above results and those reported in Lemma 3.17, one can evaluate the VEVs of the monomials of any order in the Wightman field (as well as the VEVs involving the conjugate momentum and their time derivatives). As well-known, due to Wick's theorem [142, 146], this operation reduces to the computation of the VEV of the field squared. Let us first give the following corollary descending straightforwardly from Proposition 3.21.

Corollary 3.24. Let $\mathbf{v} \in \mathfrak{D}^{\vee}(\mathcal{H})$ be the vacuum state and let $n \in \mathbb{N}$. *i)* For any pair $h \in \mathcal{H}^{-1/2+n}$, $k \in \mathcal{H}^{-1/2}$ and for any fixed $t' \in \mathbb{R}$, the map $\mathbb{R} \to \mathbb{C}$, $t \mapsto (\mathbf{v} | \hat{\varphi}_t(h) \hat{\varphi}_{t'}(k) \mathbf{v})$ is of class C^n ; for any $j \leq n$, there holds

$$\frac{d^{j}}{dt^{j}}\left(\boldsymbol{v}\,|\,\hat{\varphi}_{t}(h)\,\hat{\varphi}_{t'}(k)\,\boldsymbol{v}\right) = \left(\boldsymbol{v}\,|\,\partial_{t}^{j}\hat{\varphi}_{t}(h)\,\hat{\varphi}_{t'}(k)\,\boldsymbol{v}\right)\,.$$
(3.70)

ii) For any pair $h \in \mathcal{H}^{-1/2}$, $k \in \mathcal{H}^{-1/2+n}$ and for any fixed $t \in \mathbb{R}$, the map $\mathbb{R} \to \mathbb{C}$, $t' \mapsto (\boldsymbol{v} | \hat{\varphi}_t(h) \hat{\varphi}_{t'}(k) \boldsymbol{v})$ is of class C^n ; for any $j \leq n$, there holds

$$\frac{d^{j}}{dt^{\prime j}}\left(\boldsymbol{v}\,|\,\hat{\varphi}_{t}(h)\,\hat{\varphi}_{t^{\prime}}(k)\,\boldsymbol{v}\right) = \left(\boldsymbol{v}\,|\,\hat{\varphi}_{t}(h)\,\partial_{t^{\prime}}^{j}\hat{\varphi}_{t^{\prime}}(k)\,\boldsymbol{v}\right)\,. \tag{3.71}$$

Remark 3.16. i) Of course, it follows that if both $h \in \mathcal{H}^{-1/2+n}$ and $k \in \mathcal{H}^{-1/2+n}$ for some $n \in \mathbb{N}$, then the map $\mathbb{R}^2 \to \mathbb{C}$, $(t, t') \mapsto (\boldsymbol{v} \mid \hat{\varphi}_t(h) \, \hat{\varphi}_{t'}(k) \, \boldsymbol{v})$ is of class C^n jointly in the two variables. Moreover, for any $j, j' \leq n$, there holds

$$\partial_t^j \partial_{t'}^{j'} (\boldsymbol{v} \,|\, \hat{\varphi}_t(h) \,\hat{\varphi}_{t'}(k) \,\boldsymbol{v}) = (\boldsymbol{v} \,|\, \partial_t^j \hat{\varphi}_t(h) \,\partial_{t'}^{j'} \hat{\varphi}_{t'}(k) \,\boldsymbol{v}) \,. \tag{3.72}$$

ii) From the forthcoming proof, it appears that all the above statements can be easily generalized to any expectation value of the form $(\boldsymbol{f} \mid \partial_t^j \hat{\varphi}_t(h) \partial_{t'}^{j'} \hat{\varphi}_{t'}(k) \boldsymbol{g})$, for any given $\boldsymbol{f}, \boldsymbol{g} \in \mathfrak{D}^{\vee}(\mathcal{H})$.

Proof. As an example, we show how to prove statement i). First notice that, for any $\boldsymbol{v} \in \mathfrak{D}^{\vee}(\mathcal{H}), \ k \in \mathcal{H}^{-1/2}$ and for any fixed $t' \in \mathbb{R}$, one has $\hat{\varphi}_{t'}(k) \, \boldsymbol{v} \in \mathfrak{D}^{\vee}(\mathcal{H})$. On the other hand, due to Proposition 3.21, the map $t \mapsto \hat{\varphi}_t(h) \, \hat{\varphi}_{t'}(k) \, \boldsymbol{v}$ is of class C^n for any $h \in \mathcal{H}^{-1/2+n}$. Then, the thesis follows easily from the linearity and continuity (whence, analyticity) of the inner product (|) on $\mathfrak{F}^{\vee}(\mathcal{H})$.

Proposition 3.25. Let $h, k \in \mathcal{H}^{-1/2}$ and let $t, t' \in \mathbb{R}$; then the VEV of the Wightman field squared $(\boldsymbol{v} | \hat{\varphi}_t(h) \hat{\varphi}_{t'}(k) \boldsymbol{v})$ can be expressed as follows:

$$(\boldsymbol{v} | \hat{\varphi}_t(h) \hat{\varphi}_{t'}(k) \boldsymbol{v}) = \frac{1}{2} \langle \mathcal{J}h | (e^{-i(t-t')\sqrt{\mathcal{A}}} / \sqrt{\mathcal{A}})k \rangle , \qquad (3.73)$$

where $\langle | \rangle$ indicates the extension to $\mathcal{H}^{-1/2} \times \mathcal{H}^{1/2}$ of the inner product on \mathcal{H} (¹⁰).

Proof. First notice that, using the expression (3.61) for the Wightman field $\hat{\varphi}_t(h)$ at time $t \in \mathbb{R}$, one has

$$(\boldsymbol{v} | \hat{\varphi}_t(h) \hat{\varphi}_{t'}(k) \boldsymbol{v}) =$$

$$(\boldsymbol{v} | \left[\hat{\varphi}(\cos(\sqrt{A} t)h) \hat{\varphi}(\cos(\sqrt{A} t')k) + \hat{\pi}((\sin(\sqrt{A} t)/\sqrt{A})h) \hat{\pi}((\sin(\sqrt{A} t')/\sqrt{A})k) + \hat{\varphi}(\cos(\sqrt{A} t)h) \hat{\pi}((\sin(\sqrt{A} t')/\sqrt{A})k) + \hat{\pi}((\sin(\sqrt{A} t)/\sqrt{A})h) \hat{\varphi}(\cos(\sqrt{A} t')k) \right] \boldsymbol{v}).$$

To proceed, let us evaluate each of the terms at time zero in the right-hand side of the above equality, using the results derived in Lemma 3.17. Next, recall that the conjugation \mathcal{J} commutes with $\cos(\sqrt{\mathcal{A}}t)$, $\sin(\sqrt{\mathcal{A}}t)$ and with any real power of \mathcal{A} (see Corollary 2.26); interpreting the inner products $\langle | \rangle$ so obtained in terms of the extension (2.103) and using Corollary 2.24 for the (Banach) adjoints of $\cos(\sqrt{\mathcal{A}}t)$, $\sin(\sqrt{\mathcal{A}}t)$, \mathcal{A}^r ($r \in \mathbb{R}$), it can be infered that

$$(\boldsymbol{v} \mid \hat{\varphi}_t(h) \, \hat{\varphi}_{t'}(k) \, \boldsymbol{v}) =$$

$$\frac{1}{2} \langle \mathcal{J}h \mid \mathcal{A}^{-1/2} \left[\left(\cos(\sqrt{\mathcal{A}} t) \cos(\sqrt{\mathcal{A}} t') + \sin(\sqrt{\mathcal{A}} t) \sin(\sqrt{\mathcal{A}} t') \right) + i \left(\cos(\sqrt{\mathcal{A}} t) \sin(\sqrt{\mathcal{A}} t') - \sin(\sqrt{\mathcal{A}} t) \cos(\sqrt{\mathcal{A}} t') \right) \right] k \rangle$$

¹⁰In fact, for any $k \in \mathcal{H}^{-1/2}$, there holds $(e^{-i(t-t')\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}}) k \in \mathcal{H}^{1/2}$ (see Lemma 2.27).

CHAPTER 3. QUANTUM FIELD THEORY ON SPATIAL DOMAINS WITH BOUNDARIES

Then, the thesis follows again by elementary functional calculus $(^{11})$.

132

For the subsequent developments, only the evaluation of the VEVs at equal time is required; these are reported in the forthcoming corollary.

Corollary 3.26. For any pair $h, k \in \mathcal{H}^{3/2}$, the VEVs of the second order monomials in the Wightman field (and the related time derivatives of degree two at most), evaluated at equal times are the following:

$$(\boldsymbol{v} | \hat{\varphi}_t(h) \hat{\varphi}_t(k) \boldsymbol{v}) = \frac{1}{2} \langle \mathcal{J}h | \mathcal{A}^{-1/2} k \rangle ,$$

$$(\boldsymbol{v} | \hat{\varphi}_t(h) \partial_t \hat{\varphi}_t(k) \boldsymbol{v}) = - (\boldsymbol{v} | \partial_t \hat{\varphi}_t(h) \hat{\varphi}_t(k) \boldsymbol{v}) = \frac{i}{2} \langle \mathcal{J}h | k \rangle , \qquad (3.74)$$

$$(\boldsymbol{v} | \partial_t \hat{\varphi}_t(h) \partial_t \hat{\varphi}_t(k) \boldsymbol{v}) = -(\boldsymbol{v} | \partial_{tt} \hat{\varphi}_t(h) \hat{\varphi}_t(k) \boldsymbol{v}) = -(\boldsymbol{v} | \hat{\varphi}_t(h) \partial_{tt} \hat{\varphi}_t(k) \boldsymbol{v}) = \frac{1}{2} \langle \mathcal{J}h | \mathcal{A}^{1/2} k \rangle;$$

again, $\langle | \rangle$ indicates, in general, the extension to $\mathcal{H}^{(2)}$ of the inner product on $\mathcal{H}^{(12)}$.

Proof. First recall the identity (3.72), which allows to express the VEVs in the leaft-hand sides of Eq. (3.74) in terms of the derivatives of $(\boldsymbol{v} | \hat{\varphi}_t(h) \hat{\varphi}_{t'}(k) \boldsymbol{v})$ evaluated at equal times (t' = t); then, the thesis follows easily using the explicit expression (3.73).

Remark 3.17. All the excessions on the left-hand sides of the identities in Eq. (3.74) do, in principle, depend on the time parameter $t \in \mathbb{R}$, which however does not appear in the corresponding right-hand sides. This fact shows that the VEV of the Wick polynomials evaluated at equal times are actually time independent; as a matter of fact, this was to be expected due to the staticity features of the configuration under analysis. We will return on this topic in the following (see Proposition 3.37).

3.2 Scalar quantum field on a spatial domain.

Hereafter we employ the abstract framework developed in the previous section to analyze the case of an Hermitian scalar field living on a suitable open subset of (d+1)-dimensional Minkowski spacetime \mathcal{M}_{d+1} (with $d \in \{1, 2, 3, ...\}$ arbitrary). Let us recall that this setting was already considered in Chapter 1, where the main ideas were presented using a language somehow less precise than the mathematically rigorous one developed in Chapter 2 and in Section 3.1 of the present chapter.

We refer to the same framework described in Chapter 1; in particular, let us recall that we employ natural units (so that c = 1 and $\hbar = 1$) and that we identify Minkowski spacetime with \mathbb{R}^{d+1} using the set of global inertial coordinates

$$x = (x^{\mu})_{\mu=0,1,\dots,d} \equiv (x^0, \mathbf{x}) \equiv (t, \mathbf{x}) : \mathcal{M}_{d+1} \to \mathbb{R}^{d+1}$$
,

¹¹Just notice that $(\cos a \cos b + \sin a \sin b) + i (\cos a \sin b - \sin a \cos b) = e^{-i(a-b)}$, for all $a, b \in \mathbb{R}$.

¹²In fact, under the assumptions made for h, k, the expressions in the right-hand sides of Eq. (3.74) can also be interpreted, a posteriori, in terms of the non-extended inner product on \mathcal{H} .
in terms of which the Minkowski metric η has coefficients $(\eta_{\mu\nu}) = \text{diag}(-1, 1, ..., 1)$ (see Eq.s (1.1-1.3)). We make reference to the theory of a scalar field living on a fixed spatial domain $\Omega \subset \mathbb{R}^d$, which we assume to be open and connected but otherwise arbitrary (possibly, unbounded); time evolution is described on the open subset of Minkowski spacetime given by $\mathbb{R} \times \Omega \subset \mathbb{R}^{d+1} \simeq \mathcal{M}_{d+1}$. Prescribed boundary conditions on $\partial\Omega$ for the field are properly taken into account, as well as the interaction with an assigned scalar potential $V \in C^{\infty}(\Omega)$.

We first reconsider the canonical quantization of the classical theory; this is obtained specializing the general formulation of Section 3.1 to the explicit model under analysis. Next, we implement ZR within the present setting to define a regularized version of the Wightman field evaluated at a spacetime point $x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega$, depending on a complex parameter u introduced on purpose. This regularized field is used to construct natural quantized counterparts of the classical observables, such as the stress-energy tensor, the total energy and the pressure on the boundary. The VEVs of this regularized observables are computed explicitly using the integral kernels analyzed in Chapter 2 of this manuscript. Finally, the renormalized VEVs are defined in terms of the analytic continuation of the corresponding regularized expressions at u = 0, which is the value corresponding formally to the non-regularized field operator.

Let us recall once more that the spatial domain Ω , its boundary $\partial \Omega$ and the potential V are assigned classical objects which do not evolve in time; moreover no back-reaction exterted on them by the quantum field is ever considered in the analysis to be described in the following.

Canonical quantization (revisited).

To make contact with the standard literature [58, 130], let us specialize the abstract approach to canonical quantization described in Section 3.1 for the particular case we are considering in this section.

First of all, we define the single particle Hilbert space to be

$$\mathcal{H} = L^2(\Omega, d\mathbf{x}) \equiv L^2(\Omega) \tag{3.75}$$

(with inner product $\langle f|g\rangle_{L^2} := \int_{\Omega} d\mathbf{x} \,\overline{f}(\mathbf{x}) \,g(\mathbf{x})$), i.e., as the space of square-integrable functions on Ω with respect to the standard Lebesgue measure $d\mathbf{x}$.

The corresponding bosonic Fock space $\mathfrak{F}^{\vee}(\mathcal{H}) = \mathfrak{F}^{\vee}(L^2(\Omega))$ is the direct sum of the totally symmetric tensor powers $\mathcal{H}^{\vee n} = (L^2(\Omega))^{\vee n}$ $(n \in \mathbb{N})$; as well known, each of these tensor powers coincides with the space of square-integrable functions on the *n*-fold cartesian product $\Omega \times ... \times \Omega$ which are invariant under permutation of the coordinates. So,

$$\mathfrak{F}^{\vee}(\mathcal{H}) = \bigoplus_{n=0}^{+\infty} \mathcal{H}^{\vee n}$$
(3.76)

with $\mathcal{H}^{\vee 0} := \mathbb{C}$ and $\mathcal{H}^{\vee n} = L^2_{\vee}(\times_{i=1}^n \Omega, \bigotimes_{i=1}^n d\mathbf{x}) \equiv L^2_{\vee}(\times_{i=1}^n \Omega)$ for $n \in \{1, 2, 3, ...\}$, where $L^2_{\vee}(\times_{i=1}^n \Omega)$

$$L^{2}_{\vee}(\times_{i=1}^{n}\Omega) :=$$

$$\{f^{(n)} \in L^{2}(\times_{i=1}^{n}\Omega, \otimes_{i=1}^{n}d\mathbf{x}) \mid f^{(n)}(\mathbf{x}_{1}, ..., \mathbf{x}_{n}) = f^{(n)}(\mathbf{x}_{\pi(1)}, ..., \mathbf{x}_{\pi(n)}), \forall \pi \in \mathsf{P}_{n}\}$$

$$(3.77)$$

CHAPTER 3. QUANTUM FIELD THEORY ON SPATIAL DOMAINS WITH BOUNDARIES

(here $\mathbf{x}_i \in \Omega$ for i = 1, ..., n and \mathbb{P}_n is the permutation group of n elements). For any $h \in L^2(\Omega)$, the creation and annihilation operators $\hat{a}^-(h), \hat{a}^+(h)$ defined in Eq.s (3.17) (3.18) are easily seen to map any finite-particle element $\mathbf{f} = (f^{(n)}) \in \mathfrak{D}^{\vee}(\mathcal{H})$ (with $f^{(n)} \in L^2_{\vee}(\times_{i=1}^n \Omega)$) into $\hat{a}^{\pm}(h) \mathbf{f} = ((\hat{a}^{\pm}(h) \mathbf{f})^{(n)}) \in \mathfrak{D}^{\vee}(\mathcal{H})$, where

$$(\hat{a}^{-}(h)\boldsymbol{f})^{(n)}(\mathbf{x}_{1},...,\mathbf{x}_{n}) = \sqrt{n+1} \int_{\Omega} d\mathbf{y} \,\overline{h}(\mathbf{y}) \,f^{(n+1)}(\mathbf{y},\mathbf{x}_{1},...,\mathbf{x}_{n}) , \qquad (3.78)$$

$$(\hat{a}^{+}(h)\boldsymbol{f})^{(n)}(\mathbf{x}_{1},...,\mathbf{x}_{n}) = \frac{1}{\sqrt{n}} \sum_{i=1}^{n} h(\mathbf{x}_{i}) f^{(n-1)}(\mathbf{x}_{1},...,\mathbf{x}_{i},...,\mathbf{x}_{n}) .$$
(3.79)

The Segal field and conjugate momentum are defined again according to Eq.s (3.30) (3.31) and, of course, all the results of subsection 3.1 continue to hold.

In order to proceed, let us consider the strictly positive, self-adjoint Schrödinger-type differential operator $\mathcal{A} = (-\Delta + V) \upharpoonright \mathcal{D}_{\mathcal{A}} : \mathcal{D}_{\mathcal{A}} \subset L^2(\Omega) \to L^2(\Omega)$, defined on a dense admissible domain $\mathcal{D}_{\mathcal{A}} \subset L^2(\Omega)$. Since \mathcal{A} fulfills the general hypotheses of Section 2.5, we can consider the scale of Hilbert spaces \mathcal{H}^r $(r \in [-\infty, +\infty])$ related to it. Of course, complex conjugation on $L^2(\Omega)$ commutes with \mathcal{A} (i.e., $\overline{\mathcal{A}f} = \mathcal{A}\overline{f}$, for all $f \in \mathcal{D}_{\mathcal{A}}$), so that it can be uniquely extended to a continuous antilinear involution on $\mathcal{H}^{-\infty}$ according to Prosition 2.25. Moreover, the projectors \mathcal{P}_{\pm} related to complex conjugation (see Eq. (2.68)) map any function $f \in L^2(\Omega)$ into its real and imaginary parts; more precisely, there holds $\mathcal{P}_+ f = \Re f$ and $\mathcal{P}_- f = i \Im f$. In view of this, when considering the extensions $\mathcal{P}_{\pm} : \mathcal{H}^{-\infty} \to \mathcal{H}^{-\infty}$ of these operators defined according to Eq. (2.127), with a slight abuse of notation we will write, for any $f \in \mathcal{H}^{-\infty}$,

$$\mathcal{P}_{+}f = \Re f , \qquad \mathcal{P}_{-}f = i\,\Im f . \tag{3.80}$$

The Wightman field and the conjugate momentum at time zero can be expressed according to Eq.s (3.44) (3.45), which with the present assumptions read

$$\hat{\varphi}(h) = \hat{\Phi}_S(\Re(\mathcal{A}^{-1/4}h)) + i\,\hat{\Phi}_S(\Im(\mathcal{A}^{-1/4}h)) \qquad \text{for } h \in \mathcal{H}^{-1/2} , \qquad (3.81)$$

$$\hat{\pi}(h) = \hat{\Pi}_{S}(\Re(\mathcal{A}^{1/4}h)) + i \hat{\Pi}_{S}(\Im(\mathcal{A}^{1/4}h)) \quad \text{for } h \in \mathcal{H}^{1/2} .$$
(3.82)

Time evolution is implemented again via second quantization; this allows to define the time evolved Wightman field $\hat{\varphi}_t(h)$ $(t \in \mathbb{R})$, for any $h \in \mathcal{H}^{-1/2}$, as well as its time derivatives $\partial_t^n \hat{\varphi}_t(h)$, for any $h \in \mathcal{H}^{-1/2+n}$ $(n \in \mathbb{N})$. Moreover, there holds the strong version (3.66) of the Klein-Gordon equation.

Remark 3.18. The approach to second quantization described above generalizes the corresponding formulation that we considered in Section 1.2 of Chapter 1. Let us spend a few more words about this fact; to this purpose, let us first recall that in Section 1.2 (see, in particular, Eq.s (1.13-1.15)) we gave an expansion of the field operator in terms of the creation and annihilation operators \hat{a}_k , \hat{a}_k^{\dagger} associated to a complete orthonormal set of eigenfunctions $(F_k)_{k\in\mathcal{K}}$ of \mathcal{A} , with corresponding eigenvalues $(\omega_k^2)_{k\in\mathcal{K}}$ (where $\omega_k \geq \varepsilon$, for

some $\varepsilon > 0$, and \mathcal{K} is an unspecified set of labels). In the present framework, assuming the mentioned *proper* eigenfunctions $F_k \in \mathcal{H}$ ($k \in \mathcal{K}$) to exist, it appears that the associated operators $\hat{a}_k, \hat{a}_k^{\dagger}$ are strictly related to the maps $\hat{a}^-(), \hat{a}^+() : \mathcal{H} \to \mathcal{L}(\mathfrak{D}^{\vee}(\mathcal{H}))$; more precisely, for any $k \in \mathcal{K}$, this connection is extablished setting

$$\hat{a}_k := \hat{a}^-(\overline{F}_k) \ . \tag{3.83}$$

135

Then, since \hat{a}_k^{\dagger} is the adjoint of \hat{a}_k , due to Proposition 3.5 the above definition also yields

$$\hat{a}_k^{\dagger} = (\hat{a}^-(\overline{F}_k))^{\dagger} = \hat{a}^+(\overline{F}_k) . \qquad (3.84)$$

In view of the above correspondences, the identities in Eq.s (1.14) and (1.15) are an obvious reformulation of Eq.s (3.23) and (3.20), respectively (¹³). To conclude, let us point out that the field expansion (1.13) evaluated at time zero can be derived starting from the expression (3.47) for the Wightman field in terms of the operators \hat{a}^- , \hat{a}^+ (¹⁴).

The zeta-regularized Wightman field.

The analysis to be presented in the following will be mainly focused on the Wightman field operator. Analogous results could be easily derived also for the conjugate momentum operator; however, for brevity, we shall not report them within this manuscript since they are not strictly necessary for the applications that will be discussed in the following.

In the mathematical formulation described so far, only a suitably smeared version of the Wightman field (at time $t \in \mathbb{R}$) can be properly defined, i.e., $\hat{\varphi}_t(h)$ for $h \in \mathcal{H}^{-1/2}$. Nevertheless, in many physical applications, it is of interest to *evaluate the field at a point* $\mathbf{x} \in \Omega$; this operation could be formally accounted for by considering the Wightman field evaluated on the Dirac delta element $\delta_{\mathbf{x}} \in \mathcal{H}^{-r}$ (r > d/2), setting

$$\stackrel{\text{```}}{\varphi}(x) \equiv \hat{\varphi}_t(\mathbf{x}) := \hat{\varphi}_t(\delta_{\mathbf{x}}) \stackrel{\text{'``}}{=} (x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega) .$$
(3.85)

$$\hat{\varphi}(h) = \frac{1}{\sqrt{2}} \int_{\mathcal{K}} dk \left(\langle F_k | h \rangle_{L^2} \, \hat{a}^- (\overline{\mathcal{A}^{-1/4} F_k}) + \langle \overline{F}_k | h \rangle_{L^2} \, \hat{a}^+ (\mathcal{A}^{-1/4} \overline{F}_k) \right) \, .$$

Next, notice that $\mathcal{A}^{-1/4}F_k = \omega_k^{-1/2}F_k$ and $\mathcal{A}^{-1/4}\overline{F}_k = \omega_k^{-1/2}\overline{F}_k$, so that

$$\hat{\varphi}(h) = \int_{\mathcal{K}} \frac{dk}{\sqrt{2\omega_k}} \left(\langle F_k | h \rangle_{L^2} \ \hat{a}^-(\overline{F}_k) + \langle \overline{F}_k | h \rangle_{L^2} \ \hat{a}^+(\overline{F}_k) \right) = \int_{\mathcal{K}} \frac{dk}{\sqrt{2\omega_k}} \left(\langle F_k | h \rangle_{L^2} \ \hat{a}_k + \langle \overline{F}_k | h \rangle_{L^2} \ \hat{a}_k^\dagger \right),$$

where the second identity follows from the relations in Eq.s (3.83) (3.84). The last expression written above can be formally interpreted as the smearing of the field (1.13) with the function $h \in L^2(\Omega)$.

¹³Recalling that $\langle F_k | F_h \rangle_{L^2} = \delta(k, h)$, where $\delta(k, h)$ is the Dirac delta on the label space \mathcal{K} .

¹⁴Let us account briefly for this statement. To this purpose, for any given $h \in \mathcal{H}$ consider the representation (3.47) for the Wightmand field at time zero $\hat{\varphi}(h)$ in terms of the operators $\hat{a}^{-}(\mathcal{J}(\mathcal{A}^{-1/4}h))$ and $\hat{a}^{+}(\mathcal{A}^{-1/4}h)$. Using the eigenfunction expansions $h = \int_{\mathcal{K}} dk \langle F_k | h \rangle_{L^2} F_k$ and $h = \int_{\mathcal{K}} dk \langle \overline{F}_k | h \rangle_{L^2} \overline{F}_k$ and recalling that the maps \hat{a}^{-} and \hat{a}^{+} are respectively \mathbb{C} -antilinear and \mathbb{C} -linear, by linearity we formally obtain

On the other hand, since δ_x does not belong to the proper test-function space $\mathcal{H}^{-1/2}$ (for any $d \in \{1, 2, 3, ...\}$), it appears that no rigorous meaning can be directly attributed to expressions like $\hat{\varphi}_t(\delta_{\mathbf{x}})$ (appearing in Eq. (3.85)) within the framework developed in Section 3.1.

A natural approach to give an admissible definition of point-wise evaluation for the Wightman field operator is to introduce a family of suitably regularized Dirac delta elements, depending on a parameter; at the end the regularizing parameter has to be removed via a limiting procedure, to be properly interpreted. In the spirit of ZR, we give the forthcoming definition.

Definition 3.27. Let $\kappa > 0$ be any fixed, real parameter and put (see Eq. (1.17))

$$\mathcal{A}_{\kappa} := \mathcal{A}/\kappa^2 \ . \tag{3.86}$$

For any fixed $\mathbf{x} \in \Omega$ and for any $u \in \mathbb{C}$, the *zeta-regularized Dirac delta* at \mathbf{x} is

$$\delta_{\mathbf{x}}^{u} := \mathcal{A}_{\kappa}^{-u/4} \delta_{\mathbf{x}} . \tag{3.87}$$

Remark 3.19. i) Following the considerations of Section 1.3, we will refer to κ as mass parameter; this is introduced in the definition of the zeta-regularized Dirac delta for dimensional reasons, in order to make the rescaled operator \mathcal{A}_{κ} adimensional (¹⁵). Of course, $\delta^{u}_{\mathbf{x}}$ also depends on κ ; however, we assume κ to be assigned once and for all and, in pursue of notational simplicity, we choose to not indicate explicitly the dependence on this parameter when writing the regularized Dirac delta $\delta^{u}_{\mathbf{x}}$.

ii) Other regularized versions of the Dirac delta could, of course, be considered in alternative to (3.87). For example, having in mind an exponential type regularization, one could put (for $\mathbf{t} \in \Sigma_0$)

$$\delta_{\mathbf{x}}^{\mathbf{t}} := e^{-\mathbf{t}\mathcal{A}_{\kappa}} \delta_{\mathbf{x}} \quad \text{or} \quad \delta_{\mathbf{x}}^{\mathbf{t}} := e^{-\mathbf{t}\sqrt{\mathcal{A}_{\kappa}}} \delta_{\mathbf{x}} .$$
(3.88)

These alternative definitions will not be considered within this work. We plan to discuss them and their relation to ZR elsewhere.

Lemma 3.28. i) For any fixed $j \in \mathbb{N}$, $r \in \mathbb{R}$ and for any $u \in \Sigma_{2j+d-2r}$, the map $\delta^u : \Omega \to \mathcal{H}^{-r}$, $\boldsymbol{x} \mapsto \delta^u_{\boldsymbol{x}}$ is of class C^j ; moreover, for any multi-index α of order $\leq j$, the α -th derivative of this map is given by

$$(\partial^{\alpha}\delta^{u})_{\boldsymbol{x}} = \mathcal{A}_{\kappa}^{-u/4} (\partial^{\alpha}\delta)_{\boldsymbol{x}} \in \mathcal{H}^{-r} \qquad for \ u \in \Sigma_{2|\alpha|+d-2r} .$$
(3.89)

ii) Let $\bar{}$ indicate the extension of the complex conjugation on $\mathcal{H} = L^2(\Omega)$ to $\mathcal{H}^{-\infty}$; then, for all $u \in \mathbb{C}$ and for all $x \in \Omega$, there holds

$$\overline{\delta_{\boldsymbol{x}}^{u}} = \delta_{\boldsymbol{x}}^{\overline{u}} . \tag{3.90}$$

¹⁵In fact, by standard dimensional analysis, the Schrödinger-type operator $\mathcal{A} := -\Delta + V$ can be attributed the dimension of a momentum squared; due to the convention $c = \hbar = 1$ observed in the present manuscript, this means that, dimensionally speaking, \mathcal{A} is a mass squared.

Proof. i) By definition (see Eq. (3.87)), the map $\mathbf{x} \mapsto \delta^u_{\mathbf{x}}$ is given by the composition of the maps $\mathbf{x} \mapsto \delta_{\mathbf{x}}$ and $h \mapsto \kappa^{u/2} \mathcal{A}^{-u/4} h$. Next, recall that the map $\Omega \to \mathcal{H}^{-r'}$, $\mathbf{x} \mapsto \delta_{\mathbf{x}}$ is of class C^j for all r' > j + d/2 (see Proposition 2.48). On the other hand, due to Corollary 2.13, the map $\mathcal{A}^{-u/4} : \mathcal{H}^{-(r+\Re u/2)} \to \mathcal{H}^{-r}$ is an Hilbertian isomorphism; so, in particular, it is a linear and continuous (whence analytic) map. The above facts, employed with $r' = r + \Re u/2$, suffice to infer the thesis.

ii) In order to avoid confusion, let us temporarily indicate with \mathcal{J} the extension to $\mathcal{H}^{-\infty}$ of the complex conjugation. Then, notice that $\overline{\delta_{\mathbf{x}}^{u}} = \mathcal{J}(\kappa^{u/2}\mathcal{A}^{-u/4}\delta_{\mathbf{x}}) = \kappa^{\overline{u}/2}\mathcal{J}\mathcal{A}^{-u/4}\mathcal{J}^{-1}\mathcal{J}\delta_{\mathbf{x}}$. Due to Corollary 2.26, there holds $\mathcal{J}\mathcal{A}^{-u/4}\mathcal{J}^{-1} = \mathcal{A}^{-\overline{u}/4}$; on the other hand, Lemma 2.52 allows to infer that $\mathcal{J}\delta_{\mathbf{x}} = \delta_{\mathbf{x}}$. Summing up, one has $\overline{\delta_{\mathbf{x}}^{u}} = \kappa^{\overline{u}/2}\mathcal{A}^{-\overline{u}/4}\delta_{\mathbf{x}} = \mathcal{A}_{\kappa}^{-\overline{u}/4}\delta_{\mathbf{x}} = \delta_{\mathbf{x}}^{-\overline{u}}$, which proves Eq. (3.90).

Remark 3.20. Recall the definition (2.237) of the Dirichlet kernel $\mathcal{A}^{-u/4}(,)$. Then, for all $\mathbf{x} \in \Omega$, it appears that the above relations (3.87) (3.89) can be rephrased as follows:

$$\delta_{\mathbf{x}}^{u} = \kappa^{u/2} \mathcal{A}^{-u/4}(, \mathbf{x}) \qquad \text{for } u \in \Sigma_{2d} ;$$

$$(\partial^{\alpha} \delta^{u})_{\mathbf{x}} = \kappa^{u/2} \partial_{2}^{\alpha} \mathcal{A}^{-u/4}(, \mathbf{x}) \qquad \text{for } u \in \Sigma_{2(|\alpha|+d)} .$$

(3.91)

The above relations show that, for $\Re u$ large enough, $\delta^{u}_{\mathbf{x}}$ ($\mathbf{x} \in \Omega$) coincides with the κ -rescaled Dirichlet kernel of \mathcal{A} , with one of the second argument fixed; so, for any fixed $\mathbf{x} \in \Omega$, $\delta^{u}_{\mathbf{x}}$ is in fact an ordinary (continuous and differentiable) function.

In view of Lemma 3.28, the forthcoming definition is well-posed.

Definition 3.29. The zeta-regularized Wightman field at $x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega$ (i.e., at time $t \in \mathbb{R}$ and space position $\mathbf{x} \in \Omega$) is

$$\hat{\varphi}^{u}(x) \equiv \hat{\varphi}^{u}(t, \mathbf{x}) := \hat{\varphi}_{t}(\delta^{u}_{\mathbf{x}}) \quad \text{for } u \in \Sigma_{d-1} .$$
(3.92)

Remark 3.21. Due to Lemma 3.28, $\delta_{\mathbf{x}}^{u} \in \mathcal{H}^{-1/2}$ whenever $\Re u > d-1$; this suffices to infer that the right-hand side of Eq. (3.92) makes sense according to Definition 3.13, so that $\hat{\varphi}^{u}(x)$ defined above is well posed.

Proposition 3.30. Let $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$ be any finite particle state; then, for any $n \in \mathbb{N}$ and for all $u \in \Sigma_{2n+d-1}$, the map $\mathbb{R} \times \Omega \to \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{F}^{\vee}(\mathcal{H}), x \mapsto \hat{\varphi}^u(x) \mathbf{f}$ is of class C^n . Moreover, for any (spatial) multi-index α and any $j \in \mathbb{N}$ with $j + |\alpha| \leq n$, there holds

$$\partial_t^j \partial_{\boldsymbol{x}}^{\boldsymbol{\alpha}} (\hat{\varphi}^u(t, \boldsymbol{x}) \boldsymbol{f}) = \partial_t^j \hat{\varphi}_t ((\partial^{\boldsymbol{\alpha}} \delta^u)_{\boldsymbol{x}}) \boldsymbol{f} \quad \text{for } u \in \Sigma_{2(j+|\boldsymbol{\alpha}|)+d-1} ; \quad (3.93)$$

in addition the order of the derivatives can be interchanged arbitrarily.

Proof. First of all, recall the definition (3.92) of the zeta-regularized Wightman field $\hat{\varphi}^u(x)$. In order to prove the thesis, we show hereafter in several steps that all the partial derivatives (with respect to t and **x**) of order $\leq n$ exist and are continuous on $\mathbb{R} \times \Omega$. Step 1 - Existence and continuity of the time derivatives (of any order $j \in \mathbb{N}$ with $j \leq n$). This statement follows easily from Proposition 3.21, simply noting that $\delta^u_{\mathbf{x}} \in \mathcal{H}^{-1/2+j}$ for all $u \in \Sigma_{d-1+2j}$ (see Lemma 3.28). Step 2 - Existence and continuity of the spatial derivatives (of any order $j \in \mathbb{N}$ with $j \leq n$); proof of Eq. (3.93) for j = 0. First recall that, due to Lemma 3.14 and to Proposition 3.19, the map $\mathcal{H}^{-1/2} \to \mathfrak{D}^{\vee}(\mathcal{H}) \subset \mathfrak{F}^{\vee}(\mathcal{H})$, $h \mapsto \hat{\varphi}_t(h) \mathbf{f}$ is linear and continuous, whence analytic; on the other hand, the map $\Omega \to \mathcal{H}^{-r}$, $\mathbf{x} \mapsto \delta^u_{\mathbf{x}}$ is of class C^j for all $u \in \Sigma_{d+2j-2r}$ (see Lemma 3.28). Since $\mathbf{x} \mapsto \hat{\varphi}_t(\delta^u_{\mathbf{x}}) \mathbf{f}$ is given by the composition of the previous two maps (set r = 1/2), the regularity results discussed above for the latters suffice to infer the first part of the thesis. Eq. (3.93) with j = 0 follows easily by continuity arguments, recalling that $(\partial^\alpha \delta^u)_{\mathbf{x}}$ is the α -th derivative of the map $\mathbf{x} \mapsto \delta^u_{\mathbf{x}}$ (see Lemma 3.28).

Step 3 - Existence and continuity of the mixed derivatives of the form $\partial_t^j \partial_{\mathbf{x}}^{\alpha}(\hat{\varphi}^u(t, \mathbf{x}) \mathbf{f})$ (for $j \in \mathbb{N}$ and α a multi-index with $j+|\alpha| \leq n$). Notice that, due to the previously proved identity (3.93) with j = 0, there holds $\partial_t^j \partial_{\mathbf{x}}^{\alpha}(\hat{\varphi}^u(t, \mathbf{x}) \mathbf{f}) = \partial_t^j(\hat{\varphi}_t((\partial^{\alpha} \delta^u)_{\mathbf{x}}) \mathbf{f})$. Then, one can proceed as in Step 1 of the present proof to show that the map $(t, \mathbf{x}) \mapsto \partial_t^j(\hat{\varphi}_t((\partial^{\alpha} \delta^u)_{\mathbf{x}}) \mathbf{f})$ is well-posed and continuous; indeed, it suffices to notice that the map $\Omega \to \mathcal{H}^{-1/2+j}$, $\mathbf{x} \mapsto (\partial^{\alpha} \delta^u)_{\mathbf{x}}$ is continuous for all $u \in \Sigma_{d-1+2(j+|\alpha|)}$ (see Lemma 3.28). This suffices to infer the thesis.

Step 4 - Commutativity of the partial derivatives ∂_t^j and ∂_x^{α} . As an example, consider the expression $\partial_t \partial_{\mathbf{x}}^{\alpha}(\hat{\varphi}^u(t, \mathbf{x}) \mathbf{f})$, for $|\alpha| \leq n-1$ and recall that, using Eq. (3.93) with j = 1, this can be re-written as $\partial_t(\hat{\varphi}_t((\partial^{\alpha} \delta^u)_{\mathbf{x}}) \mathbf{f})$. Next, notice that, due to Lemma 3.21 (see, in particular, the second identity in Eq. (3.64)), there holds $\partial_t(\hat{\varphi}_t((\partial^{\alpha} \delta^u)_{\mathbf{x}}) \mathbf{f}) = \hat{\pi}_t((\partial^{\alpha} \delta^u)_{\mathbf{x}}) \mathbf{f}$. Moreover, by arguments similar to those of Step 2 it can be proved that, for all multi-indexes β with $\beta \leq \alpha$ (¹⁶), $\hat{\pi}_t((\partial^{\alpha} \delta^u)_{\mathbf{x}}) \mathbf{f} = \partial_{\mathbf{x}}^{\alpha-\beta} (\hat{\pi}_t((\partial^{\beta} \delta^u)_{\mathbf{x}}) \mathbf{f})$; recalling again Eq. (3.64), this yields $\hat{\pi}_t((\partial^{\alpha} \delta^u)_{\mathbf{x}}) \mathbf{f} = \partial_{\mathbf{x}}^{\alpha-\beta} \partial_t \partial_{\mathbf{x}}^{\beta}(\hat{\varphi}_t^u(\mathbf{x}) \mathbf{f})$. Summing up, the above facts show that $\partial_t \partial_{\mathbf{x}}^{\alpha}(\hat{\varphi}^u(t, \mathbf{x}) \mathbf{f}) = \partial_{\mathbf{x}}^{\alpha-\beta} \partial_t \partial_{\mathbf{x}}^{\beta}(\hat{\varphi}^u(t, \mathbf{x}) \mathbf{f})$), for all $\beta \leq \alpha$. Finally, one can proceed by induction to prove that analogous identities hold with higher order time derivatives; this, along with Step 3, yields Eq. (3.93).

In view of the previous proposition, the following definition is well posed.

Definition 3.31. Let α be any multi-index and let $j \in \mathbb{N}$; for any $u \in \Sigma_{d-1+2(j+|\alpha|)}$ the *j*-th time, α -th spatial derivative of the zeta-regularized Wightman field at time $t \in \mathbb{R}$ and space position $\mathbf{x} \in \Omega$ is the unique operator $\partial_t^j \partial_{\mathbf{x}}^{\alpha} \hat{\varphi}^u(t, \mathbf{x}) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ (coinciding with similar expressions obtained by exchanging the differentiation order) which fulfills

$$(\partial_t^j \partial_{\mathbf{x}}^{\alpha} \hat{\varphi}^u(t, \mathbf{x})) \mathbf{f} = \partial_t^j \partial_{\mathbf{x}}^{\alpha} (\hat{\varphi}^u(t, \mathbf{x}) \mathbf{f}) \qquad \text{for all } \mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H}) .$$
(3.94)

Corollary 3.32. For any $\mathbf{f} \in \mathfrak{D}^{\vee}(\mathcal{H})$ and for all $u \in \Sigma_{d+3}$, the zeta-regularized Wightman field fulfills the following version of the Klein-Gordon equation:

$$\begin{cases} \left. \left(\partial_{tt} - \Delta_{\boldsymbol{x}} + V(\boldsymbol{x}) \right) \hat{\varphi}^{u}(t, \boldsymbol{x}) \boldsymbol{f} = \boldsymbol{o} \right. \\ \left. \hat{\varphi}^{u}(t, \boldsymbol{x}) \boldsymbol{f} \right|_{t=0} = \hat{\varphi}(\delta^{u}_{\boldsymbol{x}}) \boldsymbol{f} , \\ \left. \left(\partial_{t} \hat{\varphi}^{u}(t, \boldsymbol{x}) \boldsymbol{f} \right) \right|_{t=0} = \hat{\pi}(\delta^{u}_{\boldsymbol{x}}) \boldsymbol{f} \end{cases}$$
(3.95)

¹⁶With the notation $\beta \leq \alpha$, we mean that $\beta_i \leq \alpha_i$ for all $i \in \{1, ..., d\}$.

where $o \in \mathfrak{D}^{\vee}(\mathcal{H})$ is the null element of the Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$.

Proof. First of all, let us notice that $\delta_{\mathbf{x}}^{u} \in \text{Dom}(\mathcal{A}) \subset \mathcal{H}$ for all $u \in \Sigma_{d} \supset \Sigma_{d+3}$ (see Lemma 3.28, here employed with r = 0), so that $\mathcal{A} \delta_{\mathbf{x}}^{u} = (-\Delta_{\mathbf{x}} + V(\mathbf{x})) \delta_{\mathbf{x}}^{u}$ (¹⁷). This fact and Proposition 3.30 (see, in particular, Eq. (3.93)) imply, by linearity and continuity, that $\hat{\varphi}_{t}(\mathcal{A} \delta_{\mathbf{x}}^{u}) \mathbf{f} = (-\Delta_{\mathbf{x}} + V(\mathbf{x})) \hat{\varphi}^{u}(t, \mathbf{x}) \mathbf{f}$ (one should also recall the definition (3.92) of the zeta-regularized Wightman field $\hat{\varphi}^{u}(t, \mathbf{x})$). Then the thesis follows straightforwardly from Corollary 3.23 (see, in particular, Eq. (3.66)), to be employed here with $h = \delta_{\mathbf{x}}^{u}$ (for $u \in \Sigma_{d-1}$).

Before proceeding, let us also point out that, once the zeta-regularized Wightman field has been defined according to Eq. (3.92), one can consider the corresponding regularized VEV of the field squared at any two (possibly coincinding) spacetime points, i.e., the *zetaregularized propagator*. Concerning this quantity, the subsequent results can be proved.

Corollary 3.33. Let $\mathbf{v} \in \mathfrak{D}^{\vee}(\mathcal{H})$ be the vacuum state and let $n \in \mathbb{N}$. Then, for any $u \in \Sigma_{d-1+2n}$, the map $(\mathbb{R} \times \Omega)^2 \to \mathbb{C}$, $(x, y) \mapsto (\mathbf{v} | \hat{\varphi}^u(x) \hat{\varphi}^u(y) \mathbf{v})$ is of class C^n . Moreover, for any $j, j' \in \mathbb{N}$ and any pair of multi-indeces α, α' with $j + |\alpha|, j' + |\alpha'| \leq n$, there holds

$$\partial_{x^{0}}^{j}\partial_{y^{0}}^{j'}\partial_{\boldsymbol{x}}^{\alpha}\partial_{\boldsymbol{y}}^{\alpha'}(\boldsymbol{v}\,|\,\hat{\varphi}^{u}(x)\,\hat{\varphi}^{u}(y)\,\boldsymbol{v}) = (\boldsymbol{v}\,|\,(\partial_{x^{0}}^{j}\partial_{\boldsymbol{x}}^{\alpha}\hat{\varphi}^{u}(x))\,(\partial_{y^{0}}^{j'}\partial_{\boldsymbol{y}}^{\alpha'}\hat{\varphi}^{u}(y))\,\boldsymbol{v}) \ . \tag{3.96}$$

Proof. Due to the results Proposition 3.30, the thesis can be easily proved moving along the same lines as in the proof of Corollary 3.24.

Lemma 3.34. For any two spacetime points $x = (x^0, \boldsymbol{x}), y = (y^0, \boldsymbol{y}) \in \mathbb{R} \times \Omega$ and any $u \in \Sigma_{d-1}$, consider the VEV $(\boldsymbol{v} | \hat{\varphi}^u(x) \hat{\varphi}^u(y) \boldsymbol{v})$; this can be expressed as follows

$$\left(\boldsymbol{v}\,|\,\hat{\varphi}^{u}(x)\,\hat{\varphi}^{u}(y)\,\boldsymbol{v}\right) = \frac{\kappa^{u}}{2}\,\left\langle\delta_{\boldsymbol{x}}\,|\,\left(e^{-i(x^{0}-y^{0})\sqrt{\mathcal{A}}}\,/\mathcal{A}^{\frac{u+1}{2}}\,\right)\delta_{\boldsymbol{y}}\right\rangle\,,\tag{3.97}$$

where $\langle | \rangle$ indicates the extension to $\mathcal{H}^{(2)}$ of the inner product on \mathcal{H} .

Remark 3.22. Of course, the identity in Eq. (3.97) could be re-expressed as follows, using the language of integral kernels:

$$(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \, \hat{\varphi}^{u}(y) \, \boldsymbol{v}) = \frac{\kappa^{u}}{2} \left(e^{-i(x^{0} - y^{0})\sqrt{\mathcal{A}}} \, \mathcal{A}^{-\frac{u+1}{2}} \right)(\mathbf{x}, \mathbf{y}) \,. \tag{3.98}$$

Nonethelss, in view of the following applications, it is advisable to keep in mind the mentioned identity in the version given in the above Lemma.

¹⁷Indeed, for any $f \in \mathcal{H}^r$ with r > 2 + d/2, there holds

$$\langle \mathcal{A}\,\delta^{u}_{\mathbf{x}}|f\rangle = \langle \delta_{\mathbf{x}}|\mathcal{A}\,\mathcal{A}^{-\overline{u}/4}f\rangle = (-\bigtriangleup + V)(\mathcal{A}^{-\overline{u}/4}f)(\mathbf{x}) = (-\bigtriangleup_{\mathbf{x}} + V(\mathbf{x}))\langle \delta^{u}_{\mathbf{x}}|f\rangle \ ,$$

where, in the last passage, we have used the fact that

$$(\mathcal{A}^{-\overline{u}/4}f)(\mathbf{x}) = \langle \delta_{\mathbf{x}} | \mathcal{A}^{-\overline{u}/4}f \rangle = \langle \mathcal{A}^{-u/4}\delta_{\mathbf{x}} | f \rangle = \langle \delta_{\mathbf{x}}^{u} | f \rangle .$$

Then, the identity $\mathcal{A} \delta^{u}_{\mathbf{x}} = (- \bigtriangleup_{\mathbf{x}} + V(\mathbf{x})) \delta^{u}_{\mathbf{x}}$ follows by linearity and continuity.

Proof. First notice that the definition (3.92) and the identity (3.73) of Proposition 3.25 allow to infer $(\boldsymbol{v} | \hat{\varphi}^u(x) \hat{\varphi}^u(y) \boldsymbol{v}) = \frac{1}{2} \langle \overline{\delta_{\mathbf{x}}^u} | (e^{-i(x^0 - y^0)\sqrt{A}} / \sqrt{A}) \delta_{\mathbf{y}}^u \rangle$; on the other hand, the definition (3.87) (along with the identity (3.90)) yields

$$\langle \overline{\delta_{\mathbf{x}}^{u}} | (e^{-i(x^{0}-y^{0})\sqrt{\mathcal{A}}} / \sqrt{\mathcal{A}}) \delta_{\mathbf{y}}^{u} \rangle = \kappa^{u} \langle \mathcal{A}^{-\overline{u}/4} \delta_{\mathbf{x}} | (e^{-i(x^{0}-y^{0})\sqrt{\mathcal{A}}} / \sqrt{\mathcal{A}}) \mathcal{A}^{-u/4} \delta_{\mathbf{y}} \rangle .$$
(3.99)

To proceed, notice that for any $u \in \Sigma_{d-1}$ there exists $r, r' \in \mathbb{R}$ such that $\Re u > d - 2r$, $\Re u > d - 2r' - 2$ and -r' > r; then, due to Lemma 3.28, one has $\mathcal{A}^{-\overline{u}/4}\delta_{\mathbf{x}} \in \mathcal{H}^{-r}$ and $(e^{-i(x^0-y^0)\sqrt{\mathcal{A}}}/\sqrt{\mathcal{A}})\mathcal{A}^{-u/4}\delta_{\mathbf{y}} \in \mathcal{H}^{-r'} \hookrightarrow \mathcal{H}^r$. Therefore, the pairing $\langle | \rangle$ in the right-hand side of Eq. (3.99) can be interpreted as the extension to $\mathcal{H}^{-r} \times \mathcal{H}^r$ of the inner product on $L^2(\Omega)$ (defined according to Proposition 2.16). Then, the thesis follows recalling that the Banach adjoint of $\mathcal{A}^{-\overline{u}/2}$ is $\mathcal{A}^{-u/2}$ (see Corollary 2.24).

Remark 3.23. In the following we will mainly consider second order derivatives of the zetaregularized Wightman field. In order to adopt a more standard and concise nomenclature, we will use for such quantities the short-hand notation

$$\partial_{\mu\nu}\hat{\varphi}^{u}(x)$$
 for $u \in \Sigma_{d+3}$ and $\mu, \nu \in \{0, 1, ..., d\}$, (3.100)

where we have introduced the following conventions, for $x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega$ and $i, j \in \{1, ..., d\}$:

$$\partial_{00}\hat{\varphi}^{u}(x) := \partial_{t}^{2}\hat{\varphi}^{u}(t, \mathbf{x}) ,$$

$$\partial_{i0}\hat{\varphi}^{u}(x) \equiv \partial_{0i}\hat{\varphi}^{u}(t, \mathbf{x}) := \partial_{t}\partial_{x^{i}}\hat{\varphi}^{u}(t, \mathbf{x}) ,$$

$$\partial_{ij}\hat{\varphi}^{u}(x) \equiv \partial_{ji}\hat{\varphi}^{u}(t, \mathbf{x}) := \partial_{x^{i}x^{j}}\hat{\varphi}^{u}(t, \mathbf{x}) .$$

(3.101)

The zeta-regularized stress-energy tensor and its VEV.

We now consider one of the main observables of interest for the applications, i.e., the stress-energy tensor. By analogy with the classical expression (1.7), we consider the following definition for the corresponding quantum version; here, in order to deal with the non-commutativity of the operators involved, we choose to replace the pointwise products with the symmetrized Jordan's product

$$A \circ B := \frac{1}{2} \left(A B + B A \right) , \qquad (3.102)$$

for all linear operators $A, B : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H}).$

Definition 3.35. For any $u \in \Sigma_{d+3}$, the zeta-regularized stress-energy tensor at $x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega$ (i.e., at time $t \in \mathbb{R}$ and evaluated at the space position $\mathbf{x} \in \Omega$) is the operator $\hat{T}^{u}_{\mu\nu}(x) : \mathfrak{D}^{\vee}(\mathcal{H}) \to \mathfrak{D}^{\vee}(\mathcal{H})$ given by

$$T^{u}_{\mu\nu}(x) :=$$

$$(1 - 2\xi) \partial_{\mu}\hat{\varphi}^{u}(x) \circ \partial_{\nu}\hat{\varphi}^{u}(x)$$

$$- \left(\frac{1}{2} - 2\xi\right) \eta_{\mu\nu} \left(\partial^{\lambda}\hat{\varphi}^{u}(x) \partial_{\lambda}\hat{\varphi}^{u}(x) + V(\mathbf{x})(\hat{\varphi}^{u}(x))^{2}\right) - 2\xi \,\hat{\varphi}^{u}(x) \circ \partial_{\mu\nu}\hat{\varphi}^{u}(x) ,$$

$$(3.103)$$

where $\xi \in \mathbb{R}$ is an assigned parameter.

The zeta-regularized stress-energy VEV at x is $(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(x) \boldsymbol{v})$, where $\boldsymbol{v} \in \mathfrak{D}^{\vee}(\mathcal{H})$ indicates as usual the vacuum state of the Fock space $\mathfrak{F}^{\vee}(\mathcal{H})$.

Remark 3.24. i) This is a proposed regularization of

$$\hat{T}_{\mu\nu}(x) := (1 - 2\xi) \,\partial_{\mu}\hat{\varphi}(x) \circ \partial_{\nu}\hat{\varphi}(x) - \left(\frac{1}{2} - 2\xi\right) \eta_{\mu\nu} \left(\partial^{\lambda}\hat{\varphi}(x) \,\partial_{\lambda}\hat{\varphi}(x) + V(\mathbf{x})(\hat{\varphi}(x))^{2}\right) - 2\xi\,\hat{\varphi}(x) \circ \partial_{\mu\nu}\hat{\varphi}(x) , \qquad (3.104)$$

which is ill-defined. See, e.g., [64] for some basic information on this object (and, in particular, on the role of the parameter ξ).

ii) Let us stress that, due to Proposition 3.30 and to the related Definition 3.31 of the zeta-regularized Wightman field (see also the notations introduced in Eq. (3.101)), all the expressions on the right-hand side of Eq. (3.103) are in fact well-defined under the assumption $u \in \Sigma_{d+3}$, since there appear only derivatives of the zeta-regularized Wightman field of second order at most.

iii) The choice we made to replace the point-wise products in the classical expression with the symmetrized Jordan's product (3.102) of the corresponding operators, grants automatically the symmetry of the zeta-regularized stress-energy tensor $\hat{T}^{u}_{\mu\nu}(x)$ under exchange of the indexes $\mu, \nu \in \{0, ..., d\}$, i.e., that

$$\hat{T}^{u}_{\mu\nu}(x) = \hat{T}^{u}_{\nu\mu}(x) . \qquad (3.105)$$

The zeta-regularized stress-energy VEV can be expressed in terms of the zeta-regularized propagator $(\boldsymbol{v} | \hat{\varphi}^u(x) \hat{\varphi}^u(y) \boldsymbol{v})$ and of its space-time derivatives, evaluated along the diagonal y = x. More precisely, there holds the following result.

Lemma 3.36. For any $u \in \Sigma_{d+3}$ and for any $x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega$, the regularized VEV of the stress-energy tensor can be expressed as

$$(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(x) \boldsymbol{v}) = \left(\frac{1}{2} - \xi \right) (\partial_{x^{\mu}y^{\nu}} + \partial_{x^{\nu}y^{\mu}}) - \left(\frac{1}{2} - 2\xi \right) \eta_{\mu\nu} \left(\partial^{x^{\lambda}} \partial_{y^{\lambda}} + V(\boldsymbol{x}) \right) - \xi \left(\partial_{x^{\mu}x^{\nu}} + \partial_{y^{\mu}y^{\nu}} \right) \Big|_{y=x} (\boldsymbol{v} | \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \boldsymbol{v}) .$$

$$(3.106)$$

Proof. The thesis can be easily proved keeping in mind the definition (3.103) for the zeta-regularized stress-energy tensor operator and using the results of Corollary 3.33.

Due to the above lemma, the zeta-regularized stress-energy VEV can be explicitly computed in terms of the Dirichlet kernel and of its derivatives evaluated along the diagonal. More precisely, there holds the forthcoming proposition. **Proposition 3.37.** For any $u \in \Sigma_{d+3}$, the components of the zeta-regularized stress-energy VEV at $x = (t, \mathbf{x}) \in \mathbb{R} \times \Omega$ admit the following representations (for $i, j \in \{1, ..., d\}$):

$$(\boldsymbol{v} | \hat{T}_{00}^{u}(t, \boldsymbol{x}) \boldsymbol{v}) = \\ \kappa^{u} \left[\left(\frac{1}{4} + \xi \right) \mathcal{A}^{-\frac{u-1}{2}}(\boldsymbol{x}, \boldsymbol{y}) + \left(\frac{1}{4} - \xi \right) \left(\partial^{x^{\ell}} \partial_{y^{\ell}} + V(\boldsymbol{x}) \right) \mathcal{A}^{-\frac{u+1}{2}}(\boldsymbol{x}, \boldsymbol{y}) \right]_{\boldsymbol{y}=\boldsymbol{x}},$$

$$(3.107)$$

$$(\boldsymbol{v} | \hat{T}_{0i}^{u}(t, \boldsymbol{x}) \boldsymbol{v}) = (\boldsymbol{v} | \hat{T}_{i0}^{u}(t, \boldsymbol{x}) \boldsymbol{v}) = 0 , \qquad (3.108)$$

$$(\boldsymbol{v} | \hat{T}_{ij}^{u}(t, \boldsymbol{x}) \boldsymbol{v}) = (\boldsymbol{v} | \hat{T}_{ji}^{u}(t, \boldsymbol{x}) \boldsymbol{v}) =$$

$$\kappa^{u} \left[\left(\frac{1}{4} - \xi \right) \delta_{ij} \left(\mathcal{A}^{-\frac{u-1}{2}}(\boldsymbol{x}, \boldsymbol{y}) - \left(\partial^{x^{\ell}} \partial_{y^{\ell}} + V(\boldsymbol{x}) \right) \mathcal{A}^{-\frac{u+1}{2}}(\boldsymbol{x}, \boldsymbol{y}) \right) + \left(\left(\frac{1}{2} - \xi \right) \partial_{x^{i}} \partial_{y^{j}} - \xi \partial_{x^{i}x^{j}} \right) \mathcal{A}^{-\frac{u+1}{2}}(\boldsymbol{x}, \boldsymbol{y}) \right]_{\boldsymbol{y}=\boldsymbol{x}}.$$

$$(3.109)$$

Here the notation $[]_{y=x}$ is used to indicate that the expressions within the square brakets must be evaluated along the diagonal y = x.

Proof. First consider the expression (3.106) for the zeta-regularized stress-energy VEV $(\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu}(x) \boldsymbol{v}) \ (\mu, \nu \in \{0, ..., d\})$ in terms of the propagator; then, Eq.s (3.107-3.109) follow by simple algebraic computations if one can prove the forthcoming identities for the VEVs of the second order monomials of the zeta-regularized field, for $i, j \in \{1, ..., d\}$:

$$\left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \, \hat{\varphi}^{u}(x) \, \boldsymbol{v} \right) = \frac{\kappa^{u}}{2} \, \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{y}=\mathbf{x}} \,,$$

$$\partial_{x^{0}y^{0}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = -\partial_{x^{0}x^{0}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = -\partial_{y^{0}y^{0}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = \frac{\kappa^{u}}{2} \, \mathcal{A}^{-\frac{u-1}{2}}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{y}=\mathbf{x}} \,,$$

$$\partial_{x^{i}y^{0}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = -\partial_{x^{0}y^{i}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = \\ \partial_{y^{0}y^{i}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = -\partial_{x^{0}x^{i}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(x) \, \boldsymbol{v} \right) = \frac{i \kappa^{u}}{2} \, \partial_{x^{i}} \mathcal{A}^{-\frac{u}{2}}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{y}=\mathbf{x}} \,,$$

$$\partial_{x^{i}y^{j}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = \partial_{y^{i}y^{j}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = \frac{\kappa^{u}}{2} \, \partial_{x^{i}x^{j}} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{y}=\mathbf{x}} \,,$$

$$\partial_{x^{i}x^{j}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = \partial_{y^{i}y^{j}} \Big|_{y=x} \left(\boldsymbol{v} \mid \hat{\varphi}^{u}(x) \hat{\varphi}^{u}(y) \, \boldsymbol{v} \right) = \frac{\kappa^{u}}{2} \, \partial_{x^{i}x^{j}} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \Big|_{\mathbf{y}=\mathbf{x}} \,.$$

All the above identities can be derived by elementary manipulations, recalling the explicit expression (3.97) given in Lemma 3.34 for the zeta-regularized propagator. As an example, we show how to compute the VEV $\partial_{x^0y^i}|_{y=x}(\boldsymbol{v} \mid \hat{\varphi}^u(x) \hat{\varphi}^u(y) \boldsymbol{v})$ (the other identities can be

obtained in a similar fashion). To this purpose first notice that, using the cited lemma, one has

$$\partial_{x^0 y^i}\Big|_{y=x} (\boldsymbol{v} \,|\, \hat{\varphi}^u(x) \hat{\varphi}^u(y) \,\boldsymbol{v}) = \frac{\kappa^u}{2} \,\partial_{x^0} \partial_{y^i} \langle \delta_{\mathbf{x}} \,|\, (e^{-i(x^0 - y^0)\sqrt{\mathcal{A}}} / \mathcal{A}^{\frac{u+1}{2}}) \,\delta_{\mathbf{y}} \rangle\Big|_{y=x} ;$$

this allows to infer that $\partial_{x^0 y^i}|_{y=x}(\boldsymbol{v} | \hat{\varphi}^u(x) \hat{\varphi}^u(y) \boldsymbol{v}) = -\frac{i\kappa^u}{2} \langle \delta_{\mathbf{x}} | \mathcal{A}^{-\frac{u}{2}} (\partial_i \delta)_{\mathbf{y}} \rangle \Big|_{y=x}$, which in terms of integral kernels (see Eq. (2.221)) gives

$$\partial_{x^0 y^i}\Big|_{y=x} (\boldsymbol{v} \,|\, \hat{\varphi}^u(x) \hat{\varphi}^u(y) \,\boldsymbol{v}) = -\frac{i\,\kappa^u}{2}\,\partial_{y^i} \mathcal{A}^{-\frac{u}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}}\,.$$

In conclusion, the thesis follows noting that $\partial_{y^i} \mathcal{A}^{-\frac{u}{2}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}} = \partial_{x^i} \mathcal{A}^{-\frac{u}{2}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$, due to item i) of Proposition 2.64 (¹⁸).

Remark 3.25. Let us stress that, in principle, the zeta-regularized stress-energy VEV $(\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu}(x) \boldsymbol{v}) \equiv (\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu}(t, \mathbf{x}) \boldsymbol{v})$ does depend on both the time $(t \in \mathbb{R})$ and spatial $(\mathbf{x} \in \Omega)$ coordinates; nevertheless, the expressions obtained in the right-hand sides of Eq.s (3.107-3.109) for the components of the mentioned VEV show that the latter is, in fact, independent of the time variable t. This property was to be expected as a natural consequence of the staticity features of the setting under analysis.

In view of the considerations discussed in the above remark, in purse of brevity, we adopt the following (slightly abusive) convention.

Notation 3.38. For all $(t, \mathbf{x}) \in \mathbb{R} \times \Omega$, $\mu, \nu \in \{0, ..., d\}$ and for any $u \in \Sigma_{d+3}$, we put

$$(\boldsymbol{v} | \hat{T}^{\boldsymbol{u}}_{\mu\nu}(\mathbf{x}) \boldsymbol{v}) \equiv (\boldsymbol{v} | \hat{T}^{\boldsymbol{u}}_{\mu\nu}(t, \mathbf{x}) \boldsymbol{v}) .$$
(3.110)

Remark 3.26. The hypothesis $u \in \Sigma_{d+3}$ made in Proposition 3.37 is necessary to grant the existence of the stress-energy tensor operator $\hat{T}^{u}_{\mu\nu}(\mathbf{x})$ appearing in the left-hand sides of Eq.s (3.107-3.109). However, keeping in mind the results of Proposition 2.64, it appears that the expressions on the right-hand sides of the cited equations continue to make sense under the weaker assumption $u \in \Sigma_{d+1}$; as a matter of fact, this is also the minimum requirement necessary to ensure the continuity of the map $\mathbf{x} \mapsto (\mathbf{v} \mid \hat{T}^{u}_{\mu\nu}(\mathbf{x}) \mathbf{v})$, thus granting the well-posedness of the point-wise evaluation of the stress-energy VEV.

The considerations of Remark 3.26 can be interpreted in the sense of analytic continuation, meaning that the right-hand sides of Eq.s (3.107-3.109) give the analytic continuation to Σ_{d+1} . More precisely, there hold the results reported in the subsequent Corollary.

$$\partial_{y^i} \mathcal{A}^{-\frac{u}{2}}(\mathbf{x}, \mathbf{y}) = \partial_{y^i} \mathcal{A}^{-\frac{u}{2}}(\mathbf{y}, \mathbf{x});$$

 $^{^{18}}$ In fact, Eq. (2.240) yields, in particular,

on the other hand, when evaluation along the diagonal $\mathbf{y} = \mathbf{x}$ is being considered, one can relabel the variables in the right-hand side of the the above identity $(\mathbf{x} \leftrightarrow \mathbf{y})$, thus obtaining $\partial_{y^i} \mathcal{A}^{-\frac{u}{2}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}} = \partial_{x^i} \mathcal{A}^{-\frac{u}{2}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$.

Corollary 3.39. There hold the following statements.

i) For any $j \in \mathbb{N}$ and $u \in \Sigma_{d+1+j}$, the map $\Omega \to \mathbb{C}$, $\boldsymbol{x} \mapsto (\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\boldsymbol{x}) \boldsymbol{v})$ is of class C^{j} (whence, in particular, continuous for $u \in \Sigma_{d+1}$). ii) For any fixed $\boldsymbol{x} \in \Omega$ and for any multi-index α of order $\leq j$, the map $\Sigma_{d+1+j} \to \mathbb{C}$, $u \mapsto \partial_{\boldsymbol{x}}^{\alpha}(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\boldsymbol{x}) \boldsymbol{v})$ is analytic.

Proof. First of all, recall the expressions (3.107-3.109) derived in Proposition 3.37 for the components of the zeta-regularized stress-energy VEV $(\boldsymbol{v} | \hat{T}^u_{\mu\nu}(\mathbf{x}) \boldsymbol{v})$ in terms of the Dirichlet kernel and of its derivatives (evaluated along the diagonal $\mathbf{y} = \mathbf{x}$). Then, both statements i) and ii) can be easily infered from the results of Proposition 2.64, granting the differentiability of the Dirichlet kernel (also along the diagonal) with respect to the spatial variables and its analyticity with respect to the complex index; one should recall, as well, that V was assumed to be smooth.

Remark 3.27. When the domain and the potential fulfill the stronger regularity assumptions (2.159) and suitable conditions are prescribed on the boundary (see the comments at the end of subsection 2.6), the results of Corollary 3.39 can be extended up to the boundary. More precisely, item i) of the above Corollary holds with Ω replaced by $\overline{\Omega} = \Omega \cup \partial \Omega$, while item ii) holds for any $\mathbf{x} \in \overline{\Omega}$.

Next, let us state two results which will turn out to be crucial for the developments to be discussed in the following (in particular, to define a renormalized version of the stress-energy VEV).

Theorem 3.40. Assume the domain Ω , its boundary $\partial\Omega$ and the potential V to be such that, for any pair of multi-indexes α, β with $|\alpha| + |\beta| \leq 2$, for any $\mathbf{x} \in \Omega$ and for some $N \in \mathbb{N}$, the diagonal heat kernel derivative $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ has the form (2.354) (or even (2.357); see pages 104 and 105, respectively). Then, for any $\mu, \nu \in \{0, ..., d\}$ and for fixed $\mathbf{x} \in \Omega$, the map $\Sigma_{d+1} \to \mathbb{C}$, $u \mapsto (\mathbf{v} | \hat{T}_{\mu\nu}^u(\mathbf{x}) \mathbf{v})$ can be analytically continued to a function, which is meromorphic on the strip Σ_{d+1-2N} and possesses only possible simple pole singularities at u = d + 1 - 2n, for $n \in \{0, ..., N\}$. In particular, if N > (d + 1)/2the mentioned analytic continuation extends to a neighbour of u = 0; moreover, whenever the spatial dimension d is even, u = 0 is a regular point for the mentioned analytic continuation.

Proof. Consider once more the representations (3.107-3.109) for the components of the zeta-regularized stress-energy VEV $(\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu}(\mathbf{x}) \boldsymbol{v})$; in particular, notice that these representations involve the functions $\mathcal{A}^{-\frac{u\pm1}{2}}(\mathbf{x},\mathbf{y})$, $\partial_{w^{i}z^{j}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})$ (with $w, z \in \{x, y\}$ and $i, j \in \{1, ..., d\}$), evaluated along the diagonal $\mathbf{y} = \mathbf{x}$. Then the thesis follows easily from Theorem 2.91, whose hypotheses are assumed to be fulfilled in the present theorem. \Box

Theorem 3.41. Assume the domain Ω , its boundary $\partial\Omega$ and the potential V to be such that, for any pair of multi-indexes α, β with $|\alpha| + |\beta| \leq 2$ and for any $\mathbf{x} \in \Omega$, the diagonal cylinder kernel derivative $\partial_1^{\alpha} \partial_2^{\beta} e^{-\mathbf{t}\sqrt{\mathcal{A}}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ has the form (2.359) for some function such that the map $[0, +\infty) \ni \mathbf{t} \mapsto J^{(\alpha,\beta)}(\mathbf{t}; \mathbf{x})$ admits an analytic extension to a complex

open neighbour $\mathcal{U} \subset \mathbb{C}$ of $[0, +\infty)$ (see page 106). Then, for any $\mu, \nu \in \{0, ..., d\}$ and for fixed $\mathbf{x} \in \Omega$, the map $\Sigma_{d+1} \to \mathbb{C}$, $u \mapsto (\mathbf{v} | \hat{T}^u_{\mu\nu}(\mathbf{x}) \mathbf{v})$ can be analytically continued to a function, which is meromorphic on the whole complex plane and possesses only possible simple pole singularities at u = k, for $k \in \{0, ..., d+1\}$.

Proof. The thesis can be derived as in the proof of in Theorem 3.40, using Theorem 2.92 in place of Theorem 2.91. $\hfill \Box$

The zeta-regularized vacuum energy and pressure.

Following the analysis presented in Section 4 of our previous work [64], we are now going to consider other physical observables (aside from the stress-energy tensor), which are of direct interest for studying the Casimir effect. In particular, keeping in mind the results derived in the previous subsection for the VEV of the zeta-regularized stress-energy tensor (recalling, especially, that it is time independent), we can define zeta-regularized versions for both the vacuum energy and the vacuum pressure on the boundary $\partial\Omega$ of the spatial domain Ω . More precisely, proceeding again by analogy with the classical field theory described in subsection 1.1, we give the following definitions, which are well-posed under the specified assumptions.

Definition 3.42. Assume the zeta-regularized VEV of the energy density $(\boldsymbol{v} | T_{00}^u \boldsymbol{v})$ to fulfill the following hypothesis:

$$\exists r_1, r_2 \in \mathbb{R} \text{ with } r_1 < r_2 \text{ s.t. } \forall u \in \Sigma_{(r_1, r_2)}$$

the map $\Omega \to \mathbb{C}, \mathbf{x} \mapsto (\boldsymbol{v} \mid \hat{T}^u_{00}(\mathbf{x}) \boldsymbol{v}) \text{ belongs to } L^1(\Omega) ,$ (3.111)

(recall that $\Sigma_{(r_1,r_2)}$ indicates the strip $\{u \in \mathbb{C} \mid r_1 < \Re u < r_2\}$). Then, for all $u \in \Sigma_{(r_1,r_2)}$ the zeta-regularized vacuum total energy is (compare with Eq. (1.8))

$$\mathcal{E}^{u} := \int_{\Omega} d\mathbf{x} \left(\boldsymbol{v} \,|\, \hat{T}^{u}_{00}(\mathbf{x}) \,\boldsymbol{v} \right) \,. \tag{3.112}$$

Remark 3.28. i) In the next paragraph we will show that the assumption (3.111) holds in the case where Ω is a bounded domain with smooth boundary $\partial\Omega$, $V \in C^{\infty}(\overline{\Omega})$ and Dirichlet boundary conditions are prescribed.

ii) Recall that the map $\Omega \ni \mathbf{x} \mapsto (\boldsymbol{v} \mid \hat{T}_{00}^u(\mathbf{x}) \boldsymbol{v})$ is granted to be continuous for all $u \in \Sigma_{d+1}$, due to Corollary 3.39 (see, in particular, item i) therein); therefore, the assumption (3.111) can be checked by studying the behaviour of $(\boldsymbol{v} \mid \hat{T}_{00}^u \boldsymbol{v})$ near the boundary $\partial\Omega$ and, when Ω is unbounded, at spatial infinity.

When the boundary $\partial \Omega$ is non-empty and sufficiently regular, one can also consider the pressure exterted on it by the field confined within the spatial domain Ω .

Definition 3.43. Let $\partial \Omega$ be piecewise differentiable; consider the subset

$$\partial \Omega' := \{ \mathbf{x} \in \partial \Omega \mid \exists \mathbf{n}(\mathbf{x}) \equiv (n^i(\mathbf{x})) = \text{unit outer normal at } \mathbf{x} \}$$
(3.113)

and assume the zeta-regularized VEV of the spatial components of the stress-energy tensor $(\boldsymbol{v} | \hat{T}_{ij}^u \boldsymbol{v}) \ (i, j \in \{1, ..., d\})$ to fulfill the following hypothesis:

$$\exists r_1, r_2 \in \mathbb{R} \text{ with } r_1 < r_2 \text{ s.t. } \forall u \in \Sigma_{(r_1, r_2)}, i \in \{1, ..., d\}$$

the map $\partial \Omega' \to \mathbb{C}, \mathbf{x} \mapsto (\boldsymbol{v} \mid \hat{T}^u_{ij}(\mathbf{x}) \boldsymbol{v}) n^j(\mathbf{x}) \text{ is locally bounded }.$ (3.114)

Then, for all $u \in \Sigma_{(r_1,r_2)}$, the zeta-regularized vacuum pressure at $\mathbf{x} \in \partial \Omega'$ is the vector $\mathbf{p}^u(\mathbf{x}) \equiv (p_i^u(\mathbf{x}))$ whose components are defined as (compare with Eq. (1.9))

$$p_i^u(\mathbf{x}) := (\mathbf{v} \mid \hat{T}_{ij}^u(\mathbf{x}) \, \mathbf{v}) \, n^j(\mathbf{x}) \qquad (i \in \{1, ..., d\}) \,. \tag{3.115}$$

The case of a bounded domain.

Let us now restrict the attention to a particular type of settings; more precisely, we consider configurations fulfilling the assumptions (2.159) and (2.209):

 $\Omega \subset \mathbb{R}^d \text{ is bounded with compact boundary } \partial\Omega \text{ of class } C^{\infty}, \\ V \in C^{\infty}(\overline{\Omega}), \text{ Dirichlet boundary conditions are prescribed on } \partial\Omega.$ (3.116)

In these cases, it can be shown that the regularized vacuum energy \mathcal{E}^u and pressure \mathbf{p}^u introduced above are well-defined; as a matter of fact, much more can be said about these observables. Hereafter we report some results which we already derived in our previous work [64]; therein we employed systematically eigenfunction expansion techniques, while the arguments presented here rely on more general considerations.

Before proceeding, let us point out that some of the forthcoming results continue hold also under assumptions more general than those in Eq. (3.116).

Proposition 3.44. Assume the hypotheses (3.116) to be fulfilled; then, the map $\Sigma_{d+1} \rightarrow \mathbb{C}$, $u \mapsto \mathcal{E}^u$ is (well-defined and) analytic.

Proof. First of all, consider the representation (3.107) which allows to express the VEV $(\boldsymbol{v}|\hat{T}_{00}^{u}(\mathbf{x})\boldsymbol{v})$ in terms of the diagonal kernels $\mathcal{A}^{-\frac{u\pm1}{2}}(\mathbf{x},\mathbf{y})|_{\mathbf{y}=\mathbf{x}}, \partial^{x^{\ell}}\partial_{y^{\ell}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})|_{\mathbf{y}=\mathbf{x}}$. Recall that, due to Proposition 2.64, these kernels are all continuous functions of \mathbf{x} on $\overline{\Omega} = \Omega \cup \partial \Omega$; this suffices to infer that the map $\Omega \ni \mathbf{x} \mapsto (\boldsymbol{v}|\hat{T}_{00}^{u}(\mathbf{x})\boldsymbol{v})$ is integrable, so that \mathcal{E}^{u} is well-defined according to Eq. (3.112).

Let us now pass to show the analyticity of the map $\Sigma_{d+1} \ni u \mapsto \mathcal{E}^u$. To this purpose, it suffices to show that there exists the complex derivative $\frac{d}{du} \mathcal{E}^u$, for all $u \in \Sigma_{d+1}$; this fact follows, in turn, from Lebesgue's dominated convergence theorem if we can prove that, for any fixed $u_0 > d + 1$, there exist $\delta > 0$ and a function $\mathcal{T} \in L^1(\Omega)$ such that $|\partial_u(\boldsymbol{v}|\hat{T}^u_{00}(\mathbf{x})\boldsymbol{v})| \leq \mathcal{T}(\mathbf{x})$ for all $u \in \Sigma_{(u_0-\delta,u_0+\delta)} \cap \Sigma_{d+1}$ and for all $\mathbf{x} \in \Omega$.

So, let us proceed to prove the last statement above. Keeping in mind the representation (3.107) and recalling item v) of Proposition 2.64, it can be easily inferred that the complex derivative $\partial_u(\boldsymbol{v}|\hat{T}^u_{00}(\mathbf{x})\boldsymbol{v})$ is given by a sum of integral kernels of the form

$$\langle \delta_{\mathbf{x}} | (\partial_u \mathcal{A}^{-\frac{u\pm 1}{2}}) \delta_{\mathbf{x}} \rangle = -\frac{1}{2} \langle \delta_{\mathbf{x}} | (\mathcal{A}^{-\frac{u\pm 1}{2}} \ln \mathcal{A}) \delta_{\mathbf{x}} \rangle ,$$

$$\langle \partial_i \delta_{\mathbf{x}} | (\partial_u \mathcal{A}^{-\frac{u+1}{2}}) \partial_i \delta_{\mathbf{x}} \rangle = -\frac{1}{2} \langle \partial_i \delta_{\mathbf{x}} | (\mathcal{A}^{-\frac{u+1}{2}} \ln \mathcal{A}) \partial_i \delta_{\mathbf{x}} \rangle \qquad (i \in \{1, ..., d\})$$

To proceed, notice that all the above mentioned kernels are of the form $\langle f | (\mathcal{A}^{-\frac{u\pm 1}{2}} \ln \mathcal{A}) f \rangle$, for $f = \delta_{\mathbf{x}}$ or $\partial_i \delta_{\mathbf{x}}$; on the other hand, it can be easily proved by functional analytic methods that, for any $\delta \in (0, (u_0 - d - 1)/2)$ and for all $u \in \Sigma_{(u_0 - \delta, u_0 + \delta)}$ ($\subset \Sigma_{d+1}$), there exists a positive constant $c_{u_0,\delta}$ such that $|\langle f | (\mathcal{A}^{-\frac{u\pm 1}{2}} \ln \mathcal{A}) f \rangle| \leq 2 c_{u_0,\delta} \langle f | \mathcal{A}^{-\frac{u_0\pm 1}{2}+\delta} f \rangle$ (¹⁹). In consequence of this, there holds

$$\begin{split} |\langle \delta_{\mathbf{x}} | (\partial_{u} \mathcal{A}^{-\frac{u\pm1}{2}}) \delta_{\mathbf{x}} \rangle| &\leq c_{u_{0},\delta,\varepsilon} \left. \mathcal{A}^{-\frac{u_{0}\pm1}{2}+\delta}(\mathbf{x},\mathbf{y}) \right|_{\mathbf{y}=\mathbf{x}}, \\ |\langle \partial_{i} \delta_{\mathbf{x}} | (\partial_{u} \mathcal{A}^{-\frac{u+1}{2}}) \partial_{i} \delta_{\mathbf{x}} \rangle| &\leq c_{u_{0},\delta,\varepsilon} \left. \partial_{x^{i}y^{i}} \mathcal{A}^{-\frac{u_{0}+1}{2}+\delta}(\mathbf{x},\mathbf{y}) \right|_{\mathbf{y}=\mathbf{x}} \quad (i \in \{1,...,d\}). \end{split}$$

which allows to infer

$$\begin{aligned} |\partial_u(\boldsymbol{v} | \hat{T}^u_{00}(\mathbf{x}) \boldsymbol{v})| &\leq \\ c_{u_0,\delta,\varepsilon} \,\kappa^{\Re u} \left[\left(\frac{1}{4} + |\xi| \right) \mathcal{A}^{-\frac{u_0-1}{2} + \delta}(\mathbf{x}, \mathbf{y}) + \left(\frac{1}{4} + |\xi| \right) \left(\partial^{x^{\ell}} \partial_{y^{\ell}} + |V(\mathbf{x})| \right) \mathcal{A}^{-\frac{u_0+1}{2} + \delta}(\mathbf{x}, \mathbf{y}) \right]_{\mathbf{y} = \mathbf{x}}. \end{aligned}$$

To conclude notice that, since $u_0 \pm 1 + 2\delta \in \Sigma_{d+1\pm 1}$, due to Proposition 2.64 (keeping in mind that the potential V is assumed to be smooth on $\overline{\Omega}$), the expression on the right-hand side of the last inequality is in fact a continuous function of \mathbf{x} on $\overline{\Omega}$. Therefore, the mentioned right-hand side yields the sought-for summable dominant; this proves the thesis.

Proposition 3.45. Assume the hypotheses (3.116) to be fulfilled; then, the regularized vacuum energy \mathcal{E}^u can be expressed as the sum

$$\mathcal{E}^u = E^u + B^u , \qquad (3.117)$$

where we introduced the zeta regularized bulk and boundary vacuum energies which are, respectively,

$$E^{\boldsymbol{u}} := \frac{\kappa^{\boldsymbol{u}}}{2} \int_{\Omega} d\boldsymbol{x} \, \mathcal{A}^{-\frac{\boldsymbol{u}-1}{2}}(\boldsymbol{x}, \boldsymbol{x}) \,, \qquad (3.118)$$

$$|\langle f|(\mathcal{A}^{-\frac{u\pm 1}{2}}\ln\mathcal{A})f\rangle| = 2\int_{K} d\mu \ w^{-(\Re u\pm 1)}|\ln w| \ |f|^{2}$$

On the other hand, for all $\Re u \in (u_0 - \delta, u_0 + \delta)$, by elementary calculus one has (for $w \in [\varepsilon, +\infty)$)

$$w^{-(\Re u \pm 1)} |\ln w| \le c_{u_0,\delta} \, w^{-(u_0 \pm 1) + \delta} \qquad \text{with} \ c_{u_0,\delta} := \max\{\varepsilon^{-\delta} |\ln \varepsilon|, \, \varepsilon^{-3\delta} |\ln \varepsilon|, \, e^{-\delta}, \, e^{-3\delta}\} \ .$$

Then, the thesis follows noting that $\int_K d\mu \; w^{-(u_0 \pm 1) + 2\delta} \; |f|^2 = \langle f | \mathcal{A}^{-\frac{u_0 \pm 1}{2} + \delta} f \rangle$.

¹⁹To prove this fact consider the realization (2.54-2.56) of \mathcal{H} and \mathcal{A} , which where often employed in Chapter 2 of the present work; this allows to infer

CHAPTER 3. QUANTUM FIELD THEORY ON SPATIAL DOMAINS WITH BOUNDARIES

$$B^{u} := \kappa^{u} \left(\frac{1}{4} - \xi\right) \int_{\partial\Omega} da(\boldsymbol{x}) \, n^{i}(\boldsymbol{x}) \, \partial_{y^{i}} \mathcal{A}^{-\frac{u+1}{2}}(\boldsymbol{x}, \boldsymbol{y}) \Big|_{\boldsymbol{y}=\boldsymbol{x}} \,. \tag{3.119}$$

Remark 3.29. In particular notice that, in view of Eq. (3.118), the regularized bulk energy can be equivalently expressed as

$$E^{u} := \frac{\kappa^{u}}{2} \operatorname{Tr} \mathcal{A}^{-\frac{u-1}{2}} .$$
(3.120)

Indeed, under the assumptions (3.116), \mathcal{A} has purely discrete spectrum and explicit estimates are available for the eigenvalues; moreover, due to Corollary 2.75, $\mathcal{A}^{-\frac{u-1}{2}}$ is of trace class and $\operatorname{Tr} \mathcal{A}^{-\frac{u-1}{2}}$ is an analytic function of u for $u \in \Sigma_{d+1}$.

Recalling that \mathcal{E}^u is analytic in the very same strip Σ_{d+1} due to Proposition 3.44, this suffices to infer that also B^u is analytic therein. Moreover, due to the results of item vi) of Proposition 2.64 (see also Proposition 2.72 and the related Remark 2.25), one readily infers that the regularized boundary energy B^u defined according to Eq. (3.119) vanishes identically whenever either Dirichlet or Neumann conditions are prescribed on the boundary of the domain Ω .

Proof. Recalling the definition (3.112) and considering again the expression (3.107) for the regularized VEV $(\boldsymbol{v} | \hat{T}_{00}^{u}(\mathbf{x}) \boldsymbol{v})$, one has

$$\mathcal{E}^{u} = \kappa^{u} \int_{\Omega} d\mathbf{x} \left[\left(\frac{1}{4} + \xi \right) \mathcal{A}^{-\frac{u-1}{2}}(\mathbf{x}, \mathbf{y}) + \left(\frac{1}{4} - \xi \right) \left(\partial^{x^{\ell}} \partial_{y^{\ell}} + V(\mathbf{x}) \right) \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right]_{\mathbf{y} = \mathbf{x}}.$$

Now, notice that $[\partial^{x^{\ell}} \partial_{y^{\ell}} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y})]_{\mathbf{y}=\mathbf{x}} = \partial^{x_{\ell}} ([\partial_{y^{\ell}} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y})]_{\mathbf{y}=\mathbf{x}}) - [\triangle_{\mathbf{y}} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y})]_{\mathbf{y}=\mathbf{x}}$ (by the Leibnitz rule); therefore, by the divergence theorem it follows that

$$\int_{\Omega} d\mathbf{x} \left[(\partial^{x^{\ell}} \partial_{y^{\ell}} + V(\mathbf{x})) \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right]_{\mathbf{y}=\mathbf{x}} = \int_{\partial\Omega} da(\mathbf{x}) \ n^{\ell}(\mathbf{x}) \left[\partial_{y^{\ell}} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right]_{\mathbf{y}=\mathbf{x}} + \int_{\Omega} d\mathbf{x} \left[(-\bigtriangleup_{\mathbf{y}} + V(\mathbf{y})) \ \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right]_{\mathbf{y}=\mathbf{x}} .$$

In view of this, the thesis follows by simple algebraic computations if we can show that $(-\bigtriangleup_{\mathbf{y}}+V(\mathbf{y}))\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y}) = \mathcal{A}^{-\frac{u-1}{2}}(\mathbf{x},\mathbf{y})$. Indeed, to prove this fact it suffices to notice that $(-\bigtriangleup_{\mathbf{y}}+V(\mathbf{y}))\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y}) = \langle \delta_{\mathbf{x}}|\mathcal{A}^{-\frac{u+1}{2}}(-\bigtriangleup_{\mathbf{y}}+V(\mathbf{y}))\delta_{\mathbf{y}}\rangle$ by continuity (see Eq. (2.221)), and to recall that $\bigtriangleup_{\mathbf{y}}\delta_{\mathbf{y}} = \bigtriangleup\delta_{\mathbf{y}}$ (compare with Eq. (2.196)), so that $(-\bigtriangleup_{\mathbf{y}}+V(\mathbf{y}))\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y}) = \langle \delta_{\mathbf{x}}|\mathcal{A}^{-\frac{u+1}{2}}\mathcal{A}\delta_{\mathbf{y}}\rangle$, which yields the thesis.

Proposition 3.46. Assume the hypotheses (3.116) to be fulfilled; then, for any boundary point $\boldsymbol{x} \in \partial \Omega$, the map $\Sigma_{d+1} \to \mathbb{C}$, $u \mapsto \boldsymbol{p}^u(\boldsymbol{x})$ is (well-defined and) analytic.

Proof. Consider the expression (3.109) giving $(\boldsymbol{v}|\hat{T}_{ij}^{u}\boldsymbol{v})$ in terms of the Dirichlet kernel and of its derivatives evaluated along the diagonal. By a simple generalization of item i) of Proposition 2.64 it follows that $\mathcal{A}^{-\frac{u\pm1}{2}}(\mathbf{x},\mathbf{y})|_{\mathbf{y}=\mathbf{x}}, \partial_{zw}\mathcal{A}^{\frac{u+1}{2}}(\mathbf{x},\mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ are continuous functions of \mathbf{x} up to the boundary; moreover, for any fixed $\mathbf{x} \in \partial\Omega$ the mentioned kernels are analytic functions of u, for $u \in \Sigma_{d+1}$, and the unit outer normal $\mathbf{n}(\mathbf{x})$ exists. This suffices to infer the thesis.

Finally, let us state two results which will turn out to be crucial for the developments to be discussed in the following (in particular, to define a renormalized version of the bulk energy).

Theorem 3.47. Assume the heat trace to fulfill the assumptions of Theorem 2.94 (see, in particular, Eq. (2.362)) for some $N \in \mathbb{N}$. Then, the map $\Sigma_{d+1} \to \mathbb{C}$, $u \mapsto E^u$ can be analytically continued to a function which is meromorphic on the strip Σ_{d+1-2N} and possesses only possible simple pole singularities at u = d + 1 - n, for $n \in \{0, ..., 2N\}$. In particular, if N > (d+1)/2 the mentioned analytic continuation extends to a neighbour of u = 0.

Proof. Consider the representation (3.120) for the zeta-regularized bulk energy E^u , in terms of the trace $\text{Tr}\mathcal{A}^{-\frac{u-1}{2}}$. Then the thesis follows easily from Theorem 2.94.

Theorem 3.48. Assume the cylinder trace to fulfill the assumptions of Theorem 2.96 (see, in particular, Eq. (2.367)). Then, the map $\Sigma_{d+1} \to \mathbb{C}$, $u \mapsto E^u$ can be analytically continued to a function which is meromorphic on the whole complex plane and possesses only possible simple pole singularities at at u = k, for $k \in \{0, ..., d+1\}$.

Proof. Consider once more the representation (3.120) for the zeta-regularized bulk energy E^u , in terms of the trace $\text{Tr}\mathcal{A}^{-\frac{u-1}{2}}$. Then the thesis follows by arguments similar to those employed in the proof of Theorem 3.40, using Theorem 2.96 in place of Theorem 2.94. \Box

Renormalized observables.

Let us consider the zeta-regularized versions of the stress-energy VEV $(\boldsymbol{v}|\hat{T}_{\mu\nu}^{u}(\mathbf{x})\boldsymbol{v})$ $(\mu,\nu \in \{0,...,d\}, \mathbf{x} \in \Omega)$, of the vacuum bulk and boundary energies E^{u}, B^{u} , and of the vacuum pressure $\mathbf{p}^{u}(\mathbf{x})$ $(\mathbf{x} \in \partial \Omega)$ introduced in the previous subsection. In view of the results derived therein for these observables (see, in particular, Corollary 3.39 and Propositions 3.44, 3.46), it appears that the mentioned observables are typically analytic functions of u, for $\Re u$ sufficiently large (i.e., for $u \in \Sigma_{d+1}$); moreover, they can be analytically continued to wider regions of the complex plane, possibly including a neighbour of the origin, where they are meromorphic and possess only simple pole singularities.

Therefore, in general, we can proceed to define the renormalized versions of the above mentioned zeta-regularized observables following the general ("extended") approach presented in [64]. To this purpose, let us indicate with F(u) any one of $(\boldsymbol{v}|\hat{T}^{u}_{\mu\nu}(\mathbf{x})\boldsymbol{v}), E^{u}, B^{u},$ $\mathbf{p}^{u}(\mathbf{x})$ and state the following definition.

Definition 3.49. (Zeta approach to renormalization). Assume there exist $r_1, r_2 \in \mathbb{R}$ with $r_1 < r_2$ such that the map $\Sigma_{(r_1,r_2)} \to \mathbb{C}$, $u \mapsto F(u)$ is analytic; moreover, assume this map to posses an analytic continuation to a function (indicated again with F) which is meromorphic on the larger strip $\Sigma_{(r_-,r_+)} \supset \Sigma_{(r_1,r_2)}$ including the origin $(0 \in \Sigma_{(r_-,r_+)})$. Then, the renormalized value of F is

$$F_{ren} := RP \Big|_{u=0} F(u) , \qquad (3.121)$$

where $RP|_{u=0}$ indicates the evaluation in u = 0 of the regular part of the Laurent expansion centered at the same point.

Remark 3.30. i) In order to avoid misunderstandings, let us mention that, for any meromorphic function F with a pole of order N ($N \in \mathbb{N}$) at u = 0, the Laurent expansion centered at the same point (u = 0) is

$$F(u) = \sum_{k=-N}^{+\infty} F_k u^k ; \qquad (3.122)$$

its regular part is given by $(RP F)(u) := \sum_{k=0}^{+\infty} F_k u^k$, so that

$$RP\Big|_{u=0}F(u) = F_0$$

ii) Let us stress that the prescription (3.121) is a quite straightforward generalization of the approach considered, e.g., in [22, 56], where F was assumed to posses a simple pole in u = 0 (i.e., it was assumed that N = 1). On the other hand it is apparent that, whenever F is analytic at u = 0, the prescription (3.121) reduces to

$$F_{ren} := F(0) , \qquad (3.123)$$

which was referred to as "restricted" zeta approach in our previous work [64].

For future reference, let us specify the definition (3.121) for the cases of main interest in the applications.

Definition 3.50. Assume the regularized expressions $(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\mathbf{x}) \boldsymbol{v}) (\mathbf{x} \in \Omega), E^{u}, B^{u}$ and $\mathbf{p}^{u}(\mathbf{x}) (\mathbf{x} \in \partial \Omega)$ to fulfill the assumptions considered in Definition 3.49. Then, we consider the corresponding renormalized versions reported hereafter.

i) For any $\mathbf{x} \in \Omega$ and $\mu, \nu \in \{0, ..., d\}$, the renormalized stress-energy VEV is

$$(\boldsymbol{v} \mid \hat{T}_{\mu\nu}(\mathbf{x}) \, \boldsymbol{v})_{ren} := RP \Big|_{\boldsymbol{u}=0} \left(\boldsymbol{v} \mid \hat{T}^{\boldsymbol{u}}_{\mu\nu}(\mathbf{x}) \, \boldsymbol{v} \right) \,. \tag{3.124}$$

ii) The renormalized bulk and boundary energies are, respectively,

$$E^{ren} := RP\Big|_{u=0} E^u$$
, $B^{ren} := RP\Big|_{u=0} B^u$. (3.125)

iii) The renormalized pressure, at any point $\mathbf{x} \in \partial \Omega$ where the outer normal $\mathbf{n}(\mathbf{x})$ is well defined, is

$$\mathbf{p}^{ren}(\mathbf{x}) := RP \Big|_{u=0} \mathbf{p}^u(\mathbf{x}) .$$
(3.126)

Remark 3.31. Let us stress that, in particular, whenever the diagonal heat or cylinder kernel derivatives fulfill the hypotheses of Theorem 2.91 (with N > (d+1)/2) or of Theorem 2.93, respectively, then Eq. (3.124) is granted to give a finite value for $(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{x}) \boldsymbol{v})_{ren}$ in consequence of Theorems 3.40 and 3.41, respectively. On the other hand, Theorems 3.47 and 3.48 show that the renormalized bulk energy E^{ren} is also finite under suitable hypothesis.

Conformal and non-conformal parts of the stress-energy tensor.

In the literature [21, 119, 153] it is customary to write the stress-energy tensor (here to be intended as the zeta-regularized operator $\hat{T}^{u}_{\mu\nu}$, its VEV ($\boldsymbol{v} | \hat{T}^{u}_{\mu\nu} \boldsymbol{v}$) or the renormalized version ($\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v}$)_{ren}) as the sum of a conformal and a non-conformal part. In order to define these quantities, let us consider for ξ the critical value

$$\xi_d := \frac{d-1}{4d} \ . \tag{3.127}$$

Remark 3.32. As well-known, when coupling of the scalar field to gravity is taken into account and no external potential is present (for V = 0), the theory is invariant under conformal transformations of the spacetime line element if ξ has the critical value (3.127) (see, e.g., [153], page 447).

In the sequel we adopt systematically the notations

$$\Diamond \equiv \text{conformal} , \qquad \blacksquare \equiv \text{non-conformal} ; \qquad (3.128)$$

in particular, in accordance with them, we give the following definition.

Definition 3.51. Consider the renormalized stress-energy VEV $(\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$; its conformal and non-conformal parts are, respectively,

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}^{(\Diamond)} \boldsymbol{v})_{ren} := (\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren} \Big|_{\boldsymbol{\xi} = \boldsymbol{\xi}_d} , \qquad (3.129)$$

$$(\boldsymbol{v} \mid \hat{T}_{\mu\nu}^{(\blacksquare)} \boldsymbol{v})_{ren} := \frac{1}{\xi - \xi_d} \left((\boldsymbol{v} \mid \hat{T}_{\mu\nu} \boldsymbol{v})_{ren} - (\boldsymbol{v} \mid \hat{T}_{\mu\nu}^{(\diamondsuit)} \boldsymbol{v})_{ren} \right) .$$
(3.130)

Remark 3.33. i) Of course, in view of the above definitions, there holds

$$(\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren} = (\boldsymbol{v} | \hat{T}_{\mu\nu}^{(\Diamond)} \boldsymbol{v})_{ren} + (\xi - \xi_d) (\boldsymbol{v} | \hat{T}_{\mu\nu}^{(\blacksquare)} \boldsymbol{v})_{ren} .$$
(3.131)

ii) In the applications to be considered in the forthcoming Chapter 4, when presenting our final results for the renormalized stress-energy VEV, we will either write them in the form (3.131) or give separately the conformal and non-conformal parts (3.129) (3.130).

Anomalies.

As we pointed out in our previous work [64], Eq.s (3.125) and (3.126) are not the only reasonable ways to define renormalized versions of the bulk/boundary energies and of the boundary pressure, respectively. For example, starting with the renormalized stressenergy VEV $(\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$, one could consider the quite natural alternative definitions described hereafter.

On the one hand, one could define the *renormalized total energy* as

$$\mathcal{E}^{ren} := \int_{\Omega} d\mathbf{x} \; (\boldsymbol{v} \,|\, \hat{T}_{00}(\mathbf{x}) \,\boldsymbol{v})_{ren} \tag{3.132}$$

(in few words: in the approach (3.112)-(3.125), one integrates over the domain Ω the regularized energy density and next renormalizes; in the approach (3.132), one first renormalizes the energy density and then integrates it over Ω).

On the other hand, the *renormalized pressure* $\mathbf{p}^{ren}(\mathbf{x}) \equiv (p_i^{ren}(\mathbf{x}))$ at any boundary point $\mathbf{x} \in \partial \Omega'$ (see Eq. (3.113)) with outer unit normal $\mathbf{n}(\mathbf{x})$ could be defined as

$$p_i^{ren}(\mathbf{x}) := \left(\lim_{\mathbf{x}' \in \Omega, \, \mathbf{x}' \to \mathbf{x}} (\boldsymbol{v} \,|\, \hat{T}_{ij}(\mathbf{x}') \,\boldsymbol{v})_{ren} \right) n^j(\mathbf{x}) \tag{3.133}$$

(in few words: in the approach (3.115)-(3.126), one stays at a point on the boundary, and performs therein the renormalization; in the approach (3.133), one renormalizes at points inside Ω , and then moves towards the boundary $(^{20})$).

Let us recall a fact that we already stressed in [64]: namely, that the possibilities (3.132) and (3.133) are not granted a priori to be equivalent, respectively, to the definitions (3.125) and (3.126) considered in the previous subsection. For example, it may happen that the integral in the right-hand side of Eq. (3.132) is divergent, while the prescription (3.125) always gives a finite result by construction (assuming \mathcal{E}^u can be analytically continued in a neighbour of u = 0). Similarly, the renormalized VEV $(\boldsymbol{v} | \hat{T}_{ij}(\mathbf{x}') \boldsymbol{v})_{ren}$ $(\mathbf{x}' \in \Omega)$ might have no finite limit for $\mathbf{x}' \to \mathbf{x} \in \partial\Omega$, thus making ill-defined the prescription (3.133) for the renormalized pressure.

As a matter of fact, in our previous series of works [64, 65, 66, 67] we showed with some explicit examples that, in general, the prescriptions (3.132) and (3.133) give infinite results for \mathcal{E}^{ren} and \mathbf{p}^{ren} due to the singular behaviours of the renormalized stress-energy VEV near the boundary. In consequence of this, there arise unavoidable ambiguities, or *anomalies*, when talking about these renormalized observables. On the other hand, let us stress that the above mentioned boundary singularities of the renormalized stress-energy VEV are not a specific consequence of the ZR approach which we are considering in the present manuscript; indeed, they also appear if one uses a point-splitting approach, as indicated by the very systematic analysis of Deutsch and Candelas [49].

For the moment, these anomalies must be accepted as a problematic aspect which is common to all the main regularization schemes; what we can do is just to record them when they appear, and hope that in the future they can be better understood. Most probably, their origin should be looked for in some excessive idealization of the physical model. For example, one could try to describe in a more realistic manner the boundaries of the spatial domain; these are "hard" and "deterministic" in the present formulation, but could perhaps be replaced with "soft" or "stochastic" boundaries, following an idea which was first proposed by Ford and Svaiter [70].

To conclude, let us also report a supposition we made in [64]. Therein, motivated by some explicit results derived in our series of papers [64, 65, 66, 67], we advanced the following

²⁰Notice that both the mentioned approaches require the existence of the normal $\mathbf{n}(\mathbf{x})$; therefore, they both lose meaning on the eventual edges and corner points of $\partial\Omega$.

conjecture about the pressure anomaly discussed before:

"at points $\mathbf{x} \in \partial \Omega$ where the normal is well defined and the approach (3.133) gives a finite pressure, the result obtained according to the latter prescription (3.134) agrees with the renormalized pressure defined by Eq. (3.126)."

Chapter 4

Some explicitly solvable models

In this conclusive chapter we describe some applications of the abstract framework developed previously in this manuscript; in particular, we analyze some models which can be treated exactly, i.e., without resorting to any perturbative method.

First of all, we briefly report some results obtained in our series of papers [64, 65, 66, 67]: more precisely, we consider the case of a scalar field interacting with a background harmonic potential and that of a field confined within a rectangular box. Next, in Section 4.3 we consider the configuration involving two parallel planes; this model was analyzed in [65] for Dirichlet, Neumann and periodic boundary conditions (we make reference to some of the results obtained for these configurations). In the present work we describe some novel results obtained for a particular subcase which, to the best of our knowledge, was never treated previously by means of zeta techniques: we assume Robin boundary conditions are prescribed on each one of the planes and we compute all the components of the renormalized stress-energy VEV. Our main motivation in the analysis of this configuration is to study the changes arising in the passage from Dirichlet to Neumann boundary conditions in a case which can be solved explicitly; in particular, we are interested in the analysis of the boundary behaviour of the renormalized stress-energy VEV which, as mentioned in subsection 3.2 and exemplified in the forthcoming Sections 4.1 and 4.2, can cause the appearence of anomalies within the physical framework under analysis.

Before moving on, let us stress the following facts.

Remark 4.1. In all the settings to be studied in the present chapter, for simplicitly, we restrict attention to the case of a massless field; nevertheless, most of the results to be reported in the sequel could be generalized to the case of a massive field, with some additional computational effort. Furthermore, the computational methods we are going to present can be employed for an arbitrary spatial dimension $d \in \{1, 2, 3, ...\}$. As examples, at the end of each section we report the explicit results which can be obtained performing the numerical computations required by the previously mentioned general methods for some specific choice of d; we typically fix d = 3 in these computations, for clear physical interest (except for Section 4.2, where we choose d = 2 for simplicity of exposition).

4.1 The case with a background harmonic potential.

The configuration under investigation in the present section is that of a scalar quantum field interacting with a background harmonic potential, in arbitrary spatial dimension $d \in \{1, 2, 3, ...\}$. Hereafter, we report some of the results which were derived in our previous work [66], to which we refer for a more detailed analysis; following the cited work, for simplicity, we restrict the attention to the case where the field is *massless* and the background potential is *isotropic*. Nevertheless, our approach could be extended with some computational effort to anisotropic harmonic potentials and to the case of a massive scalar field, as well.

The main result of our analysis is the computation of the renormalized stress-energy VEV, obtained by applying the general framework described in Chapters 2 and 3 (see also [64]) to the present configuration; we also consider the total energy, referring to both bulk and boundary contributions. For all the mentioned observables we derive, by analytic methods, fully explicit integral representations; ultimately, these representation must be computed numerically. However, they can also be used to derive asymptotic expansions (complete with remainder estimates) for the stress-energy VEV components when the radius goes to zero or to infinity.

In subjction 4.1, we report the results obtained using the framework mentioned above and performing the required explicit computations for d = 3, with the aid of Mathematica; in particular, we restrict the attention to the renormalized energy density $(\boldsymbol{v} | \hat{T}_{00} \boldsymbol{v})_{ren}$ and to the renormalized bulk energy E^{ren} .

Before proceeding, let us mention that the idea to replace sharp boundaries with suitable background potentials is well-known in the literature on the Casimir effect. Typically (see, e.g., [15, 23, 81, 105, 120]), delta-like potentials are introduced in order to mimic boundary conditions in a "physically more realistic" framework; the ultimate purpose is to obtain less singular behaviours of the renormalized quantities, avoiding, e.g., boundary divergences such as the ones mentioned in subsection 3.2 (see also [64, 65]). The case of a scalar field interacting with an external harmonic potential has been formerly considered by Actor and Bender [2, 5], who have determined the renormalized VEV of the total (bulk) energy via global zeta regularization (using ad hoc results on the analytic continuations for the special functions involved in this specific case) (¹); as far as we know, the stress-energy tensor has not been previously computed for the present configuration.

Introducing the problem.

As anticipated previously, we consider the case of a massless scalar field on \mathbb{R}^d in presence of a classical isotropic harmonic potential. More precisely, we assume

$$\Omega := \mathbb{R}^d, \qquad V(\mathbf{x}) := \lambda^4 \, |\mathbf{x}|^2 \,, \tag{4.1}$$

¹In [66] we point out that our results for the renormalized bulk energy agree with the ones of Actor and Bender [5] for $d \in \{1, 2, 3\}$.

where $\lambda > 0$ and $|\mathbf{x}| := \sqrt{(x^1)^2 + ... + (x^d)^2}$; the constant λ is, dimensionally, a mass (or an inverse length) like the parameter κ employed to define the zeta-regularized Wightman field (see Eq. (3.92) and the related comments). In this case, we put

$$\mathcal{A} := -\bigtriangleup + V ; \tag{4.2}$$

this is an *admissible operator* in the sense of definition 2.37 (see page 55). In fact, as well known, \mathcal{A} is strictly positive and self-adjoint on the single particle Hilbert space $\mathcal{H} = L^2(\mathbb{R}^d)$, with admissible domain $\mathcal{D}_{\mathcal{A}} := \{f \in L^2(\mathbb{R}^d) | (-\Delta + V)f \in L^2(\mathbb{R}^d) \}$. Moreover, \mathcal{A} has purely discrete spectrum, given by

$$\sigma(\mathcal{A}) = \left\{ \lambda^2 \left(2(n_1 + \dots + n_d) + d \right) \mid n_i \in \mathbb{N}, \ i = 1, \dots, d \right\}$$
(4.3)

 $(^{2})$. In view of the above considerations, we can employ the functional analytic framework developed in Chapter 2 to analyze the present setting; in particular, we can consider the Dirichlet and heat kernels associated to the admissible operator \mathcal{A} , as well as the related traces.

The heat kernel and the heat trace.

Notice that the configuration described in Eq. (4.1) is of product type. Therefore, the heat kernel $e^{-\mathbf{t}\mathcal{A}}(,)$ (for $\mathbf{t} \in \Sigma_0$) related to the admissible operator $\mathcal{A} = -\Delta + V$ (with $V(\mathbf{x}) = \lambda^4 |\mathbf{x}|^2$, for $\mathbf{x} \in \mathbb{R}^d$) can be easily determined, starting from the one-dimensional Mehler kernel [108] (see also [20, 33, 46, 82]); for any $\mathbf{x}, \mathbf{y} \in \mathbb{R}^d$, the final result is [66]

$$e^{-\mathbf{t}\mathcal{A}}(\mathbf{x},\mathbf{y}) = \left(\frac{\lambda}{\sqrt{2\pi\sinh(2\lambda^2\mathbf{t})}}\right)^d \exp\left[-\lambda^2 \left(\frac{|\mathbf{x}|^2 + |\mathbf{y}|^2}{2\tanh(2\lambda^2\mathbf{t})} - \frac{\mathbf{x}\cdot\mathbf{y}}{\sinh(2\lambda^2\mathbf{t})}\right)\right]$$
(4.4)

(where $\mathbf{x} \cdot \mathbf{y} := \sum_{i=1}^{d} x^{i} y^{i}$). Moreover, by a slight variation of Proposition 2.74 and of the related Corollary 2.76, it appears that the exponential operator $e^{-\mathbf{t}\mathcal{A}}$ is of trace class for all $\mathbf{t} \in \Sigma_{0}$; so, the heat trace $\operatorname{Tr} e^{-\mathbf{t}\mathcal{A}}$ exists and is finite. It can be proved by simple computations (³) that

$$\operatorname{Tr} e^{-\mathbf{t}\mathcal{A}} = \left(\frac{1}{2\sinh(k^2\mathbf{t})}\right)^d .$$
(4.5)

²Note that the parameter $\varepsilon > 0$, which is generally chosen so as to fulfill $\sigma(\mathcal{A}) \subset [\varepsilon, +\infty)$, can be fixed explicitly in the case under analysis, setting $\varepsilon := \lambda^2 d$.

$$\operatorname{Tr} e^{-\mathbf{t}\mathcal{A}} = \sum_{n_1, \dots, n_d \in \mathbb{N}} e^{-\mathbf{t}\lambda^2 (2(n_1 + \dots + n_d) + d)} = e^{-\mathbf{t}\lambda^2 d} \left(\sum_{n=0}^{+\infty} e^{-\mathbf{t}\lambda^2 n}\right)^d = \frac{e^{-\mathbf{t}\lambda^2 d}}{(1 - e^{-\mathbf{t}\lambda^2})^d} = \left(\frac{1}{2\sinh(k^2\mathbf{t})}\right)^d.$$

For an alternative derivation, using the explicit expression (4.4) of the heat kernel, see [66], page 8, footnote 4.

³Eq. (4.5) can be easily derived as follows, recalling that the spectrum of \mathcal{A} is given by (4.3):

Rescaled spherical coordinates.

Next, notice that the configuration (4.1) is patently spherically symmetric; therefore, it is natural to pass to a set of curvilinear coordinates which best fit the symmetries of the problem. To this purpose, we introduce the spherical " λ -rescaled" coordinates

$$\mathbf{x} \mapsto \mathbf{q}(\mathbf{x}) \equiv (r(\mathbf{x}), \theta_1(\mathbf{x}), \dots, \theta_{d-2}(\mathbf{x}), \theta_{d-1}(\mathbf{x})) \in (0, +\infty) \times (0, \pi) \times \dots \times (0, \pi) \times (0, 2\pi)$$
(4.6)

whose inverse map $\mathbf{q} \mapsto \mathbf{x}(\mathbf{q})$ is described by the equations

$$\lambda x^{1} = r \cos(\theta_{1}) ,$$

$$\lambda x^{2} = r \sin(\theta_{1}) \cos(\theta_{2}) ,$$

$$\vdots \qquad (4.7)$$

$$\lambda x^{d-1} = r \sin(\theta_{1}) \dots \sin(\theta_{d-2}) \cos(\theta_{d-1}) ,$$

$$\lambda x^{d} = r \sin(\theta_{1}) \dots \sin(\theta_{d-1}) .$$

Let us stress that, for any spatial dimension d, there holds

$$r = \lambda \left| \mathbf{x} \right| \,; \tag{4.8}$$

thus, the coordinate r is an adimensional radius.

In order to avoid cumbersome notations, given a function $\mathbb{R}^d \to Y$, $\mathbf{x} \mapsto f(\mathbf{x})$ (with Y any set), we indicate the composition $\mathbf{q} \mapsto f(\mathbf{x}(\mathbf{q}))$ as $\mathbf{q} \mapsto f(\mathbf{q})$; we will use similar conventions for functions on $\mathbb{R}^d \times \mathbb{R}^d$.

Of course, the curvilinear coordinates \mathbf{q} induce a set of spacetime coordinates $q \equiv (q^{\mu}) \equiv (t, \mathbf{q})$. The spatial and space-time line elements are, respectively,

$$d\ell^{2} = a_{ij}(\mathbf{q})dq^{i}dq^{j} ; \qquad ds^{2} = -dt^{2} + d\ell^{2} = g_{\mu\nu}(q) dq^{\mu}dq^{\nu} ,$$

$$g_{00} := -1 , \qquad g_{i0} = g_{0i} := 0 , \qquad g_{ij}(q) := a_{ij}(\mathbf{q}) \qquad \text{for } i, j \in \{r, \theta_{1}, ..., \theta_{d-1}\} ,$$
(4.9)

where a_{ij} indicates the flat metric on \mathbb{R}^d expressed in the curvilinear coordinates **q**. Most results of the previous Chapters 1-3 are readily rephrased in the present framework; in particular, the analogue of Eq. (3.103) in the coordinate system (q^{μ}) is

$$\hat{T}^{u}_{\mu\nu} := (1 - 2\xi)\partial_{\mu}\hat{\varphi}^{u} \circ \partial_{\nu}\hat{\varphi}^{u} - \left(\frac{1}{2} - 2\xi\right)\eta_{\mu\nu}\left(\partial^{\rho}\hat{\varphi}^{u}\partial_{\rho}\hat{\varphi}^{2} + V(\hat{\varphi}^{u})^{2}\right) - 2\xi\,\hat{\varphi}^{u} \circ \nabla_{\mu\nu}\hat{\varphi}^{u} , \quad (4.10)$$

where ∇_{μ} denotes the covariant derivative induced by the metric (4.9). Notice that, indicating with γ_{ij}^k the Christoffel symbols for the spatial metric $(a_{ij}(\mathbf{q}))$ and with D_i the corresponding covariant derivative, for any scalar function f there hold $(i, j \in \{r, \theta_1, ..., \theta_{d-1}\})$

$$\nabla_{\mu}f = \partial_{\mu}f , \qquad \nabla_{ij}f = D_{ij}f = \partial_{ij}f - \gamma_{ij}^{\kappa}\partial_{\kappa}f ,
\nabla_{0i}f = \partial_{0}(\partial_{i}f) = \partial_{i}(\partial_{0}f) = \nabla_{i0}f , \qquad \nabla_{00}f = \partial_{00}f .$$
(4.11)

Before proceeding, let us also point out that, in view of the explicit expressions (4.4) and (4.5) for the heat kernel and trace, respectively, it is natural to introduce the rescaled parameter

$$\tau := \lambda^2 \mathbf{t} \in (0, +\infty) . \tag{4.12}$$

In the sequel, for any pair

$$\mathbf{q} = (r, \theta_1, ..., \theta_{d-1}) \equiv (r, \theta) , \qquad \mathbf{p} = (r', \theta'_1, ..., \theta'_{d-1}) \equiv (r', \theta') , \qquad (4.13)$$

we write $\mathcal{A}^{-s}(\boldsymbol{q}, \boldsymbol{p})$ and $e^{-\tau \mathcal{A}}(\boldsymbol{q}, \boldsymbol{p})$, respectively, for the Dirichlet and heat kernels at two points $\boldsymbol{x}, \boldsymbol{y}$ of (rescaled) spherical coordinates $\boldsymbol{q}, \boldsymbol{p}$, and with τ related to \mathbf{t} by Eq. (4.12). In particular, Eq. (4.4) implies

$$e^{-\tau \mathcal{A}}(\mathbf{q}, \mathbf{p}) = \left(\frac{\lambda}{\sqrt{2\pi \sinh(2\tau)}}\right)^d \exp\left[-\left(\frac{r^2 + r'^2}{2\tanh(2\tau)} - \frac{r r' S(\boldsymbol{\theta}) S(\boldsymbol{\theta}')}{\sinh(2\tau)}\right)\right]$$
(4.14)

where $S(\boldsymbol{\theta})$ and $S(\boldsymbol{\theta}')$ are the products of cosines and sines of the angular coordinates $(\theta_1, ..., \theta_{d_1})$ and $(\theta'_1, ..., \theta'_{d_1})$ of Eq. (4.13), corresponding to the scalar product $\mathbf{x} \cdot \mathbf{y}$.

The zeta-regularized stress-energy VEV.

Hereafter we are going to derive integral representations for the regularized stress-energy VEV components, using the "rescaled" heat kernel $e^{-\tau \mathcal{A}}(\mathbf{q}, \mathbf{p})$ (of Eq. (4.14)). To this purpose, let us first point out that the Mellin relation (2.266) can be rephrased in the present setting as

$$\mathcal{A}^{-s}(\mathbf{q}, \mathbf{p}) = \frac{\lambda^{-2s}}{\Gamma(s)} \int_0^{+\infty} d\tau \ \tau^{s-1} \ e^{-\tau \mathcal{A}}(\mathbf{q}, \mathbf{p}) \ , \tag{4.15}$$

along with analogous relations for its (covariant) derivatives.

Next notice that, starting from Eq. (4.10), expressions analogous to those in Eq.s (3.107-3.109) can be derived for $(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu} \boldsymbol{v})$ in terms of the Dirichlet kernel and of its derivatives in curvilinear coordinates. Using these expressions for the zeta-regularized stress-energy VEV along with Eq. (4.15) and the corresponding ones for the derivatives of $\mathcal{A}^{-s}(,)$, it follows that

$$(\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu}(\mathbf{q}) \boldsymbol{v}) = \frac{\lambda^{d+1}}{\Gamma(\frac{u+1}{2})} \left(\frac{\kappa}{\lambda}\right)^{u} \int_{0}^{+\infty} d\tau \ \tau^{\frac{u-d-3}{2}} \operatorname{H}^{(u)}_{\mu\nu}(\tau;\mathbf{q}) \quad (\mu,\nu \in \{0,r,\theta_{1},...,\theta_{d-1}\}) \quad (4.16)$$

where the coefficients $\mathbf{H}_{\mu\nu}^{(u)}(\tau; \mathbf{q})$ are as follows (for $i, j, h, \ell \in \{r, \theta_1, ..., \theta_{d-1}\}$; here D is the spatial covariant derivative of Eq. (4.11)):

$$\mathbf{H}_{00}^{(u)}(\tau;\mathbf{q}) := \left(\frac{\tau}{\lambda^2}\right)^{d/2} \left[\left(\frac{1}{4} + \xi\right) \left(\frac{u-1}{2}\right) + \left(\frac{1}{4} - \xi\right) \tau \left(a^{h\ell}(\mathbf{q}) D_{q^h p^\ell} + r^2\right) \right] \Big|_{\mathbf{p}=\mathbf{q}} e^{-\tau \mathcal{A}}(\mathbf{q},\mathbf{p}) , \quad (4.17)$$

$$\mathbf{H}_{0i}^{(u)}(\tau;\mathbf{q}) = \mathbf{H}_{i0}^{(u)}(\tau;\mathbf{q}) := 0 , \qquad (4.18)$$

$$\begin{aligned}
\mathbf{H}_{ij}^{(u)}(\tau \,; \mathbf{q}) &= \mathbf{H}_{ji}^{(u)}(\tau \,; \mathbf{q}) := \\
\left(\frac{\tau}{\lambda^2}\right)^{d/2} \left[\left(\frac{1}{4} - \xi\right) a_{ij}(\mathbf{q}) \left(\frac{u-1}{2} - \tau \left(a^{h\ell}(\mathbf{q}) D_{q^h p^\ell} + r^2\right) \right) + \\
&+ \left(\frac{\tau}{\lambda^2}\right) \left(\left(\frac{1}{2} - \xi\right) D_{q^i p^j} - \xi \, D_{q^i q^j} \right) \right] \Big|_{\mathbf{p}=\mathbf{q}} e^{-\tau \mathcal{A}}(\mathbf{q}, \mathbf{p}) \,.
\end{aligned}$$
(4.19)

Here and in the remainder of this section, we are implicitly understanding the dependence on the parameter ξ for simplicity of notation: so, $\mathbf{H}_{\mu\nu}^{(u)}(\tau; \mathbf{q})$ stands for $\mathbf{H}_{\mu\nu}^{(u)}(\tau; \mathbf{q}; \xi)$. Using the explicit expression (4.14) for the heat kernel in rescaled spherical coordinates, one can infer the following notable properties i)-v) of the coefficients $\mathbf{H}_{\mu\nu}^{(u)}(\tau; \mathbf{q})$; these properties could be proved for arbitrary d (in [66] we checked them by explicit computations in the cases $d \in \{1, 2, 3\}$).

i) For any fixed τ , \mathbf{q} and any $\mu, \nu \in \{0, r, \theta_1, ..., \theta_{d-1}\}$, the map $u \mapsto \mathsf{H}^{(u)}_{\mu\nu}(\tau; \mathbf{q})$ is affine. ii) For any fixed $\mathbf{q}, u \in \mathbb{C}$ and any $\mu, \nu \in \{0, r, \theta_1, ..., \theta_{d-1}\}$, the map $\tau \mapsto \mathsf{H}^{(u)}_{\mu\nu}(\tau; \mathbf{q})$ is smooth (i.e., of class C^{∞}) on $[0, +\infty)$ and exponentially vanishing for $\tau \to +\infty$.

iii) The final expressions for the coefficients $\mathbb{H}^{(u)}_{\mu\nu}$ do not depend on the parameter λ , even though it appears in the right-hand sides of Eq.s (4.17-4.19). iv) There holds

$$H_{\mu\nu}^{(u)} = 0 \quad \text{for } \mu \neq \nu ,$$

$$H_{\theta_{d-1}\theta_{d-1}}^{(u)} = \sin^2(\theta_{d-2}) H_{\theta_{d-2}\theta_{d-2}}^{(u)} = \dots = \sin^2(\theta_{d-2}) \dots \sin^2(\theta_1) H_{\theta_1\theta_1}^{(u)} .$$
(4.20)

v) For $\mu = \nu \in \{0, r, \theta_1\}$, there hold

$$\mathbf{H}_{\mu\nu}^{(u)}(\tau;\mathbf{q}) = e^{-r^{2}\tanh\tau}\mathcal{M}_{\mu\nu}^{(u)}(\tau;r)$$
(4.21)

where $\mathcal{M}_{\mu\nu}^{(u)}(\tau; r)$ is a polynomial in r, u of degree 1 in both these variables, with coefficients depending smoothly on τ .

Before moving on, let us make a few remarks concerning the integral representation (4.16) for the regularized stress-energy VEV. First notice that, in consequence of item iii) above, this VEV only depends on the parameter λ through the multiplicative coefficient $\lambda^{d+1}(\kappa/\lambda)^u$ in front of the integral in the cited equation. On the other hand, Eq. (4.20) of point iv) indicates that the VEV $(\boldsymbol{v} \mid \hat{T}^u_{\mu\nu} \boldsymbol{v})$ is diagonal and that the only independent components are those with $\mu = \nu \in \{0, r, \theta_1\}$; finally, these components only depend on the radial coordinate r (and not on the angular ones $\{\theta_1, ..., \theta_{d-1}\}$), in accordance with the spherical symmetry of the configuration under analysis.

Analytic continuation.

Let us move on to determine the analytic continuation of the regularized stress-energy VEV. To this purpose, consider the integral representation (4.16) of $(\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu} \boldsymbol{v})$ along with the general framework of subsection 2.8, where several methods are proposed to construct

the analytic continuation of Mellin transforms. In particular, Proposition 2.88 can be employed in the case under analysis, setting $F = \mathbf{H}_{\mu\nu}^{(u)}(\mathbf{x})$ and s = (u - d - 1)/2; this allows to infer (compare with Eq. (2.336)), for any $N \in \mathbb{N}$,

$$(\boldsymbol{v} \mid T^{u}_{\mu\nu}(\mathbf{q}) \, \boldsymbol{v}) = \frac{\lambda^{d+1}}{\Gamma(\frac{u+1}{2})} \frac{(-1)^{N}}{(\frac{u-d-1}{2})\dots(\frac{u-d-1}{2}+N-1)} \left(\frac{\kappa}{\lambda}\right)^{u} \int_{0}^{+\infty} d\tau \, \tau^{\frac{u-d-3}{2}+N} \, \partial_{\tau}^{N} \mathsf{H}^{(u)}_{\mu\nu}(\tau;\mathbf{q}) \,.$$

$$(4.22)$$

Due to the features of the coefficients $\mathbf{H}_{\mu\nu}^{(u)}$ pointed out in the previous subsection, the integral in the above expression converges for $u \in \Sigma_{d+1-2N}$, so that Eq. (4.22) yields the required analytic continuation of $(\boldsymbol{v} \mid \hat{T}_{\mu\nu}^u \boldsymbol{v})$ to the very same region; this region includes the value u = 0, required for the zeta approach to renormalization (see Eq. (3.124)), whenever

$$N > \frac{d+1}{2} . (4.23)$$

Under this assumption, following Eq. (3.124), in general we define

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{q}) \boldsymbol{v})_{ren} := RP \Big|_{\boldsymbol{u}=0} (\boldsymbol{v} | \hat{T}^{\boldsymbol{u}}_{\mu\nu}(\mathbf{q}) \boldsymbol{v}) .$$

$$(4.24)$$

For N as in Eq. (4.23), consider the expression in the second line of Eq. (4.22); for any even spatial dimension d this expression is regular in u = 0, so that we can simply evaluate $(\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu}(\mathbf{q}) \boldsymbol{v})$ at this point to obtain the renormalized stress-energy VEV. On the other hand, for odd d we must in fact discard a singular contribution in the Laurent expansion at u = 0, since the function under analysis has a simple pole in u = 0. Because of this pole singularity, the procedure of evaluating the regular part of Eq. (4.22) in u = 0 in the case of odd d implies the appearance of a logarithmic term in τ in the integrand (⁴). Simple but rather lengths computations give the following regults for d either odd or even

Simple but rather lenghty computations give the following results, for d either odd or even and $\mu, \nu \in \{0, r, \theta_1\}$:

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{q}) \boldsymbol{v})_{ren} = \lambda^{d+1} \Big(T^{(0)}_{\mu\nu}(r) + M_{\kappa,\lambda} T^{(1)}_{\mu\nu}(r) \Big) , \quad \text{where}$$

$$T^{(0)}_{\mu\nu}(r) := \int_{0}^{+\infty} d\tau \ \tau^{N-\frac{d+3}{2}} \ e^{-r^{2} \tanh \tau} \Big[\mathcal{P}^{(0)}_{\mu\nu}(\tau;r) + \ln \tau \ \mathcal{P}^{(1)}_{\mu\nu}(\tau;r) \Big] , \quad (4.25)$$

$$T^{(1)}_{\mu\nu}(r) := \int_{0}^{+\infty} d\tau \ \tau^{N-\frac{d+3}{2}} \ e^{-r^{2} \tanh \tau} \ \mathcal{P}^{(1)}_{\mu\nu}(\tau;r) , \qquad M_{\kappa,\lambda} := \gamma_{EM} + 2 \ln \left(\frac{2\kappa}{\lambda}\right)$$

⁴In fact

$$\tau^{u/2} = e^{\frac{u}{2}\ln\tau} = 1 + \frac{u}{2}\ln\tau + O(u^2) \quad \text{for } u \to 0$$

The logarithmic term proportional to u is not relevant in the case of even d, where analytic continuation exists up to u = 0 and it is simply obtained setting u = 0 in (4.22). On the contrary, for odd d we must take the regular part at u = 0 of the expression (4.22) and the above term $\frac{u}{2} \ln \tau$ contributes to it, since it is multiplied by a term proportional to 1/u that comes from the $u \to 0$ expansion of the factor $1/(\frac{u-d-3}{2}+1)...(\frac{u-d-3}{2}+N)$ in (4.22).

 $(\gamma_{EM} \simeq 0.577 \text{ is the Euler-Mascheroni constant})$. In the above definitions $\mathcal{P}_{\mu\nu}^{(0)}(\tau; r)$ and $\mathcal{P}_{\mu\nu}^{(1)}(\tau; r)$ are suitable functions determined by $\mathrm{H}_{\mu\nu}^{(u)}(\tau; \mathbf{q})$; these functions are in fact polynomials in r^2 of order N + 1, with coefficients which are smooth functions of τ on $[0, +\infty)$. Let us stress that

$$\mathcal{P}^{(1)}_{\mu\nu}(\tau;r) = 0 \quad \text{and} \quad T^{(1)}_{\mu\nu}(r) = 0 \qquad \text{for } d \text{ even },$$
 (4.26)

a fact corresponding to the previous comments on the logarithmic terms.

Next note that, in consequence of the remarks at the end of subsection 4.1, the renormalized VEV $(\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$ only depends on the parameter λ through the coefficients λ^{d+1} and $M_{\kappa,\lambda}$ in the first equation of (4.25). In particular, the functions $T^{(a)}_{\mu\nu}(r)$ $(a \in \{0,1\})$ introduced therein do not depend on λ and we can evaluate them computing the integrals in Eq. (4.25) numerically, for any fixed $r \in (0, +\infty)$. In fact, the same integrals can also be used to derive explicit expressions (complete with remainder estimates) for the asymptotic expansions of $(\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$ in the limits $r \to 0^+$ and $r \to +\infty$. We refer to [66] for more details on this topic; in the present manuscript we only report the final results which can be obtained for d = 3 (see the conclusive paragraph of the present section).

To conclude, following Definition 3.51 (⁵), we introduce the conformal and non-conformal parts of the functions $r \mapsto T^{(0)}_{\mu\nu}(r), T^{(1)}_{\mu\nu}(r)$, setting

$$T^{(a,\Diamond)}_{\mu\nu} := T^{(a)}_{\mu\nu}\Big|_{\xi=\xi_d} , \quad T^{(a,\blacksquare)}_{\mu\nu} := \frac{1}{\xi-\xi_d} \left(T^{(a)}_{\mu\nu} - T^{(a,\Diamond)}_{\mu\nu}\right) \qquad (a \in \{0,1\})$$
(4.27)

where ξ_d is defined by Eq. (3.127). In the final paragraph we present the functions $\mathcal{P}_{00}^{(a)}$ ($a \in \{0, 1\}$) and the graphs (obtained via numerical integration) for the functions in Eq. (4.27) corresponding to the energy density ($\mu = \nu = 0$), for d = 3.

The total energy.

First note that, by a simple variation of Proposition 3.45, the total energy can be expressed as the sum of a bulk and a boundary contribution as in Eq.s (3.117-3.119); in the following we are going to discuss these two contributions separately.

Let us first consider the regularized bulk energy; according to Eq. (3.120), this can be expressed as

$$E^u = \frac{\kappa^u}{2} \operatorname{Tr} \mathcal{A}^{-\frac{u-1}{2}}$$

The trace appearing in the above equation is connected through Eq. (2.297) to the heat trace which, according to Eq. (4.5), has the form

Tr
$$e^{-\mathbf{t}\mathcal{A}} = \frac{1}{\mathbf{t}^d} H(\mathbf{t})$$
 with $H(\mathbf{t}) := \left(\frac{\mathbf{t}}{2\sinh(k^2\mathbf{t})}\right)^a$; (4.28)

 $\diamondsuit \equiv {\rm conformal} \ , \qquad \blacksquare \equiv {\rm non-conformal} \ .$

⁵In particular, recall that the following convention is employed:

it is patent that the map $\mathbf{t} \mapsto H(\mathbf{t})$ is smooth on $[0, +\infty)$ and exponentially vanishing for $\mathbf{t} \to +\infty$. Then, using Theorem 2.95, we obtain for the regularized bulk energy

$$E^{u} = \frac{(-1)^{N} \kappa^{u}}{2 \Gamma(\frac{u-1}{2})(\frac{u-1}{2}-d)...(\frac{u-1}{2}-d+N-1)} \int_{0}^{+\infty} d\mathbf{t} \, \mathbf{t}^{\frac{u-3}{2}-d+N} \, \frac{d^{N}}{d\mathbf{t}^{N}} \, H(\mathbf{t}) \, .$$
(4.29)

The above relation holds for any $N \in \{1, 2, 3, ...\}$ and the integral appearing therein converges for any complex $u \in \Sigma_{2(d-N)+1}$; thus, for any integer N > d + 1/2, Eq. (4.29) gives the analytic continuation of E^u in a neighborhood of u = 0. Since here no singularity appears, we can obtain the renormalized bulk energy simply by setting u = 0 in Eq. (4.29); making again the change of integration variable $\tau := k^2 \mathbf{t}$ (see Eq. (4.12)), we infer

$$E^{ren} = -\frac{k}{2^{d+2-N}\sqrt{\pi}} \left(\prod_{i=0}^{N-1} \frac{1}{2(d-i)+1} \right) \int_{0}^{+\infty} d\tau \ \tau^{N-d-\frac{3}{2}} \frac{d^{N}}{d\tau^{N}} \operatorname{H}(\tau)$$

for any $N > d + \frac{1}{2}$, with $\operatorname{H}(\tau) := \left(\frac{\tau}{\sinh \tau}\right)^{d}$. (4.30)

Now, let us move on to discuss the boundary energy \mathcal{B}^u . This can be defined, according to Eq. (3.119); however, since the spatial domain is \mathbb{R}^d , the integral over the boundary appearing therein must be properly interpreted as the limit of integrals over the boundary of suitable, bounded subdomains. By quite lenghty computations (see [66]), this procedure allows to infer that $\mathcal{B}^u = 0$ for all $u \in \Sigma_{d-3}$. Therefore, the zeta approach implies that the renormalized boundary energy vanishes identically:

$$B^{ren} := B^u \Big|_{u=0} = 0 . (4.31)$$

A remark. Before proceeding, let us point out a fact that we anticipated in subsection 3.2: the renormalized total energy \mathcal{E}^{ren} (which, due to Eq. (4.31), in the present setting is equal to E^{ren}) does not coincide with the integral $\int_{\mathbb{R}^d} (\boldsymbol{v} | \hat{T}_{00} \boldsymbol{v})_{ren}$. This fact is patently exemplified by the results to be reported in the subsequent paragraph, dealing with a scalar field in presence of an isotropic harmonic potential in spatial dimension d = 3. Indeed, on the one hand, the forthcoming Eq. (4.46) states that the renormalized total energy $\mathcal{E}^{ren} = E^{ren}$ is finite; on the other hand, Eq. (4.45) (along with Eq. (4.40)) shows that $(\boldsymbol{v} | \hat{T}_{00} \boldsymbol{v})_{ren}$ diverges in a non-integrable way for $|\mathbf{x}| (= r/k) \to +\infty$. Let us stress that the "energy anomaly" $\mathcal{E}^{ren} \neq \int (\boldsymbol{v} | \hat{T}_{00} \boldsymbol{v})_{ren}$ is not a consequence of some ultraviolet issue specific to the present setting: in fact, it also appears in configurations involving bounded spatial domains (such as the one discussed in the subsequent Section 4.2).

The previous results in spatial dimension d = 3.

Hereafter we are going to report the results which can be obtained using the general approach presented in the previous paragraphs for the case of spatial dimension d = 3. We describe this setting using the spherical coordinates $\mathbf{q} = (r, \theta_1, \theta_2) \in (0, +\infty) \times (0, \pi) \times$

 $[0, 2\pi)$, which are related to the Cartesian coordinates $\mathbf{x} \equiv (x_1, x_2, x_3) \in \mathbb{R}^3$ via (see Eq. (4.7))

$$\lambda x^1 = r \cos \theta_1 , \quad \lambda x^2 = r \sin \theta_1 \cos \theta_2 , \quad \lambda x^3 = r \sin \theta_1 \sin \theta_2 ;$$
 (4.32)

the corresponding spatial line element is

$$d\ell^2 = \lambda^{-2} (dr^2 + r^2 (d\theta_1^2 + \sin^2 \theta_1 \, d\theta_2^2)) .$$
(4.33)

After lenghty computations, the zeta-regularized stress-energy VEV can be expressed as (compare with Eq. (4.16) and recall that dependence on ξ is understood implicitly)

$$(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\mathbf{q}) \boldsymbol{v}) = \frac{\lambda^{4}}{\Gamma(\frac{u+1}{2})} \left(\frac{\kappa}{\lambda}\right)^{u} \int_{0}^{+\infty} d\tau \ \tau^{-3+\frac{u}{2}} \operatorname{H}^{(u)}_{\mu\nu}(\tau;\mathbf{q}) \qquad (\mu,\nu \in \{0,r,\theta_{1},\theta_{2}\}) \ , \quad (4.34)$$

where $\mathbf{H}_{\mu\nu}^{(u)}$ is diagonal and we only have to consider the independent components

$$\mathbf{H}_{00}^{(u)}(\tau;\mathbf{q}) := A_{3}(\tau,r) \left[-(1-u)(1+4\xi) + (1-4\xi) \left(\frac{2\tau}{\sinh 2\tau}\right) \left(3 + r^{2} \frac{\sinh 3\tau - \sinh \tau}{\cosh \tau}\right) \right],$$
(4.35)

$$\begin{aligned}
\mathbf{H}_{rr}^{(u)}(\tau;\mathbf{q}) &:= \\
A_3(\tau,r) \left[8\xi \, \frac{\tau}{\tanh \tau} - (1\!-\!4\xi) \left(1\!-\!u + \left(\frac{2\tau}{\sinh 2\tau} \right) \left(1\!+\!2r^2 \tanh \tau \right) \right) \right], \\
\mathbf{H}_{a,c}^{(u)}(\tau;\mathbf{q}) &:=
\end{aligned} \tag{4.36}$$

$$\left(\frac{r}{\lambda}\right)^2 A_3(\tau, r) \left[8\xi \frac{\tau}{\tanh \tau} - (1 - 4\xi) \left(1 - u + \left(\frac{2\tau}{\sinh 2\tau}\right) \left(1 + \frac{r^2}{\cosh^2 2\tau}\right)\right)\right]$$
(4.37)

(for the remaining diagonal component, i.e. $\mathbf{H}_{\theta_2\theta_2}^{(u)}(\tau;\mathbf{q})$, see Eq. (4.20)). In the above, for simplicity of notation we have put

$$A_3(\tau, r) := \frac{1}{64 \,\pi^{3/2}} \, e^{-r^2 \tanh \tau} \left(\frac{2\tau}{\sinh 2\tau}\right)^{3/2} \,. \tag{4.38}$$

The above expressions for the components of the tensor $\mathbf{H}_{\mu\nu}^{(u)}$ are easily seen to possess the features anticipated in Eq.s (4.20) (4.21) and in the related comments. Thus, according to the general framework developed in subsection 4.1, we can obtain the analytic continuation of $(\boldsymbol{v} | \hat{T}_{\mu\nu}^u \boldsymbol{v})$ given in Eq. (4.34) integrating by parts N times the integral appearing therein, for any N > 2 (see Eq. (4.23)). For definiteness, we fix N = 3 so that Eq. (4.22) reads

As in all cases with odd spatial dimension, the analytic continuation of the regularized stress-energy VEV given in Eq. (4.39) has a simple pole in u = 0 (recall the comments made in subsection 4.1). In consequence of this, we have to adopt the extended version of the zeta approach to define the renormalized VEV $(\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$, taking the regular part in u = 0 of Eq. (4.39) (see Eq. (4.24)); with some effort, we obtain

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{q}) \boldsymbol{v})_{ren} = \lambda^{4} \left(T^{(0)}_{\mu\nu}(r) + M_{\kappa,\lambda} T^{(1)}_{\mu\nu}(r) \right) ,$$

$$T^{(0)}_{\mu\nu}(r) := \int_{0}^{+\infty} d\tau \ e^{-r^{2} \tanh \tau} \left[\mathcal{P}^{(0)}_{\mu\nu}(\tau;r) + \ln \tau \ \mathcal{P}^{(1)}_{\mu\nu}(\tau;r) \right] ,$$

$$T^{(1)}_{\mu\nu}(r) := \int_{0}^{+\infty} d\tau \ e^{-r^{2} \tanh \tau} \ \mathcal{P}^{(1)}_{\mu\nu}(\tau;r) , \qquad M_{\kappa,\lambda} := \gamma_{EM} + 2 \ln \left(\frac{2\mu}{\lambda}\right) ,$$

(4.40)

where

$$\mathcal{P}_{\mu\nu}^{(0)}(\tau;r) := -\frac{1}{4\sqrt{\pi}} e^{r^{2} \tanh \tau} \left[3 \partial_{\tau}^{3} \mathsf{H}_{\mu\nu}^{(0)}(\tau;\mathbf{q}) + 4 \partial_{u} \Big|_{u=0} \partial_{\tau}^{3} \mathsf{H}_{\mu\nu}^{(u)}(\tau;\mathbf{q}) \right] ,$$

$$\mathcal{P}_{\mu\nu}^{(1)}(\tau;r) := -\frac{1}{2\sqrt{\pi}} e^{r^{2} \tanh \tau} \partial_{\tau}^{3} \mathsf{H}_{\mu\nu}^{(0)}(\tau;\mathbf{q}) .$$
(4.41)

It is readily found that $\mathcal{P}_{\mu\nu}^{(0)}$, $\mathcal{P}_{\mu\nu}^{(1)}$ are polynomials of fourth order in r^2 . Now, we evaluate numerically the integrals in Eq. (4.40) and distinguish between the conformal and non-conformal parts \diamondsuit , \blacksquare of each component; once more we refer to Eq. (4.27), recalling that for d = 3 we have (see Eq. (3.131))

$$\xi_3 = \frac{1}{6} \ . \tag{4.42}$$

The forthcoming Fig.s 4.1 and 4.2 show, as examples, the graphs of the functions determining the renormalized energy density $(\boldsymbol{v} | \hat{T}_{00} \boldsymbol{v})_{ren}$, i.e.,

$$r \mapsto T_{00}^{(0,\Diamond)}(r), \ T_{00}^{(1,\Diamond)}(r), \ T_{00}^{(0,\blacksquare)}(r), \ T_{00}^{(1,\blacksquare)}(r) \ .$$
 (4.43)

We refer to [66] for the graphs of the other components.



In the following we report the asymptotic expansions of the functions in Eq. (4.43) in the limit of small and large values of the radial coordinate r; these expansions can be obtained starting from the integral representations (4.40) (see [66] for more details).

On the one hand, for $r=\lambda|\mathbf{x}|\to 0^+$ there hold

$$T_{00}^{(0,\Diamond)}(r) = -0.0047 - 0.0024 r^{2} + 0.0028 r^{4} + 0.0006 r^{6} - 0.0001 r^{8} + O(r^{10}) ,$$

$$T_{00}^{(1,\Diamond)}(r) = -0.0016 r^{4} + O(r^{10}) ,$$

$$T_{00}^{(0,\blacksquare)}(r) = -0.0143 - 0.0468 r^{2} + 0.0134 r^{4} - 0.0033 r^{6} + 0.0007 r^{8} + O(r^{10}) ,$$

$$T_{00}^{(1,\blacksquare)}(r) = 0.0380 + O(r^{10}) .$$

$$(4.44)$$

Let us stress that, using the methods described in [66], all the coefficients in the above expansions could be determined explicitly, complete with quantitative remainder estimates. However the expressions involved are quite cumbersome; for this reason, we have reported here only the first four digits of their numerical evaluation. On the other hand, in the limit $r = \lambda |\mathbf{x}| \to +\infty$, the following asymptotics can be inferred:

$$T_{00}^{(0,\Diamond)}(r) = \frac{r^4}{64\pi^2} \left(\ln r^2 + \gamma_{EM} + \frac{1}{2}\right) - \frac{5}{96\pi^2} - \frac{23}{2880\pi^2 r^4} + O(r^{-8}\ln r^2) ,$$

$$T_{00}^{(1,\Diamond)}(r) = -\frac{r^4}{64\pi^2} + O(r^{-8}\ln r^2) ,$$

$$T_{00}^{(0,\bullet)}(r) = -\frac{3}{8\pi^2} \left(\ln r^2 + \gamma_{EM} + \frac{2}{3}\right) + \frac{1}{12\pi^2 r^4} + O(r^{-8}\ln r^2) ,$$

$$T_{00}^{(1,\bullet)}(r) = \frac{3}{8\pi^2} + O(r^{-8}\ln r^2) .$$
(4.45)

In passing, let us point out that the asymptotic expansions in Eq.s (4.44), (4.45) and the corresponding graph in Figure 4.2 suggest that the function $T_{00}^{(1,\blacksquare)}$ is, in fact, constant (notice that $3/8\pi^2 \simeq 0.0380...$).

Finally, Eq. (4.30) with N = 4 and numerical evaluation of the corresponding integral allow us to derive the renormalized bulk energy

$$E^{ren} = -(0.0078607119 \pm 10^{-10}) \lambda .$$
(4.46)

This result agrees with the one obtained by Actor and Bender in [5], using a different approach also related to analytic continuation techniques $(^{6})$.

Since the renormalized boundary energy B^{ren} vanishes identically in the present configuration (see Eq. (4.31)), the above result for the bulk energy E^{ren} suffices to infer that the total energy $\mathcal{E}^{ren} = E^{ren} + B^{ren}$ is finite. On the other hand, it appears from the asymptotic expansions in Eq. (4.45) that the renormalized boundary energy $(\boldsymbol{v} \mid \hat{T}_{00} \boldsymbol{v})_{ren}$ diverges in a non-integrable manner in the limit $r \to +\infty$; therefore, the integral $\int_{\mathbb{R}^d} (\boldsymbol{v} \mid \hat{T}_{00} \boldsymbol{v})_{ren}$ is infinite (or, rather, it does not exists). In view of this, the alternative definition (3.132) for the total energy \mathcal{E}^{ren} does not yield a finite result in the present case. This is one of the anomalies which we pointed out in the remark a the end of subsection 4.1 (see subsection 3.2 for a general discussion).

4.2 The case of a rectangular box.

In the present section we consider a massless field confined within a *d*-dimensional rectangular domain $\Omega = (0, a_1) \times ... \times (0, a_d)$ ($d \in \{1, 2, 3, ...\}$, arbitrary), with Dirichlet boundary conditions. Our approach could also be generalized, with some computational effort, to include the case of a massive field and to deal with Neumann or periodic boundary conditions; we choose to avoid these generalizations here for the sake of simplicity.

⁶To check this, one must compare the numerical value reported in the above Eq. (4.46) with the one reported in Eq. (4.4) of [5]. Let us stress that conventions different from ours are used therein. In fact, using our language, the bulk energy is formally defined in [5] as $E := \sum_{n} \omega_{n}$, while our general prescription (3.120) is formally equivalent to $E = \frac{1}{2} \sum_{n} \omega_{n}$; moreover the parameter α of [5] and our parameter λ are related by $\alpha = \sqrt{2} \lambda$. Summing up, the "total energy" derived in [5] has to be multiplied by 1/2 in order to obtain our E^{ren} .

We follow the analysis presented in our previous work [67]. Our starting point is the heat kernel $e^{-t\mathcal{A}}(,)$, for which we consider two different series representations describing, respectively, the behavior for small and large \mathbf{t} ; these are used to produce series expansions giving the analytic continuations of the Dirichlet kernel $\mathcal{A}^{-s}(,)$ and of its derivatives, which converge with exponential speed (we also give fully quantitative remainder estimates). These analytic continuations determine the renormalized VEVs of several observables, namely: the stress-energy tensor, the pressure on boundary points, the total energy and the total force acting on any side of the box. Our results hold for an arbitrary spatial dimension d; in the conclusive paragraph of the present section we specialize them to the subcase d = 2, producing several graphs for the renormalized VEV of the energy density and for the other observables mentioned previously.

Many of the results reported in this Section (such as those in [67]) have been derived with the aid of Mathematica for both symbolic and numerical computations.

The Casimir effect for a rectangular box configuration has been discussed in a lot of works; here we only mention some of them. Concerning total energy and boundary forces, the foremost computations were performed by Lukosz [99, 100], by means of exponential regularization and Abel-Plana formula; the same techniques were used by Mamaev and Trunov [102, 103] (also see [104, 105]). Alternative derivations based on global ZR were later given by Ruggerio, Vilanni and Zimerman [135, 136], Ambjørn and Wolfram [11, 12] and by Li, Cheng et al. [95]; equivalent results were obtained using generalized cut-off techniques by Edery [52, 53] and by Estrada, Fulling et al. [61, 77]. Finally, let us mention the monographies of Elizalde et al. [56, 57] and Bordag et al. [25]; these can be taken as standard references for the study of global aspects. Our series representations for the total energy and for the boundary forces are different, but equivalent to the ones of [25].

On the other hand, local aspects for a scalar field in a rectangular box were first analyzed by Actor in two seminal papers [3, 4]; therein d = 3, the framework is Euclidean and the author renormalizes, mostly by analytic continuation, the effective Lagrangian density and the VEV $(\boldsymbol{v} | \hat{\varphi}^2(x) \boldsymbol{v})$. It is hardly the case to point out that these observables do not determine the renormalized stress-energy VEV; the latter was instead considered in a work of Svaiter et al. [132], again by means of analytic continuation techniques. However, in [132] some additional "empty space" divergences are removed by hand, with the motivation that they are also present when there is no boundary; moreover, [132] considers only a 3-dimensional, infinite rectangular waveguide $(\Omega = (0, a_1) \times (0, a_2) \times \mathbb{R}),$ rather than a box of arbitrary dimension. Let us also stress that the methods employed in [132] yield a representation of the renormalized stress-energy VEV via series converging with polynomial speed, which is slower than the exponential convergence of our series expansions. An alternative evaluation of the stress-energy VEV was proposed by Estrada, Fulling et al. in [61, 77], where the case of a 2-dimensional rectangular box is analyzed using exponential cut-off techniques; nevertheless, the position of principle in the cited works is that the theory with a cutoff is a more realistic description of the physical system under investigation, and renormalization is only hinted at for the energy density.
Introducing the problem.

We consider the model of a massless scalar field confined within a *d*-dimensional box, with no external potential; more precisely, we assume

$$\Omega = \times_{i=1}^{d} (0, a_i) \quad \text{with } a_i > 0 \text{ for } i \in \{1, ..., d\} , \qquad V = 0 .$$
(4.47)

The boundary $\partial \Omega$ of the spatial domain is composed by the sides

$$\pi_{p,\alpha} := \{ \mathbf{x} \in \mathbb{R}^d \mid x^p = \alpha \, a_p \,, \, x^i \in [0, a_i] \text{ for } i \neq p, \, i \in \{1, ..., d\} \}$$

for $p \in \{1, ..., d\}, \, \alpha \in \{0, 1\} ;$

$$(4.48)$$

for the sake of simplicity, we restrict attention to the case where the field fulfills Dirichlet boundary conditions on each one of these sides. Under the above assumptions, we put

$$\mathcal{A} := -\Delta ; \qquad (4.49)$$

this is strictly positive and self-adjoint on $\mathcal{H} = L^2(\Omega)$, with domain $\mathcal{D}_{\mathcal{A}} := \{f \in H^1_0(\Omega) \mid \Delta f \in L^2(\Omega)\}$ (⁷). Also in this case, \mathcal{A} is an *admissible operator* in the sense of definition 2.37 and it has purely discrete spectrum, given by

$$\sigma(\mathcal{A}) = \left\{ \omega_{\mathbf{m}}^2 := \sum_{i=1}^d \frac{m_i^2 \pi^2}{a_i^2} \mid \mathbf{m} \equiv (m_i)_{i=1,\dots,d}, \, m_i \in \{1, 2, 3, \dots\} \right\}$$
(4.50)

(⁸); a complete orthonormal set of eigenfunctions of \mathcal{A} in $L^2(\Omega)$ corresponding to the eigenvalues $(\omega_{\mathbf{m}}^2)$ in Eq. (4.50) is

$$F_{\mathbf{m}}(\mathbf{x}) := \prod_{i=1}^{d} \sqrt{\frac{2}{a_i}} \sin\left(\frac{m_i \pi}{a_i} x^i\right) \qquad (\mathbf{x} \in \Omega) .$$
(4.51)

Again, we can use the general framework of Chapter 2; in particular, we can consider the Dirichlet and heat kernels associated to \mathcal{A} , along with the related traces.

The heat kernel.

Similarly to the model with a harmonic potential considered in the previous Section 4.1, in the present setting we are dealing with a product domain configuration. In particular, the heat kernel associated to $\mathcal{A} = -\Delta$ factorizes; more precisely, we have

$$e^{-\mathbf{t}\mathcal{A}}(\mathbf{x},\mathbf{y}) = \prod_{i=1}^{d} e^{-\mathbf{t}\mathcal{A}_i}(x^i, y^i) , \qquad (4.52)$$

⁷ Let us point out that, since the boundary $\partial\Omega$ is Lipschitz, one can still define a notion of trace as in Eq. (2.160) and prove that $H_0^1(\Omega) = \{f \in H^1(\Omega) \mid f \upharpoonright \partial\Omega = 0\}$ (see [7]). Moreover, it can be proven that $\mathcal{D}_{\mathcal{A}} = \{f \in H^2(\Omega) \mid f \upharpoonright \partial\Omega = 0\}$. If Ω were smooth, one could infer this fact from Theorem 2.34; in this case $\partial\Omega$ is not smooth, but the same conclusion can be infered using the eigenfunction expansion of \mathcal{A} .

⁸In this case the parameter $\varepsilon > 0$ fulfilling $\sigma(\mathcal{A}) \subset [\varepsilon, +\infty)$, can be chosen to be $\varepsilon := \sum_{i=1}^{d} (\pi/a_i)^2$.

where, for $i \in \{1, ..., d\}$, we have put $\mathcal{A}_i := -\partial_{x^i x^i}$ (with the induced Dirichlet boundary conditions in $x^i = 0$ and $x^i = a_i$) and $e^{-t\mathcal{A}_i}(,)$ indicates the corresponding heat kernel. Each of the one-dimensional kernels can be determined by elementary methods; in particular, we showed in [67] that two different representations can be derived for $e^{-t\mathcal{A}_i}(,)$, starting from the set of eigenfunctions and eigenvalues in Eq.s (4.50) (4.51) and using the Poisson summation formula. Of course, these representations determine, according to (4.52), two alternative expressions for the total heat kernel $e^{-t\mathcal{A}}(,)$, which we proceed to report hereafter. It appears that these alternative expressions are suited to describe the behaviour of this kernel for small and large **t**, respectively.

Large \mathbf{t} representation of the heat kernel. This is

$$e^{-\mathbf{t}\mathcal{A}}(\mathbf{x},\mathbf{y}) = \frac{2^d}{a_1 \dots a_d} \sum_{\mathbf{m} \in \mathbb{N}^d_{\star}} e^{-\omega_{\mathbf{m}}^2 \mathbf{t}} \mathcal{C}_{\mathbf{m}}(\mathbf{x},\mathbf{y}) , \qquad (4.53)$$

where $\omega_{\mathbf{m}}$ is as in Eq. (4.50) and, for the sake of brevity, we have put

$$\mathbb{N}_{\star} := \{1, 2, 3, \ldots\} , \qquad \mathcal{C}_{\mathbf{m}}(\mathbf{x}, \mathbf{y}) := \prod_{i=1}^{d} \sin\left(\frac{m_{i}\pi}{a_{i}} y^{i}\right) \sin\left(\frac{m_{i}\pi}{a_{i}} x^{i}\right) . \tag{4.54}$$

Small t representation of the heat kernel. This is

$$e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y}) = \frac{1}{(4\pi\mathbf{t})^{d/2}} \sum_{\mathbf{h}\in\mathbb{Z}^d, \,\mathbf{l}\in\{1,2\}^d} \delta_{\mathbf{l}} \, e^{-\frac{1}{\mathbf{t}} \, b_{\mathbf{h}\mathbf{l}}(\mathbf{x}, \mathbf{y})} \,, \tag{4.55}$$

where, for simplicity of notation, we have put

$$\mathbf{h} := (h_i)_{i=1,\dots,d} , \quad \mathbf{l} := (l_i)_{i=1,\dots,d} , \qquad \delta_{\mathbf{l}} := \prod_{i=1}^d \delta_{l_i} , \quad \delta_{l_i} := \begin{cases} 1 & \text{for } l_i = 1 \\ -1 & \text{for } l_i = 2 \end{cases} , b_{\mathbf{h}\mathbf{l}}(\mathbf{x}, \mathbf{y}) := \sum_{i=1}^d a_i^2 (h_i - \mathcal{D}_{l_i}(x^i, y^i))^2 , \qquad \mathcal{D}_{l_i}(x^i, y^i) := \begin{cases} \frac{x^i - y^i}{2a_i} & \text{for } l_i = 1 \\ \frac{x^i + y^i}{2a_i} & \text{for } l_i = 2 \end{cases} .$$

$$(4.56)$$

Before moving on, let us emphasize a number of facts on the expansions (4.53) (4.53); we will resort to them in the following subsections, when performing the analytic continuation of the Dirichlet kernel and of its derivatives.

i) The expansions in Eq.s (4.53) and (4.55) describe, respectively, the large and small **t** behaviour of $e^{-\mathbf{t}\mathcal{A}}(,)$ in the following sense. The series over $\mathbf{m} \in \mathbb{N}^d_{\star}$ in Eq. (4.53) and the one over $\mathbf{h} \in \mathbb{Z}^d$ in Eq. (4.55) are mainly determined by the terms corresponding to small values of m_i and $|h_i|$ ($i \in \{1, ..., d\}$), in the limits $\mathbf{t} \to +\infty$ and $\mathbf{t} \to 0^+$ respectively. ii) Notice that

$$\omega_{\mathbf{n}}^2 > 0 \quad \text{for all } \mathbf{n} \in \mathbb{N}^d$$
. (4.57)

iii) There holds

$$b_{\mathbf{hl}}(\mathbf{x}, \mathbf{y}) \ge 0 \quad \text{for all } \mathbf{h} \in \mathbb{Z}^d, \, \mathbf{l} \in \{1, 2\}^d, \, \mathbf{x}, \mathbf{y} \in \Omega$$
 (4.58)

In particular, since $\mathcal{D}_{l_i}(x^i, y^i) \in [-1/2, 1/2]$ for $l_i = 1$ and $\mathcal{D}_{l_i}(x^i, y^i) \in [0, 1]$ for $l_i = 2$ (see the definition of \mathcal{D}_{l_i} in Eq. (4.56)), it follows that

$$b_{\mathbf{hl}}(\mathbf{x}, \mathbf{y}) = 0 \iff \text{for each } i \in \{1, ..., d\}, \text{ one has} \begin{cases} h_i = 0, \ l_i = 1, \ y^i = x^i \in [0, a_i] \\ \text{or } h_i = 0, \ l_i = 2, \ y^i = x^i = 0; \\ \text{or } h_i = 1, \ l_i = 2, \ y^i = x^i = a_i \end{cases}$$
(4.59)

let us stress that this implies, in particular, $b_{hl}(\mathbf{x}, \mathbf{y}) \neq 0$ for $\mathbf{x} \neq \mathbf{y}$.

The Dirichlet kernel.

The analytic continuation of $\mathcal{A}^{-s}(,)$ can be constructed in the style of Minakshisundaram [111], starting from the integral representation (2.266) in terms of the heat kernel $e^{-t\mathcal{A}}(,)$; to this purpose, let us fix arbitrarily

$$T \in (0, +\infty) \tag{4.60}$$

and re-express the cited integral representation, for $\mathbf{x}, \mathbf{y} \in \Omega$, as

$$\mathcal{A}^{-s}(\mathbf{x}, \mathbf{y}) = \mathcal{A}^{-s}_{(>)}(\mathbf{x}, \mathbf{y}) + \mathcal{A}^{-s}_{(<)}(\mathbf{x}, \mathbf{y}) , \quad \text{where}$$
(4.61)

$$\mathcal{A}_{(>)}^{-s}(\mathbf{x}, \mathbf{y}) := \frac{1}{\Gamma(s)} \int_{T}^{+\infty} d\mathbf{t} \, \mathbf{t}^{s-1} \, e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y}) \,, \qquad \mathcal{A}_{(<)}^{-s}(\mathbf{x}, \mathbf{y}) := \frac{1}{\Gamma(s)} \int_{0}^{T} d\mathbf{t} \, \mathbf{t}^{s-1} \, e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y})$$

(notice that $\mathcal{A}_{(>)}^{-s}(,)$ and $\mathcal{A}_{(>)}^{-s}(,)$ depend on T, but their sum $\mathcal{A}^{-s}(,)$ does not!). To construct the analytic continuation of $\mathcal{A}^{-s}(,)$, we substitute in the definitions of $\mathcal{A}_{(>)}^{-s}(,)$ and $\mathcal{A}_{(<)}^{-s}(,)$, respectively, the large and small **t** expansions (4.53) (4.55) for the heat kernel $e^{-\mathbf{t}\mathcal{A}}(\mathbf{x}, \mathbf{y})$ (⁹). We obtain the results reported hereafter.

Series expansion for $\mathcal{A}_{(>)}^{-s}(\boldsymbol{x}, \boldsymbol{y})$. Using the representation (4.53) for the heat kernel, one can derive [67] the following, for $\mathbf{x}, \mathbf{y} \in \Omega$:

$$\mathcal{A}_{(>)}^{-s}(\mathbf{x}, \mathbf{y}) = \frac{2^d}{a_1 \dots a_d} \Gamma(s) \sum_{\mathbf{m} \in \mathbb{N}^d_{\star}} \omega_{\mathbf{m}}^{-2s} \Gamma(s, \omega_{\mathbf{m}}^2 T) \mathcal{C}_{\mathbf{m}}(\mathbf{x}, \mathbf{y}) , \qquad (4.62)$$

where, for any $s \in \mathbb{C}$ and $z \in (0, +\infty)$, $\Gamma(s, z)$ is the upper incomplete gamma function (see [122], page 174, Eq. 8.2.2). The above expression can also be used to evaluate derivatives

⁹In these manipulations (and in some related computations) we often take for granted that certain series can be integrated or differentiated term by term. In all cases under analysis, rigorous justifications could be given using the Lebesgue dominated convergence theorem or the Fubini-Tonelli theorem, but we will not go into the details.

of any order of the function $\mathcal{A}_{(>)}^{-s}(, \cdot)$. Moreover, the series in the right-hand sides of Eq. (4.62), along with analogous ones for the derivatives of $\mathcal{A}_{(>)}^{-s}(, \cdot)$, converges for all $s \in \mathbb{C}$ even for $\mathbf{y} = \mathbf{x}$ (see the subsequent paragraph); so, Eq. (4.62) yields automatically the analytic continuation of the map $s \mapsto \mathcal{A}_{(>)}^{-s}(\mathbf{x}, \mathbf{y})$ to the whole complex plane.

Series expansion for $\mathcal{A}_{(<)}^{-s}(\boldsymbol{x}, \boldsymbol{y})$. The expression (4.55) for the heat kernel can be used to infer, for $\mathbf{x}, \mathbf{y} \in \Omega$,

$$\mathcal{A}_{(<)}^{-s}(\mathbf{x}, \mathbf{y}) = \frac{T^{s-\frac{d}{2}}}{(4\pi)^{d/2} \Gamma(s)} \sum_{\mathbf{h} \in \mathbb{Z}^d, \, \mathbf{l} \in \{1,2\}^d} \delta_{\mathbf{l}} \, \mathcal{P}_{s-\frac{d}{2}}\left(\frac{b_{\mathbf{h}\mathbf{l}}(\mathbf{x}, \mathbf{y})}{T}\right) \,. \tag{4.63}$$

Here, for $\beta \geq 0$ and $s \in \mathbb{C}$ (recall that $b_{\mathbf{hl}}(\mathbf{x}, \mathbf{y}) \geq 0$ for all $\mathbf{h} \in \mathbb{Z}^d$, $\mathbf{l} \in \{1, 2\}^d$; see Eq. (4.58)), we have introduced the function

$$\mathcal{P}_{s}(\beta) := \int_{0}^{1} d\tau \ \tau^{s-1} e^{-\frac{\beta}{\tau}} \ ; \tag{4.64}$$

this map fulfills

$$\mathcal{P}_{s}(\beta) = \begin{cases} s^{-1} & \text{for } \beta = 0, \, \Re s > 0\\ \beta^{s} \, \Gamma(-s, \beta) & \text{for } \beta > 0 \end{cases}, \qquad (4.65)$$
$$\partial_{\beta}^{\ell} \mathcal{P}_{s}(\beta) = (-1)^{\ell} \, \mathcal{P}_{s-\ell}(\beta) & \text{for } \ell \in \mathbb{N} \end{cases}$$

(again, $\Gamma(,)$ denotes the upper incomplete gamma function). Let us point out that, due to the results reported in Eq. (4.59), we have

$$b_{\rm hl}(\mathbf{x}, \mathbf{y}) = 0$$
 only for $\mathbf{y} = \mathbf{x}$ and for
a finite number of terms in the series of Eq. (4.63). (4.66)

The above mentioned terms of Eq. (4.63) deserve special attention, and must be evaluated using the first identity in Eq. (4.65); it follows that

$$\mathcal{A}_{(<)}^{-s}(\mathbf{x}, \mathbf{y}) = \frac{T^{s-\frac{d}{2}}}{(4\pi)^{d/2} \Gamma(s) (s - \frac{d}{2})} \sum_{\substack{\mathbf{h} \in \mathbb{Z}^{d}, \mathbf{l} \in \{1, 2\}^{d} \\ \text{s.t. } b_{\mathbf{h} \mathbf{l}}(\mathbf{x}, \mathbf{y}) = 0}} \delta_{\mathbf{l}} + \frac{1}{(4\pi)^{d/2} \Gamma(s)} \sum_{\substack{\mathbf{h} \in \mathbb{Z}^{d}, \mathbf{l} \in \{1, 2\}^{d} \\ \text{s.t. } b_{\mathbf{h} \mathbf{l}}(\mathbf{x}, \mathbf{y}) > 0}} \delta_{\mathbf{l}} \left(b_{\mathbf{h} \mathbf{l}}^{s-\frac{d}{2}} \Gamma\left(\frac{d}{2} - s, \frac{b_{\mathbf{h} \mathbf{l}}}{T}\right) \right) (\mathbf{x}, \mathbf{y}) .$$

$$(4.67)$$

Let us repeat that the first sum in the above expression contains finitely many terms. Notice that the term in the first line of Eq. (4.67) is related to the first equality in Eq. (4.65) which, in principle, would require $\Re s > d/2$; however, this term makes sense for all complex s except s = d/2, where a simple pole appears. On the other hand, the series in the second line of Eq. (4.67) can be proved to converge for any complex s (see, again, the forthcoming paragraph). In view of these remarks, Eq. (4.67) gives automatically the analytic continuation of $\mathcal{A}_{(<)}^{-s}(\ ,\)$ to a meromorphic function of s on the whole complex plane, with a simple pole singularity only for

$$\mathbf{y} = \mathbf{x}$$
 and $s = d/2$. (4.68)

A similar analysis can be made for the derivatives of $\mathcal{A}_{(<)}^{-s}(,)$, for which series representations analogous to that in Eq. (4.65) can be derived. For brevity, we do not report here the detailed discussion of these quantities, for which we refer to [67]. Let us only mention that, for any two spatial variables z, w and for any $\mathbf{x}, \mathbf{y} \in \Omega$, the map $s \mapsto \partial_{zw} \mathcal{A}_{(<)}^{-s}(\mathbf{x}, \mathbf{y})$ can be analytically continued to a function meromorphic on the whole complex plane, with a simple pole singularity only for

$$y = x$$
 and $s = d/2 + 1$. (4.69)

Let us stress that the above facts allow to infer, in particular, that the analytic continuations of $\mathcal{A}^{-\frac{u-1}{2}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ and $\partial_{zw}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y})|_{\mathbf{y}=\mathbf{x}}$ (required for the evaluation of the regularized stress-energy VEV and pressure) are both regular at u = 0.

Convergence and remainder estimates for the series in Eq.s (4.62) and (4.63).

This subject is discussed in more detail in [67] (see, in particular, subsection 3.4 and the related Appendix A therein); here we only report the main results. To this purpose, some notations are required; first of all we put

$$a := \min_{i \in \{1, \dots, d\}} \{a_i\} , \qquad A := \max_{i \in \{1, \dots, d\}} \{a_i\} ,$$

$$|\mathbf{z}| := \left(\sum_{i=1}^d z_i^2\right)^{1/2} \quad \text{for } \mathbf{z} = \mathbf{m} \in \mathbb{N}^d \text{ or } \mathbf{z} = \mathbf{h} \in \mathbb{Z}^d .$$

$$(4.70)$$

Besides, for $N \in (2\sqrt{d}, +\infty)$, $\alpha \in (0, 1)$, $\beta \in (0, +\infty)$, $\sigma, \rho \in \mathbb{R}$, we set

$$H_N^{(d)}(\alpha,\beta;\sigma,\rho) := \tag{4.71}$$

$$\frac{\pi^{d/2}}{(1-\alpha)^{\sigma}(\alpha\beta)^{\frac{d+\rho}{2}}\Gamma(\frac{d}{2})} \left(\frac{N-\sqrt{d}}{N-2\sqrt{d}}\right)^{d-1} \Gamma(\sigma,(1-\alpha)\beta N^2) \Gamma\left(\frac{d+\rho}{2};\alpha\beta(N-2\sqrt{d})^2\right);$$

it can be shown that there holds the asymptotic expansion, for $N \to +\infty$,

$$H_N^{(d)}(\alpha,\beta;\sigma,\rho) = \frac{\pi^{d/2} \,\beta^{\sigma-2} e^{-4\alpha\beta d}}{\alpha(1-\alpha)\,\Gamma(\frac{d}{2})} \,e^{-\beta N(N-4\alpha\beta\sqrt{d})} N^{2\sigma+\rho+d-4} \left(1+O(N^{-1})\right) \,. \tag{4.72}$$

Finally, we put

$$C_{a,A}^{(d)}(\sigma,N) := \max\left[\left(a\left(1-\frac{\sqrt{d}}{N}\right)\right)^{2\sigma}, \left(A\left(1+\frac{\sqrt{d}}{N}\right)\right)^{2\sigma}\right].$$
(4.73)

Having introduced the above notations, let us proceed to give the previously mentioned remainder estimates for the series expansions of $\mathcal{A}_{(>)}^{-s}$ and $\mathcal{A}_{(<)}^{-s}$. In all cases the remainder is controlled by the function $H_N^{(d)}$; due to the exponential decay of this function for large N described in Eq. (4.72), good approximations of all the series under investigation can be obtained by just summing the first few terms.

Estimates for the series (4.62). Let $s \in \mathbb{C}$; keeping in mind Eq. (4.62), for any $N \in (0, +\infty)$ let us write

$$\mathcal{A}_{(>)}^{-s}(\mathbf{x}, \mathbf{y}) = \mathcal{A}_{(>),N}^{-s}(\mathbf{x}, \mathbf{y}) + R_{(>),N}^{-s}(\mathbf{x}, \mathbf{y}) ,$$

$$\mathcal{A}_{(>),N}^{-s}(\mathbf{x}, \mathbf{y}) := \frac{2^d}{a_1 \dots a_d \Gamma(s)} \sum_{\mathbf{m} \in \mathbb{N}_{\star}^d, |\mathbf{m}| \le N} \omega_{\mathbf{m}}^{-2s} \Gamma(s, \omega_{\mathbf{m}}^2 T) \mathcal{C}_{\mathbf{m}}(\mathbf{x}, \mathbf{y}) ,$$

$$R_{(>),N}^{-s}(\mathbf{x}, \mathbf{y}) := \frac{2^d}{a_1 \dots a_d \Gamma(s)} \sum_{\mathbf{m} \in \mathbb{N}_{\star}^d, |\mathbf{m}| > N} \omega_{\mathbf{m}}^{-2s} \Gamma(s, \omega_{\mathbf{m}}^2 T) \mathcal{C}_{\mathbf{m}}(\mathbf{x}, \mathbf{y}) .$$
(4.74)

For the remainder function $R^{-s}_{(>),N}(\mathbf{x},\mathbf{y})$ we have the following uniform estimate

$$\left| R_{s,N}^{(>)}(\mathbf{x},\mathbf{y}) \right| \leq \frac{\max(a^{2\Re s}, A^{2\Re s})}{a_1 \dots a_d \, \pi^{2\Re s} |\Gamma(s)|} \, H_N^{(d)}\left(\alpha, \frac{\pi^2 T}{A^2}; \Re s, -2\Re s\right)$$

for either $\Re s \geq 0, \, N > 2\sqrt{d}$ or $\Re s < 0, \, N > 2\sqrt{d} + \frac{A}{\pi}\sqrt{\frac{|\Re s|}{\alpha T}}$. (4.75)

In the above, α is a parameter that can be freely chosen in (0, 1); of course, the best choice is the one minimizing the right-hand side of Eq. (4.75), which depends on the other parameters (e.g., N, T) involved in these considerations.

Estimates for the series (4.63). Let $s \in \mathbb{C}$, and exclude the case (4.68); keeping in mind Eq. (4.63), for any $N \in (0, +\infty)$ we put

$$\mathcal{A}_{(<),N}^{-s}(\mathbf{x},\mathbf{y}) = \mathcal{A}_{(<),N}^{-s}(\mathbf{x},\mathbf{y}) + R_{(<),N}^{-s}(\mathbf{x},\mathbf{y}) ,$$

$$\mathcal{A}_{(<),N}^{-s}(\mathbf{x},\mathbf{y}) := \frac{T^{s-\frac{d}{2}}}{(4\pi)^{d/2}\Gamma(s)} \sum_{\mathbf{h}\in\mathbb{Z}^d, \, |\mathbf{h}|\leq N, \, \mathbf{l}\in\{1,2\}^d} \delta_{\mathbf{l}} \,\mathcal{P}_{s-\frac{d}{2}}\left(\frac{b_{\mathbf{hl}}(\mathbf{x},\mathbf{y})}{T}\right) ,$$

$$R_{(<),N}^{-s}(\mathbf{x},\mathbf{y}) := \frac{T^{s-\frac{d}{2}}}{(4\pi)^{d/2}\Gamma(s)} \sum_{\mathbf{h}\in\mathbb{Z}^d, \, |\mathbf{h}|>N, \, \mathbf{l}\in\{1,2\}^d} \delta_{\mathbf{l}} \,\mathcal{P}_{s-\frac{d}{2}}\left(\frac{b_{\mathbf{hl}}(\mathbf{x},\mathbf{y})}{T}\right) .$$
(4.76)

For the remainder function $R_{(<),N}^{-s}$, we have the following uniform estimate:

$$\left| R_{s,N}^{(<)}(\mathbf{x},\mathbf{y}) \right| \leq \frac{C_{a,A}^{(d)}(\Re s - \frac{d}{2}, N)}{\pi^{d/2} |\Gamma(s)|} H_N^{(d)} \left(\alpha, \frac{a^2 (1 - \frac{\sqrt{d}}{N})^2}{T}; \frac{d}{2} - \Re s, 2\Re s - d \right)$$
for either $\Re s \leq \frac{d}{2}, N > 2\sqrt{d}$ or $\Re s > \frac{d}{2}, N > 3\sqrt{d} + \frac{1}{a}\sqrt{\frac{(\Re s - \frac{d}{2})T}{\alpha}}.$

$$(4.77)$$

Again, the parameter α can be freely taken in (0, 1) and it is conventient to choose for it the value minimizing the right-hand sides of Eq. (4.77), keeping into account the choices made for the other parameters (in particular, for N).

The stress-energy tensor.

Consider the representations deduced in the previous paragraph for the analytic continuations of the Dirichlet kernel (and of its derivatives). Resorting to Eq.s (3.107-3.109), one can obtain series expansions for each component of the zeta-regularized stress-energy VEV ($\boldsymbol{v} \mid \hat{T}^{u}_{\mu\nu} \boldsymbol{v}$). More precisely, for any $\mathbf{x} \in \Omega$, we have

$$(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\mathbf{x}) \boldsymbol{v}) = T^{u,(>)}_{\mu\nu}(\mathbf{x}) + T^{u,(<)}_{\mu\nu}(\mathbf{x}) , \qquad (4.78)$$

where, for • equal to > or <, $T^{u,(\bullet)}_{\mu\nu}(\mathbf{x})$ has the espression corresponding to Eq.s (3.107-3.109), with $\mathcal{A}^{-s}(,)$ replaced by $\mathcal{A}^{-s}_{(\bullet)}(,)$; thus, for $i, j \in \{1, ..., d\}$, we have

$$T_{00}^{u,(\bullet)}(\mathbf{x}) = \kappa^{u} \left[\left(\frac{1}{4} + \xi \right) \mathcal{A}_{(\bullet)}^{-\frac{u-1}{2}}(\mathbf{x}, \mathbf{y}) + \left(\frac{1}{4} - \xi \right) \partial^{x^{\ell}} \partial_{y^{\ell}} \mathcal{A}_{(\bullet)}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right]_{\mathbf{y}=\mathbf{x}},$$
(4.79)

$$T_{0j}^{u,(\bullet)}(\mathbf{x}) = T_{j0}^{u,(\bullet)}(\mathbf{x}) = 0$$
, (4.80)

$$T_{ij}^{u,(\bullet)}(\mathbf{x}) = T_{ji}^{u,(\bullet)}(\mathbf{x}) = \kappa^{u} \left[\left(\frac{1}{4} - \xi \right) \partial_{ij} \left(\mathcal{A}_{(\bullet)}^{-\frac{u-1}{2}}(\mathbf{x}, \mathbf{y}) - \partial^{x^{\ell}} \partial_{y^{\ell}} \mathcal{A}_{(\bullet)}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right) + \left(\left(\frac{1}{2} - \xi \right) \partial_{x^{i}y^{j}} - \xi \partial_{x^{i}x^{j}} \right) \mathcal{A}_{(\bullet)}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right]_{\mathbf{y}=\mathbf{x}}.$$

$$(4.81)$$

In view of the results discussed in the previous paragraphs for the functions $\mathcal{A}_{(>)}^{-s}(\mathbf{x}, \mathbf{y})$ and $\mathcal{A}_{(<)}^{-s}(\mathbf{x}, \mathbf{y})$, it appears that u = 0 is a regular point for each component of $(\boldsymbol{v} \mid \hat{T}_{\mu\nu}^{u}(\mathbf{x}) \boldsymbol{v})$; so, the general zeta approach to renormalization (3.121) reduces, in this case, to

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{x}) \boldsymbol{v})_{ren} := (\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\mathbf{x}) \boldsymbol{v}) \Big|_{u=0} .$$
(4.82)

In the following, when considering the case in spatial dimension d = 2, we will use approximate expressions for all the components of $(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{x}) \boldsymbol{v})_{ren}$ obtained replacing each Dirichlet function $\mathcal{A}_{(\bullet)}^{-s}(,)$ in Eq.s (4.79-4.81) with the truncations $\mathcal{A}_{(\bullet),N}^{-s}(,)$ of a fixed (sufficiently large) order N, given by Eq.s (4.74) (4.76). Let us recall that we have explicit remainder bounds for these truncations (see Eq.s (4.75) (4.77)); these allow us to infer error estimates for the approximate expressions of $(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{x}) \boldsymbol{v})_{ren}$ described above.

The pressure on the boundary.

Let \mathbf{x} be any point interior to one of the sides $\pi_{p,\alpha}$; we exclude \mathbf{x} to be on an edge of the box (i.e., on the intersection of two or more sides), where the outer normal is ill-defined. As an example, let us assume \mathbf{x} to be an inner point of the side $\pi_{1,0}$, so that the unit outer normal at \mathbf{x} is $\mathbf{n}(\mathbf{x}) = (-1, 0, ..., 0)$. The zeta-regularized pressure at \mathbf{x} can be defined according to the general prescription (3.115), which in the present case reduces to

$$p_i^u(\mathbf{x}) := (\boldsymbol{v} \mid \hat{T}_{ij}^u(\mathbf{x}) \, \boldsymbol{v}) \, n^j(\mathbf{x}) = -(\boldsymbol{v} \mid \hat{T}_{i1}^u(\mathbf{x}) \, \boldsymbol{v}) \; ; \qquad (4.83)$$

since Dirichlet boundary conditions are prescribed, recalling the general expression (3.109) for $(\boldsymbol{v} | \hat{T}_{ij}^{u}(\mathbf{x}) \boldsymbol{v})$ and using the decomposition (4.61) for the Dirichlet kernel, the above relation yields (¹⁰)

$$p_{i}^{u}(\mathbf{x}) = -\delta_{i1} \frac{\kappa^{u}}{4} \left. \partial_{x^{1}y^{1}} \mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x}, \mathbf{y}) \right|_{\mathbf{y}=\mathbf{x}} = \\ = -\delta_{i1} \frac{\kappa^{u}}{4} \left[\left. \partial_{x^{1}y^{1}} \mathcal{A}^{-\frac{u+1}{2}}_{(>)}(\mathbf{x}, \mathbf{y}) \right|_{\mathbf{y}=\mathbf{x}} + \left. \partial_{x^{1}y^{1}} \mathcal{A}^{\frac{u+1}{2}}_{(<)}(\mathbf{x}, \mathbf{y}) \right|_{\mathbf{y}=\mathbf{x}} \right] .$$

$$(4.84)$$

Recalling that the Dirichlet kernel and and its derivatives are all regular at u = 0, the general zeta approach to renormalization (3.126) reduces in this case to

$$p_i^{ren}(\mathbf{x}) := p_i^u(\mathbf{x}) \Big|_{u=0} = -\left(\boldsymbol{v} \, | \, \hat{T}_{i1}^u(\mathbf{x}) \, \boldsymbol{v} \right) \Big|_{u=0} \,.$$
(4.85)

When dealing with the case d = 2, we will evaluate the pressure starting from Eq.s (4.84) (4.85) and substituting the functions $\partial_{x^1y^1}\mathcal{A}_{(\bullet)}^{-s}(,)$ therein with the truncations $\partial_{x^1y^1}\mathcal{A}_{(\bullet),N}^{-s}(,)$ of a sufficiently large order N; the errors of these approximants will be evaluated using the analogues of Eq.s (4.75) and (4.77), holding for the derivatives of the Dirichlet kernel.

Before proceeding, let us mention a couple of facts about the renormalized pressure on the boundary; their proof are discussed in detail in our previous work [67].

i) Let us remark that at points on the edges of the box (i.e., on the corners which appear whenever d > 1) the outer normal and, consequently, the pressure are both ill-defined. It can be proved that the renormalized pressure (4.85) evaluated at inner points of one side diverges in a non-integrable manner when moving towards anyone of the edges; this fact is of utmost importance when attempting to evaluate the total force acting on any side of the box, a topic we discussed in detail in [66].

ii) In accordance with the alternative definition (3.133), one could define the pressure at a boundary point $\mathbf{x} \in \pi_{1,0}$ as

$$p_i^{ren}(\mathbf{x}) := \left(\lim_{\mathbf{x}' \in \Omega, \, \mathbf{x}' \to \mathbf{x}} (\boldsymbol{v} \,|\, \hat{T}_{i1}(\mathbf{x}') \,\boldsymbol{v})_{ren} \right) n^j(\mathbf{x}) = -\left(\lim_{\mathbf{x}' \in \Omega, \, \mathbf{x}' \to \mathbf{x}} (\boldsymbol{v} \,|\, \hat{T}_{i1}(\mathbf{x}') \,\boldsymbol{v})_{ren} \right). \quad (4.86)$$

$$\begin{split} \mathcal{A}^{-s}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} &= 0 , \qquad \partial_{x^{i}x^{j}}\mathcal{A}^{-s}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} &= 0 \quad \text{for all } i,j \in \{1,...,d\} ; \\ \partial_{x^{i}y^{j}}\mathcal{A}^{-s}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} &= 0 \quad \text{for all } i,j \in \{1,...,d\} \text{ such that } i \neq 1 \text{ or } j \neq 1 . \end{split}$$

These follow straightforwardly from the Dirichlet conditions prescribed on the boundary of Ω and from item vi) of Proposition 2.64.

¹⁰In the application of Eq. (3.109) to the present case, we use the following identities, holding for all $\mathbf{x} \in \partial \Omega$ and suitable $s \in \mathbb{C}$:

Simple but long computations allow to infer that the different prescriptions (4.85) and (4.86) are, in fact, equivalent for the configuration under analysis; therefore, no anomaly of the type mentioned in subsection 3.2 appears in this case for the VEV of the pressure at boundary points.

The total energy.

First of all, let us point out that, by a slight generalization of Proposition 3.45, the zeta-regularized total energy \mathcal{E}^u can be expressed as the sum of both a bulk and a boundary contribution, respectively indicated with E^u and B^u . Since Dirichlet conditions are assumed on the boundary, according to the considerations of Remark 3.29 we have

$$B^u = 0 (4.87)$$

therefore, we only have to discuss the regularized bulk term E^u . Recalling the general definition (3.118) and using the expression (4.61) for the Dirichlet kernel, we readily infer

$$E^{u} = E^{u}_{(>)} + E^{u}_{(<)} \quad \text{where}$$

$$E^{u}_{(\bullet)} := \frac{\kappa^{u}}{2} \int_{(0,a_{1}) \times \ldots \times (0,a_{d})} dx^{1} \ldots dx^{d} \mathcal{A}^{-\frac{u-1}{2}}_{(\bullet)}(\mathbf{x}, \mathbf{x}) \quad \text{for } \bullet \in \{>, <\}.$$
(4.88)

Series expansion for $E_{(>)}^u$. Let us insert the expansion (4.62) for $\mathcal{A}_{(>)}^{-s}$ into Eq. (4.88); then, exchanging the order of integration and summation and evaluating each single integral thus obtained in terms of incomplete gamma functions, we obtain

$$E_{(>)}^{u} = \frac{\kappa^{u}}{2\Gamma(\frac{u-1}{2})} \sum_{\mathbf{m}\in\mathbb{N}_{\star}^{d}} \omega_{\mathbf{m}}^{1-u} \Gamma\left(\frac{u-1}{2}, \omega_{\mathbf{m}}^{2} T\right) .$$

$$(4.89)$$

The above series can be proved to converge for all $u \in \mathbb{C}$, thus giving the analytic continuation of $E^{u,(>)}$ to the whole complex plane, in particular at u = 0.

Series expansion for $E_{(<)}^u$. Similarly, inserting the expansion (4.63) for $\mathcal{A}_{(<)}^{-s}$ into Eq. (4.88), after some effort we obtain

$$E_{(<)}^{u} = \frac{\kappa^{u} T^{\frac{u-1}{2}}}{2^{d+1} \Gamma(\frac{u-1}{2})} \sum_{p=0}^{d} \frac{(-1)^{d-p}}{(d-p)! p!} \sum_{\sigma \in \mathbb{P}_{d}} \frac{\mathfrak{a}_{\sigma,p}}{(\pi T)^{p/2}} \sum_{\mathbf{h} \in \mathbb{Z}^{p}} \mathcal{P}_{\frac{u-p-1}{2}} \left(\frac{B_{\sigma,p}(\mathbf{h})}{T}\right).$$
(4.90)

In the above P_d indicates the symmetric group with d elements and we have put

$$\mathfrak{a}_{\sigma,0} := 1 , \qquad \mathbb{Z}^0 := \{\mathbf{0}\} , \qquad B_{\sigma,0}(\mathbf{0}) := 0 ,$$

$$\mathfrak{a}_{\sigma,p} := \prod_{i=1}^p a_{\sigma(i)} , \qquad B_{\sigma,p}(\mathbf{h}) := \sum_{i=1}^p (a_{\sigma(i)}h_i)^2 \qquad \text{for } \sigma \in \mathsf{P}_d, \, p \in \{1, ..., d\} ; \qquad (4.91)$$

note that the term with p = 0 in Eq. (4.90) is just $(-1)^d \mathcal{P}_{\frac{u-1}{2}}(0)$ and that all the functions \mathcal{P}_s must be evaluated according to Eq. (4.65). It can be shown that Eq. (4.90) gives

the analytic continuation of $E^u_{(<)}$ to a meromorphic function on the whole complex plane, with simple poles at

$$u \in \{1, \dots, d+1\} ; \tag{4.92}$$

in particular, the series in Eq. (4.90) converges for all $u \in \mathbb{C} \setminus \{1, ..., d+1\}$ and explicit remainder estimates can be derived, as well.

Summing up, u = 0 is a regular point for the analytic continuations of both $E_{(>)}^u$ and $E_{(<)}^u$ so that, the general prescription in Eq. (3.125) yields

$$E^{ren} = E^u_{(>)}\Big|_{u=0} + E^u_{(<)}\Big|_{u=0}$$
(4.93)

where the two addenda on the right-hand side simply indicate the expressions (4.89) and (4.90) evaluated at u = 0.

Before moving on, let us also mention that results analogous to those reported previously concerning the convergence rate of the expansions (4.62) (4.63) for the functions $\mathcal{A}_{(>)}^{-s}(,)$ and $\mathcal{A}_{(<)}^{-s}(,)$ can be derived for the series representations (4.89) (4.90) of $E_{(>)}^{u}$, $E_{(<)}^{u}$. In particular, quantitative remainder estimates can be derived for the truncation of these series at any order $N \in \mathbb{N}$; we refer to [67] for more details.

Scaling considerations.

Let us consider the d-tuple \mathbf{x}_{\star} and the (d-1)-tuple $\boldsymbol{\rho}$ with components, respectively,

$$x_{\star}^{i} := \frac{x^{i}}{a_{i}} \in (0,1) \quad \text{for } i \in \{1,...,d\} , \qquad \rho_{i} := \frac{a_{i}}{a_{1}} \quad \text{for } i \in \{2,...,d\} .$$
(4.94)

Using the decomposition (4.61) of the Dirichlet kernel $\mathcal{A}^{-s}(,)$ and recalling the series expansions (4.62), (4.63) for the functions $\mathcal{A}^{-s}_{(>)}(,)$ and $\mathcal{A}^{-s}_{(<)}(,)$, it can be shown that

$$\mathcal{A}^{-s}(\mathbf{x}, \mathbf{y}) = a_1^{-(d-2)s} \mathfrak{D}_s(\boldsymbol{\rho}; \mathbf{x}_\star, \mathbf{y}_\star) , \qquad (4.95)$$

for all $\mathbf{x}, \mathbf{y} \in \Omega$ and for some suitable function \mathfrak{D}_s (¹¹). From the above relation it follows that, for any pair z, w of spatial variables, there also holds

$$\partial_{zw} \mathcal{A}^{-s}(\mathbf{x}, \mathbf{y}) = a_1^{-(d-2)s-2} \partial_{z_\star w_\star} \mathfrak{D}_s(\boldsymbol{\rho}; \mathbf{x}_\star, \mathbf{y}_\star) .$$
(4.96)

Due to the above results, analogous considerations can be deduced for the zeta-regularized VEV of the observables mentioned in the previous paragraphs.

In particular, from Eq.s (4.78-4.81) it can be easily inferred the following relation for each component of the regularized stress-energy VEV $(\mu, \nu \in \{0, ..., d\})$:

$$(\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\mathbf{x}) \boldsymbol{v}) = a_{1}^{u-d-1} \operatorname{T}^{u}_{\mu\nu}(\boldsymbol{\rho}; \mathbf{x}_{\star}) , \qquad (4.97)$$

¹¹Let us point out that for d = 1 the variables ρ_i are not defined and $\mathfrak{D}_s(\boldsymbol{\rho}; \mathbf{x}_\star, \mathbf{y}_\star) \equiv \mathfrak{D}_s(\mathbf{x}_\star, \mathbf{y}_\star)$ only depends on the complex parameter s and on the rescaled spatial variables $\mathbf{x}_\star, \mathbf{y}_\star$.

where $T^{u}_{\mu\nu}$ is a suitable function.

Similarly, for the regularized pressure acting on any point **x** in the interior of the side $\pi_{1,0}$, we deduce from Eq.s (4.83) (4.97) that

$$p_i^u(\mathbf{x}) = a_1^{u-d-1} p_i^u(\boldsymbol{\rho}; \mathbf{x}_{\star}) \quad \text{for } i \in \{1, ..., d\} , \qquad (4.98)$$

where p_i^u are suitable functions and \mathbf{x}_{\star} is defined as in Eq. (4.94) at points on the boundary. Clearly, the same conclusions can be drawn for the pressure on any other side $\pi_{p,\alpha}$ ($p \in \{1, ..., d\}$, $\alpha \in \{0, 1\}$).

Analogous considerations hold for the regularized bulk energy E^{u} ; from the expansions (4.88-4.90), it can be easily infered that (indicating with E^{u} a suitable function)

$$E^{u} = a_{1}^{u-1} E^{u}(\boldsymbol{\rho}) . \tag{4.99}$$

By analytic continuation at u = 0, we obtain the renormalized counterparts of the above relations: more precisely, we have

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{x}) \boldsymbol{v})_{ren} = a_1^{-(d+1)} \operatorname{T}_{\mu\nu}(\boldsymbol{\rho}; \mathbf{x}_{\star}) ,$$

$$p_i^{ren}(\mathbf{x}) = a_1^{-(d+1)} \operatorname{p}_i(\boldsymbol{\rho}; \mathbf{x}_{\star}) , \qquad E^{ren} = a_1^{-1} \operatorname{E}(\boldsymbol{\rho}) ,$$
(4.100)

where the right-hand sides of the above relations are obtained evaluating at u = 0 the functions in the right-hand sides of Eq.s (4.97-4.99).

Due to the remarks of this paragraph, for any spatial dimension d the analysis of the renormalized stress-energy VEV, pressure and total energy can always be reduced to the case $a_1 = 1$; we will use this fact in the next paragraph dealing with the case d = 2.

The previous results in spatial dimension d = 2.

As an application of the general framework developed in the previous paragraphs, let us consider the 2-dimensional case where

$$d = 2$$
, $\Omega = (0, a_1) \times (0, a_2)$ $(a_1, a_2 > 0)$. (4.101)

In our computations we fix

$$a_1 = 1$$
 (4.102)

and consider different values of a_2 ; let us repeat that this choice causes no loss of generality due to the scaling considerations discussed in the previous paragraph. Moreover, we present the final results in terms of the rescaled coordinates $x_{\star}^1 := x^1/a_1 \equiv x^1, x_{\star}^2 := x^2/a_2 \in (0,1)$, defined in Eq. (4.94) (¹²).

The basic elements to compute the renormalized stress-energy VEV and the pressure are the Dirichlet functions $\mathcal{A}_{(>)}^{-s}(\ ,\)$, $\mathcal{A}_{(<)}^{-s}(\ ,\)$, along with their spatial derivatives, for

¹² Let us stress that, for $a_1 = 1$, the quantities $(\boldsymbol{v} \mid \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$, p_1^{ren} and E^{ren} (to be discussed hereafter) do in fact coincide with the rescaled analogues $T_{\mu\nu}^{ren}$, \mathbf{E}^{ren} introduced in Eq.s (4.97) (4.98) (4.99). Besides, the lenght a_2 of the second side is identified with the ratio ρ_2 (see Eq. (4.94)).

which we use the truncated expansions (4.74), (4.76) and the remainder bounds of Eq.s (4.75), (4.77) (plus similar results for the derivatives; see [67]). Needless to say, analogous considerations also hold for the renormalized bulk energy.

The renormalized stress-energy VEV. As an example, we compute this observable for the two configurations with

$$a_2 = 1$$
 and $a_2 = 5$. (4.103)

In these cases, for the parameter T of the decomposition into (>) and (<) parts and for the truncation order N, we make the following choices:

$$T = 1 , \quad N = 7 \qquad \text{for } a_2 = 1 ; T = 1 , \quad N = 9 \qquad \text{for } a_2 = 5 .$$
 (4.104)

Furthermore, the truncation errors in Eq.s (4.75-4.77) are evaluated making for the parameter α appearing therein the choice

$$\alpha = 0.04$$
 . (4.105)

The renormalized stress-energy VEV $(\boldsymbol{v} | \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$ is obtained setting u = 0 in Eq.s (4.78-4.81). We separate the conformal and nonconformal parts, respectively indicated by the superscripts (\Diamond) and (\blacksquare); recall that Eq. (3.127) gives, in the two-dimensional case,

$$\xi_2 = \frac{1}{8} \ . \tag{4.106}$$

In the following we present, as examples, the graphs for $(\boldsymbol{v} \mid \hat{T}_{00}^{(\Diamond)} \boldsymbol{v})_{ren}$ and $(\boldsymbol{v} \mid \hat{T}_{00}^{(\blacksquare)} \boldsymbol{v})_{ren}$ obtained from the previous truncated expansions; more precisely, Fig.s 4.3 and 4.4 show, respectively, the results obtained for the configurations with $a_2 = 1$ and $a_2 = 5$. In the cited figures we refer to the variables $x_{\star}^i := x^i/a_i \in (0, 1)$ and, keeping into account some obvious symmetry considerations (¹³), we only show the graphs for

$$x^i_{\star} \in (0, 1/2) \quad \text{for } i \in \{1, 2\} .$$
 (4.107)

¹³Indeed, every component of the stress-energy VEV can be shown to be symmetric under the exchange $x^i \leftrightarrow a_i - x^i$ (or $x^i_{\star} \leftrightarrow 1 - x^i_{\star}$) for $i \in \{1, 2\}$.



Figure 4.3: Graphs of $(\boldsymbol{v} \mid \hat{T}_{00}^{(\Diamond)} \boldsymbol{v})_{ren}$ and $(\boldsymbol{v} \mid \hat{T}_{00}^{(\blacksquare)} \boldsymbol{v})_{ren}$ for $a_2 = 1$ (d = 2).



Figure 4.4: Graphs of $(\boldsymbol{v} \mid \hat{T}_{00}^{(\Diamond)} \boldsymbol{v})_{ren}$ and $(\boldsymbol{v} \mid \hat{T}_{00}^{(\blacksquare)} \boldsymbol{v})_{ren}$ for $a_2 = 5$ (d = 2).

Concerning the error estimates, for $\mu, \nu \in \{0, 1, 2\}$ and $\bullet \in \{\diamondsuit, \blacksquare\}$, let us introduce the following notation:

$$\mathcal{E}_{\mu\nu} := \frac{\text{remainder corresponding to our approximation}}{\text{by truncation of } (\boldsymbol{v} \mid \hat{T}_{\mu\nu} \boldsymbol{v})_{ren} \,.}$$
(4.108)

On the one hand, for $a_2 = 1$, our choice N = 7 (T = 1, $\alpha = 0.04$) yields the uniform bound

$$|\mathcal{E}_{\mu\nu}| \le 3 \cdot 10^{-11} \quad \text{for } \mu, \nu \in \{0, 1, 2\} .$$
 (4.109)

On the other hand, for $a_2 = 5$, our choice N = 9 (T = 1, $\alpha = 0.04$) ensures

$$|\mathcal{E}_{\mu\nu}| \le 8 \cdot 10^{-11}$$
 for $\mu, \nu \in \{0, 1, 2\}$. (4.110)

The renormalized pressure on the boundary. As in the construction of the general theory we consider, as an example, the pressure $p_i^{ren}(\mathbf{x})$ at points $\mathbf{x} \equiv (0, x^2)$ in the interior of the side $\pi_{1,0}$, making reference to the prescription (4.85). Fig. 4.5 shows the graphs obtained for $p_1^{ren}(\mathbf{x})$ as a function of $x_{\star}^2 := x^2/a_2$ (again, choosing T = 1 and truncating the related expansions to order N = 7, for $a_2 = 1$, and N = 9, for $a_2 = 5$).



As for the error, indicating with ϵ_1 the remainder associated to our approximation by truncation of p_1^{ren} , we obtain the following uniform estimates (setting $\alpha = 0.04$):

$$\begin{aligned} |\epsilon_1| &\leq 2 \cdot 10^{-12} & \text{for } a_2 = 1 ; \\ |\epsilon_1| &\leq 5 \cdot 10^{-12} & \text{for } a_2 = 5 . \end{aligned}$$
(4.111)

Before moving on, let us mention that the renormalized pressure p_1^{ren} can be proved to possess the following asymptotic behaviour near the edge $\mathbf{x} = \mathbf{0}$, for all $a_2 > 0$:

$$p_1^{ren}(\mathbf{x}) = \frac{1}{32\pi (x^2)^3} + O((x^2)^2) \quad \text{for } \mathbf{x} = (0, x^2) \text{ and } x^2 \to 0^+ .$$
 (4.112)

The renormalized bulk energy. Let us fix again $a_1 = 1$ and consider this observable for different values of a_2 . Using the decomposition into (>) and (<) parts and truncating the corresponding series at order N = 50 (with T = 1, $\alpha = 0,04$), one can plot E^{ren} as a function of a_2 (see Figure 4.6); with the choices made for the parameters a_1, T, N, α , it can be proved that the error is smaller than $2 \cdot 10^{-3}$ for any $a_2 \in [0.05, 10]$). Let us discuss some facts regarding the function $a_2 \mapsto E^{ren}(a_2)$, which can be read from the graph in Fig. 4.6 (the results reported hereafter are obtained using standard numerical methods, implemented in Mathematica).

i) There is only one point of maximum a_2^{\max} ; our approximation by truncation at order N = 50 gives

$$a_2^{\max} = 0.72719110 \pm 10^{-8}$$
, $E^{ren}(a_2^{\max}) = 0.04472675 \pm 10^{-8}$. (4.113)



Figure 4.6: Graph of E^{ren} as function of a_2 (d = 2).

ii) E^{ren} vanishes for two values $\bar{a}_2^{(1)} < \bar{a}_2^{(2)}$ of a_2 ; these are found to be

$$\bar{a}_2^{(1)} = 0.36538151 \pm 10^{-8} , \qquad \bar{a}_2^{(2)} = 2.73686534 \pm 10^{-8} .$$
 (4.114)

 E^{ren} is positive for $\bar{a}_2^{(1)} < a_2 < \bar{a}_2^{(2)}$ and negative elsewhere. This feature was also pointed out in [102]; therein it is stated that $\bar{a}_2^{(2)} = (\bar{a}_2^{(1)})^{-1}$, a relation (approximately) verified by the numerical values in Eq. (4.114).

iii) For $a_2 \to 0^+$, E^{ren} has the asymptotic behaviour

$$E^{ren}(a_2) = \frac{e_0}{(a_2)^2} \left(1 + O(a_2) \right) \quad \text{with} \quad e_0 = -0.02391 \pm 10^{-5} .$$
 (4.115)

iv) There are indications that E^{ren} approaches an asymptote for $a_2 \to +\infty$; taking into account values of the abscissa up to $a_2 = 100$, we find that this asymptote is the straight line

$$y = m_E a_2 + q_E$$
 with $\begin{cases} m_E = -0.02391416 \pm 10^{-8} \\ q_E = +0.06544985 \pm 10^{-8} \end{cases}$ (4.116)

Finally, let us mention that the numerical values of E^{ren} given by our previous analysis are in good agreement with those arising from the expansions in [25], a fact strongly indicating the equivalence between our approach and [25]. Let us also mention that the results of [25] about E^{ren} are equivalent to the ones of [11, 57, 77].

4.3 The case of parallel planes with Robin boundary conditions.

In the present conclusive section we consider a variant of the original setting studied by Casimir in his seminal paper [35], involving a massless field confined between two parallel planes. Let us recall that we already analyzed this type of configuration in our previous work [65] (see also [63] for the three-dimensional case with Dirichlet boundary conditions

and [64] for the one-dimensional analogue of a segment); therein, working in arbitrary odd spatial dimension d (also greater than d = 3), we considered several kinds of boundary conditions, namely of Dirichlet, Neumann, mixed $(^{14})$ and periodic type. For each one of these specific models we computed the renormalized VEVs of several observables. To this purpose, using the methods described in [64] and reformulated more rigorously in the present manuscript, in [65] we developed a series of general computational rules, independent of the boundary conditions prescribed on the (hyper-)planes; these general rules rely, in particular, on the fact that the configuration under analysis is of *slab type* (¹⁵) (so that it suffices to study the reduced one-dimensional problem of a segment) and on the fact that the corresponding cylinder kernel possesses some general suitable features. In the subsequent paragraphs we first recall briefly the above mentioned general rules and then use them to study the case of a massless field fulfilling a particular type of Robin conditions on the boundary planes. In the specific setting under analysis, it is possible to derive an explicit integral representation for the cylinder kernel corresponding to the reduced one-dimensional model; this representation allows to infer that the mentioned kernel fulfills the hypotheses of Theorem 2.93, so that the analytic continuation of the related Dirichlet kernel at the points of interest can be computed by means of the residue theorem (see, in particular, Eq. (2.361)). Finally, these facts are used to obtain exact expressions for the renormalized VEV of the stress-energy tensor (we plan to discuss the total vacuum energy per unit area and the pressure on the boundary elsewhere [68]).

Needless to say, the literature on the Casimir effect for the configuration with two parallel planes is immense, both regarding local and global aspects; following [65], here we only cite a few references. In his seminal paper [35], using an exponential cut-off regularization along with Abel-Plana resummation, Casimir was the first to compute the total energy and the boundary forces for the case of two parallel planes; concerning local aspects, the foremost derivation of the full stress-energy tensor VEV was given by Brown and Maclay [29], using a point-splitting technique. Computation of both global and local quantities for this model was later reproposed by several authors, using various regularization techniques: see, e.g., the monographies by Milton [110], Elizalde et al. [56, 57], Bordag et al. [25] (see, as well, the works cited therein).

The first to address the case of Robin boundary conditions were Romeo and Saharian [133], who used a generalized version of the Abel-Plana formula to obtain the renormalized

$$\Omega = \Omega_1 \times \mathbb{R}^{d_2} \ni \mathbf{x} \equiv (\mathbf{x}_1, \mathbf{x}_2) , \qquad V(\mathbf{x}) = V(\mathbf{x}_1)$$

¹⁴Here the nomenclature "mixed boundary conditions" is used to indicate the case where Dirichlet conditions are prescribed on one of the planes, while Neumann conditions are prescribed on the other.

¹⁵By definition, this means that

with $\Omega_1 \subset \mathbb{R}^{d_1}$ $(d_1 + d_2 = d)$ a suitable domain, and that the boundary conditions refer to $\partial\Omega_1 \times \mathbb{R}^{d_2}$. In this case the relevant operators are $\mathcal{A}_1 := -\Delta_1 + V$ acting in $\mathcal{H}_1 := L^2(\Omega_1)$, $\mathcal{A}_2 := -\Delta_2$ acting in $\mathcal{H}_2 := L^2(\mathbb{R}^{d_2})$ and $\mathcal{A} = \mathcal{A}_1 \otimes \mathbb{I}_2 + \mathbb{I}_1 \otimes \mathcal{A}_2 = -\Delta + V(\mathbf{x}_1)$ acting in $\mathcal{H} = \mathcal{H}_1 \otimes \mathcal{H}_2 = L^2(\Omega)$ (with \mathbb{I}_1 , \mathbb{I}_2 indicating the identity operators on $L^2(\Omega_1)$ and $L^2(\mathbb{R}^{d_2})$, respectively). See [64] for more details; we refer, in particular, to subsection 3.18 and to the related Appendix E.

stress-energy VEV and zeta techniques to compute the renormalized total vacuum energy. Analogous configurations were later considered by Saharian et al. [137] (dealing with uniformly accelerated plates, described using Rindler coordinates) and by Setare [144] (treating a case with de Sitter background spacetime); see also [9]. All these works use primarly regularization techniques different from the zeta approach; on the contrary, the analysis we present hereafter only relies on ZR in the general formulation described in the previous chapters of the present manuscript.

Introducing the problem.

We consider the d-dimensional configuration where

$$\Omega := (0, a) \times \mathbb{R}^{d-1} \quad (a > 0) , \qquad V = 0 ; \qquad (4.117)$$

the above choices correspond to a massless scalar field confined between two parallel hyperplanes set at a distance a, with no background potential. The boundary $\partial\Omega$ of the spatial domain is composed by the planes

$$\pi_0 = \{ \mathbf{x} \in \mathbb{R}^d \mid x^1 = 0 \} , \qquad \pi_a = \{ \mathbf{x} \in \mathbb{R}^d \mid x^1 = a \} .$$
(4.118)

Boundary conditions of Robin type are generically extablished setting, for some given parameters $\beta_0, \beta_a \in \mathbb{R}$,

$$(1 + \beta_0 \partial_{\mathbf{n}}) \hat{\varphi} \Big|_{\pi_0} = 0 , \qquad (1 + \beta_a \partial_{\mathbf{n}}) \hat{\varphi} \Big|_{\pi_a} = 0 \qquad (4.119)$$

(here, as usual, $\partial_{\mathbf{n}}$ indicates the derivative in the outer direction normal to the boundary). The case where $\beta_0 \neq -\beta_a$ can only be treated by perturbation theory (¹⁶); on the contrary, in the following we focus the attention to the specific setting where

$$\beta_0 = -\beta_a \equiv \beta \in (0, +\infty) , \qquad (4.120)$$

since in this case it is possible to perform a fully explicit analysis. In view of the above considerations, the boundary conditions (4.119) reduce to

$$(1 - \beta \partial_{x^1}) \hat{\varphi} \Big|_{\pi_0} = 0 , \qquad (1 - \beta \partial_{x^1}) \hat{\varphi} \Big|_{\pi_a} = 0 .$$
 (4.121)

In passing, let us point out that the above constraints correspond to the usual Dirichlet conditions for $\beta = 0$, while they formally reduce to Neumann conditions in the limit of large β (¹⁷). Moreover, the case with $\beta < 0$ can be recovered at the end by obvious symmetry arguments, making the replacement $x^1 \mapsto a - x^1$.

¹⁶In fact, in this case, only implicit expressions can be derived for the eigenvalues of the Laplacian, which are a main ingredient for the following developments.

¹⁷By elementary dimensional analysis arguments, it appears that this statement must be interpreted in terms of the adimensional ratio β/a , meaning that the limit $\beta/a \to +\infty$ has to be considered.

Keeping into account the boundary conditions (4.121), also in this case we put

$$\mathcal{A} := -\Delta ; \qquad (4.122)$$

this is strictly positive and self-adjoint on $\mathcal{H} = L^2(\Omega)$, with domain $\mathcal{D}_{\mathcal{A}} := \{f \in L^2(\Omega) \mid (1 - \beta \partial_{x^1}) f \in H^1(\Omega), \Delta f \in L^2(\Omega) \text{ and } (1 - \beta \partial_{x^1}) f \upharpoonright \pi_0 = (1 - \beta \partial_{x^1}) f \upharpoonright \pi_a = 0\}$ (¹⁸). Again, \mathcal{A} is an *admissible operator* in the sense of definition 2.37; moreover, its spectrum can be explicitly determined and is given by

$$\sigma(\mathcal{A}) = \left\{ \omega_{\mathbf{k},n}^2 := |\mathbf{k}|^2 + \frac{n^2 \pi^2}{a^2} \mid \mathbf{k} \in \mathbb{R}^{d-1}, \ n \in \{1, 2, 3, ...\} \right\}$$
(4.123)

(¹⁹). In view of the above considerations, the general framework of Chapter 2 can be employed also in this case; in particular, the Dirichlet kernel associated to \mathcal{A} and its derivatives are well-defined integral kernels.

Let us stress that the setting under analysis is, in fact, a *slab* configuration; we discussed in detail this type of configurations in our previous work [64] (see in particular subjection 3.18 therein), giving some general rules allowing to compute the Dirichlet kernel and its derivatives evaluated along the diagonal in terms of the integral kernels associated to a lower dimensional problem. In the subsequent paragraph we briefly recall the main tools related to the above considerations, required for the evaluation of the stress-energy VEV to be described in the following.

Before proceeding to this topic, let us point out the following fact.

Remark 4.2. Throughout this section we assume

$$d \text{ odd}, \quad d \ge 3;$$
 (4.124)

this hypothesis is purely technical and will be motivated later (see the comments before Eq. (4.142)). The case d = 1, here excluded, could be discussed using a slight variation of the computational rules presented here (compare also with the analysis described in Section 6 of [64]); for brevity, we defer this topic to a future work [68]. Let us also anticipate that at the end of the present section we will consider more in detail the case d = 3, performing the required computations explicitly.

Reduction to a one-dimensional problem.

As mentioned previously, it appears from Eq. (4.117) that the configuration under analysis is of slab type (see the footnote 15 on page 184 and the references given therein). In particular, in the present setting, we have

$$\Omega = \Omega_1 \times \mathbb{R}^{d-1} \quad \text{with } \Omega_1 = (0, a) \subset \mathbb{R} \qquad (\text{plus, } V = 0) ; \tag{4.125}$$

186

¹⁸Similarly to what was said in the Footnote 7 of page 169, using the eigenfunction expansion of \mathcal{A} , it can be shown that there holds $\mathcal{D}_{\mathcal{A}} = \{f \in H^2(\Omega) \mid (1 - \beta \partial_{x^1})f \upharpoonright \pi_0 = (1 - \beta \partial_{x^1})f \upharpoonright \pi_a = 0\}.$

¹⁹In this case the parameter $\varepsilon > 0$ fulfilling $\sigma(\mathcal{A}) \subset [\varepsilon, +\infty)$, can be chosen to be $\varepsilon := (\pi/a)^2$.

4.3. THE CASE OF PARALLEL PLANES WITH ROBIN BOUNDARY CONDITIONS.

therefore, the admissible operator \mathcal{A} on $L^2(\Omega)$ can be expressed as (with the domain specifications provided by the subsequent considerations)

$$\mathcal{A} = \mathcal{A}_1 \otimes \mathbb{I}_2 + \mathbb{I}_1 \otimes \mathcal{A}_2 , \qquad (4.126)$$

where we have introduced the two operators

$$\mathcal{A}_1 := -\partial_{x^1 x^1} : \operatorname{Dom}(\mathcal{A}_1) \subset L^2(0, a) \to L^2(0, a) ,$$

$$\mathcal{A}_2 := -\sum_{i=2}^d \partial_{x^i x^i} : \operatorname{Dom}(\mathcal{A}_2) \subset L^2(\mathbb{R}^{d-1}) \to L^2(\mathbb{R}^{d-1}) .$$
(4.127)

Here the domains of definitions $\text{Dom}(\mathcal{A}_1), \text{Dom}(\mathcal{A}_2)$ are determined by the boundary conditions described in Eq. (4.121); more precisely, we have

$$Dom(\mathcal{A}_1) := \{ f \in L^2(0, a) \mid \partial_{x^1 x^1} f \in L^2(0, a), \ (1 - \beta \, \partial_{x^1}) f(0) = (1 - \beta \, \partial_{x^1}) f(a) = 0 \} ,$$
$$Dom(\mathcal{A}_2) := \{ f \in L^2(\mathbb{R}^{d-1}) \mid \partial_{x^1 x^1} f \in L^2(\mathbb{R}^{d-1}) \} .$$
(4.128)

$$(^{20})$$
. Let us stress that \mathcal{A}_1 is itself an admissible operator in the sense of Definition 2.37. Furthermore, it has purely discrete spectrum

$$\sigma(\mathcal{A}_1) = \left\{ \omega_n^2 := \frac{n^2 \pi^2}{a^2} \mid n \in \{1, 2, 3, ...\} \right\} , \qquad (4.129)$$

and it possesses a complete orthonormal set of eigenfunctions in $L^2(0, a)$, corresponding to the eigenvalues (ω_n^2) in Eq. (4.129), given by

$$F_n(x^1) := \sqrt{\frac{2}{a(1+\beta^2\omega_n^2)}} \left(\sin(\omega_n x^1) + \beta\,\omega_n\,\cos(\omega_n x^1)\right) \qquad (x^1 \in (0,a)) \ . \tag{4.130}$$

In view of the above considerations, we can resort to the general framework of Chapter 2, to speak about the integral kernels associated to \mathcal{A}_1 and about the related traces. In particular, in accordance with the general notations adopted in the present manuscript, we indicate the reduced Dirichlet and cylinder kernel with $\mathcal{A}_1^{-s}(x^1, y^1)$ and $e^{-t\sqrt{\mathcal{A}_1}}(x^1, y^1)$ (for $x^1, y^1 \in (0, a)$), respectively; similarly, the corresponding traces will be denoted with $\mathrm{Tr} \, \mathcal{A}_1^{-s}$ and $\mathrm{Tr} \, e^{-t\sqrt{\mathcal{A}_1}}$.

General relations between $\mathcal{A}^{-s}(,)$ and $\mathcal{A}_1^{-s}(,)$. It can be proved that the Dirichlet kernel $\mathcal{A}^{-s}(\mathbf{x}, \mathbf{y})$ evaluated along the diagonal $\mathbf{y} = \mathbf{x}$ can be expressed in terms of the reduced kernel $\mathcal{A}_1^{-s}(x^1, y^1)$ at $y^1 = x^1$; analogous relations can be derived also for the

²⁰For any $f \in \text{Dom}(\mathcal{A}_1)$ it follows that $\partial_{x^1} f \in H^1(0, a)$, so that $\partial_{x^1} f(0)$ and $\partial_{x^1} f(a)$ are well defined. Moreover, also in this case, using the eigenfunction expansion of \mathcal{A}_1 and \mathcal{A}_2 it can be shown that $\text{Dom}(\mathcal{A}_1) = \{f \in H^2(0, a) \mid (1 - \beta \partial_{x^1}) f(0) = (1 - \beta \partial_{x^1}) f(a) = 0\}.$

derivatives. Here, we are referring to Eq.s (3.115–3.118) of [64]; in the present setting, for any $u \in \Sigma_d$, these relations reduce to

$$\mathcal{A}^{-\frac{u-1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = \frac{\Gamma(\frac{u-d}{2})}{(4\pi)^{\frac{d-1}{2}}\Gamma(\frac{u-1}{2})} \mathcal{A}_{1}^{-\frac{u-d}{2}}(x^{1},y^{1})\Big|_{y^{1}=x^{1}}; \qquad (4.131)$$

$$\partial_{x^{i}y^{j}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = \partial_{x^{i}x^{j}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = \partial_{y^{i}y^{j}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = 0$$

for $i = 1$ and $j \in \{2, ..., d\}$ or $i \in \{2, ..., d\}$ and $j = 1$;
$$(4.132)$$

$$\partial_{zw}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = \frac{\Gamma(\frac{u-d+2}{2})}{(4\pi)^{\frac{d-1}{2}}\Gamma(\frac{u+1}{2})} \partial_{zw}\mathcal{A}_{1}^{-\frac{u-d+2}{2}}(x^{1},y^{1})\Big|_{y^{1}=x^{1}} \text{ for } z, w \in \{x^{1},y^{1}\}; (4.133)$$

$$\partial_{x^{i}y^{j}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = -\partial_{x^{i}x^{j}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = -\partial_{y^{i}y^{j}}\mathcal{A}^{-\frac{u+1}{2}}(\mathbf{x},\mathbf{y})\Big|_{\mathbf{y}=\mathbf{x}} = \delta_{ij} \frac{\Gamma(\frac{u-d}{2})}{2(4\pi)^{\frac{d-1}{2}}\Gamma(\frac{u+1}{2})} \mathcal{A}_{1}^{-\frac{u-d}{2}}(x^{1},y^{1})\Big|_{y^{1}=x^{1}} \quad \text{for } i,j \in \{2,3,...,d\} .$$

$$(4.134)$$

The reduced cylinder kernel $e^{-\mathbf{t}\sqrt{A_1}}(,)$. With some computational effort (see the Appendix), it is possible to express this integral kernel as follows, for all $x^1, y^1 \in (0, a)$ and all $\mathbf{t} \in (0, +\infty)$:

$$e^{-\mathbf{t}\sqrt{\mathcal{A}_{1}}}(x^{1},y^{1}) = T_{0}(\mathbf{t};x^{1},y^{1}) + T_{1}(\mathbf{t};x^{1},y^{1}) + T_{2}(\mathbf{t};x^{1},y^{1}) \quad \text{with}$$

$$T_{0}(\mathbf{t};x^{1},y^{1}) = \frac{1}{2a} \left[\frac{\cos(\frac{\pi}{a}(x^{1}-y^{1})) - e^{-\frac{\pi}{a}\mathbf{t}}}{\cosh(\frac{\pi}{a}\mathbf{t}) - \cos(\frac{\pi}{a}(x^{1}-y^{1}))} - \frac{\cos(\frac{\pi}{a}(x^{1}+y^{1})) - e^{-\frac{\pi}{a}\mathbf{t}}}{\cosh(\frac{\pi}{a}\mathbf{t}) - \cos(\frac{\pi}{a}(x^{1}+y^{1}))} \right],$$

$$T_{1}(\mathbf{t};x^{1},y^{1}) = \int_{0}^{+\infty} d\mathbf{v} \, \cos\mathbf{v} \, S_{1}(\mathbf{t}+|\beta|\mathbf{v};x^{1},y^{1}) ,$$

$$T_{2}(\mathbf{t};x^{1},y^{1}) = S_{2}(\mathbf{t};x^{1},y^{1}) - \int_{0}^{+\infty} d\mathbf{v} \, \sin\mathbf{v} \, S_{2}(\mathbf{t}+|\beta|\mathbf{v};x^{1},y^{1}) ,$$
(4.135)

where the functions S_1, S_2 are given by

$$S_1(\mathbf{t}; x^1, y^1) := \frac{1}{a} \left[\frac{\sin(\frac{\pi}{a}(x^1 + y^1))}{\cosh(\frac{\pi}{a}\mathbf{t}) - \cos(\frac{\pi}{a}(x^1 + y^1))} \right], \qquad (4.136)$$

$$S_2(\mathbf{t}; x^1, y^1) := \frac{1}{a} \left[\frac{\cos(\frac{\pi}{a}(x^1 + y^1)) - e^{-\frac{\pi}{a}\mathbf{t}}}{\cosh(\frac{\pi}{a}\mathbf{t}) - \cos(\frac{\pi}{a}(x^1 + y^1))} \right].$$
 (4.137)

Before proceeding, let us point out some facts concerning the functions T_0, T_1, T_2 introduced above to evaluate the reduced cylinder kernel $e^{-t\sqrt{A_1}}(,)$.

4.3. THE CASE OF PARALLEL PLANES WITH ROBIN BOUNDARY CONDITIONS.

i) T_0 is the cylinder kernel corresponding to the configuration of a segment (0, a) with Dirichlet boundary conditions (see [64], Eq. 6.20). Notice in particular that, for any fixed $x^1, y^1 \in (0, a)$, the map $(0, +\infty) \ni \mathbf{t} \mapsto T_0(\mathbf{t}, x^1, y^1)$ extends to a function which is meromorphic in a complex neighbour of $[0, +\infty)$, with only a pole singularity in $\mathbf{t} = 0$, and which vanishes exponentially for $\Re \mathbf{t} \to +\infty$. Analogous considerations hold for its derivatives. ii) Making the change of integration variable $\mathbf{v} = t + \beta \mathbf{w}$ and using some trivial trigonometric identities, the function T_1, T_2 defined in Eq. (4.135) can be re-expressed as follows:

$$T_1(\mathbf{t}\,;x^1,y^1) = \tag{4.138}$$

$$\frac{1}{\beta} \left[\cos\left(\frac{\mathbf{t}}{\beta}\right) \int_{\mathbf{t}}^{+\infty} d\mathbf{w} \, \cos\left(\frac{\mathbf{w}}{\beta}\right) S_1(\mathbf{w}; x^1, y^1) + \sin\left(\frac{\mathbf{t}}{\beta}\right) \int_{\mathbf{t}}^{+\infty} d\mathbf{w} \, \sin\left(\frac{\mathbf{w}}{\beta}\right) S_1(\mathbf{w}; x^1, y^1) \right] \\ T_2(\mathbf{t}; x^1, y^1) = S_2(\mathbf{t}; x^1, y^1) + \tag{4.139}$$

$$+\frac{1}{\beta} \left[\sin\left(\frac{\mathbf{t}}{\beta}\right) \int_{\mathbf{t}}^{+\infty} d\mathbf{w} \, \cos\left(\frac{\mathbf{w}}{\beta}\right) S_2(\mathbf{w}; x^1, y^1) - \cos\left(\frac{\mathbf{t}}{\beta}\right) \int_{\mathbf{t}}^{+\infty} d\mathbf{w} \, \sin\left(\frac{\mathbf{w}}{\beta}\right) S_2(\mathbf{w}; x^1, y^1) \right]$$

Using the above expressions, it can be shown (see the Appendix) that, for any fixed $x^1, y^1 \in (0, a)$, both the maps $[0, +\infty) \ni \mathbf{t} \mapsto T_1(\mathbf{t}, x^1, y^1), T_2(\mathbf{t}, x^1, y^1)$ extend to functions which are analytic in a complex neighbour of $[0, +\infty)$ and vanish exponentially for $\Re \mathbf{t} \to +\infty$; moreover, their Taylor series in $\mathbf{t} = 0$ can be determined explicitly (see Proposition .4 in the cited appendix). Analogous considerations hold for the spatial derivatives of T_1 and T_2 .

The reduced Dirichlet kernel in terms of the reduced cylinder kernels. In view of the features pointed out above for the reduced cylinder kernel $e^{-t\sqrt{A_1}}(,)$ is appears that this function fulfills the assumptions of Theorem 2.93. Therefore, the analytic continuations of the related Dirichlet kernel $\mathcal{A}_1^{-s}(,)$ and of its derivatives can be determined according to Eq. (2.360); in particular, we have

$$\mathcal{A}_{1}^{-\frac{u-d}{2}}(x^{1}, y^{1}) = \frac{e^{-i\pi(u-d)}\Gamma(d+1-u)}{2\pi i} \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{u-d-1} \, e^{-\mathbf{t}\sqrt{\mathcal{A}_{1}}}(x^{1}, y^{1}) \; ; \qquad (4.140)$$

$$\partial_{zw} \mathcal{A}_{1}^{-\frac{u-d+2}{2}}(x^{1}, y^{1}) = \frac{e^{-i\pi(u-d)} \Gamma(d-1-u)}{2\pi i} \int_{\mathfrak{H}} d\mathbf{t} \, \mathbf{t}^{u-d+1} \, \partial_{zw} e^{-\mathbf{t}\sqrt{\mathcal{A}_{1}}}(x^{1}, y^{1})$$
for $z, w \in \{x^{1}, y^{1}\}$

$$(4.141)$$

(let us recall that \mathfrak{H} denotes the Hankel contour).

The stress-energy tensor.

Recall the expressions (3.107-3.109) giving the zeta-regularized stress-energy VEV in terms of the Dirichlet kernel and of its derivatives. Eq.s (4.131-4.134), along with the expressions (4.140) (4.141) for the reduced Dirichlet functions, allow to obtain integral representations for each component of the above mentioned VEV. These integral representations give the analytic continuation of the map $\Sigma_{d+1} \ni u \mapsto (\boldsymbol{v} \mid \hat{T}_{uv}^u \boldsymbol{v})$ to the whole complex plane; due to the assumption (4.124) on the spatial dimension d, this analytic continuation is regular at u = 0. So, the general prescription (3.124) reduces in the present setting to

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{x}) \boldsymbol{v})_{ren} := (\boldsymbol{v} | \hat{T}^{u}_{\mu\nu}(\mathbf{x}) \boldsymbol{v}) \Big|_{u=0} .$$
(4.142)

Due to the meromorphic nature of the reduced cylinder kernel $e^{-t\sqrt{A_1}}(,)$ (and of its derivatives), the resulting integrals along the Hankel contour can be explicitly evaluated via the residue theorem; the final expressions for the non-vanishing components of the renormalized stress-energy VEV are $(i \in \{2, ..., d\})$

$$(\boldsymbol{v} | \hat{T}_{00}(\mathbf{x}) \boldsymbol{v})_{ren} = -(\boldsymbol{v} | \hat{T}_{ii}(\mathbf{x}) \boldsymbol{v})_{ren} = -C_d \operatorname{Res} \left(\mathbf{t}^{-(d+1)} \left[\left(\xi - \frac{d-2}{4d} \right) d \, e^{-\mathbf{t}\sqrt{\mathcal{A}_1}} (x^1, y^1) + \frac{\mathbf{t}^2}{d-1} \left(\frac{1}{4} - \xi \right) \partial_{x^1 y^1} e^{-\mathbf{t}\sqrt{\mathcal{A}_1}} (x^1, y^1) \right]_{y^1 = x^1}; 0 \right);$$

$$(4.143)$$

$$(\boldsymbol{v} \mid T_{11}(\mathbf{x}) \boldsymbol{v})_{ren} =$$

$$-C_{d} \operatorname{Res}\left(\mathbf{t}^{-(d+1)}\left[\left(\frac{1}{4}-\xi\right)d \ e^{-\mathbf{t}\sqrt{\mathcal{A}_{1}}}(x^{1},y^{1})+\frac{\mathbf{t}^{2}}{d-1}\left(\frac{1}{4} \ \partial_{x^{1}y^{1}}-\xi \ \partial_{x^{1}x^{1}}\right)e^{-\mathbf{t}\sqrt{\mathcal{A}_{1}}}(x^{1},y^{1})\right]_{y^{1}=x^{1}};0\right);$$

$$(4.144)$$

Here, for the sake of brevity, we have put

$$C_d := (-\pi)^{-\frac{d-1}{2}} \Gamma\left(\frac{d+1}{2}\right) \,. \tag{4.145}$$

We repeat that, in the above, d is an arbitrary odd dimension > 1. In the following paragraph we will report the explicit expressions for the renormalized stress-energy components arising from Eq.s (4.143-4.144) in the case of spatial dimension d = 3; again, we will give the final results in the form described in subsection 3.2, separating the conformal and non-conformal parts (see, in particular, Eq. (3.131)), noting that Eq. (3.127) gives

$$\xi_3 = \frac{1}{6} \ . \tag{4.146}$$

The previous results in spatial dimension d = 3.

As an example, let us compute the renormalized stress-energy VEV in the case where d = 3. To this purpose, recall the explicit expression (4.135) for the reduced cylinder kernel $e^{-\mathbf{t}\sqrt{\mathcal{A}_1}}(,)$ (along with the results of the Appendix) regarding its Laurent series at $\mathbf{t} = 0$; then, separating the conformal (\Diamond) and non-conformal (\blacksquare) parts, in the present setting the general identities (4.143-4.144) yield

$$(\boldsymbol{v} \mid \hat{T}_{00}^{(\Diamond)}(\mathbf{x}) \, \boldsymbol{v})_{ren} = - (\boldsymbol{v} \mid \hat{T}_{22}^{(\Diamond)}(\mathbf{x}) \, \boldsymbol{v})_{ren} = - (\boldsymbol{v} \mid \hat{T}_{33}^{(\Diamond)}(\mathbf{x}) \, \boldsymbol{v})_{ren} =$$

$$(4.147)$$

$$- \frac{\pi^2}{1440a^4} - \frac{1}{24\pi a^4} \left[\left(\frac{a}{\beta} \right)^2 \frac{\pi}{2 \sin^2(\frac{\pi x^1}{a})} - \left(\frac{a}{\beta} \right)^3 \frac{\cos(\frac{\pi x^1}{a})}{\sin(\frac{\pi x^1}{a})} +$$

$$+ \left(\frac{a}{\beta} \right)^4 \int_0^{+\infty} d\mathbf{t} \left(1 - \beta^2 \partial_{x^1 y^1} \right) \left(\cos \left(\frac{\mathbf{t}}{\beta} \right) S_2(\mathbf{t} ; x^1, y^1) + \sin \left(\frac{\mathbf{t}}{\beta} \right) S_1(\mathbf{t} ; x^1, y^1) \right)_{y^1 = x^1} \right],$$

$$(\boldsymbol{v} \mid \hat{T}_{00}^{(\bullet)}(\mathbf{x}) \, \boldsymbol{v})_{ren} = - (\boldsymbol{v} \mid \hat{T}_{22}^{(\bullet)}(\mathbf{x}) \, \boldsymbol{v})_{ren} = - (\boldsymbol{v} \mid \hat{T}_{33}^{(\bullet)}(\mathbf{x}) \, \boldsymbol{v})_{ren} =$$

$$(4.148)$$

$$- \frac{\pi^2 (3 - 2 \sin^2(\frac{\pi}{a} x^1))}{8a^4 \sin^4(\frac{\pi}{a} x^1)} + \frac{1}{2\pi a^4} \left[\left(\frac{a}{\beta} \right) \frac{\pi^2 \cos(\frac{\pi}{a} x^1)}{\sin^3(\frac{\pi}{a} x^1)} - \left(\frac{a}{\beta} \right)^2 \frac{\pi}{2 \sin^2(\frac{\pi}{a} x^1)} + \left(\frac{a}{\beta} \right)^3 \frac{\cos(\frac{\pi}{a} x^1)}{\sin(\frac{\pi}{a} x^1)} +$$

$$+ \left(\frac{a}{\beta} \right)^4 \int_0^{+\infty} d\mathbf{t} \left(1 + \beta^2 \partial_{x^1 y^1} \right) \left(\cos \left(\frac{\mathbf{t}}{\beta} \right) S_2(\mathbf{t} ; x^1, y^1) + \sin \left(\frac{\mathbf{t}}{\beta} \right) S_1(\mathbf{t} ; x^1, y^1) \right)_{y^1 = x^1} \right],$$

$$(\boldsymbol{v} \mid \hat{T}_{11}^{(\Diamond)}(\mathbf{x}) \, \boldsymbol{v})_{ren} =$$

$$(4.149)$$

$$- \frac{3\pi^2}{1440a^4} - \frac{1}{24\pi a^4} \left[\left(\frac{a}{\beta} \right)^2 \frac{\pi}{2 \sin^2(\frac{\pi x^1}{a})} - \left(\frac{a}{\beta} \right)^3 \frac{\cos(\frac{\pi x^1}{a})}{\sin(\frac{\pi x^1}{a})} +$$

$$+ \left(\frac{a}{\beta}\right)^{4} \int_{0}^{+\infty} d\mathbf{t} \left(1 - 3\beta^{2} \partial_{x^{1}y^{1}} + 2\beta^{2} \partial_{x^{1}x^{1}}\right) \left(\cos\left(\frac{\mathbf{t}}{\beta}\right) S_{2}(\mathbf{t}\,;x^{1},y^{1}) + \sin\left(\frac{\mathbf{t}}{\beta}\right) S_{1}(\mathbf{t}\,;x^{1},y^{1})\right)_{y^{1}=x^{1}} \right],$$
$$(\boldsymbol{v} \mid \hat{T}_{11}^{(\blacksquare)}(\mathbf{x}) \, \boldsymbol{v})_{ren} =$$
(4.150)

$$\frac{1}{2\pi a^4} \left[\left(\frac{a}{\beta}\right)^2 \frac{\pi}{2\sin^2(\frac{\pi}{a}x^1)} - \left(\frac{a}{\beta}\right)^3 \frac{\cos(\frac{\pi}{a}x^1)}{\sin(\frac{\pi}{a}x^1)} + \left(\frac{a}{\beta}\right)^4 \int_0^{+\infty} d\mathbf{t} \left(1 - \beta^2 \partial_{x^1x^1}\right) \left(\cos\left(\frac{\mathbf{t}}{\beta}\right) S_2(\mathbf{t}\,;x^1,y^1) + \sin\left(\frac{\mathbf{t}}{\beta}\right) S_1(\mathbf{t}\,;x^1,y^1) \right)_{y^1 = x^1} \right]$$

(here S_1 and S_2 are the functions given in Eq.s (4.136) and (4.137), respectively).

Let us point out a couple of facts appearing from the expressions (4.147-4.150) obtained above for the renormalized stress-energy VEV components. On the one hand, as was to be expected due to clear symmetry considerations, the renormalized VEV $(\boldsymbol{v} \mid \hat{T}_{\mu\nu} \boldsymbol{v})_{ren}$ only depends on the spatial coordinates $x^1 \in (0, a)$, corresponding to the direction orthogonal to the planes; so, with a slight abuse of notation, we put (for $\mu, \nu \in \{0, ..., 3\}$)

$$(\boldsymbol{v} | \hat{T}_{\mu\nu}(\mathbf{x})\boldsymbol{v})_{ren} \equiv (\boldsymbol{v} | \hat{T}_{\mu\nu}(x^1)\boldsymbol{v})_{ren} .$$
(4.151)

In fact, the same statement can be easily proven to hold for any spatial dimension d and for the regularized stress-energy VEV, as well. On the other hand, keeping only the contributions not depending on the parameter β in the above expressions (4.147-4.150), one recovers the renormalized stress-energy VEV corresponding to the case of Dirichlet boundary conditions for the same geometrical configuration with two parallel planes (²¹).



Figure 4.8: Graphs of $(\boldsymbol{v} \mid \hat{T}_{11}^{(\Diamond)} \boldsymbol{v})_{ren}$ and $(\boldsymbol{v} \mid \hat{T}_{11}^{(\blacksquare)} \boldsymbol{v})_{ren}$.

In conclusion, let us present, some results which can be obtained by numerical evaluation for the functions

 $(0,a) \ni x^1 \mapsto (\boldsymbol{v} \mid \hat{T}_{\mu\mu}^{(\bullet)}(x^1)\boldsymbol{v})_{ren} \quad \text{for } \mu \in \{0,1\} \text{ and } \boldsymbol{\bullet} \in \{\diamondsuit, \blacksquare\} .$ (4.152)

To this purpose, we fix

$$a = 1 \tag{4.153}$$

and consider several values for the parameter $\beta \in (0, +\infty)$. As an example, we report in Fig.s 4.7 and 4.8 the graphs obtained for $\beta = 0.04$, evaluating numerically the expressions

$$(\boldsymbol{v} \,|\, \hat{T}_{\mu\nu}(\mathbf{x}) \,\boldsymbol{v})_{ren} = \frac{\pi^2}{1440a^4} \begin{pmatrix} -1 & 0 & 0 & 0\\ 0 & -3 & 0 & 0\\ 0 & 0 & 1 & 0\\ 0 & 0 & 0 & 1 \end{pmatrix} - \left(\xi - \frac{1}{6}\right) \frac{\pi^2 (3 - 2\sin^2(\frac{\pi}{a} \,x^1))}{8a^4 \sin^4(\frac{\pi}{a} \,x^1)} \begin{pmatrix} -1 & 0 & 0 & 0\\ 0 & 0 & 0 & 0\\ 0 & 0 & 1 & 0\\ 0 & 0 & 0 & 1 \end{pmatrix}.$$

²¹In fact, the renormalized stress-energy VEV for a massless scalar field confined between two parallel planes with Dirichlet boundary conditions in spatial dimension d = 3 is (see, e.g., [21, 59, 64, 75, 110])

4.3. THE CASE OF PARALLEL PLANES WITH ROBIN BOUNDARY CONDITIONS.

(4.147-4.150) (including the integrals appearing therein) with the aid of Matematica (²²). In view of the above figures, all the functions $(\boldsymbol{v} \mid \hat{T}_{00}^{(\Diamond)} \boldsymbol{v})_{ren}, (\boldsymbol{v} \mid \hat{T}_{11}^{(\Diamond)} \boldsymbol{v})_{ren}$ and $(\boldsymbol{v} \mid \hat{T}_{11}^{(\blacksquare)} \boldsymbol{v})_{ren}$ appear to be constant, assuming the approximate numerical values

$$(\boldsymbol{v} | \hat{T}_{00}^{(\Diamond)} \boldsymbol{v})_{ren} \simeq -0.006853..., (\boldsymbol{v} | \hat{T}_{11}^{(\Diamond)} \boldsymbol{v})_{ren} \simeq -0.020561..., (\boldsymbol{v} | \hat{T}_{11}^{(\blacksquare)} \boldsymbol{v})_{ren} \simeq 0.$$

$$(4.154)$$

As a matter of fact, it can be checked by direct inspection that the above results continue to hold for any fixed $\beta \in (0, +\infty)$, so that all the components in Eq. (4.154) appear to be independent of the parameter β . Let us also stress that the above results agree with the exact results corresponding to both the limiting cases of Dirichlet ($\beta \rightarrow 0^+$) and Neumann ($\beta \rightarrow +\infty$) conditions (²³).

In view of the above considerations, it appears that the only term depending non-trivially on the parameter β (and on the coordinate x^1) is $(\boldsymbol{v}|\hat{T}_{00}^{(\bullet)}\boldsymbol{v})_{ren}$; in Fig. 4.9 we show the graph of this term as a function of $x^1 \in (0, 1)$ and $\beta \in (0, 1)$.



Figure 4.9: Graphs of $(\boldsymbol{v} \mid \hat{T}_{00}^{(\blacksquare)} \boldsymbol{v})_{ren}$ as a function of x^1, β .

We plan to give a more detailed analysis of the present configuration, discussing also the total energy and the boundary pressure, in a future work [68].

²³In fact, in both these cases (with the present choice a = 1), there holds (see, e.g., [65])

$$(\boldsymbol{v}|\hat{T}_{00}^{(\Diamond)}\boldsymbol{v})_{ren} = -\frac{\pi^2}{1440} \simeq -0.006853..., \quad (\boldsymbol{v}|\hat{T}_{11}^{(\Diamond)}\boldsymbol{v})_{ren} = -\frac{3\pi^2}{1440} \simeq -0.020561..., \quad (\boldsymbol{v}|\hat{T}_{11}^{(\blacksquare)}\boldsymbol{v})_{ren} = 0.020561..., \quad (\boldsymbol{v}|\hat{T$$

²²The lack of symmetry under the exchange $x^1 \leftrightarrow a - x^1$ in the graph for $(\boldsymbol{v} \mid \hat{T}_{00}^{(\blacksquare)} \boldsymbol{v})_{ren}$ (see Fig. 4.7) is due to the non-symmetric boundary conditions (4.121).

Appendix. Some results for the cylinder kernel on a segment with Robin boundary conditions.

In the present appendix we collect some results which were mentioned and employed in Section 4.3, dealing with a massless scalar field confined between two parallel planes on which Robin boundary conditions of the type (4.119) of Robin type are prescribed (recall that $\beta > 0$). Therein we argued that, since the configuration under analysis is a slab, it suffices to consider the reduced one-dimensional problem where $\Omega_1 = (0, a)$ (a > 0) and V = 0; in particular, we have the admissible operator (see Eq. (4.127))

$$\mathcal{A}_1 := -\partial_{x^1 x^1} : \operatorname{Dom}(\mathcal{A}_1) \subset L^2(0, a) \to L^2(0, a) , \qquad (155)$$

$$Dom(\mathcal{A}_1) := \{ f \in H^1(0, a) \mid \partial_{x^1 x^1} f \in L^2(0, a), \ (1 - \beta \partial_{x^1}) f(0) = (1 - \beta \partial_{x^1}) f(a) = 0 \}$$

(notice that the domain $\text{Dom}(\mathcal{A}_1)$ keeps into account the boundary conditions induced by those in Eq. (4.121)). We already mentioned (see Eq.s (4.123) (4.130)) that a complete orthonormal set of eigenfunctions of \mathcal{A}_1 with corresponding eigenvalues is given by (²⁴)

$$F_n(x^1) := \sqrt{\frac{2}{a(1+\beta^2\omega_n^2)}} \left(\sin(\omega_n x^1) + \beta \,\omega_n \,\cos(\omega_n x^1)\right) \qquad (x^1 \in (0,a)) ,$$

$$\omega_n^2 := \frac{n^2 \pi^2}{a^2} , \qquad \text{for } n \in \{1, 2, 3, ...\} .$$
 (156)

 $^{24}\mathrm{To}$ prove this fact, it suffices to consider the differential problem

$$-\partial_{x^1x^1}F_n = \omega_n^2 F_n , \quad F_n(0) - \beta F'_n(0) = 0 , \quad F_n(a) - \beta F'_n(a) = 0 , \quad \langle F_n | F_m \rangle_{L^2} = \delta_{nm} .$$

Considering the general solution $F_n(x^1) = A_n \sin(\omega_n x^1) + B_n \cos(\omega_n x^1)$ and imposing the boundary conditions yields automatically the relations

$$B_n = \beta \,\omega_n \,A_n \,, \qquad \omega_n = \frac{\pi n}{a} \,(n \in \{1, 2, 3, ...\}) \,.$$

Finally, the first identity can be used along with the normalization condition $\langle F_n | F_m \rangle_{L^2} = \delta_{nm}$ to infer

$$A_n = \sqrt{\frac{2}{a(1+\beta^2\omega_n^2)}}$$

In the following we will use these facts to compute explicitly the cylinder kernel $e^{-t\sqrt{A_1}}(,)$ and to deduce the features possessed by this function, already mentioned in Section 4.3.

Computation of $e^{-t\sqrt{A_1}}(\ ,\)$.

Lemma .1. For any fixed $t \in (0, +\infty)$, $x^1, y^1 \in (0, a)$ and any fixed $\beta \in (0, +\infty)$, the cylinder kernel can be expressed as

$$e^{-\mathbf{t}\sqrt{\mathcal{A}_{1}}}(x^{1}, y^{1}) = T_{0}(\mathbf{t}; x^{1}, y^{1}) + T_{1}(\mathbf{t}; x^{1}, y^{1}) + T_{2}(\mathbf{t}; x^{1}, y^{1}) \quad \text{where}$$

$$T_{0}(\mathbf{t}; x^{1}, y^{1}) := \frac{1}{a} \sum_{n=1}^{+\infty} e^{-\mathbf{t}\omega_{n}} \left(\cos(\omega_{n}(x^{1} - y^{1})) - \cos(\omega_{n}(x^{1} + y^{1})) \right) ,$$

$$T_{1}(\mathbf{t}; x^{1}, y^{1}) := \frac{2}{a} \sum_{n=1}^{+\infty} e^{-\mathbf{t}\omega_{n}} \frac{\beta\omega_{n}}{1 + \beta^{2}\omega_{n}^{2}} \sin(\omega_{n}(x^{1} + y^{1})) ,$$

$$T_{2}(\mathbf{t}; x^{1}, y^{1}) := \frac{2}{a} \sum_{n=1}^{+\infty} e^{-\mathbf{t}\omega_{n}} \frac{\beta^{2}\omega_{n}^{2}}{1 + \beta^{2}\omega_{n}^{2}} \cos(\omega_{n}(x^{1} + y^{1})) ,$$
(157)

where ω_n is given by (156). Moreover, all the series above are absolutely convergent.

Proof. It suffices to consider the eigenfunction expansion

$$e^{-\mathbf{t}\sqrt{\mathcal{A}_1}}(x^1, y^1) = \sum_{n=1}^{+\infty} e^{-\mathbf{t}\omega_n} F_n(x^1)\overline{F}_n(y^1)$$

along with the explicit expressions (156) for F_n and ω_n (some elementary trigonometric identities must be used, as well). Absolute convergence can be easily proved recalling again that $\omega_n = \frac{\pi n}{a}$.

Lemma .2. There holds the relations of Eq. (4.135), i.e.,

$$\begin{split} T_{0}(\mathbf{t}\,;x^{1},y^{1}) &= \frac{1}{2a} \left[\frac{\cos(\frac{\pi}{a}(x^{1}-y^{1})) - e^{-\frac{\pi}{a}\mathbf{t}}}{\cosh(\frac{\pi}{a}\mathbf{t}) - \cos(\frac{\pi}{a}(x^{1}-y^{1}))} - \frac{\cos(\frac{\pi}{a}(x^{1}+y^{1})) - e^{-\frac{\pi}{a}\mathbf{t}}}{\cosh(\frac{\pi}{a}\mathbf{t}) - \cos(\frac{\pi}{a}(x^{1}+y^{1}))} \right] ,\\ T_{1}(\mathbf{t}\,;x^{1},y^{1}) &= \int_{0}^{+\infty} d\mathbf{\mathfrak{v}} \,\cos\mathbf{\mathfrak{v}} \,S_{1}(\mathbf{t}+\beta\mathbf{\mathfrak{v}}\,;x^{1},y^{1}) \,,\\ T_{2}(\mathbf{t}\,;x^{1},y^{1}) &= S_{2}(\mathbf{t}\,;x^{1},y^{1}) - \int_{0}^{+\infty} d\mathbf{\mathfrak{v}} \,\sin\mathbf{\mathfrak{v}} \,S_{2}(\mathbf{t}+\beta\mathbf{\mathfrak{v}}\,;x^{1},y^{1}) \,, \end{split}$$

where the functions S_1, S_2 are as in Eq.s (4.136) (4.137).

Proof. We show how to compute the functions T_0, T_1, T_2 in separate steps.

1 - Computation of T_0 . Let us consider the corresponding series expansion given in Eq. (157); then, it suffices to express the trigonometric functions therein in terms of complex

4.3. THE CASE OF PARALLEL PLANES WITH ROBIN BOUNDARY CONDITIONS.

exponentials, to sum the geometric series thus obtained and to perform some elementary algebraic manipulations.

2 - Computation of T_1 . First notice that, for any $\alpha > 0$, there holds

$$\frac{\alpha}{1+\alpha^2} = \int_0^{+\infty} d\mathfrak{v} \ e^{-\alpha\mathfrak{v}} \cos\mathfrak{v} \ .$$

Using the above relation with $\alpha = \beta \omega_k$, the series expansion for T_1 in Eq. (157) can be re-expressed as follows:

$$T_1(\mathbf{t}; x^1, y^1) = \frac{2}{a} \sum_{n=1}^{+\infty} \int_0^{+\infty} d\mathbf{v} \ e^{-(\mathbf{t}+\beta\mathbf{v})\,\omega_n} \cos\mathbf{v} \ \sin(\omega_n(x^1+y^1)) \ . \tag{158}$$

Since $|e^{-(\mathbf{t}+\beta \mathfrak{v})\omega_n} \cos \mathfrak{v} \sin(\omega_n(x^1+y^1))| \leq e^{-(\mathbf{t}+\beta \mathfrak{v})\omega_n}$, by dominated convergence the sum and integral in the right-hand side of Eq. (158) can be interchanged; then, the thesis follows evaluating explicitly the sum over n = 1, 2, 3, ... (again, this is a geometric series). 3 - Computation of T_2 . First consider the identity

$$\frac{\alpha^2}{1+\alpha^2} = 1 - \int_0^{+\infty} d\mathfrak{v} \ e^{-\alpha \mathfrak{v}} \, \sin \mathfrak{v} \ ,$$

holding for any $\alpha > 0$. Then, the series expansion for T_2 given in Eq. (157) can be reformulated as

$$T_2(\mathbf{t}; x^1, y^1) = \frac{2}{a} \sum_{n=1}^{+\infty} e^{-\mathbf{t}\omega_n} \cos(\omega_n (x^1 + y^1)) - \frac{2}{a} \sum_{n=1}^{+\infty} \int_0^{+\infty} d\mathbf{v} \ e^{-(\mathbf{t} + \beta \mathbf{v})\omega_n} \sin \mathbf{v} \ \cos(\omega_n (x^1 + y^1))$$

Again, the sum and the integral in the second term can be interchanged by dominated convergence theorem; also in this case the thesis follows evaluating explicitly the two sums over n = 1, 2, 3, ... (once more, these are geometric series).

Regularity and asymptotic expansions for $e^{-t\sqrt{A_1}}(,)$.

Let us first state the following general result.

Lemma .3. Let $S : [0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto S(\mathbf{t})$ be an analytic function possessing the following properties:

i. S has an analytic estension to a complex neighbour $\mathcal{U} \subset \mathbb{C}$ of $[0, +\infty)$, with Laurent expansion in $\mathbf{t} = 0$ of the form

$$\mathcal{S}(\mathbf{t}) = \sum_{n=0}^{+\infty} \sigma_n \, \mathbf{t}^n \; . \tag{159}$$

ii. there exist $C, \alpha > 0$ such that, for all $\mathbf{t} \in [0, +\infty)$,

$$|\mathcal{S}(\mathbf{t})| \le C \, e^{-\alpha \mathbf{t}} \,. \tag{160}$$

Then, the function

$$\mathcal{T}: [0, +\infty) \to \mathbb{R} , \qquad \mathbf{t} \mapsto \mathcal{T}(\mathbf{t}) := \int_{\mathbf{t}}^{+\infty} d\mathbf{z} \, \mathcal{S}(\mathbf{z}) ,$$
 (161)

is well defined and there hold the following results: i) \mathcal{T} is analytic in \mathcal{U} and its Laurent expansion in $\mathbf{t} = 0$ is given by

$$\mathcal{T}(\mathbf{t}) = \sum_{n=0}^{+\infty} \tau_n \, \mathbf{t}^n \quad \text{with}$$

$$\tau_0 := \int_0^{+\infty} d\mathbf{z} \, \mathcal{S}(\mathbf{z}) \,, \qquad \tau_n := -\frac{\sigma_{n-1}}{n} \quad \text{for } n \in \{1, 2, 3, ..\} \,; \tag{162}$$

ii) for all $\mathbf{t} \in [0, +\infty)$, there holds

$$|\mathcal{T}(\mathbf{t})| \le \frac{C}{\alpha} e^{-\alpha \mathbf{t}} . \tag{163}$$

Proof. Let us first remark that the hypotheses on S grant, in particular, $S \in C^0([0, +\infty))$ and $S \in L^1(0, +\infty)$ ($\subset L^1(\mathbf{t}, +\infty)$, for all $\mathbf{t} \in [0, +\infty)$); then, the existence of the map (161) follows trivially. Hereafter we discuss in separate steps the proofs of statements i) and ii).

Step 1 - Proof of item i). Since $S \in C^0([0, +\infty)) \cap L^1(0, +\infty)$, the map (161) is differentiable and $\mathcal{T}'(\mathbf{t}) = -S(\mathbf{t})$ for all $\mathbf{t} \in [0, +\infty)$ (see, e.g., Thm.7.11 in [134]); then, hypothesis i. implies the analyticity of \mathcal{T} in \mathcal{U} . Moreover, by elementary results on power series, we have

$$\mathcal{T}(\mathbf{t}) = \mathcal{T}(0) + \sum_{m=0}^{+\infty} \left(-\frac{\sigma_m}{m+1} \right) \mathbf{t}^{m+1} \quad \text{for } \mathbf{t} \in \mathcal{U} ;$$

Eq. (162) follows relabeling the summation index (n := m + 1) and noting that $\tau_0 = \mathcal{T}(0) = \int_0^{+\infty} d\mathbf{z} S(\mathbf{z})$ by definition.

Step 2 - Proof of item ii). The bound (160) and the definition (161) imply, for all $\mathbf{t} \in [0, +\infty)$,

$$|\mathcal{T}(\tau)| \leq C \int_{\mathbf{t}}^{+\infty} d\mathbf{z} \ e^{-\alpha \mathbf{z}} ;$$

evaluation of the elementary integral in the right-hand side above yields Eq. (163). \Box

To proceed, let us consider again the explicit expressions (4.136) (4.137) for the functions S_1, S_2 ; in view of these expressions, it appears that

$$S_1, S_2 \in C^{\infty}([0, +\infty) \times (0, a) \times (0, a))$$
 (164)

Moreover, for any fixed $x^1, y^1 \in (0, a)$, both the maps $[0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto S_1(\mathbf{t}; x^1, y^1), S_2(\mathbf{t}; x^1, y^1)$ are analytic and vanish exponentially; more precisely, for i = 1, 2, there exist constants $C_i(x^1, y^1) > 0$ such that

$$|S_i(\mathbf{t}; x^1, y^1)| \le C_i(x^1, y^1) \ e^{-\frac{\pi}{a}\mathbf{t}} \qquad \text{for all } \mathbf{t} \in [0, +\infty) \ . \tag{165}$$

The same considerations can be drawn for the derivatives $\partial_t^l \partial_{x^1}^m \partial_{y^1}^n S_i$ (i = 1, 2), for any $l, m, n \in \{0, 1, 2, ...\}$. Let us stress that, by Lebesgue's dominated convergence theorem, the differentiation and integration orders in the expressions for the derivatives of T_1, T_2 (obtained starting from the relations in Eq. (4.135)) can be interchanged by Lebesgue dominated convergence theorem.

Then, we have the following result, mention in Section 4.3.

Proposition .4. Consider the reduced cylinder kernel $e^{-\bullet\sqrt{A_1}}(,): (0, +\infty) \times (0, a) \times (0, a), (\mathbf{t}, x^1, y^1) \mapsto e^{-\mathbf{t}\sqrt{A_1}}(x^1, y^1)$. There hold the following results: i) $e^{-\bullet\sqrt{A_1}}(,)$ is jointly smooth in all the variables, i.e., $e^{-\bullet\sqrt{A_1}}(,) \in C^{\infty}((0, +\infty) \times (0, a) \times (0, a));$ ii) for any $l, m, n \in \{0, 1, 2, ...\}$ and any fixed $x^1 \in (0, a)$, the map $(0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto \partial_{\mathbf{t}}^{l} \partial_{y^1}^{m} T(\mathbf{t}; x^1, y^1)|_{y^1=x^1}$ can be written in the form

$$\partial_{\mathbf{t}}^{l} \partial_{x^{1}}^{m} \partial_{y^{1}}^{n} T(\mathbf{t}; x^{1}, y^{1}) \Big|_{y^{1} = x^{1}} = \frac{1}{\mathbf{t}^{q}} J_{l,m,n}(\mathbf{t}; x^{1})$$
(166)

where $q \in \mathbb{N}$ (depending on l, m, n) and the function $J_{l,m,n}(; x^1) : [0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto J_{l,m,n}(\mathbf{t}; x^1)$ is analytic and vanishes exponentially (²⁵). In particular, there hold

$$T(\mathbf{t}; x^1, y^1) \Big|_{y^1 = x^1} = \frac{1}{\mathbf{t}} J_{0,0,0}(\mathbf{t}; x^1) , \qquad (167)$$

$$\partial_{x^{1}}^{m} \partial_{y^{1}}^{n} T(\mathbf{t}; x^{1}, y^{1}) \Big|_{y^{1} = x^{1}} = \frac{1}{\mathbf{t}^{3}} J_{0,m,n}(\mathbf{t}; x^{1}) \quad for \ m, n \in \{0, 1, 2\} \ with \ m+n = 2 \ . \tag{168}$$

Proof. First of all, recall that Eq. (4.135) allows to express the cylinder kernel $e^{-\mathbf{t}\sqrt{A_1}}(x^1, y^1)$ as the sum of three functions, namely $T_0(\mathbf{t}; x^1, y^1)$, $T_1(\mathbf{t}; x^1, y^1)$ and $T_2(\mathbf{t}; x^1, y^1)$. Hereafter we discuss in separate steps their features, which in conclusion yield the thesis. Step 1 - The function T_0 . Consider the explicit expression given in Eq. (4.135); it can be easily checked by direct computations that, for any $l, m, n \in \{0, 1, 2, ...\}$ and any fixed $x^1 \in (0, a)$, there holds

$$\partial_{\mathbf{t}}^{l} \partial_{x^{1}}^{m} \partial_{y^{1}}^{n} T_{0}(\mathbf{t}; x^{1}, y^{1}) \Big|_{y^{1} = x^{1}} = \frac{1}{\mathbf{t}^{q}} J_{l,m,n}^{(0)}(\mathbf{t}; x^{1})$$

199

 $^{^{25}}$ Explicit bounds could be derived but we will not report them here, since they involve cumbersome expressions and are not necessary for later developments

for some $q \in \mathbb{N}$ and some function $J_{l,m,n}^{(0)}(;x^1):[0,+\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto J_{l,m,n}(\mathbf{t};x^1)$ which is analytic and vanishes exponentially. In particular, we have q = 1 if (l,m,n) = (0,0,0)and q = 3 if $(l,m,n) \in \{(0,2,0), (0,1,1), (0,0,2)\}$.

Step 2 - The functions T_1, T_2 . We are going to show that, for any $x^1, y^1 \in (0, a)$ (including the case $y^1 = x^1$), the map $[0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto T_1(\mathbf{t}; x^1, y^1)$ is analytic and exponentially decreasing for $\mathbf{t} \to +\infty$; the very same arguments employed in the following can be used to derive analogous results for the function T_2 and for any derivative of either T_1 or T_2 . Let us consider the representation for T_1 given in Eq. (4.135); making the change of variable $\mathbf{w} := \mathbf{t} + \beta \mathbf{v} \in [\mathbf{t}, +\infty)$ and using elementary trigonometric identities, this representation can be re-expressed as (see Eq. (4.138))

$$T_1(\mathbf{t}; x^1, y^1) = \frac{1}{\beta} \sin\left(\frac{\mathbf{t}}{\beta}\right) \int_{\mathbf{t}}^{+\infty} d\mathbf{w} \sin\left(\frac{\mathbf{w}}{\beta}\right) S_1(\mathbf{w}; x^1, y^1) + \frac{1}{\beta} \cos\left(\frac{\mathbf{t}}{\beta}\right) \int_{\mathbf{t}}^{+\infty} d\mathbf{w} \cos\left(\frac{\mathbf{w}}{\beta}\right) S_1(\mathbf{w}; x^1, y^1) .$$

The properties of S_1 pointed out previously allow to infer that, for any fixed $x^1, y^1 \in (0, a)$, both the functions $[0, +\infty) \to \mathbb{R}$, $\mathbf{t} \mapsto \sin(\frac{\mathbf{t}}{\beta})S_1(\mathbf{t}; x^1, y^1), \cos(\frac{\mathbf{t}}{\beta})S_1(\mathbf{t}; x^1, y^1)$ fulfill the hypotheses of Lemma .3. In consequence of this, the integrals in the above equation are analytic functions for $\mathbf{t} \in [0, +\infty)$ which vanish exponentially; then, it follows trivially that the function $\mathbf{t} \mapsto T_1(\mathbf{t}; x^1, y^1)$ possesses the very same features. \Box

Bibliography

- [1] H. Abels, "Short Lecture Notes: Interpolation Theory and Function Spaces" (2011).
- [2] A.A. Actor, Multiple harmonic oscillator zeta functions, J.Phys.A: Math.Gen. 20, 927–936, (1987).
- [3] A.A. Actor, Local analysis of a quantum field confined within a rectangular cavity, Ann.Phys. **230**, 303–320 (1994).
- [4] A.A. Actor, Scalar quantum fields confined by rectangular boundaries, Fortsch.Phys. 43, 141–205 (1995).
- [5] A.A. Actor, I. Bender, Casimir effect for soft boundaries, Phys.Rev.D 52(6), 3581– 3590 (1995).
- [6] A.A. Actor, I. Bender, Boundaries Immersed in a Scalar Quantum Field, Fortsch.Phys. 44(4), 281–322 (1996).
- [7] R.A. Adams, J.F. Fournier, "Sobolev Spaces", Academic Press (2003).
- [8] S. Agmon, On Kernels, Eigenvalues, and Eigenfunctions of Operators Related to Elliptic Problems, Comm. Pure and Appl. Math. 18, 627–663 (1965).
- [9] L.C. de Albuquerque, R.M. Cavalcanti, *Casimir effect for the scalar field under Robin boundary conditions: A functional integral approach*, J.Phys.A **37**, 7039–7050 (2004).
- [10] C.D. Aliprantis, K.C. Border, "Infinite Dimensional Analysis: A Hitchhiker's Guide", Springer Science & Business Media (2007).
- [11] J. Ambjørn, S. Wolfram, Properties of the vacuum I. Mechanical and thermodynamic, Ann.Phys. 147(1), 1–32 (1983).
- [12] J. Ambjørn, S. Wolfram, Properties of the vacuum II. Electrodynamic, Ann.Phys. 147(1), 33–56 (1983).
- [13] C. Bär, S. Moroianu, Heat kernel asymptotics for roots of generalized Laplacians, Int.J.Math. 14, 397-412 (2003).
- [14] A.F. Beardon, "Limits: A New Approach to Real Analysis", Springer Science & Business Media (1997).

- [15] M. Beauregard, G. Fucci, K. Kirsten, P. Morales, Casimir Effect in the Presence of External Fields, J.Phys.A:Math.Theor. 46, 115401 (2013).
- [16] M. Benini, C. Dappiaggi, T.P. Hack, Quantum Field Theory on Curved Backgrounds – A Primer, Int.Jour.Mod.Phys.A 28, 1330023 (2013).
- [17] J.M. Berezanskii, "Expansion in Eigenfunctions of Self-adjoint Operators", American Math Society (1968).
- [18] Y.M. Berezansky, Z.G. Sheftel, G.F. Us, "Functional Analysis. Volumes I, II", Birkhäuser (2012).
- [19] J. Bergh, J. Löfström, "Interpolation Spaces", Springer (1976).
- [20] N. Berline, E. Getzler, M. Vergne, "Heat Kernels and Dirac Operators", Springer Science & Business Media (1992).
- [21] N.D. Birrell, P.C.W. Davies, "Quantum Fields in Curved Space", Cambridge University Press (1984).
- [22] S.K. Blau, M. Visser, A. Wipf, Zeta functions and the Casimir energy, Nucl.Phys.B 310(1), 163–180 (1988).
- [23] M. Bordag, D. Hennig, D. Robaschik, Vacuum energy in quantum field theory with external potentials concentrated on planes, J.Phys.A: Math.Gen. 25(16), 4483 (1992).
- [24] M. Bordag, U. Mohideen, V. M. Mostepanenko, New developments in the Casimir effect, Phys.Rep.-Rev.Sect.Phys.Lett. 353(1), 1–205 (2001).
- [25] M. Bordag, G.L. Klimchitskaya, U. Mohideen, V.M. Mostepanenko, "Advances in the Casimir Effect", Oxford University Press (2009).
- [26] N. Bourbaki, "General Topology: Chapters 1-4", Springer Science & Business Media (1998).
- [27] O. Bratteli, D. Robinson, "Operator Algebras and Quantum Statistical Mechanics. Volume 1: C^* - and W^* Algebras, Symmetry Groups, Decomposition of States", Springer-Verlag (2002).
- [28] G. Bressi, G. Carugno, R. Onofrio, G. Ruoso, Measurement of the Casimir Force between Parallel Metallic Surfaces, Phys.Rev.Lett. 88(4), 041804 (2002).
- [29] L.S. Brown, G.J. Maclay, Vacuum stress between conducting plates: an image solution, Phys.Rev. 184(5), 1272–1279 (1969).
- [30] Yu.A. Brychkov, A.P. Prudnikov, "Integral Transforms of Generalized Functions", CRC Press (1989).
- [31] S.T. Bunch, P.C.W. Davies, Quantum field theory in de Sitter space: renormalization by point-splitting, Proc.R.Soc.Lond.A Math.Phys.Sc. 360(1700), 117–134 (1978).

- [32] A.A. Bytsenko, G. Cognola, E. Elizalde, V. Moretti, S. Zerbini, "Analytic Aspects of Quantum Fields", World Scientific (2003).
- [33] O. Calin, D.-C. Chang, K. Furutani, C. Iwasaki, "Heat Kernels for Elliptic and Subelliptic Operators: Methods and Techniques", Springer Science & Business Media (2010).
- [34] C.G. Callan, S.R. Coleman, R. Jackiw, A new improved energy-momentum tensor, Ann.Phys. 59, 42–73, (1970).
- [35] H.B.G. Casimir, On the attraction between two perfectly conducting plates, Proc.R.Neth.Acad.Arts Sci. 51, 793–795 (1948).
- [36] I. Cavero-Peláez, K.A. Milton, K. Kirsten, Local and global Casimir energies for a semitransparent cylindrical shell, J.Phys.A-Math.Theor. 40(13), 133607 (2007).
- [37] I. Chavel, "Eigenvalues in Riemannian Geometry", Academic Press (1984).
- [38] J. Chazarain, A. Piriou, "Introduction to the Theory of Linear Partial Differential Equations", Elsevier Science Publishing Company (2011).
- [39] E.A. Coddington, On self-adjoint ordinary differential operators, Math.Scand. 4, 9– 21 (1956).
- [40] G. Cognola, E. Elizalde, S. Zerbini, Fluctuations of quantum fields via zeta function regularization, Phys.Rev.D 65, 085031 (2002).
- [41] G. Cognola, S. Zerbini, Variances of relativistic quantum field fluctuations, Int.J.Mod.Phys.A 18(12), 2067–2072 (2003).
- [42] J.B. Conway, "A Course in Functional Analysis", Springer Science & Business Media (2013).
- [43] J.M. Cook, The mathematics of second quantization, Trans.Am.Math.Soc. 74, 222– 245 (1953).
- [44] C. Dappiaggi, G. Nosari, N. Pinamonti, The Casimir effect from the point of view of algebraic quantum field theory, arXiv:1412.1409 (2014).
- [45] R. Dautray, J.L. Lions, "Mathematical Analysis and Numerical Methods for Science and Technology, Volume 2: Functional and Variational methods", Spinger (2000).
- [46] E.B. Davies, "Heat Kernels and Spectral Theory", Cambridge University Press (1990).
- [47] G. De Marco, "Analisi 2: Secondo corso di analisi matematica per l'universit. Seconda parte", Lampi di stampa (1999).

- [48] F. Demengel and G. Demengel, "Functional Spaces for the Theory of Elliptic Partial Differential Equations", Springer (2012).
- [49] D. Deutsch, P. Candelas, Boundary effects in quantum field theory, Phys.Rev.D 20(12), 3063–3080 (1979).
- [50] J. Diestel, J.J. Uhl Jr., "Vector Measures", American Mathematical Society (1977).
- [51] J.S. Dowker and R. Critchley, Effective Lagrangian and energy-momentum tensor in de Sitter space, Phys.Rev.D 13, 3224–3232 (1976).
- [52] A. Edery, Multidimensional cut-off technique, odd-dimensional Epstein zeta functions and Casimir energy of massless scalar fields, J.Phys.A:Math.Gen. **39**, 685–712 (2006).
- [53] A. Edery, Casimir piston for massless scalar fields in three dimensions, Phys.Rev.D 75, 105012 (2007).
- [54] D.E. Edmunds, W.D. Evans, "Spectral Theory and Differential Operators", Clarendon Press (1987).
- [55] Yu. V. Egorov, M.A. Shubin, *Linear Partial Differential Equations* in "Elements of the Modern Theory of Partial Differential Equations", Springer-Verlag (1999).
- [56] E. Elizalde, S.D. Odintsov, A. Romeo, A.A. Bytsenko, S. Zerbini, "Zeta regularization techniques with applications", World Scientific (1994).
- [57] E. Elizalde, "Ten Physical Applications of Spectral Zeta Functions", Springer Science & Business Media (1995).
- [58] G.G. Emch, "Algebraic Methods in Statistical Mechanics and Quantum Field Theory", Courier Corporation (2014).
- [59] G. Esposito, G.M. Napolitano, L. Rosa, Energy-momentum tensor of a Casimir apparatus in a weak gravitational field: scalar case, Phys.Rev.D 77(10), 105011 (2008).
- [60] R. Estrada, S.A. Fulling, Distributional asymptotic expansions of spectral functions and of the associated Green kernels, Elect.J.Diff.Eqs. **1999**, 1–37 (1999).
- [61] R. Estrada, S.A. Fulling, L. Kaplan, K. Kirsten, Z.H. Liu, K.A. Milton, Vacuum stressenergy density and its gravitational implications, J. Phys. A: Math. Theor. 41 (2008), 164055 (11pp).
- [62] L.C. Evans, "Partial Differential Equations", American Mathamtical Society (2010).
- [63] D. Fermi, L. Pizzocchero, Local zeta regularization and the Casimir effect, Prog.Theor.Phys. 126(3), 419–434 (2011); see also arXiv:1104.4330.
- [64] D. Fermi, L. Pizzocchero, Local zeta regularization and the scalar Casimir effect I. A general approach based on integral kernels, arXiv:1505.00711 (2015).
- [65] D. Fermi, L. Pizzocchero, Local zeta regularization and the scalar Casimir effect II. Some explicitly solvable cases, arXiv:1505.01044 (2015).
- [66] D. Fermi, L. Pizzocchero, Local zeta regularization and the scalar Casimir effect III. The case with a background harmonic potential, Int.J.Mod.Phys.A 30(35), 1550213 (2015); see also arXiv:1505.01651.
- [67] D. Fermi, L. Pizzocchero, Local zeta regularization and the scalar Casimir effect IV. The case of a rectangular box, to appear in Int.J.Mod.Phys.A **31** (2016); see also arXiv:1505.03276 (2015).
- [68] D. Fermi, L. Pizzocchero, in preparation.
- [69] J.C. Ferreira, V.A. Menegatto, Eigenvalues of Integral Operators Defined by Smooth Positive Definite Kernels, Integr.equ.oper.theory 64, 61-81 (2009).
- [70] L.H. Ford, N.F. Svaiter, Vacuum energy density near fluctuating boundaries, Phys.Rev.D 58, 065007 (1998).
- [71] G. Fucci, K. Kirsten, Conical Casimir pistons with hybrid boundary conditions, J.Phys.A-Math.Theor. 44(29), 295403 (2011).
- [72] G. Fucci, K. Kirsten, The Spectral Zeta Function for Laplace Operators on Warped Product Manifolds of the type $I \times f_N$, Comm.Math.Phys **317**(3), 635–665 (2013).
- [73] S.A. Fulling, R.A. Gustafson, Some properties of Riesz means and spectral expansions, Elect.J.Diff.Eqs., 1999(6), 1–39 (1999).
- [74] S.A. Fulling, Systematics of the relationship between vacuum energy calculations and heat kernel coefficients, J.Phys.A: Math.Gen. 36, 6857–6873 (2003).
- [75] S.A. Fulling, Mass dependence of vacuum energy, Phys.Lett.B 624, 281–286 (2005).
- [76] S.A. Fulling, Vacuum Energy as Spectral Geometry, SIGMA 3, 094 (2007).
- [77] S.A. Fulling, L. Kaplan, K. Kirsten, Z.H. Liu, K.A. Milton, Vacuum Stress and Closed Paths in Rectangles, Pistons, and Pistols, J.Phys.A 42, 155402 (2009).
- [78] I.M. Gelfand, G.E. Shilov, "Generalized Functions. Volume 3: Theory of Differential Equations", Academic Press (1967).
- [79] D. Gilbarg, N. S. Trudinger, "Elliptic Partial Differential Equations of Second Order", Springer (2001).

- [80] P.B. Gilkey, "Invariance Theory: The Heat Equation and the Atiyah-Singer Index Theorem", CRC Press (1994).
- [81] N. Graham, K.D. Olum, Negative Energy Densities in Quantum Field Theory With a Background Potential, Phys.Rev.D 67, 085014 (2003).
- [82] A. Grigor'yan, "Heat Kernel and Analysis on Manifolds", AMS/IP Studies in Advanced Mathematics 47 (2009).
- [83] P. Grisvard, "Elliptic Problems in Nonsmooth Domains, SIAM (2011).
- [84] T.P. Hack, V. Moretti, On the stress-energy tensor of quantum fields in curved spacetimes – comparison of different regularization schemes and symmetry of the Hadamard/SeeleyDeWitt coefficients, J.Phys.A:Math.Theor. 45, 374019 (2012).
- [85] J.M. Harrison, K. Kirsten, C. Texier, Spectral determinants and zeta functions of Schrödinger operators on metric graphs, J.Phys.A-Math.Theor. 45(12), 125206 (2011).
- [86] S.W. Hawking, Zeta function regularization of path integrals in curved spacetime, Commun. Math. Phys. 55(2), 133–148 (1977).
- [87] N.R. Howes, "Modern Analysis and Topology", Springer Science & Business Media (2012).
- [88] D. Iellici, V. Moretti, ζ -function regularization and one-loop renormalization of field fluctuations in curved space-time, Phys.Lett.B **425**, 33–40 (1998).
- [89] T. Kato, On some Schrödinger operators with a singular complex potential, Ann. Scuola Norm. Sup. Pisa, Classe di Scienze 4^a Ser. 5 (1), 105–114 (1978).
- [90] T. Kato, "Perturbation Theory for Linear Operators", Springer (1995).
- [91] K. Kirsten, "Spectral Functions in Mathematics and Physics", CRC Press (2001).
- [92] K. Kirsten, Basic zeta functions and some applications in physics in "A Window into Zeta and Modular Physics", Cambridge University Press (2010).
- [93] S. K. Lamoreaux, Demonstration of the Casimir Force in the 0.6 to 6 μm Range, Phys.Rev.Lett. 78, 5-8 (1997).
- [94] J.M. Lee, "Introduction to Smooth Manifolds", Springer Science & Business Media (2012).
- [95] X. Li, H. Cheng, J. Li, X. Zhai, Attractive or repulsive nature of the Casimir force for rectangular cavity, Phys.Rev.D 56(4), 2155–2162 (1997).

- [96] J.L. Lions, E. Magenes, Problemi ai limiti non omogenei (III), Ann. Scuola Norm.Sup. Pisa 15 (1-2), 41–103 (1961).
- [97] J.L. Lions, E. Magenes, "Non-Homogeneous Boundary Value Problems and Applications, Vol.I", Springer (1972).
- [98] T. Lu, T. Jeffres, K. Kirsten, Zeta function of self-adjoint operators on surfaces of revolution, J.Phys.A-Math.Theor. 48(14), 145204 (2015).
- [99] W. Lukosz, Electromagnetic Zero-Point Energy and Radiation Pressure for a Rectangular Cavity, Physica 56, 109–120 (1971).
- [100] W. Lukosz, Electromagnetic Zero-Point Energy Shift Induced by Conducting Surfaces II. The Infinite Wedge and the Rectangular Cavity, Z.Physik 262, 327–348 (1973).
- [101] A. Lunardi. Interpolation Theory. Appunti. Scuola Normale Superiore di Pisa (1999).
- [102] S.G. Mamaev, N.N. Trunov, Vacuum means of energy-momentum tensor of quantized fields on manifolds of different topology and geometry. I, Sov.Phys.J. 22(7), 766–770 (1979).
- [103] S.G. Mamaev, N.N. Trunov, Vacuum averages of the energy-momentum tensor of quantized fields on manifolds of various topology and geometry. II, Sov.Phys.J. 22(9), 966–969 (1979).
- [104] S.G. Mamaev, N.N. Trunov, Vacuum expectation values of the energy-momentum tensor of quantized fields on manifolds with different topologies and gometry. III, Sov.Phys.J. 23(7), 551-554 (1980).
- [105] S.G. Mamaev, N.N. Trunov, Vacuum expectation values of the energy-momentum tensor of quantized fields on manifolds of different topology and geometry. IV, Sov.Phys.J. 24(2), 171-174 (1981).
- [106] K. Maurin, "Methods of Hilbert Spaces", Polish Scientific Publisher (1972).
- [107] V. Maz'ya, S. Nazarov, B. Plamenevskij, "Asymptotic Theory of Elliptic Boundary Value Problems in Singularly Perturbed Domains, Volume 1", Springer Science & Business Media (2000).
- [108] F.G. Mehler, Ueber die Entwicklung einer Function von beliebig vielen Variabeln nach Laplaceschen Functionen höherer Ordnung, Journal fr Reine und Angewandte Mathematik 66(3), 161176 (1866).
- [109] V.P. Mikhailov, "Partial differential equations", Mir Publishers (1978).
- [110] K.A. Milton, "The Casimir effect Physical manifestations of zero-point energy", World Scientific Publishing Co.Pte.Ltd. (2001).

- [111] S. Minakshisundaram, A generalization of Epstein zeta functions, Canad.J.Math. 1, 320–327 (1949).
- [112] S. Minakshisundaram, A. Pleijel, Some properties of the eigenfunctions of the Laplace-Operator on Riemannian manifolds, Canad.J.Math. 1, 242–256 (1949).
- [113] U. Mohideen, A. Roy, Precision Measurement of the Casimir Force from 0.1 to 0.9 μm, Phys.Rev.Lett. 81, 004549 (1997).
- [114] V. Moretti, Direct ζ -function approach and renormalization of one-loop stress tensors in curved spacetimes, Phys.Rev.D **56**(12), 7797–7819 (1997).
- [115] V. Moretti, Local ζ -function techniques vs point-splitting procedure: a few rigorous results, Commun.Math.Phys. **201**, 327–363 (1999).
- [116] V. Moretti, One-loop stress-tensor renormalization in curved background: the relation between ζ -function and point-splitting approaches, and an improved pointsplitting procedure, J.Math.Phys. **40**, 3843–3875 (1999).
- [117] V. Moretti, A review on recent results of the ζ-function regularization procedure in curved spacetime, in D. Fortunato, M. Francaviglia, A. Masiello, "Recent developments in General Relativity", Springer-Verlag (1999); also see arXiv:gr-qc/9902056.
- [118] V. Moretti, Local ζ -functions, stress-energy tensor, field fluctuations, and all that, in curved static spacetime, Springer Proc.Phys. **137**, 323–332 (2011).
- [119] V. Moretti, "Spectral Theory and Quantum Mechanics. With an Introduction to the Algebraic Formulation", Springer (2013).
- [120] J.M. Munoz-Castaneda, J.M. Guilarte, A.M. Mosquera, Quantum vacuum energies and Casimir forces between partially transparent δ -function plates, Phys.Rev.D 87, 105020 (2013).
- [121] H.C. Ohanian, Gravitation and the new improved energy-momentum tensor, J.Math.Phys. 14(12), 1892–1897 (1973).
- [122] F.W.J. Olver, D.W. Lozier, R.F. Boisvert, and C.W. Clark, "NIST Handbook of Mathematical Functions", Cambridge University Press (2010).
- [123] L.E. Parker, D.J. Toms, "Quantum field theory in curved spacetime: Quantized Fields and Gravity", Cambridge University Press (2009).
- [124] S.J. Patterson, "An Introduction to the Theory of the Riemann Zeta-Function", Cambridge University Press (1995).
- [125] B.J. Pettis, On Integration in Vector Spaces, Trans.Am.Mat.Soc 44 (2), 277–304 (1938).

- [126] G. Plunien, B. Müller, W. Greiner, The Casimir Effect, Phys.Rep. 134(2,3), 87–193 (1986).
- [127] M.J. Radzikowski, Micro-local approach to the Hadamard condition in quantum field theory on curved space-time, Comm.Math.Phys. 179(3), 529–553 (1996).
- [128] D.B. Ray, I.M. Singer, *R*-torsion and the Laplacian on Riemannian Manifolds, Adv.Math. 7, 145–210 (1971).
- [129] M. Reed, B. Simon, "Methods of Modern Mathematical Physics I: Functional Analysis", Academic Press (1981).
- [130] M. Reed, B. Simon, "Methods of Modern Mathematical Physics II: Fourier Analysis, Self-Adjointness", Elsevier (1975).
- [131] R.B. Rodrigues, N.F. Svaiter, Vacuum Polarization of a Scalar Field in a Rectangular Waveguide, arXiv:hep-th/0111131 (2001).
- [132] R.B. Rodrigues, N.F. Svaiter, R.D.M. De Paola, Vacuum Stress Tensor of a Scalar Field in a Rectangular Waveguide, arXiv:hep-th/0110290 (2001).
- [133] A. Romeo, A.A. Saharian, Casimir effect for scalar fields under Robin boundary conditions on plates, J.Phys.A:Math.Gen. 35(5), 1297 (2002).
- [134] W. Rudin, "Real and Complex Analysis", McGraw Hill (1970).
- [135] J.R. Ruggiero, A. Villani, A.H. Zimerman, Application of Analytic Regularization to the Casimir Forces, Rev.Bras.Fis. 7(3), 663–687 (1977).
- [136] J.R. Ruggiero, A. Villani, A.H. Zimerman, Some comments on the application of analytic regularisation to the Casimir forces, J.Phys.A:Math.Gen. 13, 761–766 (1980).
- [137] A.A. Saharian, R.M. Avagyan, R. S. Davtyan, Wightman function and Casimir densities for Robin plates in the Fulling-Rindler vacuum, Int.J.Mod.Phys.A 21, 2353– 2375 (2006).
- [138] K. Schmüdgen, "Unbounded Self-adjoint Operators on Hilbert Space", Springer Science & Business Media (2012).
- [139] F. Schwabl, "Quantum Mechanics", Springer (2002).
- [140] L. Schwartz, Thorie des Noyaux, Proceedings of the International Congress of Mathematicians (Cambridge), vol.I, 220–230 (1950).
- [141] L. Schwartz, Su Alcuni Problemi della Teoria delle Equazioni Differenziali Lineari di Tipo Ellittico, Rend.Sem.Mat.Fis.Mil. 27(1), 211–249 (1957).

- [142] S.S. Schweber, An Introduction to Relativistic Quantum Field Theory, Courier Corporation (2011).
- [143] R.T. Seeley, Complex Powers of an Elliptic Operator, Am.Math.Soc.Proc.Sym.Pure Math. 10(3), 288–307 (1967).
- [144] M.R. Setare, Casimir energy-momentum tensor for the Robin Surfaces in de Sitter Spacetime, Phys.Lett.B 637, 1–4 (2006).
- [145] Z. Shen, The spectrum of Schrödinger operators with positive potentials in Riemannian manifolds, Proceedings of the Americal Mathematical Society 131(11), 3447– 3456 (2003).
- [146] B. Simon, " $P(\Phi)^2$ Euclidean (Quantum) Field Theory", Princeton University Press (2015).
- [147] M.J. Sparnaay, Attractive forces between flat plates, Nature 180, 334–335 (1957).
- [148] M.J. Sparnaay, Measurement of attractive forces between flat plates, Physica 24, 751–764 (1958).
- [149] I. Steinwart, C. Scovel, Mercers Theorem on General Domains: On the Interaction between Measures, Kernels, and RKHSs, Constr.Approx. 35, 363-417 (2012).
- [150] M.H. Stone, "Linear transformations in Hilbert space and their applications to analysis", Am.Math.Soc (1932).
- [151] F. Trèves, "Topological vector spaces, distributions and kernels", Academic Press (1967).
- [152] R.M. Wald, On the Euclidean Approach to Quantum Field Theory in Curved Spacetime, Comm.Math.Phys. 70, 221–242 (1979).
- [153] R.M. Wald, "General relativity", University of Chicago Press (2010).
- [154] G. N. Watson, "A Treatise on the Theory of Bessel Functions", Cambridge University Press (1995).
- [155] J. Weidmann, "Linear Operators in Hilbert Spaces", Springer Science & Business Media (2012).
- [156] J. Wloka, "Partial Differential Equations", Cambridge University Press (1987).
- [157] A.H. Zemanian, "Generalized Integral Transformations", John Wiley & Sons (1968).